UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): June 16, 2022

GROVE COLLABORATIVE HOLDINGS, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation)	001-40263 (Commission File Number)	88-2840659 (IRS Employer Identification No.)
1301 Sansome Street		
San Francisco, California		94111
(Address of principal executive offices)		(Zip Code)
	(800) 231-8527	
(Res	gistrant's telephone number, including area code)	
	Not Applicable	
(Former	name or former address, if changed since last report)	

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the

Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425) Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12) Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b)) Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c)) Securities registered pursuant to Section 12(b) of the Act: Name of each exchange Trading Title of each class Symbol(s) on which registered Class A common stock, par value \$0.0001 GROV New York Stock Exchange Redeemable warrants, each whole warrant **GROV.WS** New York Stock Exchange exercisable for one share of Class A common stock at an exercise price of \$11.50 per share

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company ⊠

following provisions:

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

INTRODUCTORY NOTE

Domestication and Merger Transaction

As previously announced, Virgin Group Acquisition Corp. II ("VGAC II" and, after the Domestication as described below, "New Grove"), a Cayman Islands exempted company, entered into that certain Agreement and Plan of Merger (the "Merger Agreement"), dated as of December 7, 2021, as amended and restated on March 31, 2022, by and among VGAC II, Trechouse Merger Sub, Inc., a Delaware corporation and wholly owned direct subsidiary of VGAC II ("Merger Sub I"), Trechouse Merger Sub II, LLC, a Delaware limited liability company and wholly owned direct subsidiary of VGAC II ("Merger Sub II") and Grove Collaborative, Inc., a Delaware public benefit corporation ("Grove"). On June 15, 2022, as contemplated by the Merger Agreement and described in the section titled "Domestication Proposal" beginning on page 135 of the final prospectus and definitive proxy statement, dated May 16, 2022 (the "Proxy Statement/Prospectus") and filed with the Securities and Exchange Commission (the "SEC"), VGAC II filed a notice of deregistration with the Cayman Islands Registrar of Companies, together with the necessary accompanying documents, and filed a certificate of incorporation and a certificate of corporate domestication with the Secretary of State of the State of Delaware, under which VGAC II was domesticated and continued as a Delaware public benefit corporation, changing its name to "Grove Collaborative Holdings, Inc." (the "Domestication").

As a result of and upon the effective time of the Domestication, among other things, (i) each of the then issued and outstanding Class A ordinary shares, par value \$0.0001 per share, of VGAC II (the "VGAC II Class A ordinary shares"), and Class B ordinary shares, par value \$0.0001 per share, of VGAC II (the "VGAC II Class B ordinary shares"), automatically converted, on a one-for-one basis, into shares of Class A common stock, par value \$0.0001 per share, of New Grove (the "New Grove Class A Common Stock"); (ii) each then issued and outstanding whole warrant of VGAC II (the "VGAC II Warrants") automatically converted into a warrant to acquire one share of New Grove Class A Common Stock at an exercise price of \$11.50 per share (the "New Grove Warrants"); and (iii) each of the then issued and outstanding units of VGAC II that had not been previously separated into the underlying VGAC II Class A ordinary shares and underlying VGAC II Warrants upon the request of the holder thereof, were canceled and entitled the holder thereof to one share of New Grove Class A Common Stock and one-fifth of one New Grove Warrant.

On June 16, 2022 (the "Closing Date"), as contemplated by the Merger Agreement and described in the section titled "Business Combination Proposal" beginning on page 88 of the Proxy Statement/Prospectus, New Grove consummated the merger transaction contemplated by the Merger Agreement, whereby (i) Merger Sub I merged with and into Grove (the "Initial Merger," and the time at which the Initial Merger became effective, the "Initial Effective Time"), the separate corporate existence of Merger Sub I ceased, and Grove continued as the surviving corporation and a wholly owned subsidiary of New Grove (the "Initial Surviving Corporation"), and (ii) immediately following the Initial Merger, and as part of the same overall transaction as the Initial Merger, the Initial Surviving Corporation merged with and into Merger Sub II, the separate corporate existence of the Initial Surviving Corporation ceased, and Merger Sub II continued as the surviving company and a wholly-owned subsidiary of New Grove (the "Final Merger" and, together with the Initial Merger and the Domestication, the "Business Combination").

As a result of the closing of the Business Combination (the "Closing"), at the Initial Effective Time, among other things, except as described in the next sentence (i) each share of common stock, par value \$0.0001 per share, of Grove ("Grove Common Stock") (other than the Backstop Tranche 1 Shares (as defined below)) and preferred stock, par value \$0.0001 per share, of Grove ("Grove Preferred Stock") (on an as-converted to common stock basis) (other than dissenting shares) was canceled and converted into the right to receive (a) a number of shares of Class B common stock, par value \$0.0001 per share, of New Grove (the "New Grove Class B Common Stock"), as determined pursuant to an exchange ratio set forth in the Merger Agreement and described below (the "Exchange Ratio") and (b) a number of restricted shares of New Grove Class B Common Stock that will vest upon the achievement of certain earnout thresholds prior to the tenth anniversary of Closing (the "Grove Earnout Shares"), (ii) each outstanding option to purchase Grove common stock (whether vested or unvested) was assumed by New Grove and converted into (a) comparable options that are exercisable for shares of New Grove Class B Common Stock, with a value determined in accordance with the Exchange Ratio (and, with regard to options that are intended to qualify as "incentive stock options" under Section 422 of the Code, in a manner compliant with Section 424(a) of the Code) and (b) the right to receive a number of Grove Earnout Shares, (iii) each award of restricted stock units to

acquire Grove common stock (whether vested or unvested) (collectively, "Grove RSUs") were assumed by New Grove and converted into (a) a comparable award of restricted stock units to acquire shares of New Grove Class B Common Stock and (b) the right to receive a number of Grove Earnout Shares, (iv) each warrant to acquire shares of Grove Common Stock or Grove Preferred Stock was assumed by New Grove and converted into (a) a comparable warrant to acquire shares of New Grove Class B Common Stock (the "New Grove Converted Warrants") and (b) the right to receive a number of Grove Earnout Shares, and (v) each of the Backstop Tranche 1 Shares (as defined below) issued to the Backstop Investor (as defined below) prior to the Initial Effective Time pursuant to the Backstop Subscription Agreement (as defined below) were canceled and converted into the right to receive a number of shares of New Grove Class B Common Stock equal to the Exchange Ratio (which were immediately exchanged on a one-for-one basis for shares of New Grove Class A Common Stock). Notwithstanding the foregoing, pursuant to elections made shortly before the Closing, certain holders of Grove Common Stock and Grove Preferred Stock elected to receive, and at the Closing were issued, an aggregate of 5,711,098 shares of New Grove Class A Common Stock on a share-for-share basis in lieu of the New Grove Class B Common Stock they would have otherwise been entitled to receive (the "Pre-Closing Conversion Elections"). The "Exchange Ratio" is equal to approximately 1.17604.

Holders of 39,672,045 VGAC II Class A ordinary shares elected to have their shares redeemed in connection with the Business Combination.

A description of the Merger Agreement is included in the Proxy Statement/Prospectus in the section entitled "Business Combination Proposal—The Merger Agreement" beginning on page 88, which is incorporated herein by reference.

PIPE Financing

In connection with the Merger Agreement, on December 7, 2021, VGAC II entered into subscription agreements (collectively, the **Subscription Agreements**"), with certain investors (collectively, the **'PIPE Investors**") pursuant to which, and on the terms and subject to the conditions of which, the PIPE Investors collectively subscribed for 8,707,500 shares of New Grove Class A Common Stock at a price of \$10.00 per share for aggregate proceeds of \$87,075,000 (the "**PIPE Financing**"), of which only 8,607,500 shares of New Grove Class A Common Stock have been issued for aggregate proceeds of \$86,075,000. The Company expects to issue the balance of these shares of New Grove Class A Common Stock and collect the remaining proceeds of the PIPE Financing on a subsequent date (the "**Remaining PIPE Financing**"). One of the PIPE Investors is an affiliate of VG Acquisition Sponsor II LLC (the "**Sponsor**") that subscribed for 5,000,000 shares of New Grove Class A Common Stock. In addition, the other PIPE Investors include existing equity holders of Grove that subscribed for 3,707,500 shares of New Grove Class A Common Stock in the aggregate. The PIPE Financing was consummated substantially concurrently with the Closing.

None of the shares of New Grove Class A Common Stock issued in exchange for the Subscription Agreements have been registered under the Securities Act of 1933, as amended (the "Securities Act"), in reliance upon the exemption provided in Section 4(a)(2) of the Securities Act. VGAC II granted the PIPE Investors certain customary registration rights in connection with the foregoing transactions.

A description of the Subscription Agreements is included in the Proxy Statement/Prospectus in the section entitled "Business Combination Proposal —Related Agreements—PIPE Financing" beginning on page 101, which is incorporated herein by reference.

Backstop Financing

On March 31, 2022, VGAC II entered into the a Subscription Agreement (the 'Backstop Subscription Agreement') with Corvina Holdings Limited, an affiliate of the Sponsor (the "Backstop Investor") and Grove, pursuant to which, among other things, (i) the Backstop Investor subscribed for and purchased, on the date of the Backstop Subscription Agreement, as readjusted prior to the Closing pursuant to the terms of the Backstop Subscription Agreement, 2,338,352 shares of Grove Common Stock (the "Backstop Tranche 1 Shares") for aggregate proceeds of \$27,500,000, (ii) the Backstop Investor subscribed for and purchased, on the Closing Date, 1,671,524 shares of New Grove Class A Common Stock at a purchase price of \$10.00 per share (the "Backstop

Tranche 2 Shares") for aggregate proceeds of \$16,715,240, and (iii) New Grove issued to the Backstop Investor, as of immediately following the Closing, warrants to purchase 3,875,028 shares of New Grove Class A Common Stock (each warrant exercisable to purchase one share of New Grove Class A Common Stock for \$0.01) (the "**Backstop Warrants**") (such transactions, collectively, the "**Backstop Financing**"). The Backstop Warrants are exercisable by the Backstop Investor at any time on or before June 16, 2027, and are on terms customary for warrants of such nature.

None of the shares of New Grove Class A Common Stock and New Grove Class B Common Stock issued in exchange for the Backstop Tranche 1 Shares, the Backstop Tranche 2 Shares, or the Backstop Warrants issued pursuant to the Backstop Subscription Agreement, as applicable, have been registered under the Securities Act, in reliance upon the exemption provided in Section 4(a)(2) of the Securities Act. VGAC II granted the Backstop Investor certain customary registration rights in connection with the foregoing transactions.

A description of the Backstop Subscription Agreement is included in the Proxy Statement/Prospectus in the section entitled "Business Combination Proposal—Related Agreements—Backstop Subscription Agreement" beginning on page 307, which is incorporated herein by reference.

In connection with the Closing, the registrant changed its name from Virgin Group Acquisition Corp. II to Grove Collaborative Holdings, Inc.

Immediately after giving effect to the Business Combination (notwithstanding any impact of the Pre-Closing Conversion Elections), the PIPE Financing (excluding those shares subject to the Remaining PIPE Financing) and the Backstop Financing, but excluding those shares issued pursuant to the Employee Stock Grants (as defined below), there were 23,669,479 shares of New Grove Class A Common Stock, 139,101,554 shares of New Grove Class B Common Stock, 14,750,000 New Grove Warrants, 923,857 New Grove Converted Warrants and 3,875,028 Backstop Warrants issued and outstanding. The New Grove Class A Common Stock and the New Grove Warrants trade on The New York Stock Exchange (the "NYSE") under the symbols "GROV" and "GROV.WS", respectively. The New Grove Class A Common Stock is held of record by 510 holders and the New Grove Warrants are held of record by 2 holders.

The foregoing description of each of the Merger Agreement, the Subscription Agreements and the Backstop Agreement, including the description of each in the Proxy Statement/Prospectus referenced above, is a summary only and is qualified in its entirety by the full texts of the Merger Agreement, as amended and restated, which are attached hereto as Exhibits 2.1 and 2.2, the Subscription Agreements, the form of which is attached hereto as Exhibit 10.1, and the Backstop Subscription Agreement, which is attached hereto as Exhibit 10.2, each of which is incorporated herein by reference.

Defined Terms

Terms used but not defined herein, or for which definitions are not otherwise incorporated by reference herein, shall have the meaning given to such terms in the Proxy Statement/Prospectus and such definitions are incorporated herein by reference.

As used hereafter in this Current Report on Form8-K, unless otherwise stated or the context indicates otherwise, the terms the "Company," "Registrant," "New Grove," "we," "us" and "our" refer to Grove Collaborative Holdings, Inc. (formerly known as Virgin Group Acquisition Corp. II), after giving effect to the Business Combination.

Item 1.01 Entry into a Material Definitive Agreement.

Indemnification Agreements

New Grove has entered, and expects to continue to enter into, indemnification agreements with its directors, executive officers and other key employees as determined by its board of directors (the "Board"). The

indemnification agreements will provide that New Grove will indemnify each of its directors, executive officers, and other key employees against any and all expenses incurred by such director, executive officer, or other key employee because of his or her status as one of the New Grove directors, executive officers, or other key employees, to the fullest extent permitted by Delaware Law, the Certificate of Incorporation (as defined below) and the Bylaws (as defined below). In addition, the indemnification agreements will provide that, to the fullest extent permitted by the General Corporation Law of the State of Delaware, New Grove will advance all expenses incurred by its directors, executive officers, and other key employees in connection with the legal proceeding involving his or her status as a director, executive director, or key employee.

The foregoing description of the indemnification agreements is qualified in its entirety by the full text of the form of indemnification agreement, which is attached hereto as Exhibit 10.8 to this Current Report on Form 8-K and is incorporated herein by reference.

Amended and Restated Registration Rights Agreement

On the Closing Date, in connection with the consummation of the Business Combination and as contemplated by the Merger Agreement, New Grove entered into the Amended and Restated Registration Rights Agreement (the "Registration Rights Agreement") with the Sponsor and certain other initial stockholders (collectively, with each other person who has executed and delivered a joinder thereto, the "RRA Parties"), pursuant to which the RRA Parties are entitled to registration rights in respect of certain shares of New Grove Class A Common Stock and certain other equity securities of New Grove that are held by the RRA Parties from time to time. The material terms of the Registration Rights Agreement are described in the section of the Proxy Statement/Prospectus beginning on page 102 titled "Business Combination Proposal—Related Agreements—Amended and Restated Registration Rights Agreement."

The foregoing description of the Registration Rights Agreement is qualified in its entirety by the full text of the Registration Rights Agreement, which is included as Exhibit 10.3 to this Current Report on Form 8-K and is incorporated herein by reference.

Item 2.01 Completion of Acquisition or Disposition of Assets.

The disclosure set forth in the "Introductory Note" above is incorporated into this Item 2.01 by reference. On June 14, 2022, the Business Combination was approved by the shareholders of VGAC II at the extraordinary general meeting of shareholders (the "Shareholder Meeting"). The Business Combination was completed on June 16, 2022.

FORM 10 INFORMATION

Item 2.01(f) of Form 8-K states that if the registrant was formerly a shell company, as VGAC II was immediately before the Business Combination, then following the closing of the transaction in which the registrant ceases to be a shell company, the registrant must disclose the information that would be required if the registrant were filing a general form for registration of securities on Form 10. Accordingly, New Grove is providing the information below that would be included in a Form 10 if it were to file a Form 10. Please note that the information provided below relates to the combined company after the consummation of the Business Combination, unless otherwise specifically indicated or the context otherwise requires.

Cautionary Note Regarding Forward-Looking Statements

The Company makes forward-looking statements in this Current Report on Form 8-K. All statements, other than statements of present or historical fact included in or incorporated by reference in this Current Report on Form 8-K, regarding the Company's future financial performance, as well as the Company's strategy, future operations, financial position, estimated revenues, and losses, projected costs, prospects, plans and objectives of management are forward-looking statements. When used in this Current Report on Form 8-K, the words "could," "should," "will," "may," "believe," "anticipate," "intend," "estimate," "expect," "project," the negative of such terms and other similar expressions are intended to identify forward-looking statements, although not all forward-looking statements contain such identifying words. These forward-looking statements are based on management's current expectations and assumptions about future events and are based on currently available information as to the

outcome and timing of future events. The Company cautions you that these forward-looking statements are subject to all of the risks and uncertainties, most of which are difficult to predict and many of which are beyond the control of the Company, incident to its business.

These forward-looking statements are based on information available as of the date of this Current Report on Form 8-K, and current expectations, forecasts and assumptions, and involve a number of risks and uncertainties. Accordingly, forward-looking statements should not be relied upon as representing the Company's views as of any subsequent date, and the Company does not undertake any obligation to update forward-looking statements to reflect events or circumstances after the date they were made, whether as a result of new information, future events or otherwise, except as may be required under applicable securities laws.

As a result of a number of known and unknown risks and uncertainties, the Company's actual results or performance may be materially different from those expressed or implied by these forward-looking statements. Some factors that could cause actual results to differ include:

- · statements regarding estimates and forecasts of other financial and performance metrics and projections of market opportunity;
- references with respect to the anticipated benefits of the Business Combination and the projected future financial performance of New Grove or New Grove's operating companies following the Business Combination;
- · changes in the market for Grove's products, and expansion plans and opportunities;
- · anticipated customer retention by Grove;
- · the extent to which Grove is able to protect Grove's intellectual property and not infringe on the intellectual property rights of others;
- the sources and uses of cash of the Business Combination;
- · new or adverse regulatory developments relating to automatic renewal laws; and
- other factors detailed set forth in the Proxy Statement/Prospectus in the section titled "Risk Factors" beginning on page 30 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

Business and Properties

The business and properties of VGAC II and Grove prior to the Business Combination are described in the Proxy Statement/Prospectus in the sections titled "Information About VGAC II" beginning on page 205 and "Information About Grove" beginning on page 228, which are incorporated herein by reference.

Risk Factors

The risks associated with the business and operations of New Grove and the Business Combination are described in the Proxy Statement/Prospectus in the section titled "Risk Factors" beginning on page 30, which is incorporated herein by reference.

Financial Information

Selected Historical Financial Information

The selected historical financial information of Grove for the years ended December 31, 2021, December 31, 2020 and December 31, 2019 is included in the Proxy Statement/Prospectus in the section titled "Selected Historical Financial Information of Grove" beginning on page 25, which are incorporated herein by reference.

Audited and Unaudited Condensed Financial Statements

The unaudited condensed financial statements as of March 31, 2022 and for the three months ended March 31, 2022 and March 31, 2021 of Grove have been prepared in accordance with U.S. generally accepted accounting principles and pursuant to the regulations of the SEC are included as Exhibit 99.1 hereto and are incorporated herein by reference.

The unaudited condensed financial statements of Grove should be read in conjunction with the historical audited financial statements of Grove as of December 31, 2021 and 2020 and for the years ended December 31, 2021, 2020 and 2019 and the related notes included in the Proxy Statement/Prospectus beginning on page F-31 of the Proxy Statement/Prospectus, which are incorporated herein by reference.

The unaudited financial statements for VGAC II as of March 31, 2022 and for the three months ended March 31, 2022 and March 31, 2021 are included in Item 1 of VGAC II's Quarterly Report on Form 10-Q, filed with the SEC on May 16, 2022, and are incorporated herein by reference.

The unaudited financial statements of VGAC II should be read in conjunction with the historical audited financial statements of VGAC II for the period from January 13, 2021 (inception) to December 31, 2021 and as of December 31, 2021 and the related notes included in the Proxy Statement/Prospectus beginning on page F-7 of the Proxy Statement/Prospectus, which are incorporated herein by reference.

Unaudited Pro Forma Condensed Combined Financial Information

The unaudited pro forma condensed combined financial information of the Company as of and for the three months ended March 31, 2022 and as of and for the year ended December 31, 2021 are included as Exhibit 99.2 hereto and are incorporated herein by reference.

Management's Discussion and Analysis of Financial Condition and Results of Operations

Management's discussion and analysis of the financial condition and results of operations prior to the Initial Merger is included in the Proxy Statement/Prospectus in the sections titled "VGAC II's Management's Discussion and Analysis of Financial Condition and Results of Operations" beginning on page 222 and "Grove's Management's Discussion and Analysis of Financial Condition and Results of Operations" beginning on page 261, which are incorporated herein by reference.

Management's discussion and analysis of the financial condition and results of operations of VGAC II for the three month period ended March 31, 2022 is included in Item 2 of VGAC II's Quarterly Report on Form 10-Q, filed with the SEC on May 16, 2022, and is incorporated herein by reference.

Management's discussion and analysis of the financial condition and results of operations of Grove for the three month period ended March 31, 2022 is included as Exhibit 99.3 hereto and is incorporated herein by reference.

Security Ownership of Certain Beneficial Owners and Management

The following table sets forth information known to the Company regarding the beneficial ownership of New Grove Class A Common Stock and New Grove Class B Common Stock as of June 16, 2022, after giving effect to the Business Combination (including the impact of the Pre-Closing Conversion Elections), the PIPE Financing (excluding those shares subject to the Remaining PIPE Financing), the Backstop Financing, and the Employee Stock Grants, by:

- each person who is known by New Grove to be the beneficial owner of more than five percent (5%) of the outstanding shares of New Grove Class A Common Stock and New Grove Class B Common Stock, in the aggregate;
- each current executive officer and director of New Grove; and
- all current executive officers and directors of New Grove, as a group.

Beneficial ownership for the purposes of the following table is determined in accordance with the rules and regulations of the SEC. A person is a "beneficial owner" of a security if that person has or shares "voting power," which includes the power to vote or to direct the voting of the security, or "investment power", which includes the power to dispose of or to direct the disposition of the security or has the right to acquire such powers within 60 days. New Grove Class B stock will be convertible into New Grove Class A Common Stock on a one-for-one basis. Ownership of New Grove Class B Common Stock is therefore deemed to be beneficial ownership of New Grove Class A Common Stock under SEC regulations. For purposes of the presentation of ownership of New Grove Class A Common Stock in this table, it has been assumed that each person listed therein as holding New Grove Class B Common Stock has converted into New Grove Class A Common Stock all shares of New Grove Class B Common Stock of which that person is deemed the beneficial owner. Thus, all shares of New Grove Class B Common Stock held by the reporting parties have been included in the calculation of the total amount of New Grove Class A Common Stock owned by each such person as well as in the calculation of the total amount of New Grove Class B Common Stock owned by each such person. As a result of this presentation, there are substantial duplications in the number of shares and percentages shown in the table.

The beneficial ownership percentages set forth in the table below are based on 29,412,877 shares of the Company's Class A Common Stock and 133,390,456 shares of the Company's Class B Common Stock issued and outstanding as of June 16, 2022.

Unless otherwise noted in the footnotes to the following table, and subject to applicable community property laws, New Grove believes that the persons and entities named in the table have sole voting and investment power with respect to their beneficially owned common stock.

	Number of Shares of New Grove Collaborative Class A Common		Number of Shares of New Grove Collaborative Class B Common	
Name and Address of Beneficial Owners	Stock (2)	%	Stock	<u>%</u>
Directors and current named executive officers(1):	0.00=.0=0(4)		0.004.050	6.007
Stu Landesberg(3)	9,807,070(4)	25.0%	9,804,370	6.9%
Fumbi Chima	_	_	_	_
Chris Clark	1,814,554(5)	5.9%	1,486,717	1.1%
Delida Costin(6)	924,092(7)	3.1%	144,866	*
Jennie Perry	521,039(8)	1.7%	521,039	*
David Glazer	33,672(9)	*	30,672	*
John Replogle(10)	316,906(11)	1.1%	90,436	0.1%
Rayhan Arif	_	_	_	_
Kristine Miller	_	_	_	_
Naytri Shroff Sramek	_	_	_	_
All directors and executive officers of the Company as a group (10 persons)	14,328,435	33.2%	12,225,467	8.1%
Entities associated with Virgin(12)	29,969,052	74.9%	_	_
Entities associated with Mayfield(13)	15,867,361(14)	35.2%	15,667,361	11.7%
Norwest Venture Partners XIII, LP(15)	15,990,008(16)	35.6%	15,489,908	11.6%
General Atlantic (GC), L.P.(17)	13,436,978(18)	31.7%	12,936,878	9.7%
Entities associated with Lone Pine Capital(19)	11,397,381(20)	9.9%	10,897,381	8.2%
Entities associated with MHS Capital(21)	10,535,912(22)	26.4%	10,435,612	7.8%
SCM GC Investments Limited(23)	10,893,484(24)	27.7%	9,956,536	7.5%

- The business address of each of Stuart Landesberg, Fumbi Chima, Chris Clark, Delida Costin, Jennie Perry, David Glazer, John Replogle, Rayhan Arif, Kristine Miller and Naytri Shroff Sramek is 1301 Sansome Street, San Francisco, CA 94111.
- 2 The beneficial ownership of New Grove as of the Closing Date is based on (A) 29,412,877 shares of New Grove Class A Common Stock outstanding as of as of such date and (B) 133,390,456 shares of New Grove Class B Common Stock as of such date.
- 3 Includes 679,357 shares of New Grove Class B Common Stock and 100 shares of New Grove Class A Common Stock that are held by the Landesberg Living Trust. Mr. Landesberg may be deemed to have voting and dispositive investment power with respect to the shares held by the Landesberg Living Trust.
- 4 Includes 9,904,370 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis.
- 5 Includes 1,486,717 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis
- 6 Consists of 17,515 shares of New Grove Class B Common Stock and 100 shares of New Grove Class A Common Stock that are held by the Weatherspoon Costin Family Trust. Ms. Costin may be deemed to have voting and dispositive investment power with respect to the shares held by the Weatherspoon Costin Family Trust.
- 7 Includes 144,866 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis.

- 8 Includes 521,039 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis.
- 9 Includes 30,672 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis
- 10 Consists of 24 shares of New Grove Class B Common Stock and 267 shares of New Grove Class A Common Stock that are held by the Replogle Family Trust. Mr. Replogle may be deemed to have voting and dispositive investment power with respect to the shares held by the Replogle Family Trust.
- 11 Includes 90,436 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis.
- 12 Includes 9,972,500 shares of New Grove Class A Common Stock held of record by the Sponsor, 6,700,000 warrants to acquire New Grove Class A Common Stock held of record by the Sponsor, 9,421,524 shares of New Grove Class A Common Stock held of record by the Backstop Investor, and 3,875,028 warrants to acquire New Grove Class A Common Stock held of record by the Backstop Investor. The Backstop Investor is the sole managing member and manager of the Sponsor, and is wholly owned by Virgin Group Holdings Limited ("Virgin Group Holdings"). Virgin Group Holdings is owned by Sir Richard Branson, and he has the ability to appoint and remove the management of Virgin Group Holdings and, as such, may indirectly control the decisions of Virgin Group Holdings, regarding the voting and disposition of securities held by Virgin Group Holdings. Therefore, Sir Richard Branson may be deemed to have indirect beneficial ownership of the shares held by the Sponsor and the Backstop Investor. The address of Virgin Group Holdings Limited is Craigmuir Chambers, Road Town, Tortola, VG1110, British Virgin Islands and the address of the Backstop Investor is Craigmuir Chambers, Road Town, Tortola, VG1110, British Virgin Islands.
- Consists of 13,312,440 shares of New Grove Class B Common Stock held by Mayfield XV, a Cayman Islands Exempted Limited Partnership (MF XV") and 2,354,921 shares of New Grove Class B Common Stock and 200,000 shares of New Grove Class A Common Stock held by Mayfield Select, a Cayman Islands Exempted Limited Partnership ("MF Select"). Mayfield XV Management (UGP), Ltd., a Cayman Islands Exempted Company ("MF XV UGP)" is the general partner of Mayfield XV Management (EGP), L.P., a Cayman Islands Exempted Limited Partnership, which is the general partner of MF XV. Rajeev Batra, Navin Chaddha and Urshit Parikh are the directors of MF XV UGP. As a result, each of the foregoing entities and individuals may be deemed to share beneficial ownership of the shares owned by MF XV, but each of the individuals disclaims such beneficial ownership. Mayfield Select Management (UGP), Ltd., a Cayman Islands Exempted Company ("MF Select UGP) is the general partner of Mayfield Select Management (EGP), L.P., a Cayman Islands Exempted Limited Partnership, which is the general partner of MF Select. Rajeev Batra, Navin Chaddha and Urshit Parikh are the directors of MF Select UGP. As a result, each of the foregoing entities and individuals may be deemed to share beneficial ownership of the shares owned by MF Select, but each of the individuals disclaims such beneficial ownership. The address for each of these entities and individuals is c/o Mayfield, 2484 Sand Hill Road, Menlo Park, CA 94025.
- 14 Includes 15,667,361 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis.
- 15 Includes 15,489,908 shares of New Grove Class B Common Stock and 500,100 shares of New Grove Class A Common Stock held of record by Norwest Venture Partners XIII, LP. Genesis VC Partners XIII, LLC is the general partner of Norwest Venture Partners XIII, LP, and NVP Associates, LLC is the managing member of Genesis VC Partners XIII, LLC. Each of Promod Haque, Jeffrey Crowe, and Jon Kossow, who are co-chief executive officers of NVP Associates, LLC, may be deemed to share voting and dispositive power over the shares held by Norwest Venture Partners XIII, LP. Mr. Crowe disclaims beneficial ownership of all such shares, except to the extent of his pecuniary interest therein, if any. The address for each of these entities and individuals is c/o 525 University Avenue, #800, Palo Alto, California 94301.
- 16 Includes 15,489,908 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis
- 17 Includes 12,936,878 shares of New Grove Class B Common Stock and 500,100 shares of New Grove Class A Common Stock held of record by General Atlantic (GC), L.P. ("GA GC"). The limited partners that share beneficial ownership of the shares held by GA GC are the following General Atlantic investment funds (the "GA Funds"): General Atlantic Partners 100, L.P. ("GAP 100"), General Atlantic Partners (Bermuda) EU, L.P. ("GAP Bermuda EU"), GAP Coinvestments III, LLC ("GAPCO III"), GAP Coinvestments IV, LLC ("GAPCO IV"), GAP Coinvestments V, LLC ("GAPCO V") and GAP Coinvestments CDA, L.P. ("GAPCO CDA"). The general partner of GA GC is General Atlantic (SPV) GP, LLC ("GA SPV"). The general partner of GAP 100 is ultimately controlled by General Atlantic, L.P.

("GA LP"), which is controlled by the Management Committee of GASC MGP, LLC (the "Management Committee"). The general partner of GAP Bermuda EU is ultimately controlled by GAP (Bermuda) L.P. ("GAP Bermuda"), which is also controlled by the Management Committee. GA LP is the managing member of GAPCO III, GAPCO IV and GAPCO V, the general partner of GAPCO CDA and is the sole member of GA SPV. There are nine members of the Management Committee. GA GC, GA LP, GAP Bermuda, GA SPV and the GA Funds (collectively, the "GA Group") are a "group" within the meaning of Rule 13d-5 of the Securities Exchange Act of 1934, as amended. The mailing address of the foregoing General Atlantic entities, other than GAP Bermuda EU and GAP Bermuda, is c/o General Atlantic Service Company, L.P., 55 East 52nd Street, 33rd Floor, New York, NY 10055. The mailing address of GAP Bermuda EU and GAP Bermuda is Clarendon House, 2 Church Street, Hamilton HM 11, Bermuda. Each of the members of the Management Committee disclaims ownership of the shares except to the extent that he has a pecuniary interest therein. Includes 12,936,878 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis.

- Includes 9,330,505 shares of New Grove Class B Common Stock and 100,000 shares of New Grove Class A Common Stock held of record by Lone Cypress, Ltd., 1,144,864 shares of New Grove Class B Common Stock and 100,000 shares of New Grove Class A Common Stock held of record by Lone Cascade, L.P., 391,171 shares of New Grove Class B Common Stock and 100,000 shares of New Grove Class A Common Stock held of record by Lone Spruce, L.P, 12,641 shares of New Grove Class B Common Stock and 100,000 shares of New Grove Class A Common Stock held of record by Lone Monterey Master Fund, Ltd., and 18,200 shares of New Grove Class B Common Stock and 100,000 shares of New Grove Class A Common Stock held of record by Lone Sierra, L.P. However, pursuant to a letter agreement between Lone Pine Capital and New Grove, Lone Pine Capital has relinquished its right to convert shares of New Grove Class B Common Stock held of record by the Long Pine Funds into shares of New Grove Class A Common Stock to the extent that, after giving effect to such conversion, the Lone Pine Funds, together with any other holder whose ownership may aggregated with that of the Lone Pine Funds, collectively would beneficially own in excess of 9.99%. Lone Pine Capital LLC, a Delaware limited liability company ("Lone Pine Capital"), serves as investment manager to Lone Cypress, Ltd., a Cayman Islands exempted company, Lone Cascade, L.P., a Delaware limited partnership, Lone Monterey Master Fund, Ltd., a Cayman Islands exempted company, Lone Spruce, L.P., a Delaware limited partnership, Lone Sierra, L.P., a Delaware limited partnership (collectively, the "Lone Pine Funds"), with respect to the securities of the Company directly held by each of the Lone Pine Funds and has the authority to dispose of and vote the securities of the Company directly held by the Lone Pine Funds. Each of David F. Craver, Brian F. Doherty, Kelly A. Granat and Kerry A. Tyler is an Executive Committee Member of Lone Pine Managing Member LLC, which is the Managing Member of Lone Pine Capital, with respect to the securities directly held by each of the Lone Pine Funds. Stephen F. Mandel, Jr. is the Managing Member of Lone Pine Managing Member LLC, which is the Managing Member of Lone Pine Capital, with respect to the securities directly held by each of the Lone Pine Funds. Each of David F. Craver, Brian F. Doherty, Kelly A. Granat, Kerry A. Tyler, and Stephen F. Mandel, Jr. may be deemed to beneficially own the securities held by the Lone Pine Funds and each of them disclaims beneficial ownership of these securities. The business address of Lone Pine Capital LLC and the Executive Committee Members is Two Greenwich Plaza, Greenwich, Connecticut 06830.
- 20 Includes 10,897,381 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis. However, pursuant to a letter agreement between Lone Pine Capital and New Grove, Lone Pine Capital has relinquished its right to convert shares of New Grove Class B Common Stock held of record by the Long Pine Funds into shares of New Grove Class A Common Stock to the extent that, after giving effect to such conversion, the Lone Pine Funds, together with any other holder whose ownership may aggregated with that of the Lone Pine Funds, collectively would beneficially own in excess of 9.99%.
- 21 Includes 8,093,070 shares of New Grove Class B Common Stock and 100,100 shares of New Grove Class A Common Stock held of record by MHS Capital Partners II, L.P., 1,857,640 shares of New Grove Class B Common Stock and 100 shares of New Grove Class A Common Stock held of record by MHS Capital Partners G, LLC, and 484,902 shares of New Grove Class B Common Stock and 100 shares of New Grove Class A Common Stock held of record by MHS Capital Partners G2, LLC. MHS Capital Management, LLC is the general partner of MHS Capital Partners, LP. MHS Capital Management II, LLC is the Managing Member of MHS Capital Partners G, LLC and MHS Capital Partners G2, LLC. Mark Sugarman, who is the Manager of MHS Capital Management, LLC, and MHS Capital Management II, LLC may be deemed to share voting and dispositive power over the shares held by MHS Capital Partners, LP, MHS Capital Partners G, LLC and MHS Capital Partners G2, LLC. Mr. Sugarman disclaims beneficial ownership of all such shares, except to the extent of his pecuniary interest therein, if any. The address for each of these entities and individuals is c/o 2121 S. El Camino Real, Ste 200, San Mateo, CA 94403.
- 22 Includes 10,435,612 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis.
- 23 Includes 9,956,536 shares of New Grove Class B Common Stock and 100 shares of New Grove Class A Common Stock held of record by SCM GC Investments Limited ("SCMGC"), 65,150 shares of New Grove Class A Common Stock held by Sculptor Enhanced Master Fund, Ltd. ("SCEN") and 434,850 shares of New Grove Class A Common Stock held by Sculptor Master Fund, Ltd. ("SCMF"). Additionally, based on the information provided to the Company on or prior to May 13, 2022, this number also includes 238,534 shares of New Grove Class A Common Stock and 48,633 warrants to acquire New Grove Class A Common Stock held of record by Sculptor Special Funding, LP ("NRMD"), a Cayman Islands company, 54,605 shares of New

Grove Class A Common Stock and 10,921 warrants to acquire New Grove Class A Common Stock held of record by Sculptor Credit Opportunities Master Fund, Ltd. ("SCCO"), a Cayman Islands company, 16,296 shares of New Grove Class A Common Stock and 2,333 warrants to acquire New Grove Class A Common Stock held of record by SCEN, a Cayman Islands company, and 54,605 shares of New Grove Class A Common Stock and 10,921 warrants to acquire New Grove Class A Common Stock held of record by Sculptor SC II, LP ("SCII"), a Delaware company. Sculptor Capital LP ("Sculptor"), a Delaware limited partnership, is the investment adviser to NRMD, SCCO, SCEN, and SCMF and thus may be deemed a beneficial owner of the shares in the accounts managed by Sculptor II. Sculptor II. Sculptor II. Sculptor Capital Holding Corporation, a Delaware corporation ("SCHC"), serves as the sole general partner of Sculptor and Sculptor II. As such, SCHC may be deemed to control Sculptor and Sculptor II and, therefore, may be deemed a beneficial owner of the shares in the accounts managed by Sculptor and Sculptor II. Sculptor Capital Management, Inc. ("SCU"), a Delaware corporation, is the sole shareholder of SCHC, and may be deemed a beneficial owner of the shares in the accounts managed by Sculptor and Sculptor II. SCMGC is beneficially owned by SCMF and SCEN, and may be deemed a beneficial owner of the shares. The business address of NRMD, SCCO, SCEN, SCII, Sculptor, Sculptor II, SCHC, SCMF, SCMGC and SCU is 9 West 57 Street, 39 Floor, New York, NY 10019.

24 Includes 9,956,536 shares of New Grove Class B Common Stock that are convertible into shares of New Grove Class A Common Stock on a one-for-one basis.

Executive Compensation

The information related to executive and director compensation of Grove is included in the Proxy Statement/Prospectus in the section titled "Executive and Director Compensation of Grove" beginning on page 285, which is incorporated herein by reference.

Directors and Executive Officers

Information with respect to New Grove's directors and executive officers immediately following the Closing is set forth in the Proxy Statement/Prospectus in the section titled "Management of New Grove Following the Business Combination" beginning on page 290, which is incorporated herein by reference.

Certain Relationships and Related Party Transactions

The certain relationships and related party transactions of the Company are described in the Proxy Statement/Prospectus in the section titled "Certain Relationships and Related Person Transactions—Grove Collaborative" beginning on page 303, which is incorporated herein by reference.

The information set forth in the section entitled 'Indemnification Agreements' in Item 1.01 of this Current Report is incorporated herein by reference.

Legal Proceedings

Information about legal proceedings is set forth in the Proxy Statement/Prospectus in the section titled "Information About Grove—Legal Proceedings" on page 260, which is incorporated herein by reference.

Market Price of and Dividends on the Registrant's Common Stock and Related Stockholder Matters

Market Information and Holders

The Company's Class A ordinary shares and warrants were historically quoted on the NYSE under the symbols "VGII" and "VGII.WS," respectively. On June 17, 2022, the New Grove Class A Common Stock and the New Grove Warrants began trading on the NYSE under the new trading symbols of "GROV" and "GROV.WS," respectively. VGAC II's publicly traded units automatically separated into their component securities upon the Closing, and as a result no longer trade as a separate security and were delisted from the NYSE.

Dividends

New Grove has not paid any cash dividends on the Class A Common Stock or Class B Common Stock to date. New Grove may retain future earnings, if any, for future operations, expansion and debt repayment and has no current plans to pay cash dividends for the foreseeable future. Any decision to declare and pay dividends in the future will be made at the discretion of the Board and will depend on, among other things, New Grove's results of operations, financial condition, cash requirements, contractual restrictions and other factors that the Board may deem relevant. In addition, New Grove's ability to pay dividends may be limited by covenants of any existing and future outstanding indebtedness New Grove or its subsidiaries incur. New Grove does not anticipate declaring any cash dividends to holders of the Class A Common Stock in the foreseeable future.

Recent Sales of Unregistered Securities

Information about unregistered sales of New Grove's equity securities is set forth under Item 3.02 of this Current Report on Form8-K, which is incorporated herein by reference. Immediately after the Closing, New Grove issued an aggregate of 32,300 shares of New Grove Class A Common Stock, net of taxes, to certain of its employees of the Company's fulfillment centers, none of which are officers or directors of New Grove or any of its predecessors (collectively, the "Employee Stock Grants"). These shares were issued without consideration, and as such there was no sale of unregistered securities.

Description of Registrant's Securities to be Registered

Common Stock

A description of the New Grove Class A Common Stock is included in the Proxy Statement/Prospectus in the section titled "Description of New Grove's Securities—Common Stock" beginning on page 312 which is incorporated herein by reference.

Warrants

A description of the New Grove Warrants is included in the Proxy Statement/Prospectus in the section titled "Description of New Grove's Securities —Warrants" beginning on page 318, which is incorporated herein by reference.

Indemnification of Directors and Officers

The information set forth under the heading 'Indemnification Agreements' in Item 1.01 of this Current Report on Form 8-K is incorporated herein by reference.

Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

Reference is made to the disclosure set forth under Item 4.01 of this Current Report on Form8-K concerning the changes in certifying accountant.

Financial Statements, Supplementary Data, and Exhibits

The information set forth under Item 9.01 of this Current Report on Form 8-K is incorporated herein by reference.

Item 3.02 Unregistered Sales of Equity Securities.

The disclosure set forth in the "Introductory Note" and in Item 2.01 of this Current Report on Form 8-K above is incorporated herein by reference. The securities issued in connection with the Backstop Financing and the PIPE Financing were not registered under the Securities Act of 1933, as amended (the "Securities Act") in reliance on the exemption from registration provided by Section 4(a)(2) of the Securities Act.

Item 3.03 Material Modification to Rights of Security Holders.

On the Closing Date, in connection with the consummation of the Business Combination, New Grove filed its certificate of incorporation (the **Certificate** of Incorporation") with the Secretary of State of the State of Delaware and adopted its bylaws (the **'Bylaws**").

Copies of the Certificate of Incorporation and the Bylaws are included as Exhibits 3.1 and 3.2, respectively, to this Current Report on Form8-K and are incorporated herein by reference.

Item 4.01 Changes in Registrant's Certifying Accountant.

Change of the Company's Independent Registered Public Accounting Firm

On June 16, 2022, after the recommendation of the Audit Committee of the Board, the Board approved the engagement of Ernst & Young LLP ("EY") as the Company's independent registered public accounting firm to audit the Company's financial statements for the fiscal year ending December 31, 2022. EY served as the independent registered public accounting firm of Grove prior to the Business Combination. Accordingly, WithumSmith+Brown, PC ("Withum"), VGAC II's independent registered public accounting firm prior to the Business Combination, was informed that it would be replaced by EY as the Company's independent registered public accounting firm.

Withum's report of independent registered public accounting firm, dated February 24, 2022, on VGAC II's balance sheet as of December 31, 2021 and the related statements of operations, changes in shareholders' deficit and cash flows for the period from January 13, 2021 (inception) through December 31, 2021, did not contain any adverse opinion or disclaimer of opinion, nor were they qualified or modified as to uncertainty, audit scope or accounting principles.

During the period from January 13, 2021 (inception) through December 31, 2021 and the subsequent interim period through June 16, 2022, there were no: (i) disagreements between VGAC II and Withum on any matter of accounting principles or practices, financial statement disclosures or auditing scope or procedures, which disagreements if not resolved to Withum's satisfaction would have caused Withum to make reference to the subject matter of the disagreement in connection with its report or (ii) reportable events as defined in Item 304(a)(1)(v) of Regulation S-K under the Securities Exchange Act of 1934, as amended (the "Exchange Act").

During the period from January 13, 2021 (inception) to December 31, 2021 and the subsequent interim period through June 16, 2022, VGAC II did not consult EY with respect to either (i) the application of accounting principles to a specified transaction, either completed or proposed, or the type of audit opinion that might be rendered on VGAC II's financial statements, and no written report or oral advice was provided to VGAC II by EY that EY concluded was an important factor considered by VGAC II in reaching a decision as to the accounting, auditing or financial reporting issue; or (ii) any matter that was either the subject of a disagreement, as that term is defined in Item 304(a)(1)(iv) of Regulation S-K under the Exchange Act and the related instructions to Item 304 of Regulation S-K under the Exchange Act, or a reportable event, as that term is defined in Item 304(a)(1)(v) of Regulation S-K under the Exchange Act.

The Company has provided Withum with a copy of the disclosures made by the Company in response to this Item 4.01 and has requested that Withum furnish the Company with a letter addressed to the SEC stating whether it agrees with the statements made by the registrant in response to this Item 4.01 and, if not, stating the respects in which it does not agree. A letter from Withum is attached as Exhibit 16.1 to this Current Report on Form 8-K.

Item 5.01 Changes in Control of Registrant.

The information set forth above under "Introductory Note" and in the section entitled "Security Ownership of Certain Beneficial Owners and Management" in Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

The information set forth above in the sections titled 'Directors and Executive Officers' and "Certain Relationships and Related Transactions" in Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference.

2022 Equity and Incentive Plan

At the Special Meeting, the VGAC II stockholders considered and approved the Grove Collaborative Holdings, Inc. 2022 Equity and Incentive Plan (the "Incentive Equity Plan"). The Incentive Equity Plan became effective immediately upon the Closing. The Incentive Equity Plan initially makes available a maximum number of 24,555,528 shares of New Grove Class A Common Stock. The Incentive Equity Plan is described in greater detail in the section of the Proxy Statement/Prospectus entitled "Proposal No. 8 - Incentive Equity Plan Proposal" beginning on page 151 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

The above description of the Incentive Equity Plan, including the description in the Proxy Statement/Prospectus referenced above, does not purport to be complete and is qualified in its entirety by the full text of the Incentive Equity Plan, which is included herein as Exhibit 10.9 and is incorporated herein by reference.

2022 Employee Stock Purchase Plan

At the Special Meeting, the VGAC II stockholders considered and approved the Grove Collaborative Holdings, Inc. Employee Stock Purchase Plan (the "ESPP"). The ESPP became effective immediately upon the Closing. The ESPP initially makes available for sale a maximum number of 3,274,070 shares of New Grove Class A Common Stock. The ESPP is described in greater detail in the section of the Proxy Statement/Prospectus entitled "Proposal No. 9 - The ESPP Proposal" beginning on page 160 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

The above description of the ESPP, including the description in the Proxy Statement/Prospectus referenced above, does not purport to be complete and is qualified in its entirety by the full text of the ESPP, which is included herein as Exhibit 10.10 and is incorporated herein by reference.

Appointment of New Director

On June 16, 2022, the Board elected Fumbi Chima as a director, effective immediately. Ms. Chima will serve as a Class I director whose term expires at the Company's 2023 Annual Meeting of Stockholders. Ms. Chima was also appointed as a member of the Sustainability, Nominating and Corporate Governance Committee of the Board.

The Board has determined that Ms. Chima satisfies the definition of an "independent director" under the listing standards of the New York Stock Exchange.

Ms. Chima is currently the Executive Vice President and Chief Information Officer, a role she has had since 2020, at Boeing Employees' Credit Union (BECU), a not-for-profit financial cooperative. Prior to BECU, Ms. Chima served in leadership roles at various companies in the retail and financial sectors, including as Chief Information Officer at Adidas AG, from 2019 to 2020, Chief Information Officer at Fox Networks Group, from 2017 to 2019, Chief Information Officer at Burberry Group plc, from 2015 to July 2017, and Chief Information Officer — Asia, at Walmart, Inc., from 2014 to 2015.

Ms. Chima also previously served in other leadership roles at Walmart, Inc. from 2010 to 2014, and as Vice President of Corporate Systems at American Express Co. from 2006 to 2010.

Ms. Chima currently serves on the public company boards of AZEK Company, Inc., a manufacturer of residential and commercial building products, Ted Baker plc, a British luxury clothing company, Willis Towers Watson Public Limited Company, a British-American multinational risk management, insurance brokerage and advisory company, and Whitbread plc, a British hospitality company. Previously, Ms. Chima served as an independent member of the board of directors of Africa Prudential plc, technology-driven share registration and investor service provider, from August 2020 through March 2022, and served on the board of directors of Global Sources Ltd., a business-to-business media company with a focus on the Greater China market, from September 2016 to September 2017. Ms. Chima previously served in advisory roles for SAP Executive Advisory, from 2019 to 2020, and Apptio EMEA Advisory, in 2020. Ms. Chima received her Bachelor of Arts in Politics and Philosophy from the University of Hull.

The Board has concluded that Ms. Chima should serve on the Board due to her global business and technology experience and for her leadership in diversity and inclusion efforts, including advocating for women in business globally. The Board believes that Ms. Chima's broad range of experience with public and private company leadership positions and directorships provides significant insight into the Company's global operations and information technology.

Ms. Chima will receive the standard compensation payable to non-employee directors of New Grove.

There are no arrangements or understandings between Ms. Chima and any other person pursuant to which Ms. Chima was selected as a director of New Grove. Neither Ms. Chima nor any of her immediate family members have been a party to any transaction or currently proposed transaction with New Grove that is reportable under Item 404(a) of Regulation S-K under the Exchange Act.

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

The disclosure set forth in Item 3.03 of this Current Report on Form8-K is incorporated herein by reference.

Item 5.05 Amendments to the Registrant's Code of Ethics, or Waiver of a Provision of the Code of Ethics

In connection with the Business Combination, on June 16, 2022, the Board approved and adopted a new Code of Ethics and Business Conduct applicable to all employees, officers and directors of New Grove, including the New Grove's principal executive officer, principal financial officer and principal accounting officer or controller (or persons performing similar functions to the aforementioned officers). A copy of the Code of Ethics and Business Conduct can be found in the Investors section of New Grove's website at investors.grove.co.

Item 5.06 Change in Shell Company Status.

As a result of the Business Combination, New Grove ceased being a shell company. The material terms of the Business Combination are described in the section entitled "Business Combination Proposal" of the Proxy Statement/Prospectus, and are incorporated herein by reference. Further, the information set forth in the "Introductory Note" and under Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference.

Item 9.01 Financial Statements and Exhibits.

(a) Financial Statements of Business Acquired

The unaudited condensed financial statements of Grove as of March 31, 2022 and for the three month periods ended March 31, 2022 and March 31, 2021 are included in Exhibit 99.1 hereto and are incorporated herein by reference.

The historical audited financial statements of Grove as of December 31, 2021 and December 31, 2020 and for the years ended December 31, 2021, 2020 and 2019 and the related notes are included in the Proxy Statement/Prospectus beginning on page F-23 of the Proxy Statement/Prospectus and are incorporated herein by reference.

The unaudited financial statements of VGAC II as of and for the three months ended March, 31 2022 and the related notes are included in VGAC II's Quarterly Report on Form 10-Q filed on May 16, 2022, and are incorporated herein by reference.

The audited financial statements of VGAC II as of and for the period from January 13, 2021 (inception) to December 31, 2021 and the related notes are included in the Proxy Statement/Prospectus on page F-2 of the Proxy Statement/Prospectus and are incorporated herein by reference.

(b) Pro Forma Financial Information

The unaudited pro forma condensed combined financial information of the Company for the year ended December 31, 2021 and as of and for the three months ended March 31, 2022 are included in Exhibit 99.2 hereto and are incorporated herein by reference.

(d) Exhibits.

Exhibit	
No. 2.1+	Description
2.1+	Agreement and Plan of Merger, dated as of December 7, 2021, by and among Virgin Group Acquisition Corp. II, Treehouse Merger Sub, Inc. and
	Grove Collaborative, Inc. (incorporated by reference to Exhibit 2.1 of the Company's Form 8-K (File No. 001-40263), filed with the SEC on
	December 8, 2021).
2.2+	Amended and Restated Agreement and Plan of Merger, dated as of March 31, 2022, by and among Virgin Group Acquisition Corp. II, Treehouse
	Merger Sub, Inc., Treehouse Merger Sub II, LLC and Grove Collaborative, Inc. (incorporated by reference to Exhibit 2.1 of the Company's Form
	8-K (File No. 001-40263), filed with the SEC on April 4, 2022).
3.1*	Certificate of Incorporation of Grove Collaborative Holdings, Inc.

- 3.2* Bylaws of Grove Collaborative Holdings, Inc.
- 4.1 Specimen Warrant Certificate (incorporated by reference to Exhibit 4.3 to the Registration Statement on FormS-1/A (File No. 333-253097), filed with the SEC on March 15, 2021).
- 4.2 Warrant Agreement, dated March 22, 2021, between Continental Stock Transfer & Trust Company and the Company (incorporated by reference to Exhibit 4.1 of the Company's Form 8-K (File No. 001-39587), filed with the SEC on March 25, 2021).
- 4.3* Certificate of Corporate Domestication of Virgin Group Acquisition Corp. II, dated June 15, 2022.
- 4.4* Warrant Agreement, dated June 16, 2022, between Grove Collaborative Holdings, Inc. and Corvina Holdings Limited.
- 4.5* Specimen Warrant to Purchase Shares of Common Stock of Grove Collaborative, Inc.
- 4.6* Specimen Warrant to Purchase Shares of Common Stock of Grove Collaborative, Inc.
- 4.7* Specimen Warrant to Purchase Shares of Common Stock of Grove Collaborative, Inc.
- 4.8* Specimen Warrant to Purchase Shares of Common Stock of Grove Collaborative, Inc.
- 4.9* Specimen Warrant to Purchase Shares of Series A Preferred Stock of Grove Collaborative, Inc.
- 4.10* Specimen Warrant to Purchase Shares of Series B Preferred Stock of Grove Collaborative, Inc.
 4.11* Specimen Warrant to Purchase Shares of Series C Preferred Stock of Grove Collaborative, Inc.
- 4.12* Specimen Warrant to Purchase Shares of Series D Preferred Stock of Grove Collaborative, Inc.
- 10.1 Form of Subscription Agreement (incorporated by reference to Exhibit 10.2 of the Company's Form8-K (File No. 001-40263), filed with the SEC on December 8, 2021).
- 10.2 Subscription Agreement (incorporated by reference to Exhibit 10.1 of the Company's Form8-K (File No. 001-40263), filed with the SEC on April 4, 2022).
- 10.3* Amended and Restated Registration Rights Agreement, dated as of June 16, 2022, among the Company, Virgin Group Acquisition Sponsor II, LLC, and certain other parties.
- 10.4 Form of Sponsor Agreement (incorporated by reference to Exhibit 10.1 of the Company's Form8-K (File No. 001-40263), filed with the SEC on December 8, 2021).
- 10.5 Sponsor Agreement Amendment (incorporated by reference to Exhibit 10.2 of the Company's Form8-K (File No. 001-40263), filed with the SEC on April 4, 2022).
- 10.6+ Form of Support Agreement (incorporated by reference to Exhibit 10.3 of the Company's Form8-K (File No. 001-40263), filed with the SEC on December 8, 2021).
- 10.7+ Support Agreement Amendment (incorporated by reference to Exhibit 10.3 of the Company's Form8-K (File No. 001-40263), filed with the SEC on April 4, 2022).
- 10.8# Form of Indemnity Agreement by and between the Company and its directors and officers (incorporated by reference to Exhibit 10.8 to the Registration Statement on Form S-4/A (File No. 001-40263), filed with the SEC on May 13, 2022).
- 10.9# Grove Collaborative Holdings, Inc. 2022 Incentive Equity Plan and forms of agreement thereunder (incorporated by reference to Annex I to the Registration Statement on Form S-4/A (File No. 333-262200), filed with the SEC on May 13, 2022).
- 10.10# Grove Collaborative Holdings, Inc. Employee Stock Purchase Plan (incorporated by reference to Annex J to the Registration Statement on Form S-4/A (File No. 333-262200), filed with the SEC on May 13, 2022).
- 10.11* Mezzanine Loan and Security Agreement, dated as of April 30, 2021, among Silicon Valley Bank, a California corporation, in its capacity as administrative agent and collateral agent, Silicon Valley Bank, a California corporation, as a lender, Hercules Capital, Inc., a Maryland corporation, as a lender, and Grove Collaborative, Inc., a Delaware public benefit corporation, as Borrower.
- 10.12* First Amendment to Mezzanine Loan and Security Agreement, dated as of May 9, 2022, among Silicon Valley Bank, a California corporation, in its capacity as administrative agent and collateral agent, Silicon Valley Bank, a California corporation, as a lender, Hercules Capital, Inc., a Maryland corporation, as a lender, and Grove Collaborative, Inc., a Delaware public benefit corporation, as Borrower.
- 10.13* Second Amendment to Mezzanine Loan and Security Agreement, dated as of June 16, 2022, among Silicon Valley Bank, a California corporation, in its capacity as administrative agent and collateral agent, Silicon Valley Bank, a California corporation, as a lender, Hercules Capital, Inc., a Maryland corporation, as a lender, and Grove Collaborative, Inc., a Delaware public benefit corporation, formerly known as Treehouse Merger Sub II, LLC, as Borrower.
- 10.14+* Second Amended and Restated Loan and Security Agreement, dated as of July 29, 2020, between Silicon Valley Bank, a California corporation and Grove Collaborative, Inc. a Delaware corporation.
- 10.15* First Amendment to Second Amended and Restated Loan and Security Agreement, dated as of April 30, 2021, between Silicon Valley Bank, a California corporation and Grove Collaborative, Inc. a Delaware public benefit corporation.

10.16*	Second Amendment to Second Amended and Restated Loan and Security Agreement, dated as of May 9, 2022, between Silicon Valley Bank, a California corporation and Grove Collaborative, Inc. a Delaware public benefit corporation.
10.17*	Third Amendment to Second Amended and Restated Loan and Security Agreement, dated as of June 16, 2022, between Silicon Valley Bank, a
	California corporation and Grove Collaborative, Inc. a Delaware public benefit corporation, formerly known as Treehouse Merger Sub II, LLC.
	a Delaware limited liability company.
10.18#*	Grove Collaborative, Inc. 2016 Equity Incentive Plan.
10.19#*	Form of Stock Option Grant Notice and Option Agreement under the Grove Collaborative, Inc. 2016 Equity Incentive Plan.
10.20#*	Restricted Stock Unit Agreement under the Grove Collaborative, Inc. 2016 Equity Incentive Plan.
10.21#*	Form of Restricted Stock Unit Grant Notice under the Grove Collaborative, Inc. 2016 Equity Incentive Plan.
16.1*	Letter from WithumSmith+Brown, PC to the SEC, dated June 23, 2022.
21.1*	List of Subsidiaries.
99.1	Unaudited condensed financial statements of Grove as of and for the three months ended March 31, 2022 and 2021 (incorporated by reference
	to Exhibit 99.2 of the Company's Form 8-K (File No. 001-40263), filed with the SEC on June 2, 2022).
99.2*	Unaudited pro forma condensed combined financial information of Grove Collaborative Holdings, Inc. at March 31, 2022.
99.3	Management's discussion and analysis of the financial condition and results of operation of Grove for the three month periods ended March 31,
	2022 (incorporated by reference to Exhibit 99.1 of the Company's Form 8-K (File No. 001-40263), filed with the SEC on June 2, 2022).
104	Cover Page Interactive Data File (formatted as Inline XBRL).

* Filed herewith.

- Fried nerewith.
 The schedules and exhibits to this agreement have been omitted pursuant to Item 601(a)(5) of Regulation S-K under the Exchange Act. A copy of any omitted schedule and/or exhibit will be furnished to the SEC upon request.
 Indicates management contract or compensatory plan or arrangement.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

GROVE COLLABORATIVE HOLDINGS, INC.

By: /s/ Stuart Landesberg
Name: Stuart Landesberg
Title: Chief Executive Officer

Date: June 23, 2022

CERTIFICATE OF INCORPORATION OF GROVE COLLABORATIVE HOLDINGS, INC. (A PUBLIC BENEFIT CORPORATION)

June 15, 2022

ARTICLE I NAME

The name of the corporation is Grove Collaborative Holdings, Inc. (the "Corporation").

ARTICLE II PURPOSE

The Corporation shall be a public benefit corporation, as contemplated by Subchapter XV of the General Corporate Law of the State of Delaware as the same exists or may hereafter be amended (the "DGCL"), and is to be managed in a manner that balances the Corporation's pecuniary (financial) interests, the best interests of those materially affected by the Corporation's conduct (including customers, employees, partners, and the communities in which we operate), and the public benefits identified in this Certificate of Incorporation (the "Certificate of Incorporation").

The nature of the business or purposes to be conducted or promoted is to engage in any lawful act or activity for which corporations may be organized under the DGCL.

The specific public benefits to be promoted by the Corporation are the development, promotion and distribution of consumer products as a positive force for human and environmental health globally.

ARTICLE III REGISTERED AGENT

The address of the Corporation's registered office in the State of Delaware is 251 Little Falls Drive, in the City of Wilmington, County of New Castle, State of Delaware, 19808, and the name of the Corporation's registered agent at such address is Corporation Service Company.

ARTICLE IV CAPITALIZATION

Section 4.1 <u>Authorized Capital Stock</u>. The total number of shares of all classes of capital stock, each with a par value of \$0.0001 per share, which the Corporation is authorized to issue is 900,000,000 shares, consisting of (a) 800,000,000 shares of common stock (the "Common Stock"), including (i) 600,000,000 shares of Class A Common Stock (the "Class A Common Stock"), and (ii) 200,000,000 shares of Class B Common Stock (the "Class B Common Stock"), and (b) 100,000,000 shares of preferred stock (the "Preferred Stock").

Section 4.2 <u>Preferred Stock</u>. The Board of Directors of the Corporation (the "*Board*") is hereby expressly authorized to provide out of the unissued shares of the Preferred Stock for one or more series of Preferred Stock and to establish from time to time the number of shares to be included in each such series and to fix the voting rights, if any, designations, powers, preferences and relative, participating, optional, special and other rights, if any, of each such series and any qualifications, limitations and restrictions thereof, as shall be stated in the resolution or resolutions adopted by the Board providing for the issuance of such series and included in a certificate of designation (a "*Preferred Stock Designation*") filed pursuant to the DGCL, and the Board is hereby expressly vested with the authority to the full extent provided by law, now or hereafter, to adopt any such resolution or resolutions.

Section 4.3 Common Stock.

- (a) Voting.
- (i) Except as otherwise required by law or this Certificate of Incorporation (including any Preferred Stock Designation), the holders of the shares of Common Stock shall exclusively possess all voting power with respect to the Corporation.
- (ii) Except as otherwise required by law or this Certificate of Incorporation (including any Preferred Stock Designation), the holders of shares of Class A Common Stock shall be entitled to one (1) vote for each such share of Class A Common Stock held of record by such holders and the holders of shares of Class B Common Stock shall be entitled to ten (10) votes for each such share of Class B Common Stock held of record by such holders, in each case, on each matter properly submitted to the stockholders of the Corporation on which the holders of the Common Stock are entitled to vote.
- (iii) Except as otherwise required by law or this Certificate of Incorporation (including any Preferred Stock Designation), at any annual or special meeting of the stockholders of the Corporation, holders of the Class A Common Stock and holders of the Class B Common Stock, voting together as a single class, shall have the exclusive right to vote for the election of directors and on all other matters properly submitted to a vote of the stockholders. Notwithstanding the foregoing, except as otherwise required by law or this Certificate of Incorporation (including any Preferred Stock Designation), holders of shares of any series of Common Stock shall not be entitled to vote on any amendment to this Certificate of Incorporation (including any amendment to any Preferred Stock Designation) that relates solely to the terms of one or more outstanding series of Preferred Stock or other series of Common Stock if the holders of such affected series of Preferred Stock or Common Stock, as applicable, are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Certificate of Incorporation (including any Preferred Stock Designation) or the DGCL.
- (b) *Dividends*. Subject to applicable law and the rights, if any, of the holders of any outstanding series of the Preferred Stock, the holders of shares of Common Stock shall be entitled to receive such dividends and other distributions (payable in cash, property or capital stock of the Corporation) when, as and if declared thereon by the Board from time to time out of any assets or funds of the Corporation legally available therefor and shall share equally on a per share basis in such dividends and distributions.
- (c) Liquidation, Dissolution or Winding Up of the Corporation. Subject to applicable law and the rights, if any, of the holders of any outstanding series of the Preferred Stock, in the event of any voluntary or involuntary liquidation, dissolution or winding up of the Corporation, after payment or provision for payment of the debts and other liabilities of the Corporation, the holders of shares of Common Stock shall be entitled to receive all the remaining assets of the Corporation available for distribution to its stockholders, ratably in proportion to the number of shares of Common Stock held by them.

Section 4.4 <u>Rights and Options</u>. The Corporation has the authority to create and issue rights, warrants and options entitling the holders thereof to acquire from the Corporation any shares of its capital stock of any class or classes, with such rights, warrants and options to be evidenced by or in instrument(s) approved by the Board. The Board is empowered to set the exercise price, duration, times for exercise and other terms and conditions of such rights, warrants or options; *provided, however*, that the consideration to be received for any shares of capital stock issuable upon exercise thereof may not be less than the par value thereof.

Section 4.5 No Class Vote on Changes in Authorized Number of Shares of Stock. The number of authorized shares of any class or classes of stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the voting power of the stock of the Corporation entitled to vote generally in the election of directors, irrespective of the provisions of Section 242(b)(2) of the DGCL (or any successor provision thereto), voting together as a single class, without a separate vote of the holders of the class or classes the number of authorized shares of which are being increased or decreased, unless a vote by any holders of one or more series of Preferred Stock is required by the express terms of any Preferred Stock Designation.

Section 4.6 <u>Issuance of Class B Common Stock</u>. From and after the effective date of this Certificate of Incorporation, shares of Class B Common Stock may be issued exclusively in respect of any Grove share of capital stock (a) outstanding or (b) issuable upon the exercise or conversion of any rights, options, or warrants to acquire Grove capital stock outstanding, in each case, as of immediately prior to the Effective Time.

Section 4.7 Mandatory Conversion of Class B Common Stock. Except for a Permitted Transfer, all shares of Class B Common Stock Transferred by a holder of such shares of Class B Common Stock shall automatically and without further action on the part of the Corporation or any holder of Class B Common Stock be converted into an equal number of fully paid and nonassessable shares of Class A Common Stock upon the Transfer of such shares of Class B Common Stock. All outstanding shares of Class B Common Stock shall automatically and without further action on the part of the Corporation or any holder of Class B Common Stock be converted into an equal number of fully paid and nonassessable shares of Class A Common Stock upon the earliest to occur of (A) the date which is the fifth (5th) anniversary of the Effective Time and (B) the forty-fifth day (or, if such day is not a business day in the United States, the next such business day) after the end of the first fiscal quarter of the Corporation in which the number of shares of Class B Common Stock outstanding or subject to outstanding securities convertible into or exercisable therefor, or otherwise underlying outstanding equity compensation awards, represents, in the aggregate, less than ten percent (10%) of all shares of Common Stock outstanding or subject to outstanding securities convertible into or exercisable therefor, or otherwise underlying outstanding equity compensation awards, in each case, measured on the last day of such fiscal quarter. Each outstanding stock certificate that, immediately prior to such conversion, represented one or more shares of Class B Common Stock subject to such conversion will, upon such conversion, be deemed to represent an equal number of shares of Class A Common Stock, without the need for surrender or exchange thereof. The Corporation will, upon the request of any holder whose shares of Class B Common Stock have been converted into shares of Class A Common Stock as a result of such conversion and upon surrender by such holder to the Corporation of the outstanding certificate(s) formerly representing such holder's shares of Class B Common Stock (if any), issue and deliver to such holder certificate(s) representing the shares of Class A Common Stock into which such holder's shares of Class B Common Stock were converted as a result of such conversion (if such shares are certificated) or, if such shares are uncertificated or the stockholder otherwise consents, register such shares in book-entry form.

Section 4.8. Retirement of Class B Common Stock. Following any conversion of shares of Class B Common Stock, the reissuance of such shares of Class B Common Stock shall be prohibited, and such shares of Class B Common Stock will be retired by the Corporation and cancelled in accordance with the DGCL and, as applicable, the filing with the Delaware Secretary of State required thereby. Upon such retirement and filing with respect to all shares of Class B Common Stock, all references herein to Class A Common Stock will be deemed to be references to Common Stock. Each outstanding stock certificate that, immediately prior to such retirement and filing, represented one or more shares of Class A Common Stock will, following such retirement and filing, be deemed to represent an equal number of shares of Common Stock, without the need for surrender or exchange thereof.

Section 4.9. <u>Verification of Conversion</u>. If the Corporation has reason to believe that a Transfer giving rise to a conversion of shares of Class B Common Stock into Class A Common Stock has occurred but has not theretofore been reflected on the books of the Corporation, the Corporation may request that the holder of such shares furnish affidavits or other evidence to the Corporation as the Corporation deems necessary to determine whether a conversion of shares of Class B Common Stock to Class A Common Stock has occurred, and if such holder does not within twenty-five (25) days after the date of such request furnish sufficient evidence to the Corporation (in the manner provided in the request) to enable the Corporation to determine that no such conversion has occurred, any such shares of Class B Common Stock, to the extent not previously converted, shall be automatically converted into shares of Class A Common Stock as of the date of the Transfer in question and the same will thereupon be registered on the books, records and stock ledger of the Corporation. In connection with any action of stockholders taken at a meeting or by written consent (if action by written consent of the stockholders is not prohibited at such time under this Certificate of Incorporation), the stock ledger of the Corporation shall be presumptive evidence as to who are the stockholders entitled to vote in person or by proxy at any meeting of stockholders and the class or classes or series of shares held by each such stockholder and the number of shares of each class or classes or series held by such stockholder.

- 4.10. <u>Definitions</u>. For purposes of this *Article IV* of this Certificate of Incorporation:
- (a) "Certificate of Merger" means the certificate of merger merging Merger Sub into Grove that is duly filed with the Secretary of State of the State of Delaware
- (b) "Effective Time" means the date and time that the Certificate of Merger is duly filed with the Secretary of State of the State of Delaware or any other such date and time as the Merger Sub and Grove shall agree in writing and shall specify in the Certificate of Merger.
- (c) "Family Member" means an individual's spouse, ex-spouse, domestic partners, lineal (including by adoption) descendant or antecedent, brother or sister, the adopted child or adopted grandchild, or the spouse or domestic partner of any child, adopted child, grandchild or adopted grandchild of such individual
- (d) "Grove" means Grove Collaborative, Inc.
- (e) "Merger Sub" means Treehouse Merger Sub, Inc.
- (f) "Permitted Entity" means, with respect to a Permitted Class B Owner, (i) a Permitted Trust solely for the benefit of such Permitted Class B Owner and/or any other Permitted Entity of such Permitted Class B Owner, or (ii) any general partnership, limited partnership, limited liability company, corporation or other entity exclusively owned by such Permitted Class B Owner and/or any other Permitted Entity of such Permitted Class B Owner.
- (g) "Permitted Transfer" means any Transfer of a share of Class B Common Stock by a Permitted Class B Owner to (i) any other Permitted Class B Owner, (ii) a Permitted Entity of such Class B Owner, and (iii) such Permitted Class B Owner's Family Members or to a trust of trusts or other estate planning vehicle for the benefit of such Permitted Class B Owner's Family Members, in each case so long as a Permitted Class B Owner continues to exercise Voting Control over such shares.
- (h) "Permitted Trust" means a bona fide trust where each trustee is (i) a Permitted Class B Owner or (ii) a professional in the business of providing trustee services, including private professional fiduciaries, trust companies and bank trust departments.
- (i) "Transfer" of a share of Class B Common Stock means any sale, assignment, gift, transfer, conveyance, hypothecation or other transfer or disposition, whether direct or indirect, of such share of Class B Common Stock or any legal or beneficial interest in such share of Class B Common Stock, whether or not for value and whether voluntary or involuntary or by operation of law (including by merger, consolidation or otherwise), including, without limitation, a transfer of a share of Class B Common Stock to a broker or other nominee (involving a corresponding change in beneficial ownership), or the transfer of, or entering into a binding agreement with respect to, Voting Control over such share by proxy or otherwise (other than proxy(ies), voting instruction(s) or voting agreement(s) solicited on behalf of the Board). Notwithstanding the foregoing, the pledge of shares of Class B Common Stock by a stockholder that creates a mere security interest in such shares pursuant to a bona fide loan or indebtedness transaction for so long as such stockholder continues to exercise Voting Control over such pledged shares shall not be considered a "Transfer" within the meaning of this Article IV; provided, however, that a foreclosure on such shares or other similar action by the pledgee shall constitute a "Transfer" unless such foreclosure or similar action independently qualifies as a "Permitted Transfer" at such time.
- (j) "Voting Control" with respect to a share of Class B Common Stock means the power (whether exclusive or shared) to vote or direct the voting of such share of Class B Common Stock by proxy, voting agreement or otherwise.
- 4.11. Reservation of Stock. The Corporation shall at all times reserve and keep available out of its authorized but unissued shares of Class A Common Stock, solely for the purpose of effecting the conversion of the shares of Class B Common Stock, such number of shares of Class A Common Stock as will from time to time be sufficient to effect the conversion of all outstanding shares of Class B Common Stock into shares of Class A Common Stock.
- 4.12. <u>Protective Provisions</u>. So long as any shares of Class B Common Stock remain outstanding, the Corporation shall not, whether by merger, consolidation, reclassification of capital stock or otherwise, amend, alter, change, repeal or waive any provision in this *Article IV* (or adopt any provision inconsistent therewith) that would adversely affect the rights of holders of Class B Common Stock, without first obtaining any vote required by applicable law, this Certificate of Incorporation or the Bylaws.

ARTICLE V BOARD OF DIRECTORS

Section 5.1 <u>Board Powers</u>. The business and affairs of the Corporation shall be managed by, or under the direction of, the Board. In addition to the powers and authority expressly conferred upon the Board by statute, this Certificate of Incorporation, as it may be further amended from time to time, or the Bylaws of the Corporation (as amended from time to time in accordance with the provisions hereof and thereof, the "*Bylaws*"), the Board is hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation, subject, nevertheless, to the provisions of the DGCL and this Certificate of Incorporation.

Section 5.2 Number, Election and Term.

- (a) The number of directors of the Corporation, other than those who may be elected by the holders of one or more series of the Preferred Stock voting separately by class or series, shall be fixed from time to time in the manner provided in the Bylaws.
- (b) Subject to Section 5.5 hereof, the Board shall be divided into three classes, as nearly equal in number as possible and designated Class I, Class II and Class III. The Board is authorized to assign members of the Board already in office to Class I, Class II or Class III. The term of the initial Class I Directors shall expire at the first annual meeting of the stockholders of the Corporation following the effectiveness of this Certificate of Incorporation; the term of the initial Class II Directors shall expire at the second annual meeting of the stockholders of the Corporation following the effectiveness of this Certificate of Incorporation; and the term of the initial Class III Directors shall expire at the third annual meeting of the stockholders of the Corporation following the effectiveness of this Certificate of Incorporation. At each succeeding annual meeting of the stockholders of the Corporation, beginning with the first annual meeting of the stockholders of the Corporation following the effectiveness of this Certificate of Incorporation, each of the successors elected to replace the class of directors whose term expires at that annual meeting shall be elected for a three-year term or until the election and qualification of their respective successors in office, subject to their earlier death, resignation or removal. Subject to Section 5.5 hereof, if the number of directors that constitutes the Board is changed, any increase or decrease shall be apportioned by the Board among the classes so as to maintain the number of directors in each class as nearly equal as possible, but in no case shall a decrease in the number of directors constituting the Board shorten the term of any incumbent director. Subject to the rights of the holders of one or more series of Preferred Stock, voting separately by class or series, to elect directors pursuant to the terms of one or more series of Preferred Stock, the election of directors shall be determined by a plurality of the votes cast by the stockholders present in person or represented by proxy at the meeting and entitled to vote thereon. The Board is hereby expressly authorized, by resolution or resolutions thereof, to assign members of the Board already in office to the aforesaid classes at the time this Certificate of Incorporation (and therefore such classification) becomes effective in accordance with the DGCL.
- (c) Subject to Section 5.5 hereof, a director shall hold office until the annual meeting for the year in which his or her term expires and until his or her successor has been elected and qualified, subject, however, to such director's earlier death, resignation, retirement, disqualification or removal.
- (d) Unless and except to the extent that the Bylaws shall so require, the election of directors need not be by written ballot. The holders of shares of Common Stock shall not have cumulative voting rights.

Section 5.3 Newly Created Directorships and Vacancies. Subject to Section 5.5 hereof, newly created directorships resulting from an increase in the number of directors and any vacancies on the Board resulting from death, resignation, retirement, disqualification, removal or other cause may be filled solely and exclusively by a majority vote of the remaining directors then in office, even if less than a quorum, or by a sole remaining director (and not by stockholders), and any director so chosen shall hold office for the remainder of the full term of the class of directors to which the new directorship was added or in which the vacancy occurred and until his or her successor has been elected and qualified, subject, however, to such director's earlier death, resignation, retirement, disqualification or removal.

Section 5.4 <u>Removal</u>. Subject to Section 5.5 hereof and except as otherwise required by law, any or all of the directors may be removed from office at any time, but only by the affirmative vote of holders of a majority of the voting power of all then outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class and, for so long as the Board is classified pursuant to Section 5.2(b), only for cause

Section 5.5 <u>Preferred Stock—Directors</u>. Notwithstanding any other provision of this *Article V*, and except as otherwise required by law, whenever the holders of one or more series of the Preferred Stock shall have the right, voting separately by class or series, to elect one or more directors, the term of office, the filling of vacancies, the removal from office and other features of such directorships shall be governed by the terms of such series of the Preferred Stock as set forth in this Certificate of Incorporation (including any Preferred Stock Designation) and such directors shall not be included in any of the classes created pursuant to this *Article V* unless expressly provided by such terms.

ARTICLE VI BYLAWS

In furtherance and not in limitation of the powers conferred upon it by law, the Board shall have the power and is expressly authorized to adopt, amend, alter or repeal the Bylaws by the affirmative vote of a majority of the total number of directors present at a regular or special meeting of the Board at which there is a quorum or by unanimous written consent. The Bylaws also may be adopted, amended, altered or repealed by the stockholders; provided, however, that in addition to any vote of the holders of any class or series of capital stock of the Corporation required by law or by this Certificate of Incorporation (including any Preferred Stock Designation), the affirmative vote of the holders of at least a majority of the voting power of all then outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required for the stockholders of the Corporation to adopt, amend, alter or repeal the Bylaws; and provided further, however, that no Bylaws hereafter adopted by the stockholders of the Corporation shall invalidate any prior act of the Board that would have been valid if such Bylaws had not been adopted.

ARTICLE VII MEETINGS OF STOCKHOLDERS; ACTION BY WRITTEN CONSENT

Section 7.1 <u>Advance Notice</u>. Advance notice of stockholder nominations for the election of directors and of business to be brought by stockholders before any meeting of the stockholders of the Corporation shall be given in the manner provided in the Bylaws.

Section 7.2 <u>Action by Written Consent</u>. Subject to the terms of any series of Preferred Stock then outstanding, (a) no action shall be taken by the stockholders of the Corporation except at a duly called annual or special meeting of stockholders and (b) no action shall be taken by the stockholders of the Corporation by written consent in lieu of a meeting.

ARTICLE VIII LIMITED LIABILITY; INDEMNIFICATION

Section 8.1 <u>Limitation of Director Liability</u>. A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permitted under the DGCL as the same exists or may hereafter be amended unless a director violated his or her duty of loyalty to the Corporation or its stockholders, acted in bad faith, knowingly or intentionally violated the law, authorized unlawful payments of dividends, unlawful stock purchases or unlawful redemptions, or derived improper personal benefit from his or her actions as a director. Any amendment, modification or repeal of the foregoing sentence shall not adversely affect any right or protection of a director of the Corporation hereunder in respect of any act or omission occurring prior to the time of such amendment, modification or repeal.

Section 8.2 Indemnification and Advancement of Expenses.

- (a) To the fullest extent permitted by applicable law, as the same exists or may hereafter be amended, the Corporation shall indemnify and hold harmless each person who is or was made a party or is threatened to be made a party to or is otherwise involved in any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, including an action by and in the right of the Corporation to procure a judgment in its favor (a "proceeding") by reason of the fact that he or she is or was a director or officer of the Corporation or, while a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, other enterprise or nonprofit entity, including service with respect to an employee benefit plan (an "indemnitee"), whether the basis of such proceeding is alleged action in an official capacity as a director, officer, employee or agent, or in any other capacity while serving as a director, officer, employee or agent, against all liability and loss suffered and expenses (including, without limitation, attorneys' fees and disbursement, judgments, fines, ERISA excise taxes, damages, claims and penalties and amounts paid in settlement) reasonably incurred by such indemnitee in connection with such proceeding. The Corporation shall, to the fullest extent not prohibited by applicable law, pay the expenses (including attorneys' fees) incurred by an indemnitee in defending or otherwise participating in any proceeding in advance of its final disposition; provided, however, that, to the extent required by applicable law, such payment of expenses in advance of the final disposition of the proceeding shall be made only upon receipt of an undertaking, by or on behalf of the indemnitee, to repay all amounts so advanced if it shall ultimately be determined that the indemnitee is not entitled to be indemnified under this Section 8.2 or otherwise. The rights to indemnification and advancement of expenses conferred by this Section 8.2 shall be contractual rights and such rights shall continue as to an indemnitee who has ceased to be a director, officer, employee or agent and shall inure to the benefit of his or her heirs, executors and administrators. Notwithstanding the foregoing provisions of this Section 8.2(a), except for proceedings to enforce rights to indemnification and advancement of expenses, the Corporation shall indemnify and advance expenses to an indemnitee in connection with a proceeding (or part thereof) initiated by such indemnitee only if such proceeding (or part thereof) was authorized by the Board.
- (b) The rights to indemnification and advancement of expenses conferred on any indemnitee by this Section 8.2 shall not be exclusive of any other rights that any indemnitee may have or hereafter acquire under law, this Certificate of Incorporation, as it may be further amended from time to time, the Bylaws, an agreement, vote of stockholders or disinterested directors, or otherwise.
- (c) Any repeal or amendment of this Section 8.2 by the stockholders of the Corporation or by changes in law, or the adoption of any other provision of this Certificate of Incorporation inconsistent with this Section 8.2, shall, unless otherwise required by law, be prospective only (except to the extent such amendment or change in law permits the Corporation to provide broader indemnification rights on a retroactive basis than permitted prior thereto), and shall not in any way diminish or adversely affect any right or protection existing at the time of such repeal or amendment or adoption of such inconsistent provision in respect of any proceeding (regardless of when such proceeding is first threatened, commenced or completed) arising out of, or related to, any act or omission occurring prior to such repeal or amendment or adoption of such inconsistent provision.
- (d) This Section 8.2 shall not limit the right of the Corporation, to the extent and in the manner authorized or permitted by law, to indemnify and to advance expenses to persons other than indemnitees.

ARTICLE IX CORPORATE OPPORTUNITY

Section 9.1. To the extent allowed by law, the doctrine of corporate opportunity, or any other analogous doctrine, shall not apply with respect to the Corporation or any of its officers or directors in circumstances where the application of any such doctrine to a corporate opportunity would conflict with any fiduciary duties or contractual obligations they may have as of the date of this Certificate of Incorporation or in the future, and the Corporation renounces any expectancy that any of the directors or officers of the Corporation will offer any such corporate opportunity of which he or she may become aware to the Corporation.

Section 9.2. Neither the alteration, amendment, addition to or repeal of this *Article IX*, nor the adoption of any provision of this Certificate of Incorporation (including any Preferred Stock Designation) inconsistent with this *Article IX*, shall eliminate or reduce the effect of this *Article IX* in respect of any business opportunity first identified or any other matter occurring, or any cause of action, suit or claim that, but for this *Article IX*, would accrue or arise, prior to such alteration, amendment, addition, repeal or adoption. This *Article IX* shall not limit any protections or defenses available to, or indemnification or advancement rights of, any director or officer of the Corporation under this Certificate of Incorporation, the Bylaws or applicable law.

ARTICLE X AMENDMENT OF CERTIFICATE OF INCORPORATION

The Corporation reserves the right at any time and from time to time to amend, alter, change or repeal any provision contained in this Certificate of Incorporation (including any Preferred Stock Designation), and other provisions authorized by the laws of the State of Delaware at the time in force that may be added or inserted, in the manner now or hereafter prescribed by this Certificate of Incorporation and the DGCL; and, except as set forth in *Article VIII*, all rights, preferences and privileges of whatever nature herein conferred upon stockholders, directors or any other persons by and pursuant to this Certificate of Incorporation in its present form or as hereafter amended are granted subject to the right reserved in this *Article X*. Notwithstanding any other provisions of this Certificate of Incorporation or any provision of law which might otherwise permit a lesser vote or no vote, but in addition to any affirmative vote of the holders of any particular class or series of the Corporation required by law or by this Certificate of Incorporation or any Preferred Stock Designation filed with respect to a series of Preferred Stock, the affirmative vote of the stockholders holding at least 66.7% of the voting power of all outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required to alter, amend or repeal *Article V or Article VIII*.

ARTICLE XI EXCLUSIVE FORUM FOR CERTAIN LAWSUITS; CONSENT

Section 11.1 Forum. Subject to the last sentence in this Section 11.1 and unless the Corporation consents in writing to the selection of an alternative forum, to the fullest extent permitted by the applicable law, the Court of Chancery of the State of Delaware shall be the sole and exclusive forum for any stockholder (including a beneficial owner) to bring (a) any derivative action or proceeding brought on behalf of the Corporation, (b) any action asserting a claim of breach of a fiduciary duty owed by any director, officer or other employee of the Corporation to the Corporation or the Corporation's stockholders, (c) any action asserting a claim against the Corporation, its directors, officers or employees arising pursuant to any provision of the DGCL or this Certificate of Incorporation or the Bylaws, or (d) any action asserting a claim against the Corporation, its directors, officers or employees governed by the internal affairs doctrine and, if brought outside of Delaware, the stockholder bringing the suit will be deemed to have consented to service of process on such stockholder's counsel except any action (i) as to which the Court of Chancery in the State of Delaware determines that there is an indispensable party not subject to the jurisdiction of the Court of Chancery (and the indispensable party does not consent to the personal jurisdiction of the Court of Chancery within ten days following such determination), (ii) which is vested in the exclusive jurisdiction of a court or forum other than the Court of Chancery, (iii) for which the Court of Chancery does not have subject matter jurisdiction, or (iv) any action arising under the Securities Act of 1933, as amended (the "Securities Act)", as to which the Court of Chancery and the Federal District court of the District of Delaware shall have concurrent jurisdiction. Notwithstanding the foregoing, (A) the provisions of this Section 11.1 will not apply to suits brought to enforce any liability or duty created by the Securities Exchange Act of 1934, as amended, the Securities Act or any other claim for which the federal courts have exclusive jurisdiction and (B) unless the Corporation consents in writing to the selection of an alternative forum, the federal district courts of the United States of America shall, to the fullest extent permitted by law, be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act of 1933, as amended, or the rules and regulations promulgated thereunder.

Section 11.2 <u>Consent to Jurisdiction</u>. If any action the subject matter of which is within the scope of Section 11.1 immediately above is filed in a court other than a court located within the State of Delaware (a "Foreign Action") in the name of any stockholder, such stockholder shall be deemed to have consented to (a) the personal jurisdiction of the state and federal courts located within the State of Delaware in connection with any action brought in any such court to enforce Section 11.1 immediately above (an "FSC Enforcement Action") and (b) having service of process made upon such stockholder in any such FSC Enforcement Action by service upon such stockholder's counsel in the Foreign Action as agent for such stockholder.

Section 11.3 Severability. If any provision or provisions of this Article XI shall be held to be invalid, illegal or unenforceable as applied to any person or entity or circumstance for any reason whatsoever, then, to the fullest extent permitted by law, the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Article XI (including, without limitation, each portion of any sentence of this Article XI containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) and the application of such provision to other persons or entities and circumstances shall not in any way be affected or impaired thereby.

Section 11.4 <u>Demand Notice</u>. Any person or entity purchasing or otherwise acquiring or holding any interest in any security of the Corporation shall be deemed to have notice of and consented to the provisions of this *Article XI*.

ARTICLE XII

The name and mailing address of the incorporator are:

Name	Mailing Address
Evan Lovell	65 Bleecker Street
	6th Floor
	New York, NY 10012

[Signature page follows]

IN WITNESS WHEREOF, the undersigned has executed this Certificate of Incorporation as of the date first set forth above.

/s/ Evan Lovell Evan Lovell Incorporator

[Signature Page to Certificate of Incorporation]

BYLAWS OF GROVE COLLABORATIVE HOLDINGS, INC., A PUBLIC BENEFIT CORPORATION (THE "CORPORATION")

(Adopted on June 16, 2022)

ARTICLE I OFFICES

Section 1.1 Registered Office. The registered office of the Corporation within the State of Delaware shall be located at either (a) the principal place of business of the Corporation in the State of Delaware or (b) the office of the Corporation or individual acting as the Corporation's registered agent in the State of Delaware.

Section 1.2 Additional Offices. The Corporation may, in addition to its registered office in the State of Delaware, have such other offices and places of business, both within and outside the State of Delaware, as the Board of Directors of the Corporation (the "Board") may from time to time determine or as the business and affairs of the Corporation may require.

ARTICLE II STOCKHOLDERS MEETINGS

Section 2.1 Annual Meetings. The annual meeting of stockholders of the Corporation shall be held at such place, either within or without the State of Delaware, and time and on such date as shall be determined by the Board and stated in the notice of the meeting, *provided* that the Board may in its sole discretion determine that the meeting shall not be held at any place, but may instead be held solely by means of remote communication pursuant to Section 11.5(a). The stockholders entitled to vote at each annual meeting shall elect directors of the Corporation to fill any term of a directorship that expires on the date of such annual meeting and may transact any other business as may properly be brought before the meeting.

Section 2.2 Special Meetings. Subject to the rights of the holders of any outstanding series of the preferred stock of the Corporation ('Preferred Stock') and to the requirements of applicable law, special meetings of stockholders, for any purpose or purposes, may be called only by the Chairperson of the Board, Chief Executive Officer of the Corporation, the Board pursuant to a resolution adopted by a majority of the Board, or when requested in writing by the holders of not less than 20% of all votes entitled to be cast at the meeting, and may not be called by any other person. Special meetings of stockholders shall be held at such place, either within or without the State of Delaware, and at such time and on such date as shall be determined by the Board and stated in the Corporation's notice of the meeting, provided that the Board may in its sole discretion determine that the meeting shall not be held at any place, but may instead be held solely by means of remote communication pursuant to Section 11.5(a).

Section 2.3 Notices. Written notice of each stockholders meeting stating the place, if any, date, and time of the meeting, and the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting, and the record date for determining the stockholders entitled to vote at the meeting, if such date is different from the record date for determining stockholders entitled to notice of the meeting, shall be given, in the manner permitted by Section 11.3 to each stockholder entitled to vote thereat as of the record date for determining the stockholders entitled to notice of the meeting, by the Corporation not less than 10 nor more than 60 days before the date of the meeting unless otherwise required by the General Corporation Law of the State of Delaware (the "DGCL"). If said notice is for a stockholders meeting other than an annual meeting, it shall in addition state the purpose or purposes for which the meeting is called, and the business transacted at such meeting shall be limited to the matters so stated in the Corporation's notice of meeting (or any supplement thereto). Any meeting of stockholders as to which notice has been given may be postponed, and any meeting of stockholders as to which notice has been given may be cancelled, by the Board upon public announcement (as defined in Section 2.7(c)) given before the date previously scheduled for such meeting.

Section 2.4 Quorum. Except as otherwise provided by applicable law, the Corporation's Certificate of Incorporation, as the same may be amended or restated from time to time (the "Certificate of Incorporation"), or these Bylaws, the presence, in person or by proxy, at a stockholders meeting of the holders of shares of outstanding capital stock of the Corporation representing a majority of the voting power of all outstanding shares of capital stock of the Corporation entitled to vote at such meeting shall constitute a quorum for the transaction of business at such meeting, except that when specified business is to be voted on by a class or series of stock voting as a class, the holders of shares representing a majority of the voting power of the outstanding shares of such class or series shall constitute a quorum of such class or series for the transaction of such business. If a quorum shall not be present or represented by proxy at any meeting of the stockholders of the Corporation, the chairperson of the meeting may adjourn the meeting from time to time in the manner provided in Section 2.6 until a quorum shall attend. The stockholders present at a duly convened meeting may continue to transact business until adjournment, notwithstanding the withdrawal of enough stockholders to leave less than a quorum. Shares of its own stock belonging to the Corporation or to another corporation, if a majority of the voting power of the shares entitled to vote in the election of directors of such other corporation is held, directly or indirectly, by the Corporation, shall neither be entitled to vote nor be counted for quorum purposes; provided, however, that the foregoing shall not limit the right of the Corporation or any such other corporation to vote shares held by it in a fiduciary capacity.

Section 2.5 Voting of Shares.

- (a) <u>Voting Lists</u>. The officer who has charge of the stock ledger of the Corporation shall prepare and make, at least 10 days before every meeting of stockholders, a complete list of the stockholders of record entitled to vote at such meeting; *provided*, *however*, that if the record date for determining the stockholders entitled to vote is less than 10 days before the meeting date, the list shall reflect the stockholders entitled to vote as of the tenth day before the meeting date, arranged in alphabetical order and showing the address and the number and class of shares registered in the name of each stockholder. Nothing contained in this <u>Section 2.5(a)</u> shall require the Corporation to include electronic mail addresses or other electronic contact information on such list. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours for a period of at least 10 days prior to the meeting: (i) on a reasonably accessible electronic network, *provided* that the information required to gain access to such list is provided with the notice of the meeting, or (ii) during ordinary business hours, at the principal place of business of the Corporation. In the event that the Corporation determines to make the list available on an electronic network, the Corporation may take reasonable steps to ensure that such information is available only to stockholders of the Corporation. If the meeting is to be held at a place, then the list shall be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present. If a meeting of stockholders is to be held solely by means of remote communication as permitted by <u>Section 11.5(a)</u>, the list shall be open to the examination of any stockholder during the whole time of the meeting on a reasonably accessible electronic network, and the information required to access such list shall be provided with the notice of meeting. The stock ledger sha
- (b) <u>Manner of Voting</u>. At any stockholders meeting, every stockholder entitled to vote may vote in person or by proxy. If authorized by the Board, the voting by stockholders or proxy holders at any meeting conducted by remote communication may be effected by a ballot submitted by electronic transmission (as defined in <u>Section 11.3(c)</u>), provided that any such electronic transmission must either set forth or be submitted with information from which the Corporation can determine that the electronic transmission was authorized by the stockholder or proxy holder. The Board, in its discretion, or the chairperson of the meeting of stockholders, in such person's discretion, may require that any votes cast at such meeting shall be cast by written ballot.
- (c) <u>Proxies</u>. Each stockholder entitled to vote at a meeting of stockholders or to express consent or dissent to corporate action in writing without a meeting may authorize another person or persons to act for such stockholder by proxy, but no such proxy shall be voted or acted upon after three years from its date, unless the proxy provides for a longer period. Proxies need not be filed with the Secretary of the Corporation until the meeting is called to order, but shall be filed with the Secretary before being voted. Without limiting the manner in which a stockholder may authorize another person or persons to act for such stockholder as proxy, either of the following shall constitute a valid means by which a stockholder may grant such authority:

- (i) A stockholder may execute a writing authorizing another person or persons to act for such stockholder as proxy. Execution may be accomplished by the stockholder or such stockholder's authorized officer, director, employee or agent signing such writing or causing such person's signature to be affixed to such writing by any reasonable means, including, but not limited to, by facsimile signature.
- (ii) A stockholder may authorize another person or persons to act for such stockholder as proxy by transmitting or authorizing the transmission of an electronic transmission to the person who will be the holder of the proxy or to a proxy solicitation firm, proxy support service organization or like agent duly authorized by the person who will be the holder of the proxy to receive such transmission, provided that any such electronic transmission must either set forth or be submitted with information from which it can be determined that the electronic transmission was authorized by the stockholder. Any copy, facsimile telecommunication or other reliable reproduction of the writing or transmission authorizing another person or persons to act as proxy for a stockholder may be substituted or used in lieu of the original writing or transmission could be used; provided that such copy, facsimile telecommunication or other reproduction shall be a complete reproduction of the entire original writing or transmission.
- (d) Required Vote. Subject to the rights of the holders of one or more series of Preferred Stock, voting separately by class or series, to elect directors pursuant to the terms of one or more series of Preferred Stock, at all meetings of stockholders at which a quorum is present, the election of directors shall be determined by a plurality of the votes cast by the stockholders present in person or represented by proxy at the meeting and entitled to vote thereon. All other matters presented to the stockholders at a meeting at which a quorum is present shall be determined by the vote of a majority of the votes cast by the stockholders present in person or represented by proxy at the meeting and entitled to vote thereon, unless the matter is one upon which, by applicable law, the Certificate of Incorporation, these Bylaws or applicable stock exchange rules, a different vote is required, in which case such provision shall govern and control the decision of such matter.
- (e) Inspectors of Election. The Board may, and shall if required by law, in advance of any meeting of stockholders, appoint one or more persons as inspectors of election, who may be employees of the Corporation or otherwise serve the Corporation in other capacities, to act at such meeting of stockholders or any adjournment thereof and to make a written report thereof. The Board may designate one or more persons as alternate inspectors to replace any inspector who fails to act. If no inspectors of election or alternates are appointed by the Board, the chairperson of the meeting shall appoint one or more inspectors to act at the meeting. Each inspector, before discharging their duties, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of their ability. The inspectors shall ascertain and report the number of outstanding shares and the voting power of each; determine the number of shares present in person or represented by proxy at the meeting and the validity of proxies and ballots; count all votes and ballots and report the results; determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspectors; and certify their determination of the number of shares represented at the meeting and their count of all votes and ballots. No person who is a candidate for an office at an election may serve as an inspector at such election. Each report of an inspector shall be in writing and signed by the inspector or by a majority of them if there is more than one inspector acting at such meeting. If there is more than one inspector, the report of a majority shall be the report of the inspectors.

Section 2.6 Adjournments. Any meeting of stockholders, annual or special, may be adjourned by the chairperson of the meeting, from time to time, whether or not there is a quorum, to reconvene at the same or some other place. Notice need not be given of any such adjourned meeting if the date, time, and place, if any, thereof, and the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such adjourned meeting are announced at the meeting at which the adjournment is taken. At the adjourned meeting, the stockholders, or the holders of any class or series of stock entitled to vote separately as a class, as the case may be, may transact any business that might have been transacted at the original meeting. If the adjournment is for more than 30 days, notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If after the adjournment a new record date for stockholders entitled to vote is fixed for the adjourned meeting, the Board shall fix a new record date for notice of such adjourned meeting to each stockholder of record entitled to vote at such adjourned meeting as of the record date for notice of such adjourned meeting.

Section 2.7 Advance Notice for Business.

- (a) <u>Annual Meetings of Stockholders</u>. No business may be transacted at an annual meeting of stockholders, other than business that is either (i) specified in the Corporation's notice of meeting (or any supplement thereto) given by or at the direction of the Board, (ii) otherwise properly brought before the annual meeting by or at the direction of the Board or (iii) otherwise properly brought before the annual meeting by any stockholder of the Corporation (x) who is a stockholder of record entitled to vote at such annual meeting on the date of the giving of the notice provided for in this <u>Section 2.7(a)</u> and on the record date for the determination of stockholders entitled to vote at such annual meeting and (y) who complies with the notice procedures set forth in this <u>Section 2.7(a)</u>. Notwithstanding anything in this <u>Section 2.7(a)</u> to the contrary, only persons nominated for election as a director to fill any term of a directorship that expires on the date of the annual meeting pursuant to <u>Section 3.2</u> will be considered for election at such meeting.
- (i) In addition to any other applicable requirements, for business (other than nominations) to be properly brought before an annual meeting by a stockholder, such stockholder must have given timely notice thereof in proper written form to the Secretary of the Corporation and such business must otherwise be a proper matter for stockholder action. Subject to Section 2.7(a)(iii), a stockholder's notice to the Secretary with respect to such business, to be timely, must be received by the Secretary at the principal executive offices of the Corporation not later than the Close of Business (as defined below) on the 90th day nor earlier than the Close of Business on the 120th day before the anniversary date of the immediately preceding annual meeting of stockholders; provided, however, that in the event that the annual meeting is more than 30 days before or more than 60 days after such anniversary date, notice by the stockholder to be timely must be so received not earlier than the Close of Business on the 120th day before the meeting and not later than the later of (x) the Close of Business on the 90th day before the meeting or (y) the Close of Business on the 10th day following the day on which public announcement of the date of the annual meeting is first made by the Corporation. The public announcement of an adjournment or postponement of an annual meeting shall not commence a new time period (or extend any time period) for the giving of a stockholder's notice as described in this Section 2.7(a).
- (ii) To be in proper written form, any stockholder's notice to the Secretary (the "Noticing Party") with respect to any business (other than director nominations) must set forth as to each such matter such stockholder proposes to bring before the annual meeting (A) a brief description of the business desired to be brought before the annual meeting, the text of the proposal or business (including the text of any resolutions proposed for consideration and in the event such business includes a proposal to amend these Bylaws or the Certificate of Incorporation, the language of the proposed amendment) and the reasons for conducting such business at the annual meeting, (B) all other information relating to such business that would be required to be disclosed in a proxy statement or other filing made by such Noticing Party or any Stockholder Associated Person (as defined below) in connection with the solicitation of proxies in support of such proposed business by such Noticing Party or any Stockholder Associated Person pursuant to the pursuant to Section 14 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the rules and regulations promulgated thereunder (collectively, the "Proxy Rules"), (C) the name and address of such Noticing Party and each Stockholder Associated Person (including, as applicable, as they appear on the Corporation's books and records), (D) the class, series and number of shares of each class or series of capital stock (if any) of the Corporation that are, directly or indirectly, owned beneficially and/or of record by such Noticing Party or any Stockholder Associated Person and the date or dates such shares were acquired and the investment intent of such acquisition, (E) the name of each nominee holder for, and number of, any securities of the Corporation owned beneficially but not of record by such Noticing Party or any Stockholder Associated Person and any pledge by such Noticing Party or any Stockholder Associated Person with respect to any of such securities, (F) a complete and accurate description of all agreements, arrangements or understandings, written or oral (including any derivative or short positions, profit interests, hedging transactions, forwards, futures, swaps, options, warrants, convertible securities, stock appreciation or similar rights, repurchase agreements or arrangements, borrowed or loaned shares and so-called "stock borrowing" agreements or arrangements) that have been entered into by, or on behalf of, such Noticing Party or any Stockholder Associated Person, the effect or intent of which is to mitigate loss, manage risk or benefit from changes in the price of any securities of the Corporation, or maintain, increase or decrease the voting power of such Noticing Party or any Stockholder Associated Person with respect to securities of

the Corporation, whether or not such instrument or right shall be subject to settlement in underlying shares of capital stock of the Corporation and without regard to whether such agreement, arrangement or understanding is required to be reported on a Schedule 13D in accordance with the Exchange Act (any of the foregoing, a "Derivative Instrument"), (G) any substantial interest, direct or indirect (including any existing or prospective commercial, business or contractual relationship with the Corporation), by security holdings or otherwise, of such Noticing Party or any Stockholder Associated Person in the Corporation or any affiliate (as defined below) thereof, other than an interest arising from the ownership of Corporation securities where such Noticing Party or such Stockholder Associated Person receives no extra or special benefit not shared on a pro rata basis by all other holders of the same class or series, (H) a complete and accurate description of all agreements, arrangements or understandings, written or oral, (I) between or among such Noticing Party and any of the Stockholder Associated Persons or (II) between or among such Noticing Party or any Stockholder Associated Person and any other person or entity (naming each such person or entity), including, without limitation, (x) any proxy, contract, arrangement, understanding or relationship pursuant to which such Noticing Party or any Stockholder Associated Person, directly or indirectly, has a right to vote any security of the Corporation (other than any revocable proxy given in response to a solicitation made pursuant to, and in accordance with, Section 14(a) of the Exchange Act by way of a solicitation statement filed on Schedule 14A), (y) any understanding, written or oral, that such Noticing Party or any Stockholder Associated Person may have reached with any stockholder of the Corporation (including the name of such stockholder) with respect to how such stockholder will vote such stockholder's shares in the Corporation at any meeting of the Corporation's stockholders or take other action in support of any Proposed Nominee (as defined below) or other business, or other action to be taken, by such Noticing Party or any Stockholder Associated Person and (z) any other agreements that would be required to be disclosed by such Noticing Party, any Stockholder Associated Person or any other person or entity pursuant to Item 5 or Item 6 of a Schedule 13D pursuant to Section 13 of the Exchange Act (regardless of whether the requirement to file a Schedule 13D is applicable to such Noticing Party, such Stockholder Associated Person or such other person or entity), (I) any rights to dividends on the shares of the Corporation owned beneficially by such Noticing Party or any Stockholder Associated Person that are separated or separable from the underlying shares of the Corporation, (J) any proportionate interest in shares of the Corporation or Derivative Instruments held, directly or indirectly, by a general or limited partnership, limited liability company or similar entity in which such Noticing Party or any Stockholder Associated Person is (I) a general partner or, directly or indirectly, beneficially owns an interest in a general partner of such general or limited partnership or (II) the manager, managing member or, directly or indirectly, beneficially owns an interest in the manager or managing member of such limited liability company or similar entity, (K) any significant equity interests or any Derivative Instruments or Short Interests in any principal competitor of the Corporation held by such Noticing Party or any Stockholder Associated Person, (L) any direct or indirect interest of such Noticing Party or any Stockholder Associated Person in any contract with the Corporation, any affiliate of the Corporation or any principal competitor of the Corporation (including, without limitation, any employment agreement, collective bargaining agreement or consulting agreement), (M) a description of any material interest of such Noticing Party or any Stockholder Associated Person in the business proposed by such Noticing Party, if any, or the election of any Proposed Nominee, (N) a complete an accurate description of any performance-related fees (other than an asset-based fee) to which such Noticing Party or any Stockholder Associated Person may be entitled as a result of any increase or decrease in the value of the Corporation's securities or any Derivative Instruments, including, without limitation, any such interests held by members of such Noticing Party's or any Stockholder Associated Person's immediate family sharing the same household, (O) the investment strategy or objective, if any, of such Noticing Party or any Stockholder Associated Person who is not an individual and a copy of the prospectus, offering memorandum or similar document, if any, provided to investors or potential investors in the Noticing Party or any Stockholder Associated Person and (P) all other information relating to such Noticing Party or any Stockholder Associated Person, or such Noticing Party's or any Stockholder Associated Person's associates, that would be required to be disclosed in a proxy statement or other filing in connection with the solicitation of proxies in support of the business proposed by such Noticing Party, if any; provided, however, that the disclosures in the foregoing subclauses (C) through (P) shall not include any such disclosures with respect to the ordinary course business activities of any broker, dealer, commercial bank, trust company or other nominee who is a Noticing Party solely as a result of being the stockholder directed to prepare and submit the notice required by these Bylaws on behalf of a beneficial owner (as defined below). To be in proper written form, any Noticing Party's notice with respect to any business must set forth (A) a representation that such Noticing Party intends to appear in person or by proxy at the meeting to bring such business before the meeting and an acknowledgment that, if such Noticing Party (or a Qualified Representative (as defined below) of such Noticing Party) does not appear to present such business or Proposed Nominees, as applicable, at such meeting, the Corporation need not present such business for a vote at such meeting, notwithstanding that proxies in respect of

such vote may have been received by the Corporation, (B) a complete and accurate description of any pending or, to such Noticing Party's knowledge, threatened legal proceeding in which such Noticing Party or any Stockholder Associated Person is a party or participant involving the Corporation or, to such Noticing Party's knowledge, any officer, director, affiliate or associate of the Corporation and (C) a representation from such Noticing Party as to whether such Noticing Party or any Stockholder Associated Person intends or is part of a group that intends (I) to deliver a proxy statement and/or form of proxy to a number of holders of the Corporation's voting shares reasonably believed by such Noticing Party to be sufficient to approve or adopt the business to be proposed or elect the Proposed Nominees, as applicable, or (II) engage in a solicitation (within the meaning of Exchange Act Rule 14a-1(I) with respect to the nomination or other business, as applicable, and if so, the name of each participant (as defined in Item 4 of Schedule 14A under the Exchange Act) in such solicitation.

(iii) A Noticing Party shall update such Noticing Party's notice provided under the foregoing provisions of this Section 2.7, if necessary, such that the information provided or required to be provided in such notice shall be true and correct (A) as of the record date for determining the stockholders entitled to receive notice of the meeting and (B) as of the date that is 10 business days prior to the meeting (or any postponement, rescheduling or adjournment thereof), and such update shall be received by the Secretary at the principal executive offices of the Corporation (x) not later than the Close of Business five business days after the record date for determining the stockholders entitled to receive notice of such meeting (in the case of an update required to be made under clause (A)) and (y) not later than the Close of Business seven business days prior to the date for the meeting or, if practicable, any postponement, rescheduling or adjournment thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been postponed, rescheduled or adjourned) (in the case of an update required to be made pursuant to clause (B). For the avoidance of doubt, any information provided pursuant to this Section 2.7(a)(iii) shall not be deemed to cure any deficiencies in a notice previously delivered pursuant to this Section 2.7 and shall not extend the time period for the delivery of notice pursuant to this Section 2.7. If a Noticing Party fails to provide such written update within such period, the information as to which such written update relates may be deemed not to have been provided in accordance with this Section 2.7. If any information submitted pursuant to this Section 2.7 by any Noticing Party proposing business for consideration at a stockholder meeting shall be inaccurate in any respect, such information shall be deemed not to have been provided in accordance with this Section 2.7. Any such Noticing Party shall notify the Secretary in writing at the principal executive offices of the Corporation of any inaccuracy or change in any information submitted pursuant to this Section 2.7 within two business days after becoming aware of such inaccuracy or change. Upon written request of the Secretary on behalf of the Board (or a duly authorized committee thereof), any such Noticing Party shall provide, within seven business days after delivery of such request (or such other period as may be specified in such request), (A) written verification, reasonably satisfactory to the Board, any committee thereof or any authorized officer of the Corporation, to demonstrate the accuracy of any information submitted by such Noticing Party pursuant to this Section 2.7 and (B) a written affirmation of any information submitted by such Noticing Party pursuant to this Section 2.7 as of an earlier date. If a Noticing Party fails to provide such written verification or affirmation within such period, the information as to which written verification or affirmation was requested may be deemed not to have been provided in accordance with this Section 2.7.

(iii) The foregoing notice requirements of this Section 2.7(a) shall be deemed satisfied by a stockholder as to any proposal (other than nominations) if the stockholder has notified the Corporation of such stockholder's intention to present such proposal at an annual meeting in compliance with Rule 14a-8 (or any successor thereof) of the Exchange Act, and such stockholder has complied with the requirements of such Rule for inclusion of such proposal in a proxy statement prepared by the Corporation to solicit proxies for such annual meeting. No business shall be conducted at the annual meeting of stockholders except business brought before the annual meeting in accordance with the procedures set forth in this Section 2.7(a), provided, however, that once business has been properly brought before the annual meeting in accordance with such procedures, nothing in this Section 2.7(a) shall be deemed to preclude discussion by any stockholder of any such business. If the Board or the chairperson of the annual meeting determines that any stockholder proposal was not made in accordance with the provisions of this Section 2.7(a), such proposal shall not be presented for action at the annual meeting. Notwithstanding the foregoing provisions of this Section 2.7(a), if the stockholder (or a qualified representative of the stockholder) does not appear at the annual meeting of stockholders of the Corporation to present the proposed business, such proposed business shall not be transacted, notwithstanding that proxies in respect of such matter may have been received by the Corporation.

(iv) In addition to the provisions of this Section 2.7(a), a stockholder shall also comply with all applicable requirements of state law and the Exchange Act and the rules and regulations thereunder with respect to the matters set forth herein. Nothing in this Section 2.7(a) shall be deemed to affect any rights of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act.

(v) For the purposes of these Bylaws, (A) "affiliate" and "associate" each shall have the respective meanings set forth in Rule 12b-2 under the Exchange Act, (B) "beneficial owner" or "beneficially owned" shall have the meaning set forth for such terms in Section 13(d) of the Exchange Act, (C) "Close of Business" shall mean 5:00 p.m. Eastern Time on any calendar day, whether or not the day is a business day, (D) a "Qualified Representative" of a Noticing Party means (I) a duly authorized officer, manager or partner of such Noticing Party or (II) a person authorized by a writing executed by such Noticing Party (or a reliable reproduction or electronic transmission of the writing) delivered by such Noticing Party to the Corporation prior to the making of any nomination or proposal at a stockholder meeting stating that such person is authorized to act for such Noticing Party as proxy at the meeting of stockholders, which writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, must be produced at the meeting of stockholders, and (E) "Stockholder Associated Person" shall mean, with respect to a Noticing Party, (I) any person directly or indirectly controlling, controlled by, under common control with such Noticing Party, (II) any member of the immediate family of such Noticing Party sharing the same household, (III) any person who is a member of a "group" (as such term is used in Rule 13d-5 under the Exchange Act (or any successor provision at law)) with, or is otherwise known by such Noticing Party or other Stockholder Associated Person to be acting in concert with, such Noticing Party or any other Stockholder Associated Person (other than a stockholder that is a depositary), (V) any affiliate or associate of such Noticing Party or any other Stockholder Associated Person (other than a stockholder in paragraphs (a)(ii)-(vi) of Instruction 3 to Item 4 of Schedule 14A) with such Noticing Party or any other Stockholder Associated Person with respect to any proposed busi

(b) <u>Special Meetings of Stockholders</u>. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the meeting pursuant to the Corporation's notice of meeting. Nominations of persons for election to the Board may be made at a special meeting of stockholders at which directors are to be elected pursuant to the Corporation's notice of meeting only pursuant to <u>Section 3.2</u>.

Section 2.8 Conduct of Meetings. The chairperson of each annual and special meeting of stockholders shall be the Chairperson of the Board or, in the absence (or inability or refusal to act) of the Chairperson of the Board, the Chief Executive Officer (if they shall be a director) or, in the absence (or inability or refusal to act) of the Chief Executive Officer or if the Chief Executive Officer is not a director, the President (if they shall be a director) or, in the absence (or inability or refusal to act) of the President or if the President is not a director, such other person as shall be appointed by the Board. The Board may adopt such rules and regulations for the conduct of the meeting of stockholders as it shall deem appropriate. Except to the extent inconsistent with these Bylaws or such rules and regulations as adopted by the Board, the chairperson of any meeting of stockholders shall have the right and authority to convene and to adjourn the meeting, to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such chairperson, are appropriate for the proper conduct of the meeting. Such rules, regulations or procedures, whether adopted by the Board or prescribed by the chairperson of the meeting, may include, without limitation, the following: (a) the establishment of an agenda or order of business for the meeting; (b) rules and procedures for maintaining order at the meeting and the safety of those present; (c) limitations on attendance at or participation in the meeting to stockholders of record of the Corporation, their duly authorized and constituted proxies or such other persons as the chairperson of the meeting shall determine; (d) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (e) limitations on the time allotted to questions or comments by participants. Unless and to the extent determined by the Board or the chairperson of the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure. The secretary of each annual and special meeting of stockholders shall be the Secretary or, in the absence (or inability or refusal to act) of the Secretary, an Assistant Secretary so appointed to act by the chairperson of the meeting. In the absence (or inability or refusal to act) of the Secretary and all Assistant Secretaries, the chairperson of the meeting may appoint any person to act as secretary of the meeting.

Section 2.9 Consents in Lieu of Meeting. Any action required or permitted to be taken by the stockholders of the Corporation must be effected by a duly called annual or special meeting of such stockholders and may not be effected by written consent of the stockholders of the Corporation.

ARTICLE III DIRECTORS

Section 3.1 Powers. The business and affairs of the Corporation shall be managed by or under the direction of the Board, which may exercise all such powers of the Corporation and do all such lawful acts and things as are not by statute or by the Certificate of Incorporation or by these Bylaws required to be exercised or done by the stockholders. Directors need not be stockholders or residents of the State of Delaware. Unless otherwise provided by the Certificate of Incorporation, the Board shall consist of one or more members, the number thereof to be determined from time to time by resolution of the Board.

Section 3.2 Advance Notice for Nomination of Directors.

- (a) Only persons who are nominated in accordance with the following procedures shall be eligible for election as directors of the Corporation, except as may be otherwise provided by the terms of one or more series of Preferred Stock with respect to the rights of holders of one or more series of Preferred Stock to elect directors. Nominations of persons for election to the Board at any annual meeting of stockholders, or at any special meeting of stockholders called for the purpose of electing directors as set forth in the Corporation's notice of such special meeting, may be made (i) by or at the direction of the Board or (ii) by any stockholder of the Corporation (x) who is a stockholder of record entitled to vote in the election of directors on the date of the giving of the notice provided for in this Section 3.2 and on the record date for the determination of stockholders entitled to vote at such meeting and (y) who complies with the notice procedures set forth in this Section 3.2.
- (b) In addition to any other applicable requirements, for a nomination to be made by a stockholder, such stockholder must have given timely notice thereof in proper written form to the Secretary of the Corporation. To be timely, a stockholder's notice to the Secretary must be received by the Secretary at the principal executive offices of the Corporation (i) in the case of an annual meeting, not later than the close of business on the 90th day nor earlier than the close of business on the 120th day before the anniversary date of the immediately preceding annual meeting of stockholders; provided, however, that in the event that the annual meeting is more than 30 days before or more than 60 days after such anniversary date, notice by the stockholder to be timely must be so received not earlier than the close of business on the 120th day before the meeting and not later than the later of (x) the close of business on the 90th day before the meeting or (y) the close of business on the 10th day following the day on which public announcement of the date of the annual meeting was first made by the Corporation; and (ii) in the case of a special meeting of stockholders called for the purpose of electing directors, not later than the close of business on the 10th day following the day on which public announcement of the date of the special meeting is first made by the Corporation. In no event shall the public announcement of an adjournment or postponement of an annual meeting or special meeting commence a new time period (or extend any time period) for the giving of a stockholder's notice as described in this Section 3.2.
- (c) Notwithstanding anything in Section 3.2(b) to the contrary, in the event that the number of directors to be elected to the Board at an annual meeting is greater than the number of directors whose terms expire on the date of the annual meeting and there is no public announcement by the Corporation naming all of the nominees for the additional directors to be elected or specifying the size of the increased Board before the close of business on the 90th day prior to the anniversary date of the immediately preceding annual meeting of stockholders, a stockholder's notice required by this Section 3.2 shall also be considered timely, but only with respect to nominees for the additional directorships created by such increase that are to be filled by election at such annual meeting, if it shall be received by the Secretary at the principal executive offices of the Corporation not later than the close of business on the 10th day following the date on which such public announcement was first made by the Corporation.

(d) To be in proper written form, a stockholder's notice to the Secretary must set forth (i) as to each person whom the stockholder proposes to nominate for election as a director (each, a "Proposed Nominee") (A) the name, age, business address and residence address of such Proposed Nominee, (B) the principal occupation and employment of such Proposed Nominee, (C) a written questionnaire with respect to the background and qualification of such Proposed Nominee, completed by such Proposed Nominee in the form required by the Corporation (which form such Noticing Party shall request in writing from the Secretary prior to submitting notice and which the Secretary shall provide to such Noticing Party within 10 days after receiving such request), (D) a written representation and agreement completed by such Proposed Nominee in the form required by the Corporation (which form such Noticing Party shall request in writing from the Secretary prior to submitting notice and which the Secretary shall provide to such Noticing Party within 10 days after receiving such request) providing that such Proposed Nominee: (I) is not and will not become a party to any agreement, arrangement or understanding with, and has not given any commitment or assurance to, any person or entity as to how such Proposed Nominee, if elected as a director of the Corporation, will act or vote on any issue or question (a "Voting Commitment") that has not been disclosed to the Corporation or any Voting Commitment that could limit or interfere with such Proposed Nominee's ability to comply, if elected as a director of the Corporation, with such Proposed Nominee's fiduciary duties under applicable law; (II) is not and will not become a party to any agreement, arrangement or understanding with any person or entity other than the Corporation with respect to any direct or indirect compensation, reimbursement or indemnification in connection with service or action as a director or nominee that has not been disclosed to the Corporation; (III) will, if elected as a director of the Corporation, comply with all applicable rules of any securities exchanges upon which the Corporation's securities are listed, the Certificate of Incorporation, these Bylaws, all applicable publicly disclosed corporate governance, ethics, conflict of interest, confidentiality, stock ownership and trading policies and all other guidelines and policies of the Corporation generally applicable to directors (which other guidelines and policies will be provided to such Proposed Nominee within 5 business days after the Secretary receives any written request therefor from such Proposed Nominee), and all applicable fiduciary duties under state law; (IV) consents to being named as a nominee in the Corporation's proxy statement and form of proxy for the meeting; (V) intends to serve a full term as a director of the Corporation, if elected; and (VI) will provide facts, statements and other information in all communications with the Corporation and its stockholders that are or will be true and correct in all material respects and that do not and will not omit to state a material fact necessary in order to make the statements made, in light of the circumstances under which they are made, not misleading, (E) a description of all direct and indirect compensation and other material monetary agreements, arrangements and understandings, written or oral, during the past three years, and any other material relationships, between or among such Proposed Nominee, on the one hand, and such Noticing Party or any Stockholder Associated Person (as defined below), on the other hand, including, without limitation, all information that would be required to be disclosed pursuant to Item 404 promulgated under Regulation S-K as if such Noticing Party and any Stockholder Associated Person were the "registrant" for purposes of such rule and the Proposed Nominee were a director or executive officer of such registrant and (F) all other information relating to such Proposed Nominee or such Proposed Nominee's associates that would be required to be disclosed in a proxy statement or other filing made by such Noticing Party or any Stockholder Associated Person in connection with the solicitation of proxies for the election of directors in a contested election or otherwise required pursuant to the Proxy Rules and (ii) as to the Noticing Party (A) the name and address of such Noticing Party, each Proposed Nominee and each Stockholder Associated Person (including, as applicable, as they appear on the Corporation's books and records), (B) the class, series and number of shares of each class or series of capital stock (if any) of the Corporation that are, directly or indirectly, owned beneficially and/or of record by such Noticing Party, any Proposed Nominee or any Stockholder Associated Person and the date or dates such shares were acquired and the investment intent of such acquisition, (C) the name of each nominee holder for, and number of, any securities of the Corporation owned beneficially but not of record by such Noticing Party, any Proposed Nominee or any Stockholder Associated Person and any pledge by such Noticing Party, any Proposed Nominee or any Stockholder Associated Person with respect to any of such securities, (D) a complete and accurate description of all Derivative Instruments that have been entered into by, or on behalf of, such Noticing Party, any Proposed Nominee or any Stockholder Associated Person without regard to whether such agreement, arrangement or understanding is required to be reported on a Schedule 13D in accordance with the Exchange Act, (E) any substantial interest, direct or indirect (including any existing or prospective commercial, business or contractual relationship with the Corporation), by security holdings or otherwise, of such Noticing Party, any Proposed Nominee or any Stockholder Associated Person in the Corporation or any affiliate thereof, other than an interest arising from the ownership of Corporation securities where such Noticing Party, such Proposed Nominee or such Stockholder Associated Person receives no extra or special benefit not shared on a pro rata basis by all other holders of the same class or series, (F) a complete and accurate description of all agreements, arrangements or understandings, written or oral, (I) between or among such Noticing Party and any of the Stockholder Associated Persons or (II) between or among such Noticing Party or any Stockholder Associated Person and any other person or entity (naming each such person or entity) or any Proposed Nominee, including, without

limitation, (x) any proxy, contract, arrangement, understanding or relationship pursuant to which such Noticing Party or any Stockholder Associated Person, directly or indirectly, has a right to vote any security of the Corporation (other than any revocable proxy given in response to a solicitation made pursuant to, and in accordance with, Section 14(a) of the Exchange Act by way of a solicitation statement filed on Schedule 14A), (y) any understanding, written or oral, that such Noticing Party or any Stockholder Associated Person may have reached with any stockholder of the Corporation (including the name of such stockholder) with respect to how such stockholder will vote such stockholder's shares in the Corporation at any meeting of the Corporation's stockholders or take other action in support of any Proposed Nominee or other business, or other action to be taken, by such Noticing Party or any Stockholder Associated Person and (z) any other agreements that would be required to be disclosed by such Noticing Party, any Proposed Nominee, any Stockholder Associated Person or any other person or entity pursuant to Item 5 or Item 6 of a Schedule 13D pursuant to Section 13 of the Exchange Act (regardless of whether the requirement to file a Schedule 13D is applicable to such Noticing Party, such Proposed Nominee, such Stockholder Associated Person or such other person or entity), (G) any rights to dividends on the shares of the Corporation owned beneficially by such Noticing Party, any Proposed Nominee or any Stockholder Associated Person that are separated or separable from the underlying shares of the Corporation, (H) any proportionate interest in shares of the Corporation or Derivative Instruments held, directly or indirectly, by a general or limited partnership, limited liability company or similar entity in which such Noticing Party, any Proposed Nominee or any Stockholder Associated Person is (I) a general partner or, directly or indirectly, beneficially owns an interest in a general partner of such general or limited partnership or (II) the manager, managing member or, directly or indirectly, beneficially owns an interest in the manager or managing member of such limited liability company or similar entity, (I) any significant equity interests or any Derivative Instruments or Short Interests in any principal competitor of the Corporation held by such Noticing Party, any Proposed Nominee or any Stockholder Associated Person, (J) any direct or indirect interest of such Noticing Party, any Proposed Nominee or any Stockholder Associated Person in any contract with the Corporation, any affiliate of the Corporation or any principal competitor of the Corporation (including, without limitation, any employment agreement, collective bargaining agreement or consulting agreement), (K) a description of any material interest of such Noticing Party, any Proposed Nominee or any Stockholder Associated Person in the election of any Proposed Nominee, (L) a complete an accurate description of any performance-related fees (other than an asset-based fee) to which such Noticing Party, any Proposed Nominee or any Stockholder Associated Person may be entitled as a result of any increase or decrease in the value of the Corporation's securities or any Derivative Instruments, including, without limitation, any such interests held by members of such Noticing Party's, any Proposed Nominee's or any Stockholder Associated Person's immediate family sharing the same household, (M) the investment strategy or objective, if any, of such Noticing Party, any Proposed Nominee or any Stockholder Associated Person who is not an individual and a copy of the prospectus, offering memorandum or similar document, if any, provided to investors or potential investors in the Noticing Party or any Stockholder Associated Person and (N) all other information relating to such Noticing Party or any Stockholder Associated Person, or such Noticing Party's or any Stockholder Associated Person's associates, that would be required to be disclosed in a proxy statement or other filing in connection with the solicitation of proxies for the election of any Proposed Nominee in a contested election or otherwise pursuant to the Proxy Rules; provided, however, that the disclosures in the foregoing subclauses (A) through (N) shall not include any such disclosures with respect to the ordinary course business activities of any broker, dealer, commercial bank, trust company or other nominee who is a Noticing Party solely as a result of being the stockholder directed to prepare and submit the notice required by these Bylaws on behalf of a beneficial owner. To be in proper written form, any Noticing Party's notice with respect to any Proposed Nominee must set forth (i) a representation that such Noticing Party intends to appear in person or by proxy at the meeting to nominate any Proposed Nominees, as applicable, and an acknowledgment that, if such Noticing Party (or a Qualified Representative of such Noticing Party) does not appear to present such Proposed Nominees, as applicable, at such meeting, the Corporation need not present such Proposed Nominees for a vote at such meeting, notwithstanding that proxies in respect of such vote may have been received by the Corporation, (ii) a complete and accurate description of any pending or, to such Noticing Party's knowledge, threatened legal proceeding in which such Noticing Party, any Proposed Nominee or any Stockholder Associated Person is a party or participant involving the Corporation or, to such Noticing Party's knowledge, any officer, director, affiliate or associate of the Corporation and (iii) a representation from such Noticing Party as to whether such Noticing Party or any Stockholder Associated Person intends or is part of a group that intends (A) to deliver a proxy statement and/or form of proxy to a number of holders of the Corporation's voting shares reasonably believed by such Noticing Party to be sufficient to elect the Proposed Nominees, as applicable, or (B) engage in a solicitation (within the meaning of Exchange Act Rule 14a-1(l) with respect to the nomination and if so, the name of each participant (as defined in Item 4 of Schedule 14A under the Exchange Act) in such solicitation.

- (e) In addition to the information required above, the Corporation may require any Noticing Party to furnish such other information as the Corporation may reasonably require to determine the eligibility or suitability of a Proposed Nominee to serve as a director of the Corporation or that could be material to a reasonable stockholder's understanding of the independence, or lack thereof, of such Proposed Nominee, under the listing standards of each securities exchange upon which the Corporation's securities are listed, any applicable rules of the Securities and Exchange Commission, any publicly disclosed standards used by the Board in selecting nominees for election as a director and for determining and disclosing the independence of the Corporation's directors, including those applicable to a director's service on any of the committees of the Board, or the requirements of any other laws or regulations applicable to the Corporation. If requested by the Corporation, any supplemental information required under this paragraph shall be provided by a Noticing Party within 10 days after it has been requested by the Corporation.
- (f) A Noticing Party shall update such Noticing Party's notice provided under the foregoing provisions of this Section 3.2, if necessary, such that the information provided or required to be provided in such notice shall be true and correct (i) as of the record date for determining the stockholders entitled to receive notice of the meeting and (ii) as of the date that is 10 business days prior to the meeting (or any postponement, rescheduling or adjournment thereof), and such update shall be received by the Secretary at the principal executive offices of the Corporation (A) not later than the Close of Business five business days after the record date for determining the stockholders entitled to receive notice of such meeting (in the case of an update required to be made under clause (i)) and (B) not later than the Close of Business seven business days prior to the date for the meeting or, if practicable, any postponement, rescheduling or adjournment thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been postponed, rescheduled or adjourned) (in the case of an update required to be made pursuant to clause (ii)). For the avoidance of doubt, any information provided pursuant to this Section 3.2 (f) shall not be deemed to cure any deficiencies in a notice previously delivered pursuant to this Section 3.2 and shall not extend the time period for the delivery of notice pursuant to this Section 3.2. If a Noticing Party fails to provide such written update within such period, the information as to which such written update relates may be deemed not to have been provided in accordance with this Section 3.2. If any information submitted pursuant to this Section 3.2 by any Noticing Party proposing individuals to nominate for election or reelection as a director or business for consideration at a stockholder meeting shall be inaccurate in any respect, such information shall be deemed not to have been provided in accordance with this Section 3.2. Any such Noticing Party shall notify the Secretary in writing at the principal executive offices of the Corporation of any inaccuracy or change in any information submitted pursuant to this Section 3.2 within two business days after becoming aware of such inaccuracy or change. Upon written request of the Secretary on behalf of the Board (or a duly authorized committee thereof), any such Noticing Party shall provide, within seven business days after delivery of such request (or such other period as may be specified in such request), (i) written verification, reasonably satisfactory to the Board, any committee thereof or any authorized officer of the Corporation, to demonstrate the accuracy of any information submitted by such Noticing Party pursuant to this Section 3.2 and (ii) a written affirmation of any information submitted by such Noticing Party pursuant to this Section 3.2 as of an earlier date. If a Noticing Party fails to provide such written verification or affirmation within such period, the information as to which written verification or affirmation was requested may be deemed not to have been provided in accordance with this Section 3.2.
- (g) If the Board or the chairperson of the meeting of stockholders determines that any nomination was not made in accordance with the provisions of this Section 3.2 or that the information provided in a stockholder's notice does not satisfy the information requirements of this Section 3.2, then such nomination shall not be considered at the meeting in question. Notwithstanding the foregoing provisions of this Section 3.2, if the stockholder (or a qualified representative of the stockholder) does not appear at the meeting of stockholders of the Corporation to present the nomination, such nomination shall be disregarded, notwithstanding that proxies in respect of such nomination may have been received by the Corporation. No person shall be eligible for election as a director of the Corporation unless the person is nominated by a stockholder in accordance with the procedures set forth in this Section 3.2 or the person is nominated by the Board. The number of nominees a stockholder may nominate for election at a meeting may not exceed the number of directors to be elected at such meeting.
- (h) In addition to the provisions of this <u>Section 3.2</u>, a stockholder shall also comply with all of the applicable requirements of state law and the Exchange Act and the rules and regulations thereunder with respect to the matters set forth herein. Nothing in this <u>Section 3.2</u> shall be deemed to affect any rights of the holders of Preferred Stock to elect directors pursuant to the Certificate of Incorporation.

Section 3.3 Compensation. Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, the Board shall have the authority to fix the compensation of directors, including for service on a committee of the Board and may be paid either a fixed sum for attendance at each meeting of the Board or other compensation as a director. The directors may be reimbursed their expenses, if any, of attendance at each meeting of the Board. No such payment shall preclude any director from serving the Corporation in any other capacity and receiving compensation therefor. Members of committees of the Board may be allowed like compensation and reimbursement of expenses for service on the committee.

ARTICLE IV BOARD MEETINGS

Section 4.1 Annual Meetings. The Board shall meet as soon as practicable after the adjournment of each annual stockholders meeting at the place of the annual stockholders meeting unless the Board shall fix another time and place and give notice thereof in the manner required herein for special meetings of the Board. No notice to the directors shall be necessary to legally convene this meeting, except as provided in this Section 4.1.

Section 4.2 Regular Meetings. Regularly scheduled, periodic meetings of the Board may be held without notice at such times, dates and places (within or without the State of Delaware) as shall from time to time be determined by the Board.

Section 4.3 Special Meetings. Special meetings of the Board (a) may be called by the Chairperson of the Board, the Chief Executive Officer or the Secretary and (b) shall be called by the Chairperson of the Board, Chief Executive Officer or Secretary on the written request of any two directors then in office, or the sole director, as the case may be, and shall be held at such time, date and place (within or without the State of Delaware) as may be determined by the person calling the meeting or, if called upon the request of directors or the sole director, as specified in such written request. Notice of each special meeting of the Board shall be given, as provided in Section 11.3, to each director (i) at least 24 hours before the meeting if such notice is oral notice given personally or by telephone or written notice given by hand delivery or by means of a form of electronic transmission and delivery; (ii) at least two days before the meeting if such notice is sent by a nationally recognized overnight delivery service; and (iii) at least five days before the meeting if such notice is sent through the United States mail. If the Secretary shall fail or refuse to give such notice, then the notice may be given by the officer who called the meeting or the directors who requested the meeting. Any and all business that may be transacted at a regular meeting of the Board may be transacted at a special meeting. Except as may be otherwise expressly provided by applicable law, the Certificate of Incorporation, or these Bylaws, neither the business to be transacted at, nor the purpose of, any special meeting need be specified in the notice or waiver of notice of such meeting. A special meeting may be held at any time without notice if all of the directors are present or if those not present waive notice of the meeting in accordance with Section 11.4.

Section 4.4 Quorum; Required Vote. A majority of the Board shall constitute a quorum for the transaction of business at any meeting of the Board, and the act of a majority of the directors present at any meeting at which there is a quorum shall be the act of the Board, except as may be otherwise specifically provided by applicable law, the Certificate of Incorporation or these Bylaws. If a quorum shall not be present at any meeting, a majority of the directors present may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum is present.

Section 4.5 Consent In Lieu of Meeting Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, any action required or permitted to be taken at any meeting of the Board or any committee thereof may be taken without a meeting if all members of the Board or committee, as the case may be, consent thereto in writing or by electronic transmission, and the writing or writings or electronic transmission or transmissions (or paper reproductions thereof) are filed with the minutes of proceedings of the Board or committee. Such filing shall be in paper form if the minutes are maintained in paper form and shall be in electronic form if the minutes are maintained in electronic form.

Section 4.6 Organization. The chairperson of each meeting of the Board shall be the Chairperson of the Board or, in the absence (or inability or refusal to act) of the Chairperson of the Board, the Chief Executive Officer (if they shall be a director) or, in the absence (or inability or refusal to act) of the Chief Executive Officer is not a director, the President (if they shall be a director) or in the absence (or inability or refusal to act) of the President or if the President is not a director, a chairperson elected from the directors present. The Secretary shall act as secretary of all meetings of the Board. In the absence (or inability or refusal to act) of the Secretary, an Assistant Secretary shall perform the duties of the Secretary at such meeting. In the absence (or inability or refusal to act) of the Secretary and all Assistant Secretaries, the chairperson of the meeting may appoint any person to act as secretary of the meeting.

ARTICLE V COMMITTEES OF DIRECTORS

Section 5.1 Establishment. The Board may by resolution of the Board designate one or more committees, each committee to consist of one or more of the directors of the Corporation. Each committee shall keep regular minutes of its meetings and report the same to the Board when required by the resolution designating such committee. The Board shall have the power at any time to fill vacancies in, to change the membership of, or to dissolve any such committee.

Section 5.2 Available Powers. Any committee established pursuant to Section 5.1 hereof, to the extent permitted by applicable law and by resolution of the Board, shall have and may exercise all of the powers and authority of the Board in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers that may require it.

Section 5.3 Alternate Members. The Board may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of such committee. In the absence or disqualification of a member of the committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not they constitute a quorum, may unanimously appoint another member of the Board to act at the meeting in place of any such absent or disqualified member.

Section 5.4 Procedures. Unless the Board otherwise provides, the time, date, place, if any, and notice of meetings of a committee shall be determined by such committee. At meetings of a committee, a majority of the number of members of the committee (but not including any alternate member, unless such alternate member has replaced any absent or disqualified member at the time of, or in connection with, such meeting) shall constitute a quorum for the transaction of business. The act of a majority of the members present at any meeting at which a quorum is present shall be the act of the committee, except as otherwise specifically provided by applicable law, the Certificate of Incorporation, these Bylaws or the Board. If a quorum is not present at a meeting of a committee, the members present may adjourn the meeting from time to time, without notice other than an announcement at the meeting, until a quorum is present. Unless the Board otherwise provides and except as provided in these Bylaws, each committee designated by the Board may make, alter, amend and repeal rules for the conduct of its business. In the absence of such rules, each committee shall conduct its business in the same manner as the Board is authorized to conduct its business pursuant to Article IV of these Bylaws.

ARTICLE VI OFFICERS

Section 6.1 Officers. The officers of the Corporation elected by the Board shall be a Chief Executive Officer, a Chief Financial Officer, a Secretary and such other officers (including without limitation, a Chairperson of the Board, Presidents, Vice Presidents, Assistant Secretaries and a Treasurer) as the Board from time to time may determine. Officers elected by the Board shall each have such powers and duties as generally pertain to their respective offices, subject to the specific provisions of this Article VI. Such officers shall also have such powers and duties as from time to time may be conferred by the Board. The Chief Executive Officer or President may also appoint such other officers (including without limitation one or more Vice Presidents and Controllers) as may be necessary or desirable for the conduct of the business of the Corporation. Such other officers shall have such powers and duties and shall hold their offices for such terms as may be provided in these Bylaws or as may be prescribed by the Board or, if such officer has been appointed by the Chief Executive Officer or President, as may be prescribed by the appointing officer.

- (a) Chairperson of the Board. The Chairperson of the Board shall be chosen from among the directors and may be the Chief Executive Officer. Except as otherwise provided by law, the Certificate of Incorporation or these Bylaws, the Chairperson of the Board shall preside at all meetings of stockholders and of the Board. The Chairperson of the Board shall have such other powers and duties as may from time to time be assigned by the Board.
- (b) <u>Lead Independent Director</u>. The Board may include a Lead Director. The Lead Director shall be one of the directors who has been determined by the Board to be an "independent director" (any such director, an "*Independent Director*"). The Lead Director shall preside at all meetings of the Board at which the Chairperson of the Board is not present, preside over the executive sessions of the Independent Directors, serve as a liaison between the Chairperson of the Board and the Board and have such other responsibilities, and perform such duties, as may from time to time be assigned to them by the Board. The Lead Director shall be elected by a majority of the Independent Directors.
- (c) Chief Executive Officer. The Chief Executive Officer shall, subject to the control of the Board, have general supervision over the business of the Corporation and shall direct the affairs and policies of the Corporation. The Chief Executive Officer shall also perform such other duties and may exercise such other powers as may from time to time be assigned to such officer by these Bylaws or by the Board. In the absence (or inability or refusal to act) of the Chairperson of the Board, the Chief Executive Officer (if they shall be a director) shall preside when present at all meeting so of the stockholders and the Board. The position of Chairman and Chief Executive Officer may be held by the same person.
- (d) <u>President</u>. The President shall make recommendations to the Chief Executive Officer on all operational matters that would normally be reserved for the final executive responsibility of the Chief Executive Officer. In the absence (or inability or refusal to act) of the Chairperson of the Board and Chief Executive Officer, the President (if they shall be a director) shall preside when present at all meetings of the stockholders and the Board. The President shall also perform such duties and have such powers as shall be designated by the Board. The position of President and Chief Executive Officer may be held by the same person.
- (e) <u>Vice Presidents</u>. In the absence (or inability or refusal to act) of the President, the Vice President (or in the event there be more than one Vice President, the Vice Presidents in the order designated by the Board) shall perform the duties and have the powers of the President. Any one or more of the Vice Presidents may be given an additional designation of rank or function.

(f) Secretary.

- (i) The Secretary shall attend all meetings of the stockholders, the Board and (as required) committees of the Board and shall record the proceedings of such meetings in books to be kept for that purpose. The Secretary shall give, or cause to be given, notice of all meetings of the stockholders and special meetings of the Board and shall perform such other duties as may be prescribed by the Board, the Chairperson of the Board, Chief Executive Officer or President. The Secretary shall have custody of the corporate seal of the Corporation and the Secretary, or any Assistant Secretary, shall have authority to affix the same to any instrument requiring it, and when so affixed, it may be attested by their signature or by the signature of such Assistant Secretary. The Board may give general authority to any other officer to affix the seal of the Corporation and to attest the affixing thereof by their signature.
- (ii) The Secretary shall keep, or cause to be kept, at the principal executive office of the Corporation or at the office of the Corporation's transfer agent or registrar, if one has been appointed, a stock ledger, or duplicate stock ledger, showing the names of the stockholders and their addresses, the number and classes of shares held by each and, with respect to certificated shares, the number and date of certificates issued for the same and the number and date of certificates cancelled.
- (g) <u>Assistant Secretaries</u>. The Assistant Secretary or, if there be more than one, the Assistant Secretaries in the order determined by the Board shall, in the absence (or inability or refusal to act) of the Secretary, perform the duties and have the powers of the Secretary.

- (h) <u>Chief Financial Officer</u>. The Chief Financial Officer shall perform all duties commonly incident to that office (including, without limitation, the care and custody of the funds and securities of the Corporation, which from time to time may come into the Chief Financial Officer's hands and the deposit of the funds of the Corporation in such banks or trust companies as the Board, the Chief Executive Officer or the President may authorize).
- (i) <u>Treasurer</u>. The Treasurer shall, in the absence (or inability or refusal to act) of the Chief Financial Officer, perform the duties and exercise the powers of the Chief Financial Officer.

Section 6.2 Term of Office; Removal; Vacancies. The elected officers of the Corporation shall be appointed by the Board and shall hold office until their successors are duly elected and qualified or until their earlier death, resignation, retirement, disqualification, or removal from office. Any officer may be removed, with or without cause, at any time by the Board. Any officer appointed by the Chief Executive Officer or President may also be removed, with or without cause, by the Chief Executive Officer or President, as the case may be, unless the Board otherwise provides. Any vacancy occurring in any elected of the Corporation may be filled by the Board. Any vacancy occurring in any office appointed by the Chief Executive Officer or President may be filled by the Chief Executive Officer, or President, as the case may be, unless the Board then determines that such office shall thereupon be elected by the Board, in which case the Board shall elect such officer.

Section 6.3 Other Officers. The Board may delegate the power to appoint such other officers and agents, and may also remove such officers and agents or delegate the power to remove the same, as it shall from time to time deem necessary or desirable.

Section 6.4 Multiple Officeholders; Stockholder and Director Officers. Any number of offices may be held by the same person unless the Certificate of Incorporation or these Bylaws otherwise provide. Officers need not be stockholders or residents of the State of Delaware.

ARTICLE VII SHARES

Section 7.1 Certificated and Uncertificated Shares. The shares of the Corporation may be certificated or uncertificated, subject to the sole discretion of the Board and the requirements of the DGCL. Certificates for the shares of stock, if any, of the Corporation shall note conspicuously that the Corporation is a public benefit corporation formed pursuant to Subchapter XV of the DGCL. Any notice given by the Corporation pursuant to Section 151(f) of the DGCL upon the issuance or transfer of uncertificated shares shall state conspicuously that the Corporation is a public benefit corporation formed pursuant to Subchapter XV of the DGCL.

Section 7.2 Multiple Classes of Stock. If the Corporation shall be authorized to issue more than one class of stock or more than one series of any class, the Corporation shall (a) cause the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights to be set forth in full or summarized on the face or back of any certificate that the Corporation issues to represent shares of such class or series of stock or (b) in the case of uncertificated shares, within a reasonable time after the issuance or transfer of such shares, send to the registered owner thereof a written notice containing the information required to be set forth on certificates as specified in clause (a) above; provided, however, that, except as otherwise provided by applicable law, in lieu of the foregoing requirements, there may be set forth on the face or back of such certificate or, in the case of uncertificated shares, on such written notice a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences or rights.

Section 7.3 Signatures. Each certificate representing capital stock of the Corporation shall be signed by or in the name of the Corporation by (a) the Chairperson of the Board, the Chief Executive Officer, the President or a Vice President and (b) the Treasurer, an Assistant Treasurer, the Secretary or an Assistant Secretary of the Corporation. Any or all of the signatures on the certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, such certificate may be issued by the Corporation with the same effect as if such person were such officer, transfer agent or registrar on the date of issue.

Section 7.4 Consideration and Payment for Shares.

- (a) Subject to applicable law and the Certificate of Incorporation, shares of stock may be issued for such consideration, having in the case of shares with par value a value not less than the par value thereof, and to such persons, as determined from time to time by the Board. The consideration may consist of any tangible or intangible property or any benefit to the Corporation, including but not limited to cash, promissory notes, services performed, contracts for services to be performed or other securities or any combination thereof.
- (b) Subject to applicable law and the Certificate of Incorporation, shares may not be issued until the full amount of the consideration has been paid, unless upon the face or back of each certificate issued to represent any partly paid shares of capital stock or upon the books and records of the Corporation in the case of partly paid uncertificated shares, there shall have been set forth the total amount of the consideration to be paid therefor and the amount paid thereon up to and including the time said certificate representing certificated shares or said uncertificated shares are issued.

Section 7.5 Lost, Destroyed or Wrongfully Taken Certificates

- (a) If an owner of a certificate representing shares claims that such certificate has been lost, destroyed or wrongfully taken, the Corporation shall issue a new certificate representing such shares or such shares in uncertificated form if the owner: (i) requests such a new certificate before the Corporation has notice that the certificate representing such shares has been acquired by a protected purchaser; (ii) if requested by the Corporation, delivers to the Corporation a bond sufficient to indemnify the Corporation against any claim that may be made against the Corporation on account of the alleged loss, wrongful taking or destruction of such certificate or the issuance of such new certificate or uncertificated shares; and (iii) satisfies other reasonable requirements imposed by the Corporation.
- (b) If a certificate representing shares has been lost, apparently destroyed or wrongfully taken, and the owner fails to notify the Corporation of that fact within a reasonable time after the owner has notice of such loss, apparent destruction or wrongful taking and the Corporation registers a transfer of such shares before receiving notification, the owner shall be precluded from asserting against the Corporation any claim for registering such transfer or a claim to a new certificate representing such shares or such shares in uncertificated form.

Section 7.6 Transfer of Stock.

- (a) If a certificate representing shares of the Corporation is presented to the Corporation with an endorsement requesting the registration of transfer of such shares or an instruction is presented to the Corporation requesting the registration of transfer of uncertificated shares, the Corporation shall register the transfer as requested if:
- (i) in the case of certificated shares, the certificate representing such shares has been surrendered;
- (ii) (A) with respect to certificated shares, the endorsement is made by the person specified by the certificate as entitled to such shares; (B) with respect to uncertificated shares, an instruction is made by the registered owner of such uncertificated shares; or (C) with respect to certificated shares or uncertificated shares, the endorsement or instruction is made by any other appropriate person or by an agent who has actual authority to act on behalf of the appropriate person;
- (iii) the Corporation has received a guarantee of signature of the person signing such endorsement or instruction or such other reasonable assurance that the endorsement or instruction is genuine and authorized as the Corporation may request;
- (iv) the transfer does not violate any restriction on transfer imposed by the Corporation that is enforceable in accordance with Section 7.8(a); and

- (v) such other conditions for such transfer as shall be provided for under applicable law have been satisfied.
- (b) Whenever any transfer of shares shall be made for collateral security and not absolutely, the Corporation shall so record such fact in the entry of transfer if, when the certificate for such shares is presented to the Corporation for transfer or, if such shares are uncertificated, when the instruction for registration of transfer thereof is presented to the Corporation, both the transferor and transferer request the Corporation to do so.

Section 7.7 Registered Stockholders. Before due presentment for registration of transfer of a certificate representing shares of the Corporation or of an instruction requesting registration of transfer of uncertificated shares, the Corporation may treat the registered owner as the person exclusively entitled to inspect for any proper purpose the stock ledger and the other books and records of the Corporation, vote such shares, receive dividends or notifications with respect to such shares and otherwise exercise all the rights and powers of the owner of such shares, except that a person who is the beneficial owner of such shares (if held in a voting trust or by a nominee on behalf of such person) may, upon providing documentary evidence of beneficial ownership of such shares and satisfying such other conditions as are provided under applicable law, may also so inspect the books and records of the Corporation.

Section 7.8 Effect of the Corporation's Restriction on Transfer.

- (a) A written restriction on the transfer or registration of transfer of shares of the Corporation or on the amount of shares of the Corporation that may be owned by any person or group of persons, if permitted by the DGCL and noted conspicuously on the certificate representing such shares or, in the case of uncertificated shares, contained in a notice, offering circular or prospectus sent by the Corporation to the registered owner of such shares within a reasonable time prior to or after the issuance or transfer of such shares, may be enforced against the holder of such shares or any successor or transferee of the holder including an executor, administrator, trustee, guardian or other fiduciary entrusted with like responsibility for the person or estate of the holder.
- (b) A restriction imposed by the Corporation on the transfer or the registration of shares of the Corporation or on the amount of shares of the Corporation that may be owned by any person or group of persons, even if otherwise lawful, is ineffective against a person without actual knowledge of such restriction unless: (i) the shares are certificated and such restriction is noted conspicuously on the certificate; or (ii) the shares are uncertificated and such restriction was contained in a notice, offering circular or prospectus sent by the Corporation to the registered owner of such shares within a reasonable time prior to or after the issuance or transfer of such shares.

Section 7.9 Regulations. The Board shall have power and authority to make such additional rules and regulations, subject to any applicable requirement of law, as the Board may deem necessary and appropriate with respect to the issue, transfer or registration of transfer of shares of stock or certificates representing shares. The Board may appoint one or more transfer agents or registrars and may require for the validity thereof that certificates representing shares bear the signature of any transfer agent or registrar so appointed.

Section 7.10 Lock-Up.

- (a) Subject to Section 7.10(b), the Locked-up Holders may not Transfer any Lock-up Shares until the end of the Lock-up Period. The Lock-up Shares shall carry appropriate legends indicating the restrictions on Transfer imposed by this Section 7.10, including as required by Section 151(f) of the DGCL in respect to uncertificated stock.
- (b) Notwithstanding the provisions set forth in Section 7.10(a), the Locked-up Holders or their respective Permitted Transferees may Transfer the Lock-up Shares during the Lock-up Period (i) to any other Locked-up Holder, (ii) in the case of an individual, (A) by gift to any person related to the Company Stockholder by blood, marriage, or domestic relationship ("immediate family"), a charitable organization or a trust or other entity formed for estate planning purposes for the benefit of an immediate family member, (B) by will, intestacy or by virtue of laws of descent and distribution upon the death of such individual, or (C) pursuant to a qualified domestic relations order, (iii) in the case of a corporation, limited liability company, partnership, trust or other entity, to any stockholder, member, partner or trust beneficiary as part of a distribution, or to any corporation, partnership or other entity that is an affiliate (as defined in Rule 405 of the Securities Act of 1933, as amended) of the Locked-up Holder, (iv) in the

event of a liquidation, merger, stock exchange or other similar transaction which results in all of the Corporation's stockholders having the right to exchange their shares of common stock for cash, securities or other property, or (v) to the Corporation in connection with the "net" or "cashless" exercise of options or other rights to purchase shares of common stock held by such Locked-up Holder in satisfaction of any tax withholding or exercise price obligations through cashless surrender or otherwise, *provided* that any shares of common stock issued upon exercise of such option or other rights shall remain subject to the terms of this Section 7.10; *provided*, *however*, that, in the case of clauses (ii) and (iii), such transferees shall enter into a written agreement with the Corporation agreeing to be bound by the transfer restrictions set forth herein; and *provided*, *further*, with respect to clauses (ii) and (iii), that any such transfer shall not involve a disposition for value.

- (c) For purposes of this Section 7.10:
- (i) the term "Lock-up Period" means the period beginning on the closing date of the VGAC Transaction and ending on the date that is at least 150 days after the closing date of the VGAC Transaction, provided that such date is during an open trading window, and if such date is not during an open trading window, the date of the opening of the first subsequent trading window (for the avoidance of doubt, this provision in these Bylaws shall supersede and replace any other provisions in other agreements concerning these matters);
- (ii) the term "Lock-up Shares" means the shares of common stock received by the stockholders of the Corporation on or after the date of the adoption of these Bylaws as consideration in the VGAC Transaction (other than shares of common stock received as consideration in the VGAC Transaction for shares of common stock of Grove Collaborative, Inc. issued pursuant to the Backstop Subscription Agreement (as defined in the Merger Agreement)); provided, that, for clarity, shares of common stock issued in connection with the Domestication (as defined in the Merger Agreement), the Backstop Subscription Agreement (as defined in the Merger Agreement) or the PIPE Financing (as defined in the Merger Agreement) shall not constitute Lock-up Shares;
- (iii) the term "Locked-up Holders" means the holders of Lock-up Shares;
- (iv) the term "Merger Agreement" means that certain Amended and Restated Agreement and Plan of Merger, dated as of March 31, 2022, by and among Virgin Group Acquisition Corp. II, a Cayman Islands exempted company, Treehouse Merger Sub, Inc., a Delaware corporation, Treehouse Merger Sub II, LLC, a Delaware limited liability company, and Grove Collaborative, Inc., a Delaware public benefit corporation, as amended from time to time.
- (v) the term "Transfer" means the (A) sale or assignment of, offer to sell, contract or agreement to sell, gift, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Exchange Act with respect to, any security, (B) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (C) public announcement of any intention to effect any transaction specified in clause (A) or (B); and
- (vi) the term "VGAC Transaction" means the transactions contemplated by the Merger Agreement, including, among other things, (A) the merger of Treehouse Merger Sub, Inc., a Delaware corporation, with and into Grove Collaborative, Inc., a Delaware public benefit corporation, with Grove Collaborative, Inc. surviving as a wholly owned subsidiary of the Corporation and (B) immediately following such merger, and as part of the same overall transaction, the merger of Grove Collaborative, Inc. with and into VGAC II Merger Sub II, LLC, a Delaware limited liability company, with VGAC II Merger Sub II, LLC surviving as a wholly owned subsidiary of the Corporation, each pursuant to and as contemplated by the Merger Agreement.

ARTICLE VIII INDEMNIFICATION

Section 8.1 Right to Indemnification. To the fullest extent permitted by applicable law, as the same exists or may hereafter be amended, the Corporation shall indemnify and hold harmless each person who was or is made a party or is threatened to be made a party to or is otherwise involved in any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (hereinafter a "proceeding"), by reason of the fact that they are or were a director or officer of the Corporation or, while a director or officer of the Corporation, are or were serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, other enterprise or nonprofit entity, including service with respect to an employee benefit plan (hereinafter an "Indemnitee"), whether the basis of such proceeding is alleged action in an official capacity as a director, officer, employee or agent, or in any other capacity while serving as a director, officer, employee or agent, against all liability and loss suffered and expenses (including, without limitation, attorneys' fees, judgments, fines, ERISA excise taxes and penalties and amounts paid in settlement) reasonably incurred by such Indemnitee in connection with such proceeding; provided, however, that, except as provided in Section 8.3 with respect to proceedings to enforce rights to indemnification, the Corporation shall indemnite in connection with a proceeding (or part thereof) initiated by such Indemnitee only if such proceeding (or part thereof) was authorized by the Board.

Section 8.2 Right to Advancement of Expenses. In addition to the right to indemnification conferred in Section 8.1 an Indemnitee shall also have the right to be paid by the Corporation to the fullest extent not prohibited by applicable law the expenses (including, without limitation, attorneys' fees) incurred in defending or otherwise participating in any such proceeding in advance of its final disposition (hereinafter an "advancement of expenses"); provided, however, that, if the DGCL requires, an advancement of expenses incurred by an Indemnitee in their capacity as a director or officer of the Corporation (and not in any other capacity in which service was or is rendered by such Indemnitee, including, without limitation, service to an employee benefit plan) shall be made only upon the Corporation's receipt of an undertaking (hereinafter an "undertaking"), by or on behalf of such Indemnitee, to repay all amounts so advanced if it shall ultimately be determined that such Indemnitee is not entitled to be indemnified under this Article VIII or otherwise

Section 8.3 Right of Indemnitee to Bring Suit. If a claim under Section 8.1 or Section 8.2 is not paid in full by the Corporation within 60 days after a written claim therefor has been received by the Corporation, except in the case of a claim for an advancement of expenses, in which case the applicable period shall be 20 days, the Indemnitee may at any time thereafter bring suit against the Corporation to recover the unpaid amount of the claim. If successful in whole or in part in any such suit, or in a suit brought by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, the Indemnitee shall also be entitled to be paid the expense of prosecuting or defending such suit. In (a) any suit brought by the Indemnitee to enforce a right to indemnification hereunder (but not in a suit brought by an Indemnitee to enforce a right to an advancement of expenses) it shall be a defense that, and (b) in any suit brought by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, the Corporation shall be entitled to recover such expenses upon a final judicial decision from which there is no further right to appeal that, the Indemnitee has not met any applicable standard for indemnification set forth in the DGCL. Neither the failure of the Corporation (including its directors who are not parties to such action, a committee of such directors, independent legal counsel, or its stockholders) to have made a determination prior to the commencement of such suit that indemnification of the Indemnitee is proper in the circumstances because the Indemnitee has met the applicable standard of conduct set forth in the DGCL, nor an actual determination by the Corporation (including a determination by its directors who are not parties to such action, a committee of such directors, independent legal counsel, or its stockholders) that the Indemnitee has not met such applicable standard of conduct, shall create a presumption that the Indemnitee has not met the applicable standard of conduct or, in the case of such a suit brought by the Indemnitee, shall be a defense to such suit. In any suit brought by the Indemnitee to enforce a right to indemnification or to an advancement of expenses hereunder, or by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, the burden of proving that the Indemnitee is not entitled to be indemnified, or to such advancement of expenses, under this Article VIII or otherwise shall be on the Corporation.

Section 8.4 Non-Exclusivity of Rights. The rights provided to any Indemnitee pursuant to this <u>Article VIII</u> shall not be exclusive of any other right, which such Indemnitee may have or hereafter acquire under applicable law, the Certificate of Incorporation, these Bylaws, an agreement, a vote of stockholders or disinterested directors, or otherwise.

Section 8.5 Insurance. The Corporation may maintain insurance, at its expense, to protect itself and/or any director, officer, employee or agent of the Corporation or another corporation, partnership, joint venture, trust or other enterprise against any expense, liability or loss, whether or not the Corporation would have the power to indemnify such person against such expense, liability or loss under the DGCL.

Section 8.6 Indemnification of Other Persons. This Article VIII shall not limit the right of the Corporation to the extent and in the manner authorized or permitted by law to indemnify and to advance expenses to persons other than Indemnitees. Without limiting the foregoing, the Corporation may, to the extent authorized from time to time by the Board, grant rights to indemnification and to the advancement of expenses to any employee or agent of the Corporation and to any other person who is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust or other enterprise, including service with respect to an employee benefit plan, to the fullest extent of the provisions of this Article VIII with respect to the indemnification and advancement of expenses of Indemnitees under this Article VIII.

Section 8.7 Amendments. Any repeal or amendment of this Article VIII by the Board or the stockholders of the Corporation or by changes in applicable law, or the adoption of any other provision of these Bylaws inconsistent with this Article VIII, will, to the extent permitted by applicable law, shall be prospective only (except to the extent such amendment or change in applicable law permits the Corporation to provide broader indemnification rights on a retroactive basis than permitted prior thereto), and will not in any way diminish or adversely affect any right or protection existing hereunder in respect of any act or omission occurring prior to such repeal or amendment or adoption of such inconsistent provision; provided, however, that amendments or repeals of this Article VIII shall require the affirmative vote of the stockholders holding at least two-thirds of the voting power of all outstanding shares of capital stock of the Corporation.

Section 8.8 Certain Definitions. For purposes of this Article VIII, (a) references to "other enterprise" shall include any employee benefit plan; (b) references to "fines" shall include any excise taxes assessed on a person with respect to an employee benefit plan; (c) references to "serving at the request of the Corporation" shall include any service that imposes duties on, or involves services by, a person with respect to any employee benefit plan, its participants, or beneficiaries; and (d) a person who acted in good faith and in a manner such person reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interest of the Corporation" for purposes of Section 145 of the DGCL.

Section 8.9 Contract Rights. The rights provided to Indemnitees pursuant to this <u>Article VIII</u> shall be contract rights and such rights shall continue as to an Indemnitee who has ceased to be a director, officer, agent or employee and shall inure to the benefit of the Indemnitee's heirs, executors and administrators

Section 8.10 Severability. If any provision or provisions of this Article VIII shall be held to be invalid, illegal or unenforceable for any reason whatsoever: (a) the validity, legality and enforceability of the remaining provisions of this Article VIII shall not in any way be affected or impaired thereby; and (b) to the fullest extent possible, the provisions of this Article VIII (including, without limitation, each such portion of this Article VIII containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to give effect to the intent manifested by the provision held invalid, illegal or unenforceable.

ARTICLE IX PUBLIC BENEFIT CORPORATION

Section 9.1 Required Statement in Stockholder Meeting Notice The Corporation shall include in every notice of a meeting of stockholders a statement that it is a public benefit corporation formed pursuant to Subchapter XV of the DGCL.

Section 9.2 Periodic Statements. The Corporation shall no less than biennially provide the stockholders with a statement as to the Corporation's promotion of the public benefit or public benefits identified in the Certificate of Incorporation and of the best interests of those materially affected by the Corporation's conduct. The statement shall include:

(a) the objectives the Board has established to promote such public benefit or public benefits and interests;

- (b) the standards the Board has adopted to measure the Corporation's progress in promoting such public benefit or public benefits and interests;
- (c) objective factual information based on those standards regarding the Corporation's success in meeting the objectives for promoting such public benefit or public benefits and interests; and
- (d) an assessment of the Corporation's success in meeting the objectives and promoting such public benefit or public benefits and interests.

ARTICLE X EMERGENCY BYLAWS

Section 10.1 Emergency Bylaws. This Article X shall be operative during any emergency, disaster or catastrophe, as referred to in Section 110 of the DGCL or other similar emergency condition (including, without limitation, a pandemic), as a result of which a quorum of the Board or a committee thereof cannot readily be convened for action (each, an "Emergency"), notwithstanding any different or conflicting provision of the preceding Sections of these Bylaws or in the Certificate of Incorporation. To the extent not inconsistent with the provisions of this Article X, the preceding Sections of these Bylaws and the provisions of the Certificate of Incorporation shall remain in effect during such Emergency, and upon termination of such Emergency, the provisions of this Article X shall cease to be operative unless and until another Emergency shall occur.

Section 10.2 Meetings; Notice. During any Emergency, a meeting of the Board or any committee thereof may be called by any member of the Board or such committee or the Chairperson of the Board, the Chief Executive Officer, the President or the Secretary of the Corporation. Notice of the place, date and time of the meeting shall be given by any available means of communication by the person calling the meeting to such of the directors or committee members and Designated Officers (as defined below) as, in the judgment of the person calling the meeting, it may be feasible to reach. Such notice shall be given at such time in advance of the meeting as, in the judgment of the person calling the meeting, circumstances permit.

Section 10.3 Quorum. At any meeting of the Board called in accordance with Section 10.2 above, the presence or participation of one director shall constitute a quorum for the transaction of business, and at any meeting of any committee of the Board called in accordance with Section 10.2 above, the presence or participation of one committee member shall constitute a quorum for the transaction of business. In the event that no directors are able to attend a meeting of the Board or any committee thereof, then the Designated Officers in attendance shall serve as directors, or committee members, as the case may be, for the meeting, without any additional quorum requirement and will have full powers to act as directors, or committee members, as the case may be, of the Corporation.

Section 10.4 Liability. No officer, director or employee of the Corporation acting in accordance with the provisions of this Article X shall be liable except for willful misconduct.

Section 10.5 Amendments. At any meeting called in accordance with Section 10.2 above, the Board, or any committee thereof, as the case may be, may modify, amend or add to the provisions of this Article X as it deems it to be in the best interests of the Corporation so as to make any provision that may be practical or necessary for the circumstances of the Emergency.

Section 10.6 Repeal or Change. The provisions of this <u>Article X</u> shall be subject to repeal or change by further action of the Board or by action of the stockholders, but no such repeal or change shall modify the provisions of <u>Section 10.4</u> above with regard to action taken prior to the time of such repeal or change.

Section 10.7 Definitions. For purposes of this Article X, the term "Designated Officer" means an officer identified on a numbered list of officers of the Corporation who shall be deemed to be, in the order in which they appear on the list up until a quorum is obtained, directors of the Corporation, or members of a committee of the Board, as the case may be, for purposes of obtaining a quorum during an Emergency, if a quorum of directors or committee members, as the case may be, cannot otherwise be obtained during such Emergency, which officers have been designated by the Board from time to time but in any event prior to such time or times as an Emergency may have occurred.

ARTICLE XI MISCELLANEOUS

Section 11.1 Place of Meetings. If the place of any meeting of stockholders, the Board or committee of the Board for which notice is required under these Bylaws is not designated in the notice of such meeting, such meeting shall be held at the principal business office of the Corporation; *provided, however*, if the Board has, in its sole discretion, determined that a meeting shall not be held at any place, but instead shall be held by means of remote communication pursuant to Section 11.5 hereof, then such meeting shall not be held at any place.

Section 11.2 Fixing Record Dates.

(a) In order that the Corporation may determine the stockholders entitled to notice of any meeting of stockholders or any adjournment thereof, the Board may fix a record date, which shall not precede the date upon which the resolution fixing the record date is adopted by the Board, and which record date shall not be more than 60 nor less than 10 days before the date of such meeting. If the Board so fixes a date, such date shall also be the record date for determining the stockholders entitled to vote at such meeting unless the Board determines, at the time it fixes such record date, that a later date on or before the date of the meeting shall be the date for making such determination. If no record date is fixed by the Board, the record date for determining stockholders entitled to notice of and to vote at a meeting of stockholders shall be at the close of business on the business day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the business day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board may fix a new record date for the adjourned meeting, and in such case shall also fix as the record date for stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote in accordance with the foregoing provisions of this Section 11.2(a) at the adjourned meeting.

(b) In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights or the stockholders entitled to exercise any rights in respect of any change, conversion or exchange of stock, or for the purpose of any other lawful action, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than 60 days prior to such action. If no record date is fixed, the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the Board adopts the resolution relating thereto.

Section 11.3 Means of Giving Notice.

(a) Notice to Directors. Whenever under applicable law, the Certificate of Incorporation or these Bylaws notice is required to be given to any director, such notice shall be given either (i) in writing and sent by mail, or by a nationally recognized delivery service, (ii) by means of facsimile telecommunication or other form of electronic transmission, or (iii) by oral notice given personally or by telephone. A notice to a director will be deemed given as follows: (A) if given by hand delivery, orally, or by telephone, when actually received by the director, (B) if sent through the United States mail, when deposited in the United States mail not less than five days before the date of the meeting, with postage and fees thereon prepaid, addressed to the director at the director's address appearing on the records of the Corporation, (C) if sent for next day delivery by a nationally recognized overnight delivery service, when deposited with such service not less than two days before the date of the meeting, with fees thereon prepaid, addressed to the director at the director's address appearing on the records of the Corporation, (D) if sent by facsimile telecommunication, when sent to the facsimile transmission number for such director appearing on the records of the Corporation, (E) if sent by electronic mail, when sent to the electronic mail address for such director appearing on the records of the Corporation, or (F) if sent by any other form of electronic transmission, when sent to the address, location or number (as applicable) for such director appearing on the records of the Corporation.

- (b) Notice to Stockholders. Whenever under applicable law, the Certificate of Incorporation or these Bylaws notice is required to be given to any stockholder, such notice may be given (i) in writing and sent either by hand delivery, through the United States mail, or by a nationally recognized overnight delivery service for next day delivery, or (ii) by means of a form of electronic transmission consented to by the stockholder, to the extent permitted by, and subject to the conditions set forth in Section 232 of the DGCL. A notice to a stockholder shall be deemed given as follows: (A) if given by hand delivery, when actually received by the stockholder, (B) if sent through the United States mail, when deposited in the United States mail, with postage and fees thereon prepaid, addressed to the stockholder at the stockholder's address appearing on the stock ledger of the Corporation, (C) if sent for next day delivery by a nationally recognized overnight delivery service, when deposited with such service, with fees thereon prepaid, addressed to the stockholder at the stockholder's address appearing on the stock ledger of the Corporation, and (D) if given by a form of electronic transmission consented to by the stockholder to whom the notice is given and otherwise meeting the requirements set forth above, (E) if by facsimile transmission, when directed to a number at which the stockholder has consented to receive notice, (F) if by electronic mail, when directed to an electronic mail address at which the stockholder has consented to receive notice, (G) if by a posting on an electronic network together with separate notice to the stockholder of such specified posting, upon the later of (1) such posting and (2) the giving of such separate notice, and (H) if by any other form of electronic transmission, when directed to the stockholder. A stockholder may revoke such stockholder's consent to receiving notice by means of electronic communication by giving written notice of such revocation to the Corporation. Any such consent shall be deemed revoked if (x) the Corporation is unable to deliver by electronic transmission two consecutive notices given by the Corporation in accordance with such consent and (y) such inability becomes known to the Secretary or an Assistant Secretary or to the Corporation's transfer agent, or other person responsible for the giving of notice; provided, however, the inadvertent failure to treat such inability as a revocation shall not invalidate any meeting or other action.
- (c) <u>Electronic Transmission</u>. "*Electronic transmission*" means any form of communication, not directly involving the physical transmission of paper, that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process, including but not limited to transmission by telex, facsimile telecommunication, electronic mail, telegram and cablegram.
- (d) Notice to Stockholders Sharing Same Address. Without limiting the manner by which notice otherwise may be given effectively by the Corporation to stockholders, any notice to stockholders given by the Corporation under any provision of the DGCL, the Certificate of Incorporation or these Bylaws shall be effective if given by a single written notice to stockholders who share an address if consented to by the stockholders at that address to whom such notice is given. A stockholder may revoke such stockholder's consent by delivering written notice of such revocation to the Corporation. Any stockholder who fails to object in writing to the Corporation within 60 days of having been given written notice by the Corporation of its intention to send such a single written notice shall be deemed to have consented to receiving such single written notice.
- (e) Exceptions to Notice Requirements. Whenever notice is required to be given, under the DGCL, the Certificate of Incorporation or these Bylaws, to any person with whom communication is unlawful, the giving of such notice to such person shall not be required and there shall be no duty to apply to any governmental authority or agency for a license or permit to give such notice to such person. Any action or meeting that shall be taken or held without notice to any such person with whom communication is unlawful shall have the same force and effect as if such notice had been duly given. In the event that the action taken by the Corporation is such as to require the filing of a certificate with the Secretary of State of Delaware, the certificate shall state, if such is the fact and if notice is required, that notice was given to all persons entitled to receive notice except such persons with whom communication is unlawful. Whenever notice is required to be given by the Corporation, under any provision of the DGCL, the Certificate of Incorporation or these Bylaws, to any stockholder to whom (i) notice of two consecutive annual meetings of stockholders and all notices of stockholder meetings or of the taking of action by written consent of stockholders without a meeting to such stockholder during the period between such two consecutive annual meetings, or (ii) all, and at least two payments (if sent by first-class mail) of dividends or interest on securities during a 12-month period, have been mailed addressed to such stockholder at such stockholder's address as shown on the records of the Corporation and have been returned undeliverable, the giving of such notice to such stockholder shall not be required. Any action or meeting that shall be taken or held without notice to such stockholder shall have the same force and effect as if such notice had been duly given. If any such stockholder shall deliver to the Corporation a written notice setting forth such stockholder's then current address, the requirement that notice be given to such stockholder shall be reinstated. In the event that the action taken by the Corporation is such as to require the filing of a certificate with the Secretary of State of Delaware, the certificate need not state that notice was not given to persons to whom notice was not required to be given pursuant to Section 230(b) of the DGCL. The exception in subsection (i) of the first sentence of this Section 11.3(e) to the requirement that notice be given shall not be applicable to any notice returned as undeliverable if the notice was given by electronic transmission.

Section 11.4 Waiver of Notice. Whenever any notice is required to be given under the DGCL, the Certificate of Incorporation, or these Bylaws, a written waiver of such notice, signed by the person or persons entitled to said notice, or a waiver by electronic transmission by the person entitled to said notice, whether before or after the time stated therein, shall be deemed equivalent to such required notice. All such waivers shall be kept with the books of the Corporation. Attendance at a meeting shall constitute a waiver of notice of such meeting, except where a person attends for the express purpose of objecting to the transaction of any business on the ground that the meeting was not lawfully called or convened.

Section 11.5 Meeting Attendance via Remote Communication Equipment.

- (a) <u>Stockholder Meetings</u>. If authorized by the Board in its sole discretion, and subject to such guidelines and procedures as the Board may adopt, stockholders entitled to vote at such meeting and proxy holders not physically present at a meeting of stockholders may, by means of remote communication:
- (i) participate in a meeting of stockholders; and
- (ii) be deemed present in person and vote at a meeting of stockholders, whether such meeting is to be held at a designated place or solely by means of remote communication, provided that (A) the Corporation shall implement reasonable measures to verify that each person deemed present and permitted to vote at the meeting by means of remote communication is a stockholder or proxy holder, (B) the Corporation shall implement reasonable measures to provide such stockholders and proxy holders a reasonable opportunity to participate in the meeting and to vote on matters submitted to the stockholders, including an opportunity to read or hear the proceedings of the meeting substantially concurrently with such proceedings, and (C) if any stockholder or proxy holder votes or takes other action at the meeting by means of remote communication, a record of such votes or other action shall be maintained by the Corporation.
- (b) <u>Board Meetings</u>. Unless otherwise restricted by applicable law, the Certificate of Incorporation or these Bylaws, members of the Board or any committee thereof may participate in a meeting of the Board or any committee thereof by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other. Such participation in a meeting shall constitute presence in person at the meeting, except where a person participates in the meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting was not lawfully called or convened.
- Section 11.6 Dividends. The Board may from time to time declare, and the Corporation may pay, dividends (payable in cash, property or shares of the Corporation's capital stock) on the Corporation's outstanding shares of capital stock, subject to applicable law and the Certificate of Incorporation.

Section 11.7 Reserves. The Board may set apart out of the funds of the Corporation available for dividends a reserve or reserves for any proper purpose and may abolish any such reserve.

Section 11.8 Contracts and Negotiable Instruments. Except as otherwise provided by applicable law, the Certificate of Incorporation or these Bylaws, any contract, bond, deed, lease, mortgage or other instrument may be executed and delivered in the name and on behalf of the Corporation by such officer or officers or other employee or employees of the Corporation as the Board may from time to time authorize. Such authority may be general or confined to specific instances as the Board may determine. The Chairperson of the Board, the Chief Executive Officer, the President, the Chief Financial Officer, the Treasurer or any Vice President may execute and deliver any contract, bond, deed, lease, mortgage or other instrument in the name and on behalf of the Corporation. Subject to any restrictions imposed by the Board, the Chairperson of the Board, the Chief Executive Officer, the President, the Chief Financial Officer, the Treasurer or any Vice President may delegate powers to execute and deliver any contract, bond, deed, lease, mortgage or other instrument in the name and on behalf of the Corporation to other officers or employees of the Corporation under such person's supervision and authority, it being understood, however, that any such delegation of power shall not relieve such officer of responsibility with respect to the exercise of such delegated power.

Section 11.9 Fiscal Year. The fiscal year of the Corporation shall be fixed by the Board.

Section 11.10 Seal. The Board may adopt a corporate seal, which shall be in such form as the Board determines. The seal may be used by causing it or a facsimile thereof to be impressed, affixed or otherwise reproduced.

Section 11.11 Books and Records. The books and records of the Corporation may be kept within or outside the State of Delaware at such place or places as may from time to time be designated by the Board.

Section 11.12 Resignation. Any director, committee member or officer may resign by giving notice thereof in writing or by electronic transmission to the Chairperson of the Board, the Chief Executive Officer, the President or the Secretary. The resignation shall take effect at the time it is delivered unless the resignation specifies a later effective date or an effective date determined upon the happening of an event or events. Unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.

Section 11.13 Surety Bonds. Such officers, employees and agents of the Corporation (if any) as the Chairperson of the Board, the Chief Executive Officer, the President or the Board may direct, from time to time, shall be bonded for the faithful performance of their duties and for the restoration to the Corporation, in case of their death, resignation, retirement, disqualification or removal from office, of all books, papers, vouchers, money and other property of whatever kind in their possession or under their control belonging to the Corporation, in such amounts and by such surety companies as the Chairperson of the Board, the Chief Executive Officer, the President or the Board may determine. The premiums on such bonds shall be paid by the Corporation and the bonds so furnished shall be in the custody of the Secretary.

Section 11.14 Securities of Other Corporations. Powers of attorney, proxies, waivers of notice of meeting, consents in writing and other instruments relating to securities owned by the Corporation may be executed in the name of and on behalf of the Corporation by the Chairperson of the Board, the Chief Executive Officer, the President, or any officers authorized by the Board. Any such officer, may, in the name of and on behalf of the Corporation, take all such action as any such officer may deem advisable to vote in person or by proxy at any meeting of security holders of any corporation in which the Corporation may own securities, or to consent in writing, in the name of the Corporation as such holder, to any action by such corporation, and at any such meeting or with respect to any such consent shall possess and may exercise any and all rights and power incident to the ownership of such securities and which, as the owner thereof, the Corporation might have exercised and possessed. The Board may from time to time confer like powers upon any other person or persons.

Section 11.15 Amendments. The Board shall have the power to adopt, amend, alter or repeal the Bylaws. The affirmative vote of a majority of the Board shall be required to adopt, amend, alter or repeal the Bylaws. The Bylaws also may be adopted, amended, altered or repealed by the stockholders; provided, however, that in addition to any vote of the holders of any class or series of capital stock of the Corporation required by applicable law or the Certificate of Incorporation, the affirmative vote of the holders of at least a majority of the voting (except as otherwise provided in Section 8.7) power of all outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required for the stockholders to adopt, amend, alter or repeal the Bylaws.

<u>Delaware</u> The First State

Page 1

I, JEFFREY W. BULLOCK, SECRETARY OF STATE OF THE STATE OF DELAWARE, DO HEREBY CERTIFY THE ATTACHED IS A TRUE AND CORRECT COPY OF THE CERTIFICATE OF DOMESTICATION OF "VIRGIN GROUP ACQUISITION CORP. II", FILED IN THIS OFFICE THE FIFTEENTH DAY OF JUNE, A.D. 2022, AT 6:16 O' CLOCK P.M.

AFYS OF CASE O

/s/ Jeffrey W. Bullock
Jeffrey W. Bullock, Secretary of State

6860351 8100D SR# 20222739355

You may verify this certificate online at corp.delaware.gov/authver.shtml

Date: 06-15-22

Authentication: 203691488

CERTIFICATE OF DOMESTICATION OF VIRGIN GROUP ACQUISITION CORP. II AS A PUBLIC BENEFIT CORPORATION

Pursuant to Section 388 of the General Corporation Law of the State of Delaware

Virgin Group Acquisition Corp. II, a Cayman Islands exempted company limited by its shares (the "Corporation"), which intends to domesticate as a Delaware public benefit corporation pursuant to this Certificate of Domestication, does hereby certify to the following facts relating to the domestication of the Corporation in the State of Delaware:

- 1. The Corporation was originally incorporated on the 13th day of January, 2021 under the laws of the Cayman Islands.
- 2. The name of the Corporation immediately prior to the filing of this Certificate of Domestication is Virgin Group Acquisition Corp. II.
- 3. The name of the Corporation as set forth in the Certificate of Incorporation is Grove Collaborative Holdings, Inc.
- 4. The jurisdiction that constituted the seat, siege social or principal place of business or central administration of the Corporation immediately prior to the filing of this Certificate of Domestication is the Cayman Islands.
- 5. The domestication has been approved in the manner provided for by the document, instrument, agreement or other writing, as the case may be, governing the internal affairs of the Corporation and the conduct of its business or by applicable non-Delaware law, as appropriate.
- 6. The specific public benefit purpose of the Corporation is to promote development, promotion and distribution of consumer products as a positive force for human and environmental health globally.

[Signature Page Follows]

State of Delaware Secretary of State Division of Corporations Delivered 06:16 PM 06/15/2022 FILED 06:16 PM 06/15/2022 SR 20222739355 - File Number 6860351 IN WITNESS WHEREOF, the Corporation has caused this Certificate of Domestication to be executed in its name this 15th day of June, 2022.

VIRGIN GROUP ACQUISITION CORP. II,

a Cayman Islands company

By: /s/ Evan Lovell
Name: Evan Lovell

Title: Chief Financial Officer

[Signature Page to Certificate of Corporate Domestication]

THIS WARRANT AND THE SECURITIES ISSUABLE UPON EXERCISE HEREOF HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "1933 ACT"), OR APPLICABLE STATE SECURITIES LAWS AND MAY NOT BE OFFERED FOR SALE, SOLD, TRANSFERRED OR OTHERWISE DISPOSED OF IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT FOR SUCH SECURITIES UNDER THE 1933 ACT, OR AN OPINION OF COUNSEL, REASONABLY SATISFACTORY TO THE ISSUER HEREOF, TO THE EFFECT THAT REGISTRATION IS NOT REQUIRED UNDER THE 1933 ACT AS AN EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE 1933 ACT IS AVAILABLE.

Grove Collaborative Holdings, Inc.

WARRANT TO PURCHASE CLASS A COMMON STOCK

Warrant No.: 001

Date of Issuance: June 16, 2022

Grove Collaborative Holdings, Inc., a Delaware public benefit corporation (the 'Company'), hereby certifies that, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Corvina Holdings Limited, a BVI business company limited by shares incorporated in the British Virgin Islands, the registered holder hereof or its registered permitted assigns (collectively, the "Holder"), is entitled, subject to the terms set forth below to purchase from the Company, at the Exercise Price (as defined below), at any time or times on or after the date hereof (the "Issuance Date") until the Expiration Date (as defined below), 3,875,028 fully paid nonassessable shares of Class A Common Stock (as defined below), subject to adjustment as provided herein (the "Warrant Shares"). Except as otherwise defined herein, capitalized terms in this Warrant to Purchase Class A Common Stock (including any Warrants to Purchase Class A Common Stock issued in exchange, transfer or replacement hereof, this "Warrant"), shall have the meanings set forth in Section 15 hereof. This Warrant has been issued in connection with that certain Subscription Agreement, dated as of March 31, 2022, by and between the Company and Holder (as amended, restated, supplemented, refinanced, extended or otherwise modified from time to time, the "Subscription Agreement").

1. EXERCISE OF WARRANT.

(a) Mechanics of Exercise. Subject to the terms and conditions hereof (including, without limitation, the limitations set forth in Section 1(d)), this Warrant may be exercised by the Holder at any time or times on or after the Issuance Date until the Expiration Date, in whole or in part, by (i) delivery of a written notice, in the form attached hereto as Exhibit A (the "Exercise Notice"), of the Holder's election to exercise all or part of this Warrant and (ii) (A) payment to the Company of an amount equal to the Exercise Price multiplied by the number of Warrant Shares as to which this Warrant is being exercised (the "Aggregate Exercise Price") in cash by wire transfer of immediately available funds or (B) by instructing the Company to withhold a number of Warrant Shares issuable upon such exercise of this Warrant with an aggregate Fair Market Value as of the date of the Exercise Notice equal to the Aggregate Exercise Price (a "Cashless Exercise"). The Holder shall not be required to deliver the original Warrant in order to effect an exercise hereunder, nor shall any ink-original signature or medallion guarantee (or other type of guarantee or notarization) with respect to any Exercise Notice be required. Execution and delivery of the Exercise Notice with respect to a number of Warrant Shares that is less than all of the Warrant Shares shall have the same effect as cancellation of the original Warrant and the issuance of a new Warrant, on the same terms contained herein, evidencing the right to purchase the remaining number of Warrant Shares. On or before the third (3rde) Business Day following the date on which the Company has received the Exercise Notice, the Company shall transmit by electronic mail an acknowledgment of confirmation of receipt of the Exercise Notice to the Holder and the Company's transfer agent (the "Transfer Agent") and shall provide to the Holder instructions for payment of the Aggregate Exercise Price, if applicable. On or before the fifth (5th) Business Day following the date on which the Company has received the Exercise Notice (the 'Share Delivery Date'), so long as the Holder delivers the Aggregate Exercise Price (or notice of a Cashless Exercise) on or prior to noon Eastern Time on the fourth (4th) Business Day following the date on which the Company has received the Exercise Notice (provided that if the Aggregate Exercise Price (or notice of a Cashless Exercise) has not been delivered by such date, the Share Delivery Date shall be two (2) Business Days after the Aggregate Exercise Price (or notice of a Cashless Exercise) is delivered), the Company shall credit such aggregate

number of Warrant Shares to which the Holder is entitled pursuant to such exercise to the Holder's or its designee's account with the Transfer Agent or, if permitted by the Company and requested by the Holder, by physical delivery of a certificate, registered in the Company's share register in the name of the Holder or its designee. The Company shall be responsible for all fees and expenses incurred in connection with the issuance of the Warrant Shares, including the fees and expenses of the Transfer Agent, if any. Upon delivery of the Exercise Notice and the Aggregate Exercise Price (or notice of a Cashless Exercise) therefore, the Holder shall be deemed for all corporate purposes to have become the holder of record and beneficial owner of the Warrant Shares with respect to which this Warrant has been exercised, irrespective of the date such Warrant Shares are delivered. If this Warrant is physically delivered by the Holder to the Company in connection with any exercise pursuant to this Section 1(a) and the number of Warrant Shares available for exercise pursuant to this Warrant is greater than the number of Warrant Shares that the Holder seeks to acquire pursuant to the current exercise, then the Company shall as soon as practicable and in no event later than five (5) Business Days after such exercise and at its own expense, issue a new Warrant (on the same terms contained herein and in accordance with Section 6(e)) representing the right to purchase the number of Warrant Shares issuable immediately prior to such exercise under this Warrant, less the number of Warrant Shares with respect to which this Warrant is exercised. No fractional Warrant Shares are to be issued upon the exercise of this Warrant, but rather the number of Warrant Shares to be issued shall be rounded down to the nearest whole number. The Company's obligations to issue and deliver Warrant Shares in accordance with the terms and subject to the conditions hereof are absolute and unconditional, irrespective of any action or inaction by the Holder to enforce the same, any waiver or consent with respect to any provision hereof, the recovery of any judgment against any Person or any action to enforce the same, or any setoff, counterclaim, recoupment, limitation or termination; provided, however, that the Company shall not be required to deliver Warrant Shares with respect to an exercise prior to the Holder's delivery of the Aggregate Exercise Price (or notice of a Cashless Exercise) with respect to such exercise.

- (b) Exercise Price. For purposes of this Warrant, "Exercise Price" means \$0.01 per Warrant Share, subject to adjustment as provided herein.
- (c) <u>Disputes</u>. In the case of a dispute as to the determination of the Exercise Price or the arithmetic calculation of the Warrant Shares, the Company shall promptly issue to the Holder the number of Warrant Shares that are not disputed and resolve such dispute in accordance with <u>Section 10</u> hereof

(d) Limitations on Exercise.

(i) Notwithstanding anything to the contrary contained herein, the Company shall not effect the exercise of any portion of this Warrant, and the Holder shall not have the right to exercise any portion of this Warrant, and any such exercise shall be null and void and treated as if never made, to the extent that after giving effect to such exercise, the Holder together with the other Attribution Parties collectively would beneficially own in excess of 9.99% (the "Maximum Percentage") of the shares of Common Stock outstanding immediately after giving effect to such exercise. Notwithstanding anything to the contrary set forth herein, by written notice to the Company, the Holder may, in its sole discretion, increase or decrease the Maximum Percentage to any other percentage not in excess of 19.999% of the Common Stock of the Company issued and outstanding on June 16, 2022 (when taken together with any shares issued pursuant to this Warrant unless the Company's shareholders shall have approved the transactions contemplated by this Warrant) specified in such notice and such percentage will be deemed the new Maximum Percentage for all purposes under this Warrant; provided, that any such increase will not be effective until the sixty-first (6 bt) day after such notice is delivered to the Company. For purposes of the preceding sentences, the aggregate number of shares of Common Stock beneficially owned by the Holder and the other Attribution Parties shall include the number of shares of Common Stock held by the Holder and all other Attribution Parties plus the number of shares of Common Stock issuable upon exercise of this Warrant with respect to which the determination of such sentence is being made, but shall exclude shares of Common Stock which would be issuable upon (A) exercise of the remaining, unexercised portion of this Warrant beneficially owned by the Holder or any of the other Attribution Parties and (B) exercise or conversion of the unexercised or unconverted portion of any other securities of the Company (including, without limitation, any convertible notes or convertible preferred stock or warrants, including other warrants issued concurrently herewith) beneficially owned by the Holder or any other Attribution Party subject to a limitation on conversion or exercise analogous to the limitation contained in this Section 1(d). For purposes of this Section 1(d), beneficial ownership shall be calculated in accordance with Section 13(d) of the 1934 Act. For purposes of determining the number of outstanding shares of Common Stock the Holder may acquire upon the exercise of this Warrant without exceeding the Maximum Percentage, there were 162,771,033 shares of Common Stock issued and outstanding as of the issuance of this Warrant on June 16, 2022.

- (ii) Additionally, the Holder shall not be entitled to exercise this Warrant for a number of Warrant Shares in excess of that number of Warrant Shares which, upon giving effect to such exercise, would result any required filing and clearance under the Hart-Scott-Rodino Antitrust Improvements Act (the "HSR Act") if one has not been made and pre-approval obtained. If a filing and clearance under the HSR Act would be required in connection with the exercise of this Warrant, the Holder and the Company shall make such filings promptly and to seek early termination, and only exercise such Warrants after applicable approvals are obtained.
- (iii) If the Company receives an Exercise Notice from the Holder at a time when the issuance of the Warrant Shares would not comply with applicable law, including the rules and regulations of the New York Stock Exchange (the "NYSE") (or any securities exchange on which the Common Stock is listed) ("Applicable Law"), the Company shall (A) notify the Holder in writing of the number of shares of Common Stock issued or issuable in connection with any other transaction by the Company that may be aggregated with the issuance of the Warrant Shares under the rules and regulations of the NYSE (or any other securities exchange on which the Common Stock is then listed), and to the extent that such Exercise Notice would otherwise cause the Holder's beneficial ownership, as determined pursuant to this Section 1(d), to exceed the Maximum Percentage or would not comply with Applicable Law, the Holder must notify the Company of a reduced number of Warrant Shares to be acquired pursuant to such Exercise Notice (the number of shares by which such purchase is reduced, the "Reduction Shares") and (B) as soon as reasonably practicable, the Company shall return to the Holder any Exercise Price paid by the Holder for the Reduction Shares.
- (vi) Upon the written request of the Holder, the Company shall promptly confirm in writing or by electronic mail to the Holder the number of shares of Common Stock issued or issuable in connection with any other transaction by the Company that may be aggregated with the issuance of the Warrant Shares under the rules and regulations of the NYSE (or any other securities exchange on which the Common Stock is then listed).
- (v) For purposes of clarity, (A) the shares of Class A Common Stock issuable pursuant to the terms of this Warrant in excess of the Maximum Percentage or the issuance of which would not comply with Applicable Law shall not be deemed to be beneficially owned by the Holder for any purpose including for purposes of Section 13(d) or Rule 16a-1(a)(1) of the 1934 Act.
- (vi) No prior inability to exercise this Warrant pursuant to this <u>Section 1(d)</u> shall have any effect on the applicability of the provisions of this <u>Section 1(d)</u> with respect to any subsequent determination of exercisability. The provisions of this paragraph shall be construed and implemented in a manner otherwise than in strict conformity with the terms of this <u>Section 1(d)</u> to the extent necessary to correct this paragraph or any portion of this <u>Section 1(d)</u> which may be defective or inconsistent with the intended beneficial ownership or Applicable Law limitation contained in this <u>Section 1(d)</u> or to make changes or supplements necessary or desirable to properly give effect to such limitation. The limitation contained in this <u>Section 1(d)</u> may not be waived and shall apply to a successor holder of this Warrant.
- (vii) For purposes of Rule 144 promulgated under the Securities Act of 1933, as amended, it is intended, understood and acknowledged that the Warrant Shares issued in a Cashless Exercise transaction shall be deemed to have been acquired by the Holder, and the holding period for the Warrant Shares shall be deemed to have commenced, on the date this Warrant was originally issued (provided that the U.S. Securities and Exchange Commission continues to take the position that such treatment is proper at the time of such exercise). In the event that a registration statement registering the issuance of Warrant Shares is, for any reason, not effective at the time of exercise of this Warrant, then the Warrant may only be exercised through a Cashless Exercise, as set forth in this Section 1.
- (e) <u>Required Reserve Amount</u>. So long as this Warrant remains outstanding, the Company shall at all times keep reserved for issuance, free from preemptive or any other contingent purchase rights (other than those of the Holder), under this Warrant a number of shares of Class A Common Stock equal to at least 100% of the maximum number of shares of Class A Common Stock as shall be necessary to satisfy the Company's obligation to issue shares of Class A Common Stock under this Warrant (without regard to any limitations on exercise contained herein) (the "**Required Reserve Amount**"). At no time shall the number of shares of Class A Common Stock reserved pursuant to this <u>Section 1(e)</u> be reduced other than proportionally in connection with any exercise of this Warrant or such other event covered by <u>Section 2</u> below.

- (f) Insufficient Authorized Shares. If, notwithstanding Section 1(e), and not in limitation thereof, at any time while this Warrant remains outstanding the Company does not have a sufficient number of authorized and unreserved shares of Class A Common Stock to satisfy its obligation to reserve the Required Reserve Amount (an "Authorized Share Failure"), then the Company shall immediately take all action necessary to increase the Company's authorized shares of Class A Common Stock to an amount sufficient to allow the Company to reserve the Required Reserve Amount. Without limiting the generality of the foregoing sentence, as soon as practicable after the date of the occurrence of an Authorized Share Failure, but in no event later than sixty (60) days after the occurrence of such Authorized Share Failure, the Company shall file a proxy statement for a meeting of its stockholders for the approval of an increase in the number of authorized shares of Class A Common Stock. In connection with such meeting, the Company shall provide each stockholder with a proxy statement, the Company's board of directors shall recommend each stockholder vote in favor of such increase in authorized shares of Class A Common Stock, the Company shall use its best efforts to solicit its stockholders' approval of such increase in authorized shares of Class A Common Stock and promptly after receipt of such approval the Company shall file an amendment to its certificate of incorporation with the Secretary of State of the State of Delaware reflecting the increased in authorized shares of Class A Common Stock.
- (g) Charges, Taxes and Expenses. Issuance and delivery of certificates for shares of Common Stock upon exercise of this Warrant shall be made without charge to the Holder for any issue or transfer tax, transfer agent fee or other incidental tax or expense (excluding any applicable stamp duties) in respect of the issuance of such certificates, all of which taxes and expenses shall be paid by the Company; provided, however, that the Company shall not be required to pay any tax that may be payable in respect of any transfer involved in the registration of any certificates for Warrant Shares or the Warrants in a name other than that of the Holder or an Affiliate thereof. The Holder shall be responsible for all other tax liability that may arise as a result of holding or transferring this Warrant or receiving Warrant Shares upon exercise hereof.
- 2. <u>ADJUSTMENT OF EXERCISE PRICE AND NUMBER OF WARRANT SHARES</u>. The Exercise Price and the number of Warrant Shares shall be adjusted from time to time as follows:
- (a) Adjustment Upon Dividend, Subdivision or Combination of Common Stock. If the Company shall, at any time or from time to time after the Issuance Date, (i) pay a dividend or make any other distribution upon the Class A Common Stock without consideration or (ii) reclassify or subdivide (including by any stock split, stock dividend, recapitalization, substitutions, exchange or otherwise) its outstanding shares of Class A Common Stock into a greater number of shares, the Exercise Price in effect immediately prior to any such dividend, distribution, reclassification or subdivision will be proportionately reduced and the number of Warrant Shares issuable upon exercise of this Warrant shall be proportionately increased. If the Company at any time on or after the Issuance Date combines (by combination, reverse stock split or otherwise) its outstanding shares of Class A Common Stock into a smaller number of shares, the Exercise Price in effect immediately prior to such combination will be proportionately increased and the number of Warrant Shares issuable upon exercise of this Warrant shall be proportionately decreased. Any adjustment under this Section 2(a) will become effective at the close of business Eastern Time on the date the subdivision or combination becomes effective. The provisions of this Section 2 will similarly apply to successive stock dividends, stock splits or combinations, reclassifications, exchanges, substitutions, or other events.
- (b) <u>Number of Warrant Shares; Calculations</u>. Simultaneously with any adjustment to the Exercise Price pursuant to <u>Section 2</u> or <u>Section 3</u>, the number of Warrant Shares that may be purchased upon exercise of this Warrant shall be increased or decreased proportionately, so that after such adjustment the aggregate Exercise Price payable hereunder for the increased or decreased number of Warrant Shares shall be the same as the aggregate Exercise Price in effect immediately prior to such adjustment. All calculations under this <u>Section 2 and Section 3</u> shall be made to the nearest one-thousandth of one cent or the nearest share, as applicable.

(c) <u>Certificate as to Adjustment</u>. As promptly as reasonably practicable following any adjustment of the Exercise Price or other adjustments pursuant to this <u>Section 2</u> or <u>Section 3</u>, but in any event not later than ten (10) Business Days thereafter, the Company shall furnish to the Holder a certificate of an executive officer setting forth in reasonable detail such adjustment and the facts upon which it is based and certifying the calculation thereof. As promptly as reasonably practicable following the receipt by the Company of a written request by the Holder, but in any event not later than ten (10) Business Days thereafter, the Company shall furnish to the Holder and to the Company's Transfer Agent a certificate of an executive officer certifying the Exercise Price then in effect and the number of Warrant Shares or the amount, if any, of other shares of stock, securities or assets then issuable upon exercise of the Warrant.

3. FUNDAMENTAL TRANSACTIONS; DISTRIBUTION RIGHTS; PURCHASE RIGHTS.

(a) If, at any time while this Warrant is outstanding (i) the Company effects any merger or consolidation of the Company with or into another Person, in which the Company is not the surviving entity, (ii) the Company effects any sale to another Person of all or substantially all of its assets in one transaction or a series of related transactions, (iii) pursuant to any tender offer or exchange offer (whether by the Company or another Person), holders of capital stock tender shares representing more than 50% of the voting power of the capital stock of the Company and the Company or such other Person, as applicable, accepts such tender for payment, (iv) the Company consummates a stock purchase agreement or other business combination (including, without limitation, a reorganization, recapitalization, spin-off or scheme of arrangement but not including any underwritten offering, registered direct offering, private placement or other transaction with the primary purpose of financing or fund raising for the Company) with another Person whereby such other Person acquires more than the 50% of the voting power of the capital stock of the Company (except for any such transaction in which the stockholders of the Company immediately prior to such transaction maintain, in substantially the same proportions, the voting power of such Person immediately after the transaction), or (v) the Company effects any reclassification of the Common Stock or any compulsory share exchange pursuant to which the Common Stock is effectively converted into or exchanged for other securities, cash or property (other than as a result of a subdivision or combination of shares of Common Stock covered by Section 2(a) above) (in any such case, a "Fundamental Transaction"), then upon such Fundamental Transaction the Holder shall have the right to receive, upon exercise of this Warrant, the same amount and kind of securities, cash or property as it would have been entitled to receive upon the occurrence of such Fundamental Transaction if it had been, immediately prior to such Fundamental Transaction, the holder of the number of Warrant Shares then issuable upon exercise in full of this Warrant without regard to any limitations on exercise contained herein (the "Alternate Consideration"). The Company shall not effect any Fundamental Transaction in which the Company is not the surviving entity or the Alternate Consideration includes securities of another Person unless (i) the Alternate Consideration is solely cash and the Company provides for the simultaneous Cashless Exercise of this Warrant pursuant to Section 1 or (ii) prior to or simultaneously with the consummation thereof, any successor to the Company, surviving entity, ultimate parent (if the Company is a subsidiary of another Person) or other Person (including any purchaser of assets of the Company) shall assume the obligation to deliver to the Holder such Alternate Consideration as, in accordance with the foregoing provisions, the Holder may be entitled to receive, and the other obligations under this Warrant. The provisions of this paragraph (c) shall similarly apply to subsequent transactions analogous of a Fundamental Transaction type. Notice of such adjustments shall be given in accordance with Section 2(c).

(b) In addition to any adjustments pursuant to Section 2 and this Section 3, if the Company shall declare or make any dividend or other distribution of its assets (or rights to acquire its assets) to holders of shares of Common Stock, by way of return of capital or otherwise (including, without limitation, any distribution of cash, stock or other securities, property, options, evidence of indebtedness or any other assets by way of a dividend, spin off, reclassification, corporate rearrangement, scheme of arrangement or other similar transaction) (a "Distribution"), at any time after the issuance of this Warrant, then, in each such case, the Holder shall be entitled to participate in such Distribution to the same extent that the Holder would have participated therein if the Holder had held the number of Warrant Shares acquirable upon complete exercise of this Warrant (without regard to any limitations or restrictions on exercise of this Warrant, including without limitation, the Maximum Percentage or the other limitations set forth in Section 1(d)) immediately before the date on which a record is taken for such Distribution, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the participation in such Distribution.

(c) In addition to any adjustments pursuant to Section 2 and this Section 3, if at any time the Company grants, issues or sells any Capital Stock or rights to purchase Capital Stock or other property pro rata to the record holders of any class of Common Stock (the "Purchase Rights"), then the Holder will be entitled to acquire, upon the terms applicable to such Purchase Rights, the aggregate Purchase Rights which the Holder could have acquired if the

Holder had held the number of shares of Common Stock acquirable upon complete exercise of this Warrant (without regard to any limitations or restrictions on exercise of this Warrant, including without limitation, the Maximum Percentage or the other limitations set forth in Section 1(d)) immediately before the date on which a record is taken for the grant, issuance or sale of such Purchase Rights, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the grant, issuance or sale of such Purchase Rights.

4. NONCIRCUMVENTION. The Company hereby covenants and agrees that the Company will not, by amendment of its certificate of incorporation or bylaws, or through any reorganization, transfer of assets, consolidation, merger, scheme of arrangement, dissolution, issue or sale of securities, or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Warrant, and will at all times in good faith carry out all of the provisions of this Warrant and take all action as may be required to protect the rights of the Holder. Without limiting the generality of the foregoing, the Company (i) shall take all such actions as may be necessary or appropriate in order that the Company may validly and legally issue fully paid and nonassessable shares of Class A Common Stock upon the exercise of this Warrant and (ii) shall, so long as any of the Warrants are outstanding, take all action necessary to reserve and keep available out of its authorized and unissued and otherwise unreserved shares of Class A Common Stock, solely for the purpose of effecting the exercise of the Warrants as herein provided, the number of shares of Class A Common Stock as shall from time to time be necessary to effect the exercise of the Warrants then outstanding (without regard to any limitations on exercise), free from preemptive rights or any other contingent purchase rights of persons other than the Holder (taking into account the adjustments and restrictions set forth herein).

5. HOLDER NOT DEEMED A STOCKHOLDER; LEGEND; REGISTRATION RIGHTS.

(a) No Stock Rights. Except as otherwise specifically provided herein, the Holder, solely in such Person's capacity as a holder of this Warrant, shall not be entitled to vote or receive dividends or be deemed the holder of capital stock of the Company for any purpose, nor shall anything contained in this Warrant be construed to confer upon the Holder, solely in such Person's capacity as the Holder of this Warrant, any of the rights of a stockholder of the Company or any right to vote, give or withhold consent to any corporate action (whether any reorganization, issue of stock, reclassification of stock, consolidation, merger, conveyance or otherwise), receive notice of meetings, receive dividends or subscription rights, or otherwise, prior to the issuance to the Holder of the Warrant Shares which such Person is then entitled to receive upon the due exercise of this Warrant. In addition, nothing contained in this Warrant shall be construed as imposing any liabilities on the Holder to purchase any securities (upon exercise of this Warrant or otherwise) or as a stockholder of the Company, whether such liabilities are asserted by the Company or by creditors of the Company.

(b) <u>Legend</u>. Except as otherwise provided in this <u>Section 5(b)</u>, each Warrant Share initially issued upon the exercise of this Warrant, and each Warrant Share issued to any subsequent transferee of any such Warrant Share, in either case whether in certificated or book-entry form, shall be stamped or otherwise bear a legend in substantially the following form (the "Securities Act Legend"):

THESE SECURITIES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "1933 ACT"), OR APPLICABLE STATE SECURITIES LAWS AND MAY NOT BE OFFERED FOR SALE, SOLD, TRANSFERRED OR OTHERWISE DISPOSED OF IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT FOR SUCH SECURITIES UNDER THE 1933 ACT, OR AN OPINION OF COUNSEL, REASONABLY SATISFACTORY TO THE ISSUER HEREOF, TO THE EFFECT THAT REGISTRATION IS NOT REQUIRED UNDER THE 1933 ACT AS AN EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE 1933 ACT IS AVAILABLE.

Certificates or book entries evidencing the Warrant Shares shall not contain any legend (including the Securities Act Legend), (i) while a registration statement covering the resale of such Warrant Shares is effective under the 1933 Act, (ii) following any sale of such Warrant Shares pursuant to Rule 144, (iii) if such Warrant Shares are eligible for sale under Rule 144 without volume or manner-of-sale restrictions, or (iv) if such legend is not required under applicable requirements of the 1933 Act (including judicial interpretations and pronouncements issued by the staff of the U.S. Securities and Exchange Commission). The Company agrees that following such time as such legend is no longer required under this Section 5(b) and upon the request of the Holder, the Company will promptly following the delivery by the Holder to the Company or the Transfer Agent of Warrant Shares issued with a Securities Act Legend deliver or cause to be delivered to the Holder Warrant Shares free from any Securities Act Legend or other legend. The Company will use its commercially reasonable best efforts, including delivering an opinion to the Transfer Agent at its own expense, to ensure any legend (including the Securities Act Legend) is removed in accordance with this Section 5(b).

(c) <u>Registration Rights</u>. The Warrant Shares are entitled to registration rights pursuant to the Amended and Restated Registration Rights Agreement, dated as of June 16, 2022, by and among the Company, the Holder and the other stockholders that are signatories thereto.

6. REISSUANCE OF WARRANTS.

- (a) <u>Registration of Warrant</u>. The Company shall register this Warrant, upon the records to be maintained by the Company for that purpose (the "Warrant Register"), in the name of the record Holder hereof from time to time. The Company may deem and treat the registered Holder as the absolute owner hereof for the purpose of any exercise hereof or any distribution to the Holder, and for all other purposes, absent actual notice to the contrary. The Company shall also register any transfer, exchange, reissuance or cancellation of any portion of this Warrant in the Warrant Register. No service charge shall be made to the Holder for any registration of transfer, but the Company may require payment of a sum sufficient to cover any transfer taxes required in connection with making of any such transfer.
- (b) <u>Transfer of Warrant</u>. Subject to applicable securities laws, if this Warrant is to be transferred by the Holder, the Holder shall surrender this Warrant to the Company or its Transfer Agent, as directed by the Company, together with all applicable transfer taxes, whereupon the Company will, or will cause its Transfer Agent to, forthwith issue and deliver upon the order of the Holder a new Warrant (in accordance with <u>Section 6(e)</u>), registered as the Holder may request, representing the right to purchase the number of Warrant Shares being transferred by the Holder and, if less than the total number of Warrant Shares then underlying this Warrant is being transferred, a new Warrant (in accordance with <u>Section 6(e)</u>) to the Holder representing the right to purchase the number of Warrant Shares not being transferred. The acceptance of the new Warrant by the transferee thereof shall be deemed the acceptance by such transferee of all of the rights and obligations in respect of the new Warrant that the Holder has in respect of this Warrant.
- (c) <u>Lost, Stolen or Mutilated Warrant</u>. Upon receipt by the Company of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant, and, in the case of loss, theft or destruction, of any indemnification undertaking by the Holder to the Company in customary form (but without the obligation to post a bond) and, in the case of mutilation, upon surrender and cancellation of this Warrant, the Company shall execute and deliver to the Holder a new Warrant (in accordance with <u>Section 6(e)</u>) representing the right to purchase the Warrant Shares then underlying this Warrant.
- (d) Exchangeable for Multiple Warrants; No Fractional Warrant Shares. This Warrant is exchangeable, upon the surrender hereof by the Holder at the principal office of the Company, for a new Warrant or Warrants (in accordance with Section 6(e)) representing in the aggregate the right to purchase the number of Warrant Shares then underlying this Warrant, and each such new Warrant will represent the right to purchase such portion of such Warrant Shares as is designated by the Holder at the time of such surrender; provided, however, that the Company or its Transfer Agent, as directed by the Company, shall not be required to issue Warrants for fractional shares of Class A Common Stock hereunder, but rather the number of Warrant Shares to be issued shall be rounded down to the nearest whole number.
- (e) <u>Issuance of New Warrants</u>. Whenever the Company or its Transfer Agent, as directed by the Company, is required to issue a new Warrant pursuant to the terms of this Warrant, such new Warrant shall (i) be of like tenor with this Warrant, (ii) represent, as indicated on the face of such new Warrant, the right to purchase the Warrant Shares then underlying this Warrant (or in the case of a new Warrant being issued pursuant to <u>Section 6(b)</u> or <u>Section 6(c)</u>, the Warrant Shares designated by the Holder which, when added to the number of shares of Class A Common Stock underlying the other new Warrants issued in connection with such issuance, does not exceed the number of Warrant Shares then underlying this Warrant), (iii) have an issuance date, as indicated on the face of such new Warrant which is the same as the Issuance Date and (iv) have the same rights and conditions as this Warrant.

- (f) No Fractional Shares. No fractional Warrant Shares will be issued in connection with any exercise of this Warrant. In lieu of any fractional shares that would otherwise be issuable, the number of Warrant Shares to be issued shall be rounded down to the next whole number and the Company shall pay the Holder in cash the Fair Market Value for any such fractional shares.
- 7. NOTICES. Any notice or communication required or permitted hereunder shall be in writing and either delivered personally, emailed or sent by overnight mail via a reputable overnight carrier, or sent by certified or registered mail, postage prepaid, and shall be deemed to be given and received (a) when so delivered personally, (b) when sent, with no mail undeliverable or other rejection notice, if sent by email, or (c) five (5) Business Days after the date of mailing to the address below or to such other address or addresses as such person may hereafter designate by notice given hereunder:
 - (i) if to the Company, to:

Grove Collaborative Holdings, Inc. 1301 Sansome St. San Francisco, California 94111 Attention: Nathan Francis

Email: nfrancis@grove.co with a copy (which copy shall not constitute notice) to:

1001 Page Mill Road Building 1 Palo Alto, California 94304 Attention: Martin A. Wellington Email: mwellington@sidley.com

Sidley Austin LLP 1999 Avenue of the Stars 17th Floor Los Angeles, California 90067 Attention: Joshua G. DuClos Email: jduclos@sidley.com

(ii) if to the Holder, to:

Corvina Holdings Limited Craigmuir Chambers, PO Box 71 Road Town Tortola British Virgin Islands

with a copy (which copy shall not constitute notice) to:

Virgin Management USA, Inc. 65 Bleecker Street, 6th Floor New York, NY 10012 Attention: Harold Brunink

Email: harold.brunink@virgin.com

with a copy (which copy shall not constitute notice) to:

Davis Polk & Wardwell LLP 450 Lexington Avenue New York, NY 10017

Attention: Derek Dostal, Lee Hochbaum, William Aaronson Email: derek.dostal@davispolk.com lee.hochbaum@davispolk.com william.aaronson@davispolk.com

- 8. <u>AMENDMENT AND WAIVER</u>. Except as otherwise provided herein, the provisions of this Warrant may be amended or waived and the Company may take any action herein prohibited, or omit to perform any act herein required to be performed by it, only if the Company has obtained the written consent of the Holder. No waiver by the Company or the Holder of any of the provisions hereof shall be effective unless explicitly set forth in writing and signed by the party so waiving. No waiver by any party shall operate or be construed as a waiver in respect of any failure, breach or default not expressly identified by such written waiver, whether of a similar or different character, and whether occurring before or after that waiver. No failure to exercise, or delay in exercising, any right, remedy, power or privilege arising from this Warrant shall operate or be construed as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege.
- 9. GOVERNING LAW; JURISDICTION; JURY TRIAL. This Warrant shall be governed by and construed and enforced in accordance with, and all questions concerning the construction, validity, interpretation and performance of this Warrant shall be governed by, the internal laws of the State of New York, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of New York or any other jurisdictions) that would cause the application of the laws of any jurisdictions other than the State of New York. The Company hereby irrevocably submits to the exclusive jurisdiction of the state and federal courts sitting in The City of New York, Borough of Manhattan, for the adjudication of any dispute hereunder or in connection herewith or with any transaction contemplated hereby or discussed herein, and hereby irrevocably waives, and agrees not to assert in any suit, action or proceeding, any claim that it is not personally subject to the jurisdiction of any such court, that such suit, action or proceeding is brought in an inconvenient forum or that the venue of such suit, action or proceeding is improper. The Company hereby irrevocably waives personal service of process and consents to process being served in any such suit, action or proceeding by mailing a copy thereof to the Company at the address set forth in Section 7(i) above or such other address as the Company subsequently delivers to the Holder and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law. Nothing contained herein shall be deemed or operate to preclude the Holder from bringing suit or taking other legal action against the Company in any other jurisdiction to collect on the Company's obligations to the Holder, to realize on any collateral or any other security for such obligations, or to enforce a judgment or other court ruling in favor of the Holder. THE COMPANY HEREBY IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR IN CONNECTION WITH OR ARISING OUT OF THIS WARRANT OR ANY TRANSACTION CONTEMPLATED HEREBY.
- 10. <u>DISPUTE RESOLUTION</u>. In the case of a dispute as to the determination of the Exercise Price or the arithmetic calculation of the Warrant Shares, the Company shall submit the disputed determinations or arithmetic calculations in writing (including via facsimile or email) within two (2) Business Days of receipt of the Exercise Notice giving rise to such dispute, as the case may be, to the Holder. If the Holder and the Company are unable to agree upon such determination or calculation of the Exercise Price or the Warrant Shares within three (3) Business Days of such disputed determination or arithmetic calculation being submitted to the Holder, then the Company shall, within two (2) Business Days submit in writing (including via facsimile or email) the disputed determination of the Exercise Price to an independent, reputable investment bank selected by the Company and approved by the Holder. The Company shall cause at its expense the investment bank to perform the determinations or calculations and notify the Company and the Holder of the results no later than ten (10) Business Days from the time it receives the disputed determinations or calculations. Such investment bank's determination or calculation, as the case may be, shall be binding upon all parties absent demonstrable error.
- 11. <u>REMEDIES, OTHER OBLIGATIONS, BREACHES AND INJUNCTIVE RELIEF.</u> The remedies provided in this Warrant shall be cumulative and in addition to all other remedies available under this Warrant, at law or in equity (including a decree of specific performance and/or other injunctive relief), and nothing herein shall limit the right of the Holder to pursue actual damages for any failure by the Company to comply with the terms of this Warrant. The Company acknowledges that a breach by it of its obligations hereunder will cause

irreparable harm to the Holder and that the remedy at law for any such breach may be inadequate. The Company therefore agrees that, in the event of any such breach or threatened breach, the holder of this Warrant shall be entitled, in addition to all other available remedies, to an injunction restraining any breach, without the necessity of showing economic loss and without any bond or other security being required.

12. TRANSFER

- (a) The Holder agrees not to make any sale, transfer, assignment, hypothecation or other disposal of all or any portion of this Warrant or the Warrant Shares or any beneficial interest herein or therein, unless (i) a registration statement under the 1933 Act covering the disposition of such securities is effective or the Company is provided with an opinion of counsel reasonably satisfactory to the Company to the effect that such sale, transfer, assignment, hypothecation or other disposal will be exempt from the requirements of the 1933 Act and the registration or qualification requirements of any applicable state securities laws and (ii) the transferee shall have confirmed to the Company in writing, the representations and warranties set forth in Section 14(b).
- (b) Notwithstanding Section 12(a), this Warrant or the Warrant Shares may not be transferred except to (i) any Affiliate of the Holder or (ii) any Person that is a permitted assignee of the Holder under the Subscription Agreement (as may apply to the transfer of this Warrant or the Warrant Shares).
- 13. SEVERABILITY; CONSTRUCTION; HEADINGS. If any provision of this Warrant is prohibited by law or otherwise determined to be invalid or unenforceable by a court of competent jurisdiction, the provision that would otherwise be prohibited, invalid or unenforceable shall be deemed amended to apply to the broadest extent that it would be valid and enforceable, and the invalidity or unenforceability of such provision shall not affect the validity of the remaining provisions of this Warrant so long as this Warrant as so modified continues to express, without material change, the original intentions of the parties as to the subject matter hereof and the prohibited nature, invalidity or unenforceability of the provision(s) in question does not substantially impair the respective expectations or reciprocal obligations of the parties or the practical realization of the benefits that would otherwise be conferred upon the parties. The parties will endeavor in good faith negotiations to replace the prohibited, invalid or unenforceable provision(s) with a valid provision(s), the effect of which comes as close as possible to that of the prohibited, invalid or unenforceable provision(s). This Warrant shall be deemed to be jointly drafted by the Company and the Holder and shall not be construed against any Person as the drafter hereof. The headings of this Warrant are for convenience of reference and shall not form part of, or affect the interpretation of, this Warrant.
- 14. <u>COUNTERPARTS</u>. This Agreement may be executed in any number of original or facsimile counterparts and each of such counterparts shall for all purposes be deemed to be an original, and all such counterparts shall together constitute but one and the same instrument.

15. REPRESENTATIONS AND WARRANTIES.

- (a) The Company hereby represents and warrants to the Holder and covenants that:
- (i) Organization; Authorization. The Company is a corporation duly organized, validly existing and in good standing under the laws of the state of Delaware and has all requisite corporate power and authority to carry on its business as presently conducted or proposed to be conducted. The Company further represents and warrants to the Holder that it has taken all necessary corporate action and has obtained all necessary approvals in connection with the issuance of this Warrant Shares issuable upon exercise thereof. Except as described in this Warrant, no Company shareholder approval is required in connection with the issuance of this Warrant or the Warrant Share issuable upon exercise thereof. The Company has applied to list the Warrant Shares on the NYSE (or such other securities exchange on which the Common Stock is then listed). This Warrant, when executed and delivered by the Company, shall constitute a valid and legally binding obligation of the Company, enforceable in accordance with its terms, except as enforceability thereof may be limited by bankruptcy, insolvency, moratorium and other similar laws affecting creditors' rights generally and by general equitable principles.

- (ii) <u>Capitalization</u>. The Company now has, and shall at all times keep reserved for issuance, free from preemptive or any other contingent purchase rights (other than those of the Holder), under this Warrant a number of shares of Class A Common Stock reserved for issuance equal to the Required Reserve Amount. The Warrant Shares have been duly and validly reserved for issuance, and upon issuance in accordance with the terms of the Warrant, will be duly and validly issued, fully paid and nonassessable and free of any liens or encumbrances.
- (iii) Non-Contravention. The execution and delivery of this Warrant does not, and the issuance of the Warrant Shares in accordance with the terms of this Warrant will not (A) violate the Company's certificate of incorporation or bylaws, (B) violate any law or regulation applicable to the Company or order or decree of any court or public authority having jurisdiction over the Company, or (C) result in a breach of any material mortgage, indenture, contract, agreement or undertaking to which the Company is a party or by which it is bound.
 - (b) Representations and Warranties of the Holder. The Holder hereby represents and warrants to the Company as follows:
- (i) <u>Organization</u>: Authorization. The Holder is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization with full corporate, partnership or limited liability company power and authority to enter into and to consummate the transactions contemplated by this Warrant and otherwise to carry out its obligations hereunder. The execution and delivery of this Warrant has been duly authorized by all necessary corporate, partnership or limited liability company action on the part of the Holder. This Warrant has been duly executed by the Holder, and when delivered by the Holder in accordance with the terms hereof, will constitute the valid and legally binding obligation of the Holder, enforceable against it in accordance with its terms, except as limited by general equitable principles and applicable bankruptcy, insolvency, reorganization, moratorium and other laws of general application affecting enforcement of creditors' rights generally.
- (ii) <u>Understandings or Arrangements</u>. The Holder is acquiring this Warrant as principal for its own account and has no direct or indirect arrangement or understandings with any other persons to distribute or regarding the distribution of this Warrant or the Warrant Shares (this representation and warranty not limiting the Holder's right to sell the Warrant Shares pursuant to any effective registration statement or otherwise in compliance with applicable federal and state securities laws).
- (iii) <u>Status</u>. At the time the Holder was offered this Warrant, it was, and as of the date hereof it is, either: (i) an "accredited investor" as defined in Rule 501(a)(1), (a)(2), (a)(3), (a)(7) or (a)(8) under the 1933 Act or (ii) a "qualified institutional buyer" as defined in Rule 144A(a) under the 1933 Act. The Holder is not required to be registered as a broker-dealer under Section 15 of the 1934 Act.
- (iv) Experience of the Holder. The Holder, either alone or together with its representatives, has such knowledge, sophistication and experience in business and financial matters so as to be capable of evaluating the merits and risks of the prospective investment in this Warrant, and has so evaluated the merits and risks of such investment. The Holder is able to bear the economic risk of an investment in this Warrant and, at the present time, is able to afford a complete loss of such investment.
 - 15. CERTAIN DEFINITIONS. For purposes of this Warrant, the following terms shall have the following meanings:
 - (a) "1933 Act" means the Securities Act of 1933, as amended.
 - (b) "1934 Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder.
- (c) "Affiliate" means, with respect to any Person, any other Person that directly or indirectly controls, is controlled by, or is under common control with, such Person, it being understood for purposes of this definition that "control" of a Person means the power directly or indirectly either to vote 10% or more of the stock having ordinary voting power for the election of directors of such Person or direct or cause the direction of the management and policies of such Person whether by contract or otherwise.
- (d) "Attribution Parties" means, with respect to any Person, any other Person (including any other Persons with whom the first Person is deemed to be acting with together as a group) whose beneficial ownership would be aggregated with such first Person's (including by aggregation with one or more other Attribution Parties) for purposes of Section 13(d) of the 1934 Act. For clarity, the purpose of the foregoing is to subject collectively the Holder and all other Attribution Parties to the Maximum Percentage.

- (e) "Business Day" means any day other than Saturday, Sunday or other day on which commercial banks in The City of New York are authorized or required by law to remain closed.
- (f) "Capital Stock" means (a) any shares, interests, participations or other equivalents (however designated) of capital stock of a corporation; (b) any ownership interests in a Person other than a corporation, including membership interests, partnership interests, joint venture interests and beneficial interests; and (c) any warrants, options, convertible or exchangeable securities, subscriptions, rights (including any preemptive or similar rights), calls or other rights to purchase or acquire any of the foregoing.
- (g) "Class A Common Stock" means (i) the Company's Class A common stock, par value \$0.0001 per share, and (ii) any capital stock into which such Class A Common Stock shall have been changed or any capital stock resulting from a reclassification of such Class A Common Stock.
- (h) "Common Stock" means (i) the Class A Common Stock, (ii) the Company's Class B common stock, par value \$0.0001 per share, and (iii) any capital stock into which such Common Stock shall have been changed or any capital stock resulting from a reclassification of such Common Stock.
 - (i) "Expiration Date" means the date that is the fifth anniversary of the Issuance Date.
- (j) "Fair Market Value" means, as of any particular date: (a) the volume weighted average of the closing sales prices of the Class A Common Stock for such day on all domestic securities exchanges on which the Class A Common Stock may at the time be listed; (b) if there have been no sales of the Class A Common Stock on any such exchange on any such day, the average of the highest bid and lowest asked prices for the Class A Common Stock on all such exchanges at the end of such day; (c) if on any such day the Class A Common Stock is not listed on a domestic securities exchange, the closing sales price of the Class A Common Stock as quoted on the OTC Bulletin Board, the Pink OTC Markets or similar quotation system or association for such day; or (d) if there have been no sales of the Class A Common Stock on the OTC Bulletin Board, the Pink OTC Markets or similar quotation system or association on such day, the average of the highest bid and lowest asked prices for the Class A Common Stock quoted on the OTC Bulletin Board, the Pink OTC Markets or similar quotation system or association at the end of such day; in each case, averaged over twenty (20) consecutive Business Days ending on the Business Day immediately prior to the day as of which "Fair Market Value" is being determined; provided. that if the Class A Common Stock is listed on any domestic securities exchange, the term "Business Day" as used in this sentence means Business Days on which such exchange is open for trading. If at any time the Class A Common Stock is not listed on any domestic securities exchange or quoted on the OTC Bulletin Board, the Pink OTC Markets or similar quotation system or association, the "Fair Market Value" of the Class A Common Stock shall be the fair market value per share as determined by the Company's board of directors in good faith.
- (k) "Person" means an individual or corporation, partnership, trust, incorporated or unincorporated association, joint venture, limited liability company, joint stock company, government (or an agency or subdivision thereof) or other entity of any kind.

[Signature Page Follows]

IN WITNESS WHEREOF, the undersigned have caused this Warrant to be duly executed as of the Issuance Date set out above.

GROVE COLLABORATIVE HOLDINGS, INC.

By: /s/ Stuart Landesberg
Name: Stuart Landesberg
Title: Chief Executive Officer

Accepted and agreed:

CORVINA HOLDINGS LIMITED

By: /s/ Kerry Graziola
Name: Kerry Graziola
Title: Alternate Director

[Signature Page to Warrant to Purchase Class A Common Stock]

EXHIBIT A

EXERCISE NOTICE

[To be executed only upon exercise of this Warrant]

 $To: Grove\ Collaborative\ Holdings,\ Inc.$

exercises t	undersigned registered owner of the Warrant to Purchase Class A Common St the Warrant for the purchase of shares of Class A common stock, par v tive Holdings, Inc. (the "Company") and hereby elects to pay the Exercise Prior	alue \$0.0001 per share (the "Warrant Shares") of Grove
	pursuant to Section 1(a)(ii)(A) of the Warrant by delivery to the Company Company or by wire transfer of immediately available funds to an account Aggregate Exercise Price, or	
	pursuant to the "Cashless Exercise" option set forth in Section 1(a)(ii)(B) of	f the Warrant,
	price and on the terms and conditions specified in the Warrant and requests that ities or other property issuable upon such exercise) be issued in the name of and	
	, whose ac	ldress is
new Warrant. By its deli Holder, th Schedule	ch Warrant Shares shall not include all of the Warrant Shares issuable as provide ant of like tenor and date for the balance of the Warrant Shares issuable thereus invery of this Exercise Notice, the undersigned represents and warrants to the Core Holder and its Affiliates and any other Persons whose beneficial ownership of 13D beneficially own an aggregate of shares of Common Exchange Act of 1934, as amended, and under Section 1(d) of the Warrant to section 1 (d) of the Warrant to section 2 (d) of the Warrant to section 2 (d) of the Warrant 2 (d) of t	ompany that in giving effect to the exercise evidenced hereby the of Common Stock is aggregated with the Holder's on Holder's Stock, as determined in accordance with Section 13(d) of the which this notice relates.
		(Name of Registered Owner)
		(Signature of Registered Owner)
		(Street Address)
		(City) (State) (Zip Code)

(Signature must conform in all respects to name of Holder as specified on the face of the Warrant)

THIS WARRANT AND THE UNDERLYING SECURITIES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR UNDER THE SECURITIES LAWS OF ANY STATE. THESE SECURITIES MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED EXCEPT AS PERMITTED UNDER THE ACT AND APPLICABLE STATE SECURITIES LAWS IN ACCORDANCE WITH APPLICABLE REGISTRATION REQUIREMENTS OR AN EXEMPTION THEREFROM. THE ISSUER OF THESE SECURITIES MAY REQUIRE AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE ISSUER THAT SUCH OFFER, SALE, TRANSFER, PLEDGE OR HYPOTHECATION OTHERWISE COMPLIES WITH THE ACT AND ANY APPLICABLE STATE SECURITIES LAWS. THIS WARRANT MUST BE SURRENDERED TO THE COMPANY OR ITS TRANSFER AGENT AS A CONDITION PRECEDENT TO THE SALE, TRANSFER, PLEDGE OR HYPOTHECATION OF ANY INTEREST IN ANY OF THE SECURITIES REPRESENTED HEREBY.

WARRANT TO PURCHASE SHARES OF COMMON SHARES of GROVE COLLABORATIVE, INC. (fka ePantry LLC)) Dated as of _____ Void after the date specified in Section 8

Void after the date specified in Section 8		
No.		
THIS CERTIFIES THAT, for value received,, or its registered assigns (the "Holder"), is entitled, subject to the provisions and upon the terms and conditions set forth herein, to purchase from Grove Collaborative, Inc. (fka ePantry, LLC), a Delaware corporation (the "Company"), shares of the Company's Common Stock (the "Shares"), in the amounts, at such times and at the price per share set forth in Section 1. The term Warrant' as used herein shall include this Warrant and any warrants delivered in substitution or exchange therefor as provided herein. This Warrant is issued in connection with the transactions described in the Note and Warrant Purchase Agreement, dated as of, by and among the Company and the purchasers described therein (the "Purchase Agreement"), and is one of a series of warrants referred to as the "Warrants" in the Purchase Agreement. The following is a statement of the rights of the Holder and the conditions to which this Warrant is subject, and to which Holder, by acceptance of this Warrant, agrees:		
1. Number and Price of Shares; Exercise Period.		
(a) <i>Number of Shares</i> . Subject to any previous exercise of the Warrant, the Holder shall have the right to purchase up to the number of Shares that equals the quotient obtained by dividing (x)by (y) the Exercise Price (as defined below), prior to (or in connection with) the expiration of this Warrant as provided in Section 8.		
(b) Exercise Price. The exercise price per Share shall be equal to \$, subject to adjustment pursuant hereto (the "Exercise Price")		
(c) Exercise Period. This Warrant shall be exercisable, in whole or in part, prior to (or in connection with) the expiration of this Warrant as set forth in Section 8.		

2. Exercise of the Warrant.

- (a) *Exercise*. The purchase rights represented by this Warrant may be exercised at the election of the Holder, in whole or in part, in accordance with Section 1, by:
- (i) the tender to the Company at its principal office (or such other office or agency as the Company may designate) of a notice of exercise in the form of Exhibit A (the "Notice of Exercise"), duly completed and executed by or on behalf of the Holder, together with the surrender of this Warrant; and
- (ii) the payment to the Company of an amount equal to (x) the Exercise Price multiplied by (y) the number of Shares being purchased, by wire transfer or certified, cashier's or other check acceptable to the Company and payable to the order of the Company.
- (b) No Fractional Shares or Scrip. No fractional shares or scrip representing fractional shares shall be issued upon the exercise of the rights under this Warrant. In lieu of such fractional share to which the Holder would otherwise be entitled, the Company shall make a cash payment equal to the Exercise Price multiplied by such fraction.
- (c) Conditional Exercise. The Holder may exercise this Warrant conditioned upon (and effective immediately prior to) consummation of any transaction that would cause the expiration of this Warrant pursuant to Section 8 by so indicating in the notice of exercise.
- 3. Replacement of the Warrant. Subject to the receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant and, in the case of loss, theft or destruction, on delivery of an indemnity agreement reasonably satisfactory in form and substance to the Company or, in the case of mutilation, on surrender and cancellation of this Warrant, the Company at the expense of the Holder shall execute and deliver, in lieu of this Warrant, a new warrant of like tenor and amount.

4. Transfer of the Warrant.

- (a) *Warrant Register.* The Company shall maintain a register (the "*Warrant Register*") containing the name and address of the Holder or Holders. Until this Warrant is transferred on the Warrant Register in accordance herewith, the Company may treat the Holder as shown on the Warrant Register as the absolute owner of this Warrant for all purposes, notwithstanding any notice to the contrary. Any Holder of this Warrant (or of any portion of this Warrant) may change its address as shown on the Warrant Register by written notice to the Company requesting a change.
- (b) *Warrant Agent.* The Company may appoint an agent for the purpose of maintaining the Warrant Register referred to in Section 4(a), issuing the Shares or other securities then issuable upon the exercise of the rights under this Warrant, exchanging this Warrant, replacing this Warrant or conducting related activities.
- (c) *Transferability of the Warrant.* Subject to the provisions of this Warrant with respect to compliance with the Securities Act of 1933, as amended (the "Securities Act") and limitations on assignments and transfers, including without limitation compliance with the restrictions on transfer set forth in Section 5, title to this Warrant may be transferred by endorsement (by the transferor and the transfere executing the assignment form attached as Exhibit B (the "Assignment Form")) and delivery in the same manner as a negotiable instrument transferable by endorsement and delivery.

- (d) Exchange of the Warrant upon a Transfer. On surrender of this Warrant (and a properly endorsed Assignment Form) for exchange, subject to the provisions of this Warrant with respect to compliance with the Securities Act and limitations on assignments and transfers, the Company shall issue to or on the order of the Holder a new warrant or warrants of like tenor, in the name of the Holder or as the Holder (on payment by the Holder of any applicable transfer taxes) may direct, for the number of shares issuable upon exercise hereof, and the Company shall register any such transfer upon the Warrant Register. This Warrant (and the securities issuable upon exercise of the rights under this Warrant) must be surrendered to the Company or its warrant or transfer agent, as applicable, as a condition precedent to the sale, pledge, hypothecation or other transfer of any interest in any of the securities represented hereby.
- (e) *Minimum Transfer*. This Warrant may not be transferred in part unless such transfer is to a transferee who, pursuant to such transfer, receives the right to purchase at least 10,000 Shares hereunder (as adjusted from time to time in accordance with Section 6).
- (f) *Taxes.* In no event shall the Company be required to pay any tax which may be payable in respect of any transfer involved in the issue and delivery of any certificate in a name other than that of the Holder, and the Company shall not be required to issue or deliver any such certificate unless and until the person or persons requesting the issue thereof shall have paid to the Company the amount of such tax or shall have established to the satisfaction of the Company that such tax has been paid or is not payable.
- 5. Restrictions on Transfer of the Warrant and Shares; Compliance with Securities Laws. By acceptance of this Warrant, the Holder agrees to comply with the following:
- (a) Restrictions on Transfers. Subject to Section 5(b), this Warrant may not be transferred or assigned in whole or in part without the Company's prior written consent, and any attempt by Holder to transfer or assign any rights, duties or obligations that arise under this Warrant without such permission shall be void. Any transfer of this Warrant or the Shares (the "Securities") must be in compliance with all applicable federal and state securities laws. The Holder agrees not to make any sale, assignment, transfer, pledge or other disposition of all or any portion of the Securities, or any beneficial interest therein, unless and until the transferee thereof has agreed in writing for the benefit of the Company to take and hold such Securities subject to, and to be bound by, the terms and conditions set forth in this Warrant to the same extent as if the transferee were the original Holder hereunder, and
- (i) there is then in effect a registration statement under the Securities Act covering such proposed disposition and such disposition is made in accordance with such registration statement, or
- (ii) (A) such Holder shall have given prior written notice to the Company of such Holder's intention to make such disposition and shall have furnished the Company with a detailed description of the manner and circumstances of the proposed disposition, (B) the transferee shall have confirmed to the satisfaction of the Company in writing, substantially in the form of Exhibit A-1, that the Securities are being acquired (i) solely for the transferee's own account and not as a nominee for any other party, (ii) for investment and (iii) not with a view toward distribution or resale, and shall have confirmed such other matters related thereto as may be reasonably requested by the Company, and (C) such Holder shall have furnished the Company with (i) an opinion of counsel, reasonably satisfactory to the Company, to the effect that disposition will not require registration of such Securities under the Securities Act or (ii) a "no action" letter from the Securities and Exchange Commission to the effect that the transfer of such Securities without registration will not result in a recommendation by the staff of the Securities and Exchange Commission that action be taken with respect thereto, whereupon such Holder shall be entitled to transfer such Securities in accordance with the terms of the notice delivered by the Holder to the Company.

- (b) *Permitted Transfers*. Permitted transfers include (i) a transfer not involving a change in beneficial ownership, or (ii) transactions involving the distribution without consideration of Securities by any Holder to (x) a parent, subsidiary or other affiliate of a Holder that is a corporation, (y) any of the Holder's partners, members or other equity owners, or retired partners or members, or to the estate of any of its partners, members or other equity owners or retired partners or members, or (z) a venture capital fund that is controlled by or under common control with one or more general partners or managing members of, or shares the same management company with, the Holder; *provided*, in each case, that the Holder shall give written notice to the Company of the Holder's intention to effect such disposition and shall have furnished the Company with a detailed description of the manner and circumstances of the proposed disposition.
- (c) *Investment Representation Statement.* Unless the rights under this Warrant are exercised pursuant to an effective registration statement under the Securities Act that includes the Shares with respect to which the Warrant was exercised, it shall be a condition to any exercise of the rights under this Warrant that the Holder shall have confirmed to the satisfaction of the Company in writing, substantially in the form of Exhibit A-1, that the Shares so purchased are being acquired solely for the Holder's own account and not as a nominee for any other party, for investment and not with a view toward distribution or resale and that the Holder shall have confirmed such other matters related thereto as may be reasonably requested by the Company.
- (d) Securities Law Legend. The Securities shall (unless otherwise permitted by the provisions of this Warrant) be stamped or imprinted with a legend substantially similar to the following (in addition to any legend required by state securities laws):

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR UNDER THE SECURITIES LAWS OF CERTAIN STATES. THESE SECURITIES MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED EXCEPT AS PERMITTED UNDER THE ACT AND APPLICABLE STATE SECURITIES LAWS IN ACCORDANCE WITH APPLICABLE REGISTRATION REQUIREMENTS OR AN EXEMPTION THEREFROM. THE ISSUER OF THESE SECURITIES MAY REQUIRE AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE ISSUER THAT SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION OTHERWISE COMPLIES WITH THE ACT AND ANY APPLICABLE STATE SECURITIES LAWS. THIS CERTIFICATE MUST BE SURRENDERED TO THE COMPANY OR ITS TRANSFER AGENT AS A CONDITION PRECEDENT TO THE SALE, TRANSFER, PLEDGE OR HYPOTHECATION OF ANY INTEREST IN ANY OF THE SECURITIES REPRESENTED HEREBY.

(e) Market Stand-off Legend. The Shares issued upon exercise hereof shall also be stamped or imprinted with a legend in substantially the following form:

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO RESTRICTIONS ON TRANSFERABILITY AND RESALE, INCLUDING A LOCK-UP PERIOD IN THE EVENT OF A PUBLIC OFFERING, AS SET FORTH IN THE WARRANT PURSUANT TO WHICH THESE SHARES WERE ISSUED, A COPY OF WHICH MAY BE OBTAINED AT THE PRINCIPAL OFFICE OF THE COMPANY.

(f) *Instructions Regarding Transfer Restrictions*. The Holder consents to the Company making a notation on its records and giving instructions to any transfer agent in order to implement the restrictions on transfer established in this Section 5.

- (g) Removal of Legend. The legend referring to federal and state securities laws identified in Section 5(d) stamped on a certificate evidencing the Shares and the stock transfer instructions and record notations with respect to such securities shall be removed and the Company shall issue a certificate without such legend to the holder of such securities if (i) such securities are registered under the Securities Act, or (ii) such holder provides the Company with an opinion of counsel reasonably acceptable to the Company to the effect that a sale or transfer of such securities may be made without registration or qualification.
- **6.** Adjustments. Subject to the expiration of this Warrant pursuant to Section 8, the number and kind of shares purchasable hereunder and the Exercise Price therefor are subject to adjustment from time to time, as follows:
- (a) Merger or Reorganization. If at any time there shall be any reorganization, recapitalization, merger or consolidation (a "Reorganization") involving the Company (other than as otherwise provided for herein or as would cause the expiration of this Warrant under Section 8) in which shares of the Company's stock are converted into or exchanged for securities, cash or other property, then, as a part of such Reorganization, lawful provision shall be made so that the Holder shall thereafter be entitled to receive upon exercise of this Warrant, the kind and amount of securities, cash or other property of the successor corporation resulting from such Reorganization, equivalent in value to that which a holder of the Shares deliverable upon exercise of this Warrant would have been entitled in such Reorganization if the right to purchase the Shares hereunder had been exercised immediately prior to such Reorganization. In any such case, appropriate adjustment (as determined in good faith by the Board of Directors of the successor corporation) shall be made in the application of the provisions of this Warrant with respect to the rights and interests of the Holder after such Reorganization to the end that the provisions of this Warrant shall be applicable after the event, as near as reasonably may be, in relation to any shares or other securities deliverable after that event upon the exercise of this Warrant.
- (b) Reclassification of Shares. If the securities issuable upon exercise of this Warrant are changed into the same or a different number of securities of any other class or classes by reclassification, capital reorganization or otherwise (other than as otherwise provided for herein) (a "Reclassification"), then, in any such event, in lieu of the number of Shares which the Holder would otherwise have been entitled to receive, the Holder shall have the right thereafter to exercise this Warrant for a number of shares of such other class or classes of stock that a holder of the number of securities deliverable upon exercise of this Warrant immediately before that change would have been entitled to receive in such Reclassification, all subject to further adjustment as provided herein with respect to such other shares.
- (c) Subdivisions and Combinations. In the event that the outstanding shares or securities are subdivided (by stock split, by payment of a stock dividend or otherwise) into a greater number of shares of such securities, the number of Shares issuable upon exercise of the rights under this Warrant immediately prior to such subdivision shall, concurrently with the effectiveness of such subdivision, be proportionately increased, and the Exercise Price shall be proportionately decreased, and in the event that the outstanding securities are combined (by reclassification or otherwise) into a lesser number of shares of such securities, the number of Shares issuable upon exercise of the rights under this Warrant immediately prior to such combination shall, concurrently with the effectiveness of such combination, be proportionately decreased, and the Exercise Price shall be proportionately increased.
- (d) *Notice of Adjustments*. Upon any adjustment in accordance with this Section 6, the Company shall give notice thereof to the Holder, which notice shall state the event giving rise to the adjustment, the Exercise Price as adjusted and the number of securities or other property purchasable upon the exercise of the rights under this Warrant, setting forth in reasonable detail the method of calculation of each. The Company shall, upon the written request of any Holder, furnish or cause to be furnished to such Holder a certificate setting forth (i) such adjustments, (ii) the Exercise Price at the time in effect and (iii) the number of securities and the amount, if any, of other property that at the time would be received upon exercise of this Warrant.

- 7. Notification of Certain Events. Prior to the expiration of this Warrant pursuant to Section 8, in the event that the Company shall authorize:
- (a) the issuance of any dividend or other distribution on the capital stock of the Company (other than (i) dividends or distributions otherwise provided for in Section 6, (ii) repurchases of Shares issued to or held by employees, officers, directors or consultants of the Company or its subsidiaries upon termination of their employment or services pursuant to agreements providing for the right of said repurchase; or (iii) repurchases of Shares issued to or held by employees, officers, directors or consultants of the Company or its subsidiaries pursuant to rights of first refusal or first offer contained in agreements providing for such rights), whether in cash, property, stock or other securities;
 - (b) the voluntary liquidation, dissolution or winding up of the Company; or

(a) 5:00 p.m., Pacific time, on_

(c) any transaction resulting in the expiration of this Warrant pursuant to Section 8(b) or 8(c);

the Company shall send to the Holder of this Warrant at least 10 days prior written notice of the date on which a record shall be taken for any such provi issual

8. Expiration of the Warrant. This Warrant shall expire and shall no longer be exercisable as of the earlier of:
ble upon exercise of the rights under the Warrants.
sions set forth in this section may be shortened or waived prospectively or retrospectively by the consent of the holders of a majority of the Shares
end or distribution specified in clause (a) or the expected effective date of any such other event specified in clause (b) or (c), as applicable. The notice

(b) (i) the acquisition of the Company by another entity by means of any transaction or series of related transactions to which the Company is
a party (including, without limitation, any stock acquisition, reorganization, merger or consolidation, but excluding any sale of stock for capital raising
purposes and any transaction effected primarily for purposes of changing the Company's jurisdiction of incorporation) other than a transaction or series of
related transactions in which the holders of the voting securities of the Company outstanding immediately prior to such transaction or series of related
transactions retain, immediately after such transaction or series of transactions, as a result of shares in the Company held by such holders prior to such
transaction or series of transactions, at least a majority of the total voting power represented by the outstanding voting securities of the Company or such
other surviving or resulting entity (or if the Company or such other surviving or resulting entity is a wholly-owned subsidiary immediately following such
acquisition, its parent), or (ii) a sale, lease or other disposition of all or substantially all of the assets of the Company and its subsidiaries taken as a whole
by means of any transaction or series of related transactions, except where such sale, lease or other disposition is to a wholly-owned subsidiary of the
Company; or

(c) immediately prior to the closing of a firm commitment underwritten initial public offering pursuant to an effective registration statement filed under the Securities Act covering the offering and sale of the Company's Shares.

- 9. No Rights as a Stockholder. Nothing contained herein shall entitle the Holder to any rights as a stockholder of the Company or to be deemed the holder of any securities that may at any time be issuable on the exercise of the rights hereunder for any purpose nor shall anything contained herein be construed to confer upon the Holder, as such, any right to vote for the election of directors or upon any matter submitted to stockholders at any meeting thereof, or to give or withhold consent to any corporate action (whether upon any recapitalization, issuance of stock, reclassification of stock, change of par value or change of stock to no par value, consolidation, merger, conveyance or otherwise) or to receive notice of meetings, or to receive dividends or subscription rights or any other rights of a stockholder of the Company until the rights under the Warrant shall have been exercised and the Shares purchasable upon exercise of the rights hereunder shall have become deliverable as provided herein.
- 10. Market Stand-off. The Holder of this Warrant hereby agrees that such Holder shall not sell or otherwise transfer, make any short sale of, grant any option for the purchase of, or enter into any hedging or similar transaction with the same economic effect as a sale, of any securities of the Company held by the Holder (other than those included in the registration) during the one hundred and eighty (180) day period following the effective date of the registration statement for the Company's initial public offering filed under the Securities Act (or such other period as may be requested by the Company or an underwriter to accommodate regulatory restrictions on (i) the publication or other distribution of research reports and (ii) analyst recommendations and opinions, including, but not limited to, the restrictions contained in NASD Rule 2711(f)(4) or NYSE Rule 472(f)(4), or any successor provisions or amendments thereto. The obligations described in this section shall not apply to a registration relating solely to employee benefit plans on Form S-1 or Form S-8 or similar forms that may be promulgated in the future, or a registration relating solely to a transaction on Form S-4 or similar forms that may be promulgated in the future. The Company may impose stop-transfer instructions and may stamp each certificate with a legend as substantially set forth in Section 5(e) with respect to the securities subject to the foregoing restriction until the end of such one hundred and eighty (180) day (or other) period. The Holder agrees to execute a market stand-off agreement with the underwriters in the offering in customary form consistent with the provisions of this section
 - 11. Representations and Warranties of the Holder. By acceptance of this Warrant, the Holder represents and warrants to the Company as follows:
- (a) *No Registration*. The Holder understands that the Securities have not been, and will not be, registered under the Securities Act by reason of a specific exemption from the registration provisions of the Securities Act, the availability of which depends upon, among other things, the *bona fide* nature of the investment intent and the accuracy of the Holder's representations as expressed herein or otherwise made pursuant hereto.
- (b) *Investment Intent.* The Holder is acquiring the Securities for investment for its own account, not as a nominee or agent, and not with a view to, or for resale in connection with, any distribution thereof. The Holder has no present intention of selling, granting any participation in, or otherwise distributing the Securities, nor does it have any contract, undertaking, agreement or arrangement for the same.
- (c) Investment Experience. The Holder has substantial experience in evaluating and investing in private placement transactions of securities in companies similar to the Company, and has such knowledge and experience in financial or business matters so that it is capable of evaluating the merits and risks of its investment in the Company and protecting its own interests.
- (d) *Speculative Nature of Investment.* The Holder understands and acknowledges that the Company has a limited financial and operating history and that its investment in the Company is highly speculative and involves substantial risks. The Holder can bear the economic risk of its investment and is able, without impairing its financial condition, to hold the Securities for an indefinite period of time and to suffer a complete loss of its investment.

- (e) Access to Data. The Holder has had an opportunity to ask questions of officers of the Company, which questions were answered to its satisfaction. The Holder believes that it has received all the information that it considers necessary or appropriate for deciding whether to acquire the Securities. The Holder understands that any such discussions, as well as any information issued by the Company, were intended to describe certain aspects of the Company's business and prospects, but were not necessarily a thorough or exhaustive description. The Holder acknowledges that any business plans prepared by the Company have been, and continue to be, subject to change and that any projections included in such business plans or otherwise are necessarily speculative in nature, and it can be expected that some or all of the assumptions underlying the projections will not materialize or will vary significantly from actual results.
- (f) Accredited Investor. The Holder is an "accredited investor" within the meaning of Regulation D, Rule 501(a), promulgated by the Securities and Exchange Commission and agrees to submit to the Company such further assurances of such status as may be reasonably requested by the Company.
- (g) **Residency**. The residency of Holder (or, in the case of a partnership or corporation, such entity's principal place of business) is correctly set forth on the signature page hereto.
- (h) Restrictions on Resales. The Holder acknowledges that the Securities must be held indefinitely unless subsequently registered under the Securities Act or an exemption from such registration is available. The Holder is aware of the provisions of Rule 144 promulgated under the Securities Act, which permit resale of shares purchased in a private placement subject to the satisfaction of certain conditions, which may include, among other things, the availability of certain current public information about the Company; the resale occurring not less than a specified period after a party has purchased and paid for the security to be sold; the number of shares being sold during any three-month period not exceeding specified limitations; the sale being effected through a "broker's transaction," a transaction directly with a "market maker" or a "riskless principal transaction" (as those terms are defined in the Securities Act or the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder); and the filing of a Form 144 notice, if applicable. The Holder acknowledges and understands that the Company may not be satisfying the current public information requirement of Rule 144 at the time the Holder wishes to sell the Securities and that, in such event, the Holder may be precluded from selling the Securities under Rule 144 even if the other applicable requirements of Rule 144 have been satisfied. The Holder acknowledges that, in the event the applicable requirements of Rule 144 are not met, registration under the Securities Act or an exemption from registration will be required for any disposition of the Securities. The Holder understands that, although Rule 144 is not exclusive, the Securities and Exchange Commission has expressed its opinion that persons proposing to sell restricted securities received in a private offering other than in a registered offering or pursuant to Rule 144 will have a substantial burden of proof in establishing that an exemption from registration is ava
- (i) *No Public Market.* The Holder understands and acknowledges that no public market now exists for any of the securities issued by the Company and that the Company has made no assurances that a public market will ever exist for the Company's securities.
- (j) *Brokers and Finders*. The Holder has not engaged any brokers, finders or agents in connection with the Securities, and the Company has not incurred nor will incur, directly or indirectly, as a result of any action taken by the Holder, any liability for brokerage or finders' fees or agents' commissions or any similar charges in connection with the Securities.

- (k) *Legal Counsel*. The Holder has had the opportunity to review this Warrant, the exhibits and schedules attached hereto and the transactions contemplated by this Warrant with its own legal counsel. The Holder is not relying on any statements or representations of the Company or its agents for legal advice with respect to this investment or the transactions contemplated by this Warrant.
- (l) *Tax Advisors*. The Holder has reviewed with its own tax advisors the U.S. federal, state and local and non-U.S. tax consequences of this investment and the transactions contemplated by this Warrant. With respect to such matters, the Holder relies solely on any such advisors and not on any statements or representations of the Company or any of its agents, written or oral. The Holder understands that it (and not the Company) shall be responsible for its own tax liability that may arise as a result of this investment and the transactions contemplated by this Warrant.

12. Miscellaneous.

- (a) Amendments. Except as expressly provided herein, neither this Warrant nor any term hereof may be amended, waived, discharged or terminated other than by a written instrument referencing this Warrant and signed by the Company and the holders of warrants representing not less than a majority of the Shares issuable upon exercise of any and all outstanding Warrants, which does not need to include the consent of the Holder. Any amendment, waiver, discharge or termination effected in accordance with this Section 12(a) shall be binding upon each holder of the Warrants, each future holder of such Warrants and the Company; provided, however, that no special consideration or inducement may be given to any such holder in connection with such consent that is not given ratably to all such holders, and that such amendment must apply to all such holders equally and ratably in accordance with the number of shares issuable upon exercise of the Warrants. The Company shall promptly give notice to all holders of Warrants of any amendment effected in accordance with this Section 12(a).
- (b) Waivers. No waiver of any single breach or default shall be deemed a waiver of any other breach or default theretofore or thereafter occurring.
- (c) *Notices*. All notices and other communications required or permitted hereunder shall be in writing and shall be mailed by registered or certified mail, postage prepaid, sent by electronic mail (if to the Holder) or otherwise delivered by hand, messenger or courier service addressed:
- (i) if to the Holder, to the Holder at the Holder's address or electronic mail address as shown in the Company's records, as may be updated in accordance with the provisions hereof, or until any such Holder so furnishes an address or electronic mail address to the Company, then to and at the address or electronic mail address of the last holder of this Warrant for which the Company has contact information in its records; or
- (ii) if to the Company, to the attention of the Chief Executive Officer or Chief Financial Officer of the Company at the Company's address as shown on the signature page hereto, or at such other current address as the Company shall have furnished to the Holder.

Each such notice or other communication shall for all purposes of this Warrant be treated as effective or having been given (i) if delivered by hand, messenger or courier service, when delivered (or if sent via a nationally-recognized overnight courier service, freight prepaid, specifying next-business-day delivery, one business day after deposit with the courier), or (ii) if sent by mail, at the earlier of its receipt or five days after the same has been deposited in a regularly maintained receptacle for the deposit of the United States mail, addressed and mailed as aforesaid, or (iii) if sent via electronic mail, upon confirmation of

delivery when directed to the relevant electronic mail address, if sent during normal business hours of the recipient, or if not sent during normal business hours of the recipient, then on the recipient's next business day. In the event of any conflict between the Company's books and records and this Warrant or any notice delivered hereunder, the Company's books and records will control absent fraud or error.

- (d) Governing Law. This Warrant and all actions arising out of or in connection with this Warrant shall be governed by and construed in accordance with the laws of the State of California, without regard to the conflicts of law provisions of the State of California, or of any other state.
- (e) *Jurisdiction and Venue*. Each of the Holder and the Company irrevocably consents to the exclusive jurisdiction of, and venue in, the state courts in Santa Clara County in the State of California (or in the event of exclusive federal jurisdiction, the courts of the Northern District of California), in connection with any matter based upon or arising out of this Warrant or the matters contemplated herein, and agrees that process may be served upon them in any manner authorized by the laws of the State of California for such persons.
- (f) *Titles and Subtitles*. The titles and subtitles used in this Warrant are used for convenience only and are not to be considered in construing or interpreting this Warrant. All references in this Warrant to sections, paragraphs and exhibits shall, unless otherwise provided, refer to sections and paragraphs hereof and exhibits attached hereto.
- (g) Severability. If any provision of this Warrant becomes or is declared by a court of competent jurisdiction to be illegal, unenforceable or void, portions of such provision, or such provision in its entirety, to the extent necessary, shall be severed from this Warrant, and such illegal, unenforceable or void provision shall be replaced with a valid and enforceable provision that will achieve, to the extent possible, the same economic, business and other purposes of the illegal, unenforceable or void provision. The balance of this Warrant shall be enforceable in accordance with its terms.
- (h) Waiver of Jury Trial; Judicial Reference EACH OF THE HOLDER AND THE COMPANY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATED TO THIS WARRANT. If the waiver of jury trial set forth in this paragraph is not enforceable, then any claim or cause of action arising out of or relating to this Warrant shall be settled by judicial reference pursuant to California Code of Civil Procedure Section 638 et seq. before a referee sitting without a jury, such referee to be mutually acceptable to the parties or, if no agreement is reached, by a referee appointed by the Presiding Judge of the California Superior Court for Santa Clara County. This paragraph shall not restrict the Holder or the Company from exercising remedies under the Uniform Commercial Code or from exercising pre-judgment remedies under applicable law.
- (i) California Corporate Securities Law. THE SALE OF THE SECURITIES THAT ARE THE SUBJECT OF THIS WARRANT HAS NOT BEEN QUALIFIED WITH THE COMMISSIONER OF CORPORATIONS OF THE STATE OF CALIFORNIA AND THE ISSUANCE OF SUCH SECURITIES OR THE PAYMENT OR RECEIPT OF ANY PART OF THE CONSIDERATION THEREFOR PRIOR TO SUCH QUALIFICATION IS UNLAWFUL, UNLESS THE SALE OF SECURITIES IS EXEMPT FROM QUALIFICATION BY SECTION 25100, 25102, OR 25105 OF THE CALIFORNIA CORPORATIONS CODE. THE RIGHTS OF ALL PARTIES TO THIS WARRANT ARE EXPRESSLY CONDITIONED UPON THE QUALIFICATION BEING OBTAINED, UNLESS THE SALE IS SO EXEMPT.

- (j) Rights and Obligations Survive Exercise of the Warrant. Except as otherwise provided herein, the rights and obligations of the Company and the Holder under this Warrant shall survive exercise of this Warrant.
- (k) *Entire Agreement.* Except as expressly set forth herein, this Warrant (including the exhibits attached hereto) constitutes the entire agreement and understanding of the Company and the Holder with respect to the subject matter hereof and supersede all prior agreements and understandings relating to the subject matter hereof.

(signature page follows)

The Company and the Holder sign this Warrant as of the date stated on the first page.

	GROVE COLLABORATIVE, INC. (FKA EPANTRY, LLC)	
	Ву:	
	Name:	
	Title:	
	Address:	
AGREED AND ACKNOWLEDGED,		
Ву:		
Name:		
Title:		
Address:		

EXHIBIT A

NOTICE OF EXERCISE

TO: GROVE COLLABORATIVE, INC. (the "Company")

Atte	ntion:	President	
(1)	Exerc	ise. The undersigned elects to	purchase the following pursuant to the terms of the attached warrant:
		Number of shares:	
		Type of security:	
(2)	Metho	od of Exercise. The undersign	ned elects to exercise the attached warrant pursuant to:
		A cash payment, and any.	tenders herewith payment of the purchase price for such shares in full, together with all applicable transfer taxes, if
(3)	Condi	tional Exercise. Is this a con	ditional exercise pursuant to Section 2(c):
		Yes	No
		If "Yes," indicate the applic	able condition:
(4)	Stock	Certificate. Please issue a ce	rtificate or certificates representing the shares in the name of:
		The undersigned	
		e	
		_	
(5)	Unexe	rcised Portion of the Warra	ant. Please issue a new warrant for the unexercised portion of the attached warrant in the name of:
		The undersigned	
		Other—Name:	
		_	
		Not applicable	

- (6) Investment Intent. The undersigned represents and warrants that the aforesaid shares are being acquired for investment for its own account, not as a nominee or agent, and not with a view to, or for resale in connection with, the distribution thereof, and that the undersigned has no present intention of selling, granting any participation in, or otherwise distributing the shares, nor does it have any contract, undertaking, agreement or arrangement for the same, and all representations and warranties of the undersigned set forth in Section 11 of the attached warrant are true and correct as of the date hereof.
- (7) **Investment Representation Statement and Market Stand-Off Agreement.** The undersigned has executed, and delivers herewith, an Investment Representation Statement and Market Stand-Off Agreement in a form substantially similar to the form attached to the warrant as Exhibit A-1.
- (8) Consent to Receipt of Electronic Notice. Subject to the limitations set forth in Delaware General Corporation Law §232(e), the undersigned consents to the delivery of any notice to stockholders given by the Company under the Delaware General Corporation Law or the Company's certificate of incorporation or bylaws by (i) facsimile telecommunication to the facsimile number provided below (or to any other facsimile number for the undersigned in the Company's records), (ii) electronic mail to the electronic mail address provided below (or to any other electronic mail address for the undersigned in the Company's records), (iii) posting on an electronic network together with separate notice to the undersigned of such specific posting or (iv) any other form of electronic transmission (as defined in the Delaware General Corporation Law) directed to the undersigned. This consent may be revoked by the undersigned by written notice to the Company and may be deemed revoked in the circumstances specified in Delaware General Corporation Law §232.

(Print name of the warrant holder)
(Signature)
(Name and title of signatory, if applicable
(Date)
(Fax number)
(1 ax number)

EXHIBIT A-1

INVESTMENT REPRESENTATION STATEMENT AND MARKET STAND-OFF AGREEMENT

INVESTOR:	
COMPANY:	GROVE COLLABORATIVE, INC.
SECURITIES:	THE WARRANT ISSUED ON [INSERT DATE] (THE "WARRANT") AND THE SECURITIES ISSUED OR ISSUABLE UPON EXERCISE THEREOF
DATE:	

In connection with the purchase or acquisition of the above-listed Securities, the undersigned Investor represents and warrants to, and agrees with, the Company as follows:

- 1. **No Registration.** The Investor understands that the Securities have not been, and will not be, registered under the Securities Act of 1933, as amended (the "Securities Act"), by reason of a specific exemption from the registration provisions of the Securities Act, the availability of which depends upon, among other things, the *bona fide* nature of the investment intent and the accuracy of the Investor's representations as expressed herein or otherwise made pursuant hereto.
- 2. **Investment Intent.** The Investor is acquiring the Securities for investment for its own account, not as a nominee or agent, and not with a view to, or for resale in connection with, any distribution thereof. The Investor has no present intention of selling, granting any participation in, or otherwise distributing the Securities, nor does it have any contract, undertaking, agreement or arrangement for the same.
- 3. **Investment Experience.** The Investor has substantial experience in evaluating and investing in private placement transactions of securities in companies similar to the Company, and has such knowledge and experience in financial or business matters so that it is capable of evaluating the merits and risks of its investment in the Company and protecting its own interests.
- 4. **Speculative Nature of Investment.** The Investor understands and acknowledges that the Company has a limited financial and operating history and that its investment in the Company is highly speculative and involves substantial risks. The Investor can bear the economic risk of its investment and is able, without impairing its financial condition, to hold the Securities for an indefinite period of time and to suffer a complete loss of its investment.
- 5. Access to Data. The Investor has had an opportunity to ask questions of officers of the Company, which questions were answered to its satisfaction. The Investor believes that it has received all the information that it considers necessary or appropriate for deciding whether to acquire the Securities. The Investor understands that any such discussions, as well as any information issued by the Company, were intended to describe certain aspects of the Company's business and prospects, but were not necessarily a thorough or exhaustive description. The Investor acknowledges that any business plans prepared by the Company have been, and continue to be, subject to change and that any projections included in such business plans or otherwise are necessarily speculative in nature, and it can be expected that some or all of the assumptions underlying the projections will not materialize or will vary significantly from actual results.

- 6. Accredited Investor. The Investor is an "accredited investor" within the meaning of Regulation D, Rule 501(a), promulgated by the Securities and Exchange Commission and agrees to submit to the Company such further assurances of such status as may be reasonably requested by the Company.
- 7. **Residency.** The residency of the Investor (or, in the case of a partnership or corporation, such entity's principal place of business) is correctly set forth on the signature page hereto.
- 8. **Restrictions on Resales.** The Investor acknowledges that the Securities must be held indefinitely unless subsequently registered under the Securities Act or an exemption from such registration is available. The Investor is aware of the provisions of Rule 144 promulgated under the Securities Act, which permit resale of shares purchased in a private placement subject to the satisfaction of certain conditions, which may include, among other things, the availability of certain current public information about the Company; the resale occurring not less than a specified period after a party has purchased and paid for the security to be sold; the number of shares being sold during any three-month period not exceeding specified limitations; the sale being effected through a "broker's transaction," a transaction directly with a "market maker" or a "riskless principal transaction" (as those terms are defined in the Securities Act or the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder); and the filing of a Form 144 notice, if applicable. The Investor acknowledges and understands that the Company may not be satisfying the current public information requirement of Rule 144 at the time the Investor wishes to sell the Securities and that, in such event, the Investor may be precluded from selling the Securities under Rule 144 even if the other applicable requirements of Rule 144 have been satisfied. The Investor understands and acknowledges that, in the event the applicable requirements of Rule 144 are not met, registration under the Securities Act or an exemption from registration will be required for any disposition of the Securities. The Investor understands that, although Rule 144 is not exclusive, the Securities and Exchange Commission has expressed its opinion that persons proposing to sell restricted securities received in a private offering other than in a registered offering or pursuant to Rule 144 will have a substantial burden of proof in establishing that an exe
- 9. **No Public Market.** The Holder understands and acknowledges that no public market now exists for any of the securities issued by the Company and that the Company has made no assurances that a public market will ever exist for the Company's securities.
- 10. **Brokers and Finders.** The Investor has not engaged any brokers, finders or agents in connection with the Securities, and the Company has not incurred nor will incur, directly or indirectly, as a result of any action taken by the Investor, any liability for brokerage or finders' fees or agents' commissions or any similar charges in connection with the Securities.
- 11. **Legal Counsel.** The Investor has had the opportunity to review the Warrant, the exhibits and schedules attached thereto and the transactions contemplated by the Warrant with its own legal counsel. The Investor is not relying on any statements or representations of the Company or its agents for legal advice with respect to this investment or the transactions contemplated by the Warrant.
- 12. **Tax Advisors.** The Investor has reviewed with its own tax advisors the U.S. federal, state and local and non-U.S. tax consequences of this investment and the transactions contemplated by the Warrant. With respect to such matters, the Investor relies solely on such advisors and not on any statements or representations of the Company or any of its agents, written or oral. The Investor understands that it (and not the Company) shall be responsible for its own tax liability that may arise as a result of this investment or the transactions contemplated by the Warrant.

13. **Market Stand-off.** The Investor agrees that the Investor shall not sell or otherwise transfer, make any short sale of, grant any option for the purchase of, or enter into any hedging or similar transaction with the same economic effect as a sale, of any securities of the Company held by the Investor (other than those included in the registration) during the one hundred and eighty (180) day period following the effective date of the registration statement for the Company's initial public offering filed under the Securities Act (or such other period as may be requested by the Company or an underwriter to accommodate regulatory restrictions on (i) the publication or other distribution of research reports and (ii) analyst recommendations and opinions, including, but not limited to, the restrictions contained in NASD Rule 2711(f)(4) or NYSE Rule 472(f)(4), or any successor provisions or amendments thereto). The obligations described in this section shall not apply to a registration relating solely to employee benefit plans on Form S-1 or Form S-8 or similar forms that may be promulgated in the future, or a registration relating solely to a transaction on Form S-4 or similar forms that may be promulgated in the future. The Company may impose stop-transfer instructions and may stamp each certificate with a legend with respect to securities subject to the foregoing restriction until the end of such one hundred and eighty (180) day (or other) period. The Investor agrees to execute a market stand-off agreement with the relevant underwriters in customary form consistent with the provisions of this section.

(signature page follows)

The Investor is signing this Investment Representation Statement and Market Stand-Off Agreement on the date first written above.

(Print)	name of the investor)
	(Signature)
(Name and titl	le of signatory, if applicable
(,	Street address)

INVESTOR

EXHIBIT B

ASSIGNMENT FORM

ASSIGNOR:	
COMPANY:	GROVE COLLABORATIVE, INC.
WARRANT:	THE WARRANT TO PURCHASE COMMON SHARES ISSUED ON [INSERT DATE] (THE "WARRANT")
DATE:	
the rights	t. The undersigned registered holder of the Warrant ("Assignor") assigns and transfers to the assignee named below ("Assignee") all of f Assignor under the Warrant, with respect to the number of shares set forth below: ame of Assignee:
	ddress of Assignee:
	umber of Shares Assigned:
	revocably constitute and appoint as attorney to make such transfer on the books of Grove Collaborative, Inc., maintained for the th full power of substitution in the premises.

- (2) **Obligations of Assignee.** Assignee agrees to take and hold the Warrant and any shares of stock to be issued upon exercise of the rights thereunder (the "Securities") subject to, and to be bound by, the terms and conditions set forth in the Warrant to the same extent as if Assignee were the original holder thereof.
- (3) **Investment Intent.** Assignee represents and warrants that the Securities are being acquired for investment for its own account, not as a nominee or agent, and not with a view to, or for resale in connection with, the distribution thereof, and that Assignee has no present intention of selling, granting any participation in, or otherwise distributing the shares, nor does it have any contract, undertaking, agreement or arrangement for the same, and all representations and warranties set forth in Section 11 of the Warrant are true and correct as to Assignee as of the date hereof.
- (4) **Investment Representation Statement and Market Stand-Off Agreement.** Assignee has executed, and delivers herewith, an Investment Representation Statement and Market Stand-Off Agreement in a form substantially similar to the form attached to the Warrant as Exhibit A-1.

Assignor and Assignee are signing this Assignment Form on the date first set forth above.

ASSIGNOR

ASSIGNEE

(Print name of Assignor)

(Signature of Assignor)

(Print name of signatory, if applicable)

(Print name of signatory, if applicable)

(Print title of signatory, if applicable)

(Print title of signatory, if applicable)

Address:

Address:

THIS WARRANT AND THE SECURITIES ISSUABLE UPON THE EXERCISE HEREOF HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED. THEY MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED, HYPOTHECATED, OR OTHERWISE TRANSFERRED EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY THAT REGISTRATION IS NOT REQUIRED UNDER SUCH ACT OR UNLESS SOLD PURSUANT TO RULE 144 UNDER SUCH ACT.

Date of Issuance Vo	oid after
GROVE COLLABORATIVE, INC. WARRANT TO PURCHASE SHARES OF COMMON STOCK	
For good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, this Warrant is issued to, or its assigns (the "Holder") by Grove Collaborative, Inc., a Delaware corporation (the "Company").	
1. Purchase of Shares.	
(a) <u>Number of Shares</u> . Subject to the terms and conditions set forth herein, the Holder is entitled, upon surrender of this Warrant at the principal office of the Company (or at such other place as the Company shall notify the Holder in writing), to purchase from the Company up to fully paid and nonassessable shares of the Company's Common Stock, par value \$0.0001 per share (the "Common Stock").	
(b) <u>Exercise Price</u> . The exercise price for the shares of Common Stock issuable pursuant to this Section 1 (the "Shares") shall be \$ share (the "Exercise Price"). The Shares and the Exercise Price shall be subject to adjustment pursuant to Section 9 hereof.	per
2. Exercise Period.	
(a) Notwithstanding anything to the contrary contained herein, this Warrant shall only be exercisable with respect to those Shares which have vested as of the date of exercise. One eight (1/8th) of the Shares shall vest and become exercisable each quarter during the term of this Warrant, beginning on, and continuing on the same day each quarter thereafter, such that all of the Shares issuable pursuant to this Warrant shall become fully vested as of the date years from,; provided that if that certain Agreement by and between Company and Holder dated as of, is terminated, the Shares will cease vesting and the Warrant shall only be exercisable with respect to those Shares which have vested as of such termination. Fractional share numbers resulting from the calculations required pursuant to the foregoing continues shall be rounded down to the powerest whole anyther of shares.	nt n the pect

(b) This Warrant shall be exercisable, in whole or in part, during the term commencing on,	and ending at	_p.m.
PST on, (the "Exercise Period"); provided, however, that this Warrant shall no longer be exercisable an	id become null and voice	d upon the
earlier of (a) the consummation of the Company's sale of its Common Stock or other securities in the Company's first und	derwritten public offerie	ng
pursuant to an effective registration statement under the Securities Act of 1933, as amended (other than a registration state	ement relating either to	sale of
securities to employees of the Company pursuant to its stock option, stock purchase or similar plan or a SEC Rule 145 tra	nsaction) (an "Initial Pi	ublic
Offering"), (b) the consummation of a Deemed Liquidation Event, as such term is defined in the Company's current Ame	nded and Restated Cert	ificate of
Incorporation on file with the Secretary of State of the State of Delaware, (c)months after the Consulting Agreement te	rminates or (d) five (5)	years
after the date of issuance. For purposes of this Warrant, any of the transactions described in subsection (b) shall be referre	d to herein as a "Corpo	rate
Transaction". In the event of an Initial Public Offering or Corporate Transaction, the Company shall notify the Holder at I	east() days pric	or to the
consummation of such Initial Public Offering or Corporate Transaction.		

3. Method of Exercise.

- (a) While this Warrant remains outstanding and exercisable in accordance with Section 2 above, the Holder may exercise, in whole or in part, the purchase rights evidenced hereby. Such exercise shall be effected by:
- (i) the surrender of the Warrant, together with a duly executed copy of the Notice of Exercise attached hereto, to the Secretary of the Company at its principal office (or at such other place as the Company shall notify the Holder in writing); and
 - (ii) the payment to the Company of an amount equal to the aggregate Exercise Price for the number of Shares being purchased.
- (b) Each exercise of this Warrant shall be deemed to have been effected immediately prior to the close of business on the day on which this Warrant is surrendered to the Company as provided in Section 3(a) above. At such time, the person or persons in whose name or names any certificate for the Shares shall be issuable upon such exercise as provided in Section 3(c) below shall be deemed to have become the holder or holders of record of the Shares represented by such certificate.
- (c) As soon as practicable after the exercise of this Warrant in whole or in part the Company at its expense will cause to be issued in the name of, and delivered to, the Holder, or as such Holder (upon payment by such Holder of any applicable transfer taxes) may direct:
 - (i) a certificate or certificates for the number of Shares to which such Holder shall be entitled, and
- (ii) in case such exercise is in part only, a new warrant or warrants (dated the date hereof) of like tenor, calling in the aggregate on the face or faces thereof for the number of Shares equal to the number of such Shares described in this Warrant minus the number of such Shares purchased by the Holder upon all exercises made in accordance with Section 3(a) above or Section 4 below.

(d) Notwithstanding the provisions of Section 2, if the holder has not exercised this Warrant prior to the closing of a Corporate Transaction or an Initial Public Offering, this Warrant shall automatically be deemed to be exercised in full in the manner set forth in Section 4, without any further action on behalf of the Holder immediately prior to such closing.

4. <u>Net Exercise</u>. In lieu of exercising this Warrant for cash, the Holder may elect to receive shares equal to the value of this Warrant (or the portion thereof being exercised) by surrender of this Warrant at the principal office of the Company together with notice of such election (a "Net Exercise"). A Holder who Net Exercises shall have the rights described in Sections 3(b) and 3(c) hereof, and the Company shall issue to such Holder a number of Shares computed using the following formula:

$$X \square \frac{Y(A-B)}{A}$$

Where

X = The number of Shares to be issued to the Holder.

Y = The number of Shares purchasable under this Warrant or, if only a portion of the Warrant is being exercised, the portion of the Warrant being exercised (at the date of such calculation).

A = The fair market value of one (1) Share (at the date of such calculation).

B = The Exercise Price (as adjusted to the date of such calculation).

For purposes of this Section 4, the fair market value of a Share shall mean the average of the closing prices of the Shares (or equivalent shares of Common Stock underlying this Warrant) quoted on any exchange or electronic securities market on which the Shares (or equivalent shares of Common Stock underlying the Warrant) are listed, as published in *The Wall Street Journal* for the thirty (30) trading days prior to the date of determination of fair market value (or such shorter period of time during which such Shares were traded on such exchange). In the event that this Warrant is exercised pursuant to this Section 4 in connection with the Initial Public Offering, the fair market value per Share shall be the per share offering price to the public of the Initial Public Offering. If the Shares are not traded on an exchange or an electronic securities market, the fair market value shall be the price per Share that the Company could obtain from a willing buyer for Shares sold by the Company from authorized but unissued Shares, as such prices shall be determined in good faith by the Company's Board of Directors.

- 5. Representations and Warranties of the Company. In connection with the transactions provided for herein, the Company hereby represents and warrants to the Holder that:
- (a) <u>Organization, Good Standing, and Qualification</u>. The Company is a corporation duly organized, validly existing, and in good standing under the laws of the State of Delaware and has all requisite corporate power and authority to carry on its business as now conducted. The Company is duly qualified to transact business and is in good standing in each jurisdiction in which the failure to so qualify would have a material adverse effect on its business or properties.
- (b) <u>Authorization</u>. Except as may be limited by applicable bankruptcy, insolvency, reorganization or similar laws relating to or affecting the enforcement of creditors' rights, all corporate action has been taken on the part of the Company, its officers, directors, and stockholders necessary for the authorization, execution and delivery of this Warrant. The Company has taken all corporate action required to make all the obligations of the Company reflected in the provisions of this Warrant the valid and enforceable obligations they purport to be. The issuance of this Warrant will not be subject to preemptive rights of any stockholders of the Company. The Company has authorized sufficient shares of Common Stock to allow for the exercise of this Warrant.
- (c) <u>Valid Issuance of Common Stock</u>. The Shares, when issued, sold, and delivered in accordance with the terms of the Warrants for the consideration expressed therein, will be duly and validly issued, fully paid and nonassessable and, based in part upon the representations and warranties of the Holders in this Warrant, will be issued in compliance with all applicable federal and state securities laws.
- 6. <u>Representations and Warranties of the Holder</u>. In connection with the transactions provided for herein, the Holder hereby represents and warrants to the Company that:
- (a) <u>Authorization</u>. Holder represents that it has full power and authority to enter into this Warrant. This Warrant constitutes the Holder's valid and legally binding obligation, enforceable in accordance with its terms, except as may be limited by (i) applicable bankruptcy, insolvency, reorganization, or similar laws relating to or affecting the enforcement of creditors' rights and (ii) laws relating to the availability of specific performance, injunctive relief or other equitable remedies.
- (b) <u>Purchase Entirely for Own Account.</u> The Holder acknowledges that this Warrant is entered into by the Company in reliance upon such Holder's representation to the Company that the Warrant and the Shares (collectively, the "Securities") will be acquired for investment for the Holder's own account, not as a nominee or agent, and not with a view to the resale or distribution of any part thereof, and that the Holder has no present intention of selling, granting any participation in or otherwise distributing the same. By acknowledging this Warrant, the Holder further represents that the Holder does not have any contract, undertaking, agreement, or arrangement with any person to sell, transfer or grant participations to such person or to any third person, with respect to the Securities.
- (c) <u>Disclosure of Information</u>. The Holder acknowledges that it has received all the information it considers necessary or appropriate for deciding whether to acquire the Securities. The Holder further represents that it has had an opportunity to ask questions and receive answers from the Company regarding the terms and conditions of the offering of the Securities.

- (d) Investment Experience. The Holder is an investor in securities of companies in the development stage and acknowledges that it is able to fend for itself, can bear the economic risk of its investment, and has such knowledge and experience in financial or business matters that it is capable of evaluating the merits and risks of the investment in the Securities. If other than an individual, the Holder also represents it has not been organized solely for the purpose of acquiring the Securities.
- (e) <u>Accredited Investor</u>. The Holder is an "accredited investor" within the meaning of Rule 501 of Regulation D, as presently in effect, as promulgated by the Securities and Exchange Commission (the "SEC") under the Act.
- (f) <u>Restricted Securities</u>. The Holder understands that the Securities are characterized as "restricted securities" under the federal securities laws inasmuch as they are being acquired from the Company in a transaction not involving a public offering and that under such laws and applicable regulations such securities may be resold without registration under the Act, only in certain limited circumstances. In this connection, the Holder represents that it is familiar with Rule 144, as presently in effect, as promulgated by the SEC under the Act ("Rule 144"), and understands the resale limitations imposed thereby and by the Act.
- (g) <u>Further Limitations on Disposition</u>. Without in any way limiting the representations set forth above, the Holder further agrees not to make any disposition of all or any portion of the Securities unless and until the transferee has agreed in writing for the benefit of the Company to be bound by the terms of this Warrant, including, without limitation, this Section 6, Section 23, and:
- (i) there is then in effect a registration statement under the Act covering such proposed disposition and such disposition is made in accordance with such registration statement;
- (ii) the Holder shall have notified the Company of the proposed disposition and shall have furnished the Company with a detailed statement of the circumstances surrounding the proposed disposition, and if reasonably requested by the Company, the Holder shall have furnished the Company with an opinion of counsel, reasonably satisfactory to the Company, that such disposition will not require registration of such shares under the Act. It is agreed that the Company will not require opinions of counsel for transactions made pursuant to Rule 144 except in extraordinary circumstances; or
- (iii) if other than an individual, the Holder shall not make any disposition to any of the Company's competitors as such is reasonably determined by the Company.
 - (h) Legends. It is understood that the Securities may bear the following legend:

"THESE SECURITIES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED. THEY MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED, HYPOTHECATED, OR OTHERWISE TRANSFERRED EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY THAT REGISTRATION IS NOT REQUIRED UNDER SUCH ACT OR UNLESS SOLD PURSUANT TO RULE 144 UNDER SUCH ACT."

7. Covenants of the Company.

- (a) Notices of Record Date. In the event of any taking by the Company of a record of the holders of any class of securities for the purpose of determining the holders thereof who are entitled to receive any dividend (other than a cash dividend which is the same as cash dividends paid in previous quarters and stock dividends) or other distribution, the Company shall provide the Holder, at least _____ (___) days prior to such record date, a notice specifying the date on which any such record is to be taken for the purpose of such dividend or distribution.
- (b) Covenants as to Exercise Shares. The Company covenants and agrees that all Shares that may be issued upon the exercise of the rights represented by this Warrant will, upon issuance in accordance with the terms hereof, be validly issued and outstanding, fully paid and nonassessable, and free from all taxes, liens and charges with respect to the issuance thereof. The Company further covenants and agrees that the Company will at all times during the Exercise Period, have authorized and reserved, free from preemptive rights, a sufficient number of shares of Common Stock to provide for the exercise of the rights represented by this Warrant. If at any time during the Exercise Period the number of authorized but unissued shares of Common Stock shall not be sufficient to permit exercise of this Warrant, the Company will take such corporate action as may, in the opinion of its counsel, be necessary to increase its authorized but unissued shares of Common Stock to such number of shares as shall be sufficient for such purposes.
- 8. <u>Adjustment of Exercise Price and Number of Shares</u>. The number and kind of Shares purchasable upon exercise of this Warrant and the Exercise Price shall be subject to adjustment from time to time as follows:
- (a) <u>Subdivisions</u>, <u>Combinations and Other Issuances</u>. If the Company shall at any time after the issuance but prior to the expiration of this Warrant subdivide its Common Stock, by split-up or otherwise, or combine its Common Stock, or issue additional shares of its Preferred Stock or Common Stock as a dividend with respect to any shares of its Common Stock, the number of Shares issuable on the exercise of this Warrant shall forthwith be proportionately increased in the case of a subdivision or stock dividend, or proportionately decreased in the case of a combination. Appropriate adjustments shall also be made to the Exercise Price payable per share, but the aggregate Exercise Price payable for the total number of Shares purchasable under this Warrant (as adjusted) shall remain the same. Any adjustment under this Section 9(a) shall become effective at the close of business on the date the subdivision or combination becomes effective, or as of the record date of such dividend, or in the event that no record date is fixed, upon the making of such dividend.

- (b) Reclassification, Reorganization and Consolidation. In case of any reclassification, capital reorganization or change in the capital stock of the Company (other than as a result of a subdivision, combination or stock dividend provided for in Section 9(a) above), then, as a condition of such reclassification, reorganization or change, lawful provision shall be made, and duly executed documents evidencing the same from the Company or its successor shall be delivered to the Holder, so that the Holder shall have the right at any time prior to the expiration of this Warrant to purchase, at a total price equal to that payable upon the exercise of this Warrant, the kind and amount of shares of stock and other securities or property receivable in connection with such reclassification, reorganization or change by a holder of the same number and type of securities as were purchasable as Shares by the Holder immediately prior to such reclassification, reorganization or change. In any such case appropriate provisions shall be made with respect to the rights and interest of the Holder so that the provisions hereof shall thereafter be applicable with respect to any shares of stock or other securities or property deliverable upon exercise hereof, and appropriate adjustments shall be made to the Exercise Price per Share payable hereunder, provided the aggregate Exercise Price shall remain the same.
- (c) Notice of Adjustment. When any adjustment is required to be made in the number or kind of shares purchasable upon exercise of the Warrant, or in the Exercise Price, the Company shall promptly notify the Holder of such event and of the number of Shares or other securities or property thereafter purchasable upon exercise of this Warrant.
- 9. No Fractional Shares or Scrip. No fractional shares or scrip representing fractional shares shall be issued upon the exercise of this Warrant, but in lieu of such fractional shares the Company shall make a cash payment therefor on the basis of the Exercise Price then in effect.
- 10. No Stockholder Rights. Prior to exercise of this Warrant, the Holder shall not be entitled to any rights of a stockholder with respect to the Shares, including (without limitation) the right to vote such Shares, receive dividends or other distributions thereon, exercise preemptive rights or be notified of stockholder meetings, and, except as otherwise provided in this Warrant, such Holder shall not be entitled to any stockholder notice or other communication concerning the business or affairs of the Company.
- 11. <u>Transfer of Warrant</u>. Subject to compliance with applicable federal and state securities laws and any other contractual restrictions between the Company and the Holder contained herein, this Warrant and all rights hereunder are transferable in whole or in part by the Holder to any person or entity upon written notice to the Company. Within a reasonable time after the Company's receipt of an executed Assignment Form in the form attached hereto, the transfer shall be recorded on the books of the Company upon the surrender of this Warrant, properly endorsed, to the Company at its principal offices, and the payment to the Company of all transfer taxes and other governmental charges imposed on such transfer. In the event of a partial transfer, the Company shall issue to the new holders one (1) or more appropriate new warrants. Notwithstanding the foregoing, the Shares shall also be subject to the transfer restrictions of Section 36 of the Company's Bylaws.

- 12. <u>Governing Law</u>. This Warrant shall be governed by and construed under the laws of the State of California as applied to agreements among California residents, made and to be performed entirely within the State of California.
- 13. <u>Successors and Assigns</u>. The terms and provisions of this Warrant shall inure to the benefit of, and be binding upon, the Company and the holders hereof and their respective successors and assigns.
- 14. <u>Counterparts</u>. This Warrant may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 15. <u>Titles and Subtitles</u>. The titles and subtitles used in this Warrant are used for convenience only and are not to be considered in construing or interpreting this Warrant.
- 16. Notices. All notices and other communications given or made pursuant hereto shall be in writing and shall be deemed effectively given: (a) upon personal delivery to the party to be notified, (b) when sent by confirmed electronic mail or facsimile if sent during normal business hours of the recipient, and if not so confirmed, then on the next business day, (c) ____(_) days after having been sent by registered or certified mail, return receipt requested, postage prepaid, or (d) one (1) day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written verification of receipt. All communications shall be sent to the respective parties at the following addresses (or at such other addresses as shall be specified by notice given in accordance with this Section 16):

If to the Company:

with a copy to:

Grove Collaborative, Inc. 1301 Sansome San Francisco, CA 94111 Attention: Stuart Landesberg, Chief Executive Officer

Gunderson Dettmer Stough Villeneuve Franklin & Hachigian, LLP 550 Allerton Street
Redwood City, CA 94063
Telephone: (650) 463-5340
E-Mail: igaviria@gunder.com

Attention: Ivan Gaviria

If to Holder:

At the address shown on the signature page hereto.

- 17. Finder's Fee. Each party represents that it neither is or will be obligated for any finder's fee or commission in connection with this transaction. The Holder agrees to indemnify and to hold harmless the Company from any liability for any commission or compensation in the nature of a finder's fee (and the costs and expenses of defending against such liability or asserted liability) for which the Holder or any of its officers, partners, employees or representatives is responsible. The Company agrees to indemnify and hold harmless the Holder from any liability for any commission or compensation in the nature of a finder's fee (and the costs and expenses of defending against such liability or asserted liability) for which the Company or any of its officers, employees or representatives is responsible.
- 18. Expenses. If any action at law or in equity is necessary to enforce or interpret the terms of this Warrant, the prevailing party shall be entitled to reasonable attorneys' fees, costs and necessary disbursements in addition to any other relief to which such party may be entitled.
- 19. Entire Agreement; Amendments and Waivers This Warrant and any other documents delivered pursuant hereto constitute the full and entire understanding and agreement between the parties with regard to the subjects hereof and thereof. Nonetheless, any term of this Warrant may be amended and the observance of any term of this Warrant may be waived (either generally or in a particular instance and either retroactively or prospectively), with the written consent of the Company and the Holder; or if this Warrant has been assigned in part, by the holders or rights to purchase a majority of the shares originally issuable pursuant to this Warrant.
- 20. Severability. If any provision of this Warrant is held to be unenforceable under applicable law, such provision shall be excluded from this Warrant and the balance of the Warrant shall be interpreted as if such provision were so excluded and shall be enforceable in accordance with its terms.
- 21. "Market Stand-Off" Agreement. The Holder hereby agrees that it will not, without the prior written consent of the managing underwriter, during the period commencing on the date of the final prospectus relating to the Company's Initial Public Offering and ending on the date specified by the Company and the managing underwriter (such period not to exceed ______ (____) days) (i) lend, offer, pledge, sell, contract to sell, sell any option or contract to purchase, purchase any option or contract to sell, grant any option, right or warrant to purchase, or otherwise transfer or dispose of, directly or indirectly, any shares of the Company's capital stock acquired through the exercise of this Warrant, or (ii) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of the Company's capital stock acquired through the exercise of this Warrant, whether any such transaction described in clause (i) or (ii) above is to be settled by delivery of securities, in cash or otherwise. The underwriters in connection with the Company's Initial Public Offering are intended third party beneficiaries of this Section and shall have the right, power and authority to enforce the provisions hereof as though they were a party hereto. Holder further agrees to execute such agreements as may be reasonably requested by the underwriters in the Company's Initial Public Offering that are consistent with this Section 23 or that are necessary to give further effect thereto.

In order to enforce the foregoing covenant, the Company may	y impose stop-transfer instructions with	n respect to shares of the Company's capital
stock acquired through the exercise of this Warrant (and the shares	or securities of every other person sub	ject to the foregoing restriction) until the end of
such period. Notwithstanding the foregoing, if (i) during the last	(_) days of the	()-day restricted period, the Company
issues		

an earnings release or material news or a material event relating to the Company occurs; or (ii) prior to the expiration of the()-day
restricted period, the Company announces that it will release earnings results during the ()-day period beginning on the last day of the	
()-day period, the restrictions imposed by this Section 23 shall continue to apply until the expiration of the(_)-day
period beginning on the issuance of the earnings release or the occurrence of the material news or material event.	

The Holder agrees that a legend reading substantially as follows shall be placed on all certificates representing all Shares of the Holder (and the shares or securities of every other person subject to the restriction contained in this Section 23):

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO ALOCK-UP PERIOD AFTER THE EFFECTIVE DATE OF THE ISSUER'S REGISTRATION STATEMENT FILED UNDER THE ACT, AS AMENDED, AS SET FORTH IN AN AGREEMENT BETWEEN THE COMPANY AND THE ORIGINAL HOLDER OF THESE SECURITIES, A COPY OF WHICH MAY BE OBTAINED AT THE ISSUER'S PRINCIPAL OFFICE. SUCH LOCK-UP PERIOD IS BINDING ON TRANSFEREES OF THESE SHARES.

IN WITNESS WHEREOF, the parties have executed this Warrant as of the date first written above.

GROVE COLLABORATIVE, INC.

	By: Stuart Landesberg Chief Executive Officer
ACKNOWLEDGED AND AGREED:	
Ву:	
Address:	

NOTICE OF EXERCISE

GROVE COLLABORATIVE, INC.

Attention: Corporate Secretary

The unders	igned hereby elects to purchase, pursuant to the provisions of the Warr	rant, as follows:
	shares of Common Stock pursuant to the terms of the attached Warrant, and tenders herewith payment in cash of the Exercise Price of such Shares in full, together with all applicable transfer taxes, if any.	
	Net Exercise the attached Warrant with respect to Share	res.
The unders	igned hereby represents and warrants that Representations and Warran	ties in Section 6 hereof are true and correct as of the date hereof.
		HOLDER:
Date:		Ву:
	Address:	
Name in which sh	nares should be registered:	

ASSIGNMENT FORM

(To assign the foregoing Warrant, execute this form and supply required information. Do not use this form to purchase shares.)

FOR VALUE RECEIVED, the foregoing Warrant and all rights evidenced thereby are hereby assigned to

Name:	
	(Please Print)
Address:	
	(Please Print)
Dated:	
Holder's	
Signature:	<u> </u>
Holder's	
Address:	<u> </u>

NOTE: The signature to this Assignment Form must correspond with the name as it appears on the face of the Warrant. Officers of corporations and those acting in a fiduciary or other representative capacity should provide proper evidence of authority to assign the foregoing Warrant.

THIS WARRANT AND THE SHARES ISSUABLE HEREUNDER HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AS SET FORTH IN SECTIONS 6.3 AND 6.4 BELOW, MAY NOT BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND LAWS OR, IN THE OPINION OF LEGAL COUNSEL IN FORM AND SUBSTANCE SATISFACTORY TO THE COMPANY, SUCH OFFER, SALE, PLEDGE OR OTHER TRANSFER IS EXEMPT FROM SUCH REGISTRATION.

WARRANT TO PURCHASE STOCK

Schedule I hereto (the 'Issue Date') by the certain of even date herev	TOCK (as amended and in effect from time to time, this 'Warrant') is issued as of the issue date set forth on company set forth on Schedule I hereto (the 'Company') to in connection with that with among and (collectively, and as may be set from time to time, the "Loan Agreement"). The parties agree as follows:			
SCHEDULE I. <u>WARRANT PROVISIONS</u> .				
Warrant Section Recitals - "Issue Date"	Warrant Provision			
Recitals -"Company"	Grove Collaborative, Inc., a Delaware public benefit corporation			
1.1 –"Class"	Common Stock, \$0.0001 par value per share.			
1.1 –"Exercise Price"	\$per Share.			
1.2 –"Initial Shares"				
1.3 –"Additional Shares"				
1.3 – Conditions for exercisability of Additional Shares	The making (if any) of the first Term B Loan Advance (as defined in the Loan Agreement) to the Company in any amount.			
4.1(b) – Share percentage as of the Issue Date	% of the Company's total fully-diluted issued and outstanding shares of capital stock.			
6.1(a) -"Expiration Date"				
	SECTION 1. <u>RIGHT TO PURCHASE SHARES</u> .			
assignee or transferee of this Warrant or of Company up to the number of fully paid and	able consideration, the Company hereby grants to (together with any successor or permitted any shares issued upon exercise hereof, "Holder") the right, and Holder is entitled, to purchase from the a non-assessable shares (as determined pursuant to Section 1.2 below) of the class set forth or Schedule I Share set forth on Schedule I hereto (the "Exercise Price"), subject to the provisions and upon the terms and			

1.2 Number of Shares. This Warrant shall be exercisable for the number of initial shares of the Class as set forth or Schedule I hereto (the "Initial Shares"), plus the Additional Shares (as hereinafter defined), if any (collectively, and as may be adjusted from time to time in accordance with the

provisions of this Warrant, the "Shares").

1.3 Additional Shares. This Warrant shall automatically become exercisable for the number of additional shares of the Class as set forth or Schedule I hereto (the "Additional Shares") upon the occurrence of the event set forth on Schedule I hereto. The number of Additional Shares may be adjusted from time to time in accordance with the provisions of this Warrant, including, without limitation, adjustments in respect of events occurring prior to the date, if any, on which this Warrant becomes exercisable for such shares as if they constituted "Shares" hereunder for such purpose at all times from the Issue Date.

SECTION 2. EXERCISE.

- 2.1 Method of Exercise. Holder may exercise this Warrant in whole or in part at any time and from time to time prior to the expiration or earlier termination of this Warrant, by delivering to the Company the original of this Warrant together with a duly executed Notice of Exercise in substantially the form attached hereto as Appendix 1 and, unless Holder is exercising this Warrant pursuant to a cashless exercise set forth in Section 2.2 below, a check, wire transfer of same-day funds (to an account designated by the Company), or other form of payment acceptable to the Company for the aggregate Exercise Price for the Shares being purchased. Notwithstanding any contrary provision herein, to the extent that the original of this Warrant is an electronic original, in no event shall an original ink-signed paper copy of this Warrant be required for any exercise of a Holder's rights hereunder, nor shall this Warrant or any physical copy hereof be required to be physically surrendered at the time of any exercise hereof.
- 2.2 <u>Cashless Exercise</u>. On any exercise of this Warrant, in lieu of payment of the aggregate Exercise Price in the manner specified in Section 2.1 above, Holder may elect to surrender to the Company Shares having an aggregate value equal to the aggregate Exercise Price. If Holder makes such election, the Company shall issue to Holder such number of fully paid and non-assessable Shares determined by the following formula:

X = Y(A-B)/A

where:

- X = the number of Shares to be issued to Holder;
- Y = the number of Shares with respect to which this Warrant is being exercised (inclusive of the Shares surrendered to the Company in payment of the aggregate Exercise Price);
 - A = the fair market value (as determined pursuant to Section 2.3 below) of one Share; and
 - B = the Exercise Price.
- 2.3 Fair Market Value. If shares of the Class are then traded or quoted on a nationally recognized securities exchange, inter-dealer quotation system or over-the-counter market (a "Trading Market"), the fair market value of a Share shall be the closing price or last sale price of a share of the Class reported for the Business Day immediately before the date on which Holder delivers this Warrant together with its Notice of Exercise to the Company. If shares of the Class are not then traded in a Trading Market, the Board of Directors of the Company shall determine the fair market value of a Share in its reasonable good faith judgment.
- 2.4 <u>Delivery of Certificate and New Warrant</u>. Within a reasonable time after Holder exercises this Warrant in the manner set forth in Sections 2.1 or 2.2 above, the Company shall deliver to Holder a certificate (or, in the case of uncertificated securities, provide notice of book entry) representing the Shares issued to Holder upon such exercise and, if this Warrant has not been fully exercised and has not expired, a new warrant of like tenor representing the Shares not so acquired (or surrendered in payment of the aggregate Exercise Price).

2.5 Replacement of Warrant.

(a) <u>Paper Original Warrant</u>. To the extent that the original of this Warrant is a paper original, on receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant and, in the case of loss, theft or destruction, on delivery of an indemnity agreement reasonably satisfactory in form, substance and amount to the Company or, in the case of mutilation, on surrender of this Warrant to the Company for cancellation, the Company shall, within a reasonable time, execute and deliver to Holder, in lieu of this Warrant, a new warrant of like tenor and amount.

(b) <u>Electronic Original Warrant</u>. To the extent that the original of this Warrant is an electronic original, if at any time this Warrant is rejected by any person (including, but not limited to, paying or escrow agents) or any such person fails to comply with the terms of this Warrant based on this Warrant being presented to such person as an electronic record or a printout hereof, or any signature hereto being in electronic form, the Company shall, promptly upon Holder's request and without indemnity, execute and deliver to Holder, in lieu of electronic original versions of this Warrant, a new warrant of like tenor and amount in paper form with original ink signatures.

2.6 Treatment of Warrant Upon Acquisition of Company.

- (a) Acquisition. "Acquisition" means any transaction or series of related transactions involving: (i) the sale, lease, exclusive license, or other disposition of all or substantially all of the assets of the Company; (ii) any merger or consolidation of the Company into or with another person or entity (other than a merger or consolidation effected exclusively to change the Company's domicile), or any other corporate reorganization, in which the stockholders of the Company in their capacity as such immediately prior to such merger, consolidation or reorganization, own less than a majority of the Company's (or the surviving or successor entity's) outstanding voting power immediately after such merger, consolidation or reorganization; or (iii) any sale or other transfer by the stockholders of the Company of shares representing at least a majority of the Company's then-total outstanding combined voting power. For the avoidance of doubt, "Acquisition" shall not include any sale and issuance by the Company of shares of its capital stock or of securities or instruments exercisable for or convertible into, or otherwise representing the right to acquire, shares of its capital stock to one or more investors for cash in a transaction or series of related transactions the primary purpose of which is a bona fide equity financing of the Company.
- (b) Treatment of Warrant in Cash/Public Acquisition. In the event of an Acquisition in which the consideration to be received by the holders of the outstanding shares of the Class (in their capacity as such) consists solely of cash, solely of Marketable Securities (as hereinafter defined) or a combination of cash and Marketable Securities (a "Cash/Public Acquisition"), and the fair market value of one Share as determined in accordance with Section 2.3 above would be greater than the Exercise Price in effect as of immediately prior to the closing of such Cash/Public Acquisition, and Holder has not previously exercised this Warrant in full, then, in lieu of Holder's exercise of the unexercised portion of this Warrant, this Warrant shall, as of immediately prior to such closing (but subject to the occurrence thereof) automatically cease to represent the right to purchase Shares and shall, from and after such closing, represent solely the right to receive the aggregate consideration that would have been payable in such Acquisition on and in respect of all Shares for which this Warrant was exercisable as of immediately prior to the closing thereof, net of the aggregate Exercise Price therefor, as if such Shares had been issued and outstanding to Holder as of immediately prior to such closing, as and when such consideration is paid to the holders of the outstanding shares of the Class. In the event of a Cash/Public Acquisition in which the fair market value of one Share as determined in accordance with Section 2.3 above would be equal to or less than the Exercise Price in effect as of immediately prior to such closing.
- (c) Treatment of Warrant in non-Cash/Public Acquisition. Upon the closing of any Acquisition other than a Cash/Public Acquisition, the acquiring, surviving or successor entity shall assume this Warrant and the Company's obligations hereunder, and this Warrant shall thereafter be exercisable for the same securities and/or other property as would have been paid for the Shares issuable upon exercise of the unexercised portion of this Warrant as if such Shares were outstanding on and as of the closing of such Acquisition, at an aggregate Exercise Price equal to the aggregate Exercise Price in effect as of immediately prior to such closing, all subject to further adjustment from time to time thereafter in accordance with the provisions of this Warrant
- (d) Marketable Securities. "Marketable Securities" means securities meeting all of the following requirements (determined as of immediately prior to the closing of the Acquisition): (i) the issuer thereof is then subject to the reporting requirements of Section 13 or Section 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and is then current in its filing of all required reports and other information under the Act and the Exchange Act; (ii) the class and series of shares or other security of the issuer that would be received by Holder in connection with the Acquisition were Holder to exercise this Warrant on or prior to the closing thereof is then traded in a Trading Market, and (iii) following the closing of such Acquisition, Holder would not be restricted from publicly re-selling all of the issuer's shares and/or other securities that would be received by Holder in such Acquisition were Holder to exercise this Warrant in full on or prior to the closing of such Acquisition, except to the extent that any such restriction (x) arises solely under federal or state securities laws, rules or regulations, and (y) does

not extend beyond six (6) months from the closing of such Acquisition. Notwithstanding the foregoing provisions of this Section 2.6(d), securities held in escrow or subject to holdback to cover indemnification-related claims shall be deemed to be Marketable Securities if they would otherwise be Marketable Securities but for the fact that they are held in escrow or subject to holdback to cover indemnification-related claims.

SECTION 3. CERTAIN ADJUSTMENTS TO THE SHARES, CLASS AND EXERCISEPRICE.

- 3.1 Stock Dividends, Splits, Etc. If the Company declares or pays a dividend or distribution on the outstanding shares of the Class payable in additional shares of the Class (including fractional shares) or other securities or property (other than cash), then upon exercise of this Warrant, for each Share acquired, Holder shall receive, without additional cost to Holder, the total number and kind of securities and property which Holder would have received had Holder owned the Shares of record as of the date the dividend or distribution occurred. If the Company subdivides the outstanding shares of the Class by reclassification or otherwise into a greater number of shares, the number of Shares purchasable hereunder shall be proportionately increased, even if such number would include fractional shares, and the Exercise Price shall be proportionately decreased. If the outstanding shares of the Class are combined or consolidated, by reclassification or otherwise, into a lesser number of shares, the Exercise Price shall be proportionately increased and the number of Shares shall be proportionately decreased, even if such number would include fractional shares.
- 3.2 Reclassification, Exchange, Combination or Substitution. Upon any event whereby all of the outstanding shares of the Class are reclassified, exchanged, combined, substituted, or replaced for, into, with or by Company securities of a different class and/or series, then from and after the consummation of such event, "Class" shall mean such securities and this Warrant will be exercisable for the number of such securities that Holder would have received had the Shares been outstanding on and as of the consummation of such event, at an aggregate Exercise Price equal to the aggregate Exercise Price in effect as of immediately prior to such event, all subject to further adjustment thereafter from time to time in accordance with the provisions of this Warrant. The provisions of this Section 3.2 shall similarly apply to successive reclassifications, exchanges, combinations, substitutions, replacements or other similar events.
- 3.3 Adjustment to Exercise Price on Cash Dividend. In the event that the Company at any time or from time to time prior to the exercise in full of this Warrant pays any cash dividend on the outstanding shares of the Class or makes any cash distribution on or in respect of all outstanding shares of the Class (other than a distribution of cash proceeds received by the Company in connection with an Acquisition described in Section 2.6(a)(i) above), then on and as of the date of each such dividend payment and/or distribution, the Exercise Price shall be reduced by an amount equal to the amount paid or distributed upon or in respect of each outstanding share of the Class; provided that in no event shall the Exercise Price be reduced below the then-par value, if any, of a share of the Class.
- 3.4 No Fractional Share. No fractional Share shall be issued upon exercise of this Warrant, and the number of Shares to be issued shall be rounded down to the nearest whole Share. If a fractional Share interest arises upon any exercise of this Warrant, the Company shall eliminate such fractional Share interest by paying Holder in cash an amount equal to (a) such fractional interest, multiplied by (b)(i) the fair market value (as determined in accordance with Section 2.3 above) of a full Share, less (ii) the then-effective Exercise Price (the "Fractional Share Value"), unless Holder otherwise elects, in its sole discretion, to waive such payment. Notwithstanding any contrary provision herein, if this Warrant becomes exercisable for a fractional Share interest at any time or from time to time prior to the exercise in full of this Warrant, and the Company eliminates such fractional Share interest prior to any exercise of this Warrant, then the then-effective Exercise Price shall be reduced by an amount equal to the Fractional Share Value, unless Holder otherwise elects, in its sole discretion, to waive such reduction.
- 3.5 <u>Certificate as to Adjustments</u>. Within a reasonable time following each adjustment of the Exercise Price, Class and/or number of Shares pursuant to the terms of this Warrant, the Company, at its expense, shall deliver a certificate of its Chief Financial Officer or other authorized officer to Holder setting forth the adjustments to the Exercise Price, Class and/or number of Shares and the facts upon which such adjustments are based. The Company shall, at any time and from time to time within a reasonable time following Holder's written request and at the Company's expense, furnish Holder with a certificate of its Chief Financial Officer or other authorized officer setting forth the then-current Exercise Price, Class and number of Shares and the computations or other determinations thereof.

SECTION 4. REPRESENTATIONS AND COVENANTS OF THE COMPANY.

- 4.1 Representations and Warranties. The Company represents and warrants to, and agrees with, Holder as follows:
- (a) The initial Exercise Price first set forth above is not greater than the fair market value of a share of the Class as determined by the most recently completed third-party valuation, approved or accepted by the Company's Board of Directors, of a share of the Class obtained for purposes of the Company's compliance with Section 409A of the Internal Revenue Code of 1986, as amended (or the corresponding section of any successor statute) (a "409A Valuation").
- (b) The Initial Shares set forth on Schedule I hereto plus the Additional Shares set forth on Schedule I hereto collectively represent not less than the share percentage set forth on Schedule I hereto, calculated on and as of the Issue Date hereof on a fully-diluted, common stock-equivalent basis (but without excluding shares of capital stock that are not convertible into shares of common stock) assuming (i) the conversion into common stock of all outstanding securities and instruments (including, without limitation, securities deemed to be outstanding pursuant to clause (ii) of this Section 4.1(b)) convertible by their terms into shares of common stock (regardless of whether such securities or instruments are by their terms now so convertible), (ii) the exercise in full of all outstanding options, warrants (including, without limitation, this Warrant) and other rights to purchase or acquire shares of common stock or securities exercisable for or convertible into shares of common stock (regardless of whether such options, warrants or other rights to purchase or acquire are by their terms now exercisable); and (iii) the inclusion of all shares of common stock reserved for issuance under all of the Company's incentive stock and stock option plans and not now subject to outstanding grants or options.
- (c) All Shares which may be issued upon the exercise of this Warrant, shall, upon issuance, be duly authorized, validly issued, fully paid and non-assessable, and free of any liens and encumbrances except for restrictions on transfer provided for herein or under the Company's Certificate of Incorporation or Bylaws, each as amended and in effect from time to time (the "Charter Documents"), any Stockholder Agreement (to the extent Holder is then a party thereto or otherwise subject thereto in accordance with the provisions of Section 5.4 below) or applicable federal and state securities laws. The Company covenants that it shall at all times cause to be reserved and kept available out of its authorized and unissued capital stock such number of shares of the Class and other securities as will be sufficient to permit the exercise in full of this Warrant.
 - (d) The Company's capitalization table attached hereto as Appendix 2 is true and complete, in all material respects, as of the Issue Date.
 - 4.2 Notice of Certain Events. If the Company proposes at any time to:
- (a) declare any dividend or distribution upon the outstanding shares of the Class, whether in cash, stock or other securities or property and whether or not a regular cash dividend;
- (b) offer for subscription or sale pro rata to all holders of the outstanding shares of the Class any additional securities of the Company (other than pursuant to contractual pre-emptive or first refusal rights);
- (c) effect any redemption, reclassification, exchange, combination, substitution, reorganization or recapitalization of the outstanding shares of the Class;
 - (d) effect an Acquisition, or to liquidate, dissolve or wind up the Company; or
- (e) effect its initial, underwritten offering and sale of its securities to the public pursuant to an effective registration statement under the Act (the "IPO"); then, in connection with each such event, the Company shall give Holder (pursuant to Section 6.5 below):
 - (1) in the case of the matters referred to in (a) and (b) above, at least ____ (___) Business Days prior written notice of the earlier to occur of the effective date thereof or the date on which a record will be taken for such dividend, distribution, or subscription rights (and specifying the date on which the holders of outstanding shares of the Class will be entitled thereto) or for determining rights to vote, if any;

require regarding the treatment of this Warrant in connection with such event giving rise to the notice); and
(3) with respect to the IPO, at least(_) Business Days prior written notice of the date on which the Company proposes to make the first public filling of its registration statement in connection therewith.
4.3 Certain Company Information. The Company will provide such information requested by Holder from time to time, within a reasonable time
following each such request, that is reasonably necessary to enable Holder to comply with Holder's accounting or reporting requirements. Prior to the IPO,
such information may include, but shall not be limited to, the Company's then-current summary capitalization table, the price per share for which the
Company most recently prior thereto sold or issued shares of its convertible preferred stock to investors for cash in a bona fide equity financing of the
Company, and the Company's most recent 409A Valuation. Notwithstanding the foregoing provisions of this Section 4.2, the Company shall not be
obligated to provide any such information the disclosure of which, in the written advice or opinion of counsel to the Company (a copy of which shall be

same will take place (and specifying the date on which the holders of outstanding shares of the Class will be entitled to exchange their shares

(___) Business Days prior written notice of the date when the

SECTION 5. REPRESENTATIONS AND COVENANTS OF HOLDER.

Holder represents and warrants to, and agrees with, the Company as follows:

provided to Holder), would adversely affect the attorney-client privilege between the Company and its counsel.

(2) in the case of the matters referred to in (c) and (d) above, at least_

5.1 Investment Representations.

- (a) <u>Purchase for Own Account</u>. This Warrant and the Shares to be acquired upon exercise hereof are being acquired for investment for Holder's account, not as a nominee or agent, and not with a view to the public resale or distribution within the meaning of the Act. Holder also represents that it has not been formed for the specific purpose of acquiring this Warrant or the Shares.
- (b) <u>Disclosure of Information</u>. Holder is aware of the Company's business affairs and financial condition and has received or has had full access to all the information it considers necessary or appropriate to make an informed investment decision with respect to the acquisition of this Warrant and its underlying securities. Holder further has had an opportunity to ask questions of and receive answers from the Company regarding the terms and conditions of the offering of this Warrant and its underlying securities and to obtain additional information (to the extent the Company possessed such information or could acquire it without unreasonable effort or expense) necessary to verify any information furnished to Holder or to which Holder has access.
- (c) Investment Experience. Holder understands that the purchase of this Warrant and its underlying securities involves substantial risk. Holder has experience as an investor in securities of companies in the development stage and acknowledges that Holder can bear the economic risk of such Holder's investment in this Warrant and its underlying securities for an indefinite period of time, and has such knowledge and experience in financial or business matters that Holder is capable of evaluating the merits and risks of its investment in this Warrant and its underlying securities and/or has a preexisting personal or business relationship with the Company and certain of its officers, directors or controlling persons of a nature and duration that enables Holder to be aware of the character, business acumen and financial circumstances of such persons.
 - (d) Accredited Investor Status. Holder is an "accredited investor" within the meaning of Regulation D promulgated under the Act.

- (e) The Act. Holder understands that this Warrant and the Shares issuable upon exercise hereof have not been registered under the Act or registered or qualified under the securities laws of any state, and are issued in reliance upon specific exemptions therefrom, which exemptions depend upon, among other things, the bona fide nature of the Holder's investment intent as expressed herein. Holder understands that the Company is under no obligation to so register or qualify this Warrant, the Shares or such other securities. Holder understands that this Warrant and the Shares issued upon any exercise hereof are "restricted securities" under applicable federal and state securities laws and must be held indefinitely unless subsequently registered under the Act and registered or qualified under applicable state securities laws, or unless exemptions from such registration and qualification are otherwise available. Holder is aware of the provisions of Rule 144 promulgated under the Act.
- 5.2 No Stockholder Rights. Without limiting any provision of this Warrant, Holder agrees that as a Holder of this Warrant it will not have any rights (including, but not limited to, voting rights) as a stockholder of the Company with respect to the Shares issuable hereunder unless and until the exercise of this Warrant and then only with respect to the Shares issued on such exercise.
- 5.3 Market Stand-off Agreement. The Holder agrees that the Shares shall be subject to the Market Standoff provisions in Section 2.11 of the Company's Amended and Restated Investor Rights Agreement dated November 25, 2020, as amended and in effect from time to time.
- 5.4 Stockholder Agreements. Following any exercise of this Warrant and solely with respect to the Shares issued thereupon, if the Company so requests in writing, Holder shall become a party to the Company's then-effective right of first refusal and co-sale agreement, voting agreement and/or each other agreement entered into among the Company and holders of the outstanding shares of the Class, each as may be amended and in effect from time to time (collectively, the "Stockholder Agreements"), by execution and delivery to the Company of a counterpart signature page, joinder agreement, instrument of accession or similar instrument, provided that such Stockholder Agreement is by its terms in force and effect at the time of such exercise. If the foregoing condition is met, then effective upon such exercise, Holder shall automatically become bound by, and the Shares issued upon such exercise shall automatically become subject to, such Stockholder Agreement.
- 5.5 <u>Confidential Information</u>. Holder agrees to treat and hold all information provided by the Company pursuant to this Warrant in confidence in accordance with the provisions of Section 12.9 of the Loan Agreement (regardless of whether the Loan Agreement shall then be in effect).

SECTION 6. MISCELLANEOUS.

6.1 Term; Automatic Cashless Exercise Upon Expiration.

- (a) <u>Term.</u> Subject to the provisions of Section 2.6 above, this Warrant is exercisable in whole or in part at any time and from time to time on or before 6:00 PM, Pacific time, on the expiration date set forth on Schedule I hereto (the "**Expiration Date**") and shall be void thereafter; provided that if the Company does not deliver to Holder written confirmation of the fair market value of a Share pursuant to Section 6.1(b) below, then the Expiration Date shall automatically be extended until the earlier to occur of (i) such date as the Company delivers such written confirmation and (ii) one (1) year after the Expiration Date.
- (b) <u>Automatic Cashless Exercise upon Expiration</u>. In the event that, upon the Expiration Date, the fair market value of one Share as determined in accordance with Section 2.3 above is greater than the Exercise Price in effect on such date, then this Warrant shall automatically be deemed on and as of such date to be exercised pursuant to Section 2.2 above as to all Shares for which it shall not previously have been exercised, and the Company shall, within a reasonable time following Holder's written request, deliver a certificate (or, in the case of uncertificated securities, provide notice of book entry) representing the Shares issued to Holder upon such exercise. If shares of the Class are not then traded in a Trading Market, the Company shall deliver to Holder, prior to the Expiration Date, written confirmation of the fair market value of a Share (as determined pursuant to Section 2.3 above) to be used in determining whether this Warrant shall automatically exercise on the Expiration Date pursuant to this Section 6.1(b).
- 6.2 <u>Legends</u>. Each certificate or notice of book entry evidencing Shares shall be imprinted with a legend in substantially the following form (together with such additional legends as may be required by the Charter Documents or under any Stockholder Agreement (to the extent Holder is then a party thereto or otherwise subject thereto in accordance with the provisions of Section 5.4 above)):

THE SHARES EVIDENCED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AS SET FORTH IN THAT CERTAIN WARRANT TO PURCHASE STOCK ISSUED BY THE ISSUER TO _______ DATED _______, MAY NOT BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND LAWS OR, IN THE OPINION OF LEGAL COUNSEL IN FORM AND SUBSTANCE SATISFACTORY TO THE ISSUER, SUCH OFFER, SALE, PLEDGE OR OTHER TRANSFER IS EXEMPT FROM SUCH REGISTRATION.

And, if then applicable, a legend in substantially the following form:

THE SECURITIES EVIDENCED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFERABILITY AND RESALE, INCLUDING A LOCK-UP PERIOD AFTER THE EFFECTIVE DATE OF THE ISSUER'S REGISTRATION STATEMENT FILED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AS SET FORTH IN THAT CERTAIN WARRANT TO PURCHASE STOCK ISSUED BY THE ISSUER TO ________, A COPY OF WHICH MAY BE OBTAINED AT THE ISSUER'S PRINCIPAL OFFICE. SUCH RESTRICTIONS ARE BINDING ON TRANSFEREES OF THESE SECURITIES.

- 6.3 <u>Compliance with Securities Laws on Transfer.</u> This Warrant and the Shares issued upon exercise hereof may not be transferred or assigned in whole or in part except in compliance with applicable federal and state securities laws by the transferor and the transferee (including, without limitation, the delivery of investment representation letters and legal opinions reasonably satisfactory to the Company, as reasonably requested by the Company). The Company shall not require Holder to provide an opinion of counsel if the transfer is to an affiliate of Holder; provided that such affiliate is an "accredited investor" as defined in Regulation D promulgated under the Act.
- 6.4 <u>Transfer Procedure</u>. Subject to the provisions of Section 6.3 and upon providing the Company with written notice, Holder may transfer all or part of this Warrant or the Shares issued upon exercise of this Warrant to any transferee; provided that in connection with any such transfer, Holder will give the Company notice of the portion of the Warrant and/or Shares being transferred with the name, address and taxpayer identification number of the transferee, and Holder will surrender this Warrant, or the certificates or other evidence of such Shares or other securities, to the Company for reissuance to the transferee(s) (and to Holder if applicable); and provided further, that any transferee shall make substantially the representations set forth in Section 5.1 above and shall agree in writing with the Company to be bound by all of the terms and conditions of this Warrant; and provided further, that the transfer of any Shares issued on exercise hereof shall be subject to the provisions of the Stockholder Agreements to the extent Holder is then a party thereto or otherwise subject thereto in accordance with the provisions of Section 5.4 above. Notwithstanding any contrary provision herein, at all times prior to the IPO, Holder may not, without the Company's prior written consent, transfer this Warrant or any portion hereof, or any Shares issued upon any exercise hereof to any person or entity who directly competes with the Company (as determined by the Company's Board of Directors in its reasonable good faith judgment), except in connection with an Acquisition of the Company by such a direct competitor.
- 6.5 Notices. All notices and other communications hereunder from the Company to the Holder, or vice versa, shall be deemed delivered and effective (i) when given personally, (ii) on the third (3rd) Business Day after being mailed by first-class registered or certified mail, postage prepaid, (iii) upon actual receipt if given by electronic mail and such receipt is confirmed in writing by the recipient, or (iv) on the first Business Day following delivery to a reliable overnight courier service, courier fee prepaid, in any case at such address as may have been furnished to the Company or Holder, as the case may be, in writing by the Company or such Holder from time to time in accordance with the provisions of this Section 6.5. All notices to Holder shall be addressed as follows until the Company receives notice of a change of address in connection with a transfer or otherwise:

All notices to the Company shall be addressed as follows until Holder receives notice of a change in address:

Grove Collaborative, Inc. Attn: 1301 Sansome Street San Francisco, CA 94111 Email:

With a copy (which shall not constitute notice) to:

Sidley Austin LLP

Attn: Martin A. Wellington 1001 Page Mill Road, Building 1 Palo Alto, California, 94304 Email: mwellington@sidley.com

- 6.6 <u>Amendment and Waiver</u>. Notwithstanding any contrary provision herein or in the Loan Agreement, this Warrant may be amended and any provision hereof waived (either generally or in a particular instance and either retroactively or prospectively) only by an instrument in writing signed by Holder and any party against which enforcement of such amendment or waiver is sought.
- 6.7 Counterparts; Electronic Signatures; Status as Certificated Security. This Warrant may be executed by one or more of the parties hereto in any number of separate counterparts, all of which together shall constitute one and the same instrument. The Company, Holder and any other party hereto may execute this Warrant by electronic means and each party hereto recognizes and accepts the use of electronic signatures and the keeping of records in electronic form by any other party hereto in connection with the execution and storage hereof. To the extent that this Warrant or any agreement subject to the terms hereof or any amendment hereto is executed, recorded or delivered electronically, it shall be binding to the same extent as though it had been executed on paper with an original ink signature, as provided under applicable law, including, without limitation, any state law based on the Uniform Electronic Transactions Act. The fact that this Warrant is executed, signed, stored or delivered electronically shall not prevent the transfer by any Holder of this Warrant pursuant to Section 6.4 or the enforcement of the terms hereof. To the extent that the original of this Warrant is an electronic original, this Warrant, and any copies hereof, shall NOT be deemed to be a "certificated security" within the meaning of Section 8102(a)(4) of the California Commercial Code. Physical possession of the original of this Warrant or any paper copy thereof shall confer no special status to the bearer thereof.
- 6.8 <u>Headings</u>. The headings in this Warrant are for purposes of reference only and shall not limit or otherwise affect the meaning of any provision of this Warrant.
 - 6.9 Business Days. "Business Day" means any day that is not a Saturday, Sunday or a day on which banks in California are closed.

SECTION 7. GOVERNING LAW, VENUE AND JURY TRIAL WAIVER; JUDICIAL REFERENCE

- 7.1 Governing Law. This Warrant shall be governed by and construed in accordance with the laws of the State of California, without giving effect to its principles regarding conflicts of law.
- 7.2 <u>Jurisdiction and Venue</u>. The Company and Holder each irrevocably and unconditionally submit to the exclusive jurisdiction of the State and Federal courts in Santa Clara County, California; provided, however, that nothing in this Warrant shall be deemed to operate to preclude Holder from bringing suit or taking other legal action in any other jurisdiction to enforce a judgment or other court order in favor of Holder. The Company expressly,

irrevocably and unconditionally submits and consents in advance to such jurisdiction in any action or suit commenced in any such court, and the Company hereby irrevocably and unconditionally waives, to the fullest extent permitted by applicable law, any objection that it may have based upon lack of personal jurisdiction, improper venue, or forum non conveniens and hereby irrevocably and unconditionally consents to the granting of such legal or equitable relief as is deemed appropriate by such court. The Company hereby waives personal service of the summons, complaints, and other process issued in such action or suit and agrees that service of such summons, complaints, and other process may be made by registered or certified mail addressed to the Company in accordance with Section 6.5 of this Warrant and that service so made shall be deemed completed upon the earlier to occur of the Company's actual receipt thereof of ______ (___) days after deposit in the U.S. mails, proper postage prepaid.

7.3 <u>Jury Trial Waiver</u> TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, THE COMPANY AND HOLDER EACH WAIVES ITS RIGHT TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION ARISING OUT OF OR BASED UPON THIS WARRANT, THE LOAN AGREEMENT OR ANY CONTEMPLATED TRANSACTION, INCLUDING CONTRACT, TORT, BREACH OF DUTY AND ALL OTHER CLAIMS. THIS WAIVER IS A MATERIAL INDUCEMENT FOR THE PARTIES' AGREEMENT TO THIS WARRANT. EACH PARTY HERETO HAS REVIEWED THIS WAIVER WITH ITS COUNSEL.

7.4 Judicial Reference. WITHOUT INTENDING IN ANY WAY TO LIMIT THE PARTIES AGREEMENT TO WAIVE THEIR RESPECTIVE RIGHT TO A TRIAL BY JURY, if the waiver of the right to a trial by jury in Section 7.3 above is not enforceable, the parties hereto agree that any and all disputes or controversies of any nature between them arising at any time shall be decided by a reference to a private judge, mutually selected by the parties (or, if they cannot agree, by the Presiding Judge of the Santa Clara County, California Superior Court) appointed in accordance with California Code of Civil Procedure Section 638 (or pursuant to comparable provisions of federal law if the dispute falls within the exclusive jurisdiction of the federal courts), sitting without a jury, in Santa Clara County, California; and the parties hereby submit to the jurisdiction of such court. The reference proceedings shall be conducted pursuant to and in accordance with the provisions of California Code of Civil Procedure Sections 638 through 645.1, inclusive. The private judge shall have the power, among others, to grant provisional relief, including without limitation, entering temporary restraining orders, issuing preliminary and permanent injunctions and appointing receivers. All such proceedings shall be closed to the public and confidential and all records relating thereto shall be permanently sealed. If during the course of any dispute, a party desires to seek provisional relief, but a judge has not been appointed at that point pursuant to the judicial reference procedures, then such party may apply to the Santa Clara County, California Superior Court for such relief. The proceeding before the private judge shall be conducted in the same manner as it would be before a court under the rules of evidence applicable to judicial proceedings. The parties shall be entitled to discovery which shall be conducted in the same manner as it would be before a court under the rules of discovery applicable to judicial proceedings. The private judge shall oversee discovery and may enforce all discovery rules and orders applicable to judicial proceedings in the same manner as a trial court judge. The parties agree that the selected or appointed private judge shall have the power to decide all issues in the action or proceeding, whether of fact or of law, and shall report a statement of decision thereon pursuant to California Code of Civil Procedure Section 644(a). Nothing in this Section 7.4 shall limit the right of any party at any time to exercise self-help remedies or obtain provisional remedies. The private judge shall also determine all issues relating to the applicability, interpretation, and enforceability of this Section 7.4.

7.5 Survival. This Section 7 shall survive the termination of this Warrant.

[Signature page follows]

IN WITNESS WHEREOF, the parties have caused this Warrant To Purchase Stock to be executed by their duly authorized representatives effective as of the Issue Date written above.

COMPANY:	
GROVE COLL	ABORATIVE, INC.
By:	
Name:	
Title:	
HOLDER:	
Ву:	
Name:	
Title:	

APPENDIX 1

Form of Notice of Exercise of Warrant

	ndersigned Holder hereby exercises its right to purchase		
one] Stock of	(the "Company") in accordance with the attached	Warrant To Purchase Stock,	and tenders payment of the aggregate
Exercise Price	for such shares as follows:		
[]	Check in the amount of \$ payable to order of the Compar	ny enclosed herewith	
[]	Wire transfer of immediately available funds to the Company's acc	count	
[]	Cashless exercise pursuant to Section 2.2 of the Warrant, resulting in the issuance of shares of the [Common] / [Ser Preferred] [circle one] Stock of the Company		
[]	Other [Describe]		
2 Plans	e issue a certificate or certificates (or evidence of book entry) representi	na tha Charas in tha nama s	nasified below:
Z. Flease	e issue a certificate of certificates (of evidence of book entry) represent	ing the shares in the hame s	pecified below.
	Holder's Name		
	(11)		
	(Address)		
3. By its	execution below and for the benefit of the Company, Holder hereby ma	akes each of the representat	ions and warranties set forth in Section
5.1 of the War	rant To Purchase Stock as of the date hereof.		
		HOLDER:	
		HOLDER.	
		By:	
		Name:	
		Title:	
		(Date):	
	A . P. 1	()	
	Appendix 1		

Appendix I

APPENDIX 2

Company Capitalization Table

See attached

Appendix 2

THIS WARRANT AND THE UNDERLYING SECURITIES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"). THEY MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT AS TO SUCH SECURITIES UNDER THE ACT OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY THAT SUCH REGISTRATION IS NOT REQUIRED.

GROVE COLLABORATIVE, INC.

WARRANT TO PURCHASE COMMON STOCK

No. CSW-
Void After the Termination Date
THIS CERTIFIES THAT, for value received, (the "Holder"), is entitled to purchase at the Exercise Price (defined below) from Grove Collaborative, Inc., a Delaware corporation (the "Company"), up to an aggregate of shares of the Company's Common Stock (the "Common Stock"), subject to the exercisability and other terms and conditions set forth herein. This Warrant is being issued in satisfaction of the Company's obligations pursuant to that certain Agreement between the Holder and the Company dated (as the same may be amended and/or restated from time to time, the "Holder Agreement"). 1. DEFINITIONS. As used herein, the following terms shall have the following respective meanings:
1. DEFINITIONS. As used herein, the following terms shall have the following respective meanings:
(a) "Exercise Period" shall mean the period commencing with the date hereof and ending on the date that is the year anniversary of the date the Holder Agreement terminates, unless sooner terminated as provided below, with such date being the "Termination Date".
(b) "Exercise Price" shall mean \$ per Exercise Share, subject to adjustment pursuant to Section 5 below.
(c) "Exercise Shares" shall mean the shares of Common Stock issuable upon exercise of this Warrant, subject to adjustment pursuant to the terms herein, including but not limited to adjustment pursuant to Section 5 below.
2. EXERCISE OF WARRANT.
2.1 The rights represented by this Warrant may be exercised in whole or in part at any time during the Exercise Period, by delivery of the following to the Company at its address set forth above (or at such other address as it may designate by notice in writing to the Holder):
(a) An executed Notice of Exercise in the form attached hereto;

(b) Payment of the Exercise Price either (i) in cash or by check, or (ii) by cancellation of indebtedness; and

(c) This Warrant.

Upon the exercise of the rights represented by this Warrant, a certificate or certificates for the Exercise Shares so purchased, registered in the name of the Holder or persons affiliated with the Holder, if the Holder so designates, shall be issued and delivered to the Holder within a reasonable time after the rights represented by this Warrant shall have been so exercised.

The person in whose name any certificate or certificates for Exercise Shares are to be issued upon exercise of this Warrant shall be deemed to have become the holder of record of such shares on the date on which this Warrant was surrendered and payment of the Exercise Price was made, irrespective of the date of delivery of such certificate or certificates, except that, if the date of such surrender and payment is a date when the stock transfer books of the Company are closed, such person shall be deemed to have become the holder of such shares at the close of business on the next succeeding date on which the stock transfer books are open.

2.2 Net Exercise. Notwithstanding any provisions herein to the contrary, if the fair market value of one share of the Company's Common Stock is greater than the Exercise Price (at the date of calculation as set forth below), in lieu of exercising this Warrant by payment of cash, the Holder may elect to receive Exercise Shares equal to the value (as determined below) of this Warrant (or the portion thereof being canceled) by surrender of this Warrant at the principal office of the Company together with the properly endorsed Notice of Exercise in which event the Company shall issue to the Holder a number of shares of Common Stock computed using the following formula:

$$X = \underline{Y(A-B)}_{A}$$

Where:

- X = the number of shares of Common Stock to be issued to the Holder (not to exceed the number of Exercise Shares)
- Y = the number of shares of Common Stock purchasable under the Warrant (not to exceed the number of Exercise Shares) or, if only a portion of the Warrant is being exercised, the portion (not to exceed the number of Exercise Shares) of the Warrant being canceled (at the date of such calculation)
- A = the fair market value of one share of the Company's Common Stock (at the date of such calculation)
- B = Exercise Price (as adjusted to the date of such calculation)

For purposes of the above calculation, the fair market value of one share of Common Stock shall be determined by the Company's Board of Directors in good faith; *provided*, *however*, that in the event that this Warrant is net issue exercised pursuant to this Section 2.3 in connection with the Company's initial public offering of its Common Stock, the fair market value per share shall be the per share offering price to the public of the Company's initial public offering.

3. COVENANTS OF THE COMPANY.

- 3.1 Covenants as to Exercise Shares. The Company covenants and agrees that all Exercise Shares that may be issued upon the exercise of the rights represented by this Warrant will, upon issuance, be validly issued and outstanding, fully paid and nonassessable, and free from all taxes, liens and charges with respect to the issuance thereof. The Company further covenants and agrees that the Company will at all times during the Exercise Period, have authorized and reserved, free from preemptive rights, a sufficient number of shares of its Common Stock to provide for the exercise of the rights represented by this Warrant. If at any time during the Exercise Period the number of authorized but unissued shares of Common Stock shall not be sufficient to permit exercise of this Warrant, the Company will take such corporate action as may, in the opinion of its counsel, be necessary to increase its authorized but unissued shares of Common Stock to such number of shares as shall be sufficient for such purposes.
- 3.2 Notices of Record Date. In the event of any taking by the Company of a record of the holders of Common Stock for the purpose of determining the holders thereof who are entitled to receive any dividend (other than a cash dividend which is the same as cash dividends paid in previous quarters) or other distribution, the Company shall mail to the Holder, at ____ (__) days prior to the date specified herein, a notice specifying the date on which any such record is to be taken for the purpose of such dividend or distribution.

4. REPRESENTATIONS OF HOLDER.

4.1 Acquisition of Warrant for Personal Account. The Holder represents and warrants that it is acquiring the Warrant and the Exercise Shares solely for its account for investment and not with a view to or for sale or distribution of said Warrant or Exercise Shares or any part thereof. The Holder also represents that the entire legal and beneficial interests of the Warrant and Exercise Shares the Holder is acquiring is being acquired for, and will be held for, its account only.

4.2 Securities Are Not Registered.

- (a) The Holder understands that the Warrant and the Exercise Shares have not been registered under the Securities Act of 1933, as amended (the "Act") on the basis that no distribution or public offering of the stock of the Company is to be effected. The Holder realizes that the basis for the exemption may not be present if, notwithstanding its representations, the Holder has a present intention of acquiring the securities for a fixed or determinable period in the future, selling (in connection with a distribution or otherwise), granting any participation in, or otherwise distributing the securities. The Holder has no such present intention.
- (b) The Holder recognizes that the Warrant and the Exercise Shares must be held indefinitely unless they are subsequently registered under the Act or an exemption from such registration is available. The Holder recognizes that the Company has no obligation to register the Warrant or the Exercise Shares of the Company, or to comply with any exemption from such registration.

(c) The Holder is aware that neither the Warrant nor the Exercise Shares may be sold pursuant to Rule 144 adopted under the Act unless certain conditions are met, including, among other things, the existence of a public market for the shares, the availability of certain current public information about the Company, the resale following the required holding period under Rule 144 and the number of shares being sold during any three month period not exceeding specified limitations. Holder is aware that the conditions for resale set forth in Rule 144 have not been satisfied and that the Company presently has no plans to satisfy these conditions in the foreseeable future.

4.3 Disposition of Warrant and Exercise Shares.

- (a) The Holder further agrees not to make any disposition of all or any part of the Warrant or Exercise Shares in any event unless and until:
- (i) The Company shall have received a letter secured by the Holder from the Securities and Exchange Commission stating that no action will be recommended to the Commission with respect to the proposed disposition;
- (ii) There is then in effect a registration statement under the Act covering such proposed disposition and such disposition is made in accordance with said registration statement; or
- (iii) The Holder shall have notified the Company of the proposed disposition and shall have furnished the Company with a detailed statement of the circumstances surrounding the proposed disposition, and if reasonably requested by the Company, the Holder shall have furnished the Company with an opinion of counsel, reasonably satisfactory to the Company, for the Holder to the effect that such disposition will not require registration of such Warrant or Exercise Shares under the Act or any applicable state securities laws.
- (b) The Holder understands and agrees that all certificates evidencing the shares to be issued to the Holder may bear the following legends, in addition to any other legend required by law, the Company's Bylaws or other agreements between Holder and the Company:

THESE SECURITIES HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"). THEY MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT AS TO THE SECURITIES UNDER THE ACT OR AN OPINION OF COUNSEL SATISFACTORY TO THE COMPANY THAT SUCH REGISTRATION IS NOT REQUIRED.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO A TRANSFER RESTRICTION, AS PROVIDED IN THE BYLAWS OF THE CORPORATION.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO A RIGHT OF FIRST REFUSAL OPTION IN FAVOR OF THE CORPORATION AND/OR ITS ASSIGNEE(S), AS PROVIDED IN THE BYLAWS OF THE CORPORATION.

- 4.4 Accredited Investor Status. The Holder is an "accredited investor" as such term is defined in Rule 501 under the Act.
- **4.5 Information and Sophistication.** The Holder hereby: (i) acknowledges that it has received all the information it has requested from the Company and it considers necessary or appropriate for deciding whether to acquire this Warrant and the Exercise Shares; (ii) represents that it has had an opportunity to ask questions and receive answers from the Company regarding the terms and conditions of the offering of this Warrant and the Exercise Shares and to obtain any additional information necessary to verify the accuracy of the information given the Holder; and (iii) further represents that it has such knowledge and experience in financial and business matters that it is capable of evaluating the merits and risk of this investment.
- **4.6 Restriction on Transfer.** The Holder acknowledges that the Exercise Shares, if any when issued upon exercise of this Warrant, shall be subject to restrictions on transferability as set forth in the Bylaws.
- 5. ADJUSTMENT OF EXERCISE PRICE. In the event, after the date this Warrant is issued and prior to exercise, of changes in the outstanding Common Stock of the Company by reason of stock dividends, split-ups, recapitalizations, reclassifications, combinations or exchanges of shares, separations, reorganizations, liquidations, or the like, the number and class of shares available under the Warrant in the aggregate and the Exercise Price shall be correspondingly adjusted to give the Holder of the Warrant, on exercise for the same aggregate Exercise Price, the total number, class, and kind of shares as the Holder would have owned had the Warrant been exercised prior to the event and had the Holder continued to hold such shares until after the event requiring adjustment; provided, however, that such adjustment shall not be made with respect to, and this Warrant shall terminate if not exercised prior to, the events set forth in Section 7 below. The form of this Warrant need not be changed because of any adjustment in the number of Exercise Shares subject to this Warrant.
- 6. FRACTIONAL SHARES. No fractional shares shall be issued upon the exercise of this Warrant as a consequence of any adjustment pursuant hereto. All Exercise Shares (including fractions) issuable upon exercise of this Warrant may be aggregated for purposes of determining whether the exercise would result in the issuance of any fractional share. If, after aggregation, the exercise would result in the issuance of a fractional share, the Company shall, in lieu of issuance of any fractional share, pay the Holder otherwise entitled to such fraction a sum in cash equal to the product resulting from multiplying the then current fair market value of an Exercise Share by such fraction.
- 7. EARLY TERMINATION. In the event of, at any time during the Exercise Period, an initial public offering of securities of the Company registered under the Act, or any capital reorganization, or any reclassification of the capital stock of the Company (other than a change in par value or from par value to no par value or no par value or as a result of a stock dividend or subdivision, split-up or combination of shares), or the consolidation or merger of the Company with or into another corporation (other than a merger solely to effect a reincorporation of the Company into another state), or the sale or other disposition of all or substantially all the properties and assets of the Company in its entirety to any other person, the Company shall provide to the Holder _____ (__) days advance written notice of such public offering, reorganization, reclassification, consolidation, merger or sale or other disposition of the Company's assets, and this Warrant shall terminate unless exercised prior to the date such public offering is closed or the occurrence of such reorganization, reclassification, consolidation, merger or sale or other disposition of the Company's assets.

- 8. Market Stand-Off Agreement. Holder hereby agrees that Holder shall not sell, dispose of, transfer, make any short sale of, grant any option for the purchase of, or enter into any hedging or similar transaction with the same economic effect as a sale, any shares of Common Stock (or other securities) of the Company held by Holder (other than those included in the registration) during the _ _-day period following the effective date of the initial public offering (or such longer period as the underwriters or the Company shall request in order to facilitate compliance with FINRA Rule 2241 or any successor or similar rule or regulation). Holder further agrees to execute and deliver such other agreements as may be reasonably requested by the Company or the managing underwriters that are consistent with the foregoing or that are necessary to give further effect thereto. In addition, if requested by the Company or the representative of the underwriters of Common Stock (or other securities) of the Company, Holder shall provide, within () days of such request, such information as may be required by the Company or such representative in connection with the completion of any public offering of the Company's securities pursuant to a registration statement filed under the Act. The obligations described in this Section 8 shall not apply to a registration relating solely to employee benefit plans on Form S-1 or Form S-8 or similar forms that may be promulgated in the future, or a registration relating solely to a transaction on Form S-4 or similar forms that may be promulgated in the future. In order to enforce the foregoing covenant, the Company may impose stop-transfer instructions with respect to such Common Stock (or other securities) until the end of such period. Holder agrees that any transferee of the Warrant (or other securities) of the Company held by Holder shall be bound by this Section 8. The underwriters of the Company's stock are intended third party beneficiaries of this Section 8 and shall have the right, power and authority to enforce the provisions hereof as though they were a party hereto.
- **9. NO STOCKHOLDER RIGHTS.** This Warrant in and of itself shall not entitle the Holder to any voting rights or other rights as a stockholder of the Company.
- 10. TRANSFER OF WARRANT. Subject to applicable laws, the restriction on transfer set forth on the first page of this Warrant, and any restrictions applicable to the transfer of shares set forth in the Company's Bylaws (which Holder acknowledges and agrees shall apply to transfer of this Warrant *mutatis mutandis*), as they may be amended from time to time, this Warrant and all rights hereunder are transferable, by the Holder in person or by duly authorized attorney, upon delivery of this Warrant and the form of assignment attached hereto to any transferee designated by Holder. The transferee shall sign an investment letter in form and substance satisfactory to the Company.
- 11. LOST, STOLEN, MUTILATED OR DESTROYED WARRANT. If this Warrant is lost, stolen, mutilated or destroyed, the Company may, on such terms as to indemnity or otherwise as it may reasonably impose (which shall, in the case of a mutilated Warrant, include the surrender thereof), issue a new Warrant of like denomination and tenor as the Warrant so lost, stolen, mutilated or destroyed. Any such new Warrant shall constitute an original contractual obligation of the Company, whether or not the allegedly lost, stolen, mutilated or destroyed Warrant shall be at any time enforceable by anyone.

12. NOTICES, ETC. All notices required or permitted hereunder shall be in writing and shall be deemed effectively given: (a) upon personal
delivery to the party to be notified, (b) when sent by confirmed telex or facsimile if sent during normal business hours of the recipient, if not, then on the
next business day, (c) (_) days after having been sent by registered or certified mail, return receipt requested, postage prepaid, or (d) (_)
day after deposit with a nationally recognized overnight courier, specifying next day delivery, with written verification of receipt. All communications
shall be sent to the Company and Holder at the respective address listed on the signature page, or at such other address as the Company or Holder may
designate by () days advance written notice to the other parties hereto.

13. GOVERNING LAW. This Warrant and all rights, obligations and liabilities hereunder shall be governed by the laws of the State of California.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, this Warrant has been executed as of the date first set forth above.

Agreed and Accepted:
HOLDER:
Ву:
Name:
Title:

Address:

COMPANY:

GROVE COLLABORATIVE, INC.

By: Stuart Landesberg Chief Executive Officer

Address: 1301 Sansome St

San Francisco, CA 94111

NOTICE OF EXERCISE

TO: GROVE COLLABORATIVE, INC. (THE "COMPANY")

(1) [] The undersigned hereby elects to purchasesh Warrant, and tenders herewith payment of the exercise price in full, together.	ares of Common Stock of the Company pursuant to the terms of the attached her with all applicable transfer taxes, if any.
[] The undersigned hereby elects to purchase exercise provisions set forth in Section 2.1 of the attached Warrant, and sl	_ shares of Common Stock of the Company pursuant to the terms of the net hall tender payment of all applicable transfer taxes, if any.
and not with a view to, or for resale in connection with, the distribution the reselling such shares; (ii) the undersigned is aware of the Company's busithe Company to reach an informed and knowledgeable decision regarding investments of this type and has such knowledge and background in finant and risks of this investment and protecting the undersigned's own interest upon exercise of this Warrant have not been registered under the Securitie the registration provisions of the Act, which exemption depends upon, am herein, and, because such securities have not been registered under the Act an exemption from such registration is available; (v) the undersigned is at Rule 144 adopted under the Act unless certain conditions are met and untal, that among the conditions for use of the Rule is the availability of cumade such information available and has no present plans to do so; and (v	ness affairs and financial condition and has acquired sufficient information about its investment in the Company; (iii) the undersigned is experienced in making icial and business matters that the undersigned is capable of evaluating the merits is; (iv) the undersigned understands that the shares of Common Stock issuable is Act of 1933, as amended (the "Act"), by reason of a specific exemption from long other things, the bona fide nature of the investment intent as expressed it, they must be held indefinitely unless subsequently registered under the Act or ware that the aforesaid shares of Common Stock may not be sold pursuant to ill the undersigned has held the shares for the number of years prescribed by Rule rrent information to the public about the Company and the Company has not in the undersigned agrees not to make any disposition of all or any part of the a registration statement under the Act covering such proposed disposition and in the undersigned has provided the Company with an opinion of counsel
(Date)	Holder
	Signature
	Name/Title

ASSIGNMENT FORM

(To assign the foregoing Warrant, execute this form and supply required information. Do not use this form to purchase shares.)

FOR VALUE RECEIVED, the foregoing Warrant and all rights evidenced thereby are hereby assigned to

Name:	
	(Please Print)
Address:	
	(Please Print)
Dated:, 20	
Holder's Signature:	
Holder's Address:	

NOTE: The signature to this Assignment Form must correspond with the name as it appears on the face of the Warrant, without alteration or enlargement or any change whatever. Officers of corporations and those acting in a fiduciary or other representative capacity should file proper evidence of authority to assign the foregoing Warrant.

THIS WARRANT AND THE SHARES ISSUABLE HEREUNDER HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AS SET FORTH IN SECTIONS 5.3 AND 5.4 BELOW, MAY NOT BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND LAWS OR IN FORM AND SUBSTANCE SATISFACTORY TO THE COMPANY, SUCH OFFER, SALE, PLEDGE OR OTHER TRANSFER IS EXEMPT FROM SUCH REGISTRATION.

WARRANT TO PURCHASE STOCK

Company:	GROVE COLLABORATIVE, INC., a Delaware corporation
Number of Shares:	(the " <u>Initial Shares</u> "), plus all Additional Shares which Holder is entitled to purchase pursuant to Section 1.7; provided, however, that the maximum number of Shares issuable hereunder shall not exceed
Type/Series of Stock:	Either (a) Series A Preferred Stock or (b) at Holder's election in its sole and absolute discretion, Next Round Stock; provided, however, that unless and until the Next Equity Round occurs, the Class (as defined below) shall be the Company's Series A Preferred Stock. As used herein, "Next Round Stock" means the series of the Company's preferred equity securities issued in connection with the Company's next bona fide round of preferred stock equity financing that is consummated after the Issue Date of this Warrant (the "Next Equity Round").
Warrant Price:	(a) If Series A Preferred, \$ per share, or (b) if Next Round Stock, the price per Share paid by an investor for a Share of Next Round Stock in connection with the Next Equity Round.
Issue Date:	
Expiration Date:	See also Section 1.6 and 5.1(b).
Credit Facility:	This Warrant to Purchase Stock (as the same may from time to time be amended, modified, supplemented or restated, the " <u>Warrant</u> ") is issued in connection with that certain Agreement of even date herewith between and the Company (as the same may from time to time be amended, modified, supplemented or restated, the " <u>Loan Agreement</u> ").
or transferee of this Warra non-assessable shares (the stated Warrant Price, all a	CERTIFIES THAT, for good and valuable consideration, (together with any successor or permitted assignee and or of any shares issued upon exercise hereof, "Holder") is entitled to purchase up to the number of fully paid and "Shares") of the above-stated Type/Series of Stock (the "Class") of the above-named company (the "Company") at the above-steed forth above and as adjusted pursuant to Section 2 of this Warrant, subject to the provisions and upon the terms and as Warrant. Reference is made to Section 5.4 of this Warrant whereby shall transfer this Warrant to its parent

SECTION 1. EXERCISE.

- 1.1 Method of Exercise. Holder may at any time and from time to time through the Expiration Date exercise this Warrant, in whole or in part, by delivering to the Company the original of this Warrant together with a duly executed Notice of Exercise in substantially the form attached hereto as Appendix 1 and, unless Holder is exercising this Warrant pursuant to a cashless exercise set forth in Section 1.2, a check, wire transfer of same-day funds (to an account designated by the Company), or other form of payment acceptable to the Company for the aggregate Warrant Price for the Shares being purchased.
- 1.2 <u>Cashless Exercise</u>. On any exercise of this Warrant, in lieu of payment of the aggregate Warrant Price in the manner as specified in Section 1.1 above, but otherwise in accordance with the requirements of Section 1.1, Holder may elect to receive Shares equal to the value of this Warrant, or portion hereof as to which this Warrant is being exercised. Thereupon, the Company shall issue to the Holder such number of fully paid and non-assessable Shares as are computed using the following formula:

X = Y(A-B)/A

where:

- X = the number of Shares to be issued to the Holder;
- Y = the number of Shares with respect to which this Warrant is being exercised (inclusive of the Shares surrendered to the Company in payment of the aggregate Warrant Price);
- A = the Fair Market Value (as determined pursuant to Section 1.3 below) of one Share; and
- B = the Warrant Price.
- 1.3 Fair Market Value. If the Company's common stock is then traded or quoted on a nationally recognized securities exchange, inter-dealer quotation system or over-the-counter market (a "Trading Market") and the Class is common stock, the fair market value of a Share shall be the closing price or last sale price of a share of common stock reported for the Business Day immediately before the date on which Holder delivers this Warrant together with its Notice of Exercise to the Company. If the Company's common stock is then traded in a Trading Market and the Class is a series of the Company's convertible preferred stock, the fair market value of a Share shall be the closing price or last sale price of a share of the Company's common stock reported for the Business Day immediately before the date on which Holder delivers this Warrant together with its Notice of Exercise to the Company multiplied by the number of shares of the Company's common stock into which a Share is then convertible. If the Company's common stock is not traded in a Trading Market, the Board of Directors of the Company shall determine the fair market value of a Share in its reasonable good faith judgment.

- 1.4 <u>Delivery of Certificate and New Warrant</u>. Within a reasonable time after Holder exercises this Warrant in the manner set forth in Section 1.1 or 1.2 above, the Company shall deliver to Holder a certificate representing the Shares issued to Holder upon such exercise and, if this Warrant has not been fully exercised and has not expired, a new warrant of like tenor representing the Shares not so acquired.
- 1.5 <u>Replacement of Warrant</u>. On receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant and, in the case of loss, theft or destruction, on delivery of an indemnity agreement reasonably satisfactory in form, substance and amount to the Company or, in the case of mutilation, on surrender of this Warrant to the Company for cancellation, the Company shall, within a reasonable time, execute and deliver to Holder, in lieu of this Warrant, a new warrant of like tenor and amount.

1.6 Treatment of Warrant Upon Acquisition of Company.

- (a) Acquisition. For the purpose of this Warrant, "Acquisition" means any transaction or series of related transactions involving: (i) the sale, lease, exclusive license, or other disposition of all or substantially all of the assets of the Company (ii) any merger or consolidation of the Company into or with another person or entity (other than a merger or consolidation effected exclusively to change the Company's domicile), or any other corporate reorganization, in which the stockholders of the Company in their capacity as such immediately prior to such merger, consolidation or reorganization, own less than a majority of the Company's (or the surviving or successor entity's) outstanding voting power immediately after such merger, consolidation or reorganization; or (iii) any sale or other transfer by the stockholders of the Company of shares representing at least a majority of the Company's then-total outstanding combined voting power; provided that, "Acquisition" shall not include any transaction or series of transactions in which the Company sells and issues its capital stock to investors solely for capital raising purposes in a bona fide round of equity financing.
- (b) Treatment of Warrant at Acquisition. In the event of an Acquisition in which the consideration to be received by the Company's stockholders consists solely of cash, solely of Marketable Securities or a combination of cash and Marketable Securities (a "Cash/Public Acquisition"), and the fair market value of one Share as determined in accordance with Section 1.3 above would be greater than the Warrant Price in effect on such date immediately prior to such Cash/Public Acquisition, and Holder has not exercised this Warrant pursuant to Section 1.1 above as to all Shares, then this Warrant shall automatically be deemed to be cashless exercised pursuant to Section 1.2 above as to all Shares effective immediately prior to and contingent upon the consummation of a Cash/Public Acquisition. In connection with such cashless exercise, Holder shall be deemed to have restated each of the representations and warranties in Section 4 of the Warrant as the date thereof and the Company shall promptly notify the Holder of the number of Shares (or such other securities) issued upon cashless exercise. In the event of a Cash/Public Acquisition where the fair market value of one Share as determined in accordance with Section 1.3 above would be less than the Warrant Price in effect immediately prior to such Cash/Public Acquisition, then this Warrant will expire immediately prior to the consummation of such Cash/Public Acquisition.

- (c) Upon the closing of any Acquisition other than a Cash/Public Acquisition defined above, the acquiring, surviving or successor entity shall assume the obligations of this Warrant, and this Warrant shall thereafter be exercisable for the same securities and/or other property as would have been paid for the Shares issuable upon exercise of the unexercised portion of this Warrant as if such Shares were outstanding on and as of the closing of such Acquisition, subject to further adjustment from time to time in accordance with the provisions of this Warrant.
- (d) As used in this Warrant, "Marketable Securities" means securities meeting all of the following requirements: (i) the issuer thereof is then subject to the reporting requirements of Section 13 or Section 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and is then current in its filing of all required reports and other information under the Act and the Exchange Act; (ii) the class and series of shares or other security of the issuer that would be received by Holder in connection with the Acquisition were Holder to exercise this Warrant on or prior to the closing thereof is then traded in Trading Market, and (iii) following the closing of such Acquisition, Holder would not be restricted from publicly re-selling all of the issuer's shares and/or other securities that would be received by Holder in such Acquisition were Holder to exercise or convert this Warrant in full on or prior to the closing of such Acquisition, except to the extent that any such restriction (x) arises solely under federal or state securities laws, rules or regulations, and (y) does not extend beyond six (6) months from the closing of such Acquisition.
- 1.7 Additional Shares. In addition to the right to purchase the Initial Shares granted to Holder on the Issue Date, on the Funding Date of each Growth Capital Advance under the Loan Agreement, the Company shall be deemed to have automatically granted to Holder, in addition to the number of Shares for which this Warrant can otherwise be exercised by Holder, the right to purchase, at an exercise price per share equal to the Warrant Price, that number of additional shares (rounded down to the nearest whole Share) equal to ______ multiplied by the original principal amount of each such Growth Capital Advance (all such additional Shares being called the "Additional Shares"); provided, however, that the maximum number of Shares issuable hereunder shall not exceed ______. Capitalized terms used but not defined in this Section 1.7 shall have the meanings given to them in the Loan Agreement.

SECTION 2. ADJUSTMENTS TO THE SHARES AND WARRANT PRICE

- 2.1 Stock Dividends, Splits, Etc. If the Company declares or pays a dividend or distribution on the outstanding shares of the Class payable in common stock or other securities or property (other than cash), then upon exercise of this Warrant, for each Share acquired, Holder shall receive, without additional cost to Holder, the total number and kind of securities and property which Holder would have received had Holder owned the Shares of record as of the date the dividend or distribution occurred. If the Company subdivides the outstanding shares of the Class by reclassification or otherwise into a greater number of shares, the number of Shares purchasable hereunder shall be proportionately increased and the Warrant Price shall be proportionately decreased. If the outstanding shares of the Class are combined or consolidated, by reclassification or otherwise, into a lesser number of shares, the Warrant Price shall be proportionately increased and the number of Shares shall be proportionately decreased.
- 2.2 <u>Reclassification, Exchange, Combinations or Substitution.</u> Upon any event whereby all of the outstanding shares of the Class are reclassified, exchanged, combined, substituted, or replaced for, into, with or by Company securities of a different class and/or series, then from and after the consummation of such event, this Warrant will be exercisable for the number, class and series of Company securities that Holder would have received had the Shares been outstanding on and as of the consummation of such event, and subject to further adjustment thereafter from time to time in accordance with the provisions of this Warrant. The provisions of this Section 2.2 shall similarly apply to successive reclassifications, exchanges, combinations substitutions, replacements or other similar events.

- 2.3 Conversion of Preferred Stock. If the Class is a class and series of the Company's convertible preferred stock, in the event that all outstanding shares of the Class are converted, automatically or by action of the holders thereof, into common stock pursuant to the provisions of the Company's Certificate of Incorporation, including, without limitation, in connection with the Company's initial, underwritten public offering and sale of its common stock pursuant to an effective registration statement under the Act (the "IPO"), then from and after the date on which all outstanding shares of the Class have been so converted, this Warrant shall be exercisable for such number of shares of common stock into which the Shares would have been converted had the Shares been outstanding on the date of such conversion, and the Warrant Price shall equal the Warrant Price in effect as of immediately prior to such conversion divided by the number of shares of common stock into which one Share would have been converted, all subject to further adjustment thereafter from time to time in accordance with the provisions of this Warrant.
- 2.4 <u>Adjustments for Diluting Issuances</u>. Without duplication of any adjustment otherwise provided for in this Section 2, the number of shares of common stock issuable upon conversion of the Shares shall be subject to anti-dilution adjustment from time to time in the manner set forth in the Company's Certificate of Incorporation as if the Shares were issued and outstanding on and as of the date of any such required adjustment.
- 2.5 No Fractional Share. No fractional Share shall be issuable upon exercise of this Warrant and the number of Shares to be issued shall be rounded down to the nearest whole Share. If a fractional Share interest arises upon any exercise of the Warrant, the Company shall eliminate such fractional Share interest by paying Holder in cash the amount computed by multiplying the fractional interest by (i) the fair market value (as determined in accordance with Section 1.3 above) of a full Share, less (ii) the then-effective Warrant Price.
- 2.6 Notice/Certificate as to Adjustments. Upon each adjustment of the Warrant Price, Class and/or number of Shares, the Company, at the Company's expense, shall notify Holder in writing within a reasonable time setting forth the adjustments to the Warrant Price, Class and/or number of Shares and facts upon which such adjustment is based. The Company shall, upon written request from Holder, furnish Holder with a certificate of its Chief Financial Officer, including computations of such adjustment and the Warrant Price, Class and number of Shares in effect upon the date of such adjustment.

SECTION 3. REPRESENTATIONS AND COVENANTS OF THE COMPANY.

3.1 Representations and Warranties. The Company represents and warrants to, and agrees with, the Holder as follows:

(a) The initial Warrant Price referenced on the first page of this Warrant is not greater than the price per share at which shares of the Class were last sold and issued prior to the Issue Date hereof in an arms-length transaction in which at least \$_____ of such shares were sold.

(b) All Shares which may be issued upon the exercise of this Warrant, and all securities, if any, issuable upon conversion of the Shares, shall
upon issuance, be duly authorized, validly issued, fully paid and non-assessable, and free of any liens and encumbrances except for restrictions on transfer
provided for herein or under applicable federal and state securities laws. The Company covenants that it shall at all times cause to be reserved and kept
available out of its authorized and unissued capital stock such number of shares of the Class, common stock and other securities as will be sufficient to
permit the exercise in full of this Warrant and the conversion of the Shares into common stock or such other securities; provided, however, that the
Company shall not authorize the Next Round Stock unless and until the closing of the Next Equity Round.

- (c) The Company's capitalization table attached hereto as Schedule 1 is true and complete, in all material respects, as of the Issue Date.
- 3.2 Notice of Certain Events. If the Company proposes at any time to:
- (a) declare any dividend or distribution upon the outstanding shares of the Class or common stock, whether in cash, property, stock, or other securities and whether or not a regular cash dividend;
- (b) offer for subscription or sale pro rata to the holders of the outstanding shares of the Class any additional shares of any class or series of the Company's stock (other than pursuant to contractual pre-emptive rights);
 - (c) effect any reclassification, exchange, combination, substitution, reorganization or recapitalization of the outstanding shares of the Class;
 - (d) effect an Acquisition or to liquidate, dissolve or wind up; or
 - (e) effect an IPO;

then, in connection with each such event, the Company shall give Holder:

- (1) at least _____ Business Days prior written notice of the date on which a record will be taken for such dividend, distribution, or subscription rights (and specifying the date on which the holders of outstanding shares of the Class will be entitled thereto) or for determining rights to vote, if any, in respect of the matters referred to in (a) and (b) above;
- (2) in the case of the matters referred to in (c) and (d) above at least ______ (__) Business Days prior written notice of the date when the same will take place (and specifying the date on which the holders of outstanding shares of the Class will be entitled to exchange their shares for the securities or other property deliverable upon the occurrence of such event and such reasonable information as Holder may reasonably require regarding the treatment of this Warrant in connection with such event giving rise to the notice); and

(3) with respect to the IPO, at least) Business Days prior written notice of the date on which the Company proposes to file	its
registration statement in connection therewith.		

Company will also provide information requested by Holder that is reasonably necessary to enable Holder to comply with Holder's accounting or reporting requirements.

SECTION 4. REPRESENTATIONS, WARRANTIES OF THE HOLDER.

The Holder represents and warrants to the Company as follows:

- 4.1 <u>Purchase for Own Account</u>. This Warrant and the Shares to be acquired upon exercise of this Warrant by Holder are being acquired for investment for Holder's account, not as a nominee or agent, and not with a view to the public resale or distribution within the meaning of the Act. Holder also represents that it has not been formed for the specific purpose of acquiring this Warrant or the Shares.
- 4.2 <u>Disclosure of Information</u>. Holder is aware of the Company's business affairs and financial condition and has received or has had full access to all the information it considers necessary or appropriate to make an informed investment decision with respect to the acquisition of this Warrant and its underlying securities. Holder further has had an opportunity to ask questions and receive answers from the Company regarding the terms and conditions of the offering of this Warrant and its underlying securities and to obtain additional information (to the extent the Company possessed such information or could acquire it without unreasonable effort or expense) necessary to verify any information furnished to Holder or to which Holder has access.
- 4.3 <u>Investment Experience</u>. Holder understands that the purchase of this Warrant and its underlying securities involves substantial risk. Holder has experience as an investor in securities of companies in the development stage and acknowledges that Holder can bear the economic risk of such Holder's investment in this Warrant and its underlying securities and has such knowledge and experience in financial or business matters that Holder is capable of evaluating the merits and risks of its investment in this Warrant and its underlying securities and/or has a preexisting personal or business relationship with the Company and certain of its officers, directors or controlling persons of a nature and duration that enables Holder to be aware of the character, business acumen and financial circumstances of such persons.
 - 4.4 Accredited Investor Status. Holder is an "accredited investor" within the meaning of Regulation D promulgated under the Act.
- 4.5 The Act. Holder understands that this Warrant and the Shares issuable upon exercise hereof have not been registered under the Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of the Holder's investment intent as expressed herein. Holder understands that this Warrant and the Shares issued upon any exercise hereof must be held indefinitely unless subsequently registered under the Act and qualified under applicable state securities laws, or unless exemption from such registration and qualification are otherwise available. Holder is aware of the provisions of Rule 144 promulgated under the Act.

- 4.6 <u>Market Stand-off Agreement</u>. The Holder agrees that the Shares shall be subject to the Market Standoff provisions in Section 2.11 of the Investor Rights Agreement, as may be amended and/or restated from time to time.
- 4.7 No Shareholder Rights. Holder, as a Holder of this Warrant, will not have any rights as a shareholder of the Company (including, but not limited to, voting rights) in respect of the Shares issuable hereunder unless and until the exercise of this Warrant.

SECTION 5. MISCELLANEOUS.

- 5.1 Term and Automatic Cashless Exercise Upon Expiration.
- (a) <u>Term</u>. Subject to the provisions of Section 1.6 above, this Warrant is exercisable in whole or in part at any time and from time to time on or before 6:00 PM, Pacific time, on the Expiration Date and shall be void thereafter.
- (b) <u>Automatic Cashless Exercise upon Expiration</u>. In the event that, upon the Expiration Date, the fair market value of one Share (or other security issuable upon the exercise hereof) as determined in accordance with Section 1.3 above is greater than the Warrant Price in effect on such date, then this Warrant, to the extent unexercised, shall automatically be deemed on and as of such date to be exercised pursuant to Section 1.2 above as to all Shares (or such other securities) for which it shall not previously have been exercised, and the Company shall, within a reasonable time, deliver a certificate representing the Shares (or such other securities) issued upon such exercise to Holder.
- 5.2 <u>Legends</u>. The Shares (and the securities issuable, directly or indirectly, upon conversion of the Shares, if any) shall be imprinted with a legend in substantially the following form:

5.3 <u>Compliance with Securities Laws on Transfer.</u> This Warrant and the Shares issuable upon exercise of this Warrant (and the securities issuable, directly or indirectly, upon conversion of the Shares, if any) may not be transferred or assigned in whole or in part except in compliance with applicable federal and state securities laws by the transferor and the transferee (including, without limitation, the delivery of investment representation letters and legal opinions reasonably satisfactory to the Company, as reasonably requested by the Company). The Company

shall not require Holder to provide an opinion of couthat any such transferee is an "accredited investor" as require an opinion of counsel if there is no material q	s defined in Regulation D promulgated und	or any other affiliate of Holder, provided ler the Act. Additionally, the Company shall also not romulgated under the Act.
company, By its acceptance of this set forth in Section 4 hereof and agrees to be bound by provisions of Section 5.3 and upon providing the Cotthis Warrant or the Shares issuable upon exercise of any transferee, provided, however, in connection with portion of the Warrant being transferred with the name Warrant to the Company for reissuance to the transference shall agree in writing with the Comprovision herein, at all times prior to the IPO, Holder any Shares issued upon any exercise hereof, or any significant contents.	s Warrant,hereby makes by all of the terms and conditions of this W mpany with written notice,	_ and any subsequent Holder may transfer all or part of ctly or indirectly, upon conversion of the Shares, if any) to my subsequent Holder will give the Company notice of the
effective (i) when given personally, (ii) on the (iii) upon actual receipt if given by facsimile or elect following delivery to a reliable overnight courier service.	_ () Business Day after being mailed by ronic mail and such receipt is confirmed in vice, courier fee prepaid, in any case at such yor such Holder from time to time in acc	folder, or vice versa, shall be deemed delivered and first-class registered or certified mail, postage prepaid, writing by the recipient, or (iv) on the first Business Day h address as may have been furnished to the Company or ordance with the provisions of this Section 5.5. All notices in connection with a transfer or otherwise:

Attn:		
Γelephone:		
Facsimile:		
Email address:		
inan address:	 	

Notice to the Company shall be addressed as follows until Holder receives notice of a change in address:

Grove Collaborative, Inc. Attn: Stuart Landesberg, Chief Executive Officer 1301 Sansome Street San Francisco, California 94111 Email: sl@grove.co

- 5.6 <u>Waiver</u>. This Warrant and any term hereof may be changed, waived, discharged or terminated (either generally or in a particular instance and either retroactively or prospectively) only by an instrument in writing signed by the party against which enforcement of such change, waiver, discharge or termination is sought.
- 5.7 Attorney's Fees. In the event of any dispute between the parties concerning the terms and provisions of this Warrant, the party prevailing in such dispute shall be entitled to collect from the other party all costs incurred in such dispute, including reasonable attorneys' fees.
- 5.8 <u>Counterparts: Facsimile/Electronic Signatures</u>. This Warrant may be executed in counterparts, all of which together shall constitute one and the same agreement. Any signature page delivered electronically or by facsimile shall be binding to the same extent as an original signature page with regards to any agreement subject to the terms hereof or any amendment thereto.
- 5.9 Governing Law. This Warrant shall be governed by and construed in accordance with the laws of the State of California, without giving effect to its principles regarding conflicts of law.
- 5.10 <u>Headings</u>. The headings in this Warrant are for purposes of reference only and shall not limit or otherwise affect the meaning of any provision of this Warrant.
 - 5.11 <u>Business Days</u>. "<u>Business Day</u>" is any day that is not a Saturday, Sunday or a day on which ______ is closed.

[Remainder of page left blank intentionally] [Signature page follows]

as of the Issue Date written above.	•	,	1
"COMPANY"			
GROVE COLLABORATIVE, INC.			
By: Name: Stuart Landesberg Title: Chief Executive Officer			
"HOLDER"			
By:			
Name:			
Title			

[Signature Page to Warrant to Purchase Stock]

IN WITNESS WHEREOF, the parties have caused this Warrant to Purchase Stock to be executed by their duly authorized representatives effective

APPENDIX 1

NOTICE OF EXERCISE

	ereby exercises its right purchase		
[] check in the amount of	of \$ payable to order of the Co	mpany enclosed herewith	
[] Wire transfer of imm	ediately available funds to the Company's	account	
[] Cashless Exercise pu	rsuant to Section 1.2 of the Warrant		
[] Other [Describe]			
2. Please issue a certificate or	r certificates representing the Shares in the	name specified below:	
	Holder's Name		
	(Address)		
3. By its execution below and Warrant to Purchase Stock as of the	d for the benefit of the Company, Holder he e date hereof.	ereby restates each of the representation	as and warranties in Section 4 of the
		HOLDER:	
		Ву:	
		Name:	
		Title:	
		(Date):	

Appendix 1

SCHEDULE 1

Company Capitalization Table

See attached

Schedule 1

THIS WARRANT AND THE SHARES ISSUABLE HEREUNDER HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AS SET FORTH IN SECTIONS 5.3 AND 5.4 BELOW, MAY NOT BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND LAWS OR IN FORM AND SUBSTANCE SATISFACTORY TO THE COMPANY, SUCH OFFER, SALE, PLEDGE OR OTHER TRANSFER IS EXEMPT FROM SUCH REGISTRATION.

WARRANT TO PURCHASE STOCK

Company:	GROVE COLLABORATIVE, INC., a Delaware corporation
Number of Shares:	(the " <u>Initial Shares</u> "), plus all Additional Shares which Holder is entitled to purchase pursuant to Section 1.7; provided, however, that the maximum number of Shares issuable hereunder shall not exceed
Type/Series of Stock:	Either (a) Series B Preferred Stock or (b) at Holder's election in its sole and absolute discretion, Next Round Stock; provided, however, that unless and until the Next Equity Round occurs, the Class (as defined below) shall be the Company's Series B Preferred Stock. As used herein, "Next Round Stock" means the series of the Company's preferred equity securities issued in connection with the Company's next bona fide round of preferred stock equity financing that is consummated after the Issue Date of this Warrant (the "Next Equity Round").
Warrant Price:	(a) If Series B Preferred, \$ per share, or (b) if Next Round Stock, the price per Share paid by an investor for a Share o Next Round Stock in connection with the Next Equity Round.
Issue Date:	
Expiration Date:	See also Section 1.6 and 5.1(b).
Credit Facility:	This Warrant to Purchase Stock (as the same may from time to time be amended, modified, supplemented or restated, the "Warrant") is issued in connection with that certain Agreement of even date herewith between and the Company, which amends the certain Agreement dated as of between and the Company (as the same may from time to time be amended, modified, supplemented or restated, the "Loan Agreement").
THIS WARRANT C	ERTIFIES THAT, for good and valuable consideration, (together with any successor or permitted assignee or

SECTION 1. EXERCISE.

- 1.1 Method of Exercise. Holder may at any time and from time to time through the Expiration Date exercise this Warrant, in whole or in part, by delivering to the Company the original of this Warrant together with a duly executed Notice of Exercise in substantially the form attached hereto as Appendix 1 and, unless Holder is exercising this Warrant pursuant to a cashless exercise set forth in Section 1.2, a check, wire transfer of same-day funds (to an account designated by the Company), or other form of payment acceptable to the Company for the aggregate Warrant Price for the Shares being purchased.
- 1.2 <u>Cashless Exercise</u>. On any exercise of this Warrant, in lieu of payment of the aggregate Warrant Price in the manner as specified in Section 1.1 above, but otherwise in accordance with the requirements of Section 1.1, Holder may elect to receive Shares equal to the value of this Warrant, or portion hereof as to which this Warrant is being exercised, if the Fair Market Value per Share exceeds the Warrant Price per Share. Thereupon, the Company shall issue to the Holder such number of fully paid and non-assessable Shares as are computed using the following formula, rounded down to the nearest whole Share:

X = Y(A-B)/A

where:

X = the number of Shares to be issued to the Holder;

Y = the number of Shares with respect to which this Warrant is being exercised (inclusive of the Shares surrendered to the Company in payment of the aggregate Warrant Price);

A = the Fair Market Value (as determined pursuant to Section 1.3 below) of one Share; and

B = the Warrant Price.

1.3 Fair Market Value. If the Company's common stock is then traded or quoted on a nationally recognized securities exchange, inter-dealer quotation system or over-the-counter market (a "Trading Market") and the Class is common stock, the fair market value of a Share shall be the closing price or last sale price of a share of common stock reported for the Business Day immediately before the date on which Holder delivers this Warrant together with its Notice of Exercise to the Company. If the Company's common stock is then traded in a Trading Market and the Class is a series of the Company's convertible preferred stock, the fair market value of a Share shall be the closing price or last sale price of a share of the Company's common stock reported for the Business Day immediately before the date on which Holder delivers this Warrant together with its Notice of Exercise to the Company multiplied by the number of shares of the Company's common stock into which a Share is then convertible. If the Company's common stock is not traded in a Trading Market, the Board of Directors of the Company shall determine the fair market value of a Share in its reasonable good faith judgment.

- 1.4 <u>Delivery of Certificate and New Warrant</u>. Within a reasonable time after Holder exercises this Warrant in the manner set forth in Section 1.1 or 1.2 above, the Company shall deliver to Holder a certificate representing the Shares issued to Holder upon such exercise and, if this Warrant has not been fully exercised and has not expired, a new warrant of like tenor representing the Shares not so acquired.
- 1.5 <u>Replacement of Warrant</u>. On receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant and, in the case of loss, theft or destruction, on delivery of an indemnity agreement reasonably satisfactory in form, substance and amount to the Company or, in the case of mutilation, on surrender of this Warrant to the Company for cancellation, the Company shall, within a reasonable time, execute and deliver to Holder, in lieu of this Warrant, a new warrant of like tenor and amount.

1.6 Treatment of Warrant Upon Acquisition of Company.

- (a) Acquisition. For the purpose of this Warrant, "Acquisition" means any transaction or series of related transactions involving: (i) the sale, lease, exclusive license, or other disposition of all or substantially all of the assets of the Company (ii) any merger or consolidation of the Company into or with another person or entity (other than a merger or consolidation effected exclusively to change the Company's domicile), or any other corporate reorganization, in which the stockholders of the Company in their capacity as such immediately prior to such merger, consolidation or reorganization, own less than a majority of the Company's (or the surviving or successor entity's) outstanding voting power immediately after such merger, consolidation or reorganization; or (iii) any sale or other transfer by the stockholders of the Company of shares representing at least a majority of the Company's then-total outstanding combined voting power; provided that, "Acquisition" shall not include any transaction or series of transactions in which the Company sells and issues its capital stock to investors solely for capital raising purposes in a bona fide round of equity financing.
- (b) Treatment of Warrant at Acquisition. In the event of an Acquisition in which the consideration to be received by the Company's stockholders consists solely of cash, solely of Marketable Securities or a combination of cash and Marketable Securities (a "Cash/Public Acquisition"), and the fair market value of one Share as determined in accordance with Section 1.3 above would be greater than the Warrant Price in effect on such date immediately prior to such Cash/Public Acquisition, and Holder has not exercised this Warrant pursuant to Section 1.1 above as to all Shares, then this Warrant shall automatically be deemed to be cashless exercised pursuant to Section 1.2 above as to all Shares effective immediately prior to and contingent upon the consummation of a Cash/Public Acquisition. In connection with such cashless exercise, Holder shall be deemed to have restated each of the representations and warranties in Section 4 of the Warrant as the date thereof and the Company shall promptly notify the Holder of the number of Shares (or such other securities) issued upon cashless exercise. In the event of a Cash/Public Acquisition where the fair market value of one Share as determined in accordance with Section 1.3 above would be less than the Warrant Price in effect immediately prior to such Cash/Public Acquisition, then this Warrant will expire immediately prior to the consummation of such Cash/Public Acquisition.

been paid for the Shares issuable upon exercise of the unexercised portion of this Warrant as if such Shares were outstanding on and as of the closing of such Acquisition, subject to further adjustment from time to time in accordance with the provisions of this Warrant.

(d) As used in this Warrant, "Marketable Securities" means securities meeting all of the following requirements: (i) the issuer thereof is then subject to the reporting requirements of Section 13 or Section 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and is then current in its filing of all required reports and other information under the Act and the Exchange Act; (ii) the class and series of shares or other security of the issuer that would be received by Holder in connection with the Acquisition were Holder to exercise this Warrant on or prior to the closing thereof is then traded in Trading Market, and (iii) following the closing of such Acquisition, Holder would not be restricted from publicly re-selling all of the issuer's shares and/or other securities that would be received by Holder in such Acquisition were Holder to exercise or convert this Warrant in full on or prior to the closing of such Acquisition, except to the extent that any such restriction (x) arises solely under federal or state securities laws, rules or regulations, and (y) does not extend beyond ______ (___) months from the closing of such Acquisition.

1.7 Additional Shares. In addition to the right to purchase the Initial Shares granted to Holder on the Issue Date, on the Funding Date of each Supplemental Growth Capital Advance under the Loan Agreement, the Company shall be deemed to have automatically granted to Holder, in addition to the number of Shares for which this Warrant can otherwise be exercised by Holder, the right to purchase, at an exercise price per share equal to the

shall assume the obligations of this Warrant, and this Warrant shall thereafter be exercisable for the same securities and/or other property as would have

(c) Upon the closing of any Acquisition other than a Cash/Public Acquisition defined above, the acquiring, surviving or successor entity

SECTION 2. ADJUSTMENTS TO THE SHARES AND WARRANT PRICE

of each such Growth Capital Advance (all such additional Shares being called the "<u>Additional Shares</u>"); provided, however, that the maximum number of Additional Shares issuable hereunder shall not exceed

Capitalized terms used but not defined in this Section 1.7 shall have the meanings given

multiplied by the original principal amount

Warrant Price, that number of additional shares (rounded down to the nearest whole Share) equal to

to them in the Loan Agreement.

2.1 Stock Dividends, Splits, Etc. If the Company declares or pays a dividend or distribution on the outstanding shares of the Class payable in common stock or other securities or property (other than cash), then upon exercise of this Warrant, for each Share acquired, Holder shall receive, without additional cost to Holder, the total number and kind of securities and property which Holder would have received had Holder owned the Shares of record as of the date the dividend or distribution occurred. If the Company subdivides the outstanding shares of the Class by reclassification or otherwise into a greater number of shares, the number of Shares purchasable hereunder shall be proportionately increased and the Warrant Price shall be proportionately decreased. If the outstanding shares of the Class are combined or consolidated, by reclassification or otherwise, into a lesser number of shares, the Warrant Price shall be proportionately increased and the number of Shares shall be proportionately decreased.

- 2.2 <u>Reclassification</u>, <u>Exchange</u>, <u>Combinations or Substitution</u>. Upon any event whereby all of the outstanding shares of the Class are reclassified, exchanged, combined, substituted, or replaced for, into, with or by Company securities of a different class and/or series, then from and after the consummation of such event, this Warrant will be exercisable for the number, class and series of Company securities that Holder would have received had the Shares been outstanding on and as of the consummation of such event, and subject to further adjustment thereafter from time to time in accordance with the provisions of this Warrant. The provisions of this Section 2.2 shall similarly apply to successive reclassifications, exchanges, combinations substitutions, replacements or other similar events.
- 2.3 Conversion of Preferred Stock. If the Class is a class and series of the Company's convertible preferred stock, in the event that all outstanding shares of the Class are converted, automatically or by action of the holders thereof, into common stock pursuant to the provisions of the Company's Certificate of Incorporation, including, without limitation, in connection with the Company's initial, underwritten public offering and sale of its common stock pursuant to an effective registration statement under the Act (the "IPO"), then from and after the date on which all outstanding shares of the Class have been so converted, this Warrant shall be exercisable for such number of shares of common stock into which the Shares would have been converted had the Shares been outstanding on the date of such conversion, and the Warrant Price shall equal the Warrant Price in effect as of immediately prior to such conversion divided by the number of shares of common stock into which one Share would have been converted, all subject to further adjustment thereafter from time to time in accordance with the provisions of this Warrant.
- 2.4 <u>Adjustments for Diluting Issuances</u>. Without duplication of any adjustment otherwise provided for in this Section 2, the number of shares of common stock issuable upon conversion of the Shares shall be subject to anti-dilution adjustment from time to time in the manner set forth in the Company's Certificate of Incorporation as if the Shares were issued and outstanding on and as of the date of any such required adjustment.
- 2.5 No Fractional Share. No fractional Share shall be issuable upon exercise of this Warrant and the number of Shares to be issued shall be rounded down to the nearest whole Share. If a fractional Share interest arises upon any exercise of the Warrant, the Company shall eliminate such fractional Share interest by paying Holder in cash the amount computed by multiplying the fractional interest by (i) the fair market value (as determined in accordance with Section 1.3 above) of a full Share, less (ii) the then-effective Warrant Price.
- 2.6 Notice/Certificate as to Adjustments. Upon each adjustment of the Warrant Price, Class and/or number of Shares, the Company, at the Company's expense, shall notify Holder in writing within a reasonable time setting forth the adjustments to the Warrant Price, Class and/or number of Shares and facts upon which such adjustment is based. The Company shall, upon written request from Holder, furnish Holder with a certificate of its Chief Financial Officer, including computations of such adjustment and the Warrant Price, Class and number of Shares in effect upon the date of such adjustment.

SECTION 3. REPRESENTATIONS AND COVENANTS OF THE COMPANY.

- 3.1 Representations and Warranties. The Company represents and warrants to, and agrees with, the Holder as follows:
- (a) The initial Warrant Price referenced on the first page of this Warrant is not greater than the price per share at which shares of the Class were last sold and issued prior to the Issue Date hereof in an arms-length transaction in which at least \$______ of such shares were sold.
- (b) All Shares which may be issued upon the exercise of this Warrant, and all securities, if any, issuable upon conversion of the Shares, shall, upon issuance, be duly authorized, validly issued, fully paid and non-assessable, and free of any liens and encumbrances except for restrictions on transfer provided for herein or under applicable federal and state securities laws. The Company covenants that it shall at all times cause to be reserved and kept available out of its authorized and unissued capital stock such number of shares of the Class, common stock and other securities as will be sufficient to permit the exercise in full of this Warrant and the conversion of the Shares into common stock or such other securities; provided, however, that the Company shall not authorize the Next Round Stock unless and until the closing of the Next Equity Round.
 - (c) The Company's capitalization table attached hereto as Schedule 1 is true and complete, in all material respects, as of the Issue Date.
 - 3.2 Notice of Certain Events. If the Company proposes at any time to:
- (a) declare any dividend or distribution upon the outstanding shares of the Class or common stock, whether in cash, property, stock, or other securities and whether or not a regular cash dividend;
- (b) offer for subscription or sale pro rata to the holders of the outstanding shares of the Class any additional shares of any class or series of the Company's stock (other than pursuant to contractual pre-emptive rights);
 - (c) effect any reclassification, exchange, combination, substitution, reorganization or recapitalization of the outstanding shares of the Class;
 - (d) effect an Acquisition or to liquidate, dissolve or wind up; or
 - (e) effect an IPO;

then, in connection with each such event, the Company shall give Holder:

(1) at least ____ (_) Business Days prior written notice of the date on which a record will be taken for such dividend, distribution, or subscription rights (and specifying the date on which the holders of outstanding shares of the Class will be entitled thereto) or for determining rights to vote, if any, in respect of the matters referred to in (a) and (b) above;

(2) in the case of the matters referred to in (c) and (d) above at least () Business Days prior written notice of the date when the same
will take place (and specifying the date on which the holders of outstanding shares of the Class will be entitled to exchange their shares for the
securities or other property deliverable upon the occurrence of such event and such reasonable information as Holder may reasonably require
regarding the treatment of this Warrant in connection with such event giving rise to the notice); and
(3) with respect to the IPO, at least () Business Days prior written notice of the date on which the Company proposes to file its
registration statement in connection therewith.

Company will also provide information requested by Holder that is reasonably necessary to enable Holder to comply with Holder's accounting or reporting requirements.

SECTION 4. REPRESENTATIONS, WARRANTIES OF THE HOLDER.

The Holder represents and warrants to the Company as follows:

- 4.1 <u>Purchase for Own Account</u>. This Warrant and the Shares to be acquired upon exercise of this Warrant by Holder are being acquired for investment for Holder's account, not as a nominee or agent, and not with a view to the public resale or distribution within the meaning of the Act. Holder also represents that it has not been formed for the specific purpose of acquiring this Warrant or the Shares.
- 4.2 <u>Disclosure of Information</u>. Holder is aware of the Company's business affairs and financial condition and has received or has had full access to all the information it considers necessary or appropriate to make an informed investment decision with respect to the acquisition of this Warrant and its underlying securities. Holder further has had an opportunity to ask questions and receive answers from the Company regarding the terms and conditions of the offering of this Warrant and its underlying securities and to obtain additional information (to the extent the Company possessed such information or could acquire it without unreasonable effort or expense) necessary to verify any information furnished to Holder or to which Holder has access.
- 4.3 <u>Investment Experience</u>. Holder understands that the purchase of this Warrant and its underlying securities involves substantial risk. Holder has experience as an investor in securities of companies in the development stage and acknowledges that Holder can bear the economic risk of such Holder's investment in this Warrant and its underlying securities and has such knowledge and experience in financial or business matters that Holder is capable of evaluating the merits and risks of its investment in this Warrant and its underlying securities and/or has a preexisting personal or business relationship with the Company and certain of its officers, directors or controlling persons of a nature and duration that enables Holder to be aware of the character, business acumen and financial circumstances of such persons.
 - 4.4 Accredited Investor Status. Holder is an "accredited investor" within the meaning of Regulation D promulgated under the Act.

4.5 The Act. Holder understands that this Warrant and the Shares issuable upon exercise hereof have not been registered under the Act in reliance
upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of the Holder's investment intent as
expressed herein. Holder understands that this Warrant and the Shares issued upon any exercise hereof must be held indefinitely unless subsequently
registered under the Act and qualified under applicable state securities laws, or unless exemption from such registration and qualification are otherwise
available. Holder is aware of the provisions of Rule 144 promulgated under the Act.

4.6 Market Stand-off A	greement. The Holder agrees that the Shares shall be subject to the Market Stand	doff provisions in Section of	the
	Agreement by and between the Company and the parties thereto, dated	, as may be amende	d and/or
estated from time to time			

4.7 No Shareholder Rights. Holder, as a Holder of this Warrant, will not have any rights as a shareholder of the Company (including, but not limited to, voting rights) in respect of the Shares issuable hereunder unless and until the exercise of this Warrant.

SECTION 5. MISCELLANEOUS.

5.1 Term and Automatic Cashless Exercise Upon Expiration.

- (a) <u>Term.</u> Subject to the provisions of Section 1.6 above, this Warrant is exercisable in whole or in part at any time and from time to time on or before ______ PM, Pacific time, on the Expiration Date and shall be void thereafter.
- (b) <u>Automatic Cashless Exercise upon Expiration</u>. In the event that, upon the Expiration Date, the fair market value of one Share (or other security issuable upon the exercise hereof) as determined in accordance with Section 1.3 above is greater than the Warrant Price in effect on such date, then this Warrant, to the extent unexercised, shall automatically be deemed on and as of such date to be exercised pursuant to Section 1.2 above as to all Shares (or such other securities) for which it shall not previously have been exercised, and the Company shall, within a reasonable time, deliver a certificate representing the Shares (or such other securities) issued upon such exercise to Holder.
- 5.2 <u>Legends</u>. The Shares (and the securities issuable, directly or indirectly, upon conversion of the Shares, if any) shall be imprinted with a legend in substantially the following form:

THE SHARES EVIDENCED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AS SET FORTH IN THAT CERTAIN WARRANT TO PURCHASE STOCK ISSUED BY THE ISSUER TO ______ DATED ______, MAY NOT BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED UNLESS AND UNTIL

REGISTERED UNDER SAID ACT AND LAWS OR IN FORM AND SUBSTANCE SATISFACTORY TO THE ISSUER, SUCH OFFER, SALE, PLEDGE OR OTHER TRANSFER IS EXEMPT FROM SUCH REGISTRATION.

5.3 Compliance with Securities Laws on Transfer. This Warrant and the Shares issuable upon exercise of this Warrant (and the securities issuable, directly or indirectly, upon conversion of the Shares, if any) may not be transferred or assigned in whole or in part except in compliance with applicable federal and state securities laws by the transferor and the transferee (including, without limitation, the delivery of investment representation letters and legal opinions reasonably satisfactory to the Company, as reasonably requested by the Company). The Company shall not require Holder to provide an opinion of counsel if the transfer is to (
5.4 <u>Transfer Procedure.</u> After receipt by of the executed Warrant, will transfer all of this Warrant to its parent company, By its acceptance of this Warrant, hereby makes to the Company each of the representations and warranties set forth in Section 4 hereof and agrees to be bound by all of the terms and conditions of this Warrant as if the original Holder hereof. Subject to the provisions of Section 5.3 and upon providing the Company with written notice, and any subsequent Holder may transfer all or part of this Warrant or the Shares issuable upon exercise of this Warrant (or the securities issuable directly or indirectly, upon conversion of the Shares, if any) to any transferee, provided, however, in connection with any such transfer, or any subsequent Holder will give the Company notice of the portion of the Warrant being transferred with the name, address and taxpayer identification number of the transferee and Holder will surrender this Warrant to the Company for reissuance to the transferee(s) (and Holder if applicable); and provided further, that any subsequent transferee thereof the than shall agree in writing with the Company to be bound by all of the terms and conditions of this Warrant. Notwithstanding any contrary provision herein, at all times prior to the IPO, Holder may not, without the Company's prior written consent, transfer this Warrant or any portion hereof, or any Shares issued upon any exercise hereof, or any shares or other securities issued upon any conversion of any Shares issued upon any exercise hereof, to any person or entity who directly competes with the Company, except in connection with an Acquisition of the Company by such a direct competitor.
5.5 Notices. All notices and other communications hereunder from the Company to the Holder, or vice versa, shall be deemed delivered and effective (i) when given personally, (ii) on the () Business Day after being mailed by first-class registered or certified mail, postage prepaid, (iii) upon actual receipt if given by facsimile or electronic mail and such receipt is confirmed in writing by the recipient, or (iv) on the first Business Day following delivery to a reliable overnight courier service, courier fee prepaid, in any case at such address as may have been furnished to the Company or Holder, as the case may be, in writing by the Company or such Holder from time to time in accordance with the provisions of this Section 5.5. All notices to Holder shall be addressed as follows until the Company receives notice of a change of address in connection with a transfer or otherwise:

Telepho Facsimil	ne:
Notice to the Company	y shall be addressed as follows until Holder receives notice of a change in address:
Attn: Sti 1301 Sa San Frar	follaborative, Inc. uart Landesberg, Chief Executive Officer nsome Street ncisco, California 94111 1@grove.co
	ant and any term hereof may be changed, waived, discharged or terminated (either generally or in a particular instance and ectively) only by an instrument in writing signed by the party against which enforcement of such change, waiver, discharge or
	the event of any dispute between the parties concerning the terms and provisions of this Warrant, the party prevailing in such officer from the other party all costs incurred in such dispute, including reasonable attorneys' fees.
same agreement. Company, the use of electronic signatur or any agreement subject to textent as though it had been	imile/Electronic Signatures. This Warrant may be executed in counterparts, all of which together shall constitute one and the Holder and any other party hereto may execute this Warrant by electronic means and each party hereto recognizes and accepts res and records by any other party hereto in connection with the execution and storage hereof. To the extent that this Warrant the terms hereof or any amendment hereto is executed, recorded or delivered electronically, it shall be binding to the same executed on paper with an original ink signature. The fact that this Warrant is executed, signed, stored or delivered ent the transfer by any Holder of this Warrant pursuant to Section 5.4 or the enforcement of the terms hereof.

- 5.9 Governing Law. This Warrant shall be governed by and construed in accordance with the laws of the State of California, without giving effect to its principles regarding conflicts of law.
- 5.10 <u>Headings</u>. The headings in this Warrant are for purposes of reference only and shall not limit or otherwise affect the meaning of any provision of this Warrant.
 - 5.11 <u>Business Days</u>. "<u>Business Day</u>" is any day that is not a Saturday, Sunday or a day on which ______ is closed.

 [Remainder of page left blank intentionally]

[Remainder of page left blank intentionally] [Signature page follows]

as of the Issue Date written above.	•	•	•
"COMPANY"			
GROVE COLLABORATIVE, INC.			
By: Name: Stuart Landesberg Title: Chief Executive Officer			
"HOLDER"			
By:			
Name:			

[Signature Page to Warrant to Purchase Stock]

IN WITNESS WHEREOF, the parties have caused this Warrant to Purchase Stock to be executed by their duly authorized representatives effective

APPENDIX 1

NOTICE OF EXERCISE

1. The undersigned Holder hereby exercises its right purchase shares of Grove Collaborative, Inc. (the " <u>Company</u> ") in accordance with the attached Warrant To Pu Price for such shares as follows:	the Common/SeriesPreferred [circle one] Stock of rchase Stock, and tenders payment of the aggregate Warrant
[] check in the amount of \$ payable to order of the Company	enclosed herewith
[] Wire transfer of immediately available funds to the Company's accoun	t.
[] Cashless Exercise pursuant to Section 1.2 of the Warrant	
[] Other [Describe]	
2. Please issue a certificate or certificates representing the Shares in the name specific	ed below:
Holder's Name	
(Address)	_
3. By its execution below and for the benefit of the Company, Holder hereby restates Warrant to Purchase Stock as of the date hereof.	each of the representations and warranties in Section 4 of the
	HOLDER:
	Ву:
	Name:
	Title:
	(Date):
Appendix 1	

SCHEDULE 1

Company Capitalization Table

See attached

Schedule 1

THIS WARRANT AND THE SECURITIES ISSUABLE UPON EXERCISE HEREUNDER HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 AS AMENDED (THE "1933 ACT"), OR ANY STATE SECURITIES LAWS. THEY MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED, OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT RELATED THERETO OR AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO YOU THAT SUCH REGISTRATION IS NOT REQUIRED UNDER THE 1933 ACT, OR ANY APPLICABLE STATE SECURITIES LAWS.

PLAIN ENGLISH WARRANT AGREEMEN	ľ	ľ
--------------------------------	---	---

This is a PLAIN ENGLISH WARRANT a corporation, and, a		and between GROVE COLLABORATI	VE, INC., a Delaware
The words "We", "Us", or "Our" refer to the to the issuer, which is GROVE COLLABOR "Warrant Agreement".	RATIVE, INC., and not to any individu	. The al. The words "the Parties" refers to bot INC. This Plain English Warrant Agree	h
The Parties have entered into a	Agreement dated as of	(the "Loan Agreement").	
In consideration of such Loan Agreement th			ow:
	WARRANT INFOR	RMATION	
Effective Date	Warran	t Number	Loan Facility Number
Warrant Coverage Part 1: Tranche A: \$(% of \$) or \$	Number of Shares Part 1: Tranche A:; Tranche B: Part 2: The Number of Shares for each of Part 1 and Part 2 is subject to adjustment as set forth in Sections 1 and 4 of this Warrant Agreement. OUR CONTACT INF	Price Per Share Parts 1 and 2: \$, subject to adjustment as set forth in Sections 1 and 4 of this Warrant Agreement.	Type of Stock Parts 1 and 2: Series C Preferred Stock, subject to adjustment as set forth in Sections 1 and 4 of this Warrant Agreement.
Name	YOUR CONTACT INI	For Notices FORMATION	Contact Person
Customer Name	Address I	For Notices	Contact Person
Grove Collaborative, Inc.		some Street co, CA 94111	Phil Moon Tel: 800-231-8527 email: pmoon@grove.co

1. WHAT YOU AGREE TO GRANT US			
Part 1:			
•	Tranche A. You grant to Us and We	are entitled, upon the ter	rms and subject to the conditions set forth in this Warrant Agreement, to
	purchase from You, at a price per sha	re equal to the Exercise	Price, that number of fully paid and non-assessable shares of Your Warrant
	Stock equal to either (a)	(\$), divided by the Exercise Price or (b) in the event a minimum of

	purchase from You, at a price per share equal to the Exercise Price, that number of fully paid and non-assessable shares of Your Warrant Stock equal to either (a) (\$
•	Tranche B. Upon the availability of the Part 1 Tranche B Commitment Amount under the Loan Agreement, You grant to Us and We are entitled, upon the terms and subject to the conditions set forth in this Warrant Agreement, to purchase from You, at a price per share equal to the Exercise Price, that number of fully paid and non-assessable shares of Your Warrant Stock equal to either (a)
	the Exercise Price, that number of fully paid and non-assessable shares of Your Warrant Stock equal to either (a) Dollars (\$
subject to the of fully paid a Price or (b) in	the availability of the Part 2 Commitment Amount under the Loan Agreement, You grant to Us and We are entitled, upon the terms and conditions set forth in this Warrant Agreement, to purchase from You, at a price per share equal to the Exercise Price, an additional number and non-assessable shares of Your Warrant Stock equal to either (a)
	of this Warrant Agreement, the following capitalized terms have the meanings given below:
	"Exercise Price" means (a) if the Next Round financing closes on or before the lower of (i) \$ and (ii) the lowest per share price for which Your preferred stock is sold in the Next Round or (b) if the Next Round financing does not close on or before, then \$
	"Next Round" means the next bona fide round of equity financing occurring subsequent to the Effective Date in which You issue and sell shares of Your preferred stock for aggregate gross cash proceeds of at least \$ (excluding any amounts received upon conversion or cancellation of indebtedness).
	"Warrant Stock" means (a) the class and series of Your preferred stock issued in the Next Round, if the lowest per share price for which such preferred stock is sold in the Next Round is less than \$ and the Next Round is consummated on or before, or (b) in all other cases, Your Series C Preferred Stock. For avoidance of doubt, if this Warrant Agreement is exercised prior to the Next Round then this Warrant Agreement shall be exercisable for Your Series C Preferred Stock.
of the investr	agree that this Warrant Agreement to purchase the Warrant Stock has a fair market value equal to \$ and that \$ of the issue price ment will be allocable to the Warrant Agreement and the balance shall be allocable to the Loan Agreement for income tax purposes and the e discount on the Loan Agreement shall be considered to be zero.

The term of this Warrant Agreement and Our right to purchase Warrant Stock will begin on the Effective Date and shall be available for () year through and including
Notwithstanding the foregoing, Our right to purchase the Warrant Stock shall be automatically and fully exercised via the net issuance method described below (without surrender of the Warrant Agreement) upon the occurrence of a Merger Event, as defined below, with a Person that is not one of Your affiliates, in which Your common stock is exchanged for cash and/or stock of a class that is listed on a recognized national exchange, provided that, upor consummation of the Merger Event, the consideration payable to Us pursuant to such exercise and on account of the Warrant Stock consists of (i) cash of a class that is listed on a recognized national exchange and the total per share value of such stock consideration is equal to or greater than

3. HOW WE MAY PURCHASE YOUR WARRANT STOCK.

WHEN ARE WE ENTITLED TO PURCHASE YOUR WARRANT STOCK.

We may exercise Our purchase rights, in whole or in part, at any time, or from time to time, prior to the expiration of the term of this Warrant Agreement, by giving You, except as provided in the last paragraph of this Section 3, a completed and executed **Notice of Exercise** in the form attached as **Exhibit I**. Promptly upon receipt of the Notice of Exercise and in any event no later than _____ (___) days after you have received Our Notice of Exercise and payment of the aggregate Exercise Price for the shares purchased, You will issue to Us a certificate (in electronic format, if You are issuing certificates electronically) for the number of shares of Warrant Stock that We have purchased and You will execute the **Acknowledgment of Exercise** in the form attached hereto as **Exhibit II** indicating the number of shares which will be available to Us for future purchases, if any.

We may pay for the Warrant Stock by either (i) cash or check, or (ii) by the **net issuance method** as determined below. If We elect the net issuance method, You will issue Warrant Stock using the following formula:

$$X = \underline{Y(A-B)}$$

2.

Where: X = the number of shares of Warrant Stock to be issued to Us.

Y = the number of shares of Warrant Stock We request to be exercised under this Warrant Agreement.

A = the fair market value of one share of Warrant Stock (as of the date of such calculation).

B = the Exercise Price (as of the date of such calculation).

For purposes of the above calculation and the below paragraphs, current fair market value of Warrant Stock shall mean with respect to each share of Warrant Stock:

If the exercise is in connection with the initial public offering of Your Common Stock and if Your registration statement relating to such public offering has been declared effective by the Securities and Exchange Commission, then the fair market value per share shall be the product of (x) the initial "Price to Public" specified in the final prospectus of the offering and (y) the number of shares of common stock into which each share of Warrant Stock is convertible at the time of such exercise.

If this Warrant Agreement is exercised after, and not in connection with Your initial public offering, and:

- if Your common stock is traded on a securities exchange, the fair market value of the Warrant Stock shall be the product of (x) the average of the closing prices of Your common stock over a ____ (__) day period ending ____ (__) days before the day the current fair market value of the Warrant Stock is being determined and (y) the number of shares of Your common stock into which each share of Warrant Stock is convertible at the time of such exercise; or
- if Your common stock is actively tradedover-the-counter, the fair market value of the Warrant Stock shall be the product of (x) the average of the closing bid and asked prices of Your common stock over the ____ (__) day period ending ____ (__) days before the day the current fair market value of the Warrant Stock is being determined and (y) the number of shares of Your common stock into which each share of Warrant Stock is convertible at the time of such exercise.

If this Warrant Agreement is exercised prior to or after Your initial public offering, and:

• Your common stock is not listed on any securities exchange or theover-the-counter market, the current fair market value of one share of Warrant Stock shall be the product of (x) the fair market value of a share of Your common stock (the highest price per share which You could obtain from a willing buyer (not a current employee or director) for shares of common stock sold, from authorized but unissued shares), as determined in good faith by Your Board of Directors, and (y) the number of shares of common stock into which each share of Warrant Stock is convertible at the time of such exercise. Notwithstanding the foregoing, however, if You shall become subject to a merger, acquisition or other consolidation pursuant to which holders of Warrant Stock shall be entitled to receive cash, securities or other property, then the fair market value of the Warrant Stock shall be deemed to be the value received by the holders of the Warrant Stock (on a common equivalent basis) pursuant to such merger or acquisition or other consolidation.

During the term of this Warrant Agreement, except as provided in the next sentence, You will at all times from and after the Effective Date have authorized and reserved a sufficient number of shares of (a) Warrant Stock to provide for the exercise of our rights to purchase Warrant Stock, and (b) common stock to provide for the conversion of the Warrant Stock. Notwithstanding the foregoing, in no event shall You be required to authorize or reserve any of Your preferred stock issued in the Next Round unless and until both (x) the Next Round is consummated and (y) the Warrant Stock is Your preferred stock issued in the Next Round.

If We elect to exercise part of the Warrant Agreement, You will promptly issue to Us an amended Warrant Agreement stating the remaining number of shares that are available. All other terms and conditions of that amended Warrant Agreement shall be identical to those contained in this Warrant Agreement.

If at the end of the term of this Warrant Agreement, the fair market value of one share of Warrant Stock (or other security issuable upon the exercise hereof) as determined in accordance herewith is greater than the Exercise Price in effect on such date, then this Warrant Agreement shall automatically be deemed on and as of such date to be converted pursuant hereto as to all shares of Warrant Stock (or such other securities) for which it shall not previously have been exercised or converted, and You shall promptly deliver a certificate representing the shares of Warrant Stock (or such other securities) issued upon such conversion to Us.

4. WHEN WILL THE NUMBER OF SHARES AND EXERCISE PRICE CHANGE.

• If You are Acquired. Subject to Section 2, if at any time: (i) there is a reorganization of Your stock (other than a reclassification, exchange or subdivision of Your stock otherwise provided for in this Warrant Agreement); or (ii) You consummate a Deemed Liquidation Event (as defined in Your Certificate of Incorporation as amended through the Effective Date (Your "Certificate of Incorporation")) and a waiver of such Deemed Liquidation Event by Your shareholders shall not apply to Us (each such event referred to as a

"Merger Event"), then, as a part of such Merger Event, lawful provision shall be made so that We shall thereafter be entitled to receive, upon exercise of Our rights under this Warrant Agreement, the number of shares of preferred stock or other securities of the successor or surviving person resulting from such Merger Event, equal in value to that which would have been issuable if We had exercised Our rights under this Warrant Agreement immediately prior to the Merger Event. In any such case, appropriate adjustment (as determined in good faith by Your Board of Directors) shall be made in the application of the provisions of this Warrant Agreement with respect to Our rights and interest after the Merger Event so that the provisions of this Warrant Agreement (including adjustments of the Exercise Price and number of shares of Warrant Stock purchasable) shall be applicable to the greatest extent possible.

- If You Reclassify Your Stock. If at any time, other than in connection with a Merger Event covered above, You combine, reclassify, exchange or subdivide Your securities or otherwise change any of the securities as to which purchase rights under this Warrant Agreement exist into the same or a different number of securities of any other class or classes, this Warrant Agreement will thereafter represent the right to acquire such number and kind of securities as would have been issuable as the result of such change with respect to the securities which were subject to the purchase rights under this Warrant Agreement immediately prior to such combination, reclassification, exchange, subdivision or other change.
- If You Subdivide or Combine Your Shares. If at any time, other than in connection with a Merger Event covered above, You combine or subdivide the Warrant Stock, the Exercise Price will be proportionately decreased in the case of a subdivision, or proportionately increased in the case of a combination.
- If You Pay Stock Dividends. If at any time, other than in connection with a Merger Event covered above, You pay a dividend payable in, or make any other distribution (except any distribution specifically provided for in the above paragraphs) of the Warrant Stock, then the Exercise Price shall be adjusted, from and after the record date of such dividend or distribution, to that price determined by multiplying the Exercise Price in effect immediately prior to such record date by a fraction (i) the numerator of which shall be the total number of all shares of the Warrant Stock outstanding immediately prior to such dividend or distribution, and (ii) the denominator of which shall be the total number of all shares of the Warrant Stock outstanding immediately after such dividend or distribution. We will thereafter be entitled to purchase, at the Exercise Price resulting from such adjustment, the number of shares of Warrant Stock (calculated to the nearest whole share) obtained by multiplying the Exercise Price in effect immediately prior to such adjustment by the number of shares of Warrant Stock issuable upon the exercise hereof immediately prior to such adjustment and dividing the product thereof by the Exercise Price resulting from such adjustment.
- "Pay to Play" Rights. In the event that We have not exercised this Warrant as to all the Warrant Stock and any "pay to play" terms or conditions (i.e. terms or conditions that require a holder of shares of Your preferred stock (the "Preferred Stock") to purchase securities in a future round of equity financing or else lose the benefit of anti-dilution protections applicable to shares of Preferred Stock or have such shares of Preferred Stock automatically convert into common stock or another class or series of capital stock) in Your Certificate of Incorporation are triggered in connection with any sale or issuance of securities (a "Trigger Event"), then, in each such event the purchase rights remaining under this Warrant Agreement shall automatically adjust to provide Us, upon the later exercise hereof, with the same securities and/or rights that We would have received had We (x) exercised such portion of this Warrant Agreement prior to such Trigger Event, and (y) participated in the applicable equity financing in an amount sufficient to be deemed to have fully participated for purposes of such "pay to play" provision.
- If You Change the Antidilution Rights of the Warrant Stock or Issue New Preferred or Convertible Stock. All antidilution rights applicable to the Warrant Stock purchasable under this Warrant Agreement are as set forth in Your Certificate of Incorporation. You will promptly provide Us with any restatement, amendment; modification of or waiver of any right, in each case that relates to the Warrant Stock, under Your Certificate of Incorporation.

5. WE CAN TRANSFER THIS WARRANT AGREEMENT.

Subject to the terms and conditions contained in Section 7, We (or any successor transferee) may transfer in whole or in part this Warrant Agreement and all its rights. You will record the transfer on Your books when You receive Our Notice of Transfer in the form attached hereto as Exhibit III, and Our payment of all transfer taxes and other governmental charges involved in such transfer. Such transferee shall agree to be bound by all terms and conditions of this Warrant Agreement and make the representations and covenants (including with respect to the "Market Stand-Off") in Section 7 hereof to You on the effective date of such transfer.

6. REPRESENTATIONS, WARRANTIES, AND COVENANTS FROM YOU.

- Reservation of Warrant Stock. The Warrant Stock issuable upon exercise of Our rights under this Warrant Agreement will be duly and validly reserved (with it being acknowledged that, prior to the Next Round, You have not reserved any shares of the Next Round that this Warrant Agreement may be exercisable for) and when issued in accordance with the provisions of this Warrant Agreement will be validly issued, fully paid and non-assessable, and will be free of any taxes, liens, charges or encumbrances of any nature whatsoever; provided, however, that the Warrant Stock issuable pursuant to this Warrant Agreement may be subject to restrictions on transfer under state and/or Federal securities laws. Upon Our exercise, You will issue to Us certificates for shares of Warrant Stock without charging Us any tax, or other cost incurred by You in connection with such exercise and the related issuance of shares of Warrant Stock. You will not be required to pay any tax, which may be payable in respect of any transfer involved and the issuance and delivery of any certificate in a name other than
- Due Authority. Your execution and delivery of this Warrant Agreement and the performance of Your obligations hereunder, including the issuance to Us of the right to acquire the shares of Warrant Stock, have been duly authorized by all necessary corporate action on Your part and this Warrant Agreement is not inconsistent with Your Certificate of Incorporation or Bylaws, do not contravene any law or governmental rule, regulation or order applicable to it, do not and will not contravene any provision of, or constitute a default under, any indenture, mortgage, material contract or other instrument to which You are a party or by which You are bound, and this Warrant Agreement constitutes a legal, valid and binding agreement, enforceable in accordance with its respective terms, subject to laws of general application relating to bankruptcy, insolvency and the relief of debtors and the rules of law or principles at equity governing specific performance, injunctive relief and other equitable remedies.
- Consents and Approvals. No consent or approval of, giving of notice to, registration with, or taking of any other action in respect of any
 state, Federal or other governmental authority or agency is required with respect to execution, delivery and Your performance of Your
 obligations under this Warrant Agreement, except for the corporate approvals in connection with the Next Round and the filing of any
 required notices pursuant to Federal and state securities laws, which filings will be effective by the times required thereby.
- Issued Securities. All of Your issued and outstanding shares of common stock, Warrant Stock or any other securities have been duly authorized and validly issued and are fully paid and nonassessable. All outstanding shares of common stock and Warrant Stock were issued in full compliance with all Federal and state securities laws. In addition, as of the Effective Date, and immediately prior to the issuance of this Warrant, Our capitalization is as set forth in the attached Schedule 1.
- Other Commitments to Register Securities. Except as set forth in this Warrant Agreement and the Investors' Rights Agreement, You are
 not, pursuant to the terms of any other agreement currently in existence, under any obligation to register under the 1933 Act any of Your
 presently outstanding securities or any of Your securities which may hereafter be issued.
- Exempt Transaction. Subject to the accuracy of Our representations in Section 7 hereof, the issuance of the Warrant Stock upon exercise of this Warrant Agreement will constitute a transaction exempt from (i) the registration requirements of Section 5 of the 1933 Act, in reliance upon Section 4(2) thereof, and (ii) the qualification requirements of the applicable state securities laws.

- Compliance with Rule 144. We may sell the Warrant Stock issuable hereunder in compliance with Rule 144 promulgated by the Securities and Exchange Commission. Within ___ (__) days of Our request, You agree to furnish Us, a written statement confirming Your compliance with the filing requirements of the Securities and Exchange Commission as set forth in such Rule 144, as may be amended.
- No Impairment. Except as set forth herein, You agree not to, by amendment of Your Certificate of Incorporation, by-laws or other organizational or charter documents or through a reorganization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms to be observed or performed under this Warrant Agreement by You, but shall at all times in good faith assist in carrying out of all the provisions of this Warrant Agreement and in taking all such action as may be necessary or appropriate to protect Our rights under this Warrant Agreement against impairment. However, notwithstanding the foregoing, You shall not be deemed to have impaired Our rights if You amend, restate, modify or waive any provisions of Your Certificate of Incorporation (by amendment, merger, reclassification, recapitalization or otherwise), or the holders of Your preferred stock waive their rights thereunder, in a manner that does not (individually or when considered in the context of any other actions being taken in connection with such amendments or waivers) the rights, privileges, preferences, restrictions and limitations of the Warrant Stock in a manner different from the effect that such amendments, restatements, modifications or waivers have generally on the rights, privileges, preferences, restrictions and limitations of the then outstanding securities of You that are of the same series and class as the Warrant Stock, it being expressly understood that an amendment, restatement, modification or waiver that applies equally to all holders of the class and/or series of securities then issuable upon exercise of this Warrant that has a different economic effect on You shall not be deemed to affect the rights, privileges, preferences, restrictions and limitations of the Warrant Stock in a manner different from the effect that such amendments, restatements, modification or waivers have generally on the rights, privileges, preferences, restrictions and limitations of the then outstanding securities of You that are of the same series and class as the Warrant Stock.

7. OUR REPRESENTATIONS AND COVENANTS TO YOU.

- Investment Purpose. The right to acquire Warrant Stock or the Warrant Stock issuable upon exercise of Our rights contained herein and the common stock issuable upon conversion will be acquired for investment purposes and not with a view to the sale or distribution of any part thereof, and We have no present intention of selling or engaging in any public distribution of the same in violation of the 1933 Act.
- Private Issue. We understand (i) that this Warrant Agreement, the Warrant Stock issuable upon exercise of this Warrant Agreement and the
 common stock issuable upon conversion of the Warrant Stock are not registered under the 1933 Act or qualified under applicable state
 securities laws on the ground that the issuance contemplated by this Warrant Agreement will be exempt from the registration and
 qualifications requirements thereof, and (ii) that Your reliance on such exemption is predicated on the representations set forth in this
 Section 7
- Disposition of Our Rights. In no event will We make a disposition of any of Our rights to acquire Warrant Stock or Warrant Stock issuable upon exercise of such rights or the common stock issuable upon conversion of the Warrant Stock unless and until (i) We shall have notified You in writing of the proposed disposition, and (ii) the transferee agrees to be bound in writing to the applicable terms and conditions of this Warrant Agreement, and (iii) if You request, We shall have furnished You with an opinion of counsel satisfactory to You and Your counsel to the effect that (A) appropriate action necessary for compliance with the 1933 Act has been taken, or (B) an exemption from the registration requirements of the 1933 Act is available. Notwithstanding the foregoing, the restrictions imposed upon the transferability of any of Our rights to acquire Warrant Stock or Warrant Stock issuable on the exercise of such rights or the common stock issuable upon conversion of the Warrant Stock do not apply to transfers from the beneficial owner of any of the aforementioned securities to its nominee or from such nominee to its beneficial owner, and shall terminate as to any particular share of Warrant Stock when (1) such security shall have been effectively registered under the 1933 Act and sold by the holder thereof in accordance with such registration or (2) such security shall have been sold without registration in compliance with Rule 144 under the 1933 Act, or (3) a letter shall

have been issued to You at Our request by the staff of the Securities and Exchange Commission or a ruling shall have been issued to You at Our request by such Commission stating that no action shall be recommended by such staff or taken by such Commission, as the case may be, if such security is transferred without registration under the 1933 Act in accordance with the conditions set forth in such letter or ruling and such letter or ruling specifies that no subsequent restrictions on transfer are required. Whenever the restrictions imposed hereunder shall terminate, as hereinabove provided, the holder of a share of Warrant Stock then outstanding as to which such restrictions have terminated shall be entitled to receive from You, without expense to such holder, one or more new certificates for the Warrant or for such shares of Warrant Stock not bearing any restrictive legend referring to 1933 Act registration or exemption.

- Financial Risk. We have such knowledge and experience in financial and business matters and knowledge of Your business affairs and
 financial condition as to be capable of evaluating the merits and risks of Our investment, and have the ability to bear the economic risks of
 Our investment.
- Risk of No Registration. We understand that if You do not register with the Securities and Exchange Commission pursuant to Section 12 of the Securities Exchange Act of 1934 (the "1934 Act"), or file reports pursuant to Section 15(d), of the 1934 Act, or if a registration statement covering the securities under the 1933 Act is not in effect when We desire to sell (i) the rights to purchase Warrant Stock pursuant to this Warrant Agreement, or (ii) the Warrant Stock issuable upon exercise of the right to purchase, or (iii) the common stock issuable upon conversion of the Warrant Stock, We may be required to hold such securities for an indefinite period. We also understand that any sale of Our right to purchase Warrant Stock or Warrant Stock or common stock issuable upon conversion of the Warrant Stock, which might be made by it in reliance upon Rule 144 under the 1933 Act may be made only in accordance with the terms and conditions of that Rule.
- Accredited Investor. We are an "accredited investor" within the meaning of the Securities and Exchange Rule 501 of Regulation D of the 1933 Act, as presently in effect.
- "Market Stand-Off" Agreement. With respect to this Warrant, the Warrant Stock and any shares of common stock issuable upon conversion of the Warrant Stock, We hereby agree to be bound by the "Market Stand-off" provisions in Section ______ of Your Investors' Rights Agreement (as defined below). Notwithstanding the foregoing, in no event shall such market stand-off agreement restrict Our ability to exercise Our purchase rights under this Warrant Agreement, including the transfer of common stock to You solely to satisfy the exercise price pursuant to the net issuance method and if there is any conflict between the Investors' Rights Agreement and this subsection, this subsection shall control.

0	NOTICES	VOLLA	CDEE	TO DDO	TIDE HE
ð.	NOTICES	YUU A	GKEE	ιυ γκυν	IDE US.

You agree to give Us at least () days prior written notice of the following events:

- If You pay a dividend or distribution declaration upon Your stock.
- If You offer for subscription pro-rata to the existing shareholders additional stock or other rights.
- If You consummate or sign definitive documents providing for a Merger Event.
- · If You have an initial public offering.
- If You dissolve or liquidate.

All notices in this Section must set forth details of the event, how the event adjusts either Our number of shares or Our Exercise Price and the method used for such adjustment.

Timely Notice. Your failure to timely provide such notice required above shall entitle Us to retain the benefit of the applicable notice period notwithstanding anything to the contrary contained in any insufficient notice received by Us; provided, however, that this will not prevent or delay You from consummating any of the transactions listed above.

. DOCUMENTS YOU WILL PROVIDE US.

Upon signing this Warrant Agreement You will provide Us with:

- · Executed originals of this Warrant Agreement, and all other documents and instruments that We may reasonably require
- Secretary's certificate of incumbency and authority
- Certified copy of resolutions of Your board of directors approving this Warrant Agreement
- Certified copy of Certificate of Incorporation and by-laws as amended through the Effective Date
- The Investors' Rights Agreement

So long as this Warrant Agreement is in effect, You shall provide Us with the following:

•	Within () business days after the closing of any equity financing, or extension of an existing round of equity financing, occurring after the Effective Date, in which You issue preferred stock or other securities You will provide Us with copies of the fully executed equity financing documents, including without limitation the related stock purchase agreement, investors rights agreement, voting agreement, amended or restated Certificates of Incorporation, current capitalization table and other related documents. Notwithstanding any term or condition contained in this Warrant Agreement to the contrary, Your failure to comply with this paragraph shall not constitute a default unless You have not provided the information requested within () business days after Our request.
•	Within () days after completion You shall provide Us with any 409A Valuation Reports or other similar reports prepared for You. Notwithstanding any term or condition contained in this Warrant Agreement to the contrary, Your failure to comply with this paragraph shall not constitute a default unless You have not provided the information requested within () business days after Our request.
•	After all obligations under the Loan Agreement have been finally paid in full, within () days after the end of each quarter, You will provide Us with (1) an unaudited income statement, statement of cash flows, and an unaudited balance sheet prepared in accordance with GAAP accompanied by a report detailing any material contingencies, (2) (i) at all times that Your Board of Directors requires You to prepare audited financial statements, as soon as available, but no later than () days after the last day of Your fiscal year, audited consolidated financial statements prepared under GAAP, consistently applied, together with an unqualified opinion on the financial statements from an independent certified public accounting firm reasonably acceptable to Us; and (ii) at all other times, as soon as available, but no later than () days after the last day of Your fiscal year, a company-prepared consolidated balance sheet and income statement covering Your consolidated operations during such fiscal year.
•	You shall submit to Us any other documents and other information that We may reasonably request from time to time and are necessary to implement the provisions and purposes of this Warrant Agreement. Notwithstanding any term or condition contained in this Warrant Agreement to the contrary, Your failure to comply with this paragraph shall not constitute a default unless You have not provided the information requested within () days of Our request.

10. REGISTRATION RIGHTS UNDER THE 1933 ACT.

Concurrently with the issuance of this Warrant Agreement, the Amended and Restated Investors' Rights Agreement attached hereto as Exhibit IV shall be executed and delivered by You, Us, and by such other parties as are necessary to amend and restate the Prior Agreement (as defined in therein) (the "Investors' Rights Agreement").

11. OTHER LEGAL PROVISIONS THE PARTIES WILL ABIDE BY.

Effective Date. This Warrant Agreement shall be construed and shall be given effect in all respects as if it had been executed and delivered by the Parties on the date hereof. This Warrant Agreement shall be binding upon any of the successors or assigns of the Parties.

Attorney's Fees. In any litigation, arbitration or court proceeding between the Parties relating to this Warrant Agreement, the prevailing party shall be entitled to attorneys' fees and expenses and all costs of proceedings incurred in enforcing this Warrant Agreement.

Fractional Shares. No fractional shares shall be issued upon the exercise of this Warrant Agreement as a consequence of any adjustment pursuant hereto. All Warrant Stock (including fractions) issuable upon exercise of this Warrant Agreement may be aggregated for purposes of determining whether the exercise would result in the issuance of any fractional share. If, after aggregation, the exercise would result in the issuance of a fractional share, You shall, in lieu of issuance of any fractional share, pay Us, if We are otherwise entitled to such fraction, a sum in cash equal to the product resulting from multiplying the then current fair market value of the Warrant Stock by such fraction.

Governing Law. This Warrant Agreement shall be governed by and construed for all purposes under and in accordance with the laws of the State of California without giving effect to that body of law pertaining to conflicts of laws.

Consent to Jurisdiction and Venue. All judicial proceedings arising in or under or related to this Warrant Agreement may be brought in any state or federal court of competent jurisdiction located in the State of California. By execution and delivery of this Warrant Agreement, each party hereto generally and unconditionally: (a) consents to personal jurisdiction in San Mateo County, State of California; (b) waives any objection as to jurisdiction or venue in San Mateo County, State of California; (c) agrees not to assert any defense based on lack of jurisdiction or venue in the aforesaid courts; and (d) irrevocably agrees to be bound by any judgment rendered thereby in connection with this Warrant Agreement. Service of process on any party hereto in any action arising out of or relating to this Warrant Agreement shall be effective if given in accordance with the requirements for notice set forth in this Section, and shall be deemed effective and received as set forth therein. Nothing herein shall affect the right to serve process in any other manner permitted by law or shall limit the right of either party to bring proceedings in the courts of any other jurisdiction.

Mutual Waiver of Jury Trial; Judicial Reference. Because disputes arising in connection with complex financial transactions are most quickly and economically resolved by an experienced and expert person and the Parties wish applicable state and federal laws to apply (rather than arbitration rules), the Parties desire that their disputes be resolved by a judge applying such applicable laws. EACH OF THE PARTIES SPECIFICALLY WAIVES ANY RIGHT THEY MAY HAVE TO TRIAL BY JURY OF ANY CAUSE OF ACTION, CLAIM, CROSS-CLAIM, COUNTERCLAIM, THIRD PARTY CLAIM OR ANY OTHER CLAIM (COLLECTIVELY, "CLAIMS") ASSERTED BY YOU AGAINST US OR OUR ASSIGNEE OR BY US OR OUR ASSIGNEE AGAINST YOU. IN THE EVENT THAT THE FOREGOING JURY TRIAL WAIVER IS NOT ENFORCEABLE, ALL CLAIMS, INCLUDING ANY AND ALL QUESTIONS OF LAW OR FACT RELATING THERETO, SHALL, AT THE WRITTEN REQUEST OF ANY PARTY, BE DETERMINED BY JUDICIAL REFERENCE PURSUANT TO THE CALIFORNIA CODE OF CIVIL PROCEDURE ("REFERENCE"). THE PARTIES SHALL SELECT A SINGLE NEUTRAL REFEREE, WHO SHALL BE A RETIRED STATE OR FEDERAL JUDGE. IN THE EVENT THAT THE PARTIES CANNOT AGREE UPON A REFEREE, THE REFEREE SHALL BE APPOINTED BY THE COURT. THE REFEREE SHALL REPORT A STATEMENT OF DECISION TO THE COURT. NOTHING IN THIS SECTION SHALL LIMIT THE RIGHT OF ANY PARTY AT ANY TIME TO EXERCISE LAWFUL SELF-HELP REMEDIES, FORECLOSE AGAINST COLLATERAL OR OBTAIN PROVISIONAL REMEDIES. THE PARTIES SHALL BEAR THE FEES

AND EXPENSES OF THE REFEREE EQUALLY UNLESS THE REFEREE ORDERS OTHERWISE. THE REFEREE SHALL ALSO DETERMINE ALL ISSUES RELATING TO THE APPLICABILITY, INTERPRETATION, AND ENFORCEABILITY OF THIS SECTION. THE PARTIES ACKNOWLEDGE THAT THE CLAIMS WILL NOT BE ADJUDICATED BY A JURY. This waiver extends to all such Claims, including Claims that involve persons other than You and Us; Claims that arise out of or are in any way connected to the relationship between You and Us; and any Claims for damages, breach of contract, specific performance, or any equitable or legal relief of any kind, arising out of this Warrant Agreement.

Counterparts. This Warrant Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

Notices. Any notice required or permitted under this Warrant Agreement shall be given in writing and shall be deemed effectively given upon the earlier of (1) actual receipt or ____ days after mailing if mailed postage prepaid by regular or airmail to Us or You or (2) one day after it is sent by overnight mail via nationally recognized courier or (3) on the same day as sent via confirmed facsimile transmission, provided that the original is sent by personal delivery or mail by the sending party.

Remedies. In the event of any default hereunder, the non-defaulting party may proceed to protect and enforce its rights either by suit in equity and/or by action at law, including but not limited to an action for damages as a result of any such default, and/or an action for specific performance for any default where such party will not have an adequate remedy at law and where damages will not be readily ascertainable. Each party expressly acknowledges and agrees that there is no adequate remedy at law for any breach of this Warrant Agreement and that in the event of any breach of this Warrant Agreement, the injured party shall be entitled to specific performance of any or all provisions hereof or an injunction prohibiting the other party from continuing to commit any such breach of this Warrant Agreement.

Survival. The representations, warranties, covenants, and conditions of the Parties contained herein or made pursuant to this Warrant Agreement shall survive the execution and delivery of this Warrant Agreement.

Severability. In the event any one or more of the provisions of this Warrant Agreement shall for any reason be held invalid, illegal or unenforceable, the remaining provisions of this Warrant Agreement shall be unimpaired, and the invalid, illegal or unenforceable provision shall be replaced by a mutually acceptable valid, legal and enforceable provision, which comes closest to the intention of the Parties underlying the invalid, illegal or unenforceable provision.

Entire Agreement. This Warrant Agreement constitutes the entire agreement between the Parties pertaining to the subject matter contained in it and supersedes all prior and contemporaneous agreements, representations and undertakings of the Parties, whether oral or written, with respect to such subject matter

Amendments. Any provision of this Warrant Agreement may only be amended by a written instrument signed by the Parties.

Lost Warrants or Stock Certificates. You covenant to Us that, upon receipt of evidence reasonably satisfactory to Us of the loss, theft, destruction or mutilation of this Warrant Agreement or any stock certificate and, in the case of any such loss, theft or destruction, upon receipt of an indemnity reasonably satisfactory to You, or in the case of any such mutilation upon surrender and cancellation of such Warrant Agreement or stock certificate, You will make and deliver a new Warrant Agreement or stock certificate, of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant Agreement or stock certificate.

Rights as Stockholders. We shall not, as a party to this Warrant Agreement, be entitled to vote or receive dividends or be deemed the holder of Warrant Stock or any of Your other securities which may at any time be issuable upon the exercise hereof for any purpose, nor shall anything contained herein be construed to confer upon Us any of the rights of one of Your stockholders or any right to vote for the election of directors or upon any matter submitted to stockholders at any meeting thereof, or to receive dividends or subscription rights or otherwise until this Warrant Agreement is exercised and the shares purchasable upon the exercise hereof shall have become deliverable, as provided herein.

Signatures. This Warrant Agreement may be executed and delivered by facsimile or transmitted electronically in either Tagged Image Format Files ("TIFF") or Portable Document Format ("PDF") and, upon such delivery, the facsimile, TIFF or PDF signature, as applicable, will be deemed to have the same effect as if the original signature had been delivered to the other party.

(Signature Page to Follow)

EXHIBIT I

NOTICE OF EXERCISE

To:	
1.	We hereby elect to purchase [] shares of the Series [] Preferred Stock of [], pursuant to the terms of the Plain English Warrant Agreement dated the [] day of [], [20] (the "Warrant Agreement") between You and Us, We hereby tender here payment of the purchase price for such shares in full, together with all applicable transfer taxes, if any.
2.	Method of Exercise (Please initial the applicable blank)
	a The undersigned elects to exercise the Warrant Agreement by means of a cash payment, and gives You full payment for the purchase price of the shares being purchased, together with all applicable transfer taxes, if any.
	b. The undersigned elects to exercise the Warrant Agreement by means of the net issuance method of Section 3 of the Warrant Agreement.
3.	In exercising Our rights to purchase the Series [] Preferred Stock of [], We hereby confirm and acknowledge the investment representations, warranties and covenants made in Section 7 of the Warrant Agreement.
	se issue a certificate or certificates representing these purchased shares of Series [] Preferred Stock in Our name or in such other name as is ified below.
	(Name)
	(Address)
	US:
	Ву:
	Title:
	Date:

EXHIBIT II

ACKNOWLEDGMENT OF EXERCISE

[], hereby acknowledges receipt of the "Notice shares of the Series [] Preferred Stock of [], purs	e of Exercise" from [], to purchase [] uant to the terms of the Plain English Warrant Agreement dated the			
day of [20_] (the "Warrant Agreement") and further acknowledges that [] shares remain subject to purchase under the terms of the				
wartant Agreement.	Warrant Agreement.			
YOU:	GROVE COLLABORATIVE, INC.			
	Ву:			
	Title:			
	Date:			

EXHIBIT III

TRANSFER NOTICE

FOR VALUE RECEIVED, the foregoing Warrant Agreement and all rights evidenced thereby are hereby transferred and assigned to

(Please Print)		
Whose address is		
Dated:		
Holder's Signature:		
Holder's Address:		
Transferee's Signature:	-	
Transferee's Address:		
Signature Guaranteed:		

NOTE: The signature to this Transfer Notice must correspond with the name as it appears on the face of the Warrant Agreement, without alteration or enlargement or any change whatever. Officers of corporations and those acting in a fiduciary or other representative capacity should file proper evidence of authority to assign the foregoing Warrant Agreement.

EXHIBIT IV

A&R INVESTORS' RIGHTS AGREEMENT

SCHEDULE 1

CAPITALIZATION TABLE

Outstanding Fully diluted

Common Series Seed Preferred Series A Preferred Series B Preferred Series C Preferred Common Warrants Series A Warrants Series B Warrants Options Issued Options Available Total THIS WARRANT AND THE SECURITIES ISSUABLE UPON EXERCISE HEREUNDER HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 AS AMENDED (the "1933 ACT"), OR ANY STATE SECURITIES LAWS. THEY MAY NOT BE SOLD, OFFERED FOR SALE, PLEDGED, OR HYPOTHECATED IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT RELATED THERETO OR AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO YOU THAT SUCH REGISTRATION IS NOT REQUIRED UNDER THE 1933 ACT, OR ANY APPLICABLE STATE SECURITIES LAWS.

PLAIN ENGLISH WARRANT AGREEMENT

This is a PLAIN ENGLISH WARRANT A and, a	GREEMENT dated by and be	etween GROVE COLLABORATIVE, I	NC., a Delaware corporation,		
The words "We", "Us", or "Our" refer to the warrant holder, which is The words "You" or "Your" refers to the issuer, which is GROVE COLLABORATIVE, INC., and not to any individual. The words "the Parties" refers to both and GROVE COLLABORATIVE, INC. This Plain English Warrant Agreement may be referred to as the "Warrant Agreement".					
The Parties have entered into an Amended at Agreement").	nd Restated Plain English Growth Capi	ital Loan and Security Agreement dated	as of(the "Loan		
In consideration of such Loan Agreement the Parties agree to the following mutual agreements and conditions set forth below:					
	WARRANT INFORMATION				
Effective Date	Warrant Number		Loan Facility Number		
Warrant Coverage Part 1: Tranche A: \$ (% of \$); Tranche B: \$ (% of \$), each as provided in Section 1 below. Part 2: \$ (% of \$), upon the availability of the Part 4 Commitment Amount and as provided in Section 1 below.	Number of Shares Part 1: Tranche A: ; Tranche B: Part 2: The Number of Shares for each of Part 1 and Part 2 is subject to adjustment as set forth in Sections 1 and 4 of this Warrant Agreement.	Price Per Share Parts 1 and 2: \$, subject to adjustment as set forth in Sections 1 and 4 of this Warrant Agreement.	Type of Stock Parts 1 and 2: Series D Preferred Stock, subject to adjustment as set forth in Sections 1 and 4 of this Warran Agreement.		

OUR CONTACT INFORMATION

Name	Address For Notices	Contact Person
	YOUR CONTACT INFORMATION	
Customer Name Grove Collaborative, Inc.	Address For Notices 1301 Sansome Street San Francisco, CA 94111	Contact Person Phil Moon, VP Business Operations Tel: 800-231-8527 Fax: n/a email: pmoon@grove.co
1. WHAT YOU AGREE TO GRANT US		
Part 1:		
	ntitled, upon the terms and subject to the conditions set forth in ise Price, that number of fully paid and non-assessable shares one.	
upon the terms and subject to the conditions	rt 3 Tranche B Commitment Amount under the Loan Agreeme s set forth in this Warrant Agreement, to purchase from You, a sessable shares of Your Warrant Stock equal toDollars	at a price per share equal to the Exercise
subject to the conditions set forth in this Warrant	itment Amount under the Loan Agreement, You grant to Us an Agreement, to purchase from You, at a price per share equal to Varrant Stock equal to Dollars (\$), divided by the Ex	the Exercise Price, an additional number
The number of shares of Warrant Stock and the E	xercise Price of such Warrant Stock are subject to adjustment a	as provided in Section 4 hereof.
For purposes of this Warrant Agreement, the follow	owing capitalized terms have the meanings given below:	
	nd financing closes on or before the lower of (i) \$ ext Round or (b) if the Next Round financing does not close or	
	2	

"Next Round" means the next bona fide round of equity financing occurring subsequent to the Effective Date in which You issue and sell shares of Your preferred stock for aggregate gross cash proceeds of at least \$5,000,000 (excluding any amounts received upon conversion or cancellation of indebtedness).

"Warrant Stock" means (a) the class and series of Your preferred stock issued in the Next Round, if the lowest per share price for which such preferred stock is sold in the Next Round is less than \$_____ and the Next Round is consummated on or before_____, or (b) in all other cases, Your Series D Preferred Stock. For avoidance of doubt, if this Warrant Agreement is exercised prior to the Next Round then this Warrant Agreement shall be exercisable for Your Series D Preferred Stock.

The Parties agree that this Warrant Agreement to purchase the Warrant Stock has a fair market value equal to \$100 and that \$100 of the issue price of the investment will be allocable to the Warrant Agreement and the balance shall be allocable to the Loan Agreement for income tax purposes and the original issue discount on the Loan Agreement shall be considered to be zero.

2.	WHEN ARE WE ENTITLED TO	O PURCHASE YOU	JR WARRANT STOCK.	
The	term of this Warrant Agreement and	Our right to purchase	e Warrant Stock will begin on the Effective Date and shall be available for the gro	eater of (i)
	() years through and including	and (ii)	years after the closing of Your initial public offering	

Notwithstanding the foregoing, Our right to purchase the Warrant Stock shall be automatically and fully exercised via the net issuance method described below (without surrender of the Warrant Agreement) upon the occurrence of a Merger Event, as defined below, with a Person that is not one of Your affiliates, in which Your common stock is exchanged for cash and/or stock of a class that is listed on a recognized national exchange, provided that, upon consummation of the Merger Event, the consideration payable to Us pursuant to such exercise and on account of the Warrant Stock consists of (i) cash or (ii) stock of a class that is listed on a recognized national exchange and the total per share value of such stock consideration is equal to or greater than two (2) times the aggregate Exercise Price (as adjusted). Further notwithstanding anything to the contrary contained herein, if the per share value of the consideration payable to holders of the Warrant Stock upon the consummation of a Merger Event in cash is less than the Exercise Price (as adjusted) and We have not elected to exercise this Warrant Agreement, then this Warrant Agreement shall automatically terminate as of in connection with the consummation of such Merger Event and shall be of no further force and effect (unless exercised by Us in connection with such Merger Event). No less than ten (10) days prior to any Merger Event, You shall provide Us with written notice of the proposed Merger Event together with a copy of the executed merger agreement, or other definitive documentation (and all schedules and exhibits thereto) and information concerning Your expected capitalization immediately prior to the Merger Event. Upon consummation of the Merger Event, You shall promptly provide Us with (a) a copy of any modifications or amendments to the executed merger agreement, (b) any other documents in connection therewith, (c) updated information, if any, concerning Your capitalization immediately prior to the Merger Event, and, (d) upon request, by Us any other

3. HOW WE MAY PURCHASE YOUR WARRANT STOCK.

We may exercise Our purchase rights, in whole or in part, at any time, or from time to time, prior to the expiration of the term of this Warrant Agreement, by giving You, except as provided in the last paragraph of this Section 3, a completed and executed Notice of Exercise in the form attached as Exhibit I. Promptly upon receipt of the Notice of Exercise and in any event no later than twenty-one (21) days after you have received Our Notice of Exercise and payment of the aggregate Exercise Price for the shares purchased, You will issue to Us a certificate (in electronic format, if You are issuing certificates electronically) for the number of shares of Warrant Stock that We have purchased and You will execute the Acknowledgment of Exercise in the form attached hereto as Exhibit II indicating the number of shares which will be available to Us for future purchases, if any.

We may pay for the Warrant Stock by either (i) cash or check, or (ii) by the net issuance method as determined below. If We elect the net issuance method, You will issue Warrant Stock using the following formula:

$$X = \underline{Y(A-B)}$$

the number of shares of Warrant Stock to be issued to Us.

the number of shares of Warrant Stock We request to be exercised under

this Warrant Agreement.

Y= this warrant Agreement. the fair market value of one share of Warrant Stock (as of the date of such calculation).

A=B= the Exercise Price (as of the date of such calculation).

For purposes of the above calculation and the below paragraphs, current fair market value of Warrant Stock shall mean with respect to each share of Warrant Stock:

If the exercise is in connection with the initial public offering of Your Common Stock and if Your registration statement relating to such public offering has been declared effective by the Securities and Exchange Commission, then the fair market value per share shall be the product of (x) the initial "Price to Public" specified in the final prospectus of the offering and (y) the number of shares of common stock into which each share of Warrant Stock is convertible at the time of such exercise.

If this Warrant Agreement is exercised after, and not in connection with Your initial public offering, and:

if Your common stock is traded on a securities exchange, the fair market value of the Warrant Stock shall be the product of (x) the average of the closing prices of Your common stock over a five (5) day period ending three (3) days before the day the current fair market value of the Warrant Stock is being determined and (y) the number of shares of Your common stock into which each share of Warrant Stock is convertible at the time of such exercise; or

• if Your common stock is actively tradedover-the-counter, the fair market value of the Warrant Stock shall be the product of (x) the average of the closing bid and asked prices of Your common stock over the five (5) day period ending three (3) days before the day the current fair market value of the Warrant Stock is being determined and (y) the number of shares of Your common stock into which each share of Warrant Stock is convertible at the time of such exercise.

If this Warrant Agreement is exercised prior to or after Your initial public offering, and:

• Your common stock is not listed on any securities exchange or theover-the-counter market, the current fair market value of one share of Warrant Stock shall be the product of (x) the fair market value of a share of Your common stock (the highest price per share which You could obtain from a willing buyer (not a current employee or director) for shares of common stock sold, from authorized but unissued shares), as determined in good faith by Your Board of Directors, and (y) the number of shares of common stock into which each share of Warrant Stock is convertible at the time of such exercise. Notwithstanding the foregoing, however, if You shall become subject to a merger, acquisition or other consolidation pursuant to which holders of Warrant Stock shall be entitled to receive cash, securities or other property, then the fair market value of the Warrant Stock shall be deemed to be the value received by the holders of the Warrant Stock (on a common equivalent basis) pursuant to such merger or acquisition or other consolidation.

During the term of this Warrant Agreement, except as provided in the next sentence, You will at all times from and after the Effective Date have authorized and reserved a sufficient number of shares of (a) Warrant Stock to provide for the exercise of our rights to purchase Warrant Stock, and (b) common stock to provide for the conversion of the Warrant Stock. Notwithstanding the foregoing, in no event shall You be required to authorize or reserve any of Your preferred stock issued in the Next Round unless and until both (x) the Next Round is consummated and (y) the Warrant Stock is Your preferred stock issued in the Next Round.

If We elect to exercise part of the Warrant Agreement, You will promptly issue to Us an amended Warrant Agreement stating the remaining number of shares that are available. All other terms and conditions of that amended Warrant Agreement shall be identical to those contained in this Warrant Agreement.

If at the end of the term of this Warrant Agreement, the fair market value of one share of Warrant Stock (or other security issuable upon the exercise hereof) as determined in accordance herewith is greater than the Exercise Price in effect on such date, then this Warrant Agreement shall automatically be deemed on and as of such date to be converted pursuant hereto as to all shares of Warrant Stock (or such other securities) for which it shall not previously have been exercised or converted, and You shall promptly deliver a certificate representing the shares of Warrant Stock (or such other securities) issued upon such conversion to Us.

4. WHEN WILL THE NUMBER OF SHARES AND EXERCISE PRICE CHANGE.

- If You are Acquired. Subject to Section 2, if at any time: (i) there is a reorganization of Your stock (other than a reclassification, exchange or subdivision of Your stock otherwise provided for in this Warrant Agreement); or (ii) You consummate a Deemed Liquidation Event (as defined in Your Certificate of Incorporation as amended through the Effective Date (Your "Certificate of Incorporation")) and a waiver of such Deemed Liquidation Event by Your shareholders shall not apply to Us (each such event referred to as a "Merger Event"), then, as a part of such Merger Event, lawful provision shall be made so that We shall thereafter be entitled to receive, upon exercise of Our rights under this Warrant Agreement, the number of shares of preferred stock or other securities of the successor or surviving person resulting from such Merger Event, equal in value to that which would have been issuable if We had exercised Our rights under this Warrant Agreement immediately prior to the Merger Event. In any such case, appropriate adjustment (as determined in good faith by Your Board of Directors) shall be made in the application of the provisions of this Warrant Agreement with respect to Our rights and interest after the Merger Event so that the provisions of this Warrant Agreement (including adjustments of the Exercise Price and number of shares of Warrant Stock purchasable) shall be applicable to the greatest extent possible.
- If You Reclassify Your Stock. If at any time, other than in connection with a Merger Event covered above, You combine, reclassify, exchange or subdivide Your securities or otherwise change any of the securities as to which purchase rights under this Warrant Agreement exist into the same or a different number of securities of any other class or classes, this Warrant Agreement will thereafter represent the right to acquire such number and kind of securities as would have been issuable as the result of such change with respect to the securities which were subject to the purchase rights under this Warrant Agreement immediately prior to such combination, reclassification, exchange, subdivision or other change.
- If You Subdivide or Combine Your Shares. If at any time, other than in connection with a Merger Event covered above, You combine or
 subdivide the Warrant Stock, the Exercise Price will be proportionately decreased in the case of a subdivision, or proportionately increased in the
 case of a combination.
- If You Pay Stock Dividends. If at any time, other than in connection with a Merger Event covered above, You pay a dividend payable in, or make any other distribution (except any distribution specifically provided for in the above paragraphs) of the Warrant Stock, then the Exercise Price shall be adjusted, from and after the record date of such dividend or distribution, to that price determined by multiplying the Exercise Price in effect immediately prior to such record date by a fraction (i) the numerator of which shall be the total number of all shares of the Warrant Stock outstanding immediately prior to such dividend or distribution, and (ii) the denominator of which shall be the total number of all shares of the Warrant Stock outstanding immediately after such dividend or distribution. We will thereafter be entitled to purchase, at the Exercise Price resulting from such adjustment, the number of shares of Warrant Stock (calculated to the nearest whole share) obtained by multiplying the Exercise Price in effect immediately prior to such adjustment by the number of shares of Warrant Stock issuable upon the exercise hereof immediately prior to such adjustment and dividing the product thereof by the Exercise Price resulting from such adjustment.

- "Pay to Play" Rights. In the event that We have not exercised this Warrant as to all the Warrant Stock and any "pay to play" terms or conditions (i.e. terms or conditions that require a holder of shares of Your preferred stock (the "Preferred Stock") to purchase securities in a future round of equity financing or else lose the benefit of anti-dilution protections applicable to shares of Preferred Stock or have such shares of Preferred Stock automatically convert into common stock or another class or series of capital stock) in Your Certificate of Incorporation are triggered in connection with any sale or issuance of securities (a "Trigger Event"), then, in each such event the purchase rights remaining under this Warrant Agreement shall automatically adjust to provide Us, upon the later exercise hereof, with the same securities and/or rights that We would have received had We (x) exercised such portion of this Warrant Agreement prior to such Trigger Event, and (y) participated in the applicable equity financing in an amount sufficient to be deemed to have fully participated for purposes of such "pay to play" provision.
- If You Change the Antidilution Rights of the Warrant Stock or Issue New Preferred or Convertible StockAll antidilution rights applicable to the Warrant Stock purchasable under this Warrant Agreement are as set forth in Your Certificate of Incorporation. You will promptly provide Us with any restatement, amendment, modification of or waiver of any right, in each case that relates to the Warrant Stock, under Your Certificate of Incorporation.

5. WE CAN TRANSFER THIS WARRANT AGREEMENT.

Subject to the terms and conditions contained in Section 7, We (or any successor transferee) may transfer in whole or in part this Warrant Agreement and all its rights. You will record the transfer on Your books when You receive Our Notice of Transfer in the form attached hereto as Exhibit III, and Our payment of all transfer taxes and other governmental charges involved in such transfer. Such transferee shall agree to be bound by all terms and conditions of this Warrant Agreement and make the representations and covenants (including with respect to the "Market Stand-Off") in Section 7 hereof to You on the effective date of such transfer.

6. REPRESENTATIONS, WARRANTIES, AND COVENANTS FROM YOU.

• Reservation of Warrant Stock. The Warrant Stock issuable upon exercise of Our rights under this Warrant Agreement will be duly and validly reserved (with it being acknowledged that, prior to the Next Round, You have not reserved any shares of the Next Round that this Warrant Agreement may be exercisable for) and when issued in accordance with the provisions of this Warrant Agreement will be validly issued, fully paid and non-assessable, and will be free of any taxes, liens, charges or encumbrances of any nature whatsoever; provided, however, that the Warrant Stock issuable pursuant to this Warrant Agreement may be subject to restrictions on transfer under state and/or Federal securities laws. Upon Our exercise, You will issue to Us certificates for shares of Warrant Stock without charging Us any tax, or other cost incurred by You in connection with such exercise and the related issuance of shares of Warrant Stock. You will not be required to pay any tax, which may be payable in respect of any transfer involved and the issuance and delivery of any certificate in a name other than _______.

- Due Authority. Your execution and delivery of this Warrant Agreement and the performance of Your obligations hereunder, including the issuance to Us of the right to acquire the shares of Warrant Stock, have been duly authorized by all necessary corporate action on Your part and this Warrant Agreement is not inconsistent with Your Certificate of Incorporation or Bylaws, do not contravene any law or governmental rule, regulation or order applicable to it, do not and will not contravene any provision of, or constitute a default under, any indenture, mortgage, material contract or other instrument to which You are a party or by which You are bound, and this Warrant Agreement constitutes a legal, valid and binding agreement, enforceable in accordance with its respective terms, subject to laws of general application relating to bankruptcy, insolvency and the relief of debtors and the rules of law or principles at equity governing specific performance, injunctive relief and other equitable remedies.
- Consents and Approvals. No consent or approval of, giving of notice to, registration with, or taking of any other action in respect of any state,
 Federal or other governmental authority or agency is required with respect to execution, delivery and Your performance of Your obligations under
 this Warrant Agreement, except for the corporate approvals in connection with the Next Round and the filing of any required notices pursuant to
 Federal and state securities laws, which filings will be effective by the times required thereby.
- Issued Securities. All of Your issued and outstanding shares of common stock, Warrant Stock or any other securities have been duly authorized and validly issued and are fully paid and nonassessable. All outstanding shares of common stock and Warrant Stock were issued in full compliance with all Federal and state securities laws. In addition, as of the Effective Date, and immediately prior to the issuance of this Warrant, Our capitalization is as set forth in the attached Schedule 1.
- Other Commitments to Register Securities. Except as set forth in this Warrant Agreement and the Investors' Rights Agreement, You are not, pursuant to the terms of any other agreement currently in existence, under any obligation to register under the 1933 Act any of Your presently outstanding securities or any of Your securities which may hereafter be issued.
- Exempt Transaction. Subject to the accuracy of Our representations in Section 7 hereof, the issuance of the Warrant Stock upon exercise of this Warrant Agreement will constitute a transaction exempt from (i) the registration requirements of Section 5 of the 1933 Act, in reliance upon Section 4(2) thereof, and (ii) the qualification requirements of the applicable state securities laws.
- Compliance with Rule 144. We may sell the Warrant Stock issuable hereunder in compliance with Rule 144 promulgated by the Securities and Exchange Commission. Within ten (10) days of Our request, You agree to furnish Us, a written statement confirming Your compliance with the filing requirements of the Securities and Exchange Commission as set forth in such Rule 144, as may be amended.
- No Impairment. Except as set forth herein, You agree not to, by amendment of Your Certificate of Incorporation, by-laws or other organizational or charter documents or

through a reorganization, transfer of assets, consolidation, merger, dissolution, issue or sale of securities or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms to be observed or performed under this Warrant Agreement by You, but shall at all times in good faith assist in carrying out of all the provisions of this Warrant Agreement and in taking all such action as may be necessary or appropriate to protect Our rights under this Warrant Agreement against impairment. However, notwithstanding the foregoing, You shall not be deemed to have impaired Our rights if You amend, restate, modify or waive any provisions of Your Certificate of Incorporation (by amendment, merger, reclassification, recapitalization or otherwise), or the holders of Your preferred stock waive their rights thereunder, in a manner that does not (individually or when considered in the context of any other actions being taken in connection with such amendments or waivers) the rights, privileges, preferences, restrictions and limitations of the Warrant Stock in a manner different from the effect that such amendments, restatements, modifications or waivers have generally on the rights, privileges, preferences, restrictions and limitations of the then outstanding securities of You that are of the same series and class as the Warrant Stock, it being expressly understood that an amendment, restatement, modification or waiver that applies equally to all holders of the class and/or series of securities then issuable upon exercise of this Warrant that has a different economic effect on You shall not be deemed to affect the rights, privileges, preferences, restrictions and limitations of the Warrant Stock in a manner different from the effect that such amendments, restatements, modification or waivers have generally on the rights, privileges, preferences, restrictions and limitations of the then outstanding securities of You that are of the same series and class as the Warrant Stock in a manner different

7. OUR REPRESENTATIONS AND COVENANTS TO YOU.

- Investment Purpose. The right to acquire Warrant Stock or the Warrant Stock issuable upon exercise of Our rights contained herein and the common stock issuable upon conversion will be acquired for investment purposes and not with a view to the sale or distribution of any part thereof, and We have no present intention of selling or engaging in any public distribution of the same in violation of the 1933 Act.
- **Private Issue.** We understand (i) that this Warrant Agreement, the Warrant Stock issuable upon exercise of this Warrant Agreement and the common stock issuable upon conversion of the Warrant Stock are not registered under the 1933 Act or qualified under applicable state securities laws on the ground that the issuance contemplated by this Warrant Agreement will be exempt from the registration and qualifications requirements thereof; and (ii) that Your reliance on such exemption is predicated on the representations set forth in this Section 7.
- Disposition of Our Rights. In no event will We make a disposition of any of Our rights to acquire Warrant Stock or Warrant Stock issuable upon exercise of such rights or the common stock issuable upon conversion of the Warrant Stock unless and until (i) We shall have notified You in writing of the proposed disposition, and (ii) the transferee agrees to be bound in writing to the applicable terms and conditions of this Warrant Agreement, and (iii) if You request, We shall have furnished You with an opinion of counsel satisfactory to You and Your counsel to the effect that (A) appropriate action necessary for compliance

with the 1933 Act has been taken, or (B) an exemption from the registration requirements of the 1933 Act is available. Notwithstanding the foregoing, the restrictions imposed upon the transferability of any of Our rights to acquire Warrant Stock or Warrant Stock issuable on the exercise of such rights or the common stock issuable upon conversion of the Warrant Stock do not apply to transfers from the beneficial owner of any of the aforementioned securities to its nominee or from such nominee to its beneficial owner, and shall terminate as to any particular share of Warrant Stock when (1) such security shall have been effectively registered under the 1933 Act and sold by the holder thereof in accordance with such registration or (2) such security shall have been sold without registration in compliance with Rule 144 under the 1933 Act, or (3) a letter shall have been issued to You at Our request by the staff of the Securities and Exchange Commission or a ruling shall have been issued to You at Our request by such Commission stating that no action shall be recommended by such staff or taken by such Commission, as the case may be, if such security is transferred without registration under the 1933 Act in accordance with the conditions set forth in such letter or ruling and such letter or ruling specifies that no subsequent restrictions on transfer are required. Whenever the restrictions imposed hereunder shall terminate, as hereinabove provided, the holder of a share of Warrant Stock then outstanding as to which such restrictions have terminated shall be entitled to receive from You, without expense to such holder, one or more new certificates for the Warrant or for such shares of Warrant Stock not bearing any restrictive legend referring to 1933 Act registration or exemption.

- Financial Risk. We have such knowledge and experience in financial and business matters and knowledge of Your business affairs and financial
 condition as to be capable of evaluating the merits and risks of Our investment, and have the ability to bear the economic risks of Our investment.
- Risk of No Registration. We understand that if You do not register with the Securities and Exchange Commission pursuant to Section 12 of the Securities Exchange Act of 1934 (the "1934 Act"), or file reports pursuant to Section 15(d), of the 1934 Act, or if a registration statement covering the securities under the 1933 Act is not in effect when We desire to sell (i) the rights to purchase Warrant Stock pursuant to this Warrant Agreement, or (ii) the Warrant Stock issuable upon exercise of the right to purchase, or (iii) the common stock issuable upon conversion of the Warrant Stock, We may be required to hold such securities for an indefinite period. We also understand that any sale of Our right to purchase Warrant Stock or Warrant Stock or common stock issuable upon conversion of the Warrant Stock, which might be made by it in reliance upon Rule 144 under the 1933 Act may be made only in accordance with the terms and conditions of that Rule.
- Accredited Investor. We are an "accredited investor" within the meaning of the Securities and Exchange Rule 501 of Regulation D of the 1933 Act, as presently in effect.
- "Market Stand-Off" Agreement. With respect to this Warrant, the Warrant Stock and any shares of common stock issuable upon conversion of the
 Warrant Stock, We hereby agree to be bound by the "Market Stand-off" provisions in Section 2.11 of Your Investors' Rights Agreement (as defined
 below). Notwithstanding the foregoing, in no event shall such market stand-off agreement restrict Our ability to exercise Our purchase rights under

this Warrant Agreement, including the transfer of common stock to You solely to satisfy the exercise price pursuant to the net issuance method and if there is any conflict between the Investors' Rights Agreement and this subsection, this subsection shall control.

8. NOTICES YOU AGREE TO

You agree to give Us at least ten (10) days prior written notice of the following events:

- If You pay a dividend or distribution declaration upon Your stock.
- · If You offer for subscription pro-rata to the existing shareholders additional stock or other rights.
- If You consummate or sign definitive documents providing for a Merger Event.
- · If You have an initial public offering.
- · If You dissolve or liquidate.

All notices in this Section must set forth details of the event, how the event adjusts either Our number of shares or Our Exercise Price and the method used for such adjustment.

Timely Notice. Your failure to timely provide such notice required above shall entitle Us to retain the benefit of the applicable notice period notwithstanding anything to the contrary contained in any insufficient notice received by Us; provided, however, that this will not prevent or delay You from consummating any of the transactions listed above.

9. DOCUMENTS YOU WILL PROVIDE US.

Upon signing this Warrant Agreement You will provide Us with:

- · Executed originals of this Warrant Agreement, and all other documents and instruments that We may reasonably require
- · Secretary's certificate of incumbency and authority
- · Certified copy of resolutions of Your board of directors approving this Warrant Agreement
- · Certified copy of Certificate of Incorporation and by-laws as amended through the Effective Date
- · The Investors' Rights Agreement

So long as this Warrant Agreement is in effect, You shall provide Us with the following:

• Within five (5) business days after the closing of any equity financing, or extension of an existing round of equity financing, occurring after the Effective Date, in which You issue preferred stock or other securities You will provide Us with copies of the fully executed

equity financing documents, including without limitation the related stock purchase agreement, investors rights agreement, voting agreement, amended or restated Certificates of Incorporation, current capitalization table and other related documents. Notwithstanding any term or condition contained in this Warrant Agreement to the contrary, Your failure to comply with this paragraph shall not constitute a default unless You have not provided the information requested within five (5) business days after Our request.

- Within thirty (30) days after completion You shall provide Us with any 409A Valuation Reports or other similar reports prepared for You.
 Notwithstanding any term or condition contained in this Warrant Agreement to the contrary, Your failure to comply with this paragraph shall not constitute a default unless You have not provided the information requested within five (5) business days after Our request.
- After all obligations under the Loan Agreement have been finally paid in full, within forty-five (45) days after the end of each quarter, You will provide Us with (1) an unaudited income statement, statement of cash flows, and an unaudited balance sheet prepared in accordance with GAAP accompanied by a report detailing any material contingencies, (2) (i) at all times that Your Board of Directors requires You to prepare audited financial statements, as soon as available, but no later than two hundred and ten (210) days after the last day of Your fiscal year, audited consolidated financial statements prepared under GAAP, consistently applied, together with an unqualified opinion on the financial statements from an independent certified public accounting firm reasonably acceptable to Us; and (ii) at all other times, as soon as available, but no later than sixty (60) days after the last day of Your fiscal year, a company-prepared consolidated balance sheet and income statement covering Your consolidated operations during such fiscal year.
- You shall submit to Us any other documents and other information that We may reasonably request from time to time and are necessary to
 implement the provisions and purposes of this Warrant Agreement. Notwithstanding any term or condition contained in this Warrant Agreement to
 the contrary, Your failure to comply with this paragraph shall not constitute a default unless You have not provided the information requested within
 five (5) days of Our request.

10. REGISTRATION RIGHTS UNDER THE 1933 ACT.

Concurrently with the issuance of this Warrant Agreement, the Amended and Restated Investors' Rights Agreement attached hereto as Exhibit IV shall be executed and delivered by You, Us, and by such other parties as are necessary to amend and restate the Prior Agreement (as defined in therein) (the "Investors' Rights Agreement").

11. OTHER LEGAL PROVISIONS THE PARTIES WILL ABIDE BY.

Effective Date. This Warrant Agreement shall be construed and shall be given effect in all respects as if it had been executed and delivered by the Parties on the date hereof. This Warrant Agreement shall be binding upon any of the successors or assigns of the Parties.

Attorney's Fees. In any litigation, arbitration or court proceeding between the Parties relating to this Warrant Agreement, the prevailing party shall be entitled to attorneys' fees and expenses and all costs of proceedings incurred in enforcing this Warrant Agreement.

Fractional Shares. No fractional shares shall be issued upon the exercise of this Warrant Agreement as a consequence of any adjustment pursuant hereto. All Warrant Stock (including fractions) issuable upon exercise of this Warrant Agreement may be aggregated for purposes of determining whether the exercise would result in the issuance of any fractional share. If, after aggregation, the exercise would result in the issuance of a fractional share, You shall, in lieu of issuance of any fractional share, pay Us, if We are otherwise entitled to such fraction, a sum in cash equal to the product resulting from multiplying the then current fair market value of the Warrant Stock by such fraction.

Governing Law. This Warrant Agreement shall be governed by and construed for all purposes under and in accordance with the laws of the State of California without giving effect to that body of law pertaining to conflicts of laws.

Consent to Jurisdiction and Venue. All judicial proceedings arising in or under or related to this Warrant Agreement may be brought in any state or federal court of competent jurisdiction located in the State of California. By execution and delivery of this Warrant Agreement, each party hereto generally and unconditionally: (a) consents to personal jurisdiction in San Mateo County, State of California; (b) waives any objection as to jurisdiction or venue in San Mateo County, State of California; (c) agrees not to assert any defense based on lack of jurisdiction or venue in the aforesaid courts; and (d) irrevocably agrees to be bound by any judgment rendered thereby in connection with this Warrant Agreement. Service of process on any party hereto in any action arising out of or relating to this Warrant Agreement shall be effective if given in accordance with the requirements for notice set forth in this Section, and shall be deemed effective and received as set forth therein. Nothing herein shall affect the right to serve process in any other manner permitted by law or shall limit the right of either party to bring proceedings in the courts of any other jurisdiction.

Mutual Waiver of Jury Trial; Judicial Reference. Because disputes arising in connection with complex financial transactions are most quickly and economically resolved by an experienced and expert person and the Parties wish applicable state and federal laws to apply (rather than arbitration rules), the Parties desire that their disputes be resolved by a judge applying such applicable laws. EACH OF THE PARTIES SPECIFICALLY WAIVES ANY RIGHT THEY MAY HAVE TO TRIAL BY JURY OF ANY CAUSE OF ACTION, CLAIM, CROSS-CLAIM, COUNTERCLAIM, THIRD PARTY CLAIM OR ANY OTHER CLAIM (COLLECTIVELY, "CLAIMS") ASSERTED BY YOU AGAINST US OR OUR ASSIGNEE OR BY US OR OUR ASSIGNEE AGAINST YOU. IN THE EVENT THAT THE FOREGOING JURY TRIAL WAIVER IS NOT ENFORCEABLE, ALL CLAIMS, INCLUDING ANY AND ALL QUESTIONS OF LAW OR FACT RELATING THERETO, SHALL, AT THE WRITTEN REQUEST OF ANY PARTY, BE DETERMINED BY JUDICIAL REFERENCE PURSUANT TO THE CALIFORNIA CODE OF CIVIL PROCEDURE ("REFERENCE"). THE PARTIES SHALL SELECT A SINGLE NEUTRAL REFEREE, WHO SHALL BE A RETIRED STATE OR FEDERAL JUDGE. IN THE EVENT THAT THE PARTIES CANNOT AGREE UPON A REFEREE, THE REFEREE SHALL BE APPOINTED BY THE COURT. THE REFEREE

SHALL REPORT A STATEMENT OF DECISION TO THE COURT. NOTHING IN THIS SECTION SHALL LIMIT THE RIGHT OF ANY PARTY AT ANY TIME TO EXERCISE LAWFUL SELF-HELP REMEDIES, FORECLOSE AGAINST COLLATERAL OR OBTAIN PROVISIONAL REMEDIES. THE PARTIES SHALL BEAR THE FEES AND EXPENSES OF THE REFEREE EQUALLY UNLESS THE REFEREE ORDERS OTHERWISE. THE REFEREE SHALL ALSO DETERMINE ALL ISSUES RELATING TO THE APPLICABILITY, INTERPRETATION, AND ENFORCEABILITY OF THIS SECTION. THE PARTIES ACKNOWLEDGE THAT THE CLAIMS WILL NOT BE ADJUDICATED BY A JURY. This waiver extends to all such Claims, including Claims that involve persons other than You and Us; Claims that arise out of or are in any way connected to the relationship between You and Us; and any Claims for damages, breach of contract, specific performance, or any equitable or legal relief of any kind, arising out of this Warrant Agreement.

Counterparts. This Warrant Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

Notices. Any notice required or permitted under this Warrant Agreement shall be given in writing and shall be deemed effectively given upon the earlier of (1) actual receipt or 3 days after mailing if mailed postage prepaid by regular or airmail to Us or You or (2) one day after it is sent by overnight mail via nationally recognized courier or (3) on the same day as sent via confirmed facsimile transmission, provided that the original is sent by personal delivery or mail by the sending party.

Remedies. In the event of any default hereunder, the non-defaulting party may proceed to protect and enforce its rights either by suit in equity and/or by action at law, including but not limited to an action for damages as a result of any such default, and/or an action for specific performance for any default where such party will not have an adequate remedy at law and where damages will not be readily ascertainable. Each party expressly acknowledges and agrees that there is no adequate remedy at law for any breach of this Warrant Agreement and that in the event of any breach of this Warrant Agreement, the injured party shall be entitled to specific performance of any or all provisions hereof or an injunction prohibiting the other party from continuing to commit any such breach of this Warrant Agreement.

Survival. The representations, warranties, covenants, and conditions of the Parties contained herein or made pursuant to this Warrant Agreement shall survive the execution and delivery of this Warrant Agreement.

Severability. In the event any one or more of the provisions of this Warrant Agreement shall for any reason be held invalid, illegal or unenforceable, the remaining provisions of this Warrant Agreement shall be unimpaired, and the invalid, illegal or unenforceable provision shall be replaced by a mutually acceptable valid, legal and enforceable provision, which comes closest to the intention of the Parties underlying the invalid, illegal or unenforceable provision.

Entire Agreement. This Warrant Agreement constitutes the entire agreement between the Parties pertaining to the subject matter contained in it and supersedes all prior and contemporaneous agreements, representations and undertakings of the Parties, whether oral or written, with respect to such subject matter

Amendments. Any provision of this Warrant Agreement may only be amended by a written instrument signed by the Parties.

Lost Warrants or Stock Certificates. You covenant to Us that, upon receipt of evidence reasonably satisfactory to Us of the loss, theft, destruction or mutilation of this Warrant Agreement or any stock certificate and, in the case of any such loss, theft or destruction, upon receipt of an indemnity reasonably satisfactory to You, or in the case of any such mutilation upon surrender and cancellation of such Warrant Agreement or stock certificate, You will make and deliver a new Warrant Agreement or stock certificate, of like tenor, in lieu of the lost, stolen, destroyed or mutilated Warrant Agreement or stock certificate.

Rights as Stockholders. We shall not, as a party to this Warrant Agreement, be entitled to vote or receive dividends or be deemed the holder of Warrant Stock or any of Your other securities which may at any time be issuable upon the exercise hereof for any purpose, nor shall anything contained herein be construed to confer upon Us any of the rights of one of Your stockholders or any right to vote for the election of directors or upon any matter submitted to stockholders at any meeting thereof, or to receive dividends or subscription rights or otherwise until this Warrant Agreement is exercised and the shares purchasable upon the exercise hereof shall have become deliverable, as provided herein.

Signatures. This Warrant Agreement may be executed and delivered by facsimile or transmitted electronically in either Tagged Image Format Files ("TIFF") or Portable Document Format ("PDF") and, upon such delivery, the facsimile, TIFF or PDF signature, as applicable, will be deemed to have the same effect as if the original signature had been delivered to the other party.

(Signature Page to Follow)

IN WITNESS WHEREOF, each of the Parties have caused this Warrant Agreement to be executed by its officers who are duly authorized as of the Effective Date.

You:	GROVE COLLABORATIVE, INC.
Signature:	
Print Name:	
Title:	
Us:	
Signature:	
Print Name:	-
Title:	

[SIGNATURE PAGE TO WARRANT AGREEMENT]

EXHIBIT I

NOTICE OF EXERCISE

To:						
1.	We hereby elect to purchase [] shares of the Series [] Preferred Stock of [], pursuant to the terms of the Plain English Warrant Agreement dated the [] day of [], [20] (the "Warrant Agreement") between You and Us, We hereby tender here payment of the purchase price for such shares in full, together with all applicable transfer taxes, if any.					
2.	Method of Exercise (Please initial the applicable blank)					
	a The undersigned elects to exercise the Warrant Agreement by means of a cash payment, and gives You full payment for the purchase price of the shares being purchased, together with all applicable transfer taxes, if any.					
	bThe undersigned elects to exercise the Warrant Agreement by means of the net issuance method of Section 3 of the Warrant Agreement.					
3.	In exercising Our rights to purchase the Series [] Preferred Stock of [], We hereby confirm and acknowledge the investment representations, warranties and covenants made in Section 7 of the Warrant Agreement.					
	se issue a certificate or certificates representing these purchased shares of Series [] Preferred Stock in Our name or in such other name as is ified below.					
	(Name)					
	(Address)					
	US:					
	Ву:					
	Title:					
	Data					

EXHIBIT II

ACKNOWLEDGMENT OF EXERCISE

, hereby acknowledges receipt of the "Notice of Exercise" from	to purchase shares of the Series
Preferred Stock of [] pursuant to the terms of the Plain English W	arrant Agreement dated the [] day of [], [20] (the
"Warrant Agreement") and further acknowledges that [] shares remain subject to purchase under the terms of the Warrant Agreement	
YOU: GROVE COLLABORATIVE,	
	Ву:
	Title:
	Date:

EXHIBIT III

TRANSFER NOTICE

FOR VALUE RECEIVED, the foregoing Warrant Agreement and all rights evidenced thereby are hereby transferred and assigned to

(Please Print)	
Whose address is	
Dated:	
Holder's Signature:	
Holder's Address:	
Transferee's Signature:	
Transferee's Address:	
Signature Guaranteed:	

NOTE: The signature to this Transfer Notice must correspond with the name as it appears on the face of the Warrant Agreement, without alteration or enlargement or any change whatever. Officers of corporations and those acting in a fiduciary or other representative capacity should file proper evidence of authority to assign the foregoing Warrant Agreement.

EXHIBIT IV

A&R INVESTORS' RIGHTS AGREEMENT

AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

by and among

GROVE COLLABORATIVE HOLDINGS, INC.,

and

THE STOCKHOLDERS THAT ARE SIGNATORIES HERETO

Dated as of June 16, 2022

TABLE OF CONTENTS

		Page
SECTION 1. CERTAIN DEFINITIONS		1
SECTION 2. REGISTRATION RIGHTS		6
2.1.	Demand Registrations	6
2.2.	Piggyback Registrations	10
2.3.	Allocation of Securities Included in Registration Statement	11
2.4.	Registration Procedures	13
2.5.	Registration Expenses	20
2.6.	Certain Limitations on Registration Rights	21
2.7.	Limitations on Sale or Distribution of Other Securities	21
2.8.	No Required Sale	21
2.9.	Indemnification	22
2.10.	No Inconsistent Agreements	25
SECTION	3. UNDERWRITTEN OFFERINGS	26
3.1.	Requested Underwritten Offerings	26
3.2.	Piggyback Underwritten Offerings	26
SECTION 4. GENERAL		26
4.1.	Adjustments Affecting Registrable Securities	26
4.2.	Rule 144	26
4.3.	Nominees for Beneficial Owners	27
4.4.	Amendments and Waivers	27
4.5.	Notices	27
4.6.	Successors and Assigns	28
4.7.	Termination	28
4.8.	Entire Agreement	28
4.9.	Governing Law; Jurisdiction; WAIVER OF JURY TRIAL	29
4.10.	Interpretation; Construction	29
4.11.	Counterparts	29
4.12.	Severability	29
4.13.	Specific Enforcement	30
4.14.	Further Assurances	30

TABLE OF CONTENTS

(continued)

		Page
4.15.	Confidentiality	30
4.16.	Opt-Out Requests	31
4.17.	Original Registration Rights Agreement	31
Exhibit A	Joinder Agreement	

AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT, dated as of June 16, 2022 (as amended, restated, supplemented or otherwise modified from time to time, this "Agreement"), is made and entered into by and among (i) Grove Collaborative Holdings, Inc., a Delaware public benefit corporation domesticated from Virgin Group Acquisition Corp. II, a Cayman Islands exempted company (the "Company"), (ii) the stockholders of the Company party hereto (the "Stockholders") and (iii) any person or entity who hereafter becomes a party to this Agreement pursuant to Section 4.6 of this Agreement (each, a "Holder" and collectively with the Stockholders, the "Holders").

RECITALS:

WHEREAS, the Company, Treehouse Merger Sub, Inc., a Delaware corporation and a wholly owned subsidiary of the Company ('Merger Sub I'), Treehouse Merger Sub II, LLC, a Delaware limited liability company and a wholly owned subsidiary of the Company ("Merger Sub II") and Grove Collaborative, Inc., a Delaware public benefit corporation ("Grove"), have entered into an Amended and Restated Agreement and Plan of Merger, dated as of March 31, 2022 (as amended from time to time on or prior to the date hereof, the "Merger Agreement"), pursuant to which, among other things, Merger Sub I merged with and into Grove with Grove continuing as the surviving entity and a subsidiary of the Company (the "Initial Merger") and, immediately following the Initial Merger, Grove merged with and into Merger Sub II with Merger Sub II continuing as the surviving entity and a subsidiary of the Company (the "Final Merger" and, together with the Initial Merger, the "Mergers");

WHEREAS, the Company and Virgin Group Acquisition Sponsor II LLC, a Cayman Island limited liability company and a Stockholder (the "Sponsor") are parties to that certain Registration and Shareholder Rights Agreement, dated as of March 22, 2021 (the 'Original Registration Rights Agreement"), which shall be amended and restated by this Agreement;

WHEREAS, following the closing of the Mergers (the "Closing"), the Sponsor and the other Stockholders owned shares of Class A Common Stock, par value \$0.0001 per share of the Company (the "Class A Common Stock"), Class A Common Stock Equivalents (as defined herein), shares of Class B Common Stock, par value \$0.0001 per share of the Company (the "Class B Common Stock"), which are convertible on a share for share basis into shares of Class A Common Stock, and/or Class B Common Stock Equivalents (as defined herein);

WHEREAS, each of the Stockholders (other than the Sponsor and Corvina Holdings Limited) beneficially owns at least 5% of the Common Stock; and

WHEREAS, in connection with the Mergers, the Company has agreed to provide the registration rights set forth in this Agreement.

NOW, THEREFORE, in consideration of the premises and of the mutual covenants and obligations hereinafter set forth, the parties hereto hereby agree as follows:

Section 1. Certain Definitions. As used herein, the following terms shall have the following meanings:

"Additional Piggyback Rights" has the meaning ascribed to such term in Section 2.3(a).

"Affiliate" means, with respect to any Person, any other Person that directly or indirectly controls, is controlled by or is under common control with, such Person. For the purposes of this definition "control" (including, with correlative meanings, the terms "controlling", "controlled by" and "under common control with"), with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such specified Person, whether through the ownership of voting securities, by contract or otherwise. For the avoidance of doubt, neither the Company nor any Person controlled by the Company shall be deemed to be an Affiliate of any Holder.

"Agreement" has the meaning ascribed to such term in the Preamble.

"Automatic shelf registration statement" has the meaning ascribed to such term in Section 2.4.

"Board" means the Board of Directors of the Company.

"Business Day" means a day, other than Saturday, Sunday or other day on which commercial banks in New York, New York are authorized or required by law to close.

"Claims" has the meaning ascribed to such term in Section 2.9(a).

"Class A Common Stock" has the meaning ascribed to such term in the recitals.

"Class A Common Stock Equivalents" means all shares of Class B Common Stock, all Class B Common Stock Equivalents, and all options, warrants and other securities convertible into, or exchangeable or exercisable for (at any time or upon the occurrence of any event or contingency and without regard to any vesting or other conditions to which such securities may be subject), shares of Class A Common Stock (including any note or debt security convertible into or exchangeable for shares of Class A Common Stock).

"Class B Common Stock" has the meaning ascribed to such term in the recitals.

"Class B Common Stock Equivalents" means all options, warrants and other securities convertible into, or exchangeable or exercisable for (at any time or upon the occurrence of any event or contingency and without regard to any vesting or other conditions to which such securities may be subject), shares of Class B Common Stock (including any note or debt security convertible into or exchangeable for shares of Class B Common Stock).

"Common Stock" means all shares existing or hereafter authorized of the Class A Common Stock and Class B Common Stock, and any class of common stock of the Company and any and all securities of any kind whatsoever which may be issued after the date hereof in respect of, or in exchange for, such shares of common stock of the Company pursuant to a merger, consolidation, stock split, stock dividend or recapitalization of the Company or otherwise.

"Company" has the meaning ascribed to such term in the Preamble.

"Confidential Information" has the meaning ascribed to such term in Section 4.15.

"Demand Exercise Notice" has the meaning ascribed to such term in Section 2.1(b)(i).

"Demand Registration" has the meaning ascribed to such term in Section 2.1(b)(i).

"Demand Registration Period" has the meaning ascribed to such term in Section 2.1(b)(i).

"Demand Registration Request" has the meaning ascribed to such term in Section 2.1(b)(i).

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations of the SEC issued under such Act, as they may from time to time be in effect.

"Expenses" means any and all fees and expenses incident to the Company's performance of or compliance with Section 2, including: (i) SEC, stock exchange, FINRA and all other registration and filing fees and all listing fees and fees with respect to the inclusion of securities on the Nasdaq or on any other U.S. or non-U.S. securities market on which the Registrable Securities are listed or quoted, (ii) fees and expenses of compliance with state securities or "blue sky" laws of any state or jurisdiction of the United States or compliance with the securities laws of foreign jurisdictions and in connection with the preparation of a "blue sky" survey, including reasonable fees and expenses of outside "blue sky" counsel and securities counsel in foreign jurisdictions, (iii) word processing, printing and copying expenses, (iv) messenger and delivery expenses, (v) expenses incurred in connection with any road show, (vi) fees and disbursements of counsel for the Company, (vii) with respect to each registration or underwritten offering, the reasonable fees and disbursements of one counsel for the Initiating Holder and one counsel for all other Participating Holder(s) collectively (selected by the holders of a majority of the Registrable Securities held by such other Participating Holder(s)), together in each case with any local counsel, provided that expenses payable by the Company pursuant to this clause (vii) shall not exceed (1) \$150,000 for the first registration pursuant to this Agreement and (2) \$100,000 for each subsequent registration, (viii) fees and disbursements of all independent public accountants (including the expenses of any opinion and/or audit/review and/or "comfort" letter and updates thereof) and fees and expenses of other Persons, including special experts, retained by the Company, (ix) fees and expenses payable to a Qualified Independent Underwriter (but expressly excluding any underwriting discounts and commissions), (x) fees and expenses of any transfer agent or custodian, (xi) any other fees and disbursements of underwriters, if any, customarily paid by issuers or sellers of securities, including reasonable fees and expenses of counsel for the underwriters in connection with any filing with or review by FINRA (but expressly excluding any underwriting discounts and commissions) and (xii) rating agency fees and expenses.

"Final Merger" has the meaning ascribed to such term in the Recitals.

"FINRA" means the Financial Industry Regulatory Authority, Inc.

"Initial Merger" has the meaning ascribed to such term in the Recitals.

"Initiating Holders" means (a) Holders of at least thirty percent (30%) of the Registrable Securities then outstanding or (b) the Sponsor.

- "Joinder Agreement" means a writing in the form set forth in Exhibit A hereto whereby a new Holder of Registrable Securities becomes a party to, and agrees to be bound, to the same extent as its transferor, as applicable, by the terms of this Agreement.
- "Majority Participating Holders" means Participating Holders holding more than 50% of the Registrable Securities proposed to be included in any offering of Registrable Securities by such Participating Holders pursuant to Section 2.1 or Section 2.2.
 - "Manager" means the lead managing underwriter of an underwritten offering.
 - "Merger Agreement" has the meaning ascribed to such term in the Recitals.
 - "Merger Sub I" has the meaning ascribed to such term in the Recitals.
 - "Merger Sub II" has the meaning ascribed to such term in the Recitals.
 - "Mergers" has the meaning ascribed to such term in the Recitals.
 - "Minimum Threshold" means \$50.0 million.
 - "Opt-Out Request" has the meaning ascribed to such term in Section 4.16.
- "Participating Holders" means all Holders of Registrable Securities which are proposed to be included in any offering of Registrable Securities pursuant to Section 2.1 or Section 2.2.
- "Person" means any individual, firm, corporation, company, limited liability company, partnership, trust, joint stock company, business trust, incorporated or unincorporated association, joint venture, governmental authority or other legal entity of any nature whatsoever.
 - "Piggyback Notice" has the meaning ascribed to such term in Section 2.2(a).
 - "Piggyback Shares" has the meaning ascribed to such term in Section 2.3(a)(ii).
 - "Postponement Period" has the meaning ascribed to such term in Section 2.1(c).
 - "Qualified Independent Underwriter" means a "qualified independent underwriter" within the meaning of FINRA Rule 5121.
- "Registrable Securities" means (a) any shares of Class A Common Stock held by the Holders at any time (including those held as a result of, or issuable upon, the conversion or exercise of Class A Common Stock Equivalents) or any other equity security other than Class B Common Stock or Class B Common Stock Equivalents (including warrants to purchase shares of Class A Common Stock), whether now owned or acquired by the Holders at a later time, (b) any shares of Class A Common Stock or any other equity security other than Class B Common Stock or Class B Common Stock Equivalents (including warrants to purchase shares of Class A Common Stock) issued or issuable, directly or indirectly, in exchange for or with respect to the Common Stock or any other equity security (including warrants to purchase shares of Class A Common Stock) referenced in clause (a) above by way of stock dividend, stock split or combination of shares or in

connection with a reclassification, recapitalization, merger, share exchange, consolidation or other reorganization and (c) any securities other than Class B Common Stock or Class B Common Stock Equivalents issued in replacement of or exchange for any securities described in clause (a) or (b) above. Class B Common Stock and Class B Common Stock Equivalents shall not constitute Registrable Securities hereunder, provided that the Class A Common Stock issuable upon conversion of such Class B Common Stock and underlying Class B Common Stock Equivalents are Registrable Securities for all purposes hereunder as though, in each case, such shares of Class A Common Stock were outstanding on the date hereof. For purposes of this Agreement, a Person will be deemed to be a holder of Registrable Securities whenever such Person has the right to acquire, directly or indirectly, such Registrable Securities (including upon conversion, exercise or exchange of any equity interests but disregarding any restrictions or limitations upon the exercise of such right), whether or not such acquisition has actually been effected, and such Person shall not be required to convert, exercise or exchange such equity interests (or otherwise acquire such Registrable Securities) to participate in any registered offering hereunder until the closing of such offering. As to any particular Registrable Securities, such securities shall cease to be Registrable Securities when (A) a registration statement with respect to the sale of such securities shall have been declared effective under the Securities Act and such securities shall have been disposed of in accordance with such registration statement, (B) such securities shall have been disposed of in compliance with the requirements of Rule 144, (C) such securities have been sold in a public offering of securities or (D) such securities have ceased to be outstanding.

"Rule 144" have the meaning ascribed to such term in Section 4.2.

"SEC" means the U.S. Securities and Exchange Commission or such other federal agency which at such time administers the Securities Act.

"Section 2.3(a) Sale Number" has the meaning ascribed to such term in Section 2.3(a).

"Section 2.3(b) Sale Number" has the meaning ascribed to such term in Section 2.3(b).

"Section 2.3(c) Sale Number" has the meaning ascribed to such term in Section 2.3(c).

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations of the SEC issued under such Act, as they may from time to time be in effect.

"Shelf Registrable Securities" has the meaning ascribed to such term in Section 2.1(a)(ii).

"Shelf Registration Statement" has the meaning ascribed to such term in Section 2.1(a)(i).

"Shelf Underwriting" has the meaning ascribed to such term in Section 2.1(a)(ii).

"Shelf Underwriting Initiating Holders" has the meaning ascribed to such term in Section 2.1(a)(ii).

"Shelf Underwriting Notice" has the meaning ascribed to such term in Section 2.1(a)(ii).

"Shelf Underwriting Request" has the meaning ascribed to such term in Section 2.1(a)(ii).

"Subsidiary" means any direct or indirect subsidiary of the Company on the date hereof and any direct or indirect subsidiary of the Company organized or acquired after the date hereof.

"Underwritten Block Trade" has the meaning ascribed to such term in Section 2.1(a)(ii).

"Valid Business Reason" has the meaning ascribed to such term in Section 2.1(c).

"WKSI" means a "well-known seasoned issuer" (as defined in Rule 405 of the Securities Act).

Section 2. Registration Rights.

2.1. Demand Registrations.

(a) (i) As soon as practicable but no later than thirty (30) calendar days following the closing of the Merger (the 'Filing Date'), the Company shall prepare and file with the SEC a shelf registration statement under Rule 415 of the Securities Act (such registration statement, a "Shelf Registration Statement") covering the resale of all the Registrable Securities (determined as of two business days prior to such filing) on a delayed or continuous basis and shall use its commercially reasonable efforts to have such Shelf Registration Statement declared effective as soon as practicable after the filing thereof and no later than the earlier of (x) the ninetieth (90th) calendar day following the Filing Date if the Commission notifies the Company that it will "review" the Shelf Registration Statement and (y) the tenth (10th) business day after the date the Company is notified in writing by the SEC that such Shelf Registration Statement will not be "reviewed" or will not be subject to further review. Such Shelf Registration Statement shall provide for the resale of the Registrable Securities included therein pursuant to any method or combination of methods legally available to, and requested by, any Holder named therein. The Company shall maintain the Shelf Registration Statement in accordance with the terms hereof, and shall prepare and file with the SEC such amendments, including post-effective amendments, and supplements as may be necessary to keep a Shelf Registration Statement continuously effective, available for use to permit all Holders named therein to sell their Registrable Securities included therein and in compliance with the provisions of the Securities Act until such time as there are no longer any Registrable Securities. In the event the Company files a Shelf Registration Statement on Form S-1, the Company shall use its commercially reasonable efforts to convert such Shelf Registration Statement to a Shelf Registration Statement on Form S-3 as soon as practicable after the Company is eligible to use Fo

(ii) Subject to Section 2.1(c) and the provisions below with respect to the Minimum Threshold, following the expiration of any applicable lock-up agreement, each Holder (or Holders) shall have the right at any time and from time to time to elect to sell all or any part of its Registrable Securities pursuant to an underwritten offering pursuant to the Shelf Registration Statement by delivering a written request therefor to the Company specifying the number of Registrable Securities to be included in such registration and the intended method of distribution thereof. The Holder or Holders shall make such election by delivering to the Company a written request (a "Shelf Underwriting Request") for such underwritten offering specifying the number of Registrable Securities that the Holder or Holders desire to sell pursuant to such underwritten offering (the "Shelf Underwriting"). With respect to any Shelf Underwriting Request,

the Holder or Holders making such demand shall be referred to as the 'Shelf Underwriting Initiating Holders'. As promptly as practicable, but no later than two (2) Business Days after receipt of a Shelf Underwriting Request, the Company shall give written notice (the "Shelf Underwriting Notice") of such Shelf Underwriting Request to the Holders of record of other Registrable Securities registered on such Shelf Registration Statement ("Shelf Registrable") Securities"). The Company, subject to Sections 2.3 and 2.6, shall include in such Shelf Underwriting (x) the Registrable Securities of the Shelf Underwriting Initiating Holders and (y) the Shelf Registrable Securities of any other Holder of Shelf Registrable Securities which shall have made a written request to the Company for inclusion in such Shelf Underwriting (which request shall specify the maximum number of Shelf Registrable Securities intended to be disposed of by such Holder) within five (5) days after the receipt of the Shelf Underwriting Notice. The Company shall, as expeditiously as possible (and in any event within fifteen (15) Business Days after the receipt of a Shelf Underwriting Request), but subject to Section 2.1(b), use its reasonable best efforts to effect such Shelf Underwriting. The Company shall, at the request of any Shelf Underwriting Initiating Holder or any other Holder of Registrable Securities registered on such Shelf Registration Statement, file any prospectus supplement or, if the applicable Shelf Registration Statement is an automatic shelf registration statement, any post-effective amendments and otherwise take any action necessary to include therein all disclosure and language deemed necessary or advisable by the Shelf Underwriting Initiating Holders or any other Holder of Shelf Registrable Securities to effect such Shelf Underwriting. Notwithstanding anything to the contrary in this Section 2.1(a)(ii), each Shelf Underwriting must include, in the aggregate, Registrable Securities having an aggregate market value of at least the Minimum Threshold (based on the Registrable Securities included in such Shelf Underwriting by all Participating Holders). In connection with any Shelf Underwriting (including an Underwritten Block Trade), the Company shall have the right to designate the Manager and each other managing underwriter in connection with any such Shelf Underwriting or Underwritten Block Trade, subject to Shelf Underwriting Initiating Holders' reasonable approval. Notwithstanding the foregoing, if a Shelf Underwriting Initiating Holder wishes to engage in an underwritten block trade or similar transaction or other transaction with a 2-day or less marketing period (collectively, "Underwritten Block Trade") off of a Shelf Registration Statement, then notwithstanding the foregoing time periods, such Shelf Underwriting Initiating Holder only needs to notify the Company of the Underwritten Block Trade two (2) Business Days prior to the day such offering is to commence and the Holders of record of other Registrable Securities shall not be entitled to notice of such Underwritten Block Trade and shall not be entitled to participate in such Underwritten Block Trade.

(b) (i) At any time after the first anniversary of the Closing Date that a Shelf Registration Statement as required by Section 2.1(a) is not available for use by the Holders (a "Demand Registration Period") other than pursuant to Section 2.1(c), subject to this Section 2.1(b) and Sections 2.1(c) and 2.3) and the provisions below with respect to the Minimum Threshold, at any time and from time to time during such Demand Registration Period, each Initiating Holder (or Initiating Holders) shall have the right to require the Company to effect one or more registration statements under the Securities Act covering all or any part of its Registrable Securities by delivering a written request therefor to the Company specifying the number of Registrable Securities to be included in such registration and the intended method of distribution thereof. Any such request by any Initiating Holder or Initiating Holders pursuant to this Section 2.1(b)(i) is referred to herein as a "Demand Registration Request." and the registration so requested is referred to herein as a "Demand Registration". Subject to Section 2.1(c), the Initiating

Holders shall be entitled to request (and the Company shall be required to effect) an unlimited number of Demand Registrations. The Company shall give written notice (the "Demand Exercise Notice") of such Demand Registration Request to each of the Holders of record of Registrable Securities in accordance with Section 2.2, and, subject to Sections 2.3 and 2.6, shall include in a Demand Registration (x) the Registrable Securities of the Initiating Holders and (y) the Registrable Securities of any other Holder of Registrable Securities which shall have made a written request to the Company for inclusion in such registration pursuant to Section 2.2. Notwithstanding anything to the contrary in this Section 2.1(b)(i), each Demand Registration must include, in the aggregate, Registrable Securities having an aggregate market value of at least the Minimum Threshold (based on the Registrable Securities included in such Demand Registration by all Holders participating in such Demand Registration). In connection with any Demand Registration, the Company shall have the right to designate the Manager and each other managing underwriter in connection with any underwritten offering pursuant to such registration, subject to the Initiating Holders' reasonable approval; provided that in each case, each such underwriter is reasonably satisfactory to the Company, which approval shall not be unreasonably withheld or delayed.

- (ii) The Company shall, as expeditiously as possible, but subject to Section 2.1(c), use its reasonable best efforts to (x) file or confidentially submit with the SEC (no later than (A) sixty (60) days from the Company's receipt of the applicable Demand Registration Request if the Demand Registration is on Form S-1 or similar long-form registration and or (B) thirty (30) days from the Company's receipt of the applicable Demand Registration Request if the Demand Registration is on Form S-3 or any similar short-form registration), (y) cause to be declared effective as soon as reasonably practicable such registration statement under the Securities Act that includes the Registrable Securities which the Company has been so requested to register for distribution in accordance with the intended method of distribution, and (z) if requested by the Initiating Holders, obtain acceleration of the effective date of the registration statement relating to such registration.
- (c) Notwithstanding anything to the contrary in Section 2.1(a) or Section 2.1(b), the Shelf Underwriting and Demand Registration rights granted in Section 2.1 (a) and Section 2.1(b) are subject to the following limitations: (i) the Company shall not be required to cause a registration statement filed pursuant to Section 2.1(b) to be declared effective within a period of ninety (90) days after the effective date of any other registration statement of the Company filed pursuant to the Securities Act (other than a Form S-4, Form S-8 or a comparable form or an equivalent registration form then in effect); (ii) the Company shall not be required to effect more than three (3) Demand Registrations on Form S-1 or any similar long-form registration statement at the request of the Holders in the aggregate; (iii) if the Board, in its good faith judgment, determines that any registration of Registrable Securities or Shelf Underwriting should not be made or continued because it would materially and adversely interfere with any existing or potential financing, acquisition, corporate reorganization, merger, share exchange or other transaction or event involving the Company or any of its subsidiaries or would otherwise result in the public disclosure of information that the Board in good faith has a bona fide business purpose for keeping confidential (a "Valid Business Reason"), then (x) the Company may postpone filing or confidentially submitting a registration statement relating to a Shelf Underwriting Request until five (5) Business Days after such Valid Business Reason no longer exists, but in no event for more than forty-five (45) days after the date the Board determines a Valid Business Reason exists or (y) if a registration

statement has been filed or confidentially submitted relating to a Demand Registration Request or a prospectus supplement has been filed relating to a Shelf Underwriting Request, the Company may, to the extent determined in the good faith judgment of the Board to be reasonably necessary to avoid interference with any of the transactions described above, suspend use of or, if required by the SEC, cause such registration statement to be withdrawn and its effectiveness terminated or may postpone amending or supplementing such registration statement until five (5) Business Days after such Valid Business Reason no longer exists, but in no event for more than forty-five (45) days after the date the Board determines a Valid Business Reason exists (such period of postponement or withdrawal under this clause (iv), the "Postponement Period"). The Company shall give written notice to the Initiating Holders or Shelf Underwriting Initiating Holders and any other Holders that have requested registration pursuant to Section 2.2 of its determination to postpone or suspend use of or withdraw a registration statement and of the fact that the Valid Business Reason for such postponement or suspension or withdrawal no longer exists, in each case, promptly after the occurrence thereof; provided, however, that the Company shall not be entitled to more than two (2) Postponement Periods during any twelve (12) month period.

Each Holder of Registrable Securities agrees that, upon receipt of any notice from the Company that the Company has determined to suspend use of, withdraw, terminate or postpone amending or supplementing any registration statement pursuant to clause (c)(iii) above, such Holder will discontinue its disposition of Registrable Securities pursuant to such registration statement. If the Company shall have suspended use of, withdrawn or terminated a registration statement filed under Section 2.1(b)(i) (whether pursuant to clause (c)(iii) above or as a result of any stop order, injunction or other order or requirement of the SEC or any other governmental agency or court), the Company shall not be considered to have effected a Demand Registration for the purposes of this Agreement and such request shall not count as a Demand Registration Request under this Agreement until the Company shall have permitted use of such suspended registration statement or filed a new registration statement covering the Registrable Securities covered by the withdrawn or terminated registration statement and such registration statement shall have been declared effective and shall not have been withdrawn. If the Company shall give any notice of suspension, withdrawal or postponement of a registration statement, the Company shall, not later than five (5) Business Days after the Valid Business Reason that caused such suspension, withdrawal or postponement no longer exists (but, with respect to a suspension, withdrawal or postponement pursuant to clause (c)(iii) above, in no event later than forty-five (45) days after the date of the suspension, postponement or withdrawal), as applicable, permit use of such suspended registration statement or use its reasonable best efforts to effect the registration under the Securities Act of the Registrable Securities covered by the withdrawn or postponed registration statement in accordance with this Section 2.1 (unless the Initiating Holders or Shelf Underwriting Initiating Holders shall have withdrawn such request, in which case the Company shall not be considered to have effected a Demand Registration for the purposes of this Agreement and such request shall not count as a Demand Registration Request under this Agreement), and following such permission or such effectiveness such registration shall no longer be deemed to be suspended, withdrawn or postponed pursuant to clause (iv) of Section 2.1(c) above.

(d) No Demand Registration shall be deemed to have occurred for purposes of Section 2.1(b) (i) if the registration statement relating thereto (x) does not become effective, (y) is not maintained effective for a period of at least one hundred eighty (180) days after the effective date thereof or such shorter period during which all Registrable Securities included in such Registration Statement have actually been sold (provided, however, that such period shall be extended for a period of time equal to the period any Holder of Registrable Securities refrains from selling any securities included in such Registration Statement at the request of the Company or an underwriter of the Company), or (z) is subject to a stop order, injunction, or similar order or requirement of the SEC during such period, (ii) for each Initiating Holder, if less than seventy five percent (75%) of the Registrable Securities requested by such Initiating Holder to be included in such Demand Registration are not so included pursuant to Section 2.3, (iii) if the method of disposition is a firm commitment underwritten public offering and less than seventy five percent (75%) of the applicable Registrable Securities have not been sold pursuant thereto (excluding any Registrable Securities included for sale in the underwriters' overallotment option) or (iv) if the conditions to closing specified in any underwriting agreement, purchase agreement or similar agreement entered into in connection with the registration relating to such request are not satisfied (other than as a result of a default or breach thereunder by such Initiating Holder(s) or its Affiliates or are otherwise waived by such Initiating Holder(s)).

(e) Any Initiating Holder may withdraw or revoke a Demand Registration Request delivered by such Initiating Holder at any time prior to the effectiveness of such Demand Registration by giving written notice to the Company of such withdrawal or revocation and such Demand Registration shall have no further force or effect and such request shall not count as a Demand Registration Request under this Agreement.

2.2. Piggyback Registrations.

(a) If the Company proposes or is required to register any of its equity securities for its own account or for the account of any other shareholder under the Securities Act (other than pursuant to registrations on Form S-4 or Form S-8 or any similar successor forms thereto), the Company shall give written notice (the "Piggyback Notice") of its intention to do so to each of the Holders of record of Registrable Securities, at least five (5) Business Days prior to the filing of any registration statement under the Securities Act. Notwithstanding the foregoing, the Company may delay any Piggyback Notice until after filing a registration statement, so long as all recipients of such notice have the same amount of time to determine whether to participate in an offering as they would have had if such notice had not been so delayed. Upon the written request of any such Holder, made within five (5) days following the receipt of any such Piggyback Notice (which request shall specify the maximum number of Registrable Securities intended to be disposed of by such Holder and the intended method of distribution thereof), the Company shall, subject to Sections 2.2(c), 2.3 and 2.6 hereof, use its reasonable best efforts to cause all such Registrable Securities, the Holders of which have so requested the registration thereof, to be registered under the Securities Act with the securities which the Company at the time proposes to register to permit the sale or other disposition by the Holders (in accordance with the intended method of distribution thereof) of the Registrable Securities to be so registered, including, if necessary, by filing with the SEC a post-effective amendment or a supplement to the registration statement filed by the Company or the prospectus related thereto. There is no limitation on the number of such piggyback registrations which the Company is obligated to effect pursuant to the preceding sentence. No registration of Registrable Securities effected under this Section 2.2(a) shall relieve the Company of its

- (b) Other than in connection with a Demand Registration or a Shelf Underwriting, at any time after giving a Piggyback Notice and prior to the effective date of the registration statement filed in connection with such registration, if the Company shall determine for any reason not to register or to delay registration of such equity securities, the Company may, at its election, give written notice of such determination to all Holders of record of Registrable Securities and (x) in the case of a determination not to register, shall be relieved of its obligation to register any Registrable Securities in connection with such abandoned registration, without prejudice, however, to the rights of Holders under Section 2.1, and (y) in the case of a determination to delay such registration of its equity securities, shall be permitted to delay the registration of such Registrable Securities for the same period as the delay in registering such other equity securities.
- (c) Any Holder shall have the right to withdraw its request for inclusion of its Registrable Securities in any registration statement pursuant to this <u>Section 2.2</u> by giving written notice to the Company of its request to withdraw; <u>provided</u>, <u>however</u>, that such request must be made in writing prior to the earlier of the execution by such Holder of the underwriting agreement or the execution by such Holder of the custody agreement with respect to such registration or as otherwise required by the underwriters.

2.3. Allocation of Securities Included in Registration Statement.

- (a) If any requested registration or offering made pursuant to Section 2.1 (including a Shelf Underwriting) involves an underwritten offering and the Manager of such offering shall advise the Company in good faith that, in its view, the number of securities requested to be included in such underwritten offering by the Holders of Registrable Securities, the Company or any other Persons exercising contractual registration rights ("Additional Piggyback Rights") exceeds the largest number of securities (the "Section 2.3(a) Sale Number") that can be sold in an orderly manner in such underwritten offering within a price range acceptable to the Initiating Holders and the Majority Participating Holders, the Company shall include in such underwritten offering:
- (i) first, all Registrable Securities requested to be included in such underwritten offering by the Holders thereof (including pursuant to the exercise of piggyback rights pursuant to Section 2.2); provided, however, that if the number of such Registrable Securities exceeds the Section 2.3(a) Sale Number, the number of such Registrable Securities (not to exceed the Section 2.3(a) Sale Number) to be included in such underwritten offering shall be allocated on a pro rata basis among all Holders (including each Initiating Holder) requesting that Registrable Securities be included in such underwritten offering (including pursuant to the exercise of piggyback rights pursuant to Section 2.2), based on the number of Registrable Securities then owned by each such Holder requesting inclusion in relation to the aggregate number of Registrable Securities owned by all Holders requesting inclusion; and

- (ii) second, to the extent that the number of Registrable Securities to be included pursuant to clause (i) of this Section 2.3(a) is less than the Section 2.3(a) Sale Number, any securities that the Company proposes to register for its own account, up to the Section 2.3(a) Sale Number; and (iii) third, to the extent that the number of securities to be included pursuant to clauses (i) and (ii) of this Section 2.3(a) is less than the Section 2.3(a) Sale Number, the remaining securities to be included in such underwritten offering shall be allocated on a pro rata basis among all Persons other than Holders requesting that securities be included in such underwritten offering pursuant to the exercise of Additional Piggyback Rights ("Piggyback Shares"), based on the aggregate number of Piggyback Shares then owned by each Person requesting inclusion in relation to the aggregate number of Piggyback Shares owned by all Persons requesting inclusion, up to the Section 2.3(a) Sale Number.
- (b) If any registration or offering made pursuant to Section 2.2 involves an underwritten primary offering on behalf of the Company and the Manager shall advise the Company that, in its view, the number of securities requested to be included in such underwritten offering by the Holders of Registrable Securities, the Company or any other Persons exercising Additional Piggyback Rights exceeds the largest number of securities (the "Section 2.3(b) Sale Number") that can be sold in an orderly manner in such underwritten offering within a price range acceptable to the Company, the Company shall include in such underwritten offering:
 - (i) first, all equity securities that the Company proposes to register for its own account; and
- (ii) second, to the extent that the number of securities to be included pursuant to clause (i) of this section 2.3(b) is less than the Section 2.3(b) Sale Number, the remaining Registrable Securities to be included in such underwritten offering shall be allocated on a pro rata basis among all Holders requesting that Registrable Securities be included in such underwritten offering pursuant to the exercise of piggyback rights pursuant to Section 2.2(a), based on the aggregate number of Registrable Securities then owned by each such Holder requesting inclusion in relation to the aggregate number of Registrable Securities owned by all Holders requesting inclusion, up to the Section 2.3(b) Sale Number; and (iii) third, to the extent that the number of securities to be included pursuant to clauses (i) and (ii) of this Section 2.3(b) is less than the Section 2.3(b) Sale Number, the remaining securities to be included in such underwritten offering shall be allocated on a pro rata basis among all Persons requesting that Piggyback Shares be included in such underwritten offering pursuant to the exercise of Additional Piggyback Rights, based on the aggregate number of Piggyback Shares then owned by each Person requesting inclusion in relation to the aggregate number of Piggyback Shares owned by all Persons requesting inclusion, up to the Section 2.3(b) Sale Number.
- (c) If any registration pursuant to Section 2.2 involves an underwritten offering that was initially requested by any Person(s) (other than a Holder) to whom the Company has granted registration rights which are not inconsistent with the rights granted in, and do not otherwise conflict with the terms of, this Agreement and the Manager shall advise the Company that, in its view, the number of securities requested to be included in such underwritten offering exceeds the largest number of securities (the "Section 2.3(c) Sale Number") that can be sold in an orderly manner in such underwritten offering within a price range acceptable to the Company, the Company shall include in such underwritten offering:

- (i) first, the shares requested to be included in such underwritten offering shall be allocated on a pro rata basis among such Person(s) requesting the registration and all Holders requesting that Registrable Securities be included in such underwritten offering pursuant to the exercise of piggyback rights pursuant to <u>Section 2.2(a)</u>, based on the aggregate number of securities or Registrable Securities, as applicable, then owned by each of the foregoing requesting inclusion in relation to the aggregate number of securities or Registrable Securities, as applicable, owned by all such Persons and Holders requesting inclusion, up to the Section 2.3(c) Sale Number; and
- (ii) second, to the extent that the number of securities to be included pursuant to clause (i) of this Section 2.3(c) is less than the Section 2.3(c) Sale Number, the remaining securities to be included in such underwritten offering shall be allocated on a pro rata basis among all Persons requesting that Piggyback Shares be included in such underwritten offering pursuant to the exercise of Additional Piggyback Rights, based on the aggregate number of Piggyback Shares then owned by each Person requesting inclusion in relation to the aggregate number of Piggyback Shares owned by all Persons requesting inclusion, up to the Section 2.3(c) Sale Number; and (iii) third, to the extent that the number of securities to be included pursuant to clauses (i) and (ii) of this Section 2.3(c) is less than the Section 2.3(c) Sale Number, any equity securities that the Company proposes to register for its own account, up to the Section 2.3(c) Sale Number.
- (d) If, as a result of the proration provisions set forth inclauses (a), (b) or (c) of this Section 2.3, any Holder shall not be entitled to include all Registrable Securities in an underwritten offering that such Holder has requested be included, such Holder may elect to withdraw such Holder's request to include Registrable Securities in the registration to which such underwritten offering relates or may reduce the number requested to be included; provided, however, that (x) such request must be made in writing prior to the earlier of such Holder's execution of the underwriting agreement or such Holder's execution of the custody agreement with respect to such registration and (y) such withdrawal or reduction shall be irrevocable and, after making such withdrawal or reduction, such Holder shall no longer have any right to include Registrable Securities in the registration as to which such withdrawal or reduction was made to the extent of the Registrable Securities so withdrawn or reduced.
- 2.4. <u>Registration Procedures</u>. If and whenever the Company is required by the provisions of this Agreement to effect or cause the registration of and/or participate in any offering or sale of any Registrable Securities under the Securities Act as provided in this Agreement (or use reasonable best efforts to accomplish the same), the Company shall, as expeditiously as possible:
- (a) prepare and file all filings with the SEC and FINRA as soon as practicable required for the consummation of the offering, including preparing and filing with the SEC a registration statement on an appropriate registration form of the SEC for the disposition of such Registrable Securities in accordance with the intended method of disposition thereof, which registration form (i) shall be selected by the Company (except as provided for in a Demand Registration Request) and (ii) shall, in the case of a shelf registration, be available for the sale of the Registrable Securities by the selling Holders thereof and such registration statement shall comply as to form in all material respects with the requirements of the applicable registration form

and include all financial statements required by the SEC to be filed therewith, and the Company shall use its reasonable best efforts to cause such registration statement to become effective and remain continuously effective for such period as required by this Agreement (provided, however, that as far in advance as reasonably practicable before filing a registration statement or prospectus or any amendments or supplements thereto, or comparable statements under securities or state "blue sky" laws of any jurisdiction, or any free writing prospectus related thereto, the Company will furnish to the Holders participating in the planned offering and to the Manager, if any, copies of all such documents proposed to be filed (including all exhibits thereto), which documents will be subject to their reasonable review and reasonable comment and the Company shall not file any registration statement or amendment thereto, any prospectus or supplement thereto or any free writing prospectus related thereto to which the Initiating Holders, the Majority Participating Holders or the underwriters, if any, shall reasonably object); provided, however, that, notwithstanding the foregoing, in no event shall the Company be required to file any document with the SEC which in the view of the Company or its counsel contains an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make any statement therein not misleading;

- (b) (i) prepare and file with the SEC such amendments and supplements to such registration statement and the prospectus used in connection therewith and such free writing prospectuses and Exchange Act reports as may be necessary to keep such registration statement continuously effective for such period as required by this Agreement and to comply with the provisions of the Securities Act with respect to the sale or other disposition of all Registrable Securities covered by such registration statement, and any prospectus so supplemented to be filed pursuant to Rule 424 under the Securities Act, in accordance with the intended methods of disposition by the seller or sellers thereof set forth in such registration statement and (ii) provide notice to such sellers of Registrable Securities and the Manager, if any, of the Company's reasonable determination that a post-effective amendment to a registration statement would be appropriate;
- (c) furnish, without charge, to each Participating Holder and each underwriter, if any, of the securities covered by such registration statement such number of copies of such registration statement, each amendment and supplement thereto (in each case including all exhibits), the prospectus included in such registration statement (including each preliminary prospectus and any summary prospectus) and any other prospectus filed under Rule 424 under the Securities Act, each free writing prospectus utilized in connection therewith, in each case, in conformity with the requirements of the Securities Act, and other documents, as such seller and underwriter may reasonably request in order to facilitate the public sale or other disposition of the Registrable Securities owned by such seller (the Company hereby consenting to the use in accordance with all applicable laws of each such registration statement (or amendment or post-effective amendment thereto) and each such prospectus (or preliminary prospectus or supplement thereto) or free writing prospectus by each such Participating Holder and the underwriters, if any, in connection with the offering and sale of the Registrable Securities covered by such registration statement or prospectus);

(d) use its reasonable best efforts to register or qualify the Registrable Securities covered by such registration statement under such other securities or state "blue sky" laws of such jurisdictions as any sellers of Registrable Securities or any managing underwriter, if any, shall reasonably request in writing, and do any and all other acts and things which may be reasonably necessary or advisable to enable such sellers or underwriter, if any, to consummate the disposition of the Registrable Securities in such jurisdictions (including keeping such registration or qualification in effect for so long as such registration statement remains in effect), except that in no event shall the Company be required to qualify to do business as a foreign corporation in any jurisdiction where it would not, but for the requirements of this paragraph (d), be required to be so qualified, to subject itself to taxation in any such jurisdiction or to consent to general service of process in any such jurisdiction;

(e) promptly notify each Participating Holder and each managing underwriter, if any: (i) when the registration statement, anypre-effective amendment, the prospectus or any prospectus supplement related thereto, any post-effective amendment to the registration statement or any free writing prospectus has been filed with the SEC and, with respect to the registration statement or any post-effective amendment, when the same has become effective; (ii) of any request by the SEC or state securities authority for amendments or supplements to the registration statement or the prospectus related thereto or for additional information; (iii) of the issuance by the SEC of any stop order suspending the effectiveness of the registration statement or the initiation of any proceedings for that purpose; (iv) of the receipt by the Company of any notification with respect to the suspension of the qualification of any Registrable Securities for sale under the securities or state "blue sky" laws of any jurisdiction or the initiation of any proceeding for such purpose; (v) of the existence of any fact of which the Company becomes aware which results in the registration statement or any amendment thereto, the prospectus related thereto or any supplement thereto, any document incorporated therein by reference, any free writing prospectus or the information conveyed at the time of sale to any purchaser containing an untrue statement of a material fact or omitting to state a material fact required to be stated therein or necessary to make any statement therein not misleading; and (vi) if at any time the representations and warranties contemplated by any underwriting agreement, securities sale agreement, or other similar agreement, relating to the offering shall cease to be true and correct in all material respects (unless otherwise qualified by materiality in which case such representations and warranties shall cease to be true and correct in all respects); and, if the notification relates to an event described in clause (v), unless the Company has declared that a Postponement Period exists, the Company shall promptly prepare and furnish to each such seller and each underwriter, if any, a reasonable number of copies of a prospectus supplemented or amended so that, as thereafter delivered to the purchasers of such Registrable Securities, such prospectus shall not include an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein in the light of the circumstances under which they were made not misleading;

(f) comply (and continue to comply) with all applicable rules and regulations of the SEC (including maintaining disclosure controls and procedures (as defined in Exchange Act Rule 13a-15(e)) and internal control over financial reporting (as defined in Exchange Act Rule 13a-15(f)) in accordance with the Exchange Act), and make generally available to its security holders (including by way of filings with the SEC), as soon as reasonably practicable after the effective date of the registration statement (and in any event within forty-five (45) days, or ninety (90) days if it is a fiscal year, after the end of such twelve month period described hereafter), an earnings statement (which need not be audited) covering the period of at least twelve (12) consecutive months beginning with the first day of the Company's first calendar quarter after the effective date of the registration statement, which earnings statement shall satisfy the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder;

- (g) (i) (A) use its reasonable best efforts to cause all such Registrable Securities covered by such registration statement to be listed on the principal securities exchange on which similar securities issued by the Company are then listed, if the listing of such Registrable Securities is then permitted under the rules of such exchange, or (B) if no similar securities are then so listed, use its reasonable best efforts to either cause all such Registrable Securities to be listed on a national securities exchange or to secure designation of all such Registrable Securities as a New York Stock Exchange "national market system security" within the meaning of Rule 11Aa2-1 of the Exchange Act or, failing that, secure New York Stock Exchange authorization for such shares and, without limiting the generality of the foregoing, take all actions that may be required by the Company as the issuer of such Registrable Securities in order to facilitate the managing underwriter's arranging for the registration of at least two market makers as such with respect to such shares with FINRA, and (ii) comply (and continue to comply) with the requirements of any self-regulatory organization applicable to the Company, including all corporate governance requirements;
- (h) cause its senior management, officers and employees to participate in, and to otherwise facilitate and cooperate with the preparation of the registration statement and prospectus and any amendments or supplements thereto (including participating in meetings, drafting sessions, due diligence sessions and rating agency presentations) taking into account the Company's reasonable business needs;
- (i) provide and cause to be maintained a transfer agent and registrar for all such Registrable Securities covered by such registration statement not later than the effective date of such registration statement and, in the case of any secondary equity offering, provide and enter into any reasonable agreements with a custodian for the Registrable Securities;
- (j) enter into such customary agreements (including, if applicable, an underwriting agreement) and take such other actions as the Initiating Holder or the Majority Participating Holders or the underwriters shall reasonably request in order to expedite or facilitate the disposition of such Registrable Securities (it being understood that the Holders of the Registrable Securities which are to be distributed by any underwriters shall be parties to any such underwriting agreement and may, at their option, require that the Company make for the benefit of such Holders the representations, warranties and covenants of the Company which are being made to and for the benefit of such underwriters);
- (k) use its reasonable best efforts (i) to obtain opinions from the Company's counsel, including local and/or regulatory counsel, and a "comfort" letter and updates thereof from the independent public accountants who have certified the financial statements of the Company (and/or any other financial statements) included or incorporated by reference in such registration statement, in each case, in customary form and covering such matters as are customarily covered by such opinions and "comfort" letters (including, in the case of such "comfort" letter, events subsequent to the date of such financial statements) delivered to underwriters in underwritten public offerings, which opinions and letters shall be dated the dates such opinions and "comfort" letters are customarily dated and otherwise reasonably satisfactory to the underwriters, if any, and (ii) furnish to each Participating Holder and to each underwriter, if any, a copy of such opinions and letters addressed to such underwriter;

(1) deliver promptly to counsel for the Majority Participating Holders and to each managing underwriter, if any, copies of all correspondence between the SEC and the Company, its counsel or auditors and all memoranda relating to discussions with the SEC or its staff with respect to the registration statement, and, upon receipt of such confidentiality agreements as the Company may reasonably request, make reasonably available for inspection by counsel for the Majority Participating Holders, by counsel for any underwriter participating in any disposition to be effected pursuant to such registration statement and by any attorney, accountant or other agent retained by the Majority Participating Holders or any such underwriter, during regular business hours, all pertinent financial and other records, pertinent corporate documents and properties of the Company, and cause all of the Company's officers, directors and employees to supply all information reasonably requested by any such counsel for the Majority Participating Holders, counsel for an underwriter, attorney, accountant or agent in connection with such registration statement;

(m) use its reasonable best efforts to prevent the issuance or obtain the prompt withdrawal of any order suspending the effectiveness of the registration statement, or the prompt lifting of any suspension of the qualification of any of the Registrable Securities for sale in any jurisdiction, in each case, as promptly as reasonably practicable;

- (n) provide a CUSIP number for all Registrable Securities, not later than the effective date of the registration statement;
- (o) use its reasonable best efforts to make available its senior management for participation in "road shows" and other marketing efforts and otherwise provide reasonable assistance to the underwriters (taking into account the Company's reasonable business needs and the requirements of the marketing process) in the marketing of Registrable Securities in any underwritten offering;
- (p) promptly prior to the filing of any document which is to be incorporated by reference into the registration statement or the prospectus (after the initial filing or confidential submission of such registration statement), and prior to the filing or use of any free writing prospectus, provide copies of such document to counsel for the Majority Participating Holders and to each managing underwriter, if any, and make the Company's representatives reasonably available for discussion of such document and make such changes in such document concerning the information regarding the Participating Holders contained therein prior to the filing thereof as counsel for the Majority Participating Holders or underwriters may reasonably request (provided, however, that, notwithstanding the foregoing, in no event shall the Company be required to file or confidentially submit any document with the SEC which in the view of the Company or its counsel contains an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to make any statement therein not misleading);

- (q) furnish to counsel for the Majority Participating Holders and to each managing underwriter, without charge, upon request, at least one conformed copy of the registration statement and any post-effective amendments or supplements thereto, including financial statements and schedules, all documents incorporated therein by reference, the prospectus contained in such registration statement (including each preliminary prospectus and any summary prospectus), any other prospectus and prospectus supplement filed under Rule 424 under the Securities Act and all exhibits (including those incorporated by reference) and any free writing prospectus utilized in connection therewith;
- (r) cooperate with the Participating Holders and the managing underwriter, if any, to facilitate the timely preparation and delivery of certificates not bearing any restrictive legends representing the Registrable Securities to be sold, and cause such Registrable Securities to be issued in such denominations and registered in such names in accordance with the underwriting agreement at least two (2) Business Days prior to any sale of Registrable Securities to the underwriters or, if not an underwritten offering, in accordance with the instructions of the Participating Holders at least two (2) Business Days prior to any sale of Registrable Securities and instruct any transfer agent and registrar of Registrable Securities to release any stop transfer orders in respect thereof (and, in the case of Registrable Securities registered on a Shelf Registration Statement, at the request of any Holder, prepare and deliver certificates representing such Registrable Securities not bearing any restrictive legends and deliver or cause to be delivered an opinion or instructions to the transfer agent in order to allow such Registrable Securities to be sold from time to time);
- (s) include in any prospectus or prospectus supplement if requested by any managing underwriter updated financial or business information for the Company's most recent period or current quarterly period (including estimated results or ranges of results) if required for purposes of marketing the offering in the view of the managing underwriter;
- (t) take no direct or indirect action prohibited by Regulation M under the Exchange Act; provided, however, that to the extent that any prohibition is applicable to the Company, the Company will use its reasonable best efforts to make any such prohibition inapplicable;
- (u) use its reasonable best efforts to cause the Registrable Securities covered by the applicable registration statement to be registered with or approved by such other governmental agencies or authorities as may be necessary to enable the Participating Holders or the underwriters, if any, to consummate the disposition of such Registrable Securities;
- (v) take all such other commercially reasonable actions as are necessary or advisable in order to expedite or facilitate the disposition of such Registrable Securities;
- (w) take all reasonable action to ensure that any free writing prospectus utilized in connection with any registration covered by Section 2.1 or 2.2 complies in all material respects with the Securities Act, is filed in accordance with the Securities Act to the extent required thereby, is retained in accordance with the Securities Act to the extent required thereby and, when taken together with the related prospectus, prospectus supplement and related documents, will not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading;

(x) in connection with any underwritten offering, if at any time the information conveyed to a purchaser at the time of sale includes any untrue statement of a material fact or omits to state any material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading, promptly file with the SEC such amendments or supplements to such information as may be necessary so that the statements as so amended or supplemented will not, in the light of the circumstances, be misleading;

(y) to the extent required by the rules and regulations of FINRA, retain a Qualified Independent Underwriter acceptable to the managing underwriter; and

(z) use reasonable best efforts, in good faith,to cooperate with the managing underwriters, Participating Holders, any indemnitee of the Company and their respective counsel in connection with the preparation and filing of any applications, notices, registrations and responses to requests for additional information with FINRA, Nasdaq, or any other national securities exchange on which the shares of Class A Common Stock are listed.

To the extent the Company is a WKSI at the time any Demand Registration Request is submitted to the Company, the Company shall file an automatic shelf registration statement (as defined in Rule 405 under the Securities Act) (an "automatic shelf registration statement") on Form S-3 which covers those Registrable Securities which are requested to be registered. The Company shall not take any action that would result in it not remaining a WKSI or would result in it becoming an ineligible issuer (as defined in Rule 405 under the Securities Act) during the period during which such automatic shelf registration statement is required to remain effective. If the Company does not pay the filing fee covering the Registrable Securities at the time the automatic shelf registration statement is filed, the Company agrees to pay such fee at such time or times as the Registrable Securities are to be sold in compliance with the SEC rules. If the automatic shelf registration statement has been outstanding for at least three (3) years, at or prior to the end of the third year the Company shall refile a new automatic shelf registration statement covering the Registrable Securities. If at any time when the Company is required to re-evaluate its WKSI status the Company determines that it is not a WKSI, the Company shall use its reasonable best efforts to refile the shelf registration statement on Form S-3 and, if such form is not available, Form S-1 and keep such registration statement effective during the period which such registration statement is required to be kept effective.

If the Company files any shelf registration statement for the benefit of the holders of any of its securities other than the Holders, and the Holders do not request that their Registrable Securities be included in such Shelf Registration Statement, the Company agrees that it shall include in such registration statement such disclosures as may be required by Rule 430B under the Securities Act (referring to the unnamed selling security holders in a generic manner by identifying the initial offering of the securities to the Holders) in order to ensure that the Holders may be added to such shelf registration statement at a later time through the filing of a prospectus supplement rather than a post-effective amendment.

The Company may require as a condition precedent to the Company's obligations under this Section 2.4 that each Participating Holder as to which any registration is being effected (i) furnish the Company such information regarding such seller and the distribution of such securities as the Company may from time to time reasonably request (including as required under state securities laws), provided that such information is necessary for the Company to consummate such registration and shall be used only in connection with such registration and (ii) provide any underwriters participating in the distribution of such securities such information as the underwriters may request and execute and deliver any agreements, certificates or other documents as the underwriters may request.

Each Holder of Registrable Securities agrees that upon receipt of any notice from the Company of the happening of any event of the kind described in clause (v) of paragraph (e) of this Section 2.4, such Holder will discontinue such Holder's disposition of Registrable Securities pursuant to the registration statement covering such Registrable Securities until such Holder's receipt of the copies of the supplemented or amended prospectus contemplated by paragraph (e) of this Section 2.4 and, if so directed by the Company, will deliver to the Company (at the Company's expense) all copies, other than permanent file copies, then in such Holder's possession of the prospectus covering such Registrable Securities that was in effect at the time of receipt of such notice. In the event the Company shall give any such notice, the applicable period mentioned in paragraph (b) of this Section 2.4 shall be extended by the number of days during such period from and including the date of the giving of such notice to and including the date when each Participating Holder covered by such registration statement shall have received the copies of the supplemented or amended prospectus contemplated by paragraph (e) of this Section 2.4.

The Company agrees not to file or make any amendment to any registration statement with respect to any Registrable Securities, or any amendment of or supplement to the prospectus, or any free writing prospectus, which amendment refers to any Holder covered thereby by name, or otherwise identifies such Holder, without the consent of such Holder, such consent not to be unreasonably withheld or delayed, unless such disclosure is required by law, in which case the Company shall provide written notice to such Holders no less than five (5) Business Days prior to the filing.

2.5. Registration Expenses.

- (a) The Company shall pay all Expenses with respect to any registration or offering of Registrable Securities pursuant to Section 2, whether or not a registration statement becomes effective or the offering is consummated.
- (b) Notwithstanding the foregoing, (x) the provisions of this Section 2.5 shall be deemed amended to the extent necessary to cause these expense provisions to comply with state "blue sky" laws of each state in which the offering is made and (y) in connection with any underwritten offering hereunder, each Participating Holder shall pay all underwriting discounts and commissions and any transfer taxes, if any, attributable to the sale of such Registrable Securities, pro rata with respect to payments of discounts and commissions in accordance with the number of shares sold in the offering by such Participating Holder.

2.6. Certain Limitations on Registration Rights. In the case of any registration under Section 2.1 involving an underwritten offering, or, in the case of a registration under Section 2.2, if the Company has determined to enter into an underwriting agreement in connection therewith, all securities to be included in such underwritten offering shall be subject to such underwriting agreement and no Person may participate in such underwritten offering unless such Person (i) agrees to sell such Person's securities on the basis provided therein and completes and executes all reasonable questionnaires, and other customary documents (including custody agreements, powers of attorney, indemnities, lock-up agreements) which must be executed in connection therewith; provided, however, that all such documents shall be consistent with the provisions hereof and (ii) provides such other information to the Company or the underwriter as may be necessary to register such Person's securities.

2.7. Limitations on Sale or Distribution of Other Securities.

- (a) Each Holder that is a director or officer of the Company agrees, to the extent requested by the Manager of any underwritten public offering pursuant to a registration or offering effected pursuant to Section 2.1 (including any Shelf Underwriting pursuant to Section 2.1) or Section 2.2 (including any offering effected by the Company for its own account), not to sell, transfer or otherwise dispose of, including any sale pursuant to Rule 144, any Class A Common Stock or Class A Common Stock Equivalents (other than as part of such underwritten public offering) during the time period reasonably requested by the Manager, not to exceed the period from seven days prior to the pricing date of such offering until ninety (90) days after the pricing date of such offering or such shorter period as the Manager, the Company or any executive officer or director of the Company shall agree to.
- (b) The Company hereby agrees that, in connection with an offering pursuant to Section 2.1 (including any Shelf Underwriting pursuant to Section 2.1(e)) or 2.2, the Company shall not sell, transfer, or otherwise dispose of, any Class A Common Stock or Class A Common Stock Equivalent (other than as part of such underwritten public offering, a registration on Form S- 4 or Form S-8 or any successor or similar form which is (x) then in effect or (y) shall become effective upon the conversion, exchange or exercise of any then outstanding Class A Common Stock Equivalent), until a period from seven days prior to the pricing date of such offering until ninety (90) days after the pricing date of such offering or such shorter period as the Manager, the Company or any executive officer or director of the Company shall agree to and the Company shall so provide in any registration rights agreements hereafter entered into with respect to any of its securities.
- 2.8. No Required Sale. Nothing in this Agreement shall be deemed to create an independent obligation on the part of any Holder to sell any Registrable Securities pursuant to any effective registration statement. A Holder is not required to include any of its Registrable Securities in any registration statement, is not required to sell any of its Registrable Securities which are included in any effective registration statement, and may sell any of its Registrable Securities in any manner in compliance with applicable law (subject to applicable lock-up restrictions) even if such shares are already included on an effective registration statement.

2.9. Indemnification.

(a) In the event of any registration or offer and sale of any securities of the Company under the Securities Act pursuant to this Section 2, the Company will (without limitation as to time), and hereby agrees to, and hereby does, indemnify and hold harmless, to the fullest extent permitted by law, each Participating Holder, its directors, officers, employees, stockholders, members, general and limited partners, agents, affiliates, representatives, successors and assigns (and the directors, officers, employees, stockholders, members, general and limited partners, agents, affiliates, representatives, successors and assigns thereof), each other Person who participates as a seller (and its directors, officers, employees, stockholders, members, general and limited partners, agents, affiliates, representatives, successors and assigns), underwriter or Qualified Independent Underwriter, if any, in the offering or sale of such securities, each officer, director, employee, stockholder, managing director, agent, affiliate, representative, successor, assign or partner of such underwriter or Qualified Independent Underwriter, and each other Person, if any, who controls (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) such seller or any such underwriter or Qualified Independent Underwriter and each director, officer, employee, stockholder, managing director, agent, affiliate, representative, successor, assign or partner of such controlling Person, from and against any and all losses, claims, damages or liabilities, joint or several, actions or proceedings (whether commenced or threatened) and expenses (including reasonable fees of counsel and any amounts paid in any settlement effected with the Company's consent, which consent shall not be unreasonably withheld or delayed) to which each such indemnified party may become subject under the Securities Act or otherwise in respect thereof (collectively, "Claims"), insofar as such Claims arise out of, are based upon, relate to or are in connection with (i) any untrue statement or alleged untrue statement of a material fact contained in any registration statement under which such securities were registered under the Securities Act or the omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading, (ii) any untrue statement or alleged untrue statement of a material fact contained in any preliminary, final or summary prospectus or any amendment or supplement thereto, together with the documents incorporated by reference therein, or any free writing prospectus utilized in connection therewith, or the omission or alleged omission to state therein a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading, or (iii) any untrue statement or alleged untrue statement of a material fact in the information conveyed by the Company or any underwriter to any purchaser at the time of the sale to such purchaser, or the omission or alleged omission to state therein a material fact required to be stated therein, or (iv) any violation by the Company of any federal, state or common law rule or regulation applicable to the Company and relating to any action required of or inaction by the Company in connection with any such offering of Registrable Securities, and the Company will reimburse any such indemnified party for any legal or other expenses reasonably incurred by such indemnified party in connection with investigating or defending any such Claim as such expenses are incurred; provided, however, that the Company shall not be liable to any such indemnified party in any such case to the extent such Claim arises out of or is based upon any untrue statement or alleged untrue statement of a material fact or omission or alleged omission of a material fact made in such registration statement or amendment thereof or supplement thereto or in any such prospectus or any preliminary, final or summary prospectus or free writing prospectus in reliance upon and in conformity with written information furnished to the Company by or on behalf of such indemnified party specifically for use therein. Such indemnity and reimbursement of expenses shall remain in full force and effect regardless of any investigation made by or on behalf of such indemnified party and shall survive the transfer of such securities by such seller.

(b) Each Participating Holder (and, if the Company requires as a condition to including any Registrable Securities in any registration statement filed in accordance with Section 2.1 or 2.2, any underwriter and Qualified Independent Underwriter, if any) shall, severally and not jointly, indemnify and hold harmless (in the same manner and to the same extent as set forth in paragraph (a) of this Section 2.9) to the extent permitted by law the Company, its officers and its directors, each Person controlling the Company within the meaning of the Securities Act and all other prospective sellers and their directors, officers, stockholders, fiduciaries, managing directors, agents, affiliates, representatives, successors, assigns or general and limited partners and respective controlling Persons with respect to any untrue statement or alleged untrue statement of any material fact in, or omission or alleged omission of any material fact from, such registration statement, any preliminary, final or summary prospectus contained therein, or any amendment or supplement thereto, or any free writing prospectus utilized in connection therewith, if such statement or alleged statement or omission or alleged omission was made in reliance upon and in conformity with written information furnished to the Company or its representatives by or on behalf of such Participating Holder or underwriter or Qualified Independent Underwriter, if any, specifically for use therein, and each such Participating Holder, underwriter or Qualified Independent Underwriter, if any, shall reimburse such indemnified party for any legal or other expenses reasonably incurred by such indemnified party in connection with investigating or defending any such Claim as such expenses are incurred; provided, however, that the aggregate amount which any such Participating Holder shall be required to pay pursuant to this Section 2.9 (including pursuant to indemnity, contribution or otherwise) shall in no case be greater than the amount of the net proceeds received by such Participating Holder upon the sale of the Registrable Securities pursuant to the registration statement giving rise to such Claim; provided, further, that such Participating Holder shall not be liable in any such case to the extent that prior to the filing or confidential submission of any such registration statement or prospectus or amendment thereof or supplement thereto, or any free writing prospectus utilized in connection therewith, such Participating Holder has furnished in writing to the Company information expressly for use in such registration statement or prospectus or any amendment thereof or supplement thereto or free writing prospectus which corrected or made not misleading information previously furnished to the Company. The Company and each Participating Holder hereby acknowledge and agree that, unless otherwise expressly agreed to in writing by such Participating Holders to the contrary, for all purposes of this Agreement, the only information furnished or to be furnished to the Company for use in any such registration statement, preliminary, final or summary prospectus or amendment or supplement thereto, or any free writing prospectus, are statements specifically relating to (i) the beneficial ownership of shares of Common Stock by such Participating Holder and its Affiliates as disclosed in the section of such document entitled "Selling Stockholders" or "Principal and Selling Stockholders" and (ii) the name and address of such Participating Holder. If any additional information about such Holder or the plan of distribution (other than for an underwritten offering) is required by law to be disclosed in any such document, then such Holder shall not unreasonably withhold its agreement referred to in the immediately preceding sentence. Such indemnity and reimbursement of expenses shall remain in full force and effect regardless of any investigation made by or on behalf of such indemnified party and shall survive the transfer of such securities by such Holder.

(c) Indemnification similar to that specified in the preceding paragraphs (a) and (b) of this Section 2.9 (with appropriate modifications) shall be given by the Company and each Participating Holder with respect to any required registration or other qualification of securities under any applicable securities and state "blue sky" laws.

(d) Any Person entitled to indemnification under this Agreement shall notify promptly the indemnifying party in writing of the commencement of any action or proceeding with respect to which a claim for indemnification may be made pursuant to this Section 2.9, but the failure of any indemnified party to provide such notice shall not relieve the indemnifying party of its obligations under the preceding paragraphs of this Section 2.9. except to the extent the indemnifying party is materially and actually prejudiced thereby and shall not relieve the indemnifying party from any liability which it may have to any indemnified party otherwise than under this Section 2.9. In case any action or proceeding is brought against an indemnified party and such indemnified party shall have notified the indemnifying party of the commencement thereof (as required above), the indemnifying party shall be entitled to participate therein and, unless in the reasonable opinion of outside counsel to the indemnified party a conflict of interest between such indemnified and indemnifying parties exists in respect of such Claim, to assume the defense thereof jointly with any other indemnifying party similarly notified, to the extent that it chooses, with counsel reasonably satisfactory to such indemnified party, and after notice from the indemnifying party to such indemnified party that it so chooses, the indemnifying party shall not be liable to such indemnified party for any legal or other expenses subsequently incurred by such indemnified party in connection with the defense thereof other than reasonable costs of investigation; provided, however, that (i) if the indemnifying party fails to take reasonable steps necessary to defend diligently the action or proceeding within twenty (20) days after receiving notice from such indemnified party that the indemnified party believes it has failed to do so; or (ii) if such indemnified party who is a defendant in any action or proceeding which is also brought against the indemnifying party reasonably shall have concluded that there may be one or more legal or equitable defenses available to such indemnified party which are not available to the indemnifying party or which may conflict with or be different from those available to another indemnified party with respect to such Claim; or (iii) if representation of both parties by the same counsel is otherwise inappropriate under applicable standards of professional conduct, then, in any such case, the indemnified party shall have the right to assume or continue its own defense as set forth above (but with no more than one firm of counsel for all indemnified parties in each jurisdiction, except to the extent any indemnified party or parties reasonably shall have made a conclusion described in clause (ii) or (iii) above) and the indemnifying party shall be liable for any expenses therefor. No indemnifying party shall be liable for any settlement of any proceeding effected without its written consent (which consent shall not be unreasonably withheld or delayed), but if settled with such consent or if there be a final judgment for the plaintiff, such indemnifying party agrees to indemnify each indemnified party from and against any loss, claim, damage, liability or expense by reason of such settlement or judgment. No indemnifying party shall, without the written consent of the indemnified party, effect the settlement or compromise of, or consent to the entry of any judgment with respect to, any pending or threatened action or claim in respect of which indemnification or contribution may be sought hereunder (whether or not the indemnified party is an actual or potential party to such action or claim) unless such settlement, compromise or judgment (A) includes an unconditional release of the indemnified party from all liability arising out of such action or claim and (B) does not include a statement as to or an admission of fault or culpability, by or on behalf of any indemnified party.

- (e) If for any reason the foregoing indemnity is unavailable, unenforceable or is insufficient to hold harmless an indemnified party under Sections 2.9(a), (b) or (c), then each applicable indemnifying party shall contribute to the amount paid or payable to such indemnified party as a result of any Claim in such proportion as is appropriate to reflect the relative fault of the indemnifying party, on the one hand, and the indemnified party, on the other hand, with respect to such Claim. The relative fault shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the indemnifying party or the indemnified party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such untrue statement or omission. If, however, the allocation provided in the second preceding sentence is not permitted by applicable law, then each indemnifying party shall contribute to the amount paid or payable by such indemnified party in such proportion as is appropriate to reflect not only such relative faults but also the relative benefits of the indemnifying party and the indemnified party as well as any other relevant equitable considerations. The parties hereto agree that it would not be just and equitable if any contribution pursuant to this Section 2.9(e) were to be determined by pro rata allocation or by any other method of allocation which does not take account of the equitable considerations referred to in the preceding sentences of this Section 2.9(e). The amount paid or payable in respect of any Claim shall be deemed to include any legal or other expenses reasonably incurred by such indemnified party in connection with investigating or defending any such Claim. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation. Notwithstanding anything in this Section 2.9(e) to the contrary, no indemnifying party (other than the Company) shall be required pursuant to this Section 2.9(e) to contribute any amount greater than the amount of the net proceeds received by such indemnifying party from the sale of Registrable Securities pursuant to the registration statement giving rise to such Claim, less the amount of any indemnification payment made by such indemnifying party pursuant to Sections 2.9(b) and (c). In addition, no Holder of Registrable Securities or any Affiliate thereof shall be required to pay any amount under this Section 2.9(e) unless such Person or entity would have been required to pay an amount pursuant to Section 2.9(b) if it had been applicable in accordance with its terms.
- (f) The indemnity and contribution agreements contained herein shall be in addition to any other rights to indemnification or contribution which any indemnified party may have pursuant to law or contract and shall remain operative and in full force and effect regardless of any investigation made or omitted by or on behalf of any indemnified party and shall survive the transfer of the Registrable Securities by any such party.
- (g) The indemnification and contribution required by this <u>Section 2.9</u> shall be made by periodic payments of the amount thereof during the course of the investigation or defense, as and when bills are received or expense, loss, damage or liability is incurred.
- 2.10. No Inconsistent Agreements. The Company shall not hereafter enter into any agreement with respect to its securities that is inconsistent in any material respects with the rights granted to the Holders in this Agreement.

Section 3. Underwritten Offerings.

- 3.1. Requested Underwritten Offerings. If requested by the underwriters for any underwritten offering pursuant to a registration requested under Section 2.1, the Company shall enter into a customary underwriting agreement with the underwriters. Such underwriting agreement shall (i) be satisfactory in form and substance to the Initiating Holders and the Majority Participating Holders, (ii) contain terms not inconsistent with the provisions of this Agreement and (iii) contain such representations and warranties by, and such other agreements on the part of, the Company and such other terms as are generally prevailing in agreements of that type, including indemnities and contribution agreements on substantially the same terms as those contained herein or as otherwise customary for the lead underwriter. Every Participating Holder shall be a party to such underwriting agreement. Each Participating Holder shall not be required to make any representations or warranties to or agreements with the Company or the underwriters other than customary representations of a selling shareholder, including representations, warranties or agreements regarding its ownership of and title to the Registrable Securities, any written information specifically provided by such Participating Holder for inclusion in the registration statement and its intended method of distribution; and any liability of such Participating Holder to any underwriter or other Person under such underwriting agreement for indemnity, contribution or otherwise shall in no case be greater than the amount of the net proceeds received by such Participating Holder upon the sale of Registrable Securities pursuant to such registration statement and in no event shall relate to anything other than information about such Holder specifically provided by such Holder for use in the registration statement and prospectus.
- 3.2. Piggyback Underwritten Offerings. In the case of a registration pursuant to Section 2.2, if the Company shall have determined to enter into an underwriting agreement in connection therewith, all of the Participating Holders' Registrable Securities to be included in such registration shall be subject to such underwriting agreement. Each such Participating Holder shall not be required to make any representations or warranties to or agreements with the Company or the underwriters other than customary representations of a selling shareholder, including representations, warranties or agreements regarding its ownership of and title to the Registrable Securities, any written information specifically provided by such Participating Holder for inclusion in the registration statement and its intended method of distribution; and any liability of such Participating Holder to any underwriter or other Person under such underwriting agreement shall in no case be greater than the amount of the net proceeds received by such Participating Holder upon the sale of Registrable Securities pursuant to such registration statement and in no event shall relate to anything other than information about such Holder specifically provided by such Holder for use in the registration statement and prospectus.

Section 4. General.

- 4.1. Adjustments Affecting Registrable Securities. The provisions of this Agreement shall apply, to the full extent set forth herein with respect to the Registrable Securities, to any and all shares of capital stock of the Company, any successor or assign of the Company (whether by merger, share exchange, consolidation, sale of assets or otherwise) or any Subsidiary or parent company of the Company which may be issued in respect of, in exchange for or in substitution of, Registrable Securities and shall be appropriately adjusted for any stock dividends, splits, reverse splits, combinations, recapitalizations and the like occurring after the date hereof.
- 4.2. Rule 144. The Company covenants that (i) so long as it remains subject to the reporting provisions of the Exchange Act, it will timely file the reports required to be filed by it under the Securities Act or the Exchange Act (including, but not limited to, the reports under Sections 13 and 15(d) of the Exchange Act referred to in subparagraph (c)(1)(i) of Rule 144 under the Securities Act, as such Rule may be amended ("Rule 144")) or, if the Company is not required

to file such reports, it will, upon the request of any Holder, make publicly available other information so long as necessary to permit sales by such Holder under Rule 144, or any similar rules or regulations hereafter adopted by the SEC, and (ii) it will take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell Registrable Securities without registration under the Securities Act within the limitation of the exemptions provided by Rule 144, or any similar rule or regulation hereafter adopted by the SEC. Upon the request of any Holder of Registrable Securities, the Company will promptly deliver to such Holder a written statement as to whether it has complied with such requirements.

- 4.3. <u>Nominees for Beneficial Owners</u> If Registrable Securities are held by a nominee for the beneficial owner thereof, the beneficial owner thereof may, at its option, be treated as the Holder of such Registrable Securities for purposes of any request or other action by any Holder or Holders of Registrable Securities pursuant to this Agreement (or any determination of any number or percentage of shares constituting Registrable Securities held by any Holder or Holders of Registrable Securities contemplated by this Agreement); <u>provided, however</u>, that the Company shall have received evidence reasonably satisfactory to it of such beneficial ownership.
- 4.4. Amendments and Waivers. Except as otherwise provided herein, no modification, amendment or waiver of any provision of this Agreement shall be effective against the Company or any Holder unless such modification, amendment or waiver is approved in writing by the Company and the Holders holding a majority of the Registrable Securities then held by all Holders; provided that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects one Holder, solely in its capacity as a Holder of Registrable Securities, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of the Holder so affected. No waiver of any of the provisions of this Agreement shall be deemed to or shall constitute a waiver of any other provision hereof (whether or not similar). No failure or delay on the part of any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof or of any other or future exercise of any such right, power or privilege.
- 4.5. Notices. All notices, demands and other communications to be given or delivered under or by reason of the provisions of this Agreement shall be in writing and shall be deemed to have been given (i) if personally delivered, on the date of delivery, (ii) if delivered by express courier service of national standing (with charges prepaid), on the Business Day following the date of delivery to such courier service, (iii) if deposited in the United States mail, first-class postage prepaid, on the fifth (5th) Business Day following the date of such deposit, (iv) if delivered by facsimile transmission, upon confirmation of successful transmission, (x) on the date of such transmission is completed at or prior to 5:00 p.m., local time of the recipient party on a Business Day, and (y) on the next Business Day following the date of transmission, if such transmission is completed after 5:00 p.m., local time of the recipient party, or is transmitted on a day that is not a Business Day, or (v) if via e-mail communication, on the date of delivery. All notices, demands and other communications hereunder shall be delivered as set forth below and to any subsequent holder of Stock subject to this Agreement at such address as indicated by the Company's records, or pursuant to such other instructions as may be designated in writing by the party to receive such notice:

if to the Company, to:

if to any Holder, to the address set forth opposite the name of such Holder on the signature pages hereto or such other address indicated in the records of the Company.

4.6. Successors and Assigns. Except as otherwise provided herein, this Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and the respective successors, permitted assigns, heirs and personal representatives of the parties hereto, whether so expressed or not. This Agreement may not be assigned by the Company without the prior written consent of the Holders. No Holder shall have the right to assign all or part of its or his rights and obligations under this Agreement to any Person without the consent of the Company and unless such Person duly executes and delivers to the Company a Joinder Agreement. Upon any such assignment, such assignee shall have and be able to exercise and enforce all rights of the assigning Holder which are assigned to it and, to the extent such rights are assigned, any reference to the assigning Holder shall acquire additional Registrable Securities, such Registrable Securities shall be subject to all of the terms, and entitled to all the benefits, of this Agreement. Additional Persons may become parties to this Agreement as Holders with the consent of the Company (not to be unreasonably withheld or delayed), by executing and delivering to the Company the Joinder Agreement.

4.7. Termination.

- (a) The obligations of the Company and a Holder under this Agreement, in each case solely with respect to such Holder, will terminate upon the earlier of:
 - (i) the date on which such Holder no longer holds any Registrable Securities; or
- (ii) the later of (A) the date on which such Holder no longer beneficially owns at least 1% of the then outstanding Class A Common Stock or Class A Common Stock Equivalents, and such Holder (notwithstanding any beneficial ownership of Class A Common Stock or Class A Common Stock Equivalents by such Holder) is not an Affiliate of the Company and (B) the date on which such the Holder is eligible to sell its Registrable Securities pursuant to Rule 144 (without limitation as to volume or manner of sale).
 - (b) This Agreement shall terminate on the date that is five (5 years from date hereof.
- (c) Notwithstanding clauses (a) and (b) above, Section 2.5, Section 2.9, Section 4.9 and Section 4.13 shall survive termination of this Agreement.
- 4.8. Entire Agreement. This Agreement and the other documents referred to herein or delivered pursuant hereto which form part hereof constitute the entire agreement and understanding between the parties hereto and supersedes all prior agreements and understandings relating to the subject matter hereof

4.9. Governing Law; Jurisdiction; Waiver of Jury Trial.

- (a) This Agreement will be governed by, and construed in accordance with, the laws of the State of New York, without giving effect to the principles of conflict of laws thereof.
- (b) Any suit, action or proceeding seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement may be brought against any of the parties in the United States District Court for the Southern District of New York or any New York state court located in New York, New York, and each of the parties hereby consents to the exclusive jurisdiction of such court (and of the appropriate appellate courts) in any such suit, action or proceeding and waives any objection to venue laid therein. Process in any such suit, action or proceeding may be served on any party anywhere in the world, whether within or without the jurisdiction of any such court. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT.

4.10. Interpretation; Construction.

- (a) The table of contents and headings in this Agreement are for convenience of reference only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof. Where a reference in this Agreement is made to a Section, such reference shall be to a Section of this Agreement unless otherwise indicated. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation."
- (b) The parties have participated jointly in negotiating and drafting this Agreement. In the event that an ambiguity or a question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.
- 4.11. <u>Counterparts</u>. This Agreement may be executed and delivered in any number of separate counterparts (including by facsimile or electronic mail), each of which shall be an original, but all of which together shall constitute one and the same agreement.
- 4.12. Severability. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Agreement and the application of such provision to other persons or circumstances shall not be affected by such invalidity or unenforceability, nor shall such invalidity or unenforceability affect the validity or enforceability of such provision, or the application thereof, in any other jurisdiction.

- 4.13. Specific Enforcement. It is agreed and understood that monetary damages would not adequately compensate an injured party for the breach of this Agreement by any party hereto and, accordingly, that this Agreement shall be specifically enforceable, in addition to any other remedy to which such injured party is entitled at law or in equity, and that any breach of this Agreement shall be the proper subject of a temporary or permanent injunction or restraining order. Further, each party hereto waives any claim or defense that there is an adequate remedy at law for such breach or threatened breach or an award of specific performance is not an appropriate remedy for any reason at law or equity and agrees that a party's rights would be materially and adversely affected if the obligations of the other parties under this Agreement were not carried out in accordance with the terms and conditions hereof. Each party further agrees that no party shall be required to obtain, furnish or post any bond or similar instrument in connection with or as a condition to obtain any remedy referred to in this Section 4.13, and each party irrevocably waives any right it may have to require the obtaining, furnishing or posting of any such bond or similar instrument.
- 4.14. <u>Further Assurances</u>. Each party hereto shall do and perform or cause to be done and performed all such further acts and things and shall execute and deliver all such other agreements, certificates, instruments, and documents as any other party hereto reasonably may request in order to carry out the intent and accomplish the purposes of this Agreement and the consummation of the transactions contemplated hereby.
- 4.15. Confidentiality. Each Holder agrees that any non-public information which they may receive relating to the Company and its Subsidiaries (the "Confidential Information") including notices of proposed offerings or any suspension thereof will be held strictly confidential and will not be disclosed by it to any Person without the express written permission of the Company; provided, however, that the Confidential Information may be disclosed (i) in the event of any compulsory legal process or compliance with any applicable law, subpoena or other legal process, as required by an administrative requirement, order, decree or the rules of any relevant stock exchange or in connection with any filings that the Holder may be required to make with any regulatory authority; provided, however, that in the event of compulsory legal process, unless prohibited by applicable law or that process, each Holder agrees (A) to give the Company prompt notice thereof and to cooperate with the Company in securing a protective order in the event of compulsory disclosure and (B) that any disclosure made pursuant to public filings will be subject to the prior reasonable review of the Company, (ii) to any foreign or domestic governmental or quasi-governmental regulatory authority, including any stock exchange or other self-regulatory organization having jurisdiction over such party, (iii) to each Holder's or its Affiliate's, officers, directors, employees, partners, accountants, lawyers and other professional advisors for use relating solely to management of the investment or administrative purposes with respect to such Holder and (iv) to a proposed transferee of securities of the Company held by a Holder; provided, however, that the Holder informs the proposed transferee of the confidential nature of the information and the proposed transferee agrees in writing to comply with the restrictions in this Section 4.15 and delivers a copy of such writing to the Company.

- 4.16. Opt-Out Requests. Each Holder shall have the right, at any time and from time to time (including after receiving information regarding any potential public offering), to elect to not receive any notice that the Company or any other Holders otherwise are required to deliver pursuant to this Agreement by delivering to the Company a written statement signed by such Holder that it does not want to receive any notices hereunder (an "Opt-Out Request"); in which case and notwithstanding anything to the contrary in this Agreement the Company and other Holders shall not be required to, and shall not, deliver any notice or other information required to be provided to Holders hereunder to the extent that the Company or such other Holders reasonably expect would result in a Holder acquiring material non-public information within the meaning of Regulation FD promulgated under the Exchange Act. An Opt-Out Request may state a date on which it expires or, if no such date is specified, shall remain in effect indefinitely. A Holder who previously has given the Company an Opt-Out Request may revoke such request at any time, and there shall be no limit on the ability of a Holder to issue and revoke subsequent Opt-Out Requests; provided that each Holder shall use commercially reasonable efforts to minimize the administrative burden on the Company arising in connection with any such Opt-Out Requests.
- 4.17. Original Registration Rights Agreement. The Sponsor hereby agrees that upon execution of this Agreement by the Sponsor, the Original Registration Rights Agreement shall be automatically terminated and superseded in its entirety by this Agreement.

[Remainder of Page Intentionally Left Blank]

THE COMPANY:

Grove Collaborative Holdings, Inc., a Delaware public benefit corporation

By: /s/ Stuart Landesberg
Name: Stuart Landesberg
Title: Chief Executive Officer

Address: 1301 Sansome Street San Francisco, CA 94111

HOLDER:

VIRGIN GROUP ACQUISITION SPONSOR II LLC

By Corvina Holdings Limited, its Member Manager

By: /s/ Kerry Graziola Name: Kerry Graziola Title: Alternate Director

HOLDER:

Mayfield XV, a Cayman Islands exempted limited partnership

By: MAYFIELD SELECT MANAGEMENT (EGP), L.P., a Cayman Islands Exempted Limited Partnership Its: General Partner

By: MAYFIELD XV MANAGEMENT (UGP), LTD., a Cayman Islands Exempted Company Its: General Partner

By: /s/ Rishi Garg Name: Rishi Garg Title: Partner

Address: 2484 Sand Hill Rd. Menlo Park, CA 94705 USA

HOLDER:

Mayfield Select, a Cayman Islands exempted limited partnership

By: MAYFIELD SELECT MANAGEMENT (EGP), L.P., a Cayman Islands Exempted Limited Partnership Its: General Partner

By: MAYFIELD SELECT MANAGEMENT (UGP), LTD., a Cayman Islands Exempted Company Its: General Partner

By: /s/ Rishi Garg Name: Rishi Garg Title: Partner

Address: 2484 Sand Hill Rd. Menlo Park, CA 94705 USA

HOLDER:

Norwest Venture Partners XIII, LP

By: Genesis VC Partners XIII, LLC, General Partner By: NVP Associates, LLC, Managing Member

By: /s/ Jeff Crowe

Name: Jeff Crowe Title: Managing Partner

Address: 525 University Ave Palo Alto, CA 94301

HOLDER:

General Atlantic (GC), L.P.

By: General Atlantic (SPV) GP, LLC, its general partner

By: /s/ Kelly Pettit

Name: Kelly Pettit Title: Managing Director

Address:

General Atlantic Service Company, L.P. 55 East 52nd Street, Floor 33 New York, NY, 10055

HOLDER:

Lone Cypress, Ltd.

By: Lone Pine Capital LLC, its investment advisor

By: /s/ Kerry A. Tyler

Name: Kerry A. Tyler Title: Authorized Signatory

Address:

Lone Pine Capital LLC, Attn: Legal Department Two Greenwich Plaza Greenwich, CT 06830

HOLDER:

Lone Cascade, L.P.

By: Lone Pine Capital LLC, its investment advisor

By: /s/ Kerry A. Tyler

Name: Kerry A. Tyler Title: Authorized Signatory

Address:

Lone Pine Capital LLC, Attn: Legal Department Two Greenwich Plaza Greenwich, CT 06830

HOLDER:

Lone Spruce, L.P.

By: Lone Pine Capital LLC, its investment advisor

By: /s/ Kerry A. Tyler

Name: Kerry A. Tyler Title: Authorized Signatory

Address:

Lone Pine Capital LLC, Attn: Legal Department Two Greenwich Plaza Greenwich, CT 06830

HOLDER:

Lone Monterey Master Fund, Ltd.

By: Lone Pine Capital LLC, its investment advisor

By: /s/ Kerry A. Tyler

Name: Kerry A. Tyler Title: Authorized Signatory

Address:

Lone Pine Capital LLC, Attn: Legal Department Two Greenwich Plaza Greenwich, CT 06830

HOLDER:

Lone Sierra, L.P.

By: Lone Pine Capital LLC, its investment advisor

By: /s/ Kerry A. Tyler

Name: Kerry A. Tyler Title: Authorized Signatory

Address:

Lone Pine Capital LLC, Attn: Legal Department Two Greenwich Plaza Greenwich, CT 06830

HOLDER:

MHS Capital Partners II, L.P.

By: MHS Capital Management II, LLC

Its: General Partner

By: /s/ Mark Sugarman

Name: Mark Sugarman Title: Managing Member

MHS Capital
2121 South El Camino Real suite 200
San MATEO, CA 99403

HOLDER:

MHS Capital Partners G, LLC

By: MHS Capital Management II, LLC Its: General Partner

By: /s/ Mark Sugarman

Name: Mark Sugarman Title: Managing Member

Address: MHS Capital 2121 El Camino Real suite 200 San Mateo, CA 99403

HOLDER:

MHS Capital Partners G2, LLC

By: MHS Capital Management II, LLC Its: General Partner

By: /s/ Mark Sugarman

Name: Mark Sugarman Title: Managing Member

Address: MHS Capital 2121 El Camino Real suite 200 San Mateo, CA 99403

HOLDERS:

SCM GC Investments Limited

By: /s/ Wayne Cohen

Name: Wayne Cohen Title: Authorized Signatory

Address:

9 West 57th Street, 39th Floor

New York, NY 10019

Attn: Legal

Sculptor Master Fund, Ltd.

By: Sculptor Capital LP, its investment manager By: Sculptor Capital Holding Corporation, its General Partner

By: /s/ Wayne Cohen

Name: Wayne Cohen

Title: President and Chief Operating Officer

Sculptor Enhanced Master Fund, Ltd.

By: Sculptor Capital LP, its investment manager By: Sculptor Capital Holding Corporation, its General Partner

By: /s/ Wayne Cohen

Name: Wayne Cohen

Title: President and Chief Operating Officer

Address:

9 West 57th Street, 39th Floor New York, NY 10019

Attn: Legal

HOLDER:

CORVINA HOLDINGS LIMITED

By: /s/ Kerry Graziola Name: Kerry Graziola Title: Alternate Director

HOLDER:

/s/ Stuart Landesberg

Stuart Landesberg

Address: 7 Heuters lane Mill valley ca 94941 Usa

HOLDER:

THE LANDESBERG LIVING TRUST, DATED OCTOBER 15, 2021

By: /s/ Stuart A. Landesberg /s/ Caitlin Landesberg

Name: Stuart A. Landesberg and Caitlin Landesberg, as
co-trustees of The Landesberg Living Trust, dated
October 15, 2021

Address: 7 Heuters Lane Mill valley ca 94941

JOINDER AGREEMENT

This Joinder Agreement (this "Joinder Agreement") is made as of [], by [and among [] (the "<u>Transferring Holder</u>") and [] (the "<u>New Holder</u>"), in accordance with that certain Amended and Restated Registration Rights Agreement, dated as of June 16, 2022 (as amended from time to time, the "<u>Agreement</u>"), by and among Grove Collaborative Holdings, Inc. (the "<u>Company</u>") and the other Holders party thereto.

WHEREAS, the Agreement requires the New Holder to become a party to the Agreement by executing this Joinder Agreement, and upon the New Holder signing this Joinder Agreement, the Agreement will be deemed to be amended to include the New Holder as a Holder thereunder;

NOW, THEREFORE, in consideration of the foregoing, and of the representations, warranties, covenants and agreements contained herein, and intending to be legally bound hereby, the parties hereto agree as follows:

Section 1. Party to the Agreement. By execution of this Joinder Agreement, as of the date hereof the New Holder is hereby made a party to the Agreement as a Holder. The New Holder hereby agrees to become a party to the Agreement and to be bound by, and subject to, all of the representations, covenants, terms and conditions of the Agreement in the same manner as if the New Holder were an original signatory to the Agreement. Execution and delivery of this Joinder Agreement by the New Holder shall also constitute execution and delivery by the New Holder of the Agreement, without further action of any party.

- Section 2. <u>Defined Terms</u>. Capitalized terms used but not defined herein shall have the meanings set forth in the Agreement unless otherwise noted.
- Section 3. Representations and Warranties of the New Holder.
- 3.1. <u>Authorization</u>. The New Holder has all requisite power and authority and has taken all action necessary in order to duly and validly approve the New Holder's execution and delivery of, and performance of its obligations under, this Joinder Agreement. This Joinder Agreement has been duly executed and delivered by the New Holder and constitutes a legal, valid and binding agreement of the New Holder, enforceable against the New Holder in accordance with its terms.
- 3.2. No Conflict. The New Holder is not under any obligation or restriction, nor shall it assume any such obligation or restriction, that does or would materially interfere or conflict with the performance of its obligations under this Joinder Agreement.

Section 4. <u>Further Assurances</u>. The parties agree to execute and deliver any further instruments or perform any acts which are or may become necessary to effectuate the purposes of this Joinder Agreement.

Exhibit A-1

Section 5. <u>Governing Law</u>. This Joinder Agreement will be governed by, and construed in accordance with, the laws of the State of New York, without giving effect to the principles of conflict of laws thereof.

Section 6. Counterparts. This Joinder Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same amendatory instrument.

Section 7. Entire Agreement. This Joinder Agreement and the Agreement contain the entire understanding, whether oral or written, of the parties hereto with respect to the matters covered hereby. Any amendment or change in this Joinder Agreement shall not be valid unless made in writing and signed by each of the parties hereto.

[Signature pages follow]

Exhibit A-2

IN WITNESS WHEREOF, intending to be legally bound hereby, the undersigned parties have executed this Joinder Agreement as of the date first above written.

	[TRANSFERRING HOLDER]
	[]
	By:
	Name:
	Title:
	NEW HOLDER
	[]
	By:
	Name:
	Title:
	Notice Address: [] [] [] Attn: [] Facsimile: []
Accepted and Agreed to as of the date first written above:	
the date first written above.	
COMPANY	
Grove Collaborative Holdings, Inc.	
Ву:	
Name:	
Title:	

Exhibit A-3

MEZZANINE LOAN AND SECURITY AGREEMENT

THIS MEZZANINE LOAN AND SECURITY AGREEMENT (this "Agreement") dated as of April 30, 2021 (the "Effective Date"), among (a) SILICON VALLEY BANK, a California corporation ("SVB"), in its capacity as administrative agent and collateral agent ("Agent"), (b) SILICON VALLEY BANK, a California corporation, as a lender, (c) HERCULES CAPITAL, INC., a Maryland corporation ("Hercules"), as a lender (SVB and Hercules and each of the other "Lenders" from time to time a party hereto are referred to herein collectively as the "Lenders" and each individually as a "Lender"), and (d) GROVE COLLABORATIVE, INC., a Delaware public benefit corporation ("Borrower"), provides the terms on which Agent and the Lenders shall lend to Borrower, and Borrower shall repay Agent and the Lenders. The parties agree as follows:

1 ACCOUNTING AND OTHER TERMS

Accounting terms not defined in this Agreement shall be construed following GAAP. Calculations and determinations must be made following GAAP, except, in each case applicable hereunder, with respect to unaudited financial statements, for the absence of footnotes and subject to year-end adjustments. Capitalized terms not otherwise defined in this Agreement shall have the meanings set forth in Section 14 of this Agreement. All other terms contained in this Agreement, unless otherwise indicated, shall have the meaning provided by the Code to the extent such terms are defined therein.

2 LOAN AND TERMS OF PAYMENT

2.1 Promise to Pay. Borrower hereby unconditionally promises to pay to Agent, for the ratable benefit of each Lender, the outstanding principal amount of all Credit Extensions advanced to Borrower by such Lender and accrued and unpaid interest thereon, together with any fees as and when due in accordance with this Agreement.

2.2 Term Loan Advances.

- (a) Availability. Subject to the terms and conditions of this Agreement, upon Borrower's request, the Lenders, severally and not jointly, during the Draw Period A, shall make term loan advances to Borrower in an aggregate original principal amount equal to Sixty Million Dollars (\$60,000,000.00) according to each Lender's Term A Loan Commitment as set forth on Schedule 1 hereto (each, a "Term A Loan Advance"), and collectively, the "Term A Loan Advances"); provided that the initial Term A Loan Advance shall be made on or about the Effective Date, in an original principal amount of Twenty-Five Million Dollars (\$25,000,000.00). Subject to the terms and conditions of this Agreement, upon Borrower's request, during the Draw Period B, the Lenders, severally and not jointly, shall make term loan advances available to Borrower in an aggregate original principal amount of up to Twenty Million Dollars (\$20,000,000.00) according to each Lender's Term B Loan Commitment as set forth on Schedule 1 hereto (each, a "Term B Loan Advance", and collectively, the "Term B Loan Advances"). Each Term Loan Advance shall be in a minimum amount of at least Five Million Dollars (\$5,000,000.00). The Term A Loan Advances and Term B Loan Advances are hereinafter referred to singly as a "Term Loan Advance" and collectively as the "Term Loan Advances". After repayment, no Term Loan Advance (or any portion thereof) may be reborrowed.
- (b) Interest Payment. Commencing on the first (1st) Payment Date of the month following the month in which the Funding Date of the applicable Term Loan Advance occurs and continuing on the Payment Date of each month thereafter until such Term Loan Advance is repaid (without duplication as to part (c) of this Section as to interest payments), Borrower shall make monthly payments of accrued interest to (i) Agent, for the account of Lenders (other than Hercules) in accordance with each such Lender's Pro Rata Share, and (ii) Hercules for its own account in accordance with its Pro Rata Share, in each case in arrears, on the outstanding principal amount of the Term Loan Advance, at the rate set forth in Section 2.3(a).
- (c) Repayment. Commencing on November 1, 2022, and continuing on each Payment Date thereafter until payment in full, Borrower shall repay the aggregate outstanding Term Loan Advances to (x) Agent, for the account of Lenders (other than Hercules) in accordance with each such Lender's Pro Rata Share, and (y) Hercules for its own account in accordance with its Pro Rata Share, in (i) thirty (30) consecutive equal monthly installments of principal, plus (ii) monthly payments of accrued interest at the rate set forth in Section 2.3(a). All outstanding principal and accrued and unpaid interest with respect to the Term Loan Advances, and all other outstanding Obligations under the Term Loan Advances are due and payable in full to each Lender in accordance with such Lender's Pro Rata Share on the Term Loan Maturity Date.

- (d) Permitted Prepayment. Borrower shall have the option to prepay all, or any portion not less than Five Million Dollars (\$5,000,000), of the Term Loan Advances advanced by the Lenders under this Agreement, provided Borrower (i) delivers written notice to Agent of its election to prepay the Term Loan Advances at least seven (7) Business Days prior to such prepayment, including the principal amount that will be prepaid, and (ii) pays (y) to Agent, for the account of the Lenders (other than Hercules) in accordance with each Lender's respective Pro Rata Share and (z) Hercules for its own account in accordance with its Pro Rata Share, on the date of such prepayment (A) the outstanding principal amount of Term Loan Advances being prepaid, plus accrued and unpaid interest with respect to the Term Loan Advances being prepaid, (B) the Prepayment Premium, (C) the Final Payment, and (D) all other sums, if any, that shall have become due and payable with respect to the Term Loan Advances, including Lenders' Expenses and interest at the Default Rate with respect to any past due amounts.
- (e) Mandatory Prepayment Upon an Acceleration. If the Term Loan Advances are accelerated by Agent, following the occurrence and during the continuance of an Event of Default, Borrower shall immediately pay to (y) Agent, for the account of the Lenders (other than Hercules) in accordance with each Lender's respective Pro Rata Share and (z) Hercules for its own account in accordance with its Pro Rata Share, an amount equal to the sum of (i) all outstanding principal plus accrued and unpaid interest with respect to the Term Loan Advances, (ii) the Prepayment Premium, (iii) the Final Payment, and (iv) all other sums, if any, that shall have become due and payable with respect to the Term Loan Advances, including Lenders' Expenses and interest at the Default Rate with respect to any past due amounts.

2.3 Payment of Interest on the Credit Extensions.

- (a) Interest Rate. Subject to Section 2.3(b), the principal amount outstanding under each Term Loan Advance shall accrue interest at a floating per annum rate equal to the greater of (i) eight and three quarters of one percent (8.75%) and (ii) five and one half of one percent (5.50%) above the Prime Rate, which interest, in each case, shall be payable monthly in accordance with Section 2.3(d) below.
- (b) <u>Default Rate</u>. Immediately upon the occurrence and during the continuance of an Event of Default, outstanding Obligations shall bear interest at a rate per annum which is five percent (5.0%) above the rate that is otherwise applicable thereto (the "**Default Rate**") unless the Lenders otherwise elect from time to time in their sole discretion to impose a smaller increase. Fees and expenses which are required to be paid by Borrower pursuant to the Loan Documents (including, without limitation, Lenders' Expenses) but are not paid when due shall bear interest until paid at a rate equal to the highest rate applicable to the Obligations. Payment or acceptance of the increased interest rate provided in this Section 2.3(b) is not a permitted alternative to timely payment and shall not constitute a waiver of any Event of Default or otherwise prejudice or limit any rights or remedies of Agent or any Lender.
- (c) Adjustment to Interest Rate. Changes to the interest rate of any Credit Extension based on changes to the Prime Rate shall be effective on the effective date of any change to the Prime Rate and to the extent of any such change.
- (d) <u>Payment; Interest Computation</u>. Interest is payable monthly in arrears on the Payment Date and shall be computed on the basis of a three hundred sixty (360) day year for the actual number of days elapsed. In computing interest, (i) all payments received after 1:00 p.m. Pacific time on any day shall be deemed received at the opening of business on the next Business Day, and (ii) the date of the making of any Credit Extension shall be included and the date of payment shall be excluded; provided, however, that if any Credit Extension is repaid on the same day on which it is made, such day shall be included in computing interest on such Credit Extension.

- 2.4 Fees. Borrower shall pay (i) to Agent, for the account Lenders (other than Hercules) and (ii) Hercules for its own account (without duplication as to the amounts below):
- (a) <u>Commitment Fee</u>. (i) A fully earned, non-refundable commitment fee of Six Hundred Thousand Dollars (\$600,000.00), on the Effective Date, to be shared between the Lenders pursuant to their respective Term Loan Commitment Percentages and (ii) a commitment fee of Two Hundred Thousand Dollars (\$200,000.00), on the Funding Date of the initial draw of the Term B Loan Advance, which shall be fully earned, non-refundable commitment fee at such time, in each case to be shared between the Lenders pursuant to their respective Term Loan Commitment Percentages;
 - (b) Prepayment Premium. The Prepayment Premium, when due hereunder;
- (c) <u>Final Payment</u>. The Final Payment, when due hereunder, to be shared between the Lenders pursuant to their respective Term Loan Commitment Percentages;
- (d) <u>Good Faith Deposit</u>. Borrower has paid to Agent a good faith deposit of Eighty Thousand Dollars (\$80,000.00) (the **'Good Faith Deposit'**) prior to the date hereof which shall be applied toward Lender Expenses (and the Commitment Fee) incurred through the Effective Date; and
- (e) <u>Lenders' Expenses</u>. All Lenders' Expenses (including reasonable attorneys' fees and expenses for documentation and negotiation of this Agreement) incurred through and after the Effective Date, when due (or, if no stated due date, upon demand by Agent), provided that Agent and Lenders agree that Lender Expenses incurred through the Effective Date, <u>plus</u> Bank Expenses (as defined in the Senior Loan Agreement), incurred in amending the Senior Loan Agreement around the date of this Agreement, shall not exceed Forty Thousand Dollars (\$40,000) in the aggregate.

Unless otherwise provided in this Agreement or in a separate writing by Agent, Borrower shall not be entitled to any credit, rebate, or repayment of any fees earned by Agent or any Lender pursuant to this Agreement notwithstanding any termination of this Agreement or the suspension or termination of any Lender's obligation to make loans and advances hereunder. Agent may deduct amounts owing by Borrower under the clauses of this Section 2.4 pursuant to the terms of Section 2.5(e). Agent shall provide Borrower written notice of deductions made from the Designated Deposit Account pursuant to the terms of the clauses of this Section 2.4.

2.5 Payments; Pro Rata Treatment; Application of Payments; Debit of Accounts.

- (a) All payments (including prepayments) to be made by Borrower under any Loan Document shall be made to (i) Agent for the account of Lenders (other than Hercules) in accordance with each such Lender's Pro Rata Share, and (ii) Hercules for its own account in accordance with its Pro Rata Share, in immediately available funds in Dollars, without setoff or counterclaim, before 1:00 p.m. Pacific time on the date when due. Agent shall distribute such payments to Lenders (other than Hercules) in like funds as set forth in Section 2.6. Payments of principal and/or interest received after 1:00 p.m. Pacific time are considered received at the opening of business on the next Business Day. When a payment is due on a day that is not a Business Day, the payment shall be due the next Business Day, and additional fees or interest, as applicable, shall continue to accrue until paid.
- (b) Each borrowing by Borrower from Lenders hereunder shall be made according to the respective Term Loan Commitment Percentages of the relevant Lenders.
- (c) Except as otherwise provided herein, each payment (including each prepayment) by Borrower on account of principal or interest on the Term Loan Advances shall be applied according to each Lender's Pro Rata Share of the outstanding principal amount of the Term Loan Advances. The amount of each principal prepayment of the Term Loan Advances shall be applied to reduce the then remaining installments of the Term Loan Advances based upon each Pro Rata Share of Term Loan Advances.
- (d) Agent has the exclusive right to determine the order and manner in which all payments with respect to the Obligations may be applied. Borrower shall have no right to specify the order or the accounts to which Agent shall allocate or apply any payments required to be made by Borrower to Agent or Hercules or otherwise received by Agent or any Lender under this Agreement when any such allocation or application is not specified elsewhere in this Agreement.

(e) Agent may debit any of Borrower's deposit accounts, including the Designated Deposit Account, for principal and interest payments or any other amounts Borrower owes Agent or any Lender arising under this Agreement and the other Loan Documents when due and owing. These debits shall not constitute a set-off.

Unless Agent shall have been notified in writing by Borrower prior to the date of any payment due to be made by Borrower hereunder that Borrower will not make such payment to Agent, Agent may assume that Borrower is making such payment, and Agent may, but shall not be required to, in reliance upon such assumption, make available to Lenders their respective Pro Rata Share of a corresponding payment amount. If such payment is not made to Agent by Borrower within three (3) Business Days after such due date, Agent shall be entitled to recover, on demand, from each Lender to which any amount which was made available pursuant to the preceding sentence, such amount with interest thereon at the rate per annum equal to the daily average Federal Funds Effective Rate. Nothing herein shall be deemed to limit the rights of Agent or any Lender against Borrower.

2.6 Settlement Procedures. If Agent receives any payment for the account of Lenders on or prior to 1:00 p.m. (Pacific time) on any Business Day, Agent shall pay to each applicable Lender such Lender's Pro Rata Share of such payment on such Business Day. If Agent receives any payment for the account of Lenders after 1:00 p.m. (Pacific time) on any Business Day, Agent shall pay to each applicable Lender such Lender's Pro Rata Share of such payment on the next Business Day.

2.7 Withholding by Borrower.

(a) Payments received by Agent or any Lender from Borrower under this Agreement will be made free and clear of and without deduction for any present or future taxes, levies, imposts, duties, deductions, withholdings, assessments, fees or other similar charges imposed by any Governmental Authority (including any interest, additions to tax or penalties applicable thereto), in each case other than in respect of Excluded Taxes. Specifically, however, if at any time any Governmental Authority, applicable law, regulation or international agreement requires Borrower to make any withholding or deduction from any such payment or other sum payable hereunder to Agent or any Lender, in each case other than in respect of Excluded Taxes, Borrower hereby covenants and agrees that the amount due from Borrower with respect to such payment or other sum payable hereunder will be increased to the extent necessary to ensure that, after the making of such required withholding or deduction, Agent and/or Lender receives a net sum equal to the sum which it would have received had no withholding or deduction been required, and Borrower shall pay the full amount withheld or deducted to the relevant Governmental Authority provided however that to the extent Agent and/or any Lender is refunded any portion of such excess Agent and/or such Lender shall remit such amount to Borrower. Borrower will, upon request, furnish Agent with proof reasonably satisfactory to Agent indicating that Borrower has made such withholding payment; provided, however, that Borrower need not make any withholding payment if the amount or validity of such withholding payment is contested in good faith by appropriate and timely proceedings and as to which payment in full is bonded or reserved against by Borrower. The agreements and obligations of Borrower contained in this Section 2.7 shall survive the termination of this Agreement.

(b) If any assignee of a Lender's rights under Section 12.2 of this Agreement is not a "United States Person" as defined in Section 7701(a) (30) of the Internal Revenue Code of 1986, as amended from time to time (such assignee, a "Non-U.S. Lender"), such Non-U.S. Lender shall, upon becoming party to this Agreement, to the extent that such Non-U.S. Lender is entitled to an exemption from U.S. withholding tax on interest, deliver to Borrower a complete and properly executed IRS Form W-8BEN, W-8ECI or W-8IMY, as appropriate, or any successor form prescribed by the IRS, certifying that such Non-U.S. Lender is entitled to such exemption from U.S. withholding tax on interest. Notwithstanding Section 2.7(a) above, Borrower shall not be required to pay any additional amount to any Non-U.S. Lender under Section 2.7(a) if such Non-U.S. Lender fails or is unable to deliver the forms, certificates or other evidence described in the preceding sentence, unless such Non-U.S. Lender's failure or inability to deliver such forms is the result of any change in any applicable law, treaty or governmental rule, or any change in the interpretation thereof after such Non-U.S. Lender became a party to this Agreement.

3 CONDITIONS OF LOANS

- **3.1 Conditions Precedent to Initial Credit Extension**. Each Lender's obligation to make the initial Credit Extension hereunder is subject to the condition precedent that Agent shall have received, in form and substance satisfactory to Agent and the Lenders, such documents, and completion of such other matters, as Agent shall have reasonably requested, including, without limitation:
 - (a) duly executed amendment to the Senior Loan Agreement and satisfaction of all conditions precedent therein;
 - (b) duly executed signatures to the Loan Documents;
 - (c) duly executed original signatures to the Warrant;
 - (d) duly executed signatures to the IP Agreement;
- (e) (i) the Operating Documents, (ii) a long-form good standing certificate of Borrower certified by the Secretary of State of Delaware and (iii) certificates of foreign qualification, each certified, respectively by the Secretary of the Commonwealth of the Commonwealth of California, the Secretary of State of the State of New York, and the Secretary of State of the State of North Carolina, each as of a date no earlier than thirty (30) days prior to the Effective Date:
- (f) a secretary's corporate borrowing certificate of Borrower with respect to Borrower's Operating Documents, incumbency and resolutions authorizing the execution and delivery of this Agreement and the other Loan Documents;
 - (g) duly executed signatures to the completed Borrowing Resolutions for Borrower;
 - (h) duly executed signature to a payoff letter from TRIPLEPOINT VENTURE GROWTH BDC CORP.;
- (i) evidence that (i) the Liens securing Indebtedness owed by Borrower to TRIPLEPOINT VENTURE GROWTH BDC CORP. will be terminated and (ii) the documents and/or filings evidencing the perfection of such Liens, including without limitation any financing statements and/or control agreements, have or will, concurrently with the initial Credit Extension, be terminated;
- (j) certified copies, dated as of a recent date, of Lien searches (including without limitation, UCC searches), as Agent may reasonably request, accompanied by written evidence (including any UCC termination statements and other Lien releases) that the Liens indicated in any such financing statements or other filings either constitute Permitted Liens or have been or, in connection with the initial Credit Extension hereunder, will be terminated or released;
 - (k) the Perfection Certificate of Borrower, together with the duly executed signature thereto;
 - (l) Intellectual Property search results and completed exhibits to the IP Agreement;
 - (m) a legal opinion of Borrower's counsel dated as of the Effective Date, together with the duly executed signature thereto;
- (n) payment in full of the Growth Capital Advances (as defined in the Senior Loan Agreement) and accrued interest and fees in connection therewith; and
 - (o) payment of the fees and Lenders' Expenses then due as specified in Section 2.4 hereof.
- 3.2 Conditions Precedent to all Credit Extensions. Each Lender's obligation to make each Credit Extension, including the initial Credit Extension, is subject to the following conditions precedent:
 - (a) timely receipt by the Lenders of (i) an executed Disbursement Letter; and (ii) an executed Payment/Advance Form;

- (b) the representations and warranties in this Agreement shall be true, accurate, and complete in all material respects on the date of the Disbursement Letter (and the Payment/Advance Form) and on the Funding Date of each Credit Extension; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects as of such date, and no Event of Default shall have occurred and be continuing or result from the Credit Extension. Each Credit Extension is Borrower's representation and warranty on that date that the representations and warranties in this Agreement are true, accurate, and complete in all material respects as of such date; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects as of such date; and
 - (c) Agent and each Lender determine to its satisfaction that there has not been a Material Adverse Change.
- 3.3 Covenant to Deliver. Except as otherwise provided in Section 6.11 hereof, Borrower agrees to deliver to Agent and each Lender each item required to be delivered to Agent and each Lender under this Agreement as a condition precedent to any Credit Extension. Borrower expressly agrees that a Credit Extension made prior to the receipt by Agent and each Lender of any such item shall not constitute a waiver by Agent or Lenders of Borrower's obligation to deliver such item, and the making of any Credit Extension in the absence of a required item shall be in each Lender's sole discretion.

3.4 Procedures for Borrowing.

- (a) <u>Term Loan Advances</u>. Subject to the prior satisfaction of all other applicable conditions to the making of a Credit Extension set forth in this Agreement, to obtain a Credit Extension, Borrower shall notify Agent (which notice shall be irrevocable) by electronic mail, facsimile, or telephone by 1:00 p.m. Pacific time at least five (5) Business Days before the proposed Funding Date of such Credit Extension. Together with any such electronic or facsimile notification, Borrower shall deliver to Agent by electronic mail or facsimile a completed Disbursement Letter (and Payment/Advance Form) executed by an Authorized Signer. Agent may rely on any telephone notice given by a person whom Agent reasonably believes is an Authorized Signer. On the Funding Date, Agent shall credit the Credit Extensions to the Designated Deposit Account. Agent may make Credit Extensions under this Agreement based on instructions from an Authorized Signer or without instructions if the Credit Extensions are necessary to meet Obligations which have become past due.
- (b) Funding. In determining compliance with any condition hereunder to the making of a Credit Extension that, by its terms, must be fulfilled to the satisfaction of a Lender, Agent may presume that such condition is satisfactory to such Lender unless Agent shall have received notice to the contrary from such Lender prior to the making of such Credit Extension. Unless Agent shall have been notified in writing by any Lender prior to the date of any Credit Extension, that such Lender will not make the amount that would constitute its share of such borrowing available to Agent, Agent may assume that such Lender is making such amount available to Agent, and Agent may, in reliance upon such assumption, make available to Borrower a corresponding amount; provided that, Agent shall notify Borrower that Agent has funded such Lender's share based upon such assumption. If such amount is not made available to Agent by the required time on the Funding Date therefor, such Lender shall pay to Agent, on demand, such amount with interest thereon, at a rate equal to the greater of (i) the Federal Funds Effective Rate or (ii) a rate determined by Agent in accordance with banking industry rules on interbank compensation, for the period until such Lender makes such amount immediately available to Agent. If such Lender's share of such Credit Extension is not made available to Agent by such Lender within five (5) Business Days after such Funding Date, Agent shall also be entitled to recover such Lender's share of such Credit Extension with interest thereon at the rate per annum applicable to the Term Loan Advances, on demand, from Borrower, only within ten (10) Business Days after such Funding Date.

4 CREATION OF SECURITY INTEREST

4.1 Grant of Security Interest. Borrower hereby grants Agent, for the ratable benefit of the Lenders, to secure the payment and performance in full of all of the Obligations, a continuing security interest in, and pledges to Agent, for the ratable benefit of the Lenders, the Collateral, wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof. For clarity, any reference to "Agent's Lien" or any granting of collateral to Agent in this Agreement or any Loan Document means the Lien granted to Agent for the ratable benefit of the Lenders.

Borrower acknowledges that it previously has entered, and/or may in the future enter, into Bank Services Agreements with SVB. Regardless of the terms of any Bank Services Agreement, Borrower agrees that any amounts Borrower owes SVB thereunder shall be deemed to be Obligations hereunder and that it is the intent of Borrower and SVB to have all such Obligations secured by the first priority perfected security interest in the Collateral granted herein (subject only to Permitted Liens that are permitted pursuant to the terms of this Agreement to have superior priority to Agent's Lien in this Agreement, including, without limitation, the senior liens granted pursuant to the Senior Loan Agreement), and by any and all other security agreements, or other collateral (that constitute Loan Documents in connection with this Agreement) granted to Agent by Borrower as security for the Obligations, now or in the future.

If this Agreement is terminated, Agent's Lien in the Collateral shall continue until the Obligations (other than inchoate indemnity or other obligations which by their terms, survive termination of this Agreement) are repaid in full in cash. Upon payment in full in cash of the Obligations (other than inchoate indemnity or other obligations which by their terms, survive termination of this Agreement) and at such time as the Lenders' obligation to make Credit Extensions has terminated, Agent shall, at the sole cost and expense of Borrower, promptly release its Liens in the Collateral and all rights therein shall revert to Borrower, and Agent shall, upon reasonable request from Borrower and at Borrower's sole cost and expense, promptly deliver to Borrower written evidence of the termination of such liens and any other documents reasonably necessary to terminate such liens. In the event (x) all Obligations (other than inchoate indemnity or other obligations which by their terms, survive termination of this Agreement), except for Bank Services, are satisfied in full, and (y) this Agreement is terminated, Agent shall terminate the security interest granted herein upon Borrower providing to SVB cash collateral acceptable to SVB in its good faith business judgment for Bank Services, if any. In the event such Bank Services consist of outstanding Letters of Credit, Borrower shall provide to SVB cash collateral in an amount equal to (x) if such Letters of Credit are denominated in Dollars, then at least one hundred tre percent (105.0%); and (y) if such Letters of Credit plus, in each case, all interest, fees, and costs due or to become due in connection therewith (as estimated by SVB in its business judgment), to secure all of the Obligations relating to such Letters of Credit.

Agent's Lien on the assets of Borrower securing the Obligations of Borrower to Lenders under this Agreement and granted pursuant to the terms of this Agreement shall be junior and subordinated to SVB's security interest in the assets of Borrower securing the Obligations of Borrower to SVB under the Senior Loan Agreement.

- 4.2 Priority of Security Interest. Borrower represents, warrants, and covenants that the security interests granted herein are and shall at all times continue to be a first priority perfected security interests in the Collateral (subject only to Permitted Liens that are permitted pursuant to the terms of this Agreement to have superior priority to Agent's Lien under this Agreement, including, without limitation, the senior liens granted pursuant to the Senior Loan Agreement)). If Borrower shall acquire a commercial tort claim reasonably likely to exceed Twenty-Five Thousand Dollars (\$25,000), Borrower shall promptly notify Agent in a writing signed by Borrower of the general details thereof and grant to Agent, for the ratable benefit of the Lenders, in such writing a security interest therein and in the proceeds thereof, all upon the terms of this Agreement, with such writing to be in form and substance reasonably satisfactory to Agent. Notwithstanding the foregoing, or anything to the contrary herein, or in the other Loan Documents, Borrower shall not be required to provide a notation of the Agent's security interest granted hereunder on the certificate(s) of title for any vehicles (including motor vehicles and trailers) owned by Borrower.
- 4.3 Authorization to File Financing Statements. Borrower hereby authorizes Agent, on behalf of the Lenders, to file financing statements covering the Collateral, without notice to Borrower, with all appropriate jurisdictions to perfect or protect Agent's and Lenders' interest or rights hereunder, including a notice that any disposition of the Collateral (except as permitted under this Agreement), by Borrower or any other Person, shall be deemed to violate the rights of Agent under the Code. Such financing statements may indicate the Collateral as "all assets of the Debtor" or words of similar effect, or as being of an equal or lesser scope, or with greater detail, all in Agent's discretion.

5 REPRESENTATIONS AND WARRANTIES

Borrower represents and warrants as follows:

5.1 Due Organization, Authorization; Power and Authority. Borrower is duly existing and in good standing as a Registered Organization in its jurisdiction of formation and is qualified and licensed to do business and is in good standing in any jurisdiction in which the conduct of its business or its ownership of property requires that it be qualified except where the failure to do so could not reasonably be expected to have a material adverse effect on Borrower's business. In connection with this Agreement, Borrower has delivered to Agent a completed certificate signed by Borrower, entitled "Perfection Certificate" (the "Perfection Certificate"). Borrower represents and warrants to Agent and each Lender that (a) Borrower's exact legal name is that indicated on the Perfection Certificate and on the signature page hereof; (b) Borrower is an organization of the type and is organized in the jurisdiction set forth in the Perfection Certificate; (c) the Perfection Certificate accurately sets forth Borrower's organizational identification number or accurately states that Borrower has none; (d) the Perfection Certificate accurately sets forth Borrower's place of business, or, if more than one, its chief executive office as well as Borrower's mailing address (if different than its chief executive office); (e) Borrower (and each of its predecessors) has not, in the past five (5) years, changed its jurisdiction of formation, organizational structure or type, or any organizational number assigned by its jurisdiction; and (f) all other information set forth on the Perfection Certificate pertaining to Borrower and each of its Subsidiaries is accurate and complete in all material respects (it being understood and agreed that Borrower may from time to time update certain information in the Perfection Certificate after the Effective Date to the extent permitted by one or more specific provisions in this Agreement).

The execution, delivery and performance by Borrower of the Loan Documents to which it is a party have been duly authorized, and do not (i) conflict with any of Borrower's organizational documents, (ii) contravene, conflict with, constitute a default under or violate any material Requirement of Law, (iii) contravene, conflict or violate any applicable order, writ, judgment, injunction, decree, determination or award of any Governmental Authority by which Borrower or any of its Subsidiaries or any of their property or assets may be bound or affected, (iv) require any action by, filing, registration, or qualification with, or Governmental Approval from, any Governmental Authority (except such Governmental Approvals which have already been obtained and are in full force and effect, and filings and registrations contemplated by this Agreement), or (v) conflict with, contravene, constitute a default or breach under, or result in or permit the termination or acceleration of, any material agreement by which Borrower is bound. Borrower is not in default under any agreement to which it is a party or by which it is bound in which the default could reasonably be expected to have a material adverse effect on Borrower's business.

5.2 Collateral. Borrower has good title to, rights in, and the power to transfer each item of the Collateral upon which it purports to grant a Lien under this Agreement and other Loan Documents, free and clear of any and all Liens except Permitted Liens. Borrower has no Collateral Accounts at or with any bank or financial institution other than SVB or SVB's Affiliates except for the Collateral Accounts described in the Perfection Certificate delivered to Agent in connection herewith and Borrower has taken such actions as are necessary to give Agent, for the ratable benefit of the Lenders, a perfected security interest therein, to the extent perfection is required in accordance with the terms of Section 6.6(b). The Accounts are bona fide, existing obligations of the Account Debtors.

No Collateral valued in excess of Five Hundred Thousand Dollars (\$500,000) (other than laptops and other portable electronic items) is in the possession of any third party bailee (such as a warehouse) except as otherwise provided in the Perfection Certificate or as permitted pursuant to Section 7.2. None of the components of the Collateral valued in excess of Five Hundred Thousand Dollars (\$500,000) (other than laptops and other portable electronic items) shall be maintained at locations other than as provided in the Perfection Certificate or as permitted pursuant to Section 7.2.

All Inventory is in all material respects of good and marketable quality, free from material defects.

Borrower is the sole owner of the Intellectual Property which it owns or purports to own except for(a) non-exclusive licenses granted to its customers in the ordinary course of business, (b) over-the-counter software that is commercially available to the public, and (c) material Intellectual Property licensed to Borrower and noted on the Perfection Certificate (as the same may be updated from time to time pursuant to Section 5.1). Each Patent which it owns or purports to own and which is material to Borrower's business is valid and enforceable, and no part of the Intellectual Property which Borrower owns or purports to own and which is material to Borrower's business has been judged invalid or unenforceable, in whole or in part. To the best of Borrower's knowledge, no claim has been made in writing alleging that any part of the Intellectual Property material to Borrower's business violates the rights of any third party except to the extent such claim would not reasonably be expected to have a material adverse effect on Borrower's business.

Except for non-exclusive licenses granted to Borrower in the ordinary course of business or as noted on the Perfection Certificate (as the same may be updated from time to time), Borrower is not a party to, nor is it bound by, any Restricted License.

- **5.3 Litigation**. Except as set forth in the Perfection Certificate dated on or around the Effective Date, there are no actions or proceedings pending or, to the knowledge of any Responsible Officer, threatened in writing by or against Borrower or any of its Subsidiaries that could reasonably be expected to involve more than, individually or in the aggregate, One Million Five Hundred Thousand Dollars (\$1,500,000.00).
- 5.4 Financial Statements; Financial Condition. All consolidated financial statements for Borrower and any of its Subsidiaries delivered to Agent and the Lenders by submission to the Financial Statement Repository or otherwise submitted to Agent and the Lenders fairly present in all material respects Borrower's consolidated financial condition as of the date thereof and Borrower's consolidated results of operations for the period covered thereby. There has not been any material deterioration in Borrower's consolidated financial condition since the date of the most recent financial statements submitted to the Financial Statement Repository or otherwise submitted to Agent and the Lenders.
- **5.5 Solvency**. The fair salable value of Borrower's consolidated assets (including goodwill minus disposition costs) exceeds the fair value of Borrower's liabilities; Borrower is not left with unreasonably small capital after the transactions in this Agreement; and Borrower is able to pay its debts (including trade debts) as they mature.
- 5.6 Regulatory Compliance. Borrower is not an "investment company" or a company "controlled" by an "investment company" under the Investment Company Act of 1940, as amended. Borrower is not engaged as one of its important activities in extending credit for margin stock (under Regulations X, T and U of the Federal Reserve Board of Governors). Borrower (a) has complied in all material respects with all Requirements of Law, and (b) has not violated any Requirements of Law the violation of which could reasonably be expected to have a material adverse effect on its business. None of Borrower's or any of its Subsidiaries' properties or assets has been used by Borrower or any Subsidiary or, to best of Borrower's knowledge, by previous Persons, in disposing, producing, storing, treating, or transporting any hazardous substance other than legally. Borrower and each of its Subsidiaries have obtained all consents, approvals and authorizations of, made all declarations or filings with, and given all notices to, all Governmental Authorities that are necessary to continue their respective businesses as currently conducted, except where failure to obtain or make such consents, declarations, filings or notices could not reasonably be expected to have a material adverse effect on Borrower's business.
- 5.7 Subsidiaries; Investments. Borrower does not own any stock, partnership, or other ownership interest or other equity securities except for Permitted Investments.
- 5.8 Tax Returns and Payments; Pension Contributions. Borrower has timely filed all required tax returns and reports (or extensions therefor), and Borrower has timely paid (or timely filed extensions for) all foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower except (a) to the extent such taxes are being contested in good faith by appropriate proceedings promptly instituted and diligently conducted, so long as such reserve or other appropriate provision, if any, as shall be required in conformity with GAAP shall have been made therefor, or (b) if such taxes, assessments, deposits and contributions do not, individually or in the aggregate, exceed Two Hundred Fifty Thousand Dollars (\$250,000).

To the extent Borrower defers payment of any contested taxes or has a tax matter that exceeds Two Hundred Fifty Thousand Dollars (\$250,000), Borrower shall (i) notify Agent in writing of the commencement of, and any material development in, the proceedings, and (ii) post bonds or take any other steps required to prevent the Governmental Authority levying such contested taxes from obtaining a Lien upon any of the Collateral that is other than a "Permitted Lien." Borrower is unaware of any claims or adjustments proposed for any of Borrower's prior tax years which could result in additional taxes becoming due and payable by Borrower in excess of Two Hundred Fifty Thousand Dollars (\$250,000). Borrower has paid all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms, and Borrower has not withdrawn from participation in, and has not permitted partial or complete termination of, or permitted the occurrence of any other event with respect to, any such plan which could reasonably be expected to result in any liability of Borrower, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency.

- **5.9** Use of Proceeds. Borrower shall use the proceeds of the Credit Extensions as working capital, to refinance existing Indebtedness of Borrower, and to fund its general business requirements and not for personal, family, household or agricultural purposes.
- 5.10 Full Disclosure. No written representation, warranty or other statement of Borrower in any certificate or written statement submitted to the Financial Statement Repository or otherwise submitted to Agent or any Lender by Borrower in connection with the Loan Documents, or the transactions contemplated thereby, as of the date such representation, warranty, or other statement was made, taken together with all such written reports, written certificates and written statements submitted to the Financial Statement Repository or otherwise submitted to Agent or any Lender by Borrower, contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements contained in the reports, certificates or written statements not misleading in light of the circumstances under which they were made (it being recognized by Agent and each Lender that the projections and forecasts provided by Borrower in good faith and based upon reasonable assumptions are not viewed as facts and that actual results during the period or periods covered by such projections and forecasts may differ from the projected or forecasted results).
- **5.11 Definition of "Knowledge."** For purposes of the Loan Documents, whenever a representation or warranty is made to Borrower's knowledge or awareness, to the "best of' Borrower's knowledge, or with a similar qualification, knowledge or awareness means the actual knowledge, after reasonable investigation, of any Responsible Officer.

6 AFFIRMATIVE COVENANTS

Borrower shall do all of the following:

6.1 Government Compliance.

- (a) Maintain its and all its Subsidiaries' legal existence and good standing in their respective jurisdictions of formation and maintain qualification in each other jurisdiction in which the failure to so qualify would reasonably be expected to have a material adverse effect on Borrower's business or operations. Borrower shall comply, and have each Subsidiary comply, in all material respects, with all laws, ordinances and regulations to which it is subject.
- (b) Obtain all of the Governmental Approvals necessary for the performance by Borrower of its obligations under the Loan Documents to which it is a party and the grant of a security interest to Agent, for the ratable benefit of the Lenders, in all of its property. Borrower shall promptly provide copies of any such obtained Governmental Approvals to Agent.
- **6.2 Financial Statements, Reports, Certificates**. Provide Agent and each Lender with the following by submitting to the Financial Statement Repository:
- (a) Monthly Financial Statements. As soon as available, but no later than thirty (30) days after the last day of each month, a company-prepared consolidated and consolidating (if applicable) balance sheet and income statement covering Borrower's consolidated operations and key performance indicators for such month in a form acceptable to Agent (the "Monthly Financial Statements");

- (b) Monthly Compliance Statement. Within thirty (30) days after the last day of each month and together with the Monthly Financial Statements, a completed Compliance Statement confirming that, as of the end of such month, Borrower was in full compliance with all of the terms and conditions of this Agreement, and setting forth calculations showing compliance with the financial covenants (if any) set forth in this Agreement and such other information as Agent or the Lenders may reasonably request;
- (c) <u>Board Projections</u>. As soon as available but no later than the earlier of forty-five (45) days after the last day of the fiscal year of Borrower or forty-five (45) days after approval by the Board, and within seven (7) days of any Board-approved updates or amendments thereto, (1) annual operating budgets (including income statements, balance sheets and cash flow statements, by month) for the upcoming fiscal year of Borrower, and (2) annual financial projections for the following fiscal year (on a monthly or quarterly basis), in each case as approved by the Board and commensurate in form and substance with those provided to the Board, together with any related business forecasts used in the preparation of such annual financial projections;
- (d) <u>Annual Audited Financial Statements</u>. As soon as available, and in any event within two hundred ten (210) days after the last day of Borrower's fiscal year, audited consolidated financial statements prepared under GAAP, consistently applied, together with an unqualified opinion (provided, however, Borrower's unqualified opinion on financial statements may contain a qualification as to going concern typical for venture backed companies similar to Borrower) on the financial statements from an independent certified public accounting firm reasonably acceptable to Agent;
- (e) <u>Board Decks</u> Within forty-five (45) days of presentation to the Board, copies of Board decks and similar materials for standard scheduled board meetings, provided that such Board decks and materials may be redacted by Borrower to exclude (i) documents and matters that may be subject to attorney-client privilege; (ii) information provided in connection with or for closed sessions of the Board or its committees; (iii) information relating to Lenders, and (iv) any material confidential proprietary information and trade secrets;
- (f) 409A Valuation. Not less frequently than annually, within thirty (30) days after approval by the Board, 409A Valuation reports conducted with respect to Borrower's common stock;
- (g) Other Statements. Within five (5) Business Days of delivery by Borrower, copies of all material statements, reports and notices made generally available by borrower to Borrower's security holders or to any holders of Subordinated Debt (but excluding, for the avoidance of doubt, one-off or specific reporting requests);
- (h) <u>SEC Filings</u>. In the event that Borrower becomes subject to the reporting requirements under the Exchange Act, within five (5) days of filing, copies of all periodic and other reports, proxy statements and other materials filed by Borrower and/or any Guarantor with the SEC, any Governmental Authority succeeding to any or all of the functions of the SEC or with any national securities exchange, or distributed to its shareholders, as the case may be. Documents required to be delivered pursuant to the terms hereof (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date on which Borrower posts such documents, or provides a link thereto, on Borrower's website on the internet at Borrower's website address; provided, however, Borrower shall promptly notify Agent and the Lenders in writing (which may be by electronic mail) of the posting of any such documents;
- (i) <u>Legal Action Notice</u>. A prompt report of any legal actions pending or threatened in writing against Borrower or any of its Subsidiaries that could reasonably be expected to result in damages or costs to Borrower or any of its Subsidiaries of, individually or in the aggregate, One Million Five Hundred Thousand Dollars (\$1,500,000.00) or more (except for items set forth in the Perfection Certificate dated on or around the Effective Date);
- (j) <u>Governmental Approvals</u>. Within five (5) Business Days after the same are sent by Borrower or received by Borrower, copies of all correspondence, reports, documents and other filings with any Governmental Authority regarding compliance with or maintenance of Governmental Approvals or Requirements of Law that could reasonably be expected to have a material adverse effect on any of the Governmental Approvals or otherwise on the operations of Borrower or any of its Subsidiaries;

- (k) <u>Beneficial Ownership Information</u>. Prompt written notice of any changes to the beneficial ownership information set out in Section 2 of the Perfection Certificate. Borrower understands and acknowledges that each Lender relies on such true, accurate and up-to-date beneficial ownership information to meet such Lender's regulatory obligations to obtain, verify and record information about the beneficial owners of its legal entity customers; and
- (1) Other Financial Information. Other financial information regarding Borrower or compliance with the terms of any Loan Documents as reasonably requested by Agent or any Lender.

Any submission by Borrower of a Compliance Statement, or any other financial statement submitted to the Financial Statement Repository pursuant to this Section 6.2 or otherwise submitted to Agent and the Lenders shall be deemed to be a representation by Borrower that (a) as of the date of such Compliance Statement, or other financial statement, the information and calculations set forth therein are true, accurate and correct in all material respect, (b) as of the end of the compliance period set forth in such submission, Borrower is in compliance with all required covenants except as noted in such Compliance Statement or other financial statement, as applicable; (c) as of the date of such submission, no Events of Default have occurred or are continuing, except as noted in such Compliance Statement or other financial statement, as applicable; (d) all representations and warranties other than any representations or warranties that are made as of a specific date in Section 5 are true and correct in all material respects as of the date of such submission except as noted in such Compliance Statement, or other financial statement, as applicable; (e) as of the date of such submission, Borrower and each of its Subsidiaries has timely filed all required tax returns and reports, and Borrower has timely paid all foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower except, in each case, as otherwise permitted pursuant to the terms of Section 5.8; and (f) as of the date of such submission, no Liens have been levied or claims made against Borrower or any of its Subsidiaries relating to unpaid employee payroll or benefits of which Borrower has not previously provided written notification to Agent and the Lenders.

- **6.3 Taxes; Pensions**. Timely file and require each of its Subsidiaries to timely file, or timely file all required extensions therefor, all required tax returns and reports and timely pay, and require each of its Subsidiaries to timely pay, all foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower and each of its Subsidiaries, except for deferred payment of any taxes contested pursuant to the terms of Section 5.8 hereof, and other taxes which are less than Two Hundred Fifty Thousand Dollars (\$250,000), and shall deliver to Agent, on reasonable demand, appropriate certificates attesting to such payments, and pay all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms.
- **6.4 Inventory; Returns.** Keep all Inventory in good and marketable condition, free from material defects. Returns and allowances between Borrower and its Account Debtors shall follow Borrower's customary practices as they exist at the Effective Date. Borrower must promptly notify Agent and the Lenders of all returns, recoveries, disputes and claims that involve more than One Hundred Thousand Dollars (\$100,000.00) as to any single claim.

6.5 Insurance.

(a) Keep its business and the Collateral insured for risks and in amounts standard for companies in Borrower's industry and location and as Agent may reasonably request. Insurance policies shall be in a form, with financially sound and reputable insurance companies that are not Affiliates of Borrower, and in amounts that are reasonably satisfactory to Agent. All property policies shall have a lender's loss payable endorsement showing Agent as the sole lender loss payee. All liability policies shall show, or have endorsements showing, Agent as an additional insured. Agent shall be named as lender loss payee and/or additional insured with respect to any such insurance providing coverage in respect of any Collateral. Upon Borrower's request, Agent will confirm that the insurance that Borrower maintains is satisfactory to Agent as of such date upon receipt of evidence of insurance from Borrower.

- (b) Ensure that proceeds payable under any property policy are, at Agent's option, payable to Agent for the ratable benefit of the Lenders on account of the Obligations. Notwithstanding the foregoing, (a) so long as no Event of Default has occurred and is continuing, Borrower shall have the option of applying the proceeds of any casualty policy up to Two Million Dollars (\$2,000,000) in the aggregate per fiscal year for all losses under all casualty policies in any one (1) year, toward the replacement or repair of destroyed or damaged property; provided that any such replaced or repaired property (i) shall be of equal or like value as the replaced or repaired Collateral and (ii) shall be deemed Collateral in which Agent has been granted a first priority security interest (subject to Permitted Liens which may be superior under the Senior Loan Agreement or by operation of law), and (b) after the occurrence and during the continuance of an Event of Default, all proceeds payable under such casualty policy shall, at the option of Agent, be payable to Agent on account of the Obligations.
- (c) At Agent's reasonable request, Borrower shall deliver certified copies of insurance policies and evidence of all premium payments. Each provider of any such insurance required under this Section 6.5 shall agree, by endorsement upon the policy or policies issued by it or by independent instruments furnished to Agent, that it will give Agent twenty (20) days prior written notice before any such policy or policies shall be canceled, provided that only ten (10) days prior notice is required for cancellation due to non-payment of premium. If Borrower fails to obtain insurance as required under this Section 6.5 or to pay any amount or furnish any required proof of payment to third persons and Agent, Agent may make all or part of such payment or obtain such insurance policies required in this Section 6.5, and take any action under the policies Agent deems prudent.

6.6 Operating Accounts

- (a) Borrower, and any Subsidiary of Borrower shall maintain all of its operating accounts and excess cash with SVB or SVB's Affiliates; provided, however, that Borrower may maintain its accounts with merchant processors (including, without limitation, PayPal, Stripe, and other similar processors) ("Merchant Processing Accounts"), so long as Borrower transfers all the funds which are in excess of One Hundred Thousand Dollars (\$100,000) in any such account or Three Hundred Thousand Dollars (\$300,000) in the aggregate in an all such accounts to an account maintained with SVB within five (5) Business Days of any such account exceeding One Hundred Thousand Dollars (\$100,000) individually or such accounts exceeding Three Hundred Thousand Dollars (\$300,000) in the aggregate.
- (b) In addition to and without limiting the restrictions in (a), Borrower shall provide Agent five (5) days prior written notice before establishing any Collateral Account at or with any bank or financial institution other than Agent or Agent's Affiliates. For each Collateral Account that Borrower at any time maintains, Borrower shall cause the applicable bank or financial institution (other than Agent) at or with which any Collateral Account is maintained to execute and deliver a Control Agreement or other appropriate instrument with respect to such Collateral Account to perfect Agent's Lien in such Collateral Account in accordance with the terms hereunder which Control Agreement may not be terminated without the prior written consent of Agent. The provisions of the previous sentence shall not apply to (i) deposit accounts exclusively used for payroll, payroll taxes, and other employee wage and benefit payments to or for the benefit of Borrower's employees and identified to Agent by Borrower as such or (ii) the Merchant Processing Accounts so long as Borrower transfers applicable funds therein to an account maintained with SVB in accordance with Section 6.6(a).
- (c) Borrower, and any Subsidiary of Borrower shall obtain any business credit card exclusively from SVB, except as permitted in the defined term "Permitted Indebtedness" part (g) of this Agreement.

6.7 Protection and Registration of Intellectual Property Rights.

(a) Use commercially reasonable efforts to protect, defend and maintain the validity and enforceability of its Intellectual Property that is material to Borrower's business; (ii) promptly advise Agent in writing of material infringements or any other event that could reasonably be expected to materially and adversely affect the value of its material Intellectual Property; and (iii) not allow any Intellectual Property material to Borrower's business to be abandoned, forfeited or dedicated to the public without Agent's written consent.

- (b) To the extent not already disclosed to Agent, if Borrower (i) obtains any Patent, registered Trademark, registered Copyright, registered mask work, or any pending application for any of the foregoing, whether as owner, licensee or otherwise, or (ii) applies for any Patent or the registration of any Trademark, then (A) as to Copyrights, Borrower shall promptly provide written notice thereof to Agent and (B) as to any other new Intellectual Property, Borrower shall provide notice to Agent in the Compliance Statement next due, and in each case shall execute such intellectual property security agreements and other documents and take such other actions as Agent may request in its good faith business judgment to perfect and maintain a first priority perfected security interest in favor of Agent in such property (subject to Permitted Liens which may be superior under the Senior Loan Agreement or by operation of law). If Borrower decides to register any Copyrights or mask works in the United States Copyright Office, Borrower shall: (x) provide Agent with at least seven (7) days prior written notice of Borrower's intent to register such Copyrights or mask works together with a copy of the application it intends to file with the United States Copyright Office (excluding exhibits thereto); (y) execute an intellectual property security agreement and such other documents and take such other actions as Agent may request in its good faith business judgment to perfect and maintain a first priority perfected security interest in favor of Agent (subject to Permitted Liens which may be superior under the Senior Loan Agreement or by operation of law) in the Copyrights or mask works intended to be registered with the United States Copyright Office; and (z) record such intellectual property security agreement with the United States Copyright Office contemporaneously with filing the Copyright or mask work application(s) with the United States Copyright Office. Upon request of Agent, Borrower shall provide to Agent copies of all applications that it files for Patents or for the registration of Trademarks, Copyrights or mask works, together with evidence of the recording of the intellectual property security agreement reasonably requested by Agent and required for Agent to perfect and maintain a first priority perfected security interest in such property (subject to Permitted Liens which may be superior under the Senior Loan Agreement or by operation of law).
- (c) Provide written notice to Agent within ten (10) days of entering or becoming bound by any Restricted License (other than over-the-counter software that is commercially available to the public). Borrower shall take such commercially reasonable steps as Agent reasonably requests to obtain the consent of, or waiver by, any person whose consent or waiver is necessary for (i) any Restricted License to be deemed "Collateral" and for Agent to have a security interest in it that might otherwise be restricted or prohibited by law or by the terms of any such Restricted License, whether now existing or entered into in the future, and (ii) Agent to have the ability in the event of a liquidation of any Collateral to dispose of such Collateral in accordance with Agent's rights and remedies under this Agreement and the other Loan Documents.
- **6.8 Litigation Cooperation.** From the date hereof and continuing through the termination of this Agreement, make available to Agent, without expense to Agent or any Lender and upon at least three (3) Business Day's notice (provided no notice is required if an Event of Default has occurred and is continuing), Borrower and its officers, employees and agents and Borrower's books and records, to the extent that Agent and/or the Lenders may deem them reasonably necessary to prosecute or defend any third-party suit or proceeding instituted by or against Agent and/or any Lender with respect to any Collateral or relating to Borrower.
- 6.9 Access to Collateral; Books and Records. Allow Agent, or its agents, at reasonable times, on five (5) Business Days' notice (provided no notice is required if an Event of Default has occurred and is continuing), to inspect the Collateral and audit and copy Borrower's Books. Such inspections or audits shall be conducted no more often than once every twelve (12) months unless an Event of Default has occurred and is continuing in which case such inspections and audits shall occur as often as Agent shall determine is necessary. The foregoing inspections and audits shall be at Borrower' expense and the charge therefor shall be One Thousand Dollars (\$1,000.00) per person per day (or such higher amount as shall represent Agent's then-current standard charge for the same), plus reasonable out-of-pocket expenses. In the event Borrower and Agent schedule an audit more than eight (8) days in advance, and Borrower cancels or seeks to reschedule the audit with less than eight (8) days written notice to Agent, then (without limiting any of Agent's or any Lender's rights or remedies) Borrower shall pay Agent a fee of Two Thousand Dollars (\$2,000.00) plus any documented out-of-pocket expenses incurred by Agent to compensate Agent for the anticipated costs and expenses of the cancellation or rescheduling.
- **6.10 Further Assurances**. Execute any further instruments and take further action as Agent reasonably requests to perfect or continue Agent's Lien in the Collateral or to effect the purposes of this Agreement.

- **6.11 Post-Closing Requirements**. Deliver or cause to be delivered to Agent, in form and substance satisfactory to Agent, within thirty (30) days after the Effective Date (or in the case of clause (d) below within sixty (60) days of the Effective Date):
- (a) duly executed Control Agreements for all Collateral Accounts of Borrower except those expressly excluded from such requirement pursuant to Section 6.6(b);
 - (b) evidence that the insurance policies and endorsements required by Section 6.5 hereof are in full force and effect;
- (c) written evidence (including a filed UCC termination statement) that the Liens of Venture Lending & Leasing VIII, Inc. on the assets of Darby Smart, Inc. acquired by Borrower have been terminated and released;
- (d) duly executed landlord's consents in favor of Agent for each of Borrower's leased locations containing in excess of One Million Dollars (\$1,000,000) of Borrower's assets or property, by the respective landlord thereof; and
- (e) a duly executed termination letter for Borrower's securities account with SVB Asset Management identified as Account Number 19-SV1474 containing the wet ink "pdf" signature of TRIPLEPOINT VENTURE GROWTH BDC CORP.

7 NEGATIVE COVENANTS

Borrower shall not do any of the following without the prior written consent of the Lenders:

- 7.1 Dispositions. Convey, sell, lease, transfer, assign, or otherwise dispose of (including, without limitation, pursuant to a Division) (collectively, "Transfer"), or permit any of its Subsidiaries to Transfer, all or any part of its business or property, except for Transfers (a) of Inventory in the ordinary course of business; (b) of worn-out, surplus, or obsolete Equipment that is, in the reasonable judgment of Borrower, no longer economically practicable to maintain or useful in the ordinary course of business of Borrower; (c) consisting of Permitted Liens and Permitted Investments; (d) consisting of the sale or issuance of any stock of Borrower permitted under Section 7.2 of this Agreement; (e) consisting of Borrower's use or transfer of money or Cash Equivalents in the ordinary course of its business for the payment of ordinary course business expenses in a manner that is not prohibited by the terms of this Agreement or the other Loan Documents; (f) of non-exclusive licenses for the use of the property of Borrower or its Subsidiaries in the ordinary course of business; (g) of other property not to exceed Five Hundred Thousand Dollars (\$500,000) in the aggregate in any twelve (12) month period.
- 7.2 Changes in Business, Management, Control, or Business Locations. (a) Engage in or permit any of its Subsidiaries to engage in any business other than the businesses currently engaged in by Borrower and such Subsidiary, as applicable, or reasonably related thereto; (b) liquidate or dissolve (except that a Subsidiary may liquidate or dissolve provided simultaneous therewith all of its assets are transferred to Borrower); (c) fail to provide notice to Agent and Lenders of any Key Person departing from or ceasing to be employed by Borrower within five (5) Business Days after such Key Person's departure from Borrower; or (d) permit or suffer any Change in Control.

Borrower shall not, without at least fifteen (15) days prior written notice to Agent: (A) add any new offices or business locations, including warehouses (unless such new offices or business locations contain less than Five Hundred Thousand Dollars (\$500,000) in Borrower's assets or property) or deliver any portion of the Collateral valued, individually or in the aggregate, in excess of Five Hundred Thousand Dollars (\$500,000) to a bailee at a location other than to a bailee and at a location already disclosed in the Perfection Certificate, (B) change its jurisdiction of organization, (C) change its organizational structure or type, (D) change its legal name, or (E) change any organizational number (if any) assigned by its jurisdiction of organization. If Borrower intends to add any new offices or business locations, including warehouses, containing in excess of One Million Dollars (\$1,000,000) of Borrower's assets or property, then Borrower will first receive the written consent of Agent, and the landlord of any such new offices or business locations, including warehouses, shall execute and deliver a landlord consent in form and substance satisfactory to Agent. If Borrower intends to deliver any portion of the Collateral valued, individually or in the aggregate, in excess of One Million Dollars (\$1,000,000) to a bailee, and Agent and such bailee are not already parties to a bailee agreement governing both the Collateral and the location to which Borrower intends to deliver the Collateral, then Borrower will first receive the written consent of Agent, and such bailee shall execute and deliver a bailee agreement in form and substance satisfactory to Agent.

- 7.3 Mergers or Acquisitions. Merge or consolidate, or permit any of its Subsidiaries to merge or consolidate, with any other Person, or acquire, or permit any of its Subsidiaries to acquire, all or substantially all of the capital stock or property of another Person (including, without limitation, by the formation of any Subsidiary or pursuant to a Division) except if (A) Borrower has complied with the notice requirements applicable to prepayments hereunder, and (B) prior to or contemporaneously with the closing of such transaction, all Obligations are paid in full, and all of Bank's obligations to lend to Borrower are terminated. A Subsidiary may merge or consolidate into another Subsidiary or into Borrower; provided that, if a Borrower is involved in such transaction, such Borrower shall be the surviving legal entity.
 - 7.4 Indebtedness. Create, incur, assume, or be liable for any Indebtedness, or permit any Subsidiary to do so, other than Permitted Indebtedness.
- 7.5 Encumbrance. Create, incur, allow, or suffer any Lien on any of its property, or assign or convey any right to receive income, including the sale of any Accounts, or permit any of its Subsidiaries to do so, except, in each case as to the foregoing, for Permitted Liens, permit any Collateral not to be subject to the first priority security interest granted herein, other than Permitted Liens which may be superior under the Senior Loan Agreement or by operation of law, or enter into any agreement, document, instrument or other arrangement (except with or in favor of Agent, for the ratable benefit of the Lenders or pursuant to the Senior Loan Agreement) with any Person which directly or indirectly prohibits or has the effect of prohibiting Borrower or any Subsidiary from assigning, mortgaging, pledging, granting a security interest in or upon, or encumbering any of Borrower's or any Subsidiary's Intellectual Property, except as is otherwise permitted in (i) Section 7.1 hereof; (ii) in the definition of "Permitted Liens" herein. (iii) the licensors of in-licensed property with respect to such property; and (iv) the lessors of specific leased equipment with respect to such specific leased equipment.
 - 7.6 Maintenance of Collateral Accounts. Maintain any Collateral Account except pursuant to the terms of Section 6.6(b) hereof.
- 7.7 Distributions; Investments. (a) Pay any dividends or make any distribution or payment or redeem, retire or purchase any capital stock or other equity interests; provided that (i) Borrower may convert any of its convertible securities into other securities pursuant to the terms of such convertible securities or otherwise in exchange thereof, (ii) Borrower may pay dividends solely in common stock and repurchase de minimis stock in lieu of issuing fractional shares; (iii) Borrower may repurchase the stock of former employees, directors or consultants (including by the cancellation of indebtedness owed by such former Persons) pursuant to stock repurchase agreements so long as an Event of Default does not exist at the time of such repurchase and would not exist after giving effect to such repurchase, provided that the aggregate amount of all such repurchases does not exceed Two Hundred Fifty Thousand Dollars (\$250,000) per fiscal year; (iv) Borrower may distribute equity securities to former or current employees, officers, consultants or directors on the exercise of employee issuances pursuant to Borrower's stock incentive plan approved by the Board; and (v) Borrower may repurchase common shares of Borrower's stock from its employees, consultants and directors in connection with a bona fide equity financing of Borrower provided that (A) Borrower sells common or preferred shares to investors for an aggregate price equal to the repurchase price of such common shares repurchased by Borrower, and (B) no Event of Default exists at the time of such series of transactions and would not exist after giving effect to such transactions; or (b) directly or indirectly make any Investment (including, without limitation, by the formation of any Subsidiary) other than Permitted Investments, or permit any of its Subsidiaries to do so.
- 7.8 Transactions with Affiliates. Directly or indirectly enter into or permit to exist any material transaction with any Affiliate of Borrower, except for (i) transactions that are in the ordinary course of Borrower's business, and that are upon fair and reasonable terms that are no less favorable to Borrower than would be obtained in an arm's length transaction with a non-affiliated Person, (ii) sales of Borrower's equity securities to, and Subordinated Debt financing transactions with, Borrower's investors (in each case to the extent not otherwise prohibited under this Agreement), (iii) reasonable and customary compensation arrangements and benefit plans for officers and other employees of Borrower and its Subsidiaries entered into or maintained in the ordinary course of business, (iv) transactions permitted pursuant to the terms of Sections 7.2 and 7.7 hereof, and (v) compensation of outside directors in the ordinary course of business.

7.9 Subordinated Debt. (a) Make or permit any payment on any Subordinated Debt, except under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject, or (b) amend any provision in any document relating to the Subordinated Debt which would increase the amount thereof, provide for earlier or greater principal, interest, or other payments thereon, or adversely affect the subordination thereof to Obligations owed to Agent and the Lenders, except under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject.

7.10 Compliance. Become an "investment company" or a company controlled by an "investment company", under the Investment Company Act of 1940, as amended, or undertake as one of its important activities extending credit to purchase or carry margin stock (as defined in Regulation U of the Board of Governors of the Federal Reserve System), or use the proceeds of any Credit Extension for that purpose; fail to meet the minimum funding requirements of ERISA, permit a Reportable Event or Prohibited Transaction, as defined in ERISA, to occur; fail to comply with the Federal Fair Labor Standards Act or violate any other law or regulation, if the failure or violation could reasonably be expected to have a material adverse effect on Borrower's business, or permit any of its Subsidiaries to do so; withdraw or permit any Subsidiary to withdraw from participation in, permit partial or complete termination of, or permit the occurrence of any other event with respect to, any present pension, profit sharing and deferred compensation plan which could reasonably be expected to result in any liability of Borrower, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency.

8 EVENTS OF DEFAULT

Any one of the following shall constitute an event of default (an "Event of Default") under this Agreement:

8.1 Payment Default. Borrower fails to (a) make any payment of principal or interest on any Credit Extension when due, or (b) pay any other Obligations within three (3) Business Days after such Obligations are due and payable (which three (3) Business Day cure period shall not apply to payments due on the Term Loan Maturity Date). During the cure period, the failure to make or pay any payment specified under clause (b) hereunder is not an Event of Default (but no Credit Extension will be made during the cure period);

8.2 Covenant Default.

- (a) Borrower fails or neglects to perform any obligation in Sections 6.2, 6.3, 6.5, 6.6, 6.7(b), or 6.9, or violates any covenant in Section 7; or
- (b) Borrower fails or neglects to perform, keep, or observe any other term, provision, condition, covenant or agreement contained in this Agreement or any Loan Documents, and as to any default (other than those specified in this Section 8) under such other term, provision, condition, covenant or agreement that can be cured, has failed to cure the default within ten (10) days after the occurrence thereof; provided, however, that if the default cannot by its nature be cured within the ten (10) day period or cannot after diligent attempts by Borrower be cured within such ten (10) day period, and such default is likely to be cured within a reasonable time, then Borrower shall have an additional period (which shall not in any case exceed thirty (30) days) to attempt to cure such default, and within such reasonable time period the failure to cure the default shall not be deemed an Event of Default (but no Credit Extensions shall be made during such cure period). Cure periods provided under this section shall not apply, among other things, to financial covenants, if any, or any other covenants set forth in clause (a) above;
 - 8.3 Material Adverse Change. A Material Adverse Change occurs;

8.4 Attachment; Levy; Restraint on Business.

(a) (i) The service of process seeking to attach, by trustee or similar process, any funds of Borrower or of any entity under the control of Borrower (including a Subsidiary) in excess Five Hundred Thousand Dollars (\$500,000); or (ii) a notice of lien or levy is filed against any of Borrower's material assets by any Governmental Authority, and the same under subclauses (i) and (ii) hereof are not, within ten (10) days after the occurrence thereof, discharged or stayed (whether through the posting of a bond or otherwise); provided, however, no Credit Extensions shall be made during any ten (10) day cure period; or

- (b) (i) any material portion of Borrower's assets is attached, seized, levied on, or comes into possession of a trustee or receiver, or (ii) any court order enjoins, restrains, or prevents Borrower from conducting all or any material part of its business;
- **8.5 Insolvency.** (a) Borrower or any of its Subsidiaries is unable to pay its debts (including trade debts) as they become due or otherwise becomes insolvent as described in Section 5.6; (b) Borrower or any of its Subsidiaries begins an Insolvency Proceeding; or (c) an Insolvency Proceeding is begun against Borrower or any of its Subsidiaries and is not dismissed or stayed within forty-five (45) days (but no Credit Extensions shall be made while any of the conditions described in clause (a) exist and/or until any Insolvency Proceeding is dismissed);
- 8.6 Other Agreements. There is, under any agreement to which Borrower is a party with a third party or parties, (a) any default resulting in a right by such third party or parties, whether or not exercised, to accelerate the maturity of any Indebtedness in an amount individually or in the aggregate in excess of Seven Hundred Fifty Thousand Dollars (\$750,000); or (b) any breach or default by Borrower, the result of which could reasonably be expected to have a material adverse effect on Borrower's business; provided, however, that the Event of Default under this Section 8.6 caused by the occurrence of a breach or default under such other agreement shall be automatically cured or waived for purposes of this Agreement upon Agent receiving written notice from the party asserting such breach or default of such cure or waiver of the breach or default under such other agreement, if at the time of such cure or waiver under such other agreement (x) Bank has not declared an Event of Default under this Agreement and/or exercised any rights with respect thereto; (y) any such cure or waiver does not result in an Event of Default under any other provision of this Agreement or any Loan Documents; and (z) in connection with any such cure or waiver under such other agreement, the terms of any agreement with such third party are not modified or amended in any manner which could in the good faith business judgment of Bank be materially less advantageous to Borrower; further provided that, notwithstanding the foregoing, or anything to the contrary herein or in the Loan Documents, any failure by Borrower to comply with any financial covenant set forth in Section 6.9 of the Senior Loan Agreement (or any other financial covenant from time to time in the Senior Loan Agreement) shall not in and of itself at any time constitute an Event of Default under this Agreement;
- **8.7 Judgments; Penalties.** One or more fines, penalties or final judgments, orders or decrees for the payment of money in an amount, individually or in the aggregate, of at least Seven Hundred Fifty Thousand Dollars (\$750,000) (not covered by independent third-party insurance as to which liability has been accepted by such insurance carrier) shall be rendered against Borrower by any Governmental Authority, and the same are not, within ten (10) days after the entry, assessment or issuance thereof, discharged, satisfied, or paid, or after execution thereof, stayed or bonded pending appeal, or such judgments are not discharged prior to the expiration of any such stay (provided that no Credit Extensions will be made prior to the satisfaction, payment, discharge, stay, or bonding of such fine, penalty, judgment, order or decree);
- **8.8 Misrepresentations**. Borrower or any Person acting for Borrower makes any representation, warranty, or other statement now or later in this Agreement, any Loan Document or in any writing delivered to Agent or any Lender or to induce Agent or any Lender to enter this Agreement or any Loan Document, and such representation, warranty, or other statement is incorrect in any material respect when made;
- **8.9 Subordinated Debt.** Any document, instrument, or agreement evidencing any Subordinated Debt shall for any reason be revoked or invalidated or otherwise cease to be in full force and effect (unless permitted by the terms of such agreement), any Person shall be in breach thereof or contest in any manner the validity or enforceability thereof or deny that it has any further liability or obligation thereunder, or the Obligations shall for any reason be subordinated or shall not have the priority contemplated by this Agreement or any applicable subordination or intercreditor agreement; or
- **8.10 Senior Loan Agreement.** The occurrence of an Event of Default (as defined in the Senior Loan Agreement) under the Senior Loan Agreement; provided that, notwithstanding the foregoing, or anything to the contrary herein or in the Loan Documents, any failure by Borrower to comply with any financial covenant set forth in Section 6.9 of the Senior Loan Agreement (or any other financial covenant from time to time in the Senior Loan Agreement) shall not in and of itself at any time constitute an Event of Default under this Agreement.

9 RIGHTS AND REMEDIES

- **9.1 Rights and Remedies.** Upon the occurrence and during the continuance of an Event of Default, Agent, in accordance with the Lender Intercreditor Agreement or, if such rights and remedies are not addressed in the Lender Intercreditor Agreement, as directed by Lenders having a majority of the Obligations, may, without notice or demand, do any or all of the following:
- (a) declare all Obligations immediately due and payable (but if an Event of Default described in Section 8.5 occurs all Obligations are immediately due and payable without any action by Agent or any Lender);
- (b) stop advancing money or extending credit for Borrower's benefit under this Agreement or under any other agreement among Borrower, Agent, and/or any Lenders;
- (c) demand that Borrower (i) deposit cash with SVB in an amount equal to at least (A) one hundred five percent (105.0%) of the Dollar Equivalent of the aggregate face amount of all Letters of Credit denominated in Dollars remaining undrawn, and (B) one hundred ten percent (110.0%) of the Dollar Equivalent of the aggregate face amount of all Letters of Credit denominated in a Foreign Currency remaining undrawn (plus, in each case, all interest, fees, and costs due or to become due in connection therewith (as estimated by SVB in its good faith business judgment)), to secure all of the Obligations relating to such Letters of Credit, as collateral security for the repayment of any future drawings under such Letters of Credit, and Borrower shall forthwith deposit and pay such amounts, and (ii) pay in advance all letter of credit fees scheduled to be paid or payable over the remaining term of any Letters of Credit;
 - (d) terminate any FX Contracts;
- (e) verify the amount of, demand payment of and performance under, and collect any Accounts and General Intangibles, settle or adjust disputes and claims directly with Account Debtors for amounts on terms and in any order that Agent and/or the Lenders consider advisable, and notify any Person owing Borrower money of Agent's security interest in such funds. Borrower shall collect all such payments in trust for Agent, for the ratable benefit of the Lenders and, if requested by Agent, immediately deliver the payments to Agent, for the ratable benefit of the Lenders in the form received from the Account Debtor, with proper endorsements for deposit;
- (f) make any payments and do any acts Agent or any Lender considers necessary or reasonable to protect the Collateral and/or its security interest in the Collateral. Borrower shall assemble the Collateral if Agent requests and make it available as Agent designates. Agent may peaceably enter premises where the Collateral is located, take and maintain possession of any part of the Collateral, and pay, purchase, contest, or compromise any Lien which appears to be prior or superior to its security interest and pay all expenses incurred. Borrower grants Agent a license to enter and occupy any of its premises, without charge by Borrower, to exercise any of Agent's rights or remedies;
- (g) apply to the Obligations then due (i) any balances and deposits of Borrower it holds, or (ii) any amount held by Agent owing to or for the credit or the account of Borrower;
- (h) ship, reclaim, recover, store, finish, maintain, repair, prepare for sale, advertise for sale, and sell the Collateral. Agent, for the benefit of the Lenders is hereby granted a non-exclusive, royalty-free license or other right to use, without charge, Borrower's labels, Patents, Copyrights, mask works, rights of use of any name, trade secrets, trade names, Trademarks, and advertising matter, or any similar property as it pertains to the Collateral, in completing production of, advertising for sale, and selling any Collateral and, in connection with Agent's exercise of its rights under this Section, Borrower's rights under all licenses and all franchise agreements inure to Agent, for the ratable benefit of the Lenders;
- (i) place a "hold" on any account maintained with Agent or Lenders and/or deliver a notice of exclusive control, any entitlement order, or other directions or instructions pursuant to any Control Agreement or similar agreements providing control of any Collateral;

- (j) demand and receive possession of Borrower's Books; and
- (k) exercise all rights and remedies available to Agent and the Lenders under the Loan Documents or at law or equity, including all remedies provided under the Code (including disposal of the Collateral pursuant to the terms thereof).

SVB's right of payment, lien priority and ability to exercise rights and remedies, in each case under this Agreement, shall be subordinate to its right of payment, lien priority and ability to exercise rights and remedies, in each case under the Senior Loan Agreement.

- 9.2 Power of Attorney. Borrower hereby irrevocably appoints Agent, for the benefit of the Lenders, as its lawfulattorney-in-fact, exercisable upon the occurrence and during the continuance of an Event of Default to: (a) endorse Borrower's name on any checks, payment instruments, or other forms of payment or security; (b) sign Borrower's name on any invoice or bill of lading for any Account or drafts against Account Debtors; (c) demand, collect, sue, and give releases to any Account Debtor for monies due, settle and adjust disputes and claims about the Accounts directly with Account Debtors, and compromise, prosecute, or defend any action, claim, case, or proceeding about any Collateral (including filing a claim or voting a claim in any bankruptcy case in Agent's or Borrower's name, as Agent chooses); (d) make, settle, and adjust all claims under Borrower's insurance policies; (iv) pay, contest or settle any Lien, charge, encumbrance, security interest, or other claim in or to the Collateral, or any judgment based thereon, or otherwise take any action to terminate or discharge the same; and (f) transfer the Collateral into the name of Agent, for the ratable benefit of the Lenders, or a third party as the Code permits. Borrower hereby appoints Agent, for the benefit of the Lenders, as its lawful attorney-in-fact to sign Borrower's name on any documents necessary to perfect or continue the perfection of Agent's, for the benefit of the Lenders, security interest in the Collateral regardless of whether an Event of Default has occurred until all Obligations (other than inchoate indemnity or other obligations which, by their terms, survive termination of this Agreement) have been satisfied in full and the Loan Documents have been terminated. Agent's foregoing appointment as Borrower's attorney in fact, and all of Agent's rights and powers, coupled with an interest, are irrevocable until all Obligations (other than inchoate indemnity or other obligations which, by their terms, survive termination of this Agreement) hav
- 9.3 Protective Payments. If Borrower fails to obtain the insurance called for by Section 6.5 or fails to pay any premium thereon or fails to pay any other amount which Borrower is obligated to pay under this Agreement or any other Loan Document or which may be required to preserve the Collateral, Agent may obtain such insurance or make such payment, and all amounts so paid by Agent are Lenders' Expenses and immediately due and payable, bearing interest at the then highest rate applicable to the Obligations, and secured by the Collateral. Agent will make reasonable efforts to provide Borrower with notice of Agent obtaining such insurance at the time it is obtained or within a reasonable time thereafter. No payments by Agent are deemed an agreement to make similar payments in the future or Agent's and/or Lenders' waiver of any Event of Default.
- 9.4 Application of Payments and Proceeds Upon Default. If an Event of Default has occurred and is continuing, Agent shall have the right to apply in any order any funds in its possession, whether from Borrower's account balances, payments, proceeds realized as the result of any collection of Accounts or other disposition of the Collateral, or otherwise, to the Obligations. Agent shall pay any surplus to Borrower by credit to the Designated Deposit Account or to other Persons legally entitled thereto; Borrower shall remain liable to Agent and the Lenders for any deficiency. If Agent, directly or indirectly, enters into a deferred payment or other credit transaction with any purchaser at any sale of Collateral, Agent shall have the option, exercisable at any time, of either reducing the Obligations by the principal amount of the purchase price or deferring the reduction of the Obligations until the actual receipt by Agent of cash therefor.
- 9.5 Liability for Collateral. So long as Agent and Lenders comply with applicable law and with reasonable banking practices regarding the safekeeping of the Collateral in their possession or under the control of Agent and/or Lenders, Agent and Lenders shall not be liable or responsible for:
 (a) the safekeeping of the Collateral; (b) any loss or damage to the Collateral; (c) any diminution in the value of the Collateral; or (d) any act or default of any carrier, warehouseman, bailee, or other Person. Except as set forth in the preceding sentence, Borrower bears all risk of loss, damage or destruction of the Collateral.

- 9.6 No Waiver; Remedies Cumulative. Agent's and any Lender's failure, at any time or times, to require strict performance by Borrower of any provision of this Agreement or any other Loan Document shall not waive, affect, or diminish any right of Agent or any Lender thereafter to demand strict performance and compliance herewith or therewith. No waiver hereunder shall be effective unless signed by the party granting the waiver and then is only effective for the specific instance and purpose for which it is given. Agent's and each Lender's rights and remedies under this Agreement and the other Loan Documents are cumulative. Agent and each Lender have all rights and remedies provided under the Code, by law, or in equity. Agent's or any Lender's exercise of one right or remedy is not an election and shall not preclude Agent or any Lender from exercising any other remedy under this Agreement or any other Loan Document or other remedy available at law or in equity, and Agent's or any Lender's waiver of any Event of Default is not a continuing waiver. Agent's or any Lender's delay in exercising any remedy is not a waiver, election, or acquiescence.
- **9.7 Demand Waiver**. Borrower waives demand, notice of default or dishonor, notice of payment and nonpayment, notice of any default, nonpayment at maturity, release, compromise, settlement, extension, or renewal of accounts, documents, instruments, chattel paper, and guarantees held by Agent on which Borrower is liable.

10 AGENT

10.1 Appointment and Authority.

Each Lender hereby irrevocably appoints SVB to act on its behalf as Agent hereunder and under the other Loan Documents and authorizes Agent to take such actions on its behalf and to exercise such powers as are delegated to Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto.

The provisions of this Section 10 are solely for the benefit of Agent and Lenders, and Borrower shall not have rights as a third party beneficiary of any of such provisions. Notwithstanding any provision to the contrary elsewhere in this Agreement, Agent shall not have any duties or responsibilities to any Lender or any other Person, except those expressly set forth herein, or any fiduciary relationship with any Lender, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against Agent.

- 10.2 Delegation of Duties. Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents appointed by Agent. Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Indemnified Persons. The exculpatory provisions of this Section 10.2 shall apply to any such sub-agent and to the Indemnified Persons of Agent and any such sub-agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Agent.
- **10.3 Exculpatory Provisions.** Agent shall have no duties or obligations except those expressly set forth herein and in the other Loan Documents, including without limitation Section 8.1 of the Lender Intercreditor Agreement. Without limiting the generality of the foregoing, Agent shall not:
- (a) be subject to any fiduciary, trust, agency or other similar duties, regardless of whether any Event of Default has occurred and is continuing;
- (b) have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that Agent is required to exercise as directed in writing by the Lenders, as applicable; provided that Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose Agent to liability or that is contrary to any Loan Document or applicable law; and
- (c) except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and Agent shall not be liable for the failure to disclose, any information relating to Borrower or any of its Affiliates that is communicated to or obtained by any Person serving as Agent or any of its Affiliates in any capacity.

Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Lenders (or as Agent shall believe in good faith shall be necessary, under the circumstances as provided in Section 13.7) or (ii) in the absence of its own gross negligence or willful misconduct.

Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Event of Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document or (v) the satisfaction of any condition set forth in Section 3 or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to Agent.

10.4 Reliance by Agent. Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. Agent may consult with legal counsel (who may be counsel for Borrower), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts. In determining compliance with any condition hereunder to the making of a Credit Extension that, by its terms, must be fulfilled to the satisfaction of a Lender, Agent may presume that such condition is satisfactory to such Lender unless Agent shall have received notice to the contrary from such Lender prior to the making of such Credit Extension. Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement and the other Loan Documents in accordance with a request of the Lenders, and such request and any action taken or failure to act pursuant thereto shall be binding upon Lenders and all future holders of the Credit Extensions.

10.5 Notice of Default. Agent shall not be deemed to have knowledge or notice of the occurrence of any Event of Default (except with respect to defaults in the payment of principal, interest or fees required to be paid to Agent for the account of Lenders), unless Agent has received notice from a Lender or Borrower referring to this Agreement, describing such Event of Default and stating that such notice is a "notice of default". In the event that Agent receives such a notice, Agent shall give notice thereof to Lenders. Agent shall take such action with respect to such Event of Default as shall be reasonably directed by the Lenders.

10.6 Non-Reliance on Agent and Other Lenders Each Lender expressly acknowledges that neither Agent nor any of its officers, directors, employees, agents, attorneys in fact or affiliates has made any representations or warranties to it and that no act by Agent hereafter taken, including any review of the affairs of a Group Member or any Affiliate of a Group Member, shall be deemed to constitute any representation or warranty by Agent to any Lender. Each Lender represents to Agent that it has, independently and without reliance upon Agent or any other Lender, and based on such documents and information as it has deemed appropriate, made its own appraisal of, and investigation into, the business, operations, property, financial and other condition and creditworthiness of the Group Members and their Affiliates and made its own decision to make its Credit Extensions hereunder and enter into this Agreement. Each Lender also represents that it will, independently and without reliance upon Agent or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make such investigation as it deems necessary to inform itself as to the business, operations, property, financial and other condition and creditworthiness of the Group Members and their Affiliates. Except for notices, reports and other documents expressly required to be furnished to Lenders by Agent hereunder, Agent shall have no duty or responsibility to provide any Lender with any credit or other information concerning the business, operations, property, condition (financial or otherwise), prospects or creditworthiness of any Group Member or any Affiliate of a Group Member that may come into the possession of Agent or any of its officers, directors, employees, agents, attorneys in fact or Affiliates.

10.7 Indemnification. Each Lender agrees to indemnify Agent in its capacity as such (to the extent not reimbursed by Borrower and without limiting the obligation of Borrower to do so in accordance with the terms hereof, according to its Term Loan Commitment Percentage in effect on the date on which indemnification is sought under this Section 10.7 (or, if indemnification is sought after the date upon which the Commitments shall have terminated and the Obligations shall have been paid in full, in accordance with its Term Loan Commitment Percentage immediately prior to such date), from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind whatsoever that may at any time (whether before or after the payment of the Credit Extensions) be imposed on, incurred by or asserted against Agent in any way relating to or arising out of, the Commitments, this Agreement, any of the other Loan Documents or any documents contemplated by or referred to herein or therein or the transactions contemplated hereby or thereby or any action taken or omitted by Agent under or in connection with any of the foregoing; provided that no Lender shall be liable for the payment of any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements that are found by a final and nonappealable decision of a court of competent jurisdiction to have resulted primarily from Agent's gross negligence or willful misconduct. The agreements in this Section shall survive the payment of the Credit Extensions and all other amounts payable hereunder.

10.8 Agent in Its Individual Capacity. The Person serving as Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include each such Person serving as Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with Borrower, any Guarantor or any Subsidiary or other Affiliate thereof as if such Person were not Agent hereunder and without any duty to account therefor to Lenders.

10.9 Successor Agent. Agent may at any time give notice of its resignation to Lenders and Borrower, which resignation shall not be effective until the time at which the majority of the Lenders have delivered to Agent their written consent to such resignation. Upon receipt of any such notice of resignation, the Lenders shall have the right, in consultation with Borrower, to appoint a successor, which shall be a financial institution with an office in the State of California, or an Affiliate of any such bank with an office in the State of California. If no such successor shall have been so appointed by the Lenders and shall have accepted such appointment within thirty (30) days after the retiring Agent has received the written consent of the majority of the Lenders to such resignation, then the retiring Agent may on behalf of Lenders, appoint a successor Agent meeting the qualifications set forth above; provided that in no event shall any such successor Agent be a Defaulting Lender and provided further that if the retiring Agent shall notify Borrower and Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (1) the retiring Agent shall be discharged from its duties and obligations hereunder and under the other Loan Documents (except that in the case of any collateral security held by Agent on behalf of the Lenders under any of the Loan Documents, the retiring Agent shall continue to hold such collateral security until such time as a successor Agent is appointed and such collateral security is assigned to such successor Agent) and (2) all payments, communications and determinations provided to be made by, to or through Agent shall instead be made by or to each Lender directly, until such time as the Lenders appoint a successor Agent as provided for above in this Section 10.9. Upon the acceptance of a successor's appointment as Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Agent, and the retiring Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section 10.9). The fees payable by Borrower to a successor Agent shall be the same as those payable to its predecessor unless otherwise agreed between Borrower and such successor. After the retiring Agent's resignation hereunder and under the other Loan Documents, the provisions of this Section 10 shall continue in effect for the benefit of such retiring Agent, its sub-agents and their respective Indemnified Persons in respect of any actions taken or omitted to be taken by any of them while the retiring Agent was acting as Agent.

10.10 Defaulting Lender.

- (a) <u>Defaulting Lender Adjustments</u>. Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as such Lender is no longer a Defaulting Lender, to the extent permitted by applicable law:
 - (i) <u>Waivers and Amendments</u>. Such Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as long as said Lender is a Defaulting Lender.
 - (ii) Defaulting Lender Waterfall. Any payment of principal, interest, fees or other amounts received by the Agent for the account of such Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Section 8 or otherwise, and including any amounts made available to the Agent by such Defaulting Lender pursuant to Section 13.10), shall be applied at such time or times as may be determined by the Agent as follows: first, to the payment of any amounts owing by such Defaulting Lender to the Agent hereunder; second, as Borrower may request (so long as no Event of Default exists), to the funding of any Term Loan Advance in respect of which such Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Agent; third, if so determined by the Agent and Borrower, to be held in a Deposit Account and released pro rata to satisfy such Defaulting Lender's potential future funding obligations with respect to Term Loan Advances under this Agreement; fourth, so long as no Event of Default has occurred and is continuing, to the payment of any amounts owing to Borrower as a result of any judgment of a court of competent jurisdiction obtained by Borrower against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; and fifth, to such Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (A) such payment is a payment of the principal amount of any Term Loan Advances in respect of which such Defaulting Lender has not fully funded its appropriate share and (B) such Term Loan Advances were made at a time when the conditions set forth in Section 3.1 were satisfied or waived, such payment shall be applied solely to pay the Term Loan Advances of all non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Term Loan Advances of such Defaulting Lender until such time as all Term Loan Advances are held by the Lenders pro rata in accordance with the Term Loan Commitments under this Agreement. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender pursuant to this Section 10.10(a)(ii) shall be deemed paid to and redirected by such Defaulting Lender, and each Lender irrevocably consents hereto.
 - (iii) <u>Certain Fees.</u> No Defaulting Lender shall be entitled to receive any fee pursuant to Section 2.4(a), Section 2.4(b), or Section 2.4(c) for any period during which such Lender is a Defaulting Lender (and Borrower shall not be required to pay any such fee that otherwise would have been required to have been paid to such Defaulting Lender).
- (b) <u>Defaulting Lender Cure</u>. If Borrower and Agent agree in writing that a Lender is no longer a Defaulting Lender, Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein, such Lender will, to the extent applicable, purchase at par that portion of outstanding Term Loan Advances of the other Lenders or take such other actions as Agent may determine to be necessary to cause the Term Loan Advances to be held on a *pro rata* basis by the Lenders in accordance with their respective Term Loan Commitment Percentages, whereupon such Lender will cease to be a Defaulting Lender; <u>provided</u> that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of Borrower while such Lender was a Defaulting Lender; and <u>provided further</u> that, except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from such Lender having been a Defaulting Lender.
- (c) <u>Termination of Defaulting Lender</u>. Borrower may terminate the unused amount of the Term Loan Commitment of any Lender that is a Defaulting Lender upon not less than ten (10) Business Days' prior notice to Agent (which shall promptly notify the Lenders thereof), and in such event the provisions of Section 10.10(a)(ii) will apply to all amounts thereafter paid by Borrower for the account of such Defaulting Lender under this Agreement (whether on account of principal, interest, fees, indemnity or other amounts); <u>provided</u> that (i) no Event of Default shall have occurred and be continuing, and (ii) such termination shall not be deemed to be a waiver or release of any claim Borrower, Agent or any Lender may have against such Defaulting Lender.

(d) If the Person serving as Agent is a Defaulting Lender pursuant to clause (d) of the definition thereof, thenon-Defaulting Lenders may, to the extent permitted by applicable law, by notice in writing to Borrower and such Person, remove such Person as Agent and, in consultation with Borrower, appoint a successor. If no such successor shall have been so appointed by the non-Defaulting Lenders and shall have accepted such appointment within thirty (30) days (or such earlier day as shall be agreed by the non-Defaulting Lenders) (the "Removal Effective Date"), then such removal shall nonetheless become effective in accordance with such notice on the Removal Effective Date.

11 NOTICES

All notices, consents, requests, approvals, demands, or other communication by any party to this Agreement or any other Loan Document must be in writing and shall be deemed to have been validly served, given, or delivered: (a) upon the earlier of actual receipt and three (3) Business Days after deposit in the U.S. mail, first class, registered or certified mail return receipt requested, with proper postage prepaid; (b) upon transmission, when sent by electronic mail or facsimile transmission; (c) one (1) Business Day after deposit with a reputable overnight courier with all charges prepaid; or (d) when delivered, if hand-delivered by messenger, all of which shall be addressed to the party to be notified and sent to the address, facsimile number, or email address indicated below. Agent or Borrower may change its mailing or electronic mail address or facsimile number by giving the other party written notice thereof in accordance with the terms of this Section 11.

If to Borrower: Grove Collaborative, Inc.

1301 Sansome Street

San Francisco, California 94111 Attn: Janae De Crescenzo Email: janae@grove.co

If to Agent or SVB: Silicon Valley Bank

505 Howard Street, Third Floor San Francisco, California 94105

Attn: Jeff Sereno Email: JSereno@svb.com

with a copy to: Morrison & Foerster LLP

200 Clarendon Street
Boston, Massachusetts 02116
Attn: Charles W. Stavros, Esquire
Email: CStavros@mofo.com

If to Hercules: Hercules Capital, Inc.

400 Hamilton Avenue, Suite 310 Palo Alto, California 94301

Attn: Chief Legal Officer and Greg Peterson Email: legal@htgc.com and gpeterson@htgc.com

with a copy to: Laura Blakely

GCA Law

2570 W. El Camino Real, Suite #400

Mountain View, CA 94040

12 CHOICE OF LAW, VENUE, JURY TRIAL WAIVER AND JUDICIAL REFERENCE

Except as otherwise expressly provided in any of the Loan Documents, California law governs the Loan Documents without regard to principles of conflicts of law. Borrower, Agent, and Lenders each submit to the exclusive jurisdiction of the State and Federal courts in Santa Clara County, California; provided, however, that nothing in this Agreement shall be deemed to operate to preclude Agent or Lenders from bringing suit or taking other legal action in any other jurisdiction to realize on the Collateral or any other security for the Obligations, or to enforce a judgment or other court order in favor of Agent or any Lender. Borrower expressly submits and consents in advance

to such jurisdiction in any action or suit commenced in any such court, and Borrower hereby waives any objection that it may have based upon lack of personal jurisdiction, improper venue, or forum non conveniens and hereby consents to the granting of such legal or equitable relief as is deemed appropriate by such court. Borrower hereby waives personal service of the summons, complaints, and other process issued in such action or suit and agrees that service of such summons, complaints, and other process may be made by registered or certified mail addressed to Borrower at the address set forth in, or subsequently provided by Borrower in accordance with, Section 11 of this Agreement and that service so made shall be deemed completed upon the earlier to occur of Borrower's actual receipt thereof or three (3) days after deposit in the U.S. mails, proper postage prepaid.

TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, BORROWER, AGENT AND EACH LENDER EACH WAIVE THEIR RIGHT TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION ARISING OUT OF OR BASED UPON THIS AGREEMENT, THE LOAN DOCUMENTS OR ANY CONTEMPLATED TRANSACTION, INCLUDING CONTRACT, TORT, BREACH OF DUTY AND ALL OTHER CLAIMS. THIS WAIVER IS A MATERIAL INDUCEMENT FOR ALL PARTIES TO ENTER INTO THIS AGREEMENT. EACH PARTY HAS REVIEWED THIS WAIVER WITH ITS COUNSEL.

WITHOUT INTENDING IN ANY WAY TO LIMIT THE PARTIES' AGREEMENT TO WAIVE THEIR RESPECTIVE RIGHT TO A TRIAL BY JURY, if the above waiver of the right to a trial by jury is not enforceable, the parties hereto agree that any and all disputes or controversies of any nature between them arising at any time shall be decided by a reference to a private judge, mutually selected by the parties (or, if they cannot agree, by the Presiding Judge of the Santa Clara County, California Superior Court) appointed in accordance with California Code of Civil Procedure Section 638 (or pursuant to comparable provisions of federal law if the dispute falls within the exclusive jurisdiction of the federal courts), sitting without a jury, in Santa Clara County, California; and the parties hereby submit to the jurisdiction of such court. The reference proceedings shall be conducted pursuant to and in accordance with the provisions of California Code of Civil Procedure Sections 638 through 645.1, inclusive. The private judge shall have the power, among others, to grant provisional relief, including without limitation, entering temporary restraining orders, issuing preliminary and permanent injunctions and appointing receivers. All such proceedings shall be closed to the public and confidential and all records relating thereto shall be permanently sealed. If during the course of any dispute, a party desires to seek provisional relief, but a judge has not been appointed at that point pursuant to the judicial reference procedures, then such party may apply to the Santa Clara County, California Superior Court for such relief. The proceeding before the private judge shall be conducted in the same manner as it would be before a court under the rules of evidence applicable to judicial proceedings. The parties shall be entitled to discovery which shall be conducted in the same manner as it would be before a court under the rules of discovery applicable to judicial proceedings. The private judge shall oversee discovery and may enforce all discovery rules and orders applicable to judicial proceedings in the same manner as a trial court judge. The parties agree that the selected or appointed private judge shall have the power to decide all issues in the action or proceeding, whether of fact or of law, and shall report a statement of decision thereon pursuant to California Code of Civil Procedure Section 644(a). Nothing in this paragraph shall limit the right of any party at any time to exercise self-help remedies, foreclose against collateral, or obtain provisional remedies. The private judge shall also determine all issues relating to the applicability, interpretation, and enforceability of this paragraph.

This Section 12 shall survive the termination of this Agreement.

13 GENERAL PROVISIONS

13.1 Termination Prior to Term Loan Maturity Date; Survival. All covenants, representations and warranties made in this Agreement continue in full force until this Agreement has terminated pursuant to its terms and all Obligations (other than inchoate indemnity obligations, any other obligations which, by their terms, are to survive the termination of this Agreement, and any Obligations under Bank Services Agreements that are cash collateralized in accordance with Section 4.1 of this Agreement) have been satisfied. So long as Borrower has satisfied the Obligations (other than inchoate indemnity obligations, any other obligations which, by their terms, are to survive the termination of this Agreement, and any Obligations under Bank Services Agreements that are cash collateralized in accordance with Section 4.1 of this Agreement), this Agreement may be terminated prior to the Term Loan Maturity Date by Borrower, effective three (3) Business Days after written notice of termination is given to Agent. Those obligations that are expressly specified in this Agreement as surviving this Agreement's termination shall continue to survive notwithstanding this Agreement's termination. No termination of this Agreement shall in any way affect or

impair any right or remedy of Agent or any Lender, nor shall any such termination relieve Borrower of any Obligation to any Lender, until all of the Obligations (other than inchoate indemnity obligations, any other obligations which, by their terms, are to survive the termination of this Agreement, and any Obligations under Bank Services Agreements that are cash collateralized in accordance with Section 4.1 of this Agreement) have been paid and performed in full. Those Obligations that are expressly specified in this Agreement as surviving this Agreement's termination shall continue to survive notwithstanding this Agreement's termination and payment in full of the Obligations (other than inchoate indemnity obligations, any other obligations which, by their terms, are to survive the termination of this Agreement, and any Obligations under Bank Services Agreements that are cash collateralized in accordance with Section 4.1 of this Agreement) then outstanding.

- 13.2 Successors and Assigns. This Agreement binds and is for the benefit of the successors and permitted assigns of each party. Borrower may not assign this Agreement or any rights or obligations under it without Agent and Lenders' prior written consent (which may be granted or withheld in Agent's and Lenders' sole discretion). Agent and each Lender has the right, without the consent of or notice to Borrower, to sell, transfer, assign, negotiate, or grant participation in all or any part of, or any interest in, such Lender's obligations, rights, and benefits under this Agreement and the other Loan Documents (other than the Warrant, as to which assignment, transfer and other such actions are governed by the terms thereof). Notwithstanding the foregoing, (a) in connection with any assignment by a Lender as a result of a forced divestiture at the request of any regulatory agency, the restrictions set forth herein shall not apply and Agent and the Lenders may assign, transfer or indorse its rights hereunder and under the other Loan Documents to any Person or party and (b) in connection with a Lender's own financing or securitization transactions, the restrictions set forth herein shall not apply and Agent and the Lenders may assign, transfer or indorse its rights hereunder and under the other Loan Documents to any Person or party providing such financing or formed to undertake such securitization transaction and any transferee of such Person or party upon the occurrence of a default, event of default or similar occurrence with respect to such financing or securitization transaction. Notwithstanding the foregoing, prior to the occurrence and continuation of an Event of Default, Agent and the Lenders shall not assign any interest in the Loan Documents to any person who is (a) a direct competitor of Borrower, whether as an operating company or direct or indirect parent with voting control over such operating company, or (b) a vulture fund or distressed debt fund (as determined by Agent and the Lenders in their good fa
- 13.3 Indemnification. Borrower agrees to indemnify, defend and hold Agent, each Lender and their respective directors, officers, employees, agents, attorneys, or any other Person affiliated with or representing Agent or any Lender (each, an "Indemnified Person") harmless against: (i) all obligations, demands, claims, and liabilities (collectively, "Claims") claimed or asserted by any other party in connection with the transactions contemplated by the Loan Documents; and (ii) all losses or expenses (including Lenders' Expenses) in any way suffered, incurred, or paid by such Indemnified Person as a result of, following from, consequential to, or arising from transactions between Agent, Lenders and Borrower pursuant to or in connection with the Loan Documents (including reasonable attorneys' fees and expenses), except for Claims and/or losses directly caused by such Indemnified Person's gross negligence or willful misconduct.

This Section 13.3 shall survive until all statutes of limitation with respect to the Claims, losses, and documented expenses for which indemnity is given shall have run.

- 13.4 Time of Essence. Time is of the essence for the performance of all Obligations in this Agreement.
- 13.5 Severability of Provisions. Each provision of this Agreement is severable from every other provision in determining the enforceability of any provision.
- 13.6 Correction of Loan Documents. Agent may correct patent errors and fill in any blanks in the Loan Documents consistent with the agreement of the parties so long as Agent provides Borrower with written notice of such correction and allows Borrower at least ten (10) days to object to such correction. In the event of such objection, such correction shall not be made except by an amendment signed by both Agent, each Lender and Borrower.

- 13.7 Amendments in Writing; Waiver; Integration. No purported amendment or modification of any Loan Document, or waiver, discharge or termination of any obligation under any Loan Document, or release, or subordinate Agent's and/or Lenders' security interest in, or consent to the transfer of, any Collateral shall be enforceable or admissible unless, and only to the extent, expressly set forth in a writing signed by Agent, with the consent of the Lenders in accordance with the Lender Intercreditor Agreement or, if such item is not addressed in the Lender Intercreditor Agreement, as consented to by a majority of the Lenders, and Borrower. Without limiting the generality of the foregoing, no oral promise or statement, nor any action, inaction, delay, failure to require performance or course of conduct shall operate as, or evidence, an amendment, supplement or waiver or have any other effect on any Loan Document. Any waiver granted shall be limited to the specific circumstance expressly described in it, and shall not apply to any subsequent or other circumstance, whether similar or dissimilar, or give rise to, or evidence, any obligation or commitment to grant any further waiver. The Loan Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of the Loan Documents merge into the Loan Documents. In the event any provision of any other Loan Document is inconsistent with the provisions of this Agreement, the provisions of this Agreement shall exclusively control.
- 13.8 Counterparts. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, is an original, and all taken together, constitute one Agreement.
- 13.9 Confidentiality. Agent and each Lender agrees to maintain the confidentiality of Information (as defined below) and shall exercise the same degree of care with respect to the Information that they exercise for their own respective proprietary information, except that Information may be disclosed (a) to Agent and/or any Lender's subsidiaries or Affiliates, and their respective employees, directors, investors or potential investors (and each of their respective affiliates), agents, attorneys, accountants and other professional advisors (collectively, "Representatives" and, together with Agent and the Lenders, collectively, "Lender Entities"); (b) to prospective transferees, assignees, credit providers or purchasers of any of Agent's or Lenders' interests under or in connection with this Agreement and their Representatives (provided, however, that any prospective transferee, assignees, credit providers or purchasers shall have entered into an agreement containing provisions substantially the same as those in this Section 13.9); (c) as required by law, regulation, subpoena, or other order; (d) to Agent's or any Lender's regulators or as otherwise required in connection with Agent's or any Lender's examination or audit; (e) as Agent or any Lender considers appropriate in exercising remedies under the Loan Documents; and (f) to third-party service providers of Agent and/or any Lender so long as such service providers have executed a confidentiality agreement with Agent or the Lenders, as applicable, with terms no less restrictive than those contained herein. The term "Information" means all information received from Borrower and its agents regarding Borrower or its business, in each case other than information that is either: (i) in the public domain or in Agent's or any Lender's possession when disclosed to Agent or such Lender, or becomes part of the public domain (other than as a result of its disclosure by Agent or a Lender in violation of this Agreement) after disclosure to Agent and/or the Lenders; or (ii)

Lender Entities may use anonymous forms of confidential information for aggregate datasets, for analyses or reporting, and for any other uses not expressly prohibited in writing by Borrower. The provisions of the immediately preceding sentence shall survive the termination of this Agreement.

13.10 Right of Setoff. Borrower hereby grants to Agent, for the ratable benefit of the Lenders, a Lien, security interest, and a right of setoff as security for all Obligations to Agent and the Lenders, whether now existing or hereafter arising upon and against all deposits, credits, collateral and property, now or hereafter in the possession, custody, safekeeping or control of Agent or any entity under the control of Agent (including a subsidiary of Agent) or in transit to any of them. At any time after the occurrence and during the continuance of an Event of Default, without demand or notice, Agent or any Lender may setoff the same or any part thereof and apply the same to any Obligation of Borrower then due regardless of the adequacy of any other collateral securing the Obligations. ANY AND ALL RIGHTS TO REQUIRE AGENT OR ANY LENDER TO EXERCISE ITS RIGHTS OR REMEDIES WITH RESPECT TO ANY OTHER COLLATERAL WHICH SECURES THE OBLIGATIONS, PRIOR TO EXERCISING ITS RIGHT OF SETOFF WITH RESPECT TO SUCH DEPOSITS, CREDITS OR OTHER PROPERTY OF BORROWER, ARE HEREBY KNOWINGLY, VOLUNTARILY AND IRREVOCABLY WAIVED.

- **13.11 Electronic Execution of Documents.** The words "execution," "signed," "signature" and words of like import in any Loan Document shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity and enforceability as a manually executed signature or the use of a paper-based recordkeeping systems, as the case may be, to the extent and as provided for in any applicable law, including, without limitation, any state law based on the Uniform Electronic Transactions Act.
- 13.12 Attorneys' Fees, Costs and Expenses. In any action or proceeding among Borrower, Agent, and the Lenders arising out of or relating to the Loan Documents, the prevailing party shall be entitled to recover its reasonable and documented out-of-pocket attorneys' fees and other costs and expenses incurred, in addition to any other relief to which it may be entitled.
 - 13.13 Captions. The headings used in this Agreement are for convenience only and shall not affect the interpretation of this Agreement.
- 13.14 Construction of Agreement. The parties mutually acknowledge that they and their attorneys have participated in the preparation and negotiation of this Agreement. In cases of uncertainty this Agreement shall be construed without regard to which of the parties caused the uncertainty to exist.
- 13.15 Relationship. The relationship of the parties to this Agreement is determined solely by the provisions of this Agreement. The parties do not intend to create any agency, partnership, joint venture, trust, fiduciary or other relationship with duties or incidents different from those of parties to an arm's-length contract.
- 13.16 Third Parties. Nothing in this Agreement, whether express or implied, is intended to: (a) confer any benefits, rights or remedies under or by reason of this Agreement on any persons other than the express parties to it and their respective permitted successors and assigns; (b) relieve or discharge the obligation or liability of any person not an express party to this Agreement; or (c) give any person not an express party to this Agreement any right of subrogation or action against any party to this Agreement.
- 13.17 Patriot Act. Each Lender hereby notifies Borrower that pursuant to the requirements of the USA PATRIOT Act, it is required to obtain, verify and record information that identifies Borrower and each of its Subsidiaries, which information includes the names and addresses of Borrower and each of its Subsidiaries and other information that will allow Lender, as applicable, to identify Borrower and each of its Subsidiaries in accordance with the USA PATRIOT Act.

14 DEFINITIONS

- **14.1 Definitions.** As used in the Loan Documents, the word "shall" is mandatory, the word "may" is permissive, the word "or" is not exclusive, the words "includes" and "including" are not limiting, the singular includes the plural, and numbers denoting amounts that are set off in brackets are negative. As used in this Agreement, the following capitalized terms have the following meanings:
- "Account" is, as to any Person, any "account" of such Person as "account" is defined in the Code with such additions to such term as may hereafter be made, and includes, without limitation, all accounts receivable and other sums owing to such Person.
 - "Account Debtor" is any "account debtor" as defined in the Code with such additions to such term as may hereafter be made.
- "Affiliate" is, with respect to any Person, each other Person that owns or controls directly or indirectly the Person, any Person that controls or is controlled by or is under common control with the Person, and each of that Person's senior executive officers, directors, partners and, for any Person that is a limited liability company, that Person's managers and members.
 - "Agent" is defined in the preamble hereof.
 - "Agreement" is defined in the preamble hereof.

"Authorized Signer" is any individual listed in Borrower's Borrowing Resolutions who is authorized to execute the Loan Documents, including making (and executing if applicable) any Credit Extension request, on behalf of Borrower.

"Bank Services" are any products, credit services, and/or financial accommodations previously, now, or hereafter provided to Borrower or any of its Subsidiaries by SVB or any SVB Affiliate, including, without limitation, any letters of credit, cash management services (including, without limitation, merchant services, direct deposit of payroll, business credit cards, and check cashing services), interest rate swap arrangements, and foreign exchange services as any such products or services may be identified in SVB's various agreements related thereto (each, a "Bank Services Agreement").

"Bank Services Agreement" is defined in the definition of Bank Services.

"Board" means Borrower's board of directors.

"Borrower" is defined in the preamble hereof.

"Borrower's Books" are all Borrower's books and records including ledgers, federal and state tax returns, records regarding Borrower's assets or liabilities, the Collateral, business operations or financial condition, and all computer programs or storage or any equipment containing such information.

"Borrowing Resolutions" are, with respect to any Person, those resolutions adopted by such Person's board of directors (and, if required under the terms of such Person's Operating Documents, stockholders) and delivered by such Person to Agent approving the Loan Documents to which such Person is a party and the transactions contemplated thereby, together with a certificate executed by its secretary on behalf of such Person certifying (a) such Person has the authority to execute, deliver, and perform its obligations under each of the Loan Documents to which it is a party, (b) that set forth as a part of or attached as an exhibit to such certificate is a true, correct, and complete copy of the resolutions then in full force and effect authorizing and ratifying the execution, delivery, and performance by such Person of the Loan Documents to which it is a party, (c) the name(s) of the Person(s) authorized to execute the Loan Documents, including making (and executing if applicable) any Credit Extension request, on behalf of such Person, together with a sample of the true signature(s) of such Person(s), and (d) that Agent and the Lenders may conclusively rely on such certificate unless and until such Person shall have delivered to Agent and the Lenders a further certificate canceling or amending such prior certificate.

"Business Day" is any day that is not a Saturday, Sunday or a day on which Agent is close...

"Cash Equivalents" means (a) marketable direct obligations issued or unconditionally guaranteed by the United States or any agency or any State thereof having maturities of not more than one (1) year from the date of acquisition; (b) commercial paper maturing no more than one (1) year after its creation and having the highest rating from either Standard & Poor's Ratings Group or Moody's Investors Service, Inc.; (c) SVB's certificates of deposit issued maturing no more than one (1) year after issue; and (d) money market funds at least ninety-five per cent. (95.0%) of the assets of which constitute Cash Equivalents of the kinds described in clauses (a) through (c) of this definition.

"Change in Control" means (a) at any time, any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act), shall become, or obtain rights (whether by means of warrants, options or otherwise) to become, the "beneficial owner" (as defined in Rules 13(d)-3 and 13(d)-5 under the Exchange Act), directly or indirectly, of forty-nine percent (49%) or more of the ordinary voting power for the election of directors of Borrower (determined on a fully diluted basis) other than by the sale of Borrower's equity securities in a public offering or to venture capital or private equity investors so long as Borrower identifies to the Agent and the Lenders the venture capital or private equity investors at least seven (7) Business Days prior to the closing of the transaction and provides to Agent and the Lenders a description of the material terms of the transaction; (b) during any period of twelve (12) consecutive months, a majority of the members of the Board or other equivalent governing body of Borrower cease to be composed of individuals (i) who were members of that board or equivalent governing body on the first day of such period, (ii) whose election or nomination to that board or equivalent governing body was approved

by individuals referred to in clause (i) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body or (iii) whose election or nomination to that board or other equivalent governing body was approved by individuals referred to in clauses (i) and (ii) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body; or (c) at any time, Borrower shall cease to own and control, of record and beneficially, directly or indirectly, one hundred percent (100.0%) of each class of outstanding capital stock of each Subsidiary of Borrower free and clear of all Liens (except Liens created by this Agreement), unless such Subsidiary merges with and into Borrower.

"Claims" is defined in Section 13.3.

"Code" is the Uniform Commercial Code, as the same may, from time to time, be enacted and in effect in the State of California; provided, that, to the extent that the Code is used to define any term herein or in any Loan Document and such term is defined differently in different Articles or Divisions of the Code, the definition of such term contained in Article or Division 9 shall govern; provided further, that in the event that, by reason of mandatory provisions of law, any or all of the attachment, perfection, or priority of, or remedies with respect to, Agent's Lien on any Collateral is governed by the Uniform Commercial Code in effect in a jurisdiction other than the State of California, the term "Code" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes of the provisions thereof relating to such attachment, perfection, priority, or remedies and for purposes of definitions relating to such provisions.

"Collateral" is any and all properties, rights and assets of Borrower described on Exhibit A.

"Collateral Account" is any Deposit Account, Securities Account, or Commodity Account.

"Commitment" and "Commitments" means the Term Loan Commitment(s).

"Commodity Account" is any "commodity account" as defined in the Code with such additions to such term as may hereafter be made.

"Compliance Statement" is that certain statement in the form attached hereto as Exhibit B.

"Contingent Obligation" is, for any Person, any direct or indirect liability, contingent or not, of that Person for (a) any indebtedness, lease, dividend, letter of credit or other obligation of another such as an obligation, in each case, directly or indirectly guaranteed, endorsed, co-made, discounted or sold with recourse by that Person, or for which that Person is directly or indirectly liable; (b) any obligations for undrawn letters of credit for the account of that Person; and (c) all obligations from any interest rate, currency or commodity swap agreement, interest rate cap or collar agreement, or other agreement or arrangement designated to protect a Person against fluctuation in interest rates, currency exchange rates or commodity prices; but "Contingent Obligation" does not include endorsements in the ordinary course of business. The amount of a Contingent Obligation is the stated or determined amount of the primary obligation for which the Contingent Obligation is made or, if not determinable, the maximum reasonably anticipated liability for it determined by the Person in good faith; but the amount may not exceed the maximum of the obligations under any guarantee or other support arrangement.

"Contribution Profit" is, as determined by Agent in its reasonable discretion, Borrower's net revenue minus product costs minus variable order costs (including, without limitation, all costs incurred in operating and staffing Borrower's fulfillment centers, including costs attributable to receiving, inspecting and warehousing inventories, picking, packaging and preparing customer orders for shipment, shipping and handling expenses, packaging materials costs and payment processing and related transaction costs but excluding, for the avoidance of doubt, facility and building operating costs and expenses and general overhead costs and expenses).

"Control Agreement" is any control agreement entered into among the depository institution at which Borrower maintains a Deposit Account or the securities intermediary or commodity intermediary at which Borrower maintains a Securities Account or a Commodity Account, Borrower, and Agent pursuant to which Agent obtains control (within the meaning of the Code) for the benefit of the Lenders over such Deposit Account, Securities Account, or Commodity Account.

"Copyrights" are any and all copyright rights, copyright applications, copyright registrations and like protections in each work of authorship and derivative work thereof, whether published or unpublished and whether or not the same also constitutes a trade secret.

"Credit Extension" is any Term Loan Advance, or any other extension of credit by any Lender for Borrower's benefit under the Loan Documents.

"Default Rate" is defined in Section 2.3(b).

"Defaulting Lender" is, subject to Section 10.10(b), any Lender that (a) has failed to (i) fund all or any portion of its Term Loan Advances within two (2) Business Days of the date such Term Loan Advances were required to be funded hereunder unless such Lender notifies Agent and Borrower in writing that such failure is the result of such Lender's reasonable determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied, or (ii) pay to Agent or any other Lender any other amount required to be paid by it hereunder within two (2) Business Days of the date when due, (b) has notified Borrower or Agent in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect (unless such writing or public statement relates to such Lender's obligation to fund a Term Loan Advance hereunder and states that such position is based on such Lender's reasonable determination that a condition precedent to funding (which condition precedent, together with any applicable default, shall be specifically identified in such writing or public statement) cannot be satisfied), (c) has failed, within three (3) Business Days after written request by Agent or Borrower, to confirm in writing to Agent and Borrower that it will comply with its prospective funding obligations hereunder (provided that such Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon receipt of such written confirmation by Agent and Borrower), or (d) has, or has a direct or indirect parent company that has, (i) become the subject of an Insolvency Proceeding, or (ii) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity; provided that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority so long as such ownership interest does not result in or provide such Lender with immunity from the jurisdiction of courts within the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Authority) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lender. Any determination by Agent that a Lender is a Defaulting Lender under any one or more of clauses (a) through (d) above shall be conclusive and binding absent manifest error, and such Lender shall be deemed to be a Defaulting Lender (subject to Section 10.10(b)) upon delivery of written notice of such determination to Borrower and each Lender.

"Deposit Account" is any "deposit account" as defined in the Code with such additions to such term as may hereafter be made.

"Designated Deposit Account" is the account number ending 042 (last three digits) maintained by Borrower with SVB (provided, however, if no such account number is included, then the Designated Deposit Account shall be any deposit account of Borrower maintained with SVB as chosen by the Lenders).

"Disbursement Letter" is that certain form attached hereto as Exhibit D.

"Division" means, in reference to any Person which is an entity, the division of such Person into two (2) or more separate Persons, with the dividing Person either continuing or terminating its existence as part of such division, including, without limitation, as contemplated under Section 18-217 of the Delaware Limited Liability Company Act for limited liability companies formed under Delaware law, or any analogous action taken pursuant to any other applicable law with respect to any corporation, limited liability company, partnership or other entity.

"Dollar Equivalent" is, at any time, (a) with respect to any amount denominated in Dollars, such amount, and (b) with respect to any amount denominated in a Foreign Currency, the equivalent amount therefor in Dollars as determined by Agent at such time on the basis of the then-prevailing rate of exchange in San Francisco, California, for sales of the Foreign Currency for transfer to the country issuing such Foreign Currency.

- "Dollars," "dollars" or use of the sign "\$" means only lawful money of the United States and not any other currency, regardless of whether that currency uses the "\$" sign to denote its currency or may be readily converted into lawful money of the United States.
- "Domestic Subsidiary" means a Subsidiary organized under the laws of the United States or any state or territory thereof or the District of Columbia.
- "Draw Period A" is the period of time commencing on the Effective Date and continuing through the earlier to occur of (a) March 31, 2022, and (b) an Event of Default.
- "Draw Period B" is the period of time commencing upon the occurrence of Performance Milestone and continuing through the earlier to occur of (a) April 30, 2022, and (b) an Event of Default.
 - "Effective Date" is defined in the preamble hereof.
- "Equipment" is all "equipment" as defined in the Code with such additions to such term as may hereafter be made, and includes without limitation all machinery, fixtures, goods, vehicles (including motor vehicles and trailers), and any interest in any of the foregoing.
 - "ERISA" is the Employee Retirement Income Security Act of 1974, and its regulations.
 - "Event of Default" is defined in Section 8.
 - "Exchange Act" is the Securities Exchange Act of 1934, as amended.
- "Excluded Taxes" means any of the following taxes imposed on or with respect to Bank or required to be withhold or deducted from a payment to Bank: taxes imposed on or measured by net income (however denominated), franchise taxes, and branch profits taxes, in each case imposed as a result of Bank being organized under the laws of, or having its principal office or its applicable lending office located in, the jurisdiction imposing such tax (or any political subdivision thereof).
- "Federal Funds Effective Rate" means, for any day, the weighted average of the rates on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers, as published on the next succeeding Business Day by the Federal Reserve Bank of New York, or, if such rate is not so published for any day that is a Business Day, the average of the quotations for the day of such transactions received by SVB from three federal funds brokers of recognized standing selected by it.
- "Final Payment" is a payment (in addition to and not in substitution for the regular monthly payments of principal plus accrued interest) equal to the aggregate original principal amount of the Term Loan Advances extended by the Lenders to Borrower hereunder multiplied by six and three quarters of one percent (6.75%) due on the earliest to occur of (a) the Term Loan Maturity Date, (b) the payment in full of the Term Loan Advances, (c) as required by Section 2.2(d) or Section 2.2(e), or (d) the termination of this Agreement.
- "Financial Statement Repository" is C4344e@svb.com or such other means of collecting information approved and designated by Agent and each Lender after providing notice thereof to Borrower from time to time.
 - "Foreign Currency" means lawful money of a country other than the United States.
 - "Foreign Subsidiary" means any Subsidiary which is not a Domestic Subsidiary.
 - "Funding Date" is any date on which a Credit Extension is made to or for the account of Borrower which shall be a Business Day.
- "FX Contract" is any foreign exchange contract by and between Borrower and SVB under which Borrower commits to purchase from or sell to SVB a specific amount of Foreign Currency on a specified date.

"GAAP" is generally accepted accounting principles set forth in the opinions and pronouncements of the Accounting Principles Board of the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or in such other statements by such other Person as may be approved by a significant segment of the accounting profession, which are applicable to the circumstances as of the date of determination.

"General Intangibles" is all "general intangibles" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation, all Intellectual Property, claims, income and other tax refunds, security and other deposits, payment intangibles, contract rights, options to purchase or sell real or personal property, rights in all litigation presently or hereafter pending (whether in contract, tort or otherwise), insurance policies (including without limitation key man, property damage, and business interruption insurance), payments of insurance and rights to payment of any kind.

"Good Faith Deposit" is defined in Section 2.4(d).

"Governmental Approval" is any consent, authorization, approval, order, license, franchise, permit, certificate, accreditation, registration, filing or notice, of, issued by, from or to, or other act by or in respect of, any Governmental Authority.

"Governmental Authority" is any nation or government, any state or other political subdivision thereof, any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative functions of or pertaining to government, any securities exchange and any self-regulatory organization.

"Group Member" means Borrower and its Subsidiaries.

"Guarantor" is any Person providing a Guaranty in favor of Bank.

"Guaranty" is any guarantee of all or any part of the Obligations, as the same may from time to time be amended, restated, modified or otherwise supplemented.

"Hercules" is defined in the preamble hereof.

"Indebtedness" is (a) indebtedness for borrowed money or the deferred price of property or services, such as reimbursement and other obligations for surety bonds and letters of credit, (b) obligations evidenced by notes, bonds, debentures or similar instruments, (c) capital lease obligations, and (d) Contingent Obligations.

"Indemnified Person" is defined in Section 13.3.

"Information" is defined in Section 13.9.

"Insolvency Proceeding" is any proceeding by or against any Person under the United States Bankruptcy Code, or any other bankruptcy or insolvency law, including assignments for the benefit of creditors, compositions, extensions generally with its creditors, or proceedings seeking reorganization, arrangement, or other relief.

"Intellectual Property" means, with respect to any Person, all of such Person's right, title, and interest in and to the following:

- (a) its Copyrights, Trademarks and Patents;
- (b) any and all trade secrets and trade secret rights, including, without limitation, any rights to unpatented inventions, know-how and operating manuals;
 - (c) any and all source code;

- (d) any and all design rights which may be available to such Person;
- (e) any and all claims for damages by way of past, present and future infringement of any of the foregoing, with the right, but not the obligation, to sue for and collect such damages for said use or infringement of the Intellectual Property rights identified above; and
 - (f) all amendments, renewals and extensions of any of the Copyrights, Trademarks or Patents.
- "Inventory" is all "inventory" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation all merchandise, raw materials, parts, supplies, packing and shipping materials, work in process and finished products, including without limitation such inventory as is temporarily out of Borrower's custody or possession or in transit and including any returned goods and any documents of title representing any of the above.
- "Investment" is any beneficial ownership interest in any Person (including stock, partnership interest or other securities), and any loan, advance or capital contribution to any Person.
- "IP Agreement" is that certain Intellectual Property Security Agreement between Borrower and Agent dated as of the Effective Date, as may be amended, modified or restated from time to time.
 - "Key Person" is Borrower's Chief Financial Officer, who is Servaes Tholen as of the Effective Date.
 - "Lender" and "Lenders" is defined in the preamble.
 - "Lender Entities" is defined in Section 13.9.
- "Lender Intercreditor Agreement" is that certain Intercreditor Agreement dated as of the Effective Date by and between Hercules and SVB, as may be amended from time to time in accordance with the provisions thereof.
- "Lenders' Expenses" are all of Agent's and the Lenders' documented audit fees and expenses, costs, and expenses (including reasonable and documented attorneys' fees and out-of-pocket expenses) for preparing, amending, negotiating, administering, defending and enforcing the Loan Documents (including, without limitation, those incurred in connection with appeals or Insolvency Proceedings) or otherwise incurred with respect to Borrower.
- "Letter of Credit" is a standby or commercial letter of credit issued by SVB upon request of Borrower based upon an application, guarantee, indemnity, or similar agreement.
- "Lien" is a claim, mortgage, deed of trust, levy, charge, pledge, security interest or other encumbrance of any kind, whether voluntarily incurred or arising by operation of law or otherwise against any property.
- "Loan Documents" are, collectively, this Agreement and any schedules, exhibits, certificates, notices, and any other documents related to this Agreement, the Warrant, the Perfection Certificate, each Disbursement Letter, the IP Agreement, the Lender Intercreditor Agreement, the Subordination Agreement, any Bank Services Agreement, any Control Agreement, any subordination agreement, any note, or notes or guaranties executed by Borrower, and any other present or future agreement by Borrower with or for the benefit of Agent and the Lenders in connection with this Agreement or Bank Services, all as amended, restated, or otherwise modified.
- "Material Adverse Change" is (a) a material impairment in the perfection or priority of Agent's, for the ratable benefit of the Lenders, Lien in the Collateral or in the value of such Collateral; (b) a material adverse change in the business, operations, or financial condition of Borrower; or (c) a material impairment of the prospect of repayment of any portion of the Obligations.
 - "Merchant Processing Accounts" is defined in Section 6.8.
 - "Monthly Financial Statements" is defined in Section 6.2(a).

"Non-U.S. Lender" is defined in Section 2.7(b).

"Obligations" are Borrower's obligations to pay when due any debts, principal, interest, fees, Lenders' Expenses, the Prepayment Premium, the Final Payment, and other amounts Borrower owes Agent or any Lender now or later, whether under this Agreement, the other Loan Documents (other than the Warrant), or otherwise, including, without limitation, all obligations relating to Bank Services, if any, and interest accruing after Insolvency Proceedings begin and debts, liabilities, or obligations of Borrower assigned to Agent and/or the Lenders, and to perform Borrower's duties under the Loan Documents (other than the Warrant).

"Operating Documents" are, for any Person, such Person's formation documents, as certified by the Secretary of State (or equivalent agency) of such Person's jurisdiction of organization on a date that is no earlier than thirty (30) days prior to the Effective Date, and, (a) if such Person is a corporation, its bylaws in current form, (b) if such Person is a limited liability company, its limited liability company agreement (or similar agreement), and (c) if such Person is a partnership, its partnership agreement (or similar agreement), each of the foregoing with all current amendments or modifications thereto.

"Patents" means all patents, patent applications and like protections including without limitation improvements, divisions, continuations, renewals, reissues, extensions and continuations-in-part of the same.

"Payment/Advance Form" is that certain form attached hereto as Exhibit C.

"Payment Date" is the first (1st) calendar day of each month.

"Perfection Certificate" is defined in Section 5.1.

"Performance Milestone" means delivery by Borrower to Agent and each Lender, on or prior to April 30, 2022, of evidence satisfactory to Agent and each Lender in Agent's and each Lender's sole and absolute discretion, that Borrower has achieved Contribution Profit of at least One Hundred Ten Million Dollars (\$110,000,000.00) on a trailing twelve-month basis as of a month ending on or prior to February 28, 2022.

"Permitted Indebtedness" is:

- (a) Borrower's Indebtedness to Agent and the Lenders under this Agreement and the other Loan Documents;
- (b) Indebtedness existing on the Effective Date which is shown on the Perfection Certificate;
- (c) Subordinated Debt;
- (d) unsecured Indebtedness to trade creditors incurred in the ordinary course of business;
- (e) Indebtedness incurred as a result of endorsing negotiable instruments received in the ordinary course of business;
- (f) Indebtedness secured by Liens permitted under clauses (a) and (c) of the definition of "Permitted Liens" hereunder;
- (g) Borrower's credit cards maintained with third parties in the ordinary course of Borrower's business, provided that the aggregate credit limits available thereunder do not exceed One Million Five Hundred Thousand Dollars (\$1,500,000) at any time;
 - (h) other Indebtedness not otherwise enumerated herein not to exceed Three Hundred Thousand Dollars (\$300,000) at any time;
 - (i) Indebtedness in connection with the Senior Loan Agreement;

(j) extensions, refinancings, modifications, amendments, renewals, and restatements of any items of Permitted Indebtedness (a) through (i) above, provided that the principal amount thereof is not increased or the terms thereof are not modified to impose more burdensome terms upon Borrower or its Subsidiary, as the case may be.

"Permitted Investments" are:

- (a) Investments (including, without limitation, Subsidiaries) existing on the Effective Date which are shown on the Perfection Certificate;
- (b) Investments consisting of Cash Equivalents;
- (c) Investments consisting of the endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of business;
- (d) Investments consisting of deposit accounts (but only to the extent that Borrower is permitted to maintain such accounts pursuant to Section 6.7 of this Agreement) in which Agent, for the ratable benefit of the Lenders, has a first priority perfected security interest, except as provided and excepted by Section 6.6(b);
 - (e) Investments accepted in connection with Transfers permitted by Section 7.1;
- (f) Investments consisting of the creation of a Subsidiary for the purpose of consummating a merger transaction permitted by Section 7.3 of this Agreement, which is otherwise a Permitted Investment;
- (g) Investments (i) by Borrower in Subsidiaries that are not a co-borrower or Guarantor not to exceed Two Hundred Fifty Thousand Dollars (\$250,000) in the aggregate in any fiscal year, (ii) by Subsidiaries (that are not a co-borrower or Guarantor) in other Subsidiaries not to exceed Two Hundred Fifty Thousand Dollars (\$250,000) in the aggregate in any fiscal year or in Borrower, and (iii) by any Borrower in any other co-borrower or Guarantor;
- (h) Investments consisting of (i) travel advances and employee relocation loans and other employee loans and advances in the ordinary course of business, and (ii) loans to employees, officers or directors relating to the purchase of equity securities of Borrower or its Subsidiaries pursuant to employee stock purchase plans or agreements approved by the Board
- (i) Investments (including debt obligations) received in connection with the bankruptcy or reorganization of customers or suppliers and in settlement of delinquent obligations of, and other disputes with, customers or suppliers arising in the ordinary course of business;
- (j) Investments in joint ventures or strategic alliances in the ordinary course of Borrower's business, provided that any cash investments by Borrower do not exceed Five Hundred Thousand Dollars (\$500,000) in the aggregate in any fiscal year; and
- (k) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions, to customers and suppliers who are not Affiliates, in the ordinary course of business; provided that this paragraph (j) shall not apply to Investments of Borrower in any Subsidiary.

"Permitted Liens" are:

- (a) Liens existing on the Effective Date which are shown on the Perfection Certificate or arising under this Agreement and the other Loan Documents:
- (b) Liens for taxes, fees, assessments or other government charges or levies, either (i) not due and payable or (ii) being contested in good faith and for which Borrower maintains adequate reserves on its Books, <u>provided</u> that no notice of any such Lien has been filed or recorded under the Internal Revenue Code of 1986, as amended, and the Treasury Regulations adopted thereunder;

- (c) purchase money Liens or capital leases (i) on Equipment acquired or held by Borrower incurred for financing the acquisition of the Equipment securing no more than Five Million Dollars (\$5,000,000) in the aggregate amount outstanding, or (ii) existing on Equipment when acquired, if the Lien is confined to the property and improvements and the proceeds of the Equipment;
- (d) Liens of carriers, warehousemen, suppliers, or other Persons that are possessory in nature arising in the ordinary course of business so long as such Liens attach only to Inventory, securing liabilities in the aggregate amount not to exceed Two Hundred Fifty Thousand Dollars (\$250,000) and which are not delinquent or remain payable without penalty or which are being contested in good faith and by appropriate proceedings which proceedings have the effect of preventing the forfeiture or sale of the property subject thereto;
- (e) Liens to secure payment of workers' compensation, employment insurance, old-age pensions, social security and other like obligations incurred in the ordinary course of business (other than Liens imposed by ERISA);
- (f) Liens incurred in the extension, renewal or refinancing of the Indebtedness secured by Liens described in (a) through (c), but any extension, renewal or replacement Lien must be limited to the property encumbered by the existing Lien and the principal amount of the indebtedness may not increase
- (g) leases or subleases of real property granted in the ordinary course of Borrower's business (or, if referring to another Person, in the ordinary course of such Person's business), and leases, subleases, non-exclusive licenses or sublicenses of personal property (other than Intellectual Property) granted in the ordinary course of Borrower's business (or, if referring to another Person, in the ordinary course of such Person's business), if the leases, subleases, licenses and sublicenses do not prohibit granting Agent, for the ratable benefit of the Lenders, a security interest therein;
- (h) non-exclusive licenses of Intellectual Property granted to third parties in the ordinary course of business, and licenses of Intellectual Property that could not result in a legal transfer of title of the licensed property that may be exclusive in respects other than territory and that may be exclusive as to territory only as to discrete geographical areas outside of the United States;
- (i) Liens arising from attachments or judgments, orders, or decrees in circumstances not constituting an Event of Default under Sections 8.4 and 8.7; and
 - (j) Liens in connection with the Senior Loan Agreement.
- "Person" is any individual, sole proprietorship, partnership, limited liability company, joint venture, company, trust, unincorporated organization, association, corporation, institution, public benefit corporation, firm, joint stock company, estate, entity or government agency.
- "Prepayment Premium" shall be an additional fee, payable (i) to Agent, for the account Lenders (other than Hercules) and (ii) Hercules for its own account, based on each Lender's Pro Rata Share, with respect to the Term Loan Advances, in an amount equal to:
- (a) for a prepayment of the Term Loan Advances made on or prior to the first (\$\frac{1}{2}\$) anniversary of the Effective Date, three percent (3.0%) of the principal amount of the Term Loan Advances being prepaid;
- (b) for a prepayment of the Term Loan Advances made after the first (\$\forall 1\) anniversary of the Effective Date, but on or prior to the second (2nd) anniversary of the Effective Date, two percent (2.0%) of the principal amount of the Term Loan Advances being prepaid; and

(c) for a prepayment of the Term Loan Advances made after the second (2nd) anniversary of the Effective Date but prior to the Term Loan Maturity Date, one percent (1.0%) of the principal amount of the Term Loan Advances being prepaid; provided that no Prepayment Premium shall be charged if the credit facility hereunder is replaced with a new facility from all of the Lenders that are party hereto in accordance with their respective Term Loan Commitment Percentages immediately prior to the date of such replacement; provided further that, SVB's Pro Rata Share of the Prepayment Premium shall not be charged if the credit facility hereunder is replaced with a new facility by SVB.

"Prime Rate" is the rate of interest per annum from time to time published in the money rates section of The Wall Street Journal or any successor publication thereto as the "prime rate" then in effect; provided that, in the event such rate of interest is less than zero, such rate shall be deemed to be zero for purposes of this Agreement; and provided further that if such rate of interest, as set forth from time to time in the money rates section of The Wall Street Journal, becomes unavailable for any reason as determined by Agent, the "Prime Rate" shall mean the rate of interest per annum announced by SVB as its prime rate in effect at its principal office in the State of California (such SVB announced Prime Rate not being intended to be the lowest rate of interest charged by SVB in connection with extensions of credit to debtors); provided that, in the event such rate of interest is less than zero, such rate shall be deemed to be zero for purposes of this Agreement.

"Pro Rata Share" is, as of any date of determination, with respect to each Lender, a percentage (expressed as a decimal, rounded to the ninth decimal place) determined by dividing the outstanding principal amount of Term Loan Advances held by such Lenderby the aggregate outstanding principal amount of all Term Loan Advances.

"Registered Organization" is any "registered organization" as defined in the Code with such additions to such term as may hereafter be made.

"Removal Effective Date" is defined in Section 10.10(d).

"Representatives" is defined in Section 13.9.

"Requirement of Law" is as to any Person, the organizational or governing documents of such Person, and any law (statutory or common), treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.

"Responsible Officer" is any of the Chief Executive Officer, President, Chief Financial Officer, VP of Finance and Controller of Borrower.

"Restricted License" is any material license or other similar material agreement with respect to which Borrower is the licensee (a) that prohibits or otherwise restricts Borrower from granting a security interest in Borrower's interest in such license or material agreement or any other property, or (b) for which a default under or termination of could reasonably be expected to interfere with the Agent's right to sell any Collateral.

"SEC" shall mean the Securities and Exchange Commission, any successor thereto, and any analogous Governmental Authority.

"Securities Account" is any "securities account" as defined in the Code with such additions to such term as may hereafter be made.

"Senior Loan Agreement" is that certain Second Amended and Restated Loan and Security Agreement by and between Borrower and SVB dated as of July 29, 2020, as amended, modified, supplemented and/or restated from time to time, including, without limitation, by that certain First Amendment to Second Amended and Restated Loan and Security Agreement dated of even date herewith.

"Subordinated Debt" is indebtedness incurred by Borrower subordinated to all of Borrower's now or hereafter indebtedness to Agent and the Lenders (pursuant to a subordination, intercreditor, or other similar agreement in form and substance satisfactory to Agent and the Lenders entered into between Agent, the Lenders and the other creditor), on terms acceptable to Agent and the Lenders.

"Subordination Agreement" is that certain Intercreditor and Subordination Agreement by and among SVB and Bank (as such term is defined in the Senior Loan Agreement), and the Lenders dated as of the Effective Date (as from time to time be amended, modified, supplemented and/or restated).

"Subsidiary" is, as to any Person, a corporation, partnership, limited liability company or other entity of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person. Unless the context otherwise requires, each reference to a Subsidiary herein shall be a reference to a Subsidiary of Borrower.

"SVB" is defined in the preamble hereof.

"Term A Loan Advance" and "Term A Loan Advances" is defined in Section 2.2(a).

"Term B Loan Advance" and "Term B Loan Advances" are each defined in Section 2.2(a).

"Term Loan Advance" and "Term Loan Advances" are each defined in Section 2.2(a).

"Term A Loan Commitment" means, for any Lender, the obligation of such Lender to make the Term A Loan Advance as and when available, up to the principal amount shown on Schedule 1.

"Term B Loan Commitment" means, for any Lender, the obligation of such Lender to make the Term B Loan Advance as and when available, up to the principal amount shown on Schedule 1.

"Term Loan Commitments" means the aggregate amount of the Term A Loan Commitment and the Term B Loan Commitment of all Lenders.

"Term Loan Commitment Percentage" means, as to any Lender at any time, the percentage (carried out to the fourth decimal place) of the Term Loan Commitments represented by such Lender's Term Loan Commitment at such time. The initial Term Loan Commitment Percentage of each Lender is set forth opposite the name of such Lender on Schedule 1.

"Term Loan Maturity Date" is April 1, 2025.

"Trademarks" means any trademark and servicemark rights, whether registered or not, applications to register and registrations of the same and like protections, and the entire goodwill of the business of Borrower connected with and symbolized by such trademarks.

"Transfer" is defined in Section 7.1.

"Warrant" means, collectively, (a) that certain warrant to purchase stock dated as of the Effective Date between Borrower and SVB, (b) that certain warrant to purchase stock dated as of the Effective Date between Borrower and SVB INNOVATION CREDIT FUND VIII, L.P. and (c) that certain warrant to purchase stock dated as of the Effective Date between Borrower and Hercules, in each case, as may be amended, modified, supplemented and/or restated from time to time.

[Signature Page Follows.]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the Effective Date.

BORROWER:

GROVE COLLABORATIVE, INC.

By /s/ Servaes Tholen
Name: Servaes Tholen
Title: Chief Financial Officer

AGENT:

SILICON VALLEY BANK, as Agent

 By
 /s/ Jeff Sereno

 Name:
 Jeff Sereno

 Title:
 Vice President

LENDERS:

SILICON VALLEY BANK

 By
 /s/ Jeff Sereno

 Name:
 Jeff Sereno

 Title:
 Vice President

HERCULES CAPITAL, INC.

By /s/ Zhuo Huang

Name: Zhuo Huang

Title: Associate General Counsel

[Signature Page to Mezzanine Loan and Security Agreement]

SCHEDULE 1

LENDERS AND COMMITMENTS

TERM A LOAN COMMITMENTS

		Term A Loan
Lender	Term A Loan Commitment	Commitment Percentage
Silicon Valley Bank	\$ 30,600,000.00	51.0000%
Hercules Capital, Inc.	\$ 29,400,000.00	49.0000%
TOTAL	\$ 60,000,000.00	100.0000%

TERM B LOAN COMMITMENTS

			Term B Loan
Lender	Term	B Loan Commitment	Commitment Percentage
Silicon Valley Bank	\$	10,200,000.00	51.0000%
Hercules Capital, Inc.	\$	9,800,000.00	49.0000%
<u>TOTAL</u>	\$	20,000,000.00	100.0000%

EXHIBIT A - COLLATERAL DESCRIPTION

The Collateral consists of all of Borrower's right, title and interest in and to the following personal property:

All goods, Accounts (including health-care receivables), Equipment, Inventory, contract rights or rights to payment of money, leases, license agreements, franchise agreements, General Intangibles, Intellectual Property, commercial tort claims, documents, instruments (including any promissory notes), chattel paper (whether tangible or electronic), cash, deposit accounts, certificates of deposit, fixtures, letters of credit rights (whether or not the letter of credit is evidenced by a writing), securities, and all other investment property, supporting obligations, and financial assets, whether now owned or hereafter acquired, wherever located; and

all Borrower's Books relating to the foregoing, and any and all claims, rights and interests in any of the above and all substitutions for, additions, attachments, accessories, accessions and improvements to and replacements, products, proceeds and insurance proceeds of any or all of the foregoing.

Notwithstanding the foregoing, the Collateral does not include (a) more than 65% of the presently existing and hereafter arising issued and outstanding shares of capital stock owned by Borrower of any Foreign Subsidiary which shares entitle the holder thereof to vote for directors or any other matter; and (b) any interest of Borrower as a lessee under an equipment or other lease if Borrower is prohibited by the terms of such lease from granting a security interest in such lease or under which such an assignment or Lien would cause a default to occur under such lease; *provided, however*, that upon termination of such prohibition, such interest shall immediately become Collateral without any action by Borrower or Bank.

EXHIBIT B COMPLIANCE STATEMENT

TO:	SILICON VALLEY BANK, as Agent, SVB, and HERCULES	Date:
FROM:	GROVE COLLABORATIVE, INC.	

Under the terms and conditions of the Mezzanine Loan and Security Agreement among Borrower, SVB, and Hercules (as amended, the **Loan**

Under the terms and conditions of the Mezzanine Loan and Security Agreement among Borrower, SVB, and Hercules (as amended, the **Loan** Agreement"), Borrower is in compliance for the period ending _____ with all required covenants except as noted below. Attached are the required documents evidencing such compliance, setting forth calculations prepared in accordance with GAAP consistently applied from one period to the next except as explained in an accompanying letter or footnotes, and except, in each case applicable hereunder, with respect to unaudited financial statements, for the absence of footnotes and subject to year-end adjustments Capitalized terms used but not otherwise defined herein shall have the meanings given them in the Agreement.

Please indicate compliance status by circling Yes/No under "Complies" column.

Reporting Covenants	Required		Complies	
Monthly financial statements with Compliance Statement	Monthly within 30 days	Yes	No	
Annual financial statement (CPA Audited)	FYE within 210 days	Yes	No	
Filed 10-Q, 10-K and 8-K (if applicable)	Within 5 days after filing with SEC	Yes	No	
Board-Approved Projections	Within the earlier to (i) occur 45 days of Borrower's FYE and (ii) 45 days after approval by the Board, and within 7 days of Board approval of any amendment/updates	Yes	No	
409A Valuation	Within 30 days of Board approval	Yes	No	
Board decks	Within 45 days of presentation to the Board (subject to exceptions set forth in Section 6.2(e))	Yes	No	
The following Intellectual Property was registered after the Effective D otherwise in updates to Agent/Lenders (if no registrations, state "None"	tate, and has not otherwise been provided with previous Compliance Certi	ficates	or	
-				

Other Matters

Have there been any material amendments of or other changes to the Operating Documents of Borrower or any of its Subsidiaries? If yes, provide Yes No copies of any such amendments or changes with this Compliance Certificate.

Have there been any material changes to the capitalization table of Borrower? If yes, provide copies of any such amendments or changes with this Yes No Compliance Certificate.

The following are the exceptions with respect to the certification above: (If no exceptions exist, state "No exceptions to note.")

EXHIBIT C

LOAN PAYMENT/ADVANCE REQUEST FORM DEADLINE FOR SAME DAY PROCESSING IS 1:00 PM PACIFIC TIME

Fax To:	Date:
LOAN PAYMENT: GROVE COLLABORATIVE, INC.	
From Account #	To Account #
From Account #(Deposit Account #)	To Account #(Loan Account #)
Principal \$	and/or Interest \$
Authorized Signature:Print Name/Title:	Phone Number:
LOAN ADVANCE:	
Complete Outgoing Wire Request section below if all or a portion of the	ne funds from this loan advance are for an outgoing wire.
From Account #	To Account #
From Account #(Loan Account #)	To Account #(Deposit Account #)
Amount of Term Loan Advance \$	
All Borrower's representations and warranties in the Mezzanine Loan an advance.	and Security Agreement are true, correct and complete on the date of the request for
Authorized Signature:Print Name/Title:	Phone Number:
OUTGOING WIRE REQUEST: Complete only if all or a portion of funds from the loan advance all Deadline for same day processing is 1:00 PM, Pacific Time	pove is to be wired.
Beneficiary Name:	Amount of Wire: \$
Beneficiary Bank: City and State:	Account Number:
Beneficiary Bank Transit (ABA) #:	Beneficiary Bank Code (Swift, Sort, Chip, etc.):
Intermediary Bank: For Further Credit to:	(For International Wire Only) Transit (ABA) #:
Special Instruction:	
	ransfer request shall be processed in accordance with and subject to the terms and se(s), which agreements(s) were previously received and executed by me (us).
Authorized Signature:	2nd Signature (if required):
Print Name/Title:	Print Name/Title:
Telephone #:	Telephone #:

EXHIBIT D

Form of Disbursement Letter

[see attached]

DISBURSEMENT LETTER

[DATE]

The undersigned, being the duly elected and acting ______ of ____ GROVE COLLABORATIVE, INC., a Delaware public benefit corporation ("Borrower"), does hereby certify to (a) SILICON VALLEY BANK, a California corporation ("SVB"), in its capacity as administrative agent and collateral agent ("Agent"), (b) SILICON VALLEY BANK, a California corporation, as a lender, (c) HERCULES CAPITAL, INC., a Maryland corporation ("Hercules"), as a lender (SVB and Hercules and each of the other "Lenders" from time to time a party hereto are referred to herein collectively as the "Lenders" and each individually as a "Lender") in connection with that certain Mezzanine Loan and Security Agreement dated as of April 30, 2021, by and among Borrower, Agent and the Lenders from time to time party thereto (the "Loan Agreement"; with other capitalized terms used below having the meanings ascribed thereto in the Loan Agreement) that:

- 1. The representations and warranties made by Borrower in Section 5 of the Loan Agreement and in the other Loan Documents are true and correct in all material respects as of the date hereof.
- 2. No event or condition has occurred and is continuing that would constitute an Event of Default under the Loan Agreement or any other Loan Document.
 - 3. Borrower is in compliance with the covenants and requirements contained in Sections 4, 6 and 7 of the Loan Agreement.
- 4. All conditions referred to in Section 3 of the Loan Agreement to the making of a Credit Extension to be made on or about the date hereof have been satisfied or waived by Agent.
 - 5. No Material Adverse Change has occurred.
 - 6. The undersigned is an Authorized Signer.

[Balance of Page Intentionally Left Blank]

7A. The proceeds of the Term Loan Advance shall be disbursed as follows:

Disbursement from SVB:	
Loan Amount	\$
Plus:	
—Deposit Received	\$
Less:	
—Commitment Fee	(\$)
—[Interim Interest]	(\$)
—Lender's Legal Fees	(\$)*
Net Proceeds due from SVB:	\$
Disbursement from Hercules:	
Loan Amount:	\$
Less:	
—Commitment Fee	(\$)
—[Interim Interest]	(\$)
Net Proceeds due from Hercules:	\$
TOTAL TERM LOAN ADVANCE NET PROCEEDS FROM	
LENDERS	\$
[Disbursement from AGENT:] [Cumulative and not duplicative of the	
amount set forth above]	
Loan Amount	\$
Plus:	
—Deposit Received	\$
Less:	
—Commitment Fee	(\$)
—[Interim Interest]	(\$)
—Lender's Legal Fees	(\$)
Net Proceeds due from Agent	\$

[7B. Funds from Borrower Designated Deposit Account shall be disbursed as follows:]

SVB:		
Commitment Fee	\$	
Lender's Legal Fees	\$	
Hercules: Designated Deposit Account:		
Commitment Fee	\$	
Funds due from Borrower ("Total Funds") \$		

8A. The aggregate net proceeds of the Term Loan Advance shall be transferred to the Designated Deposit Account as follows:

Account Name:		<u> </u>
Bank Name:	Silicon Valley Bank	
Bank Address:	3003 Tasman Drive	
	Santa Clara, California 95054	
Account Number:		
ABA Number:		
Account Name:	VB to debit the Total Funds from th	e Designated Deposit Account set forth below:
Bank Name:	Silicon Valley Bank	
Bank Address:	3003 Tasman Drive	
	Santa Clara, California 95054	
Account Number:		
ABA Number:		

BORROWER:		
GROVE COLLABORATIVE, INC.		
By		
Name:		
Title:		
AGENT:		
SILICON VALLEY BANK, as Agent		
Ву		
Name:		
Title:		
LENDER:		
SILICON VALLEY BANK		
Ву		
Name:		
Title:		
LENDER:		
HERCULES CAPITAL, INC.		
By		
Name:		
Title:		

Dated as of the date first set forth above.

[Signature page to Disbursement Letter]

FIRST AMENDMENT TO MEZZANINE LOAN AND SECURITY AGREEMENT

This First Amendment to Mezzanine Loan and Security Agreement (this "Amendment") is entered into this 9th day of May, 2022, by and among (a) SILICON VALLEY BANK, a California corporation ("SVB"), in its capacity as administrative agent and collateral agent ("Agent"), (b) SILICON VALLEY BANK, a California corporation, as a lender, (c) HERCULES CAPITAL, INC., a Maryland corporation ("Hercules"), as a lender (SVB and Hercules and each of the other "Lenders" from time to time a party hereto are referred to herein collectively as the "Lenders" and each individually as a "Lender") and (d) GROVE COLLABORATIVE, INC., a Delaware public benefit corporation ("Borrower").

RECITALS

- A. Agent, Lenders and Borrower have entered into that certain Mezzanine Loan and Security Agreement dated as of April 30, 2021 (as the same may from time to time be further amended, modified, supplemented or restated, the "Loan Agreement").
 - **B.** The Lenders have extended credit to Borrower for the purposes permitted in the Loan Agreement.
- C. Borrower has entered into that certain Amended and Restated Agreement and Plan of Merger dated as of March 31, 2022 (the "Merger Agreement"), by and among (a) Borrower, (b) Virgin Group Acquisition Corp. II, a Cayman Islands exempted company ("Virgin Cayman"), (c) Treehouse Merger Sub, Inc., a Delaware corporation ("Merger Sub I"), and (d) Treehouse Merger Sub II, LLC, a Delaware limited liability company ("Merger Sub II"), which provides for, among other things, (i) the domestication of Virgin Cayman as a Delaware public benefit corporation ("Newco"), (ii) the merger of Merger Sub I with and into Borrower (the "Initial Merger"), with Borrower continuing as the surviving corporation (the "Initial Surviving Corporation") and (iii) immediately following the Initial Merger, the merger of the Initial Surviving Corporation with and into Merger Sub II, with Merger Sub II continuing as the surviving limited liability company, following the consummation of which, Merger Sub II will become a direct, wholly-owned subsidiary of Newco (steps (i), (ii) and (iii), collectively with the other transactions described in the Merger Agreement, the "Business Combination Transaction").
- **D.** Pursuant to the terms of the Merger Agreement, Virgin Cayman will cause to be filed with the U.S. Securities and Exchange Commission, an effective Registration Statement on Form S-4 (including the related proxy statement/prospectus) for the registration under the Securities Act of 1933, as amended, of Newco securities to be issued in connection with the Business Combination Transaction.
- E. Borrower has requested that the Lenders and Agent amend the Loan Agreement to make certain revisions to the Loan Agreement as more fully set forth herein.
- **F.** The Lenders and Agent have agreed to so amend certain provisions of the Loan Agreement, but only to the extent, in accordance with the terms, subject to the conditions and in reliance upon the representations and warranties set forth below.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing recitals and other good and valuable consideration, the receipt and adequacy of which is hereby acknowledged, and intending to be legally bound, the parties hereto agree as follows:

- 1. Definitions. Capitalized terms used but not defined in this Amendment shall have the meanings given to them in the Loan Agreement.
- 2. Amendments to Loan Agreement.
- **2.1 Section 6.2 (Financial Statements, Reports).** Clause (g) of Section 6.2 of the Loan Agreement is deleted in its entirety and replaced with the following:

"(g) prior to the consummation of the Business Combination Transaction, within five (5) Business Days of delivery by Borrower, copies of all material statements, reports and notices made generally available by Borrower to Borrower's security holders or to any holders of Subordinated Debt;"

- **2.2 Section 7.2 (Changes in Business, Management, Control or Business Locations).** Clause (c) of Section 7.2 of the Loan Agreement is deleted in its entirety and replaced with the following:
 - "(c) [reserved];"
- **2.3 Section 7.2 (Changes in Business, Management, Control, or Business Locations).** The second paragraph of Section 7.2 is deleted in its entirety and replaced with the following:

"Borrower shall not, without at least fifteen (15) days prior written notice to Agent: (A) add any new offices or business locations, including warehouses (unless such new offices or business locations contain less than Five Hundred Thousand Dollars (\$500,000) in Borrower's assets or property) or deliver any portion of the Collateral valued, individually or in the aggregate, in excess of Five Hundred Thousand Dollars (\$500,000) to a bailee at a location other than to a bailee and at a location already disclosed in the Perfection Certificate, (B) change its jurisdiction of organization, (C) change its organizational structure or type, (D) change its legal name, or (E) change any organizational number (if any) assigned by its jurisdiction of organization; provided that Borrower may, and need not deliver any prior written notice to Agent in order to, change its name, its organizational structure or type and its organizational number assigned by its jurisdiction of organization in connection with the Business Combination Transaction. If Borrower intends to add any new offices or business locations, including warehouses, containing in excess of One Million Dollars (\$1,000,000) of Borrower's assets or property, then Borrower will first receive the written consent of Agent, and the landlord of any such new offices or business locations, including warehouses, shall execute and deliver a landlord consent in form and substance satisfactory to Agent. If Borrower intends to deliver any portion of the Collateral valued, individually or in the aggregate, in excess of One Million Dollars (\$1,000,000) to a bailee, and Agent and such bailee are not already parties to a bailee agreement governing both the Collateral and the location to which Borrower intends to deliver the Collateral, then Borrower will first receive the written consent of Agent, and such bailee shall execute and deliver a bailee agreement in form and substance satisfactory to Agent."

2.4 Section 7.3 (Mergers or Acquisitions). Section 7.3 of the Loan Agreement is deleted in its entirety and replaced with the following:

"7.3 Mergers or Acquisitions. Merge or consolidate, or permit any of its Subsidiaries to merge or consolidate, with any other Person, or acquire, or permit any of its Subsidiaries to acquire, all or substantially all of the capital stock or property of another Person (including, without limitation, by the formation of any Subsidiary or pursuant to a Division), other than (i) a merger, consolidation or acquisition in which (A) Borrower has complied with the notice requirements applicable to prepayments hereunder, and (B) prior to or contemporaneously with the closing of such transaction, all Obligations are paid in full, and all of the Lenders' obligations to lend to Borrower are terminated, (ii) a Subsidiary may merge or consolidate into another Subsidiary or into Borrower and (iii) the Business Combination Transaction, so long as (A) the Business Combination Transaction is consummated on or prior to July 31, 2022 in accordance with the Merger Agreement without any amendments or modifications thereto or waivers to any conditions or terms set forth therein that, in each case, could reasonably be expected to be materially adverse to the Lenders and (B) promptly and substantially concurrent with the consummation of the Business Combination Transaction, (a) Merger Sub II shall provide to Agent a reaffirmation of its assumption of the obligations of the "Borrower" under this Agreement and the other Loan Documents by reason of the Business Combination Transaction and Holdings shall provide Agent a guaranty to become a Guarantor hereunder, together with documentation sufficient to grant Agent a first priority Lien (subject to Permitted Liens) in and to the assets of Merger Sub II and Holdings, all in form and substance reasonably satisfactory to Agent, (b) Holdings shall provide to Agent appropriate stock certificates and powers (if applicable) and financing statements, pledging all of the direct or beneficial ownership interest in Merger Sub II, in form and substance reasonably satisfactory to Agent; and (c) Merger Sub II and Holdings shall provide to Agent certificates duly executed by a Responsible Officer or secretary of each of Merger Sub II and Holdings with respect to their respective Operating Documents and Borrowing Resolutions, documentation and other information required to comply with applicable "know your customer" and anti-money-laundering rules and regulations, and long-form good standing certificates of Merger Sub II and Holdings certified by the Secretary of State of their states of incorporation or formation, in each case as of a date no earlier than 30 days prior to the date of the joinder of guaranty, as applicable."

2.5 Section 7.7 (Distributions; Investments). Section 7.7 of the Loan Agreement is deleted in its entirety and replaced with the following:

"7.7 Distributions; Investments. (a) Pay any dividends or make any distribution or payment or redeem, retire or purchase any capital stock or other equity interests; provided that (i) Borrower may convert any of its convertible securities into other securities pursuant to the terms of such convertible securities or otherwise in exchange thereof, (ii) Borrower may pay dividends solely in common stock and repurchase de minimis stock in lieu of issuing fractional shares; (iii) Borrower may repurchase, or, following the Business Combination Transaction, may pay dividends or make distributions to Holdings to enable it to repurchase, the stock of former employees, directors or consultants (including by the cancellation of indebtedness owed by such former Persons) pursuant to stock repurchase agreements so long as an Event of Default does not exist at the time of such repurchase and would not exist after giving effect to such repurchase, provided that the aggregate amount of all such repurchases does not exceed Two Hundred Fifty Thousand Dollars (\$250,000) per fiscal year; (iv) Borrower may distribute equity securities to former or current employees, officers, consultants or directors on the exercise of employee issuances pursuant to Borrower's stock incentive plan approved by the Board; (v) Borrower may repurchase common shares of Borrower's stock from its employees and directors in

connection with a bona fide equity financing of Borrower provided that (A) Borrower sells common or preferred shares to investors for an aggregate price equal to the repurchase price of such common shares repurchased by Borrower, and (B) no Event of Default exists at the time of such series of transactions and would not exist after giving effect to such transactions; and (vi) following the Business Combination Transaction, Borrower may make cash dividends on the equity interests of Borrower solely for the purpose of funding the following: (A) to permit Holdings to pay, in the event Borrower is a passthrough entity or files a consolidated, combined, unitary or similar type tax return with Holdings, U.S. federal and state and local income taxes then due and payable pursuant to those returns solely to the extent such taxes are imposed solely with respect to the income of Borrower and its Subsidiaries; (B) ordinary operating expenses and other transaction expenses of Holdings, including overhead and administrative expenses and fees as well as directors' or managers' fees which are reasonable and customary and incurred in the ordinary course of business and attributable to the ownership or operations of Borrower and its Subsidiaries; (C) organizational, operating and other transaction costs and expenses (including overhead and administrative costs and expenses, professional fees and Public Company Costs) of Holdings which are reasonable and customary and incurred in the ordinary course of business and attributable to the ownership or operations of Borrower and its Subsidiaries; or (D) to pay reasonable and customary salary, bonus, severance and other benefits payable to current and former directors, officers, employees, members of management, managers and/or consultants of Borrower, any Subsidiary or Holdings to the extent such salaries, bonuses, severance and other benefits are attributable to the ownership or operation of Borrower or the Subsidiaries; or (b) directly or indirectly make any Investment (including, without limitation, by the formation of any Subsidiary) other than (A) Permitted Investments and (B) loans and advances to Holdings in lieu of, and not in excess of the amount of, dividends permitted to be made to such Holdings in accordance with clause (a) above, or permit any of its Subsidiaries to do so."

2.6 Section 7.11 (Holding Company). New Section 7.11 of the Loan Agreement is inserted following Section 7.10 as follows:

"7.11 Holding Company. At all times following the Business Combination Transaction, Holdings shall not (a) conduct, transact or otherwise engage in, or commit to conduct, transact or otherwise engage in, any business or operations, (b) incur, create, assume or suffer to exist any Indebtedness or other liabilities or financial obligations or (c) own, lease, manage or otherwise operate any properties or assets, other than the following activities: (i) ownership of equity interests of the Borrowers and the exercise of rights and performance of obligations in connection therewith, (ii) the issuance of its own equity interests, (iii) the maintenance of its entity existence, (iv) activities and contractual rights incidental to maintenance of its entity existence, (v) participating in administrative, legal, tax, management and accounting matters as the parent of a consolidated group of companies including the Borrower and its Subsidiaries, (vi) entering into and performing its obligations under the Loan Documents and the Loan Documents (as defined in the Senior Loan Agreement) to which it is a party (including any amendments, restatements, amendments and restatements, supplements or modifications thereto, or consents in respect thereof), (vii) adoption of an option plan, equity incentive plan, employee stock purchase plan and related agreements and activities related thereto, (viii) guarantees of operating leases of its Subsidiaries, (ix) with respect to insurance policies and related contracts and agreements, or (x) activities otherwise expressly permitted to be taken by Holdings hereunder or reasonably incidental to any of the foregoing."

- 2.7 Section 8 (Events of Default). The "." following Section 8.10 is deleted and replaced with "; or", and new Section 8.11 is inserted following Section 8.10 as follows:
 - "8.11 Guaranty. (a) Any guaranty of any Obligations terminates or ceases for any reason to be in full force and effect; (b) any Guarantor does not perform any obligation or covenant under any guaranty of the Obligations that can be cured and has failed to cure the default within ten (10) days after the occurrence thereof; provided, however, that if the default cannot by its nature be cured within the ten (10) day period or cannot after diligent attempts by Guarantor be cured within such ten (10) day period, and such default is likely to be cured within a reasonable time, then Guarantor shall have an additional period (which shall not in any case exceed thirty (30) days) to attempt to cure such default, and within such reasonable time period the failure to cure the default shall not be deemed an Event of Default (but no Credit Extensions shall be made during such cure period); provided further, that a cure period shall not be applicable to a payment default arising from a failure to make a principal or interest payment when due; (c) any circumstance described in Sections 8.3 through 8.8 of this Agreement occurs with respect to any Guarantor."
 - **2.8 Section 13 (Definitions)**. The following defined terms set forth in Section 13.1 are deleted in their entirety and replaced with the

following:

- ""Final Payment" is a payment (in addition to and not in substitution for the regular monthly payments of principal plus accrued interest) equal to the aggregate original principal amount of the Term Loan Advances extended by the Lenders to Borrower hereunder multiplied by eight and three quarters of one percent (8.75%) due on the earliest to occur of (a) the Term Loan Maturity Date, (b) the payment in full of the Term Loan Advances, (c) as required by Section 2.2(d) or Section 2.2(e), or (d) the termination of this Agreement."
- 2.9 Section 13 (Definitions). The following new defined terms are hereby inserted alphabetically in Section 13.1:
- ""Business Combination Transaction" has the meaning given to such term in the First Amendment."
- ""Holdings" is Virgin Group Acquisition Corp. II, a Cayman Islands exempted company, which shall be domesticated as a Delaware public benefit corporation in connection with the Business Combination Transaction, and its successors and permitted assigns."
- ""Merger Agreement" has the meaning given to such term in the First Amendment."
- ""Public Company Costs" is (i) costs, fees and expenses associated with, or in anticipation of, or preparation for (A) any initial public offering (or any initial public offering proposed and not consummated) and (B) compliance with the requirements of the Sarbanes-Oxley Act of 2002 and the rules and regulations promulgated in connection therewith and (ii) costs, fees and expenses relating to compliance with the provisions of the Securities Act and the Exchange Act (and any similar Requirements of Law under any other applicable jurisdiction), as applicable to companies with equity or debt securities held by the public, the rules of national securities exchanges applicable to companies with listed equity or debt securities, directors', managers' and/or employees' compensation or other costs to the extent attributable to being a public company, officer and director fee and expense reimbursement to the extent attributable to being a public company, costs, fees and expenses relating to investor relations, shareholder meetings and reports to shareholders or debtholders associated with being a public company, directors' and officers' insurance and other legal and other professional fees, listing fees and other costs and/or expenses associated with being a public company."

- ""First Amendment" is the First Amendment to Mezzanine Loan and Security Agreement, dated as of May 9, 2022, by and among Borrower, Agent and the Lenders."
- **2.10 Section 13 (Definitions)**. The following defined term set forth in Section 13.1 is deleted in its entirety:
- ""Key Person" is Borrower's Chief Financial Officer, who is Servaes Tholen as of the Effective Date."

3. Limitation of Amendments.

- **3.1** The amendments set forth in Section 2, above, are effective for the purposes set forth herein and shall be limited precisely as written and shall not be deemed to (a) be a consent to any amendment, waiver or modification of any other term or condition of any Loan Document, or (b) otherwise prejudice any right or remedy which Agent may now have or may have in the future under or in connection with any Loan Document.
- 3.2 This Amendment shall be construed in connection with and as part of the Loan Documents and all terms, conditions, representations, warranties, covenants and agreements set forth in the Loan Documents, except as herein amended, are hereby ratified and confirmed and shall remain in full force and effect.
 - **4. Representations and Warranties.** To induce the Lenders and Agent to enter into this Amendment, Borrower hereby represents and warrants to the Lenders and Agent as follows:
- **4.1** Immediately after giving effect to this Amendment (a) the representations and warranties contained in the Loan Documents are true, accurate and complete in all material respects as of the date hereof (except to the extent such representations and warranties relate to an earlier date, in which case they are true and correct as of such date), and (b) no Event of Default has occurred and is continuing;
- **4.2** Borrower has the power and authority to execute and deliver this Amendment and to perform its obligations under the Loan Agreement, as amended by this Amendment;
- **4.3** The organizational documents of Borrower delivered to Agent on or around April 30, 2021 remain true, accurate and complete and have not been amended, supplemented or restated and are and continue to be in full force and effect;
- **4.4** The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, have been duly authorized;
- **4.5** The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, do not and will not contravene (a) any law or regulation binding on or affecting Borrower, (b) any contractual restriction with a Person binding on Borrower, (c) any order, judgment or decree of any court or other governmental or public body or authority, or subdivision thereof, binding on Borrower, or (d) the organizational documents of Borrower, or applicable consents or waivers have been obtained;

- 4.6 The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, do not require any order, consent, approval, license, authorization or validation of, or filing, recording or registration with, or exemption by any governmental or public body or authority, or subdivision thereof, binding on Borrower, except as already has been obtained or made: and
- **4.7** This Amendment has been duly executed and delivered by Borrower and is the binding obligation of Borrower, enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, liquidation, moratorium or other similar laws of general application and equitable principles relating to or affecting creditors' rights.
- 5. Ratification of Loan Documents. Borrower has read this Amendment and consents to the terms hereof and further acknowledges and confirms that all of its obligations under the Loan Agreement and the other Loan Documents are in full force and effect in accordance with their respective terms without setoff, defense, counter-claim or claims in recoupment.
- 6. Senior Loan Agreement. The Lenders and Agent hereby consent to the amendments and waivers contained in the Second Amendment to the Senior Loan Agreement executed in connection herewith.
- **7. Integration**. This Amendment and the Loan Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of this Amendment and the Loan Documents merge into this Amendment and the Loan Documents.
- **8. Counterparts.** This Amendment may be executed in any number of counterparts and all of such counterparts taken together shall be deemed to constitute one and the same instrument. Each party hereto may execute this Amendment by electronic means and recognizes and accepts the use of electronic signatures and records by any other party hereto in connection with the execution and storage hereof.
- **9.** Effectiveness. As a condition precedent to the effectiveness of this Amendment, Agent shall have received the following documents prior to or concurrently with this Amendment, each in form and substance reasonably satisfactory to Agent:
- 9.1 This Amendment and the Supplement to Intellectual Property Security Agreement duly executed on behalf of the parties hereto and thereto:
- 9.2 The duly executed Second Amendment to Senior Loan Agreement and satisfaction of all conditions precedent to the effectiveness thereof therein;
- **9.3** Long-form good standing certificate of Borrower certified by the Secretary of State of Delaware, as of a date no earlier than thirty (30) days prior to the date hereof;
- 9.4 Borrower's payment of (i) a fully earned, nonrefundable amendment fee in the amount of One Hundred Fifty Thousand Dollars (\$150,000.00) to be shared between the Lenders pursuant to their respective Term Loan Commitment Percentages and (ii) Lenders' reasonable legal fees and expenses incurred in connection with this Amendment, to the extent Borrower shall have received invoices documenting such fees or expenses on or prior to the date hereof (or such later date as the Borrower shall reasonably agree); and

9.5 such other documents as Agent may reasonably request to effectuate the terms of this Amendment.

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed and delivered as of the date first written above.

AGENT BORROWER SILICON VALLEY BANK GROVE COLLABORATIVE, INC. By: /s/ Trefor Bacon By: /s/ Stuart Landesberg Name: Trefor Bacon Name: Stuart Landesberg Title: Director Title: Chief Executive Officer LENDERS SILICON VALLEY BANK By: /s/ Trefor Bacon Name: Trefor Bacon Title: Director HERCULES CAPITAL, INC. By: /s/ Zhuo Huang Name: Zhuo Huang

[Signature Page to First Amendment to Mezzanine Loan and Security Agreement]

Title: Associate General Counsel

SECOND AMENDMENT TO MEZZANINE LOAN AND SECURITY AGREEMENT

This Second Amendment to Mezzanine Loan and Security Agreement (this "Amendment") is entered into this 16th day of June, 2022, by and between (a) SILICON VALLEY BANK, a California corporation ("SVB"), in its capacity as administrative agent and collateral agent ("Agent"), (b) SILICON VALLEY BANK, a California corporation, as a lender, (c) HERCULES CAPITAL, INC., a Maryland corporation ("Hercules"), as a lender (SVB and Hercules and each of the other "Lenders" from time to time a party hereto are referred to herein collectively as the "Lenders" and each individually as a "Lender") and (d) GROVE COLLABORATIVE, INC., a Delaware public benefit corporation, formerly known as Treehouse Merger Sub II, LLC, a Delaware limited liability company ("Borrower").

RECITALS

- **A.** Agent, Lenders and Borrower have entered into that certain Mezzanine Loan and Security Agreement dated as of April 30, 2021, as amended by that certain First Amendment to Mezzanine Loan and Security Agreement dated as of May 9, 2022 and as affected by that certain Assumption Agreement, dated as of the date hereof (as the same may from time to time be further amended, modified, supplemented or restated, the "Loan Agreement").
 - B. Lenders has extended credit to Borrower for the purposes permitted in the Loan Agreement.
- C. Borrower has requested that Agent and Lenders amend the Loan Agreement to make certain revisions to the Loan Agreement as more fully set forth herein.
- **D.** Agent and Lenders have agreed to so amend certain provisions of the Loan Agreement, but only to the extent, in accordance with the terms, subject to the conditions and in reliance upon the representations and warranties set forth below.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing recitals and other good and valuable consideration, the receipt and adequacy of which is hereby acknowledged, and intending to be legally bound, the parties hereto agree as follows:

- 1. Definitions. Capitalized terms used but not defined in this Amendment shall have the meanings given to them in the Loan Agreement.
- 2. Amendments to Loan Agreement.
- **2.1 Section 6.2 (Financial Statements, Reports).** Clauses (a), (c), (d), (e), (g), (h), (i), and (l) of Section 6.2 of the Loan Agreement are deleted in their entirety and replaced with the following:

- "(a) Monthly Financial Statements. As soon as available, but no later than thirty (30) days after the last day of each month, a company-prepared consolidated and consolidating (if applicable) balance sheet and income statement covering Holdings' consolidated operations and key performance indicators for such month in a form acceptable to Agent (the "Monthly Financial Statements"); provided, that, for the month ending on May 31, 2022, the Monthly Financial Statements shall cover the Borrower's (rather than Holdings') consolidated operations and key performance indicators for such month;"
- "(c) <u>Board Projections</u>. As soon as available but no later than the earlier of forty-five (45) days after the last day of the fiscal year of Holdings or forty-five (45) days after approval by the Board, and within seven (7) days of any Board-approved updates or amendments thereto, (1) annual consolidated operating budgets (including income statements, balance sheets and cash flow statements, by month) for the upcoming fiscal year of Holdings, and (2) annual consolidated financial projections for the following fiscal year (on a monthly or quarterly basis), in each case as approved by the Board and commensurate in form and substance with those provided to the Board, together with any related business forecasts used in the preparation of such annual financial projections;"
- "(d) <u>Annual Audited Financial Statements</u>. As soon as available, and in any event within two hundred ten (210) days after the last day of Holdings' fiscal year, audited consolidated financial statements prepared under GAAP, consistently applied, together with an unqualified opinion (provided, however, Holdings' unqualified opinion on financial statements may contain a qualification as to going concern typical for venture backed companies similar to Holdings) on the financial statements from an independent certified public accounting firm reasonably acceptable to Agent;"
- "(e) <u>Board Decks</u>. Within forty-five (45) days of presentation to the board of directors of Holdings, copies of board decks and similar materials for standard scheduled board meetings, provided that such board decks and materials may be redacted by Holdings to exclude (i) documents and matters that may be subject to attorney-client privilege; (ii) information provided in connection with or for closed sessions of the board or its committees; (iii) information relating to Lenders, and (iv) any material confidential proprietary information and trade secrets;"
- "(g) Other Statements. Within five (5) Business Days of delivery by Holdings, copies of all material statements, reports and notices made generally available by Holdings to Holdings' security holders or to any holders of Subordinated Debt;"
- "(h) <u>SEC Filings</u>. In the event that Borrower becomes subject to the reporting requirements under the Exchange Act within five (5) days of filing, copies of all periodic and other reports, proxy statements and other materials filed by Borrower and/or any Guarantor with the SEC, any Governmental Authority succeeding to any or all of the functions of the SEC or with any national securities exchange, or

distributed to its shareholders, as the case may be. Documents required to be delivered pursuant to the terms hereof (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date on which Borrower and/or any Guarantor posts such documents, or provides a link thereto, on Borrower's and/or any Guarantor's website address; provided, however, Borrower and/or such Guarantor, as applicable, shall promptly notify Agent and Lenders in writing (which may be by electronic mail) of the posting of any such documents;"

- "(i) <u>Legal Action Notice</u>. Prompt report of any legal actions pending or threatened in writing against Holdings or any of its Subsidiaries that could reasonably be expected to result in damages or costs to Borrower or any of its Subsidiaries of, individually or in the aggregate, One Million Five Hundred Thousand Dollars (\$1,500,000) or more (except for items set forth in the Perfection Certificate dated on or around the Effective Date):"
- "(I) Other Financial Information. Promptly, from time to time, such other information regarding Holdings, Borrower or compliance with the terms of any Loan Documents as reasonably requested by Agent or any Lender."
- **2.2 Section 13 (Definitions).** The defined term "Permitted Indebtedness" set forth in Section 13.1 of the Loan Agreement is amended to delete clause (g) in its entirety and replace it with the following:
 - "(g) (i) Borrower's credit cards maintained with third parties in the ordinary course of Borrower's business, provided that the aggregate credit limits available thereunder do not exceed One Million Five Hundred Thousand Dollars (\$1,500,000) at any time and (ii) Indebtedness owed to any Person consisting of unpaid premiums for insurance (including property, casualty, business interruption, directors and officers or liability insurance) of Holdings, the Borrower or any of their respective Subsidiaries; provided such insurance financing arrangement has been approved in writing by Agent;"
- **2.3 Section 13 (Definitions).** The defined term "Permitted Liens" set forth in Section 13.1 of the Loan Agreement is amended to add the following new clause (k) thereto immediately following clause (j):
 - "(k) Liens on rights under insurance policies and the premiums and proceeds thereof securing obligations permitted by clause (g)(ii) of the definition of "Permitted Indebtedness"."
- **2.4 Section 13 (Definitions).** The defined term "Holdings" set forth in Section 13.1 of the Loan Agreement is deleted in its entirety and replaced with the following:

""Holdings" is Grove Collaborative Holdings, Inc., a Delaware public benefit corporation, formerly known as Virgin Group Acquisition Corp. II, a Cayman Islands exempted company, and its successors and permitted assigns."

3. Limitation of Amendments.

- 3.1 The amendments set forth in Section 2, above, are effective for the purposes set forth herein and shall be limited precisely as written and shall not be deemed to (a) be a consent to any amendment, waiver or modification of any other term or condition of any Loan Document, or (b) otherwise prejudice any right or remedy which Agent may now have or may have in the future under or in connection with any Loan Document.
- 3.2 This Amendment shall be construed in connection with and as part of the Loan Documents and all terms, conditions, representations, warranties, covenants and agreements set forth in the Loan Documents, except as herein amended, are hereby ratified and confirmed and shall remain in full force and effect.
 - **4. Representations and Warranties.** To induce the Lenders and Agent to enter into this Amendment, Borrower hereby represents and warrants to the Lenders and Agent as follows:
- **4.1** Immediately after giving effect to this Amendment (a) the representations and warranties contained in the Loan Documents are true, accurate and complete in all material respects as of the date hereof (except to the extent such representations and warranties relate to an earlier date, in which case they are true and correct as of such date), and (b) no Event of Default has occurred and is continuing;
- **4.2** Borrower has the power and authority to execute and deliver this Amendment and to perform its obligations under the Loan Agreement, as amended by this Amendment;
- **4.3** The organizational documents of Borrower delivered to Agent on or around the date hereof are true, accurate and complete and have not been amended, supplemented or restated and are and continue to be in full force and effect;
- **4.4** The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, have been duly authorized;
- 4.5 The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, do not and will not contravene (a) any law or regulation binding on or affecting Borrower, (b) any contractual restriction with a Person binding on Borrower, (c) any order, judgment or decree of any court or other governmental or public body or authority, or subdivision thereof, binding on Borrower, or (d) the organizational documents of Borrower, or applicable consents or waivers have been obtained;

- 4.6 The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, do not require any order, consent, approval, license, authorization or validation of, or filing, recording or registration with, or exemption by any governmental or public body or authority, or subdivision thereof, binding on Borrower, except as already has been obtained or made; and
- **4.7** This Amendment has been duly executed and delivered by Borrower and is the binding obligation of Borrower, enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, liquidation, moratorium or other similar laws of general application and equitable principles relating to or affecting creditors' rights.
 - **5. Ratification of Loan Documents**. Borrower has read this Amendment and consents to the terms hereof and further acknowledges and confirms that all of its obligations under the Loan Agreement and the other Loan Documents are in full force and effect in accordance with their respective terms without setoff, defense, counter-claim or claims in recoupment.
 - 6. Senior Loan Agreement. Agent and Lenders hereby consent to the amendments and waivers contained in the Third Amendment to Second Amended and Restated Loan and Security Agreement executed in connection herewith.
 - **7. Integration**. This Amendment and the Loan Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of this Amendment and the Loan Documents merge into this Amendment and the Loan Documents.
 - 8. Counterparts. This Amendment may be executed in any number of counterparts and all of such counterparts taken together shall be deemed to constitute one and the same instrument. Each party hereto may execute this Amendment by electronic means and recognizes and accepts the use of electronic signatures and records by any other party hereto in connection with the execution and storage hereof.
 - 9. Effectiveness. As a condition precedent to the effectiveness of this Amendment, Agent shall have received the following documents prior to or concurrently with this Amendment, each in form and substance reasonably satisfactory to Agent:
- **9.1** This Amendment, the Unconditional Secured Guaranty and Pledge Agreement, the Assumption Agreement, the Corporate Borrowing Certificate and the Guarantor's Certificate to Guaranty, each duly executed on behalf of the parties hereto and thereto;
- 9.2 the duly executed Third Amendment to Second Amended and Restated Loan and Security Agreement and satisfaction of all conditions precedent to the effectiveness thereof therein; and
 - 9.3 such other documents as Bank may reasonably request to effectuate the terms of this Amendment.

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed and delivered as of the date first written above.

AGENT

SILICON VALLEY BANK

/s/ Jeff Sereno By:

Name: Jeff Sereno Title: Vice President

LENDERS

SILICON VALLEY BANK

By: /s/ Jeff Sereno Name: Jeff Sereno Title: Vice President

HERCULES CAPITAL, INC.

/s/ ZhuoHuang By:

Zhuo Huang Name:

Title: Associate General Counsel BORROWER

GROVE COLLABORATIVE, INC.

By: /s/ Stuart Landesberg Name: Stuart Landesberg Title: Chief Executive Officer

[Signature Page to Second Amendment to Mezzanine Loan and Security Agreement]

SECOND AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT

THIS SECOND AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT (this "Agreement") dated as of July 29, 2020 (the "Effective Date") between SILICON VALLEY BANK, a California corporation ("Bank"), and GROVE COLLABORATIVE, INC., a Delaware corporation ("Borrower"), provides the terms on which Bank shall lend to Borrower and Borrower shall repay Bank. The parties agree as follows:

RECITALS

- A. Bank and Borrower entered into that certain Amended and Restated Loan and Security Agreement dated as of April 2, 2018 (as the same has been amended, modified, supplemented, renewed, or otherwise modified, from to time, the "Prior Loan Agreement"). Pursuant to the Prior Loan Agreement, Bank made certain loans and other credit accommodations available to Borrower.
 - B. Borrower has requested that Bank, and Bank has agreed to, replace, amend, and restate the Prior Loan Agreement in its entirety.

1. ACCOUNTING AND OTHER TERMS

Accounting terms not defined in this Agreement shall be construed following GAAP. Calculations and determinations under this Agreement must be made following GAAP, except, in each case applicable hereunder, with respect to unaudited financial statements, for the absence of footnotes and subject to year-end adjustments. Notwithstanding any terms in this Agreement to the contrary, for purposes of any financial covenant and other financial calculations in this Agreement (other than for purposes of updating the Borrowing Base) which are made in whole or in part based upon the Availability Amount as of the last day of a particular month, calculations relying on information from a Borrowing Base Statement shall be derived from the Borrowing Base Statement delivered within seven (7) days of month end pursuant to Section 6.2(a) (and not, for clarity, any more recent Borrowing Base Statement delivered after such period), and the actual delivery date of such Borrowing Base Statement shall be deemed to be the last day of the applicable month. Capitalized terms not otherwise defined in this Agreement shall have the meanings set forth in Section 13. All other terms contained in this Agreement, unless otherwise indicated, shall have the meaning provided by the Code to the extent such terms are defined therein.

2. LOAN AND TERMS OF PAYMENT

2.1 Promise to Pay. Borrower hereby unconditionally promises to pay Bank the outstanding principal amount of all Credit Extensions and accrued and unpaid interest thereon as and when due in accordance with this Agreement.

2.2 Revolving Line.

(a) <u>Availability</u>. Subject to the terms and conditions of this Agreement and to deduction of Reserves, Bank shall make Advances upon Borrower's request, not exceeding the Availability Amount. Amounts borrowed under the Revolving Line may be repaid and, prior to the Revolving Line Maturity Date, reborrowed, subject to the applicable terms and conditions precedent herein.

(b) Letters of Credit Sublimit.

(i) As part of the Revolving Line, Bank shall issue or has issued Letters of Credit denominated in Dollars or a Foreign Currency for Borrower's account. The aggregate Dollar Equivalent amount utilized for the issuance of Letters of Credit shall at all times reduce the amount otherwise available for Advances under the Revolving Line. The aggregate Dollar Equivalent of the face amount of outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit and any Letter of Credit Reserve) may not exceed Six Million Dollars (\$6,000,000).

(ii) If, on the Revolving Line Maturity Date (or the effective date of any termination of this Agreement), there are any outstanding Letters of Credit, then on such date Borrower shall provide to Bank cash collateral in an amount equal to at least one hundred five percent (105.0%) for Letters of Credit denominated in Dollars or at least one hundred ten percent (110.0%) for Letters of Credit denominated in a Foreign Currency, in each case of the aggregate Dollar Equivalent of the face amount of all such Letters of Credit plus all interest, fees, and costs due or estimated by Bank to become due in connection therewith, to secure all of the Obligations relating to such Letters of Credit. All Letters of Credit shall be in form and substance acceptable to Bank in its sole discretion and shall be subject to the terms and conditions of Bank's standard Application and Letter of Credit Agreement (the "Letter of Credit Application"). Borrower agrees to execute any further documentation in connection with the Letters of Creditas Bank may reasonably request. Borrower further agrees to be bound by the regulations and interpretations of the issuer of any Letters of Credit guaranteed by Bank and opened for Borrower's account or by Bank's interpretations of any Letter of Credit issued by Bank for Borrower's account, and Borrower understands and agrees that Bank shall not be liable for any error, negligence, or mistake, whether of omission or commission, in following Borrower's instructions or those contained in the Letters of Credit or any modifications, amendments, or supplements thereto.

- (iii) The obligation of Borrower to immediately reimburse Bank for drawings made under Letters of Credit shall be absolute, unconditional, and irrevocable, and shall be performed strictly in accordance with the terms of this Agreement, such Letters of Credit, and the Letter of Credit Application.
- (iv) Borrower may request that Bank issue a Letter of Credit payable in a Foreign Currency. If a demand for payment is made under any such Letter of Credit, Bank shall treat such demand as an Advance to Borrower of the Dollar Equivalent of the amount thereof (plus fees and charges in connection therewith such as wire, cable, SWIFT or similar charges).
- (v) To guard against fluctuations in currency exchange rates, upon the issuance of any Letter of Credit payable in a Foreign Currency, Bank shall create a reserve (the "Letter of Credit Reserve") under the Revolving Line in an amount equal to a percentage (which percentage shall be determined by Bank in its good faith business discretion) of the face amount of such Letter of Credit. The amount of the Letter of Credit Reserve may be adjusted by Bank from time to time to account for fluctuations in the exchange rate. The availability of funds under the Revolving Line shall be reduced by the amount of such Letter of Credit Reserve for as long as such Foreign Currency Letter of Credit remains outstanding.
- (c) <u>Termination; Repayment</u>. The Revolving Line terminates on the Revolving Line Maturity Date, when the principal amount of all Advances, the unpaid interest thereon, and all other Obligations relating to the Revolving Line shall be immediately due and payable.

2.3 Growth Capital Advances.

- (a) Growth Capital Advances. Bank extended certain Growth Capital Advances (as defined in the Prior Loan Agreement and hereinafter referred to as the "Growth Capital Advances") in the aggregate original principal amount of Three Million Nine Hundred Thousand Dollars (\$3,900,000). Borrower acknowledges and agrees that as of the Effective Date, the aggregate outstanding principal balance on the Growth Capital Advances is Two Million Eight Hundred Sixteen Thousand Six Hundred Sixty Six Dollars and Eighty Cents (\$2,816,666.80), which remains outstanding and is continued as an Obligation hereunder as of the Effective Date, and that such sum is not subject to any offset or defense of any kind whatsoever, and in the event Borrower has any offsets or defenses thereto, Borrower hereby irrevocably waives all such offsets and defenses. The Obligations owing with respect to the Growth Capital Advances have not been extinguished or discharged hereby and the execution of this Agreement is not intended to and shall not cause or result in a novation with respect to the Growth Capital Advances. Borrower will continue to repay the outstanding balance of the Growth Capital Advances (including interest on the outstanding balance) in accordance with the terms set forth herein.
- (b) Interest Payments. With respect to each Growth Capital Advance, on the Payment Date of each month, Borrower shall make monthly payments of interest, in arrears, on the principal amount of such Growth Capital Advance at the rate set forth in Section 2.5(a)(i).

- (c) Repayment. Commencing on the Growth Capital Amortization Date and continuing on each Payment Date thereafter, Borrower shall repay each Growth Capital Advance in (i) twenty-six (26) equal monthly installments of principal, plus (ii) monthly payments of accrued interest at the rate set forth in Section 2.5(a)(i). All outstanding principal and accrued and unpaid interest under the Growth Capital Advances, and all other outstanding Obligations with respect to such Growth Capital Advances, are due and payable in full on the Growth Capital Maturity Date.
- (d) <u>Permitted Prepayment</u>. Borrower shall have the option to prepay the entire outstanding amount of an Growth Capital Advance, provided Borrower (i) delivers written notice to Bank of its election to prepay such Growth Capital Advance at least five (5) days prior to such prepayment, and (ii) pays, on the date of such prepayment (A) the applicable amount of outstanding principal being prepaid with respect to such Growth Capital Advance plus accrued and unpaid interest thereon, (B) the Prepayment Fee, (C) the Final Payment (without duplication) with respect to the outstanding principal amount of Growth Capital Advance being prepaid, (D) the Deferred Final Payment, and (E) all other sums, if any, that shall have become due and payable with respect to such Growth Capital Advance, including interest at the Default Rate with respect to any past due amounts.
- (e) <u>Mandatory Prepayment Upon an Acceleration</u>. If the Growth Capital Advances are accelerated by Bank following the occurrence and during the continuance of an Event of Default, Borrower shall immediately pay to Bank an amount equal to the sum of (i) all outstanding principal plus accrued and unpaid interest with respect to the Growth Capital Advances, (ii) the Prepayment Fee, (iii) the Final Payment, (iv) the Deferred Final Payment, and (v) all other sums, if any, that shall have become due and payable with respect to the Growth Capital Advances, including interest at the Default Rate with respect to any past due amounts.
- 2.4 Overadvances. If, at any time, the sum of (a) the outstanding principal amount of any Advances, plus (b) the face amount of any outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit and any Letter of Credit Reserve), exceeds the lesser of either the Revolving Line or the Borrowing Base, Borrower shall immediately pay to Bank in cash the amount of such excess (such excess, the "Overadvance"). Without limiting Borrower's obligation to repay Bank any Overadvance, in the event that Borrower does not repay any Overadvances within one (1) Business Day, Borrower agrees to pay Bank interest on the outstanding amount of such Overadvance, on demand, at a per annum rate equal to the rate that is otherwise applicable to Advances plus five percent (5.0%).

2.5 Payment of Interest on the Credit Extensions.

(a) Interest Rate.

- (i) <u>Advances</u>. Subject to Section 2.5(b), the outstanding principal amount of each Advance shall accrue interest at a floating per annum rate equal to the Prime Rate. Interest shall be payable monthly in accordance with Section 2.5(d) below.
- (ii) <u>Growth Capital Advances</u>. Subject to Section 2.5(b), the outstanding principal amount of the Growth Capital Advance shall accrue interest at a floating per annum rate equal to the Prime Rate. Interest shall be payable monthly in accordance with Section 2.5(d) below.
- (b) <u>Default Rate</u>. Immediately upon the occurrence and during the continuance of an Event of Default, Obligations shall bear interest at a rate per annum which is five percent (5.0%) above the rate that is otherwise applicable thereto (the "**Default Rate**") unless Bank otherwise elects to impose a lesser rate in its sole discretion. Fees and expenses which are required to be paid by Borrower pursuant to the Loan Documents (including, without limitation, Bank Expenses) but are not paid when due shall bear interest until paid at a rate equal to the highest rate applicable to the Obligations. Payment or acceptance of the increased interest rate provided in this Section 2.5(b) is not a permitted alternative to timely payment and shall not constitute a waiver of any Event of Default or otherwise prejudice or limit any rights or remedies of Bank.
- (c) Adjustment to Interest Rate. Changes to the interest rate of any Credit Extension based on changes to the Prime Rate shall be effective on the effective date of any change to the Prime Rate and to the extent of any such change.

(d) <u>Payment; Interest Computation</u>. Interest is payable monthly in arrears on the Payment Date of each month and shall be computed on the basis of a three hundred sixty (360)-day year for the actual number of days elapsed. In computing interest, (i) all payments received after 12:00 p.m. Pacific time on any day shall be deemed received at the opening of business on the next Business Day, and (ii) the date of the making of any Credit Extension shall be included and the date of payment shall be excluded; provided, however, that if any Credit Extension is repaid on the same day on which it is made, such day shall be included in computing interest on such Credit Extension.

2.6 Fees. Borrower shall pay to Bank:

- (a) <u>Revolving Line Commitment Fee</u>. A fully earned, non-refundable commitment fee in connection with the Revolving Line in the amount of Twenty Thousand Dollars (\$20,000) on the Effective Date;
- (b) <u>Termination Fee</u>. Upon termination of this Agreement or the termination of the Revolving Line for any reason prior to the Revolving Line Maturity Date, in addition to the payment of any other amounts then-owing, a termination fee in an amount equal to Three Hundred Seventy-Five Thousand Dollars (\$375,000), provided that no termination fee shall be charged if the Revolving Line is replaced or refinanced with a new facility from Bank:
 - (c) Prepayment Fee. The Prepayment Fee, if due hereunder;
 - (d) Final Payment. The Final Payment, when due hereunder;
 - (e) Deferred Final Payment. The Deferred Final Payment, when due hereunder;
- (f) <u>Bank Expenses</u>. All Bank Expenses (including reasonable attorneys' fees and expenses for documentation and negotiation of this Agreement) incurred through and after the Effective Date, when due (or, if no stated due date, upon demand by Bank).
- (g) Fees Fully Earned. Unless otherwise provided in this Agreement or in a separate writing by Bank, Borrower shall not be entitled to any credit, rebate, or repayment of any fees earned by Bank pursuant to this Agreement notwithstanding any termination of this Agreement or the suspension or termination of Bank's obligation to make loans and advances hereunder. Bank may deduct amounts owing by Borrower under the clauses of this Section 2.6 pursuant to the terms of Section 2.7(c). Bank shall provide Borrower written notice of deductions made from the Designated Deposit Account pursuant to the terms of the clauses of this Section 2.6.
- (h) Letter of Credit Fee. Bank's customary fees and expenses for the issuance or renewal of Letters of Credit, upon the issuance of such Letter of Credit, each anniversary of the issuance during the term of such Letter of Credit, and upon the renewal of such Letter of Credit by Bank.
- (i) <u>Unused Revolving Line Facility Fee</u>. Payable monthly in arrears on the last day of each calendar month prior to the Revolving Line Maturity Date, and on the Revolving Line Maturity Date, a fee (the "**Unused Revolving Line Facility Fee**") in an amount equal to one fifth of one percent (0.20%) per annum of the average unused portion of the Revolving Line, as calculated by Bank, computed on the basis of a year with the applicable number of days as set forth in Section 2.5(d). The unused portion of the Revolving Line, for purposes of this calculation, shall be calculated on a calendar year basis and shall equal the difference between (i) the Revolving Line, and (ii) the average for the applicable monthly period of the daily closing balance of the Revolving Line outstanding plus the sum of the aggregate amount of outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit and any Letter of Credit Reserve). For the avoidance of doubt, no Unused Revolving Line Facility Fees shall accrue for any period of time following termination of the Revolving Line.

2.7 Payments; Application of Payments; Debit of Accounts.

(a) All payments to be made by Borrower under any Loan Document shall be made in immediately available funds in Dollars, without setoff or counterclaim, before 12:00 p.m. Pacific time on the date when due. Payments of principal and/or interest received after 12:00 p.m. Pacific time are considered received at the opening of business on the next Business Day. When a payment is due on a day that is not a Business Day, the payment shall be due the next Business Day, and additional fees or interest, as applicable, shall continue to accrue until paid.

- (b) Bank has the exclusive right to determine the order and manner in which all payments with respect to the Obligations may be applied. Borrower shall have no right to specify the order or the accounts to which Bank shall allocate or apply any payments required to be made by Borrower to Bank or otherwise received by Bank under this Agreement when any such allocation or application is not specified elsewhere in this Agreement.
- (c) Bank may debit any of Borrower's deposit accounts, provided that so long as no Event of Default has occurred and is continuing, Bank shall first debit the Designated Deposit Account, and then, if funds are insufficient, any other deposit accounts of Borrower for principal and interest payments or any other amounts Borrower owes Bank when due. Bank shall endeavor to promptly notify Borrower when it debits Borrower's accounts for any payments other than on account of principal or interest, but its failure to do so shall not be a breach hereunder. These debits shall not constitute a set-off.
- 2.8 Withholding. Payments received by Bank from Borrower under this Agreement will be made free and clear of and without deduction for any and all present or future taxes, levies, imposts, duties, deductions, withholdings, assessments, fees or other charges imposed by any Governmental Authority (including any interest, additions to tax or penalties applicable thereto), in each case other than in respect of Excluded Taxes. Specifically, however, if at any time any Governmental Authority, applicable law, regulation or international agreement requires Borrower to make any withholding or deduction from any such payment or other sum payable hereunder to Bank, in each case, other than in respect of Excluded Taxes, Borrower hereby covenants and agrees that the amount due from Borrower with respect to such payment or other sum payable hereunder will be increased to the extent necessary to ensure that, after the making of such required withholding or deduction, Bank receives a net sum equal to the sum which it would have received had no withholding or deduction been required, and Borrower shall pay the full amount withheld or deducted to the relevant Governmental Authority. Borrower will, upon request, furnish Bank with proof reasonably satisfactory to Bank indicating that Borrower has made such withholding payment; provided, however, that Borrower need not make any withholding payment if the amount or validity of such withholding payment is contested in good faith by appropriate and timely proceedings and as to which payment in full is bonded or reserved against by Borrower. The agreements and obligations of Borrower contained in this Section 2.8 shall survive the termination of this Agreement.

3. CONDITIONS OF LOANS

- 3.1 Conditions Precedent to Initial Credit Extension. Bank's obligation to make the initial Credit Extension is subject to the condition precedent that Bank shall have received, in form and substance satisfactory to Bank, such documents, and completion of such other matters, as Bank may reasonably deem necessary or appropriate, including, without limitation:
 - (a) duly executed signatures to the Loan Documents;
 - (b) duly executed signatures to the completed Borrowing Resolutions for Borrower;
 - (c) the Perfection Certificate of Borrower, together with the duly executed signature thereto; and
 - (d) payment of the fees and Bank Expenses then due as specified in Section 2.6 hereof.
- **3.2 Conditions Precedent to all Credit Extensions.** Bank's obligations to make each Credit Extension, including the initial Credit Extension, is subject to the following conditions precedent:
- (a) timely receipt of (i) the Credit Extension request and any materials and documents required by Section 3.4 and (ii) with respect to the request for a Growth Capital Advance, an executed Payment/Advance Form and any materials and documents required by Section 3.4;

- (b) the representations and warranties in this Agreement shall be true, accurate, and complete in all material respects on the date of the proposed Credit Extension and/or of the Payment/Advance Form, as applicable, and on the Funding Date of each Credit Extension; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects as of such date, and no Event of Default shall have occurred and be continuing or result from the Credit Extension. Each Credit Extension is Borrower's representation and warranty on that date that the representations and warranties in this Agreement remain true, accurate, and complete in all material respects; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects as of such date; and
 - (c) Bank determines to its satisfaction that there has not been any Material Adverse Change.

3.3 Covenant to Deliver.

- (a) Except as otherwise provided in Section 3.3(b), Borrower agrees to deliver to Bank each item required to be delivered to Bank under this Agreement as a condition precedent to any Credit Extension. Borrower expressly agrees that a Credit Extension made prior to the receipt by Bank of any such item shall not constitute a waiver by Bank of Borrower's obligation to deliver such item, and the making of any Credit Extension in the absence of a required item shall be in Bank's sole discretion.
- (b) Unless otherwise provided in writing, within forty-five (45) days after the Effective Date, Bank shall have received, in form and substance satisfactory to Bank, the following:
- (i) evidence satisfactory to Bank that the insurance policies and endorsements required by Section 6.7 hereof are in full force and effect, together with appropriate evidence showing lender loss payable and/or additional insured clauses or endorsements in favor of Bank; and
- (ii) a landlord's consent in favor of Bank for each of Borrower's leased locations at 35 Conewago Road, Unit B, Elizabethtown, PA 17022, by the respective landlord thereof, together with the duly executed signatures thereto.

3.4 Procedures for Borrowing.

- (a) Advances. Subject to the prior satisfaction of all other applicable conditions to the making of an Advance set forth in this Agreement, to obtain an Advance, Borrower (via an individual duly authorized by an Administrator) shall notify Bank (which notice shall be irrevocable) by electronic mail by 12:00 p.m. Pacific time on the Funding Date of the Advance. Such notice shall be made by Borrower through Bank's online banking program, provided, however, if Borrower is not utilizing Bank's online banking program, then such notice shall be in a written format acceptable to Bank that is executed by an Authorized Signer. Bank shall have received satisfactory evidence that the Board has approved that such Authorized Signer may provide such notices and request Advances. In connection with any such notification, Borrower must promptly deliver to Bank by electronic mail or through Bank's online banking program such reports and information, including without limitation, sales journals, cash receipts journals, a Borrowing Base Statement, and accounts receivable aging reports, as Bank may request in its sole discretion. Bank shall credit proceeds of an Advance to the Designated Deposit Account. Bank may make Advances under this Agreement based on instructions from an Authorized Signer or without instructions if the Advances are necessary to meet Obligations which have become due.
 - 3.5 Cash Pledge Agreement. Bank hereby agrees that the Cash Pledge Agreement (as defined in the Prior Loan Agreement) is hereby terminated.

4. CREATION OF SECURITY INTEREST.

4.1 Grant of Security Interest. Borrower hereby grants Bank, to secure the payment and performance in full of all of the Obligations, a continuing security interest in, and pledges to Bank, the Collateral, wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof.

Borrower acknowledges that it previously has entered, and/or may in the future enter, into Bank Services Agreements with Bank. Regardless of the terms of any Bank Services Agreement, Borrower agrees that any amounts Borrower owes Bank thereunder shall be deemed to be Obligations hereunder and that it is the intent of Borrower and Bank to have all such Obligations secured by the first priority perfected security interest in the Collateral granted herein (subject only to Permitted Liens that are permitted pursuant to the terms of this Agreement to have superior priority to Bank's Lien in this Agreement).

If this Agreement is terminated, Bank's Lien in the Collateral shall continue until the Obligations (other than inchoate indemnity obligations) are repaid in full in cash. Upon payment in full in cash of the Obligations (other than inchoate indemnity obligations) and at such time as Bank's obligation to make Credit Extensions has terminated, Bank shall, at the sole cost and expense of Borrower, promptly release its Liens in the Collateral and all rights therein shall revert to Borrower, and Bank shall, upon reasonable request from Borrower and at Borrower's sole cost and expense, deliver to Borrower written evidence of the termination of such liens and any other documents reasonably necessary to terminate such liens. In the event (x) all Obligations (other than inchoate indemnity obligations), except for Bank Services, are satisfied in full, and (y) this Agreement is terminated, Bank shall terminate the security interest granted herein upon Borrower providing cash collateral acceptable to Bank in its good faith business judgment for Bank Services, if any. In the event such Bank Services consist of outstanding Letters of Credit, Borrower shall provide to Bank cash collateral in an amount equal to (x) if such Letters of Credit are denominated in Dollars, then at least one hundred five percent (105.0%); and (y) if such Letters of Credit are denominated in a Foreign Currency, then at least one hundred ten percent (110.0%), of the Dollar Equivalent of the face amount of all such Letters of Credit plus all interest, fees, and costs due or to become due in connection therewith (as estimated by Bank in its business judgment), to secure all of the Obligations relating to such Letters of Credit.

- 4.2 Priority of Security Interest. Borrower represents, warrants, and covenants that the security interest granted herein is and shall at all times continue to be a first priority perfected security interest in the Collateral (subject only to Permitted Liens that are permitted pursuant to the terms of this Agreement to have superior priority to Bank's Lien under this Agreement). If Borrower shall acquire a commercial tort claim which is reasonably likely to exceed Twenty-Five Thousand Dollars (\$25,000), Borrower shall promptly notify Bank in a writing signed by Borrower of the general details thereof and grant to Bank in such writing a security interest therein and in the proceeds thereof, all upon the terms of this Agreement, with such writing to be in form and substance reasonably satisfactory to Bank.
- 4.3 Authorization to File Financing Statements. Borrower hereby authorizes Bank to file financing statements, without notice to Borrower, with all appropriate jurisdictions to perfect or protect Bank's interest or rights in the Collateral hereunder, including a notice that any disposition of the Collateral (except as permitted under this Agreement), by either Borrower or any other Person, shall be deemed to violate the rights of Bank under the Code. Such financing statements may indicate the Collateral as "all assets of the Debtor" or words of similar effect, or as being of an equal or lesser scope, or with greater detail, all in Bank's discretion.

5. REPRESENTATIONS AND WARRANTIES

Borrower represents and warrants as follows:

5.1 Due Organization, Authorization; Power and Authority. Borrower is duly existing and in good standing as a Registered Organization in its jurisdiction of formation and is qualified and licensed to do business and is in good standing in any other jurisdiction in which the conduct of its business or its ownership of property requires that it be qualified except where the failure to do so could not reasonably be expected to have a material adverse effect on Borrower's business. In connection with this Agreement, Borrower has delivered to Bank a completed certificate signed by Borrower, entitled "Perfection Certificate" (the "Perfection Certificate"). Borrower represents and warrants to Bank that (a) Borrower's exact legal name is that indicated on the Perfection Certificate and on the

signature page hereof; (b) Borrower is an organization of the type and is organized in the jurisdiction set forth in the Perfection Certificate; (c) the Perfection Certificate accurately sets forth Borrower's organizational identification number or accurately states that Borrower has none; (d) the Perfection Certificate accurately sets forth Borrower's place of business, or, if more than one, its chief executive office as well as Borrower's mailing address (if different than its chief executive office); (e) Borrower (and each of its predecessors) has not, in the past five (5) years, changed its jurisdiction of formation, organizational structure or type, or any organizational number assigned by its jurisdiction; and (f) all other information set forth on the Perfection Certificate pertaining to Borrower and each of its Subsidiaries is accurate and complete in all material respects (it being understood and agreed that Borrower may from time to time update certain information in the Perfection Certificate after the Effective Date to the extent permitted by one or more specific provisions in this Agreement).

The execution, delivery and performance by Borrower of the Loan Documents to which it is a party have been duly authorized, and do not (i) conflict with any of Borrower's organizational documents, (ii) contravene, conflict with, constitute a default under or violate any material Requirement of Law, (iii) contravene, conflict or violate any applicable order, writ, judgment, injunction, decree, determination or award of any Governmental Authority by which Borrower or any of its Subsidiaries or any of their property or assets may be bound or affected, (iv) require any action by, filing, registration, or qualification with, or Governmental Approval from, any Governmental Authority (except such Governmental Approvals which have already been obtained and are in full force and effect) or (v) conflict with, contravene, constitute a default or breach under, or result in or permit the termination or acceleration of, any material agreement by which Borrower is bound. Borrower is not in default under any agreement to which it is a party or by which it is bound in which the default could reasonably be expected to have a material adverse effect on Borrower's business.

5.2 Collateral. Borrower has good title to, rights in, and the power to transfer each item of the Collateral upon which it purports to grant a Lien hereunder, free and clear of any and all Liens except Permitted Liens. Borrower has no Collateral Accounts at or with any bank or financial institution other than Bank or Bank's Affiliates except for the Collateral Accounts described in the Perfection Certificate delivered to Bank in connection herewith and which Borrower has taken such actions as reasonably requested by Bank to give Bank a perfected security interest therein, pursuant to the terms (and subject to the exceptions) of Section 6.8(b). The Accounts are bona fide, existing obligations of the Account Debtors.

No Collateral valued in excess of Two Hundred Fifty Thousand Dollars (\$250,000) (other than laptops and other portable electronic items) is in the possession of any third party bailee (such as a warehouse) except as otherwise provided in the Perfection Certificate or as permitted pursuant to Section 7.2. None of the components of the Collateral valued in excess of Two Hundred Fifty Thousand Dollars (\$250,000) (other than laptops and other portable electronic items) shall be maintained at locations other than as provided in the Perfection Certificate or as permitted pursuant to Section 7.2.

All Inventory is in all material respects of good and marketable quality, free from material defects.

Borrower is the sole owner of the Intellectual Property which it owns or purports to own except for(a) non-exclusive licenses granted to its customers in the ordinary course of business, (b) over-the-counter software that is commercially available to the public, and (c) material Intellectual Property licensed to Borrower and noted on the Perfection Certificate (as the same may be updated from time to time pursuant to Section 5.1). Each Patent which it owns or purports to own and which is material to Borrower's business is valid and enforceable, and no part of the Intellectual Property which Borrower owns or purports to own and which is material to Borrower's business has been judged invalid or unenforceable, in whole or in part. To the best of Borrower's knowledge, no claim has been made that any part of the Intellectual Property violates the rights of any third party except to the extent such claim would not reasonably be expected to have a material adverse effect on Borrower's business.

Except for non-exclusive licenses granted to Borrower in the ordinary course of business or as noted on the Perfection Certificate (as the same may be updated from time to time), Borrower is not a party to, nor is it bound by, any Restricted License.

- 5.3 Inventory. For any item of Inventory consisting of Eligible Inventory in any Borrowing Base Statement, such Inventory (i) consists of finished goods, in good, new, and salable condition and packaging or shipping materials, which is not perishable, returned, consigned, obsolete, not sellable, damaged, or defective, and is not comprised of demonstrative or custom inventory (provided, however, Borrower's private label brand inventory shall not constitute custom inventory), works in progress, or supplies; (ii) meets all applicable governmental standards; (iii) has been manufactured in compliance with the Fair Labor Standards Act; (iv) is not subject to any Liens, except the first priority Liens granted or in favor of Bank under this Agreement or any of the other Loan Documents and Permitted Liens; (v) is located in the United States at the locations identified by Borrower in the Perfection Certificate where it maintains Inventory (or at any location permitted under Section 7.2, as updated from time to time in accordance with this Agreement), including, without limitation, Borrower's warehouse premises located in Reno, Nevada and Harrisburg, Pennsylvania; and (vi) the Funding Date for the corresponding Advance is less than one hundred eighty (180) days after the Purchase Date.
- **5.4 Litigation**. There are no actions or proceedings pending or, to the knowledge of any Responsible Officer, threatened in writing by or against Borrower or any of its Subsidiaries that could reasonably be expected to involve more than, individually or in the aggregate, Three Hundred Seventy-Five Thousand Dollars (\$375,000).
- 5.5 Financial Statements; Financial Condition. All consolidated financial statements for Borrower and any of its Subsidiaries delivered to Bank by submission to the Financial Statement Repository or otherwise submitted to Bank fairly present in all material respects Borrower's consolidated financial condition and Borrower's consolidated results of operations as of the dates and periods covered thereby. There has not been any material deterioration in Borrower's consolidated financial condition since the date of the most recent financial statements submitted to the Financial Statement Repository or otherwise submitted to Bank.
- **5.6 Solvency.** The fair salable value of Borrower's consolidated assets (including goodwill minus disposition costs) exceeds the fair value of Borrower's liabilities; Borrower is not left with unreasonably small capital after the transactions in this Agreement; and Borrower is able to pay its debts (including trade debts) as they mature.
- 5.7 Regulatory Compliance. Borrower is not an "investment company" or a company "controlled" by an "investment company" under the Investment Company Act of 1940, as amended. Borrower is not engaged as one of its important activities in extending credit for margin stock (under Regulations X, T and U of the Federal Reserve Board of Governors). Borrower (a) has complied in all material respects with all Requirements of Law, and (b) has not violated any Requirements of Law the violation of which could reasonably be expected to have a material adverse effect on its business. None of Borrower's or any of its Subsidiaries' properties or assets has been used by Borrower or any Subsidiary or, to the best of Borrower's knowledge, by previous Persons, in disposing, producing, storing, treating, or transporting any hazardous substance other than legally. Borrower and each of its Subsidiaries have obtained all consents, approvals and authorizations of, made all declarations or filings with, and given all notices to, all Governmental Authorities that are necessary to continue their respective businesses as currently conducted, except where failure to obtain or make such consents, declarations, filings or notices could not reasonably be expected to have a material adverse effect on Borrower's businesses.
- **5.8 Subsidiaries; Investments.** Borrower does not own any stock, partnership, or other ownership interest or other equity securities except for Permitted Investments.
- 5.9 Tax Returns and Payments; Pension Contributions Borrower has timely filed all required tax returns and reports (or extensions therefor), and Borrower has timely paid (or timely filed extensions for) for all other foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower except (a) to the extent such taxes are being contested in good faith by appropriate proceedings promptly instituted and diligently conducted, so long as such reserve or other appropriate provision, if any, as shall be required in conformity with GAAP shall have been made therefor, or (b) if such taxes, assessments, deposits and contributions do not, individually or in the aggregate, exceed Two Hundred Fifty Thousand Dollars (\$250,000).

To the extent Borrower defers payment of any contested taxes or has a tax matter that exceeds Two Hundred Fifty Thousand Dollars (\$250,000), Borrower shall (i) notify Bank in writing of the commencement of, and any material development in, the proceedings, and (ii) post bonds or take any other steps required to prevent the Governmental Authority levying such contested taxes from obtaining a Lien upon any of the Collateral that is other than a "Permitted Lien." Borrower is unaware of any claims or adjustments proposed for any of Borrower's prior tax years which could result in additional taxes becoming due and payable by Borrower in excess of Two Hundred Fifty

Thousand Dollars (\$250,000). Borrower has paid all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms, and Borrower has not withdrawn from participation in, and has not permitted partial or complete termination of, or permitted the occurrence of any other event with respect to, any such plan which could reasonably be expected to result in any liability of Borrower, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency.

- **5.10 Use of Proceeds.** Borrower shall use the proceeds of the Credit Extensions solely as working capital and to fund its general business requirements and not for personal, family, household or agricultural purposes.
- 5.11 Full Disclosure. No written representation, warranty or other statement of Borrower in any report, certificate, or written statement submitted to the Financial Statement Repository or otherwise submitted to Bank, as of the date such written representation, warranty, or other statement was made, taken together with all such written reports, written certificates and written statements submitted the Financial Statement Repository or otherwise submitted to Bank, contains any untrue statement of a material fact or omits to state a material fact necessary to make the written statements contained in the reports, certificates, or written statements not misleading (it being recognized by Bank that the projections and forecasts provided by Borrower in good faith and based upon reasonable assumptions are not viewed as facts and that actual results during the period or periods covered by such projections and forecasts may differ from the projected or forecasted results).
- **5.12 Definition of "Knowledge."** For purposes of the Loan Documents, whenever a representation or warranty is made to Borrower's knowledge or awareness, to the "best of" Borrower's knowledge, or with a similar qualification, knowledge or awareness means the actual knowledge, after reasonable investigation, of any Responsible Officer.

6. AFFIRMATIVE COVENANTS

Borrower shall do all of the following:

6.1 Government Compliance.

- (a) Maintain its and all its Subsidiaries' legal existence and good standing in their respective jurisdictions of formation and maintain qualification in each other jurisdiction in which the failure to so qualify would reasonably be expected to have a material adverse effect on Borrower's business or operations. Borrower shall comply, and have each Subsidiary comply, in all material respects, with all laws, ordinances and regulations to which it is subject.
- (b) Obtain all of the Governmental Approvals necessary for the performance by Borrower of its obligations under the Loan Documents to which it is a party and the grant of a security interest to Bank in all of its property. Borrower shall promptly provide copies of any such obtained Governmental Approvals to Bank.
- **6.2 Financial Statements, Reports.** Provide Bank with the following by submitting to the Financial Statement Repository or otherwise submitting to Bank:
- (a) a Borrowing Base Statement (and any schedules related thereto and including any other information requested by Bank with respect to Borrower's Accounts) upon each request for an Advance and (i) within thirty (30) days after the last day of each month when a Streamline Period is in effect, and (ii) no later than Friday of each week when a Streamline Period is not in effect;
- (b) (i) monthly accounts receivable agings, aged by invoice date, (ii) monthly accounts payable agings, aged by invoice date, and outstanding or held check registers, if any, and (iii) inventory reports for the Inventory valued on a reasonable efforts first-in, first-out basis at the lower of cost or market (in accordance with GAAP) or such other inventory reports as are requested by Bank in its good faith business judgment, (x) within thirty (30) days after the last day of each month when a Streamline Period is in effect, and (y) no later than Friday of each week when a Streamline Period is not in effect;

- (c) as soon as available, but no later than thirty (30) days after the last day of each month, a company-prepared consolidated and consolidating (if applicable) balance sheet and income statement covering Borrower's consolidated operations and key performance indicators for such month in a form acceptable to Bank (the "Monthly Financial Statements");
- (d) within thirty (30) days after the last day of each month and together with the Monthly Financial Statements, a completed Compliance Statement, confirming that, as of the end of such month, Borrower was in full compliance with all of the terms and conditions of this Agreement, and setting forth calculations showing compliance with the financial covenants set forth in this Agreement and such other information as Bank may reasonably request, including, without limitation, a statement that at the end of such month there were no held checks;
- (e) as soon as available but no later than the earlier of forty-five (45) days after the last day of the fiscal year of Borrower or forty-five (45) days after approval by the Board, and within seven (7) days of any Board-approved updates or amendments thereto, (1) annual operating budgets (including income statements, balance sheets and cash flow statements, by month) for the upcoming fiscal year of Borrower, and (2) annual financial projections for the following fiscal year (on a monthly or quarterly basis), in each case as approved by the Board and commensurate in form and substance with those provided to the Board, together with any related business forecasts used in the preparation of such annual financial projections;
- (f) as soon as available, and in any event within two hundred ten (210) days after the last day of Borrower's fiscal year, audited consolidated financial statements prepared under GAAP, consistently applied, together with an unqualified opinion (provided, however, Borrower's unqualified opinion on financial statements may contain a qualification as to going concern typical for venture backed companies similar to Borrower) on the financial statements from an independent certified public accounting firm reasonably acceptable to Bank;
- (g) in the event that Borrower becomes subject to the reporting requirements under the Exchange Act within five (5) days of filing, copies of all periodic and other reports, proxy statements and other materials filed by Borrower and/or any Guarantor with the SEC, any Governmental Authority succeeding to any or all of the functions of the SEC or with any national securities exchange, or distributed to its shareholders, as the case may be. Documents required to be delivered pursuant to the terms hereof (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date on which Borrower posts such documents, or provides a link thereto, on Borrower's website on the internet at Borrower's website address; provided, however, Borrower shall promptly notify Bank in writing (which may be by electronic mail) of the posting of any such documents;
- (h) within five (5) Business Days of delivery by Borrower, copies of all material statements, reports and notices made generally available by Borrower to Borrower's security holders or to any holders of Subordinated Debt;
- (i) prompt report of any legal actions pending or threatened in writing against Borrower or any of its Subsidiaries that could reasonably be expected to result in damages or costs to Borrower or any of its Subsidiaries of, individually or in the aggregate, Three Hundred Seventy-Five Thousand Dollars (\$375,000) or more;
- (j) promptly, from time to time, such other information regarding Borrower or compliance with the terms of any Loan Documents as reasonably requested by Bank; and
- (k) prompt written notice of any changes to the beneficial ownership information set out in the Beneficial Ownership Information Disclosure Form. Borrower understands and acknowledges that Bank relies on such true, accurate and up-to-date beneficial ownership information to meet Bank's regulatory obligations to obtain, verify and record information about the beneficial owners of its legal entity customers.

Any submission by Borrower of a Compliance Statement, a Borrowing Base Statement, or any other financial statement submitted to the Financial Statement Repository pursuant to this Section 6.2 or otherwise submitted to Bank shall be deemed to be a representation by Borrower that (i) as of the date of such Compliance Statement, Borrowing Base Statement, or other financial statement, the information and calculations set forth therein are true, accurate, and

correct, (ii) as of the end of the compliance period forth in such submission, Borrower is in complete compliance with all required covenants except as noted in such Compliance Statement, Borrowing Base Statement, or other financial statement, as applicable; (iii) as of the date of such submission, no Events of Default have occurred or are continuing, except as noted in such Compliance Statement; (iv) all representations and warranties other than any representations or warranties that are made as of a specific date in Section 5 remain true and correct in all material respects as of the date of such submission except as noted in such Compliance Statement, Borrowing Base Statement, or other financial statement, as applicable; (v) as of the date of such submission, Borrower and each of its Subsidiaries have timely filed all required tax returns and reports, and Borrower has timely paid all foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower except as otherwise permitted pursuant to the terms of Section 5.9; and (vi) as of the date of such submission, no Liens have been levied or claims made against Borrower or any of its Subsidiaries relating to unpaid employee payroll or benefits of which Borrower has not previously provided written notification to Bank.

6.3 Accounts Receivable.

- (a) Schedules and Documents Relating to Accounts. Borrower shall deliver to Bank transaction reports and schedules of collections, as provided in Section 6.2, on Bank's standard forms; provided, however, that Borrower's failure to execute and deliver the same shall not affect or limit Bank's Lien and other rights in all of Borrower's Accounts, nor shall Bank's failure to advance or lend against a specific Account affect or limit Bank's Lien and other rights therein. If requested by Bank, Borrower shall furnish Bank with copies (or, at Bank's request, originals) of all contracts, orders, invoices, and other similar documents, and all shipping instructions, delivery receipts, bills of lading, and other evidence of delivery, for any goods the sale or disposition of which gave rise to such Accounts. In addition, Borrower shall deliver to Bank, on its request, the originals of all instruments, chattel paper, security agreements, guarantees and other documents and property evidencing or securing any Accounts, in the same form as received, with all necessary indorsements, and copies of all credit memos.
- (b) <u>Disputes</u>. Borrower shall promptly notify Bank of all disputes or claims that would reasonably be expected to exceed Two Hundred Fifty Thousand Dollars (\$250,000) relating to Accounts. Borrower may forgive (completely or partially), compromise, or settle any Account for less than payment in full, or agree to do any of the foregoing so long as (i) Borrower does so in good faith, in a commercially reasonable manner, in the ordinary course of business, in arm's length transactions, and reports the same to Bank in the regular reports provided to Bank; (ii) no Event of Default has occurred and is continuing; and (iii) after taking into account all such discounts, settlements, and forgiveness, the total outstanding Advances will not exceed the lesser of the Revolving Line or the Borrowing Base.
- (c) Collection of Accounts. Borrower has established a "blocked account" as specified by Bank (the 'Cash Collateral Account'). Amounts from the Merchant Processing Accounts shall be transferred into the Cash Collateral Account in accordance with Section 6.8. Whether or not an Event of Default has occurred and is continuing, Borrower shall immediately deliver all payments on and proceeds of Accounts to the Cash Collateral Account. Subject to Bank's right to maintain a reserve pursuant to Section 6.3(d), all amounts received in the Cash Collateral Account shall be (i) applied to immediately pay off or pay down the Overadvance in the event, and in the amount, that the outstanding principal amount of any Advances exceeds the lesser of either the Revolving Line or the Borrowing Base, (ii) at Bank's option in its sole discretion, applied to immediately reduce the Obligations while an Event of Default is continuing or a Streamline Period is not in effect, and (iii) if a Streamline Period is in effect, transferred by Bank on a daily basis to Borrower's operating account with Bank. Borrower hereby authorizes Bank to transfer to the Cash Collateral Account any amounts that Bank reasonably determines are proceeds of the Accounts (provided that Bank is under no obligation to do so and this allowance shall in no event relieve Borrower of its obligations hereunder).
- (d) <u>Reserves</u>. Notwithstanding any terms in this Agreement to the contrary, at times when an Event of Default exists, Bank may hold any proceeds of the Accounts and any amounts in the Cash Collateral Account that are not applied to the Obligations pursuant to Section 6.3(c) above as a reserve to be applied to any Obligations regardless of whether such Obligations are then due and payable.

- (e) <u>Returns</u>. Provided no Event of Default has occurred and is continuing, if any Account Debtor returns any Inventory to Borrower valued in excess of One Hundred Thousand Dollars (\$100,000), Borrower shall promptly (i) determine the reason for such return, (ii) issue a credit memorandum to the Account Debtor in the appropriate amount, and (iii) provide a copy of such credit memorandum to Bank, upon request from Bank. In the event any attempted return occurs after the occurrence and during the continuance of any Event of Default, Borrower shall hold the returned Inventory in trust for Bank, and immediately notify Bank of the return of the Inventory.
- (f) <u>Verifications</u>; <u>Confirmations</u>; <u>Credit Quality</u>; <u>Notifications</u>. Bank may, from time to time, (i) verify and confirm directly with the respective Account Debtors the validity, amount and other matters relating to the Accounts, either in the name of Borrower or Bank or such other name as Bank may choose, and notify any Account Debtor of Bank's security interest in such Account and/or (ii) conduct a credit check of any Account Debtor to approve any such Account Debtor's credit.
- (g) No Liability. Bank shall not be responsible or liable for any shortage or discrepancy in, damage to, or loss or destruction of, any goods, the sale or other disposition of which gives rise to an Account, or for any error, act, omission, or delay of any kind occurring in the settlement, failure to settle, collection or failure to collect any Account, or for settling any Account in good faith for less than the full amount thereof, nor shall Bank be deemed to be responsible for any of Borrower's obligations under any contract or agreement giving rise to an Account. Nothing herein shall, however, relieve Bank from liability for its own gross negligence or willful misconduct.
- **6.4 Remittance of Proceeds.** Except as otherwise provided in Section 6.3(c), Section 7.2, and Permitted Investments, deliver, in kind, all proceeds arising from the disposition of any Collateral to Bank in the original form in which received by Borrower not later than the following Business Day after receipt by Borrower, to be applied to the Obligations (a) prior to an Event of Default, pursuant to the terms of Section 6.3(c) hereof, and (b) after the occurrence and during the continuance of an Event of Default, pursuant to the terms of Section 9.4 hereof; provided that, if no Event of Default has occurred and is continuing, Borrower shall not be obligated to remit to Bank the proceeds of the sale of worn out or obsolete Equipment disposed of by Borrower in good faith in an arm's length transaction for an aggregate purchase price of One Hundred Thousand Dollars (\$100,000) or less (for all such transactions in any fiscal year). Borrower agrees that it will not commingle proceeds of Collateral with any of Borrower's other funds or property, but will hold such proceeds separate and apart from such other funds and property and in an express trust for Bank. Nothing in this Section 6.4 limits the restrictions on disposition of Collateral set forth elsewhere in this Agreement, including, without limitation, Section 7.2 and Permitted Investments.
- 6.5 Taxes; Pensions. Timely file, and require each of its Subsidiaries to timely file, or timely filed all required extensions therefor, all required tax returns and reports and timely pay, and require each of its Subsidiaries to timely pay, all foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower and each of its Subsidiaries, except for deferred payment of any taxes contested pursuant to the terms of Section 5.9 hereof, or other taxes which are less than Two Hundred Fifty Thousand Dollars (\$250,000) and shall deliver to Bank, on demand, appropriate certificates attesting to such payments, and pay all amounts necessary to fund all present pension, profit sharing and deferred compensation plans in accordance with their terms
- 6.6 Access to Collateral; Books and Records. At reasonable times, on five (5) Business Days' notice (provided no notice is required if an Event of Default has occurred and is continuing), Bank, or its agents, shall have the right to inspect the Collateral and the right to audit and copy Borrower's Books. The foregoing inspections and audits shall be conducted no more often than (i) once every twelve (12) months when a Streamline Period is in effect, or (ii) once every six (6) months when a Streamline Period is not in effect (or more frequently as Bank in its good faith business discretion determines that conditions warrant), unless an Event of Default has occurred and is continuing in which case such inspections and audits shall occur as often as Bank shall determine is necessary. The foregoing inspections and audits shall be conducted at Borrower's expense and the charge therefor shall be One Thousand Dollars (\$1,000) per person per day (or such higher amount as shall represent Bank's then-current standard charge for the same), plus reasonable out-of-pocket expenses. In the event Borrower and Bank schedule an audit more than eight (8) days in advance, and Borrower cancels or seeks to or reschedules the audit with less than eight (8) days written notice to Bank, then (without limiting any of Bank's rights or remedies) Borrower shall pay Bank a fee of Two Thousand Dollars (\$2,000) plus any reasonable and documented out-of-pocket expenses incurred by Bank to compensate Bank for the anticipated costs and expenses of the cancellation or rescheduling. Bank and Borrower hereby acknowledge that inventory appraisals conducted by a third party satisfactory to Bank, at Borrower's sole cost and expense, shall be conducted no more often than (x) once every twelve (12) months when a Streamline Period is in effect, on (y) once every six (6) months when a Streamline Period is not in effect, unless an Event of Default has occurred and is continuing in which case such inventory appraisals shall occur as often as Bank shall dete

6.7 Insurance.

- (a) Keep its business and the Collateral insured for risks and in amounts standard for companies in Borrower's industry and location and as Bank may reasonably request. Insurance policies shall be in a form, with financially sound and reputable insurance companies that are not Affiliates of Borrower, and in amounts that are reasonably satisfactory to Bank. All property policies shall have a lender's loss payable endorsement showing Bank as the sole lender loss payee. All liability policies shall show, or have endorsements showing, Bank as an additional insured. Bank shall be named as lender loss payee and/or additional insured with respect to any such insurance providing coverage in respect of any Collateral. Upon Borrower's request, Bank will confirm that the insurance that Borrower maintains is satisfactory to Bank as of such date upon receipt of evidence of insurance from Borrower.
- (b) Ensure that proceeds payable under any property policy are, at Bank's option, payable to Bank on account of the Obligations. Notwithstanding the foregoing, (a) so long as no Event of Default has occurred and is continuing, Borrower shall have the option of applying the proceeds of any casualty policy up to Five Hundred Thousand Dollars (\$500,000) in the aggregate per fiscal year for all losses under all casualty policies in any one (1) year, toward the replacement or repair of destroyed or damaged property; provided that any such replaced or repaired property (i) shall be of equal or like value as the replaced or repaired Collateral and (ii) shall be deemed Collateral in which Bank has been granted a first priority security interest, and (b) after the occurrence and during the continuance of an Event of Default, all proceeds payable under such casualty policy shall, at the option of Bank, be payable to Bank on account of the Obligations.
- (c) At Bank's request, Borrower shall deliver certified copies of insurance policies and evidence of all premium payments. Each provider of any such insurance required under this Section 6.7 shall agree, by endorsement upon the policy or policies issued by it or by independent instruments furnished to Bank, that it will give Bank twenty (20) days prior written notice (ten (10) days for non-payment of premium) before any such policy or policies shall be materially altered or canceled. If Borrower fails to obtain insurance as required under this Section 6.7 or to pay any amount or furnish any required proof of payment to third persons and Bank, Bank may make all or part of such payment or obtain such insurance policies required in this Section 6.7, and take any action under the policies Bank deems prudent.

6.8 Accounts.

- (a) Borrower, and any Subsidiary of Borrower shall maintain all of its operating accounts and excess cash with Bank or Bank's Affiliates; provided, however, that Borrower may maintain its accounts with merchant processors (including, without limitation, PayPal, Stripe, and other similar processors) ("Merchant Processing Accounts") so long as Borrower transfers all the funds therein to the Cash Collateral Account within five (5) Business Days.
- (b) In addition to and without limiting the restrictions in (a), Borrower shall provide Bank five (5) days prior written notice before establishing any Collateral Account at or with any bank or financial institution other than Bank or Bank's Affiliates. For each Collateral Account that Borrower at any time maintains, Borrower shall cause the applicable bank or financial institution (other than Bank) at or with which any Collateral Account is maintained to execute and deliver a Control Agreement or other appropriate instrument with respect to such Collateral Account to perfect Bank's Lien in such Collateral Account in accordance with the terms hereunder which Control Agreement may not be terminated without the prior written consent of Bank. The provisions of the previous sentence shall not apply to (i) deposit accounts exclusively used for payroll, payroll taxes, and other employee wage and benefit payments to or for the benefit of Borrower's employees and identified to Bank by Borrower as such or (ii) its accounts with merchant processors so long as Borrower transfers all the funds therein to the Cash Collateral Account.
- (c) Borrower, and any Subsidiary of Borrower shall obtain any business credit card exclusively from Bank, except as permitted in the defined term "Permitted Indebtedness" part (g) of this Agreement.

- **6.9 Financial Covenants.** Maintain the following financial covenant, subject to periodic reporting, on a consolidated basis with respect to Borrower and its Subsidiaries:
- (a) Minimum Liquidity. Borrower shall maintain, at all times, to be tested on the last day of each month, Liquidity of not less than Forty-Five Million Dollars (\$45,000,000).

Notwithstanding the foregoing, Borrower's failure to comply with the financial covenants set forth in this Section 6.9 shall not in and of itself constitute an Event of Default with respect to the Growth Capital Advances and shall only apply as to and with respect to any corresponding Event of Default under this Agreement caused by such failure to comply, including, without limitation, pursuant to Section 8.2(a), solely with respect to the Advances under the Revolving Line, provided that upon the request of Borrower to implement a new financial covenant structure, Bank agrees to evaluate such request in its good faith business discretion.

6.10 Protection and Registration of Intellectual Property Rights.

- (a) (i) Use commercially reasonable efforts to protect, defend and maintain the validity and enforceability of its Intellectual Property that is material to Borrower's business; (ii) promptly advise Bank in writing of material infringements or any other event that could reasonably be expected to materially and adversely affect the value of its material Intellectual Property; and (iii) not allow any Intellectual Property material to Borrower's business to be abandoned, forfeited or dedicated to the public without Bank's written consent.
- (b) To the extent not already disclosed to Bank, if Borrower (i) obtains any Patent, registered Trademark, registered Copyright, registered mask work, or any pending application for any of the foregoing, whether as owner, licensee or otherwise, or (ii) applies for any Patent or the registration of any Trademark, then (A) as to Copyrights, Borrower shall promptly provide written notice thereof to Bank and (B) as to any other new Intellectual Property, Borrower shall provide notice to Bank in the Compliance Statement next due, and in each case shall execute such intellectual property security agreements and other documents and take such other actions as Bank may request in its good faith business judgment to perfect and maintain a first priority perfected security interest in favor of Bank in such property. If Borrower decides to register any Copyrights or mask works in the United States Copyright Office, Borrower shall: (x) provide Bank with at least seven (7) days prior written notice of Borrower's intent to register such Copyrights or mask works together with a copy of the application it intends to file with the United States Copyright Office (excluding exhibits thereto); (y) execute an intellectual property security agreement and such other documents and take such other actions as Bank may request in its good faith business judgment to perfect and maintain a first priority perfected security interest in favor of Bank in the Copyrights or mask works intended to be registered with the United States Copyright Office; and (z) record such intellectual property security agreement with the United States Copyright Office contemporaneously with filling the Copyright or mask work application(s) with the United States Copyrights or mask works, together with evidence of the recording of the intellectual property security agreement reasonably requested by Bank and required for Bank to perfect and maintain a first priority perfected security interest in such property.
- (c) Provide written notice to Bank within ten (10) days of entering or becoming bound by any Restricted License (other than over-the-counter software that is commercially available to the public). Borrower shall take such commercially reasonable steps as Bank reasonably requests to obtain the consent of, or waiver by, any person whose consent or waiver is necessary for (i) any Restricted License to be deemed "Collateral" and for Bank to have a security interest in it that might otherwise be restricted or prohibited by law or by the terms of any such Restricted License, whether now existing or entered into in the future, and (ii) Bank to have the ability in the event of a liquidation of any Collateral to dispose of such Collateral in accordance with Bank's rights and remedies under this Agreement and the other Loan Documents.
- **6.11 Litigation Cooperation.** From the date hereof and continuing through the termination of this Agreement, make available to Bank (which, prior to the occurrence of an Event of Default, shall be at reasonable times and upon reasonable notice from Bank), without expense to Bank, Borrower and its officers, employees and agents and Borrower's books and records, to the extent that Bank may deem them reasonably necessary to prosecute or defend any third-party suit or proceeding instituted by or against Bank with respect to any Collateral or relating to Borrower.

6.12 Online Banking.

- (a) Utilize Bank's online banking platform for all matters requested by Bank which shall include, without limitation (and without request by Bank for the following matters), uploading information pertaining to Accounts and Account Debtors, requesting approval for exceptions, requesting Credit Extensions, and uploading financial statements and other reports required to be delivered by this Agreement (including, without limitation, those described in Section 6.2 of this Agreement).
- (b) Comply with the terms of the Bank's Online Banking Agreement as in effect from time to time and ensure that all persons utilizing Bank's online banking platform are duly authorized to do so by an Administrator. Bank shall be entitled to assume the authenticity, accuracy and completeness on any information, instruction or request for a Credit Extension submitted via Bank's online banking platform and to further assume that any submissions or requests made via Bank's online banking platform have been duly authorized by an Administrator.
- 6.13 Formation or Acquisition of Subsidiaries. Notwithstanding and without limiting the negative covenants contained in Sections 7.3 and 7.7 hereof, around the time that Borrower forms any direct or indirect Subsidiary or acquires any direct or indirect Subsidiary after the Effective Date, upon Bank's request in its good faith business discretion (including, without limitation, pursuant to a Division), Borrower and such Guarantor shall (a) unless such new Subsidiary is a Foreign Subsidiary, cause such new Subsidiary to provide to Bank a joinder to this Agreement to become a co-borrower hereunder or Secured Guarantor hereunder, together with such appropriate financing statements and/or Control Agreements covering Collateral of such new entity, all in form and substance satisfactory to Bank (including being sufficient to grant Bank a first priority Lien (subject to Permitted Liens) in and to the Collateral of such newly formed or acquired Subsidiary), (b) provide to Bank appropriate certificates and powers and financing statements, pledging all of the direct or beneficial ownership interest in such new Subsidiary, in form and substance satisfactory to Bank (or if such new Subsidiary is a Foreign Subsidiary, not more than sixty-five percent (65%) of the presently existing and hereafter arising issued and outstanding shares of capital stock owned by Borrower of any such Foreign Subsidiary which shares entitle the holder thereof to vote for directors or any other matter); and (c) provide to Bank all other documentation in form and substance reasonably satisfactory to Bank, which in its opinion reasonable and is appropriate with respect to the execution and delivery of the applicable documentation referred to above. Any document, agreement, or instrument executed or issued pursuant to this Section 6.13 shall be a Loan Document.
- **6.14 Further Assurances**. Execute any further instruments and take further action as Bank reasonably requests to perfect or continue Bank's Lien in the Collateral or to effect the purposes of this Agreement.

7. NEGATIVE COVENANTS

Borrower shall not do any of the following without Bank's prior written consent:

- 7.1 Dispositions. Convey, sell, lease, transfer, assign, or otherwise dispose of (including, without limitation, pursuant to a Division) (collectively, "Transfer"), or permit any of its Subsidiaries to Transfer, all or any part of its business or property, except for Transfers (a) of Inventory in the ordinary course of business; (b) of worn-out, surplus, or obsolete Equipment that is, in the reasonable judgment of Borrower, no longer economically practicable to maintain or useful in the ordinary course of business of Borrower; (c) consisting of Permitted Liens and Permitted Investments; (d) consisting of the sale or issuance of any stock of Borrower permitted under Section 7.2 of this Agreement; (e) consisting of Borrower's use or transfer of money or Cash Equivalents in the ordinary course of its business for the payment of ordinary course business epneses in a manner that is not prohibited by the terms of this Agreement or the other Loan Documents; (f) of non-exclusive licenses for the use of the property of Borrower or its Subsidiaries in the ordinary course of business; and (g) of other property with an aggregate value not to exceed Two Hundred Fifty Thousand Dollars (\$250,000) in any twelve (12) month period.
- 7.2 Changes in Business, Management, Control, or Business Locations. (a) Engage in or permit any of its Subsidiaries to engage in any business other than the businesses currently engaged in by Borrower and such Subsidiary, as applicable, or reasonably related thereto; (b) liquidate or dissolve (except that a Subsidiary may liquidate or dissolve provided simultaneous therewith all of its assets are transferred to Borrower); (c) fail to provide notice to Bank of any Key Person departing from or ceasing to be employed by Borrower within five (5) Business Days after his or her departure from Borrower; or (d) permit or suffer any Change in Control.

Borrower shall not, without at least fifteen (15) days prior written notice to Bank: (A) add any new offices or business locations, including warehouses (unless such new offices or business locations contain less than Two Hundred Fifty Thousand Dollars (\$250,000) in Borrower's assets or property) or deliver any portion of the Collateral valued, individually or in the aggregate, in excess of Two Hundred Fifty Thousand Dollars (\$250,000) to a bailee at a location other than to a bailee and at a location already disclosed in the Perfection Certificate, (B) change its jurisdiction of organization, (C) change its organizational structure or type, (D) change its legal name, or (E) change any organizational number (if any) assigned by its jurisdiction of organization. If Borrower intends to add any new offices or business locations, including warehouses, containing in excess of Two Hundred Fifty Thousand Dollars (\$250,000.00) of Borrower's assets or property, then Borrower will first receive the written consent of Bank, and the landlord of any such new offices or business locations, including warehouses, shall execute and deliver a landlord consent in form and substance satisfactory to Bank. If Borrower intends to deliver any portion of the Collateral valued, individually or in the aggregate, in excess of Two Hundred Fifty Thousand Dollars (\$250,000) to a bailee, and Bank and such bailee are not already parties to a bailee agreement governing both the Collateral and the location to which Borrower intends to deliver the Collateral, then Borrower will first receive the written consent of Bank, and such bailee shall execute and deliver a bailee agreement in form and substance satisfactory to Bank.

- 7.3 Mergers or Acquisitions. Merge or consolidate, or permit any of its Subsidiaries to merge or consolidate, with any other Person, or acquire, or permit any of its Subsidiaries to acquire, all or substantially all of the capital stock or property of another Person (including, without limitation, by the formation of any Subsidiary or pursuant to a Division) except if (A) Borrower has complied with the notice requirements applicable to prepayments hereunder, and (B) prior to or contemporaneously with the closing of such transaction, all Obligations are paid in full, and all of Bank's obligations to lend to Borrower are terminated. A Subsidiary may merge or consolidate into another Subsidiary or into Borrower.
 - 7.4 Indebtedness. Create, incur, assume, or be liable for any Indebtedness, or permit any Subsidiary to do so, other than Permitted Indebtedness.
- 7.5 Encumbrance. Create, incur, allow, or suffer any Lien on any of its property, or assign or convey any right to receive income, including the sale of any Accounts, or permit any of its Subsidiaries to do so, except in each case for Permitted Liens, permit any Collateral not to be subject to the first priority security interest granted herein, except for Permitted Liens, or enter into any agreement, document, instrument or other arrangement (except with or in favor of Bank and Permitted Liens) with any Person which directly or indirectly prohibits or has the effect of prohibiting Borrower or any Subsidiary from assigning, mortgaging, pledging, granting a security interest in or upon, or encumbering any of Borrower's or any Subsidiary's Intellectual Property, except as is otherwise permitted in Section 7.1 hereof and the definition of "Permitted Liens" herein and other than (i) the licensors of in-licensed property with respect to such property; and (ii) the lessors of specific leased equipment with respect to such specific leased equipment.
 - 7.6 Maintenance of Collateral Accounts. Maintain any Collateral Account except pursuant to the terms of Section 6.8(b) hereof.
- 7.7 Distributions; Investments. (a) Pay any dividends or make any distribution or payment or redeem, retire or purchase any capital stock or other equity interests; provided that (i) Borrower may convert any of its convertible securities into other securities pursuant to the terms of such convertible securities or otherwise in exchange thereof, (ii) Borrower may pay dividends solely in common stock and repurchase de minimis stock in lieu of issuing fractional shares; (iii) Borrower may repurchase the stock of former employees, directors or consultants (including by the cancellation of indebtedness owed by such former Persons) pursuant to stock repurchase agreements so long as an Event of Default does not exist at the time of such repurchase and would not exist after giving effect to such repurchase, provided that the aggregate amount of all such repurchases does not exceed One Hundred Thousand Dollars (\$100,000) per fiscal year; (iv) Borrower may distribute equity securities to former or current employees, officers, consultants or directors on the exercise of employee issuances pursuant to Borrower's stock incentive plan approved by the Board; and (v) Borrower may repurchase common shares of Borrower's stock from its employees and directors in connection with a bona fide equity financing of Borrower provided that (A) Borrower sells common or preferred

shares to investors for an aggregate price equal to the repurchase price of such common shares repurchased by Borrower, and (B) no Event of Default exists at the time of such series of transactions and would not exist after giving effect to such transactions; or (b) directly or indirectly make any Investment (including, without limitation, by the formation of any Subsidiary) other than Permitted Investments, or permit any of its Subsidiaries to do so.

- 7.8 Transactions with Affiliates. Directly or indirectly enter into or permit to exist any material transaction with any Affiliate of Borrower, except for (i) transactions that are in the ordinary course of Borrower's business, and that are upon fair and reasonable terms that are no less favorable to Borrower than would be obtained in an arm's length transaction with a non-affiliated Person, (ii) sales of Borrower's equity securities to, and Subordinated Debt financing transactions with, Borrower's investors (in each case to the extent not otherwise prohibited under this Agreement), (iii) reasonable and customary compensation arrangements and benefit plans for officers and other employees of Borrower and its Subsidiaries entered into or maintained in the ordinary course of business, (iv) transactions permitted pursuant to the terms of Sections 7.2 and 7.7 hereof, and (v) compensation of outside directors in the ordinary course of business.
- 7.9 Subordinated Debt. (a) Make or permit any payment on any Subordinated Debt, except under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject, or (b) amend any provision in any document relating to the Subordinated Debt which would increase the amount thereof, provide for earlier or greater principal, interest, or other payments thereon, or adversely affect the subordination thereof to Obligations owed to Bank, except under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject.
- 7.10 Compliance. Become an "investment company" or a company controlled by an "investment company", under the Investment Company Act of 1940, as amended, or undertake as one of its important activities extending credit to purchase or carry margin stock (as defined in Regulation U of the Board of Governors of the Federal Reserve System), or use the proceeds of any Credit Extension for that purpose; fail to meet the minimum funding requirements of ERISA, permit a Reportable Event or Prohibited Transaction, as defined in ERISA, to occur; fail to comply with the Federal Fair Labor Standards Act or violate any other law or regulation, if the violation could reasonably be expected to have a material adverse effect on Borrower's business, or permit any of its Subsidiaries to do so; withdraw or permit any Subsidiary to withdraw from participation in, permit partial or complete termination of, or permit the occurrence of any other event with respect to, any present pension, profit sharing and deferred compensation plan which could reasonably be expected to result in any liability of Borrower, including any liability to the Pension Benefit Guaranty Corporation or its successors or any other governmental agency.

8. EVENTS OF DEFAULT

Any one of the following shall constitute an event of default (an "Event of Default") under this Agreement:

8.1 Payment Default. Borrower fails to (a) make any payment of principal or interest on any Credit Extension when due, or (b) pay any other Obligations within three (3) Business Days after such Obligations are due and payable (which three (3) Business Day cure period shall not apply to payments due on the Revolving Line Maturity Date or the Growth Capital Maturity Date). During the cure period, the failure to make or pay any payment specified under clause (b) hereunder is not an Event of Default (but no Credit Extension will be made during the cure period);

8.2 Covenant Default.

(a) Borrower fails or neglects to perform any obligation in Sections 6.2, 6.3, 6.4, 6.6, 6.7, 6.8, 6.9, 6.10(b), 6.12, or violates any covenant in Section 7; provided, however, notwithstanding the foregoing, Borrower's failure to comply with the financial covenant set forth in Section 6.9 shall not in and of itself constitute an Event of Default with respect to the Growth Capital Advances or other Obligations, and shall apply solely with respect to the Advances under the Revolving Line.

- (b) Borrower fails or neglects to perform, keep, or observe any other term, provision, condition, covenant or agreement contained in this Agreement or any Loan Documents, and as to any default (other than those specified in this Section 8) under such other term, provision, condition, covenant or agreement that can be cured, has failed to cure the default within ten (10) days after the occurrence thereof; provided, however, that if the default cannot by its nature be cured within the ten (10) day period or cannot after diligent attempts by Borrower be cured within such ten (10) day period, and such default is likely to be cured within a reasonable time, then Borrower shall have an additional period (which shall not in any case exceed thirty (30) days) to attempt to cure such default, and within such reasonable time period the failure to cure the default shall not be deemed an Event of Default (but no Credit Extensions shall be made during such cure period). Cure periods provided under this section shall not apply, among other things, to financial covenants or any other covenants set forth in clause (a) above;
- **8.3 Investor Abandonment; Lien Priority.** (a) Bank determines, in its good faith judgment, that it is the clear intention of Borrower's investors to not continue to fund the Borrower in the amounts and timeframe necessary to enable Borrower to satisfy its financial obligations as they become due and payable; or (b) there is a material impairment in the priority of Bank's security interest in the Collateral;

8.4 Attachment; Levy; Restraint on Business.

- (a) (i) The service of process seeking to attach, by trustee or similar process, any funds of Borrower or of any entity under the control of Borrower (including a Subsidiary) in excess of Two Hundred Fifty Thousand Dollars (\$250,000), or (ii) a notice of lien or levy is filed against any of Borrower's assets by any Governmental Authority, and the same under subclauses (i) and (ii) hereof are not, within ten (10) days after the occurrence thereof, discharged or stayed (whether through the posting of a bond or otherwise); provided, however, no Credit Extensions shall be made during any ten (10) day cure period; or
- (b) (i) any material portion of Borrower's assets is attached, seized, levied on, or comes into possession of a trustee or receiver, or (ii) any court order enjoins, restrains, or prevents Borrower from conducting all or any material part of its business;
- **8.5 Insolvency.** (a) Borrower or any of its Subsidiaries is unable to pay its debts (including trade debts) as they become due or otherwise becomes insolvent as described in Section 5.6; (b) Borrower or any of its Subsidiaries begins an Insolvency Proceeding; or (c) an Insolvency Proceeding is begun against Borrower or any of its Subsidiaries and is not dismissed or stayed within forty-five (45) days (but no Credit Extensions shall be made while any of the conditions described in clause (a) exist and/or until any Insolvency Proceeding is dismissed);
- **8.6 Other Agreements.** There is, under any agreement to which Borrower is a party with a third party or parties, (a) any default resulting in a right by such third party or parties, whether or not exercised, to accelerate the maturity of any Indebtedness in an amount individually or in the aggregate in excess of Two Hundred Fifty Thousand Dollars (\$250,000); or (b) any breach or default by Borrower, the result of which could reasonably be expected to have a material adverse effect on Borrower's business; provided, however, that the Event of Default under this Section 8.6 caused by the occurrence of a breach or default under such other agreement shall be automatically cured or waived for purposes of this Agreement upon Bank receiving written notice from the party asserting such breach or default of such cure or waiver of the breach or default under such other agreement, if at the time of such cure or waiver under such other agreement (x) Bank has not declared an Event of Default under this Agreement and/or exercised any rights with respect thereto; (y) any such cure or waiver does not result in an Event of Default under any other provision of this Agreement or any Loan Documents; and (z) in connection with any such cure or waiver under such other agreement, the terms of any agreement with such third party are not modified or amended in any manner which could in the good faith business judgment of Bank be materially less advantageous to Borrower;
- **8.7 Judgments; Penalties.** One or more fines, penalties or final judgments, orders or decrees for the payment of money in an amount, individually or in the aggregate, of at least Two Hundred Fifty Thousand Dollars (\$250,000) (not covered by independent third-party insurance as to which liability has been accepted by such insurance carrier) shall be rendered against Borrower by any Governmental Authority, and the same are not, within ten (10) days after the entry, assessment or issuance thereof, discharged, satisfied, or paid, or after execution thereof, stayed or bonded pending appeal, or such judgments are not discharged prior to the expiration of any such stay (provided that no Credit Extensions will be made prior to the satisfaction, payment, discharge, stay, or bonding of such fine, penalty, judgment, order or decree);

- **8.8** Misrepresentations. Borrower or any Person acting for Borrower makes any representation, warranty, or other statement now or later in this Agreement, any Loan Document or in any writing delivered to Bank or to induce Bank to enter this Agreement or any Loan Document, and such representation, warranty, or other statement is incorrect in any material respect when made;
- **8.9 Subordinated Debt.** Any document, instrument, or agreement evidencing any Subordinated Debt shall for any reason be revoked or invalidated or otherwise cease to be in full force and effect (unless permitted by the terms of such agreement), any Person shall be in breach thereof or contest in any manner the validity or enforceability thereof or deny that it has any further liability or obligation thereunder, or the Obligations shall for any reason be subordinated or shall not have the priority contemplated by this Agreement or any applicable subordination or intercreditor agreement;
- **8.10 Cross-Default with TriplePoint Loan Documents.** An event of default (as such term is defined in the TriplePoint Loan Documents) shall occur and be continuing under the TriplePoint Loan Documents and such event of default is not cured within any applicable grace period provided therein, provided however, for the avoidance of doubt, that this Section 8.10 is subject to the cure provisions set forth in Section 8.6 of this Agreement.

9. BANK'S RIGHTS AND REMEDIES

- **9.1 Rights and Remedies.** Upon the occurrence and during the continuance of an Event of Default (and following the expiration of any applicable cure periods), Bank may, without notice or demand, do any or all of the following:
- (a) declare all Obligations immediately due and payable (but if an Event of Default described in Section 8.5 occurs all Obligations are immediately due and payable without any action by Bank);
- (b) stop advancing money or extending credit for Borrower's benefit under this Agreement or under any other agreement between Borrower and Bank:
- (c) demand that Borrower (i) deposit cash with Bank in an amount equal to at least (A) one hundred five percent (105.0%) of the Dollar Equivalent of the aggregate face amount of all Letters of Credit denominated in Dollars remaining undrawn, and (B) one hundred ten percent (110.0%) of the Dollar Equivalent of the aggregate face amount of all Letters of Credit denominated in a Foreign Currency remaining undrawn (plus, in each case, all interest, fees, and costs due or to become due in connection therewith (as estimated by Bank in its good faith business judgment)), to secure all of the Obligations relating to such Letters of Credit, as collateral security for the repayment of any future drawings under such Letters of Credit, and Borrower shall forthwith deposit and pay such amounts, and (ii) pay in advance all letter of credit fees scheduled to be paid or payable over the remaining term of any Letters of Credit;
 - (d) terminate any FX Contracts;
- (e) verify the amount of, demand payment of and performance under, and collect any Accounts and General Intangibles, settle or adjust disputes and claims directly with Account Debtors for amounts on terms and in any order that Bank considers advisable, and notify any Person owing Borrower money of Bank's security interest in such funds. Borrower shall collect all payments in trust for Bank and, if requested by Bank, immediately deliver the payments to Bank in the form received from the Account Debtor, with proper endorsements for deposit;
- (f) make any payments and do any acts it considers necessary or reasonable to protect the Collateral and/or its security interest in the Collateral. Borrower shall assemble the Collateral if Bank requests and make it available as Bank designates. Bank may enter premises where the Collateral is located, take and maintain possession of any part of the Collateral, and pay, purchase, contest, or compromise any Lien which appears to be prior or superior to its security interest and pay all expenses incurred. Borrower grants Bank a license to enter and occupy any of its premises, without charge, to exercise any of Bank's rights or remedies;

- (g) apply to the Obligations any (i) balances and deposits of Borrower it holds, or (ii) amount held by Bank owing to or for the credit or the account of Borrower;
- (h) ship, reclaim, recover, store, finish, maintain, repair, prepare for sale, advertise for sale, and sell the Collateral. Bank is hereby granted a non-exclusive, royalty-free license or other right to use, without charge, Borrower's labels, Patents, Copyrights, mask works, rights of use of any name, trade secrets, trade names, Trademarks, and advertising matter, or any similar property as it pertains to the Collateral, in completing production of, advertising for sale, and selling any Collateral and, in connection with Bank's exercise of its rights under this Section 9.1, Borrower's rights under all licenses and all franchise agreements inure to Bank's benefit;
- (i) place a "hold" on any account maintained with Bank and/or deliver a notice of exclusive control, any entitlement order, or other directions or instructions pursuant to any Control Agreement or similar agreements providing control of any Collateral;
 - (j) demand and receive possession of Borrower's Books; and
- (k) exercise all rights and remedies available to Bank under the Loan Documents or at law or equity, including all remedies provided under the Code (including disposal of the Collateral pursuant to the terms thereof).
- 9.2 Power of Attorney. Borrower hereby irrevocably appoints Bank as its lawfulattorney-in-fact, exercisable following the occurrence and during the continuance of an Event of Default, to: (a) endorse Borrower's name on any checks, payment instruments, or other forms of payment or security; (b) sign Borrower's name on any invoice or bill of lading for any Account or drafts against Account Debtors; (c) demand, collect, sue, and give releases to any Account Debtor for monies due, settle and adjust disputes and claims about the Accounts directly with Account Debtors, and compromise, prosecute, or defend any action, claim, case, or proceeding about any Collateral (including filing a claim or voting a claim in any bankruptcy case in Bank's or Borrower's name, as Bank chooses); (d) make, settle, and adjust all claims under Borrower's insurance policies; (e) pay, contest or settle any Lien, charge, encumbrance, security interest, or other claim in or to the Collateral, or any judgment based thereon, or otherwise take any action to terminate or discharge the same; and (f) transfer the Collateral into the name of Bank or a third party as the Code permits. Borrower hereby appoints Bank as its lawful attorney-in-fact to sign Borrower's name on any documents necessary to perfect or continue the perfection of Bank's security interest in the Collateral regardless of whether an Event of Default has occurred until all Obligations (other than inchoate indemnity obligations) have been satisfied in full and the Loan Documents have been terminated. Bank's foregoing appointment as Borrower's attorney in fact, and all of Bank's rights and powers, coupled with an interest, are irrevocable until all Obligations (other than inchoate indemnity obligations) have been fully repaid and performed and the Loan Documents have been terminated.
- 9.3 Protective Payments. If Borrower fails to obtain the insurance called for by Section 6.7 or fails to pay any premium thereon or fails to pay any other amount which Borrower is obligated to pay under this Agreement or any other Loan Document or which may be required to preserve the Collateral, Bank may obtain such insurance or make such payment, and all amounts so paid by Bank are Bank Expenses and immediately due and payable, bearing interest at the then highest rate applicable to the Obligations, and secured by the Collateral. Bank will make reasonable efforts to provide Borrower with notice of Bank obtaining such insurance at the time it is obtained or within a reasonable time thereafter. No payments by Bank are deemed an agreement to make similar payments in the future or Bank's waiver of any Event of Default.
- 9.4 Application of Payments and Proceeds. If an Event of Default has occurred and is continuing, Bank shall have the right to apply in any order any funds in its possession, whether from Borrower account balances, payments, proceeds realized as the result of any collection of Accounts or other disposition of the Collateral, or otherwise, to the Obligations. Bank shall pay any surplus to Borrower by credit to the Designated Deposit Account or to other Persons legally entitled thereto; Borrower shall remain liable to Bank for any deficiency. If an Event of Default has occurred and is continuing, and Bank, directly or indirectly, enters into a deferred payment or other credit transaction with any purchaser at any sale of Collateral, Bank shall have the option, exercisable at any time, of either reducing the Obligations by the principal amount of the purchase price or deferring the reduction of the Obligations until the actual receipt by Bank of cash therefor.

9.5 Bank's Liability for Collateral. So long as Bank complies with reasonable banking practices regarding the safekeeping of the Collateral in the possession or under the control of Bank, Bank shall not be liable or responsible for: (a) the safekeeping of the Collateral; (b) any loss or damage to the Collateral; (c) any diminution in the value of the Collateral; or (d) any act or default of any carrier, warehouseman, bailee, or other Person. Except as set forth in the preceding sentence, Borrower bears all risk of loss, damage or destruction of the Collateral.

9.6 No Waiver; Remedies Cumulative Bank's failure, at any time or times, to require strict performance by Borrower of any provision of this Agreement or any other Loan Document shall not waive, affect, or diminish any right of Bank thereafter to demand strict performance and compliance herewith or therewith. No waiver hereunder shall be effective unless signed by the party granting the waiver and then is only effective for the specific instance and purpose for which it is given. Bank's rights and remedies under this Agreement and the other Loan Documents are cumulative. Bank has all rights and remedies provided under the Code, by law, or in equity. Bank's exercise of one right or remedy is not an election and shall not preclude Bank from exercising any other remedy under this Agreement or other remedy available at law or in equity, and Bank's waiver of any Event of Default is not a continuing waiver. Bank's delay in exercising any remedy is not a waiver, election, or acquiescence.

9.7 Demand Waiver. Borrower waives demand, notice of default or dishonor, notice of payment and nonpayment, notice of any default, nonpayment at maturity, release, compromise, settlement, extension, or renewal of accounts, documents, instruments, chattel paper, and guarantees held by Bank on which Borrower is liable.

10. NOTICES

All notices, consents, requests, approvals, demands, or other communication by any party to this Agreement or any other Loan Document must be in writing and shall be deemed to have been validly served, given, or delivered: (a) upon the earlier of actual receipt and three (3) Business Days after deposit in the U.S. mail, first class, registered or certified mail return receipt requested, with proper postage prepaid; (b) upon transmission, when sent by electronic mail or facsimile transmission; (c) one (1) Business Day after deposit with a reputable overnight courier with all charges prepaid; or (d) when delivered, if hand-delivered by messenger, all of which shall be addressed to the party to be notified and sent to the address, facsimile number, or email address indicated below. Bank or Borrower may change its mailing or electronic mail address or facsimile number by giving the other party written notice thereof in accordance with the terms of this Section 10.

If to Borrower: Grove Collaborative, Inc.

1301 Sansome Street

San Francisco, California 94111 Attn: Janae De Crescenzo Email: janae@grove.co

If to Bank: Silicon Valley Bank

505 Howard Street, Third Floor San Francisco, California 94105

Attn: Marc Diaz Email: mdiaz@svb.com

11. CHOICE OF LAW, VENUE, JURY TRIAL WAIVER AND JUDICIAL REFERENCE

Except as otherwise expressly provided in any of the Loan Documents, California law governs the Loan Documents without regard to principles of conflicts of law. Borrower and Bank each submit to the exclusive jurisdiction of the State and Federal courts in Santa Clara County, California; provided, however, that nothing in this Agreement shall be deemed to operate to preclude Bank from bringing suit or taking other legal action in any other jurisdiction to realize on the Collateral or any other security for the Obligations, or to enforce a judgment or other court order in favor of Bank. Borrower expressly submits and consents in advance to such jurisdiction in any action or suit commenced in any such court, and Borrower hereby waives any objection that it may have based upon lack of personal jurisdiction, improper venue, or forum non conveniens and hereby consents to the granting of such legal or equitable relief as is deemed appropriate by such court. Borrower hereby waives personal service of the summons,

complaints, and other process issued in such action or suit and agrees that service of such summons, complaints, and other process may be made by registered or certified mail addressed to Borrower at the address set forth in, or subsequently provided by Borrower in accordance with, Section 10 of this Agreement and that service so made shall be deemed completed upon the earlier to occur of Borrower's actual receipt thereof or three (3) days after deposit in the U.S. mails, proper postage prepaid.

TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, BORROWER AND BANK EACH WAIVE THEIR RIGHT TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION ARISING OUT OF OR BASED UPON THIS AGREEMENT, THE LOAN DOCUMENTS OR ANY CONTEMPLATED TRANSACTION, INCLUDING CONTRACT, TORT, BREACH OF DUTY AND ALL OTHER CLAIMS. THIS WAIVER IS A MATERIAL INDUCEMENT FOR BOTH PARTIES TO ENTER INTO THIS AGREEMENT. EACH PARTY HAS REVIEWED THIS WAIVER WITH ITS COUNSEL.

WITHOUT INTENDING IN ANY WAY TO LIMIT THE PARTIES' AGREEMENT TO WAIVE THEIR RESPECTIVE RIGHT TO A TRIAL BY JURY, if the above waiver of the right to a trial by jury is not enforceable, the parties hereto agree that any and all disputes or controversies of any nature between them arising at any time shall be decided by a reference to a private judge, mutually selected by the parties (or, if they cannot agree, by the Presiding Judge of the Santa Clara County, California Superior Court) appointed in accordance with California Code of Civil Procedure Section 638 (or pursuant to comparable provisions of federal law if the dispute falls within the exclusive jurisdiction of the federal courts), sitting without a jury, in Santa Clara County, California; and the parties hereby submit to the jurisdiction of such court. The reference proceedings shall be conducted pursuant to and in accordance with the provisions of California Code of Civil Procedure Sections 638 through 645.1, inclusive. The private judge shall have the power, among others, to grant provisional relief, including without limitation, entering temporary restraining orders, issuing preliminary and permanent injunctions and appointing receivers. All such proceedings shall be closed to the public and confidential and all records relating thereto shall be permanently sealed. If during the course of any dispute, a party desires to seek provisional relief, but a judge has not been appointed at that point pursuant to the judicial reference procedures, then such party may apply to the Santa Clara County, California Superior Court for such relief. The proceeding before the private judge shall be conducted in the same manner as it would be before a court under the rules of evidence applicable to judicial proceedings. The parties shall be entitled to discovery which shall be conducted in the same manner as it would be before a court under the rules of discovery applicable to judicial proceedings. The private judge shall oversee discovery and may enforce all discovery rules and orders applicable to judicial proceedings in the same manner as a trial court judge. The parties agree that the selected or appointed private judge shall have the power to decide all issues in the action or proceeding, whether of fact or of law, and shall report a statement of decision thereon pursuant to California Code of Civil Procedure Section 644(a). Nothing in this paragraph shall limit the right of any party at any time to exercise self-help remedies, foreclose against collateral, or obtain provisional remedies. The private judge shall also determine all issues relating to the applicability, interpretation, and enforceability of this paragraph.

This Section 11 shall survive the termination of this Agreement.

12. GENERAL PROVISIONS

- 12.1 Termination Prior to Maturity Date; Survival. All covenants, representations and warranties made in this Agreement shall continue in full force until this Agreement has terminated pursuant to its terms and all Obligations have been satisfied (other than inchoate indemnity obligations). So long as Borrower has satisfied the Obligations (other than inchoate indemnity obligations, and any other obligations which, by their terms, are to survive the termination of this Agreement, and any Obligations under Bank Services Agreements that are cash collateralized in accordance with Section 4.1 of this Agreement), this Agreement may be terminated by Borrower prior to (a) the Revolving Line Maturity Date, effective three (3) Business Days after written notice of termination is given to Bank and (b) the Growth Capital Maturity Date pursuant to Section 2.3(d). Those obligations that are expressly specified in this Agreement as surviving this Agreement's termination shall continue to survive notwithstanding this Agreement's termination.
- 12.2 Successors and Assigns. This Agreement binds and is for the benefit of the successors and permitted assigns of each party. Borrower may not assign this Agreement or any rights or obligations under it without Bank's prior written consent (which may be granted or withheld in Bank's discretion). Bank has the right, without the consent of or notice to Borrower, to sell, transfer, assign, negotiate, or grant participation in all or any part of, or any interest in, Bank's obligations, rights, and benefits under this Agreement and the other Loan Documents (other than the Warrant, as to which assignment, transfer and other such actions are governed by the terms thereof).

12.3 Indemnification. Borrower agrees to indemnify, defend and hold Bank and its directors, officers, employees, agents, attorneys, or any other Person affiliated with or representing Bank (each, an "Indemnified Person") harmless against: (i) all obligations, demands, claims, and liabilities (collectively, "Claims") claimed or asserted by any other party in connection with the transactions contemplated by the Loan Documents; and (ii) all losses or expenses (including Bank Expenses) in any way suffered, incurred, or paid by such Indemnified Person as a result of, following from, consequential to, or arising from transactions between Bank and Borrower contemplated by the Loan Documents (including reasonable attorneys' fees and expenses), except for Claims and/or losses and/or expenses directly caused by such Indemnified Person's gross negligence or willful misconduct.

This Section 12.3 shall survive until all statutes of limitation with respect to the Claims, losses, and expenses for which indemnity is given shall have run.

- 12.4 Time of Essence. Time is of the essence for the performance of all Obligations in this Agreement.
- 12.5 Severability of Provisions. Each provision of this Agreement is severable from every other provision in determining the enforceability of any provision.
- 12.6 Correction of Loan Documents. Bank may correct patent errors and fill in any blanks in the Loan Documents consistent with the agreement of the parties so long as Bank provides Borrower with written notice of such correction and allows Borrower at least ten (10) days to object to such correction. In the event of such objection, such correction shall not be made except by an amendment signed by both Bank and Borrower.
- 12.7 Amendments in Writing; Waiver; Integration. No purported amendment or modification of any Loan Document, or waiver, discharge or termination of any obligation under any Loan Document, shall be enforceable or admissible unless, and only to the extent, expressly set forth in a writing signed by the party against which enforcement or admission is sought. Without limiting the generality of the foregoing, no oral promise or statement, nor any action, inaction, delay, failure to require performance or course of conduct shall operate as, or evidence, an amendment, supplement or waiver or have any other effect on any Loan Document. Any waiver granted shall be limited to the specific circumstance expressly described in it, and shall not apply to any subsequent or other circumstance, whether similar or dissimilar, or give rise to, or evidence, any obligation or commitment to grant any further waiver. The Loan Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of the Loan Documents merge into the Loan Documents.
- 12.8 Counterparts. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, is an original, and all taken together, constitute one Agreement.
- 12.9 Confidentiality. In handling any confidential information, Bank shall exercise the same degree of care that it exercises for its own proprietary information, but disclosure of information may be made: (a) to Bank's Subsidiaries or Affiliates (such Subsidiaries and Affiliates, together with Bank, collectively, "Bank Entities"); (b) to prospective transferees or purchasers of any interest in the Credit Extensions (provided, however, that any prospective transferee or purchaser shall have entered into an agreement containing provisions substantially the same as those in this Section); (c) as required by law, regulation, subpoena, or other order; (d) to Bank's regulators or as otherwise required in connection with Bank's examination or audit; (e) as Bank considers appropriate in exercising remedies under the Loan Documents; and (f) to third-party service providers of Bank so long as such service providers have executed a confidentiality agreement with Bank with terms no less restrictive than those contained herein. Confidential information does not include information that is either: (i) in the public domain or in Bank's possession when disclosed to Bank (through no fault of Bank), or becomes part of the public domain (other than as a result of its disclosure by Bank in violation of this Agreement) after disclosure to Bank; or (ii) disclosed to Bank by a third party, if Bank does not know that the third party is prohibited from disclosing the information.

Bank Entities may use anonymous forms of confidential information for aggregate datasets, for analyses or reporting, and for any other uses not expressly prohibited in writing by Borrower. The provisions of the immediately preceding sentence shall survive the termination of this Agreement.

- 12.10 Attorneys' Fees, Costs and Expenses In any action or proceeding between Borrower and Bank arising out of or relating to the Loan Documents, the prevailing party shall be entitled to recover its reasonable attorneys' fees and other costs and expenses incurred, in addition to any other relief to which it may be entitled.
- 12.11 Electronic Execution of Documents. The words "execution," "signed," "signature" and words of like import in any Loan Document shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity and enforceability as a manually executed signature or the use of a paper-based recordkeeping systems, as the case may be, to the extent and as provided for in any applicable law, including, without limitation, any state law based on the Uniform Electronic Transactions Act.
- 12.12 Right of Setoff. Following the occurrence and during the continuance of an Event of Default, Borrower hereby grants to Bank a Lien and a right of setoff as security for all Obligations to Bank, whether now existing or hereafter arising upon and against all deposits, credits, collateral and property, now or hereafter in the possession, custody, safekeeping or control of Bank or any entity under the control of Bank (including a subsidiary of Bank) or in transit to any of them. At any time after the occurrence and during the continuance of an Event of Default, without demand or notice, Bank may setoff the same or any part thereof and apply the same to any liability or Obligation of Borrower even though unmatured and regardless of the adequacy of any other collateral securing the Obligations. ANY AND ALL RIGHTS TO REQUIRE BANK TO EXERCISE ITS RIGHTS OR REMEDIES WITH RESPECT TO ANY OTHER COLLATERAL WHICH SECURES THE OBLIGATIONS, PRIOR TO EXERCISING ITS RIGHT OF SETOFF WITH RESPECT TO SUCH DEPOSITS, CREDITS OR OTHER PROPERTY OF BORROWER, ARE HEREBY KNOWINGLY, VOLUNTARILY AND IRREVOCABLY WAIVED.
 - 12.13 Captions. The headings used in this Agreement are for convenience only and shall not affect the interpretation of this Agreement.
- 12.14 Construction of Agreement. The parties mutually acknowledge that they and their attorneys have participated in the preparation and negotiation of this Agreement. In cases of uncertainty this Agreement shall be construed without regard to which of the parties caused the uncertainty to exist.
- 12.15 Relationship. The relationship of the parties to this Agreement is determined solely by the provisions of this Agreement. The parties do not intend to create any agency, partnership, joint venture, trust, fiduciary or other relationship with duties or incidents different from those of parties to an arm's-length contract.
- 12.16 Third Parties. Nothing in this Agreement, whether express or implied, is intended to: (a) confer any benefits, rights or remedies under or by reason of this Agreement on any persons other than the express parties to it and their respective permitted successors and assigns; (b) relieve or discharge the obligation or liability of any person not an express party to this Agreement; or (c) give any person not an express party to this Agreement any right of subrogation or action against any party to this Agreement.
- 12.17 Transitional Arrangements. On the Effective Date, this Agreement shall amend, restate and supersede the Prior Loan Agreement in its entirety, and the Prior Loan Agreement shall be terminated, except as provided in this Section. On the Effective Date, the rights and obligations of the parties evidenced by the Prior Loan Agreement shall be evidenced by this Agreement and the other Loan Documents and the grant of security interest in the Collateral by the Borrower under the Prior Loan Agreement and the other "Loan Documents" (as defined in the Prior Loan Agreement) shall continue without novation under this Agreement and the other Loan Documents, and such security interest and any other rights and obligations which by their express terms survive the termination of the Loan Documents shall not in any event be terminated, extinguished or annulled but shall hereafter be governed by this Agreement and the other Loan Documents. All references to the Prior Loan Agreement in any Loan Document or other document or instrument delivered in connection therewith shall be deemed to refer to this Agreement and the provisions hereof as amended, restated, or otherwise modified from time to time.

13. **DEFINITIONS**

- **13.1 Definitions.** As used in the Loan Documents, the word "shall" is mandatory, the word "may" is permissive, the word "or" is not exclusive, the words "includes" and "including" are not limiting, the singular includes the plural, and numbers denoting amounts that are set off in brackets are negative. As used in this Agreement, the following capitalized terms have the following meanings:
- "Account" is, as to any Person, any "account" of such Person as "account" is defined in the Code with such additions to such term as may hereafter be made, and includes, without limitation, all accounts receivable and other sums owing to such Person.
 - "Account Debtor" is any "account debtor" as defined in the Code with such additions to such term as may hereafter be made.
 - "Administrator" is an individual that is named:
- (b) as an "Administrator" in the "SVB Online Services" form completed by Borrower with the authority to determine who will be authorized to use SVB Online Services (as defined in Bank's Online Banking Agreement as in effect from time to time) on behalf of Borrower; and
 - (c) as an Authorized Signer of Borrower in an approval by the Board.
 - "Advance" or "Advances" means a revolving credit loan (or revolving credit loans) under the Revolving Line.
- "Advance Rate" is sixty percent (60%); provided, however, that Bank has the right to decrease the Advance Rate in its good faith business judgment to mitigate the impact of events, conditions, contingencies, or risks which may adversely affect the Collateral or its value or in the event that the percentage assigned to the net orderly liquidation value of the Eligible Inventory is less than the Advance Rate, provided that prior to an Event of Default, Bank shall endeavor to provide Borrower with notice of any such changes; provided further than any failure to so notify shall not be deemed a breach of this Agreement.
- "Affiliate" is, with respect to any Person, each other Person that owns or controls directly or indirectly the Person, any Person that controls or is controlled by or is under common control with the Person, and each of that Person's senior executive officers, directors, partners and, for any Person that is a limited liability company, that Person's managers and members.
 - "Agreement" is defined in the preamble hereof.
- "Authorized Signer" is any individual listed in Borrower's Borrowing Resolution who is authorized to execute the Loan Documents, including making (and executing if applicable) any Credit Extension request, on behalf of Borrower.
- "Availability Amount" is (a) the lesser of (i) the Revolving Line or (ii) the amount available under the Borrowing Base<u>minus</u> (b) the aggregate Dollar Equivalent amount of all outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit) plus an amount equal to any Letter of Credit Reserve, <u>minus</u> (c) the outstanding principal balance of any Advances.
 - "Bank" is defined in the preamble hereof.
 - "Bank Entities" is defined in Section 12.9.
- "Bank Expenses" are all audit fees and expenses, costs, and expenses (including reasonable attorneys' fees and expenses) for preparing, amending, negotiating, administering, defending and enforcing the Loan Documents (including, without limitation, those incurred in connection with appeals or Insolvency Proceedings) or otherwise incurred with respect to Borrower or any Guarantor.

"Bank Services" are any products, credit services, and/or financial accommodations previously, now, or hereafter provided to Borrower or any of its Subsidiaries by Bank or any Bank Affiliate, including, without limitation, any letters of credit, cash management services (including, without limitation, merchant services, direct deposit of payroll, business credit cards, and check cashing services), interest rate swap arrangements, and foreign exchange services as any such products or services may be identified in Bank's various agreements related thereto (each, a "Bank Services Agreement") and shall include, without limitation, any Letters of Credit issued pursuant to Section 2.2(b).

"Bank Services Agreement" is defined in the definition of Bank Services.

"Beneficial Ownership Information Disclosure Form" means the form attached hereto as Exhibit D.

"Board" is Borrower's board of directors.

"Borrower" is defined in the preamble hereof.

"Borrower's Books" are all Borrower's books and records including ledgers, federal and state tax returns, records regarding Borrower's assets or liabilities, the Collateral, business operations or financial condition, and all computer programs or storage or any equipment containing such information.

"Borrowing Base" is an amount equal to the Advance Rate multiplied by the value of Borrower's Eligible Inventory (valued at the lower of cost or wholesale fair market value) as determined by Bank from Borrower's most recent Borrowing Base Statement.

"Borrowing Base Statement" is that certain report of the value of certain Collateral in the form specified by Bank to Borrower from time to time.

"Borrowing Resolutions" are, with respect to any Person, those resolutions adopted by such Person's board of directors (and, if required under the terms of such Person's Operating Documents, stockholders) and delivered by such Person to Bank approving the Loan Documents to which such Person is a party and the transactions contemplated thereby, together with a certificate executed by its secretary on behalf of such Person certifying (a) such Person has the authority to execute, deliver, and perform its obligations under each of the Loan Documents to which it is a party, (b) that set forth as a part of or attached as an exhibit to such certificate is a true, correct, and complete copy of the resolutions then in full force and effect authorizing and ratifying the execution, delivery, and performance by such Person of the Loan Documents to which it is a party, (c) the name(s) of the Person(s) authorized to execute the Loan Documents, including making (and executing if applicable) any Credit Extension request, on behalf of such Person, together with a sample of the true signature(s) of such Person(s), and (d) that Bank may conclusively rely on such certificate unless and until such Person shall have delivered to Bank a further certificate canceling or amending such prior certificate.

"Business Day" is any day that is not a Saturday, Sunday or a day on which Bank is closed.

"Cash Collateral Account" is defined in Section 6.3(c).

"Cash Equivalents" means (a) marketable direct obligations issued or unconditionally guaranteed by the United States or any agency or any State thereof having maturities of not more than one (1) year from the date of acquisition; (b) commercial paper maturing no more than one (1) year after its creation and having the highest rating from either Standard & Poor's Ratings Group or Moody's Investors Service, Inc.; (c) Bank's certificates of deposit issued maturing no more than one (1) year after issue; and (d) money market funds at least ninety-five percent (95%) of the assets of which constitute Cash Equivalents of the kinds described in clauses (a) through (c) of this definition.

"Change in Control" means (a) at any time, any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act), shall become, or obtain rights (whether by means of warrants, options or otherwise) to become, the "beneficial owner" (as defined in Rules 13(d)-3 and 13(d)-5 under the Exchange Act), directly or indirectly, of forty-nine percent (49%) or more of the ordinary voting power for the election of directors of Borrower (determined on a fully diluted basis) other than by the sale of Borrower's equity securities in a public offering or to venture capital or private equity investors so long as Borrower identifies to Bank the venture capital or private equity investors at least seven (7) Business Days prior to the closing of the transaction and provides to Bank a description of the material terms of the transaction; (b) during any period of twelve (12) consecutive months, a majority of the members of the Board or other equivalent governing body of Borrower cease to be composed of individuals (i) who were members of that board or equivalent governing body on the first day of such period, (ii) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in clause (i) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body or (iii) whose election or nomination to that board or equivalent governing body; or (c) at any time, Borrower shall cease to own and control, of record and beneficially, directly or indirectly, one hundred percent (100.0%) of each class of outstanding capital stock of each subsidiary of Borrower free and clear of all Liens (except Liens created by this Agreement), unless such Subsidiary merges with and into Borrower.

"Claims" is defined in Section 12.3.

"Code" is the Uniform Commercial Code, as the same may, from time to time, be enacted and in effect in the State of California; provided, that, to the extent that the Code is used to define any term herein or in any Loan Document and such term is defined differently in different Articles or Divisions of the Code, the definition of such term contained in Article or Division 9 shall govern; provided further, that in the event that, by reason of mandatory provisions of law, any or all of the attachment, perfection, or priority of, or remedies with respect to, Bank's Lien on any Collateral is governed by the Uniform Commercial Code in effect in a jurisdiction other than the State of California, the term "Code" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes of the provisions thereof relating to such attachment, perfection, priority, or remedies and for purposes of definitions relating to such provisions.

"Collateral" is any and all properties, rights and assets of Borrower described on Exhibit A.

"Collateral Account" is any Deposit Account, Securities Account, or Commodity Account.

"Commodity Account" is any "commodity account" as defined in the Code with such additions to such term as may hereafter be made.

"Compliance Statement" is that certain statement in the form attached hereto as Exhibit B.

"Contingent Obligation" is, for any Person, any direct or indirect liability, contingent or not, of that Person for (a) any indebtedness, lease, dividend, letter of credit or other obligation of another such as an obligation, in each case, directly or indirectly guaranteed, endorsed, co made, discounted or sold with recourse by that Person, or for which that Person is directly or indirectly liable; (b) any obligations for undrawn letters of credit for the account of that Person; and (c) all obligations from any interest rate, currency or commodity swap agreement, interest rate cap or collar agreement, or other agreement or arrangement designated to protect a Person against fluctuation in interest rates, currency exchange rates or commodity prices; but "Contingent Obligation" does not include endorsements in the ordinary course of business. The amount of a Contingent Obligation is the stated or determined amount of the primary obligation for which the Contingent Obligation is made or, if not determinable, the maximum reasonably anticipated liability for it determined by the Person in good faith; but the amount may not exceed the maximum of the obligations under any guarantee or other support arrangement.

"Control Agreement" is any control agreement entered into among the depository institution at which Borrower maintains a Deposit Account or the securities intermediary or commodity intermediary at which Borrower maintains a Securities Account or a Commodity Account, Borrower, and Bank pursuant to which Bank obtains control (within the meaning of the Code) over such Deposit Account, Securities Account, or Commodity Account.

- "Copyrights" are any and all copyright rights, copyright applications, copyright registrations and like protections in each work of authorship and derivative work thereof, whether published or unpublished and whether or not the same also constitutes a trade secret.
- "Credit Extension" is any Advance, any Growth Capital Advance, any Letter of Credit, any Overadvance, or any other extension of credit by Bank for Borrower's benefit under this Agreement.
- "Current Liabilities" are (a) all obligations and liabilities of Borrower to Bank, plus, (b) without duplication, the aggregate amount of Borrower's Total Liabilities that mature within one (1) year, including all of Borrower's liabilities identified as current liabilities in the Monthly Financial Statements.
 - "Default Rate" is defined in Section 2.5(b).
- "Deferred Final Payment" is a payment (in addition to and not a substitution for the regular monthly payments of principal plus accrued interest) due on the earliest to occur of (a) the Growth Capital Maturity Date, or (b) the acceleration of the Growth Capital Advances, or (c) the prepayment of any Growth Capital Advance in full pursuant to Section 2.3(d) or 2.3(e), equal to approximately Forty-Eight Thousand Eighty-Nine and 68/100 Dollars (\$48,089.68) (which is the unaccrued portion of the "Final Payment" due to Bank in connection with the "Growth Capital Advances" (as defined in the Prior Loan Agreement)).
 - "Deposit Account" is any "deposit account" as defined in the Code with such additions to such term as may hereafter be made.
- "Designated Deposit Account" is the account number ending 042 (last three digits) maintained by Borrower with Bank (provided, however, if no such account number is included, then the Designated Deposit Account shall be any deposit account of Borrower maintained with Bank as chosen by Bank).
- "Division" means, in reference to any Person which is an entity, the division of such Person into two (2) or more separate Persons, with the dividing Person either continuing or terminating its existence as part of such division, including, without limitation, as contemplated under Section 18-217 of the Delaware Limited Liability Company Act for limited liability companies formed under Delaware law, or any analogous action taken pursuant to any other applicable law with respect to any corporation, limited liability company, partnership or other entity.
- "Dollar Equivalent" is, at any time, (a) with respect to any amount denominated in Dollars, such amount, and (b) with respect to any amount denominated in a Foreign Currency, the equivalent amount therefor in Dollars as determined by Bank at such time on the basis of the then-prevailing rate of exchange in San Francisco, California, for sales of the Foreign Currency for transfer to the country issuing such Foreign Currency.
- "Dollars," "dollars" or use of the sign "\$" means only lawful money of the United States and not any other currency, regardless of whether that currency uses the "\$" sign to denote its currency or may be readily converted into lawful money of the United States.
- "Domestic Subsidiary" means a Subsidiary organized under the laws of the United States or any state or territory thereof or the District of Columbia.
 - "Effective Date" is defined in the preamble hereof.
- "Eligible Inventory" means Inventory that meets all of Borrower's representations and warranties in Section 5.3 and is otherwise reasonably acceptable to Bank in all material respects.
- "Equipment" is all "equipment" as defined in the Code with such additions to such term as may hereafter be made, and includes without limitation all machinery, fixtures, goods, vehicles (including motor vehicles and trailers), and any interest in any of the foregoing.
 - "ERISA" is the Employee Retirement Income Security Act of 1974, and its regulations.

- "Event of Default" is defined in Section 8.
- "Exchange Act" is the Securities Exchange Act of 1934, as amended.
- "Excluded Taxes" means any of the following taxes imposed on or with respect to Bank or required to be withhold or deducted from a payment to Bank: taxes imposed on or measured by net income (however denominated), franchise taxes, and branch profits taxes, in each case imposed as a result of Bank being organized under the laws of, or having its principal office or its applicable lending office located in, the jurisdiction imposing such tax (or any political subdivision thereof).
- "Final Payment" is a payment (in addition to and not a substitution for the regular monthly payments of principal plus accrued interest) due on the earliest to occur of (a) the Growth Capital Maturity Date, or (b) the acceleration of the Growth Capital Advances, or (c) the prepayment of the Growth Capital Advance in full pursuant to Section 2.3(d) or 2.3(e), equal to the original aggregate principal amount of the Growth Capital Advances multiplied by the Final Payment Percentage.
 - "Final Payment Percentage" is five percent (5.00%).
- "Financial Statement Repository" is each of (a) wd8b0c@svb.com, or such other means of collecting information approved and designated by Bank after providing notice thereof to Borrower from time to time, and (b) Bank's online banking platform as described in Section 6.12.
 - "Foreign Currency" means lawful money of a country other than the United States.
 - "Foreign Subsidiary" means any Subsidiary which is not a Domestic Subsidiary.
 - "Funding Date" is any date on which a Credit Extension is made to or for the account of Borrower which shall be a Business Day.
- "FX Contract" is any foreign exchange contract by and between Borrower and Bank under which Borrower commits to purchase from or sell to Bank a specific amount of Foreign Currency on a specified date.
- "GAAP" is generally accepted accounting principles set forth in the opinions and pronouncements of the Accounting Principles Board of the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or in such other statements by such other Person as may be approved by a significant segment of the accounting profession, which are applicable to the circumstances as of the date of determination.
- "General Intangibles" is all "general intangibles" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation, all Intellectual Property, claims, income and other tax refunds, security and other deposits, payment intangibles, contract rights, options to purchase or sell real or personal property, rights in all litigation presently or hereafter pending (whether in contract, tort or otherwise), insurance policies (including without limitation key man, property damage, and business interruption insurance), payments of insurance and rights to payment of any kind.
- "Governmental Approval" is any consent, authorization, approval, order, license, franchise, permit, certificate, accreditation, registration, filing or notice, of, issued by, from or to, or other act by or in respect of, any Governmental Authority.
- "Governmental Authority" is any nation or government, any state or other political subdivision thereof, any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative functions of or pertaining to government, any securities exchange and any self-regulatory organization.
 - "Growth Capital Advances" is defined in Section 2.3(a).

- "Growth Capital Amortization Date" is November 1, 2020.
- "Growth Capital Maturity Date" is December 1, 2022.
- "Guarantor" is any Person providing a Guaranty in favor of Bank.
- "Guaranty" is any guarantee of all or any part of the Obligations, as the same may from time to time be amended, restated, modified or otherwise supplemented.
- "Indebtedness" is (a) indebtedness for borrowed money or the deferred price of property or services, such as reimbursement and other obligations for surety bonds and letters of credit, (b) obligations evidenced by notes, bonds, debentures or similar instruments, (c) capital lease obligations, and (d) Contingent Obligations.
 - "Indemnified Person" is defined in Section 12.3.
- "Initial Audit" is Bank's inspection of Borrower's Accounts, the Collateral, and Borrower's Books, with results satisfactory to Bank in its good faith business discretion.
- "Insolvency Proceeding" is any proceeding by or against any Person under the United States Bankruptcy Code, or any other bankruptcy or insolvency law, including assignments for the benefit of creditors, compositions, extensions generally with its creditors, or proceedings seeking reorganization, arrangement, or other relief.
 - "Intellectual Property" means, with respect to any Person, all of such Person's right, title, and interest in and to the following:
 - (a) its Copyrights, Trademarks and Patents;
- (b) any and all trade secrets and trade secret rights, including, without limitation, any rights to unpatented inventions,know-how, and operating manuals;
 - (c) any and all source code;
 - (d) any and all design rights which may be available to such Person;
- (e) any and all claims for damages by way of past, present and future infringement of any of the foregoing, with the right, but not the obligation, to sue for and collect such damages for said use or infringement of the Intellectual Property rights identified above; and
 - (f) all amendments, renewals and extensions of any of the Copyrights, Trademarks or Patents.
- "Inventory" is all "inventory" as defined in the Code in effect on the date hereof with such additions to such term as may hereafter be made, and includes without limitation all merchandise, raw materials, parts, supplies, packing and shipping materials, work in process and finished products, including without limitation such inventory as is temporarily out of Borrower's custody or possession or in transit and including any returned goods and any documents of title representing any of the above.
- "Investment" is any beneficial ownership interest in any Person (including stock, partnership interest or other securities), and any loan, advance or capital contribution to any Person.
- "IP Agreement" is that certain Intellectual Property Security Agreement between Borrower and Bank dated as of December 1, 2016, as may be amended, modified or restated from time to time.
 - "Key Person" is Borrower's Chief Financial Officer, who is Servaes Tholen as of the Effective Date.

"Letter of Credit" means a standby letter of credit issued by Bank or another institution based upon an application, guarantee, indemnity or similar agreement on the part of Bank as set forth in Section 2.2(b).

"Letter of Credit Application" is defined in Section 2.2(b)(ii).

"Letter of Credit Reserve" is defined in Section 2.2(b)(v).

"Lien" is a claim, mortgage, deed of trust, levy, charge, pledge, security interest or other encumbrance of any kind, whether voluntarily incurred or arising by operation of law or otherwise against any property.

"Liquidity" is, at any time, the sum of (a) the aggregate amount of unrestricted and unencumbered cash at Bank (including amounts in the Cash Collateral Account), and (b) the Availability Amount.

"Loan Documents" are, collectively, this Agreement and any schedules, exhibits, certificates, notices, and any other documents related to this Agreement, the Warrant, the IP Agreement, the TriplePoint Subordination Agreement, any Bank Services Agreement, any subordination agreement, any note, or notes or guaranties executed by Borrower or any guarantor, and any other present or future agreement by Borrower and/or any guarantor with or for the benefit of Bank in connection with this Agreement, all as amended, restated, or otherwise modified.

"Material Adverse Change" is (a) a material impairment in the perfection or priority of Bank's Lien in the Collateral or in the value of such Collateral; (b) a material adverse change in the business, operations, or financial condition of Borrower; or (c) a material impairment of the prospect of repayment of any portion of the Obligations.

"Merchant Processing Accounts" is defined in Section 6.8.

"Minimum Liquidity Threshold" means Fifty Five Million Dollars (\$55,000,000).

"Modified Current Ratio" means a ratio of (a) Quick Assets plus Inventory (valued at cost) to (b) Current Liabilities minus Deferred Revenue.

"Monthly Financial Statements" is defined in Section 6.2(c).

"Obligations" are Borrower's obligations to pay when due any debts, principal, interest, fees, Bank Expenses, the Prepayment Fee, the Final Payment, and other amounts Borrower owes Bank now or later, whether under this Agreement, the other Loan Documents (other than the Warrant), or otherwise, including, without limitation, all obligations relating to Bank Services and interest accruing after Insolvency Proceedings begin and debts, liabilities, or obligations of Borrower assigned to Bank, and to perform Borrower's duties under the Loan Documents (other than the Warrant).

"Operating Documents" are, for any Person, such Person's formation documents, as certified by the Secretary of State (or equivalent agency) of such Person's jurisdiction of organization on a date that is no earlier than thirty (30) days prior to the Effective Date, and, (a) if such Person is a corporation, its bylaws in current form, (b) if such Person is a limited liability company, its limited liability company agreement (or similar agreement), and (c) if such Person is a partnership, its partnership agreement (or similar agreement), each of the foregoing with all current amendments or modifications thereto

"Overadvance" is defined in Section 2.4.

"Patents" means all patents, patent applications and like protections including without limitation improvements, divisions, continuations, renewals, reissues, extensions and continuations-in-part of the same.

"Payment/Advance Form" is that certain form in the form attached hereto as Exhibit C.

"Payment Date" is (a) with respect to the Growth Capital Advances, the first (1st) calendar day of each month and (b) with respect to Advances, the last calendar day of each month.

"Perfection Certificate" is defined in Section 5.1.

"Permitted Indebtedness" is:

- (a) Borrower's Indebtedness to Bank under this Agreement and the other Loan Documents;
- (b) Indebtedness existing on the Effective Date which is shown on the Perfection Certificate;
- (c) Subordinated Debt;
- (d) unsecured Indebtedness to trade creditors incurred in the ordinary course of business; and
- (e) Indebtedness incurred as a result of endorsing negotiable instruments received in the ordinary course of business;
- (f) Indebtedness secured by Liens permitted under clauses (a) and (c) of the definition of "Permitted Liens" hereunder;
- (g) Borrower's credit cards maintained with third parties in the ordinary course of Borrower's business, provided that the aggregate credit limits available thereunder do not exceed One Million Five Hundred Thousand Dollars (\$1,500,000) at any time;
 - (h) other Indebtedness not otherwise enumerated herein not to exceed One Hundred Thousand Dollars (\$100,000) at any time;
 - (i) the TriplePoint Indebtedness provided the aggregate principal amount outstanding does not exceed Fifty-Five Million Dollars (\$55,000,000); and
- (j) extensions, refinancings, modifications, amendments and restatements of any items of Permitted Indebtedness (a) through (i) above, provided that the principal amount thereof is not increased or the terms thereof are not modified to impose more burdensome terms upon Borrower or its Subsidiary, as the case may be.

"Permitted Investments" are:

- (a) Investments (including, without limitation, Subsidiaries) existing on the Effective Date which are shown on the Perfection Certificate;
- (b) Investments consisting of Cash Equivalents;
- (c) Investments consisting of the endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of Borrower;
- (d) Investments consisting of deposit accounts or securities accounts in which Bank has a perfected security interest except as provided and excepted by Section 6.8(b);
 - (e) Investments accepted in connection with Transfers permitted by Section 7.1;
- (f) Investments consisting of the creation of a Subsidiary for the purpose of consummating a merger transaction permitted by Section 7.3 of this Agreement, which is otherwise a Permitted Investment;
- (g) Investments (i) by Borrower in Subsidiaries not to exceed Two Hundred Fifty Thousand Dollars (\$250,000) in the aggregate in any fiscal year, (ii) by Subsidiaries in other Subsidiaries not to exceed Two Hundred Fifty Thousand Dollars (\$250,000) in the aggregate in any fiscal year or in Borrower, and (iii) by any Borrower in any other co-borrower or guarantor;

- (h) Investments consisting of (i) travel advances and employee relocation loans and other employee loans and advances in the ordinary course of business, and (ii) loans to employees, officers or directors relating to the purchase of equity securities of Borrower or its Subsidiaries pursuant to employee stock purchase plans or agreements approved by the Board;
- (i) Investments (including debt obligations) received in connection with the bankruptcy or reorganization of customers or suppliers and in settlement of delinquent obligations of, and other disputes with, customers or suppliers arising in the ordinary course of business;
- (j) Investments in joint ventures or strategic alliances in the ordinary course of Borrower's business, provided that any cash investments by Borrower do not exceed Two Hundred Fifty Thousand Dollars (\$250,000) in the aggregate in any fiscal year;
- (k) joint ventures or strategic alliances in the ordinary course of Borrower's business, provided that any cash investments by Borrower do not exceed Five Hundred Thousand Dollars (\$500,000) in the aggregate in any fiscal year; and
- (1) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions, to customers and suppliers who are not Affiliates, in the ordinary course of business; provided that this paragraph (k) shall not apply to Investments of Borrower in any Subsidiary.

"Permitted Liens" are:

- (a) Liens existing on the Effective Date which are shown on the Perfection Certificate or arising under this Agreement or the other Loan Documents;
- (b) Liens for taxes, fees, assessments or other government charges or levies, either (i) not due and payable or (ii) being contested in good faith and for which Borrower maintains adequate reserves on Borrower's Books, <u>provided</u> that no notice of any such Lien has been filed or recorded under the Internal Revenue Code of 1986, as amended, and the Treasury Regulations adopted thereunder;
- (c) purchase money Liens (i) on Equipment and/or operating leases acquired or held by Borrower incurred for financing the acquisition or operating leases of the Equipment securing no more than Five Million Dollars (\$5,000,000) in the aggregate amount outstanding, or (ii) existing on Equipment when acquired, if the Lien is confined to the property and improvements and the proceeds of the Equipment;
- (d) Liens of carriers, warehousemen, suppliers, or other Persons that are possessory in nature arising in the ordinary course of business so long as such Liens attach only to Inventory, securing liabilities in the aggregate amount not to exceed Two Hundred Fifty Thousand Dollars (\$250,000) and which are not delinquent or remain payable without penalty or which are being contested in good faith and by appropriate proceedings which proceedings have the effect of preventing the forfeiture or sale of the property subject thereto;
- (e) Liens to secure payment of workers' compensation, employment insurance, old-age pensions, social security and other like obligations incurred in the ordinary course of business (other than Liens imposed by ERISA);
- (f) Liens incurred in the extension, renewal or refinancing of the Indebtedness secured by Liens described in (a) through (c) but any extension, renewal or replacement Lien must be limited to the property encumbered by the existing Lien and the principal amount of the indebtedness may not increase: and
- (g) leases or subleases of real property granted in the ordinary course of Borrower's business (or, if referring to another Person, in the ordinary course of such Person's business), and leases, subleases, non-exclusive licenses or sublicenses of personal property (other than Intellectual Property) granted in the ordinary course of Borrower's business (or, if referring to another Person, in the ordinary course of such Person's business), if the leases, subleases, licenses and sublicenses do not prohibit granting Bank a security interest therein;

- (h) non-exclusive licenses of Intellectual Property granted to third parties in the ordinary course of business and licenses of Intellectual Property that could not result in a legal transfer of title of the licensed property that may be exclusive in respects other than territory and that may be exclusive as to territory only as to discrete geographical areas outside of the United States; and
 - (i) Liens arising from attachments or judgments, orders, or decrees in circumstances not constituting an Event of Default under Sections 8.4 and 8.7.
- (j) Liens in favor of TriplePoint securing the TriplePoint Indebtedness permitted under clause (i) of the definition of "Permitted Indebtedness" and subject to the TriplePoint Subordination Agreement.
- "Person" is any individual, sole proprietorship, partnership, limited liability company, joint venture, company, trust, unincorporated organization, association, corporation, institution, public benefit corporation, firm, joint stock company, estate, entity or government agency.
- "Prepayment Fee" shall be an additional fee, payable to Bank, with respect to each Growth Capital Advance, in an amount equal to one and one half of one percent (1.50%) of the aggregate original principal amount of the Growth Capital Advance being prepaid if the prepayment is made on or before July 1, 2021, provided, however, that the Prepayment Fee shall be waived by Bank in the event Borrower refinances any Growth Capital Advance with Bank.
- "Prime Rate" is the rate of interest per annum from time to time published in the money rates section of The Wall Street Journal or any successor publication thereto as the "prime rate" then in effect; provided that, in the event such rate of interest is less than zero, such rate shall be deemed to be zero for purposes of this Agreement; and provided further that if such rate of interest, as set forth from time to time in the money rates section of The Wall Street Journal, becomes unavailable for any reason as determined by Bank, the "Prime Rate" shall mean the rate of interest per annum announced by Bank as its prime rate in effect at its principal office in the State of California (such Bank announced Prime Rate not being intended to be the lowest rate of interest charged by Bank in connection with extensions of credit to debtors); provided that, in the event such rate of interest is less than zero, such rate shall be deemed to be zero for purposes of this Agreement.
- "Purchase Date" is, with respect to specific Inventory, the date on which a purchase order for such Inventory is confirmed received and deliverable to Borrower by the supplier of Borrower.
 - "Quick Assets" is, on any date, Borrower's unrestricted and unencumbered cash with Bank plus net accounts receivable.
 - "Registered Organization" is any "registered organization" as defined in the Code with such additions to such term as may hereafter be made.
- "Requirement of Law" is as to any Person, the organizational or governing documents of such Person, and any law (statutory or common), treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.
- "Reserves" means, as of any date of determination, such amounts as Bank may from time to time establish and revise in its good faith business judgment, reducing the amount of Advances and other financial accommodations which would otherwise be available to Borrower (a) to reflect events, conditions, contingencies or risks which, as determined by Bank in its good faith business judgment, do or may adversely affect (i) the Collateral or any other property which is security for the Obligations or its value (including without limitation any increase in delinquencies of Accounts), (ii) the assets, business or prospects of Borrower or any Guarantor, or (iii) the security interests and other rights of Bank in the Collateral (including the enforceability, perfection and priority thereof); or (b) to reflect Bank's reasonable belief that any collateral report or financial information furnished by or on behalf of Borrower or any Guarantor to Bank is or may have been incomplete, inaccurate or misleading in any material respect; or (c) in respect of any state of facts which Bank determines constitutes an Event of Default or may, with notice or passage of time or both, constitute an Event of Default.

"Responsible Officer" is any of the Chief Executive Officer, President, Chief Financial Officer, VP Business Operations and Controller of Borrower.

"Restricted License" is any material license or other material agreement with respect to which Borrower is the licensee (a) that prohibits or otherwise restricts Borrower from granting a security interest in Borrower's interest in such license or agreement or any other property, or (b) for which a default under or termination of could materially interfere with Bank's right to sell any Collateral.

"Revolving Line" is an aggregate principal amount equal to Twenty-Five Million Dollars (\$25,000,000).

"Revolving Line Maturity Date" is March 31, 2022.

"SEC" shall mean the Securities and Exchange Commission, any successor thereto, and any analogous Governmental Authority.

"Securities Account" is any "securities account" as defined in the Code with such additions to such term as may hereafter be made.

"Secured Guarantor" is any Guarantor who has (a) executed and delivered to Bank a security agreement in form and substance reasonably satisfactory to Bank pursuant to which such Guarantor has granted Bank a first priority perfected Lien in the types of assets substantially similar to the Collateral to secure the Obligations, free and clear of all Liens other than Permitted Liens; (b) delivered to Bank such appropriate Control Agreements in form and substance reasonably satisfactory to Bank if and to the extent required under Section 6.8; and (c) provided to Bank all other documentation in form and substance satisfactory to Bank in its reasonable discretion which in its opinion is appropriate with respect to the execution and delivery of the applicable documentation referred to above and which Bank has reasonably requested.

"Streamline Period" is, provided no Event of Default has occurred and is continuing, the period (a) commencing on the first day of the Subject Month following the day that either Borrower's (i) Modified Current Ratio is equal to or greater than 1.15 to 1.00, or (ii) Liquidity is equal to or greater than the Minimum Liquidity Threshold (collectively, the "Streamline Threshold"); and (b) terminating on the earlier to occur of (i) the occurrence of an Event of Default, and (ii) the first day thereafter in which Borrower fails to maintain the Streamline Threshold, as determined by Bank in its good faith business discretion. Upon the termination of a Streamline Period, Borrower must maintain the Streamline Threshold each consecutive day for one (1) calendar month as determined by Bank in its good faith business discretion, prior to entering into a subsequent Streamline Period. Borrower shall automatically enter into any such eligible Streamline Period, and each such Streamline Period shall commence on the first day of the monthly period following the date Bank determines, in its reasonable good faith business discretion, that the Streamline Threshold has been achieved.

"Streamline Threshold" is defined in the definition of Streamline Period.

"Subject Month" is the month which is two (2) calendar months after any Testing Month. For example, when testing Borrower's Modified Quick Ratio for May 2020, the Testing Month would be May 2020 and the Subject Month would be July 2020, and determination of whether a Streamline Period is in effect for July 2020 would be based on Borrower's Modified Quick Ratio as of the last day of May 2020.

"Subordinated Debt" is indebtedness incurred by Borrower subordinated to all of Borrower's now or hereafter indebtedness to Bank (pursuant to a subordination, intercreditor, or other similar agreement in form and substance satisfactory to Bank entered into between Bank and the other creditor), on terms acceptable to Bank.

"Subsidiary" is, as to any Person, a corporation, partnership, limited liability company or other entity of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person. Unless the context otherwise requires, each reference to a Subsidiary herein shall be a reference to a Subsidiary of Borrower or Guarantor.

"Testing Month" is any month with respect to which Bank has tested Borrower's Modified Quick Ratio to determine whether a Streamline Period is in effect.

"Total Liabilities" is on any day, obligations that should, under GAAP, be classified as liabilities on Borrower's consolidated balance sheet, including all Indebtedness.

"Trademarks" means any trademark and servicemark rights, whether registered or not, applications to register and registrations of the same and like protections, and the entire goodwill of the business of Borrower connected with and symbolized by such trademarks.

"TriplePoint" means TRIPLEPOINT VENTURE GROWTH BDC CORP., a Maryland corporation.

"TriplePoint Indebtedness" is Indebtedness in the principal amount not to exceed Fifty Five Million Dollars (\$55,000,000) under the TriplePoint Loan Documents.

"TriplePoint Loan Documents" means that certain Plain English Growth Capital Loan and Security Agreement dated as of April 2, 2018 between Borrower and TriplePoint, and any other agreement, document, promissory note, financing statement, or instrument executed by Borrower in favor of TriplePoint pursuant to or in connection with the TriplePoint Indebtedness, as the same may from time to time be amended, modified, supplemented, extended, renewed, restated or replaced.

"TriplePoint Subordination Agreement" is that certain Subordination Agreement by and between Bank and TriplePoint dated as of April 2, 2018, and that certain Amended and Restated Subordination Agreement by and between Bank and TriplePoint dated as of May 22, 2019, each as amended, affirmed and/or restated from time to time.

"Transfer" is defined in Section 7.1.

"Unused Revolving Line Facility Fee" is defined in Section 2.6(i).

"Warrant" is, individually and collectively, (a) the Warrant to Purchase Stock dated as of December 1, 2016 executed by Borrower in favor of Bank, and (b) the Warrant to Purchase Stock dated as of July 10, 2017 executed by Borrower in favor of Bank, each as the same may be amended, restated, or otherwise modified from time to time.

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the Effective Date.

BORROWER:

GROVE COLLABORATIVE, INC.

By: /s/ Servaes Tholen

Name: Servaes Tholen Title: Chief Financial Officer

BANK:

SILICON VALLEY BANK

By: /s/ Marc Diaz

Name: Marc Diaz Title: Vice President

[Signature Page to Second Amended and Restated Loan and Security Agreement]

FIRST AMENDMENT TO SECOND AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT

This First Amendment to Second Amended and Restated Loan and Security Agreement (this "Amendment") is entered into this 30th day of April, 2021, by and between Silicon Valley Bank ("Bank") and **GROVE COLLABORATIVE**, **INC.**, a Delaware public benefit corporation ("Borrower").

RECITALS

- A. Bank and Borrower have entered into that certain Second Amended and Restated Loan and Security Agreement dated as of July 29, 2020 (as the same may from time to time be further amended, modified, supplemented or restated, the "Loan Agreement").
 - B. Bank has extended credit to Borrower for the purposes permitted in the Loan Agreement.
 - C. Borrower has requested that Bank amend the Loan Agreement to make certain revisions to the Loan Agreement as more fully set forth herein.
- **D.** Bank has agreed to so amend certain provisions of the Loan Agreement, but only to the extent, in accordance with the terms, subject to the conditions and in reliance upon the representations and warranties set forth below.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing recitals and other good and valuable consideration, the receipt and adequacy of which is hereby acknowledged, and intending to be legally bound, the parties hereto agree as follows:

- 1. Definitions. Capitalized terms used but not defined in this Amendment shall have the meanings given to them in the Loan Agreement.
- 2. Amendments to Loan Agreement.
 - 2.1 Section 2.2 (Revolving Line). Subsection (b)(i) of Section 2.2 is deleted in its entirety and replaced with the following:
 - "(i) As part of the Revolving Line, Bank shall issue or has issued Letters of Credit denominated in Dollars or a Foreign Currency for Borrower's account. The aggregate Dollar Equivalent amount utilized for the issuance of Letters of Credit shall at all times reduce the amount otherwise available for Advances under the Revolving Line. The aggregate Dollar Equivalent of the face amount of outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit and any Letter of Credit Reserve) may not exceed (A) Fifteen Million Dollars (\$15,000,000) minus (i) the sum of all amounts used for Cash Management Services, and minus (ii) the FX Reduction Amount or (B) the lesser of the Revolving Line or the Borrowing Base, minus (i) the sum of all outstanding principal amounts of any Advances (including any amounts used for Cash Management Services) and minus (ii) the FX Reduction Amount."

2.2 Section 2.2 (Revolving Line). New Subsections (d) and (e) are inserted following Section 2.2(c) as follows:

"(d) Foreign Exchange Sublimit. As part of the Revolving Line, Borrower may enter into foreign exchange contracts with Bank under which Borrower commits to purchase from or sell to Bank a specific amount of Foreign Currency (each, a "FX Contract") on a specified date (the "Settlement Date"). FX Contracts shall have a Settlement Date of at least one (1) FX Business Day after the contract date. The aggregate FX Reduction Amount at any one time may not exceed the lesser of (A) Fifteen Million Dollars (\$15,000,000), minus (i) the sum of all amounts used for Cash Management Services, and minus (ii) the aggregate Dollar Equivalent of the face amount of any outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit and any Letter of Credit Reserve) or (B) the lesser of the Revolving Line or the Borrowing Base, minus (ii) the sum of all outstanding principal amounts of any Advances (including any amounts used for Cash Management Services), and minus (ii) the aggregate Dollar Equivalent of the face amount of any outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit and any Letter of Credit Reserve). The amount otherwise available for Credit Extensions under the Revolving Line shall be reduced by the FX Reduction Amount. Any amounts needed to fully reimburse Bank for any amounts not paid by Borrower in connection with FX Contracts will be treated as Advances under the Revolving Line and will accrue interest at the interest rate applicable to Advances.

(e) Cash Management Services Sublimit. Borrower may use the Revolving Line in an aggregate amount not to exceed the lesser of (A) Fifteen Million Dollars (\$15,000,000), minus (i) the aggregate Dollar Equivalent of the face amount of any outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit and any Letter of Credit Reserve), and minus (ii) the FX Reduction Amount, or (B) the lesser of the Revolving Line or the Borrowing Base, minus (i) the sum of all outstanding principal amounts of any Advances, (ii) minus the aggregate Dollar Equivalent of the face amount of any outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit and any Letter of Credit Reserve), and minus (iii) the FX Reduction Amount for Bank's cash management services, which may include merchant services, direct deposit of payroll, business credit card, and check cashing services identified in Bank's various cash management services agreements (collectively, the "Cash Management Services"). Any amounts Bank pays on behalf of Borrower for any Cash Management Services will be treated as Advances under the Revolving Line and will accrue interest at the interest rate applicable to Advances."

- 2.3 Section 2.4 (Overadvance). Section 2.4 is deleted in its entirety and replaced with the following:
- "2.4 Overadvance. If, at any time, the sum of (a) the outstanding principal amount of any Advances (including any amounts used for Cash Management Services), plus (b) the face amount of any outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit and any Letter of Credit Reserve), plus (c) the FX Reduction Amount, exceeds the lesser of either the Revolving Line or the Borrowing Base, Borrower shall immediately pay to Bank in cash the amount of such excess (such excess, the "Overadvance"). Without limiting Borrower's obligation to repay Bank any Overadvance, in the event that Borrower does not repay any Overadvances within one (1) Business Day, Borrower agrees to pay Bank interest on the outstanding amount of such Overadvance, on demand, at a per annum rate equal to the rate that is otherwise applicable to Advances plus five percent (5.0%)."
- **2.4 Section 2.5 (Payment of Interest on the Credit Extensions**). Subsection (a)(i) of Section 2.5 is deleted in its entirety and replaced with the following:
 - "(a) Interest Rate.
 - (i) <u>Advances</u>. Subject to Section 2.5(b), the outstanding principal amount of each Advance shall accrue interest at a floating per annum rate equal to the greater of (i) the Prime Rate, and (ii) three and one-quarter of one percent (3.25%). Interest shall be payable monthly in accordance with Section 2.5(d) below."
 - 2.5 Section 2.6 (Fees). Subsections (a) and (b) of Section 2.6 are deleted in their entirety and replaced with the following:
 - (a) <u>Revolving Line Commitment Fee</u>. A fully earned, non-refundable commitment fee in connection with the Revolving Line in the amount of Twenty Thousand Dollars (\$20,000) on the First Amendment Effective Date; which, for the avoidance of doubt, is in addition to any commitment fee paid prior to the First Amendment Effective Date;
 - (b) <u>Termination Fee</u>. Upon termination of this Agreement or the termination of the Revolving Line, in each case, for any reason prior to the Revolving Line Maturity Date, in addition to the payment of any other amounts then-owing, a termination fee in an amount equal to one and one half of one percent (1.50%) of then applicable Revolving Line, provided that no termination fee shall be charged if the Revolving Line is replaced or refinanced with a new facility from Bank;"
- **2.6 Section 3.4 (Procedures for Borrowing)**. The first sentence of Subsection (a) of Section 3.4 is deleted in its entirety and replaced with the following:

"Subject to the prior satisfaction of all other applicable conditions to the making of an Advance set forth in this Agreement, to obtain an Advance (other than Advances under Sections 2.2(b) or 2.2(e)), Borrower (via an individual duly authorized by an Administrator) shall notify Bank (which notice shall be irrevocable) by electronic mail by 12:00 p.m. Pacific time on the Funding Date of the Advance."

2.7 Section 5.2 (Collateral). The second paragraph of Section 5.2 is deleted in its entirety and replaced with the following:

"No Collateral valued in excess of Five Hundred Thousand Dollars (\$500,000) (other than laptops and other portable electronic items) is in the possession of any third party bailee (such as a warehouse) except as otherwise provided in the Perfection Certificate or as permitted pursuant to Section 7.2. None of the components of the Collateral valued in excess of Five Hundred Thousand Dollars (\$500,000) (other than laptops and other portable electronic items) shall be maintained at locations other than as provided in the Perfection Certificate or as permitted pursuant to Section 7.2."

2.8 Section 5.3 (Inventory). Section 5.3 is deleted in its entirety and replaced with the following:

"5.3 Inventory.

For any item of Inventory consisting of Eligible Inventory in any Borrowing Base Statement, such Inventory (i) consists of finished goods, in good, new, and salable condition and packaging or shipping materials, which is not perishable, returned, consigned, obsolete, not sellable, damaged, or defective, and is not comprised of demonstrative or custom inventory (provided, however, Borrower's private label brand inventory shall not constitute custom inventory), works in progress, or supplies; (ii) meets all applicable governmental standards; (iii) has been manufactured in compliance with the Fair Labor Standards Act; (iv) is not subject to any Liens, except the first priority Liens granted or in favor of Bank under this Agreement or any of the other Loan Documents and Permitted Liens; (v) is located in the United States at the locations identified by Borrower in the Perfection Certificate where it maintains Inventory (or at any location permitted under Section 7.2, as updated from time to time in accordance with this Agreement), including, without limitation, Borrower's warehouse premises located in Reno, Nevada, St. Peters, Missouri, and Harrisburg, Pennsylvania; (vi) is aged less than one hundred eighty (180) days from Borrower's receipt of such Inventory; and (vii) is located at locations with respect to which Bank has received a fully-executed satisfactory landlord's consent or bailee's waiver, as applicable."

2.9 Section 5.4 (Litigation). Section 5.4 is deleted in its entirety and replaced with the following:

"5.4 Litigation. Except as set forth in the Perfection Certificate dated on or around the Effective Date, there are no actions or proceedings pending or, to the knowledge of any Responsible Officer, threatened in writing by or against Borrower or any of its Subsidiaries that could reasonably be expected to involve more than, individually or in the aggregate, One Million Five Hundred Thousand Dollars (\$1,500,000.00)."

2.10 Section 6.2 (Financial Statements, Reports). Subsection (i) of Section 6.2 is deleted in its entirety and replaced with the following:

"prompt report of any legal actions pending or threatened in writing against Borrower or any of its Subsidiaries that could reasonably be expected to result in damages or costs to Borrower or any of its Subsidiaries of, individually or in the aggregate, One Million Five Hundred Thousand Dollars (\$1,500,000.00) or more (except as set forth in the Perfection Certificate dated on or around the Effective Date);"

- 2.11 Section 6.7 (Insurance). Subsection (b) of Section 6.7 is deleted in its entirety and replaced with the following:
- "(b) Ensure that proceeds payable under any property policy are, at Bank's option, payable to Bank on account of the Obligations. Notwithstanding the foregoing, (a) so long as no Event of Default has occurred and is continuing, Borrower shall have the option of applying the proceeds of any casualty policy up to Two Million Dollars (\$2,000,000) in the aggregate per fiscal year for all losses under all casualty policies in any one (1) year, toward the replacement or repair of destroyed or damaged property; provided that any such replaced or repaired property (i) shall be of equal or like value as the replaced or repaired Collateral and (ii) shall be deemed Collateral in which Bank has been granted a first priority security interest (subject to Permitted Liens which may be superior by operation of law), and (b) after the occurrence and during the continuance of an Event of Default, all proceeds payable under such casualty policy shall, at the option of Bank, be payable to Bank on account of the Obligations."
- 2.12 Section 6.8 (Accounts). Subsections (a) of Section 6.8 are deleted in their entirety and replaced with the following:
- "(a) Borrower, and any Subsidiary of Borrower shall maintain all of its operating accounts and excess cash with Bank or Bank's Affiliates; provided, however, that Borrower may maintain its accounts with merchant processors (including, without limitation, PayPal, Stripe, and other similar processors) ("Merchant Processing Accounts") so long as Borrower transfers all the funds which are in excess of One Hundred Thousand Dollars (\$100,000) in any such account or Three Hundred Thousand Dollars (\$300,000) in the aggregate in an all such accounts to the Cash Collateral Account within five (5) Business Days of any such account exceeding One Hundred Thousand Dollars (\$100,000) individually or such accounts exceeding Three Hundred Thousand Dollars (\$300,000) in the aggregate."
- **2.13 Section 6.9 (Financial Covenants).** Section 6.9 is deleted in its entirety and replaced with the following:

"6.9 Financial Covenants. Maintain the following financial covenant, subject to periodic reporting, on a consolidated basis with respect to Borrower and its Subsidiaries:

(a) Minimum Liquidity. Borrower shall maintain (i) at all times prior to the occurrence of the Performance Milestone, to be tested on the last day of each month, Liquidity of not less than Forty-Five Million Dollars (\$45,000,000) and (ii) upon the occurrence of the Performance Milestone and at all times thereafter, to be tested on the last day of each month, Liquidity of not less than Fifty Million Dollars (\$50,000,000).

Notwithstanding the foregoing, upon the request of Borrower to implement a new financial covenant structure, Bank agrees to evaluate such request in its good faith business discretion."

- 2.14 Section 7.1 (Dispositions). Clause (g) of Section 7.1 is deleted in its entirety and replaced with the following:
- "(g) of other property with an aggregate value not to exceed Five Hundred Thousand Dollars (\$500,000) in any twelve (12) month period."
- **2.15 Section 7.2 (Changes in Business, Management, Control, or Business Locations)**. The second paragraph of Section 7.2 is deleted in its entirety and replaced with the following:

"Borrower shall not, without at least fifteen (15) days prior written notice to Bank: (A) add any new offices or business locations, including warehouses (unless such new offices or business locations contain less than Five Hundred Thousand Dollars (\$500,000) in Borrower's assets or property) or deliver any portion of the Collateral valued, individually or in the aggregate, in excess of Five Hundred Thousand Dollars (\$500,000) to a bailee at a location other than to a bailee and at a location already disclosed in the Perfection Certificate, (B) change its jurisdiction of organization, (C) change its organizational structure or type, (D) change its legal name, or (E) change any organizational number (if any) assigned by its jurisdiction of organization. If Borrower intends to add any new offices or business locations, including warehouses, containing in excess of One Million Dollars (\$1,000,000) of Borrower's assets or property, then Borrower will first receive the written consent of Bank, and the landlord of any such new offices or business locations, including warehouses, shall execute and deliver a landlord consent in form and substance satisfactory to Bank. If Borrower intends to deliver any portion of the Collateral valued, individually or in the aggregate, in excess of One Million Dollars (\$1,000,000) to a bailee, and Bank and such bailee are not already parties to a bailee agreement governing both the Collateral and the location to which Borrower intends to deliver the Collateral, then Borrower will first receive the written consent of Bank, and such bailee shall execute and deliver a bailee agreement in form and substance satisfactory to Bank."

- 2.16 Section 7.7 (Distributions; Investments). Clause (a)(iii) of Section 7.7 is deleted in its entirety and replaced with the following:
- "(iii) Borrower may repurchase the stock of former employees, directors or consultants (including by the cancellation of indebtedness owed by such former Persons) pursuant to stock repurchase agreements so long as an Event of Default does not exist at the time of such repurchase and would not exist after giving effect to such repurchase, provided that the aggregate amount of all such repurchases does not exceed Two Hundred Fifty Thousand Dollars (\$250,000) per fiscal year;"
- 2.17 Section 8.3 (Investor Abandonment; Lien Priority). Section 8.3 is deleted in its entirety and replaced with the following:
- "8.3 Material Adverse Change. A Material Adverse Change occurs;"
- 2.18 Section 8.4 (Attachment; Levy; Restraint on Business). Clause (a)(i) of Section 8.4 is deleted in its entirety and replaced with the

following:

- "(i) The service of process seeking to attach, by trustee or similar process, any funds of Borrower or of any entity under the control of Borrower (including a Subsidiary) in excess of Five Hundred Thousand Dollars (\$500,000), or"
- 2.19 Section 8.6 (Other Agreements). Clause (a) of Section 8.6 is deleted in its entirety and replaced with the following:
- "(a) any default resulting in a right by such third party or parties, whether or not exercised, to accelerate the maturity of any Indebtedness in an amount individually or in the aggregate in excess of Seven Hundred Fifty Thousand Dollars (\$750,000); or"
- 2.20 Section 8.7 (Judgments; Penalties). Section 8.7 is deleted in its entirety and replaced with the following:
- "8.7 Judgments; Penalties. One or more fines, penalties or final judgments, orders or decrees for the payment of money in an amount, individually or in the aggregate, of at least Seven Hundred Fifty Thousand Dollars (\$750,000) (not covered by independent third-party insurance as to which liability has been accepted by such insurance carrier) shall be rendered against Borrower by any Governmental Authority, and the same are not, within ten (10) days after the entry, assessment or issuance thereof, discharged, satisfied, or paid, or after execution thereof, stayed or bonded pending appeal, or such judgments are not discharged prior to the expiration of any such stay (provided that no Credit Extensions will be made prior to the satisfaction, payment, discharge, stay, or bonding of such fine, penalty, judgment, order or decree);"
- 2.21 Section 8 (Events of Default). Section 8.10 is deleted in its entirety and replaced with the following:

- "8.10 Mezzanine Loan Agreement. The occurrence of an Event of Default (as defined in the Mezzanine Loan Agreement) under the Mezzanine Loan Agreement."
- **2.22 Section 13 (Definitions)**. The following terms and their respective definitions set forth in Section 13.1 are deleted in their entirety and replaced with the following:
 - ""Advance Rate" is sixty-five percent (65%); provided, however, that Bank has the right to decrease the Advance Rate in its good faith business judgment to mitigate the impact of events, conditions, contingencies, or risks which may adversely affect the Collateral or its value or in the event that the percentage assigned to the net orderly liquidation value of the Eligible Inventory is less than the Advance Rate, provided that prior to an Event of Default, Bank shall endeavor to provide Borrower with notice of any such changes; provided further than any failure to so notify shall not be deemed a breach of this Agreement. Net orderly liquidation value shall be determined by a third party inventory appraiser, as chosen by Bank, and shall be reset after each inventory appraisal."
 - ""Availability Amount" is (a) the lesser of (i) the Revolving Line or (ii) the amount available under the Borrowing Base minus (b) the aggregate Dollar Equivalent amount of all outstanding Letters of Credit (including drawn but unreimbursed Letters of Credit) plus an amount equal to the Letter of Credit Reserve, minus (c) the FX Reduction Amount, minus (d) any amounts used for Cash Management Services, and minus (e) the outstanding principal balance of any Advances."
 - ""Bank Services" are any products, credit services, and/or financial accommodations previously, now, or hereafter provided to Borrower or any of its Subsidiaries by Bank or any Bank Affiliate, including, without limitation, any letters of credit, cash management services (including, without limitation, merchant services, direct deposit of payroll, business credit cards, and check cashing services), interest rate swap arrangements, and foreign exchange services as any such products or services may be identified in Bank's various agreements related thereto (each, a "Bank Services Agreement") and shall include, without limitation, any Letters of Credit issued pursuant to Section 2.2(b), FX Contracts pursuant to Section 2.2(d), and Cash Management Services pursuant to Section 2.2(e)."
 - ""Credit Extension" is any Advance, any Growth Capital Advance, any Letter of Credit, FX Contract, amount utilized for Cash Management Services, any Overadvance, or any other extension of credit by Bank for Borrower's benefit under this Agreement."
 - ""Current Liabilities" are (a) all obligations and liabilities of Borrower to Bank, plus, (b) without duplication, the aggregate amount of Borrower's Total Liabilities that mature within one (1) year, including all of Borrower's liabilities identified as current liabilities in the Monthly Financial Statements, in each case excluding (i) Borrower's obligations for any principal, interest, fees, expenses and other amounts arising under the Mezzanine Loan Agreement and (ii) ASC 842 lease liabilities and obligations."

- ""Financial Statement Repository" is C4344e@svb.com or such other means of collecting information approved and designated by Bank and each Lender after providing notice thereof to Borrower from time to time."
- ""FX Contract" is defined in Section 2.2(d)."
- ""Loan Documents" are, collectively, this Agreement and any schedules, exhibits, certificates, notices, and any other documents related to this Agreement, the Warrant, the IP Agreement, the Subordination Agreement, any Bank Services Agreement, any subordination agreement, any note, or notes or guaranties executed by Borrower or any guarantor, and any other present or future agreement by Borrower and/or any guarantor with or for the benefit of Bank in connection with this Agreement, all as amended, restated, or otherwise modified."
- ""Revolving Line" is (i) from the First Amendment Effective Date until the occurrence of the Performance Milestone, an aggregate principal amount equal to Twenty-Five Million Dollars (\$25,000,000) and (ii) upon the occurrence of the Performance Milestone and thereafter, an aggregate principal amount equal to Thirty Million Dollars (\$30,000,000)."
- ""Revolving Line Maturity Date" is March 31, 2023."
- 2.23 Section 13 (Definitions). The following new defined terms are hereby inserted alphabetically in Section 13.1:
- ""Cash Management Services" is defined in Section 2.2(e)."
- ""Contribution Profit" is, as determined by Bank in its reasonable discretion, Borrower's net revenue minus product costs minus variable order costs (including, without limitation, all costs incurred in operating and staffing Borrower's fulfillment centers, including costs attributable to receiving, inspecting and warehousing inventories, picking, packaging and preparing customer orders for shipment, shipping and handling expenses, packaging materials costs and payment processing and related transaction costs, but excluding, for the avoidance of doubt, facility and building operating costs and expenses and general overhead costs and expenses)."
- ""First Amendment Effective Date" is April 30, 2021."
- ""FX Reduction Amount" means, with respect to a given FX Contract, the notional amount thereof multiplied by the currency exchange risk factor for the currencies involved in the FX Contract, multiplied by the current foreign exchange spot rates, in each instance as determined and calculated by Bank in its commercially reasonable discretion."

- ""Liquidity" is, at any time, the sum of (a) the aggregate amount of unrestricted and unencumbered cash and Cash Equivalents held at such time by Borrower in Deposit Accounts or Securities Accounts maintained with Bank and/or its Affiliates (including amounts in the Cash Collateral Account), and (b) the Availability Amount."
- ""Mezzanine Loan Agreement" is that certain Mezzanine Loan and Security Agreement by and among Bank, as a lender and in its capacity as administrative agent and collateral agent, HERCULES CAPITAL, INC., a Maryland corporation, as a lender, and Borrower dated as of the First Amendment Effective Date, as the same may be amended, restated, supplemented or otherwise modified from time to time."
- ""Performance Milestone" means delivery by Borrower to Bank, on or prior to April 30, 2022, (i) of evidence satisfactory to Bank on Bank's sole and absolute discretion, that Borrower has achieved Contribution Profit of at least One Hundred Ten Million Dollars (\$110,000,000.00) on a trailing twelve-month basis as of a month ending on or prior to February 28, 2022 and (ii) Borrower's irrevocable notice of intention to increase the Revolving Line in accordance with this Agreement."
- ""Settlement Date" is defined in Section 2.2(d)."
- ""Subordination Agreement" is that certain Intercreditor and Subordination Agreement by and among SVB (as defined in the Mezzanine Loan Agreement), and the Lenders (as defined in the Mezzanine Loan Agreement) dated as of the First Amendment Effective Date (as from time to time be amended, modified, supplemented and/or restated)."
- **2.24 Section 13 (Definitions)**. The following defined terms set forth in Section 13.1 are deleted in their entirety:
- ""Purchase Date" is, with respect to specific Inventory, the date on which a purchase order for such Inventory is confirmed received and deliverable to Borrower by the supplier of Borrower."
- ""TriplePoint" means TRIPLEPOINT VENTURE GROWTH BDC CORP., a Maryland corporation."
- ""TriplePoint Indebtedness" is Indebtedness in the principal amount not to exceed Fifty Five Million Dollars (\$55,000,000) under the TriplePoint Loan Documents."
- ""TriplePoint Loan Documents" means that certain Plain English Growth Capital Loan and Security Agreement dated as of April 2, 2018 between Borrower and TriplePoint, and any other agreement, document, promissory note, financing statement, or instrument executed by Borrower in favor of TriplePoint pursuant to or in connection with the TriplePoint Indebtedness, as the same may from time to time be amended, modified, supplemented, extended, renewed, restated or replaced."

- ""TriplePoint Subordination Agreement" is that certain Subordination Agreement by and between Bank and TriplePoint dated as of April 2, 2018, and that certain Amended and Restated Subordination Agreement by and between Bank and TriplePoint dated as of May 22, 2019, each as amended, affirmed and/or restated from time to time."
- 2.25 Section 13 (Definitions). Clauses (g) through (i) of the definition of "Permitted Indebtedness" are deleted in their entirety and replaced with the following:
 - "(g) Borrower's credit cards maintained with third parties in the ordinary course of Borrower's business, provided that the aggregate credit limits available thereunder do not exceed One Million Five Hundred Thousand Dollars (\$1,500,000) at any time;
 - "(h) other Indebtedness not otherwise enumerated herein not to exceed Three Hundred Thousand Dollars (\$300,000) at any time;"
 - (i) Borrower's Indebtedness under the Mezzanine Loan Agreement and the other Loan Documents (as defined in the Mezzanine Loan Agreement); and"
- 2.26 Section 13 (Definitions). Clauses (j) and (k) of the definition of "Permitted Investments" is deleted in its entirety and replaced with the following:
 - "(j) Investments in joint ventures or strategic alliances in the ordinary course of Borrower's business, provided that any cash investments by Borrower do not exceed Five Hundred Thousand Dollars (\$500,000) in the aggregate in any fiscal year;
 - (k) [reserved]; and"
 - 2.27 Section 13 (Definitions). Clause (j) of the definition of "Permitted Liens" is deleted in its entirety and replaced with the following:
 - "(j) Liens arising under the Mezzanine Loan Agreement and the other Loan Documents (as defined in the Mezzanine Loan Agreement)."
- 2.28 Exhibit B (Compliance Certificate). The Compliance Certificate appearing as <u>Exhibit B</u> to the Loan Agreement is deleted in its entirety and replaced with the Compliance Certificate attached as <u>Schedule 1</u> attached hereto.

3. Limitation of Amendments.

- **3.1** The amendments set forth in Section 2, above, are effective for the purposes set forth herein and shall be limited precisely as written and shall not be deemed to (a) be a consent to any amendment, waiver or modification of any other term or condition of any Loan Document, or (b) otherwise prejudice any right or remedy which Bank may now have or may have in the future under or in connection with any Loan Document.
- 3.2 This Amendment shall be construed in connection with and as part of the Loan Documents and all terms, conditions, representations, warranties, covenants and agreements set forth in the Loan Documents, except as herein amended, are hereby ratified and confirmed and shall remain in full force and effect.
 - 4. Representations and Warranties. To induce Bank to enter into this Amendment, Borrower hereby represents and warrants to Bank as follows:
- **4.1** Immediately after giving effect to this Amendment (a) the representations and warranties contained in the Loan Documents are true, accurate and complete in all material respects as of the date hereof (except to the extent such representations and warranties relate to an earlier date, in which case they are true and correct as of such date), and (b) no Event of Default has occurred and is continuing;
- **4.2** Borrower has the power and authority to execute and deliver this Amendment and to perform its obligations under the Loan Agreement, as amended by this Amendment;
- **4.3** The organizational documents of Borrower delivered to Bank on or around the First Amendment Effective Date remain true, accurate and complete and have not been amended, supplemented or restated and are and continue to be in full force and effect;
- **4.4** The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, have been duly authorized;
- **4.5** The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, do not and will not contravene (a) any law or regulation binding on or affecting Borrower, (b) any contractual restriction with a Person binding on Borrower, (c) any order, judgment or decree of any court or other governmental or public body or authority, or subdivision thereof, binding on Borrower, or (d) the organizational documents of Borrower, or applicable consents or waivers have been obtained;
- 4.6 The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, do not require any order, consent, approval, license, authorization or validation of, or filing, recording or registration with, or exemption by any governmental or public body or authority, or subdivision thereof, binding on Borrower, except as already has been obtained or made; and
- **4.7** This Amendment has been duly executed and delivered by Borrower and is the binding obligation of Borrower, enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, liquidation, moratorium or other similar laws of general application and equitable principles relating to or affecting creditors' rights.

- 5. Ratification of Intellectual Property Security Agreement. Borrower hereby ratifies, confirms and reaffirms, all and singular, the terms and conditions of a certain Intellectual Property Security Agreement dated as of December 1, 2016 between Borrower and Bank, as the same may have been amended, modified, supplement or restated from time to time, including, without limitation, by that certain First Supplement to Intellectual Property Security Agreement dated as of the date of this Amendment (the "First Supplement to IPSA"). and acknowledges, confirms and agrees that said Intellectual Property Security Agreement (a) contains an accurate and complete listing of all Intellectual Property Collateral (as defined therein) and (b) shall remain in full force and effect.
- **6. Ratification of Perfection Certificate**. Borrower hereby ratifies, confirms and reaffirms, all and singular, the terms and disclosures contained in a certain Perfection Certificate dated as of First Amendment Effective Date, and acknowledges, confirms and agrees that the disclosures and information Borrower provided to Bank in such Perfection Certificate have not changed, as of the date hereof.
- **7. Integration**. This Amendment and the Loan Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of this Amendment and the Loan Documents merge into this Amendment and the Loan Documents.
- 8. Counterparts. This Amendment may be executed in any number of counterparts and all of such counterparts taken together shall be deemed to constitute one and the same instrument. Each party hereto may execute this Amendment by electronic means and recognizes and accepts the use of electronic signatures and records by any other party hereto in connection with the execution and storage hereof.
- **9.** Effectiveness. As a condition precedent to the effectiveness of this Amendment, Bank shall have received the following documents prior to or concurrently with this Amendment, each in form and substance reasonably satisfactory to Bank:
 - 9.1 This Amendment and the First Supplement to IPSA duly executed on behalf of Borrower;
- **9.2** A secretary's certificate of Borrower with respect to Borrower's Operating Documents, incumbency, specimen signatures and resolutions authorizing the execution and delivery of this Amendment and the other Loan Documents to which it is a party;
 - 9.3 Duly executed signatures to the completed Borrowing Resolutions for Borrower;
 - 9.4 The duly executed Mezzanine Loan Agreement and satisfaction of all conditions precedent to the effectiveness thereof therein;

- 9.5 The duly executed Subordination Agreement;
- **9.6** Long-form good standing certificates of Borrower certified by the Secretary of State of Delaware and each other jurisdiction in which Borrower is qualified to conduct business, each as of a date no earlier than thirty (30) days prior to the date hereof;
- 9.7 Certified copies, dated as of a recent date, of financing statement and other lien searches of Borrower, which shall be obtained by Bank, accompanied by written evidence (including any Uniform Commercial Code termination statements) that the Liens revealed in any such searched either (i) will be terminated prior to or in connection with this Amendment, or (ii) will constitute Permitted Liens;
 - **9.8** duly executed signature to a payoff letter from TRIPLEPOINT VENTURE GROWTH BDC CORP.;
- 9.9 evidence that (i) the Liens securing Indebtedness owed by Borrower to TRIPLEPOINT VENTURE GROWTH BDC CORP. will be terminated and (ii) the documents and/or filings evidencing the perfection of such Liens, including without limitation any financing statements and/or control agreements, have or will, concurrently with the initial Credit Extension, be terminated;
 - 9.10 payment in full of the Growth Capital Advances and accrued interest and fees in connection therewith;
- **9.11** Borrower's payment of (i) the Deferred Final Payment in the amount of Forty-Eight Thousand Eighty-Nine and 68/100 Dollars (\$48,089.68); (ii) the Final Payment of One Hundred Ninety-Five Thousand Dollars (\$195,000); (iii) the commitment fee due and payable in connection with this Agreement in the amount of Twenty Thousand Dollars (\$20,000.00); and (iv) Bank's legal fees and expenses incurred in connection with this Amendment (subject to the aggregate limit on such legal fees and expenses set forth in the Mezzanine Loan Agreement); and
 - 9.12 such other documents as Bank may reasonably request to effectuate the terms of this Amendment.
- 10. Post-Closing Requirements. Within thirty (30) days after the date hereof (or in the case of clause (iii) below within sixty (60) days of the Effective Date), Borrower shall deliver or cause to be delivered to Bank, in form and substance satisfactory to Bank, (i) evidence that the insurance policies and endorsements required pursuant to Section 6.7 of the Loan Agreement are in full force and effect, (ii) written evidence (including a filed UCC termination statement) that the Liens of Venture Lending & Leasing VIII, Inc. on the assets of Darby Smart, Inc. acquired by Borrower have been terminated and released, (iii) duly executed landlord's consents in favor of Bank for each of Borrower's leased locations containing in excess of One Million Dollars (\$1,000,000) of Borrower's assets or property, by the respective landlord thereof, (iv) a duly executed termination letter for Borrower's securities account with SVB Asset Management identified as Account Number 19-SV1474 containing the wet ink "pdf' signature of TRIPLEPOINT VENTURE GROWTH BDC CORP. Failure to comply with the foregoing requirements within the time period noted shall constitute an Event of Default for which no grace or cure period shall apply, and (v) a duly executed Control Agreement for Borrower's securities account with SVB Asset Management identified as Account Number 19-SV1474.

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed and delivered as of the date first written above.

BANK BORROWER

SILICON VALLEY BANK GROVE COLLABORATIVE, INC.

 By:
 /s/ Jeff Sereno
 By:
 /s/ Servaes Tholen

 Name:
 Jeff Sereno
 Name:
 Servaes Tholen

 Title:
 Vice President
 Title:
 Chief Financial Officer

Schedule 1

EXHIBIT B

COMPLIANCE CERTIFICATE

TO:	SILICON VALLEY BANK	Date:	
FROM:	GROVE COLLABORATIVE, INC.		

Under the terms and conditions of the Second Amended and Restated Loan and Security Agreement between GROVE COLLABORATIVE, INC., a Delaware public benefit corporation ("Borrower"), and Bank (the "Agreement"), I certify on behalf of Borrower that Borrower is in complete compliance for the period ending with all required covenants except as noted below. Attached are the required documents evidencing such compliance, setting forth calculations prepared in accordance with GAAP consistently applied from one period to the next except as explained in an accompanying letter or footnote and except, in each case applicable hereunder, with respect to unaudited financial statements, for the absence of footnotes and subject to year-end adjustments. Capitalized terms used but not otherwise defined herein shall have the meanings given them in the Agreement.

Please indicate compliance status by circling Yes/No under "Complies" column.

Reporting Covenants	Required	Com	plies	
Monthly financial statements (consolidated and consolidating) with Compliance Statement; Key Performance Indicators	Monthly within 30 days	Yes	No	
Annual financial statement (CPA Audited)	FYE within 210 days	Yes	No	
10-Q, 10-K and 8-K	Within 5 days after filing with SEC	Yes	No	
A/R & A/P Agings and Inventory Reports	(a) monthly within 30 days during Streamline Period, and (b) no later than Friday of each week at all other times	Yes	No	
Borrowing Base Statements	(a) monthly within 30 days during Streamline Period, and (b) no later than Friday of each week at all other times	Yes	No	
Board-Approved Projections	Within the earlier to (i) occur 45 days of Borrower's FYE and (ii) 45 days after approval by the Board, and within 7 days of Board approval of any amendment/updates	Yes	Yes No	
Financial Covenant	Required	Actual	Complies	
Maintain as indicated:				
Minimum Liquidity (monthly)	(i) at all times prior to the occurrence of the Performance Milestone, \$45,000,000 and (ii) upon the occurrence of the Performance Milestone and at all times thereafter, to be tested \$50,000,000		Yes No	

The following financial covenant analyses and information set forth in Schedule 1 attached hereto are true and accurate as of the date of this Certificate.

The following are the exceptions with respect to the certification above: (If no exceptions exist, state "No exceptions to note.")

Schedule 1 to Compliance Certificate

Financial Covenants of Borrower and Streamline Period

In the event of a conflict between this Schedule and the Agreement, the terms of the Agreement shall govern. Dated: I. Liquidity (Section 6.9(a)) Required: Borrower shall maintain (i) at all times prior to the occurrence of the Performance Milestone, to be tested on the last day of each month, Liquidity of not less than Forty-Five Million Dollars (\$45,000,000) and (ii) upon the occurrence of the Performance Milestone and at all times thereafter, to be tested on the last day of each month, Liquidity of not less than Fifty Million Dollars (\$50,000,000). Actual: Aggregate amount of unrestricted and unencumbered cash and Cash Equivalents held at such time by Borrower in Deposit Accounts \$_____ A. or Securities Accounts maintained with Bank and/or its Affiliates (including amounts in the Cash Collateral Account) В. Availability Amount C. Liquidity (line A plus line B) Is line C equal to or greater than \$____? No, not in compliance Yes, in compliance

I. Streamline Period

(a) Modified Current Ratio \geq 1.15 to 1.00; or (b) Liquidity \geq \$55,000,000

Actual Modified Current Ratio:

A.	Aggregate amount of unrestricted and unencumbered cash and Cash Equivalents held at such time by Borrower in Deposit Accounts or Securities Accounts maintained with Bank and/or its Affiliates (including amounts in the Cash Collateral Account)	\$
B.	Aggregate amount of Borrower's net accounts receivable	\$
C.	Quick Assets (line A plus line B)	\$
D.	Aggregate value of Inventory (valued at the lower of cost or wholesale fair market value)	\$
E.	Aggregate amount of all obligations and liabilities of Borrower to Bank (excluding Borrower's obligations arising under the Mezzanine Loan Agreement)	\$
F.	Without duplication with line E, aggregate amount of Borrower's obligations that should, under GAAP, be classified as liabilities on Borrower's consolidated balance sheet, including all Indebtedness, that mature within 1 year, including all of Borrower's liabilities identified as current liabilities in the Monthly Financial Statements (excluding Borrower's obligations arising under the Mezzanine Loan Agreement and excluding ASC 842 lease liabilities and obligations)	\$
G.	Current Liabilities (line E plus line F)	\$
H.	Aggregate value of all amounts received or invoiced by Borrower in advance of performance under contracts and not yet recognized as revenue	\$
I.	Modified Current Ratio (line C plus line D, divided by line G plus line H)	:1.00
	Actual Liquidity:	
A.	Aggregate amount of unrestricted and unencumbered cash and Cash Equivalents held at such time by Borrower in Deposit Accounts or Securities Accounts maintained with Bank and/or its Affiliates (including amounts in the Cash Collateral Account)	\$
B.	Availability Amount	\$
C.	Liquidity (line A plus line B)	\$

SECOND AMENDMENT TO SECOND AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT

This Second Amendment to Second Amended and Restated Loan and Security Agreement (this "Amendment") is entered into this 9th day of May, 2022, by and between SILICON VALLEY BANK, a California corporation ("Bank") and GROVE COLLABORATIVE, INC., a Delaware public benefit corporation ("Borrower").

RECITALS

- A. Bank and Borrower have entered into that certain Second Amended and Restated Loan and Security Agreement dated as of July 29, 2020, as amended by that certain First Amendment to Second Amended and Restated Loan and Security Agreement dated as of April 30, 2021 (as the same may from time to time be further amended, modified, supplemented or restated, the "Loan Agreement").
 - **B.** Bank has extended credit to Borrower for the purposes permitted in the Loan Agreement.
- C. Borrower has entered into that certain Amended and Restated Agreement and Plan of Merger dated as of March 31, 2022 (the "Merger Agreement"), by and among (a) Borrower, (b) Virgin Group Acquisition Corp. II, a Cayman Islands exempted company ("Virgin Cayman"), (c) Treehouse Merger Sub, Inc., a Delaware corporation ("Merger Sub I"), and (d) Treehouse Merger Sub II, LLC, a Delaware limited liability company ("Merger Sub II"), which provides for, among other things, (i) the domestication of Virgin Cayman as a Delaware public benefit corporation ("Newco"), (ii) the merger of Merger Sub I with and into Borrower (the "Initial Merger"), with Borrower continuing as the surviving corporation (the "Initial Surviving Corporation") and (iii) immediately following the Initial Merger, the merger of the Initial Surviving Corporation with and into Merger Sub II (the "Final Merger"), with Merger Sub II continuing as the surviving limited liability company, following the consummation of which, Merger Sub II will become a direct, whollyowned subsidiary of Newco (steps (i), (ii) and (iii), collectively with the other transactions described in the Merger Agreement, the "Business Combination Transaction").
- **D.** Pursuant to the terms of the Merger Agreement, Virgin Cayman will cause to be filed with the U.S. Securities and Exchange Commission (the "SEC"), an effective Registration Statement on Form S-4 (including the related proxy statement/prospectus) for the registration under the Securities Act of 1933, as amended, of Newco securities to be issued in connection with the Business Combination Transaction.
 - E. Borrower has requested that Bank amend the Loan Agreement to make certain revisions to the Loan Agreement as more fully set forth herein.
- **F.** Bank has agreed to so amend certain provisions of the Loan Agreement, but only to the extent, in accordance with the terms, subject to the conditions and in reliance upon the representations and warranties set forth below.

AGREEMENT

- **NOW**, **THEREFORE**, in consideration of the foregoing recitals and other good and valuable consideration, the receipt and adequacy of which is hereby acknowledged, and intending to be legally bound, the parties hereto agree as follows:
 - 1. Definitions. Capitalized terms used but not defined in this Amendment shall have the meanings given to them in the Loan Agreement.
 - 2. Amendments to Loan Agreement.
- **2.1 Section 6.2 (Financial Statements, Reports).** Clause (h) of Section 6.2 of the Loan Agreement is deleted in its entirety and replaced with the following:
 - "(h) prior to the consummation of the Business Combination Transaction, within five (5) Business Days of delivery by Borrower, copies of all material statements, reports and notices made generally available by Borrower to Borrower's security holders or to any holders of Subordinated Debt;"
- **2.2 Section 7.2 (Changes in Business, Management, Control or Business Locations).** Clause (c) of Section 7.2 of the Loan Agreement is deleted in its entirety and replaced with the following:
 - "(c) [reserved];"
- **2.3 Section 7.2 (Changes in Business, Management, Control, or Business Locations).** The second paragraph of Section 7.2 is deleted in its entirety and replaced with the following:

"Borrower shall not, without at least fifteen (15) days prior written notice to Bank: (A) add any new offices or business locations, including warehouses (unless such new offices or business locations contain less than Five Hundred Thousand Dollars (\$500,000) in Borrower's assets or property) or deliver any portion of the Collateral valued, individually or in the aggregate, in excess of Five Hundred Thousand Dollars (\$500,000) to a bailee at a location other than to a bailee and at a location already disclosed in the Perfection Certificate, (B) change its jurisdiction of organization, (C) change its organizational structure or type, (D) change its legal name, or (E) change any organizational number (if any) assigned by its jurisdiction of organization; provided that Borrower may, and need not deliver any prior written notice to Bank in order to, change its name, its organizational structure or type and its organizational number assigned by its jurisdiction of organization in connection with the Business Combination Transaction. If Borrower intends to add any new offices or business locations, including warehouses, containing in excess of One Million Dollars (\$1,000,000) of Borrower's assets or property, then Borrower will first receive the written consent of Bank, and the landlord of any such new offices or business locations, including warehouses, shall execute and deliver a landlord consent in form and substance satisfactory to Bank. If Borrower intends to deliver

any portion of the Collateral valued, individually or in the aggregate, in excess of One Million Dollars (\$1,000,000) to a bailee, and Bank and such bailee are not already parties to a bailee agreement governing both the Collateral and the location to which Borrower intends to deliver the Collateral, then Borrower will first receive the written consent of Bank, and such bailee shall execute and deliver a bailee agreement in form and substance satisfactory to Bank."

2.4 Section 7.3 (Mergers or Acquisitions). Section 7.3 of the Loan Agreement is deleted in its entirety and replaced with the following:

"7.3 Mergers or Acquisitions. Merge or consolidate, or permit any of its Subsidiaries to merge or consolidate, with any other Person, or acquire, or permit any of its Subsidiaries to acquire, all or substantially all of the capital stock or property of another Person (including, without limitation, by the formation of any Subsidiary or pursuant to a Division), other than (i) a merger, consolidation or acquisition in which (A) Borrower has complied with the notice requirements applicable to prepayments hereunder, and (B) prior to or contemporaneously with the closing of such transaction, all Obligations are paid in full, and all of Bank's obligations to lend to Borrower are terminated, (ii) a Subsidiary may merge or consolidate into another Subsidiary or into Borrower and (iii) the Business Combination Transaction, so long as (A) the Business Combination Transaction is consummated on or prior to July 31, 2022 in accordance with the Merger Agreement without any amendments or modifications thereto or waivers to any conditions or terms set forth therein that, in each case, could reasonably be expected to be materially adverse to Bank and (B) promptly and substantially concurrent with the consummation of the Business Combination Transaction, (a) Merger Sub II shall provide to Bank a reaffirmation of its assumption of the obligations of the "Borrower" under this Agreement and the other Loan Documents by reason of the Business Combination Transaction and Holdings shall provide Bank a guaranty to become a Guarantor hereunder, together with documentation sufficient to grant Bank a first priority Lien (subject to Permitted Liens) in and to the assets of Merger Sub II and Holdings, all in form and substance reasonably satisfactory to Bank, (b) Holdings shall provide to Bank appropriate stock certificates and powers (if applicable) and financing statements, pledging all of the direct or beneficial ownership interest in Merger Sub II, in form and substance reasonably satisfactory to Bank; and (c) Merger Sub II and Holdings shall provide to Bank certificates duly executed by a Responsible Officer or secretary of each of Merger Sub II and Holdings with respect to their respective Operating Documents and Borrowing Resolutions, documentation and other information required to comply with applicable "know your customer" and anti-money-laundering rules and regulations, and long-form good standing certificates of Merger Sub II and Holdings certified by the Secretary of State of their states of incorporation or formation, in each case as of a date no earlier than 30 days prior to the date of the joinder of guaranty, as applicable."

2.5 Section 7.7 (Distributions; Investments). Section 7.7 of the Loan Agreement is deleted in its entirety and replaced with the following:

"7.7 Distributions; Investments. (a) Pay any dividends or make any distribution or payment or redeem, retire or purchase any capital stock or other equity interests; provided that (i) Borrower may convert any of its convertible securities into other securities pursuant to the terms of such convertible securities or otherwise in exchange thereof, (ii) Borrower may pay dividends solely in common stock and repurchase de minimis stock in lieu of issuing fractional shares; (iii) Borrower may repurchase, or, following the Business Combination Transaction, may pay dividends or make distributions to Holdings to enable it to repurchase, the stock of former employees, directors or consultants (including by the cancellation of indebtedness owed by such former Persons) pursuant to stock repurchase agreements so long as an Event of Default does not exist at the time of such repurchase and would not exist after giving effect to such repurchase, provided that the aggregate amount of all such repurchases does not exceed Two Hundred Fifty Thousand Dollars (\$250,000) per fiscal year; (iv) Borrower may distribute equity securities to former or current employees, officers, consultants or directors on the exercise of employee issuances pursuant to Borrower's stock incentive plan approved by the Board; (v) Borrower may repurchase common shares of Borrower's stock from its employees and directors in connection with a bona fide equity financing of Borrower provided that (A) Borrower sells common or preferred shares to investors for an aggregate price equal to the repurchase price of such common shares repurchased by Borrower, and (B) no Event of Default exists at the time of such series of transactions and would not exist after giving effect to such transactions; and (vi) following the Business Combination Transaction, Borrower may make cash dividends on the equity interests of Borrower solely for the purpose of funding the following: (A) to permit Holdings to pay, in the event Borrower is a passthrough entity or files a consolidated, combined, unitary or similar type tax return with Holdings, U.S. federal and state and local income taxes then due and payable pursuant to those returns solely to the extent such taxes are imposed solely with respect to the income of Borrower and its Subsidiaries; (B) ordinary operating expenses and other transaction expenses of Holdings, including overhead and administrative expenses and fees as well as directors' or managers' fees which are reasonable and customary and incurred in the ordinary course of business and attributable to the ownership or operations of Borrower and its Subsidiaries; (C) organizational, operating and other transaction costs and expenses (including overhead and administrative costs and expenses, professional fees and Public Company Costs) of Holdings which are reasonable and customary and incurred in the ordinary course of business and attributable to the ownership or operations of Borrower and its Subsidiaries; or (D) to pay reasonable and customary salary, bonus, severance and other benefits payable to current and former directors, officers, employees, members of management, managers and/or consultants of Borrower, any Subsidiary or Holdings to the extent such salaries, bonuses, severance and other benefits are attributable to the ownership or operation of Borrower or the Subsidiaries; or (b) directly or indirectly make any Investment (including, without limitation, by the formation of any Subsidiary) other than (A) Permitted Investments and (B) loans and advances to Holdings in lieu of, and not in excess of the amount of, dividends permitted to be made to such Holdings in accordance with clause (a) above, or permit any of its Subsidiaries to do so."

2.6 Section 7.11 (Holding Company). New Section 7.11 of the Loan Agreement is inserted following Section 7.10 as follows:

"7.11 Holding Company. At all times following the Business Combination Transaction, Holdings shall not (a) conduct, transact or otherwise engage in, or commit to conduct, transact or otherwise engage in, any business or operations, (b) incur, create, assume or suffer to exist any Indebtedness or other liabilities or financial obligations or (c) own, lease, manage or otherwise operate any properties or assets, other than the following activities: (i) ownership of equity interests of the Borrowers and the exercise of rights and performance of obligations in connection therewith, (ii) the issuance of its own equity interests, (iii) the maintenance of its entity existence, (iv) activities and contractual rights incidental to maintenance of its entity existence, (v) participating in administrative, legal, tax, management and accounting matters as the parent of a consolidated group of companies including the Borrower and its Subsidiaries, (vi) entering into and performing its obligations under the Loan Documents and the Mezzanine Loan Documents to which it is a party (including any amendments, restatements, amendments and restatements, supplements or modifications thereto, or consents in respect thereof), (vii) adoption of an option plan, equity incentive plan, employee stock purchase plan and related agreements and activities related thereto, (viii) guarantees of operating leases of its Subsidiaries, (ix) with respect to insurance policies and related contracts and agreements, or (x) activities otherwise expressly permitted to be taken by Holdings hereunder or reasonably incidental to any of the foregoing."

2.7 Section 8 (Events of Default). The "." following Section 8.10 is deleted and replaced with "; or", and new Section 8.11 is inserted following Section 8.10 as follows:

"8.11 Guaranty. (a) Any guaranty of any Obligations terminates or ceases for any reason to be in full force and effect; (b) any Guarantor does not perform any obligation or covenant under any guaranty of the Obligations that can be cured and has failed to cure the default within ten (10) days after the occurrence thereof; provided, however, that if the default cannot by its nature be cured within the ten (10) day period or cannot after diligent attempts by Guarantor be cured within such ten (10) day period, and such default is likely to be cured within a reasonable time, then Guarantor shall have an additional period (which shall not in any case exceed thirty (30) days) to attempt to cure such default, and within such reasonable time period the failure to cure the default shall not be deemed an Event of Default (but no Credit Extensions shall be made during such cure period); provided further, that a cure period shall not be applicable to a payment default arising from a failure to make a principal or interest payment when due; (c) any circumstance described in Sections 8.3 through 8.8 of this Agreement occurs with respect to any Guarantor."

2.8 Section 13 (Definitions). The following defined terms set forth in Section 13.1 are deleted in their entirety and replaced with the

following:

- ""Subject Month" is the month which is two (2) calendar months after any Testing Month. For example, when testing Borrower's Modified Current Ratio for May 2020, the Testing Month would be May 2020 and the Subject Month would be July 2020, and determination of whether a Streamline Period is in effect for July 2020 would be based on Borrower's Modified Current Ratio as of the last day of May 2020."
- ""Testing Month" is any month with respect to which Bank has tested Borrower's Modified Current Ratio to determine whether a Streamline Period is in effect."
- 2.9 Section 13 (Definitions). The following new defined terms are hereby inserted alphabetically in Section 13.1:
- ""Business Combination Transaction" has the meaning given to such term in the Second Amendment."
- ""Holdings" is Virgin Group Acquisition Corp. II, a Cayman Islands exempted company, which shall be domesticated as a Delaware public benefit corporation in connection with the Business Combination Transaction, and its successors and permitted assigns."
- ""Merger Agreement" has the meaning given to such term in the Second Amendment."
- ""Public Company Costs" is (i) costs, fees and expenses associated with, or in anticipation of, or preparation for (A) any initial public offering (or any initial public offering proposed and not consummated) and (B) compliance with the requirements of the Sarbanes-Oxley Act of 2002 and the rules and regulations promulgated in connection therewith and (ii) costs, fees and expenses relating to compliance with the provisions of the Securities Act and the Exchange Act (and any similar Requirements of Law under any other applicable jurisdiction), as applicable to companies with equity or debt securities held by the public, the rules of national securities exchanges applicable to companies with listed equity or debt securities, directors', managers' and/or employees' compensation or other costs to the extent attributable to being a public company, officer and director fee and expense reimbursement to the extent attributable to being a public company, costs, fees and expenses relating to investor relations, shareholder meetings and reports to shareholders or debtholders associated with being a public company, directors' and officers' insurance and other legal and other professional fees, listing fees and other costs and/or expenses associated with being a public company."
- ""Second Amendment" is the Second Amendment to Second Amended and Restated Loan and Security Agreement, dated as of May 9, 2022, by and among Borrower and Bank."

- **2.10 Section 13 (Definitions).** The following defined term set forth in Section 13.1 is deleted in its entirety:
- ""Key Person" is Borrower's Chief Financial Officer, who is Servaes Tholen as of the Effective Date."

3. Limitation of Amendments.

- **3.1** The amendments set forth in Section 2, above, are effective for the purposes set forth herein and shall be limited precisely as written and shall not be deemed to (a) be a consent to any amendment, waiver or modification of any other term or condition of any Loan Document, or (b) otherwise prejudice any right or remedy which Agent may now have or may have in the future under or in connection with any Loan Document.
- 3.2 This Amendment shall be construed in connection with and as part of the Loan Documents and all terms, conditions, representations, warranties, covenants and agreements set forth in the Loan Documents, except as herein amended, are hereby ratified and confirmed and shall remain in full force and effect.
 - 4. Representations and Warranties. To induce Bank to enter into this Amendment, Borrower hereby represents and warrants to Bank as follows:
- **4.1** Immediately after giving effect to this Amendment (a) the representations and warranties contained in the Loan Documents are true, accurate and complete in all material respects as of the date hereof (except to the extent such representations and warranties relate to an earlier date, in which case they are true and correct as of such date), and (b) no Event of Default has occurred and is continuing;
- **4.2** Borrower has the power and authority to execute and deliver this Amendment and to perform its obligations under the Loan Agreement, as amended by this Amendment;
- **4.3** The organizational documents of Borrower delivered to Bank on or around April 30, 2021 remain true, accurate and complete and have not been amended, supplemented or restated and are and continue to be in full force and effect;
- **4.4** The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, have been duly authorized;
- **4.5** The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, do not and will not contravene (a) any law or regulation binding on or affecting Borrower, (b) any contractual restriction with a Person binding on Borrower, (c) any order, judgment or decree of any court or other governmental or public body or authority, or subdivision thereof, binding on Borrower, or (d) the organizational documents of Borrower, or applicable consents or waivers have been obtained;

- 4.6 The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, do not require any order, consent, approval, license, authorization or validation of, or filing, recording or registration with, or exemption by any governmental or public body or authority, or subdivision thereof, binding on Borrower, except as already has been obtained or made; and
- **4.7** This Amendment has been duly executed and delivered by Borrower and is the binding obligation of Borrower, enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, liquidation, moratorium or other similar laws of general application and equitable principles relating to or affecting creditors' rights.
 - **5. Ratification of Loan Documents.** Borrower has read this Amendment and consents to the terms hereof and further acknowledges and confirms that all of its obligations under the Loan Agreement and the other Loan Documents are in full force and effect in accordance with their respective terms without setoff, defense, counter-claim or claims in recoupment.
 - **6. Mezzanine Loan Agreement.** Bank hereby consents to the amendments and waivers contained in the First Amendment to Mezzanine Loan Agreement executed in connection herewith.
 - **7. Integration**. This Amendment and the Loan Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of this Amendment and the Loan Documents merge into this Amendment and the Loan Documents.
 - 8. Counterparts. This Amendment may be executed in any number of counterparts and all of such counterparts taken together shall be deemed to constitute one and the same instrument. Each party hereto may execute this Amendment by electronic means and recognizes and accepts the use of electronic signatures and records by any other party hereto in connection with the execution and storage hereof.
 - **9.** Effectiveness. As a condition precedent to the effectiveness of this Amendment, Bank shall have received the following documents prior to or concurrently with this Amendment, each in form and substance reasonably satisfactory to Bank:
 - 9.1 This Amendment and the Second Supplement to IPSA duly executed on behalf of the parties hereto and thereto;
- **9.2** The duly executed First Amendment to Mezzanine Loan Agreement and satisfaction of all conditions precedent to the effectiveness thereof therein;
- **9.3** Long-form good standing certificate of Borrower certified by the Secretary of State of Delaware, as of a date no earlier than thirty (30) days prior to the date hereof;

9.4 Borrower's payment of (i) a fully earned, nonrefundable amendment fee in the amount of Ten Thousand Dollars (\$10,000.00) and (ii) Bank's reasonable legal fees and expenses incurred in connection with this Amendment to the extent Borrower shall have received invoices documenting such fees or expenses on or prior to the date hereof (or such later date as the Borrower shall reasonably agree); and

9.5 such other documents as Bank may reasonably request to effectuate the terms of this Amendment.

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed and delivered as of the date first written above.

BANK BORROWER

SILICON VALLEY BANK GROVE COLLABORATIVE, INC.

 By:
 /s/ Trefor Bacon
 By:
 /s/ Stuart Landesberg

 Name:
 Trefor Bacon
 Name:
 Stuart Landesberg

 Title:
 Director
 Title:
 Chief Executive Officer

[Signature Page to Second Amendment to Second Amended and Restated Loan and Security Agreement]

THIRD AMENDMENT TO SECOND AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT

This Third Amendment to Second Amended and Restated Loan and Security Agreement (this "Amendment") is entered into this 16th day of June, 2022, by and between **SILICON VALLEY BANK**, a California corporation ("Bank") and **GROVE COLLABORATIVE**, **INC.**, a Delaware public benefit corporation, formerly known as Treehouse Merger Sub II, LLC, a Delaware limited liability company ("Borrower").

RECITALS

- A. Bank and Borrower have entered into that certain Second Amended and Restated Loan and Security Agreement dated as of July 29, 2020, as amended by that certain First Amendment to Second Amended and Restated Loan and Security Agreement dated as of April 30, 2021, as further amended by that certain Second Amendment to Second Amended and Restated Loan and Security Agreement dated as of May 9, 2022, and as affected by that certain Assumption Agreement dated as of the date hereof (as the same may from time to time be further amended, modified, supplemented or restated, the "Loan Agreement").
 - B. Bank has extended credit to Borrower for the purposes permitted in the Loan Agreement.
 - C. Borrower has requested that Bank amend the Loan Agreement to make certain revisions to the Loan Agreement as more fully set forth herein.
- **D.** Bank has agreed to so amend certain provisions of the Loan Agreement, but only to the extent, in accordance with the terms, subject to the conditions and in reliance upon the representations and warranties set forth below.

AGREEMENT

NOW, **THEREFORE**, in consideration of the foregoing recitals and other good and valuable consideration, the receipt and adequacy of which is hereby acknowledged, and intending to be legally bound, the parties hereto agree as follows:

- 1. Definitions. Capitalized terms used but not defined in this Amendment shall have the meanings given to them in the Loan Agreement.
- 2. Amendments to Loan Agreement.
- **2.1 Section 6.2 (Financial Statements, Reports).** Clauses (c), (e), (f), (g), (h), (i), and (j) of Section 6.2 of the Loan Agreement are deleted in their entirety and replaced with the following:

"(c) as soon as available, but no later than thirty (30) days after the last day of each month, a company-prepared consolidated and consolidating (if applicable) balance sheet and income statement covering Holdings' consolidated operations and key performance indicators for such month in a form acceptable to Bank (the "Monthly Financial Statements"); provided, that, for the month ending on May 31, 2022, the Monthly Financial Statements shall cover the Borrower's (rather than Holdings') consolidated operations and key performance indicators for such month:"

"(e) as soon as available but no later than the earlier of forty-five (45) days after the last day of the fiscal year of Holdings or forty-five (45) days after approval by the Board, and within seven (7) days of any Board-approved updates or amendments thereto, (1) annual consolidated operating budgets (including income statements, balance sheets and cash flow statements, by month) for the upcoming fiscal year of Holdings, and (2) annual consolidated financial projections for the following fiscal year (on a monthly or quarterly basis), in each case as approved by the Board and commensurate in form and substance with those provided to the Board, together with any related business forecasts used in the preparation of such annual financial projections;"

"(f) as soon as available, and in any event within two hundred ten (210) days after the last day of Holdings' fiscal year, audited consolidated financial statements prepared under GAAP, consistently applied, together with an unqualified opinion (provided, however, Holdings' unqualified opinion on financial statements may contain a qualification as to going concern typical for venture backed companies similar to Holdings) on the financial statements from an independent certified public accounting firm reasonably acceptable to Bank;"

"(g) in the event that Borrower becomes subject to the reporting requirements under the Exchange Act within five (5) days of filing, copies of all periodic and other reports, proxy statements and other materials filed by Borrower and/or any Guarantor with the SEC, any Governmental Authority succeeding to any or all of the functions of the SEC or with any national securities exchange, or distributed to its shareholders, as the case may be. Documents required to be delivered pursuant to the terms hereof (to the extent any such documents are included in materials otherwise filed with the SEC) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date on which Borrower and/or any Guarantor posts such documents, or provides a link thereto, on Borrower's and/or any Guarantor's website on the internet at Borrower's and/or such Guarantor's website address; provided, however, Borrower and/or such Guarantor, as applicable, shall promptly notify Bank in writing (which may be by electronic mail) of the posting of any such documents;"

"(h) within five (5) Business Days of delivery by Holdings, copies of all material statements, reports and notices made generally available by Holdings to Holdings' security holders or to any holders of Subordinated Debt;"

- "(i) prompt report of any legal actions pending or threatened in writing against Holdings or any of its Subsidiaries that could reasonably be expected to result in damages or costs to Borrower or any of its Subsidiaries of, individually or in the aggregate, One Million Five Hundred Thousand Dollars (\$1,500,000) or more;"
- "(j) promptly, from time to time, such other information regarding Holdings, Borrower or compliance with the terms of any Loan Documents as reasonably requested by Bank; and"
- **2.2 Section 13 (Definitions).** The defined term "Permitted Indebtedness" set forth in Section 13.1 of the Loan Agreement is amended to delete clause (g) in its entirety and replace it with the following:
 - "(g) (i) Borrower's credit cards maintained with third parties in the ordinary course of Borrower's business, provided that the aggregate credit limits available thereunder do not exceed One Million Five Hundred Thousand Dollars (\$1,500,000) at any time and (ii) Indebtedness owed to any Person consisting of unpaid premiums for insurance (including property, casualty, business interruption, directors and officers or liability insurance) of Holdings, the Borrower or any of their respective Subsidiaries; provided such insurance financing arrangement has been approved in writing by Bank;"
- **2.3 Section 13 (Definitions).** The defined term "Permitted Liens" set forth in Section 13.1 of the Loan Agreement is amended to add the following new clause (k) thereto immediately following clause (j):
 - "(k) Liens on rights under insurance policies and the premiums and proceeds thereof securing obligations permitted by clause (g)(ii) of the definition of "Permitted Indebtedness"."
- **2.4 Section 13 (Definitions)**. The defined term "Holdings" set forth in Section 13.1 of the Loan Agreement is deleted in its entirety and replaced with the following:
 - ""Holdings" is Grove Collaborative Holdings, Inc., a Delaware public benefit corporation, formerly known as Virgin Group Acquisition Corp. II, a Cayman Islands exempted company, and its successors and permitted assigns."

3. Limitation of Amendments.

- 3.1 The amendments set forth in Section 2, above, are effective for the purposes set forth herein and shall be limited precisely as written and shall not be deemed to (a) be a consent to any amendment, waiver or modification of any other term or condition of any Loan Document, or (b) otherwise prejudice any right or remedy which Bank may now have or may have in the future under or in connection with any Loan Document.
- 3.2 This Amendment shall be construed in connection with and as part of the Loan Documents and all terms, conditions, representations, warranties, covenants and agreements set forth in the Loan Documents, except as herein amended, are hereby ratified and confirmed and shall remain in full force and effect.

- 4. Representations and Warranties. To induce Bank to enter into this Amendment, Borrower hereby represents and warrants to Bank as follows:
- **4.1** Immediately after giving effect to this Amendment (a) the representations and warranties contained in the Loan Documents are true, accurate and complete in all material respects as of the date hereof (except to the extent such representations and warranties relate to an earlier date, in which case they are true and correct as of such date), and (b) no Event of Default has occurred and is continuing;
- **4.2** Borrower has the power and authority to execute and deliver this Amendment and to perform its obligations under the Loan Agreement, as amended by this Amendment;
- **4.3** The organizational documents of Borrower delivered to Bank on or around the date hereof are true, accurate and complete and have not been amended, supplemented or restated and are and continue to be in full force and effect;
- **4.4** The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, have been duly authorized;
- 4.5 The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, do not and will not contravene (a) any law or regulation binding on or affecting Borrower, (b) any contractual restriction with a Person binding on Borrower, (c) any order, judgment or decree of any court or other governmental or public body or authority, or subdivision thereof, binding on Borrower, or (d) the organizational documents of Borrower, or applicable consents or waivers have been obtained;
- 4.6 The execution and delivery by Borrower of this Amendment and the performance by Borrower of its obligations under the Loan Agreement, as amended by this Amendment, do not require any order, consent, approval, license, authorization or validation of, or filing, recording or registration with, or exemption by any governmental or public body or authority, or subdivision thereof, binding on Borrower, except as already has been obtained or made; and
- **4.7** This Amendment has been duly executed and delivered by Borrower and is the binding obligation of Borrower, enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, liquidation, moratorium or other similar laws of general application and equitable principles relating to or affecting creditors' rights.
 - **5. Ratification of Loan Documents**. Borrower has read this Amendment and consents to the terms hereof and further acknowledges and confirms that all of its obligations under the Loan Agreement and the other Loan Documents are in full force and effect in accordance with their respective terms without setoff, defense, counter-claim or claims in recoupment.

- 6. Mezzanine Loan Agreement. Bank hereby consents to the amendments and waivers contained in the Second Amendment to Mezzanine Loan and Security Agreement executed in connection herewith.
- 7. Integration. This Amendment and the Loan Documents represent the entire agreement about this subject matter and supersede prior negotiations or agreements. All prior agreements, understandings, representations, warranties, and negotiations between the parties about the subject matter of this Amendment and the Loan Documents merge into this Amendment and the Loan Documents.
- 8. Counterparts. This Amendment may be executed in any number of counterparts and all of such counterparts taken together shall be deemed to constitute one and the same instrument. Each party hereto may execute this Amendment by electronic means and recognizes and accepts the use of electronic signatures and records by any other party hereto in connection with the execution and storage hereof.
- **9.** Effectiveness. As a condition precedent to the effectiveness of this Amendment, Bank shall have received the following documents prior to or concurrently with this Amendment, each in form and substance reasonably satisfactory to Bank:
- **9.1** This Amendment, the Unconditional Secured Guaranty and Pledge Agreement, the Assumption Agreement, the Corporate Borrowing Certificate and the Guarantor's Certificate to Guaranty, each duly executed on behalf of the parties hereto and thereto;
- 9.2 the duly executed Second Amendment to Mezzanine Loan and Security Agreement and satisfaction of all conditions precedent to the effectiveness thereof therein; and
 - 9.3 such other documents as Bank may reasonably request to effectuate the terms of this Amendment.

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed and delivered as of the date first written above.

BANK
SILICON VALLEY BANK
GROVE COLLABORATIVE, INC.

By: /s/ Jeff Sereno
By: /s/ Stuart Landesberg
Name: Jeff Sereno
Name: Stuart Landesberg

[Signature Page to Third Amendment to Second Amended and Restated Loan and Security Agreement]

Title:

Chief Executive Officer

Vice President

GROVE COLLABORATIVE, INC.

2016 EQUITY INCENTIVE PLAN

ADOPTED BY THE BOARD OF DIRECTORS: July 18, 2016 APPROVED BY THE STOCKHOLDERS: July 18, 2016 TERMINATION DATE: July 17, 2026

1. GENERAL.

- (a) Eligible Stock Award Recipients. Employees, Directors and Consultants are eligible to receive Stock Awards.
- (b) Available Stock Awards. The Plan provides for the grant of the following types of Stock Awards: (i) Incentive Stock Options, (ii) Nonstatutory Stock Options, (iii) Stock Appreciation Rights, (iv) Restricted Stock Awards, (v) Restricted Stock Unit Awards and (vi) Other Stock Awards.
- (c) Purpose. The Plan, through the grant of Stock Awards, is intended to help the Company secure and retain the services of eligible award recipients, provide incentives for such persons to exert maximum efforts for the success of the Company and any Affiliate and provide a means by which the eligible recipients may benefit from increases in value of the Common Stock.

2. ADMINISTRATION.

- (a) Administration by the Board. The Board will administer the Plan. The Board may delegate administration of the Plan to a Committee or Committees, as provided in Section 2(c).
 - (b) Powers of the Board. The Board will have the power, subject to, and within the limitations of, the express provisions of the Plan:
- (i) To determine (A) who will be granted Stock Awards; (B) when and how each Stock Award will be granted; (C) what type of Stock Award will be granted; (D) the provisions of each Stock Award (which need not be identical), including when a person will be permitted to exercise or otherwise receive cash or Common Stock under the Stock Award; (E) the number of shares of Common Stock subject to, or the cash value of, a Stock Award; and (F) the Fair Market Value applicable to a Stock Award.
- (ii) To construe and interpret the Plan and Stock Awards granted under it, and to establish, amend and revoke rules and regulations for administration of the Plan and Stock Awards. The Board, in the exercise of these powers, may correct any defect, omission or inconsistency in the Plan or in any Stock Award Agreement, in a manner and to the extent it will deem necessary or expedient to make the Plan or Stock Award fully effective.
 - (iii) To settle all controversies regarding the Plan and Stock Awards granted under it.
- (iv) To accelerate, in whole or in part, the time at which a Stock Award may be exercised or vest (or the time at which cash or shares of Common Stock may be issued in settlement thereof).

- (v) To suspend or terminate the Plan at any time. Except as otherwise provided in the Plan or a Stock Award Agreement, suspension or termination of the Plan will not impair a Participant's rights under the Participant's then-outstanding Stock Award without the Participant's written consent except as provided in subsection (viii) below.
- (vi) To amend the Plan in any respect the Board deems necessary or advisable, including, without limitation, by adopting amendments relating to Incentive Stock Options and certain nonqualified deferred compensation under Section 409A of the Code and/or bringing the Plan or Stock Awards granted under the Plan into compliance with the requirements for Incentive Stock Options or ensuring that they are exempt from, or compliant with, the requirements for nonqualified deferred compensation under Section 409A of the Code, subject to the limitations, if any, of applicable law. If required by applicable law or listing requirements, and except as provided in Section 9(a) relating to Capitalization Adjustments, the Company will seek stockholder approval of any amendment of the Plan that (A) materially increases the number of shares of Common Stock available for issuance under the Plan, (B) materially expands the class of individuals eligible to receive Stock Awards under the Plan, (C) materially increases the benefits accruing to Participants under the Plan, (D) materially reduces the price at which shares of Common Stock may be issued or purchased under the Plan, (E) materially extends the term of the Plan, or (F) materially expands the types of Stock Awards available for issuance under the Plan. Except as otherwise provided in the Plan or a Stock Award Agreement, no amendment of the Plan will materially impair a Participant's rights under an outstanding Stock Award without the Participant's written consent.
- (vii) To submit any amendment to the Plan for stockholder approval, including, but not limited to, amendments to the Plan intended to satisfy the requirements of Section 422 of the Code regarding Incentive Stock Options.
- (viii) To approve forms of Stock Award Agreements for use under the Plan and to amend the terms of any one or more Stock Awards, including, but not limited to, amendments to provide terms more favorable to the Participant than previously provided in the Stock Award Agreement, subject to any specified limits in the Plan that are not subject to Board discretion; provided however, that a Participant's rights under any Stock Award will not be impaired by any such amendment unless (A) the Company requests the consent of the affected Participant, and (B) such Participant consents in writing. Notwithstanding the foregoing, (1) a Participant's rights will not be deemed to have been impaired by any such amendment if the Board, in its sole discretion, determines that the amendment, taken as a whole, does not materially impair the Participant's rights, and (2) subject to the limitations of applicable law, if any, the Board may amend the terms of any one or more Stock Awards without the affected Participant's consent (A) to maintain the qualified status of the Stock Award as an Incentive Stock Option under Section 422 of the Code; (B) to change the terms of an Incentive Stock Option, if such change results in impairment of the Stock Award solely because it impairs the qualified status of the Stock Award as an Incentive Stock Option under Section 422 of the Code; (C) to clarify the manner of exemption from, or to bring the Stock Award into compliance with, Section 409A of the Code; (D) to comply with other applicable laws.
- (ix) Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company and that are not in conflict with the provisions of the Plan or Stock Awards.

- (x) To adopt such procedures and sub-plans as are necessary or appropriate to permit participation in the Plan by Employees, Directors or Consultants who are foreign nationals or employed outside the United States (provided that Board approval will not be necessary for immaterial modifications to the Plan or any Stock Award Agreement that are required for compliance with the laws of the relevant foreign jurisdiction).
- (xi) To effect, with the consent of any adversely affected Participant, (A) the reduction of the exercise, purchase or strike price of any outstanding Stock Award; (B) the cancellation of any outstanding Stock Award and the grant in substitution therefor of a new (1) Option or SAR, (2) Restricted Stock Award, (3) Restricted Stock Unit Award, (4) Other Stock Award, (5) cash and/or (6) other valuable consideration determined by the Board, in its sole discretion, with any such substituted award (x) covering the same or a different number of shares of Common Stock as the cancelled Stock Award and (y) granted under the Plan or another equity or compensatory plan of the Company; or (C) any other action that is treated as a repricing under generally accepted accounting principles.
- (c) Delegation to Committee. The Board may delegate some or all of the administration of the Plan to a Committee or Committees. If administration of the Plan is delegated to a Committee, the Committee will have, in connection with the administration of the Plan, the powers theretofore possessed by the Board that have been delegated to the Committee, including the power to delegate to a subcommittee of the Committee any of the administrative powers the Committee is authorized to exercise (and references in this Plan to the Board will thereafter be to the Committee or subcommittee, as applicable). Any delegation of administrative powers will be reflected in resolutions, not inconsistent with the provisions of the Plan, adopted from time to time by the Board or Committee (as applicable). The Board may retain the authority to concurrently administer the Plan with the Committee and may, at any time, revest in the Board some or all of the powers previously delegated.
- (d) Delegation to an Officer. The Board may delegate to one or more Officers the authority to do one or both of the following: (i) designate Employees who are not Officers to be recipients of Options and SARs (and, to the extent permitted by applicable law, other Stock Awards) and, to the extent permitted by applicable law, the terms of such Stock Awards, and (ii) determine the number of shares of Common Stock to be subject to such Stock Awards granted to such Employees; provided, however, that the Board resolutions regarding such delegation will specify the total number of shares of Common Stock that may be subject to the Stock Awards granted by such Officer and that such Officer may not grant a Stock Award to himself or herself. Any such Stock Awards will be granted on the form of Stock Award Agreement most recently approved for use by the Committee or the Board, unless otherwise provided in the resolutions approving the delegation authority. The Board may not delegate authority to an Officer who is acting solely in the capacity of an Officer (and not also as a Director) to determine the Fair Market Value pursuant to Section 13(t) below.
- (e) Effect of Board's Decision. All determinations, interpretations and constructions made by the Board in good faith will not be subject to review by any person and will be final, binding and conclusive on all persons.

3. SHARES SUBJECT TO THE PLAN.

(a) Share Reserve.

(i) Subject to Section 9(a) relating to Capitalization Adjustments, the aggregate number of shares of Common Stock that may be issued pursuant to Stock Awards from and after the Effective Date will not exceed 3,043,710 shares (the "Share Reserve").

- (ii) For clarity, the Share Reserve in this Section 3(a) is a limitation on the number of shares of Common Stock that may be issued pursuant to the Plan. Accordingly, this Section 3.(a) does not limit the granting of Stock Awards except as provided in Section 7(a).
- (b) Reversion of Shares to the Share Reserve. If a Stock Award or any portion thereof (i) expires or otherwise terminates without all of the shares covered by such Stock Award having been issued or (ii) is settled in cash (i.e., the Participant receives cash rather than stock), such expiration, termination or settlement will not reduce (or otherwise offset) the number of shares of Common Stock that may be available for issuance under the Plan. If any shares of Common Stock issued pursuant to a Stock Award are forfeited back to or repurchased by the Company because of the failure to meet a contingency or condition required to vest such shares in the Participant, then the shares that are forfeited or repurchased will revert to and again become available for issuance under the Plan. Any shares reacquired by the Company in satisfaction of tax withholding obligations on a Stock Award or as consideration for the exercise or purchase price of a Stock Award will again become available for issuance under the Plan.
- (c) Incentive Stock Option Limit. Subject to the Share Reserve and Section 9(a) relating to Capitalization Adjustments, the aggregate maximum number of shares of Common Stock that may be issued pursuant to the exercise of Incentive Stock Options will be a number of shares of Common Stock equal to three multiplied by the Share Reserve.
- (d) Source of Shares. The stock issuable under the Plan will be shares of authorized but unissued or reacquired Common Stock, including shares repurchased by the Company on the open market or otherwise.

4. ELIGIBILITY.

- (a) Eligibility for Specific Stock Awards. Incentive Stock Options may be granted only to employees of the Company or a "parent corporation" or "subsidiary corporation" thereof (as such terms are defined in Sections 424(e) and 424(f) of the Code). Stock Awards other than Incentive Stock Options may be granted to Employees, Directors and Consultants; provided, however, that Stock Awards may not be granted to Employees, Directors and Consultants who are providing Continuous Service only to any "parent" of the Company, as such term is defined in Rule 405, unless (i) the stock underlying such Stock Awards is treated as "service recipient stock" under Section 409A of the Code (for example, because the Stock Awards are granted pursuant to a corporate transaction such as a spin off transaction), (ii) the Company, in consultation with its legal counsel, has determined that such Stock Awards are otherwise exempt from Section 409A of the Code, or (iii) the Company, in consultation with its legal counsel, has determined that such Stock Awards comply with the distribution requirements of Section 409A of the Code.
- **(b) Ten Percent Stockholders.** A Ten Percent Stockholder will not be granted an Incentive Stock Option unless the exercise price of such Option is at least 110% of the Fair Market Value on the date of grant and the Option is not exercisable after the expiration of five years from the date of grant.
- (c) Consultants. A Consultant will not be eligible for the grant of a Stock Award if, at the time of grant, either the offer or sale of the Company's securities to such Consultant is not exempt under Rule 701 because of the nature of the services that the Consultant is providing to the Company, because the Consultant is not a natural person, or because of any other provision of Rule 701, unless the Company determines that such grant need not comply with the requirements of Rule 701 and will satisfy another exemption under the Securities Act as well as comply with the securities laws of all other relevant jurisdictions.

5. PROVISIONS RELATING TO OPTIONS AND STOCK APPRECIATION RIGHTS.

Each Option or SAR will be in such form and will contain such terms and conditions as the Board deems appropriate. All Options will be separately designated Incentive Stock Options or Nonstatutory Stock Options at the time of grant, and, if certificates are issued, a separate certificate or certificates will be issued for shares of Common Stock purchased on exercise of each type of Option. If an Option is not specifically designated as an Incentive Stock Option, or if an Option is designated as an Incentive Stock Option but some portion or all of the Option fails to qualify as an Incentive Stock Option under the applicable rules, then the Option (or portion thereof) will be a Nonstatutory Stock Option. The provisions of separate Options or SARs need not be identical; provided, however, that each Stock Award Agreement will conform to (through incorporation of provisions hereof by reference in the applicable Stock Award Agreement or otherwise) the substance of each of the following provisions:

- (a) Term. Subject to the provisions of Section 4(b) regarding Ten Percent Stockholders, no Option or SAR will be exercisable after the expiration of 10 years from the date of its grant or such shorter period specified in the Stock Award Agreement.
- (b) Exercise Price. Subject to the provisions of Section 4(b) regarding Ten Percent Stockholders, the exercise or strike price of each Option or SAR will be not less than 100% of the Fair Market Value of the Common Stock subject to the Option or SAR on the date the Stock Award is granted. Notwithstanding the foregoing, an Option or SAR may be granted with an exercise or strike price lower than 100% of the Fair Market Value of the Common Stock subject to the Stock Award if such Stock Award is granted pursuant to an assumption of or substitution for another option or stock appreciation right pursuant to a Corporate Transaction and in a manner consistent with the provisions of Section 409A of the Code and, if applicable, Section 424(a) of the Code. Each SAR will be denominated in shares of Common Stock equivalents.
- (c) Purchase Price for Options. The purchase price of Common Stock acquired pursuant to the exercise of an Option may be paid, to the extent permitted by applicable law and as determined by the Board in its sole discretion, by any combination of the methods of payment set forth below. The Board will have the authority to grant Options that do not permit all of the following methods of payment (or otherwise restrict the ability to use certain methods) and to grant Options that require the consent of the Company to use a particular method of payment. The permitted methods of payment are as follows:
 - (i) by cash, check, bank draft or money order payable to the Company;
- (ii) pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board that, prior to the issuance of the stock subject to the Option, results in either the receipt of cash (or check) by the Company or the receipt of irrevocable instructions to pay the aggregate exercise price to the Company from the sales proceeds;
 - (iii) by delivery to the Company (either by actual delivery or attestation) of shares of Common Stock;
- (iv) if an Option is a Nonstatutory Stock Option, by a "net exercise" arrangement pursuant to which the Company will reduce the number of shares of Common Stock issuable upon exercise by the largest whole number of shares with a Fair Market Value that does not exceed the aggregate exercise price; provided, however, that the Company will accept a cash or other payment from the Participant to the extent of any remaining balance of the aggregate exercise price not satisfied by such reduction in the number of whole shares to be issued. Shares of Common Stock will no longer be subject to an Option and will not be exercisable thereafter to the extent that (A) shares issuable upon exercise are used to pay the exercise price pursuant to the "net exercise," (B) shares are delivered to the Participant as a result of such exercise, and (C) shares are withheld to satisfy tax withholding obligations;

- (v) according to a deferred payment or similar arrangement with the Optionholder; provided, however, that interest will compound at least annually and will be charged at the minimum rate of interest necessary to avoid (A) the imputation of interest income to the Company and compensation income to the Optionholder under any applicable provisions of the Code, and (B) the classification of the Option as a liability for financial accounting purposes; or
 - (vi) in any other form of legal consideration that may be acceptable to the Board and specified in the applicable Stock Award Agreement.
- (d) Exercise and Payment of a SAR. To exercise any outstanding SAR, the Participant must provide written notice of exercise to the Company in compliance with the provisions of the Stock Appreciation Right Agreement evidencing such SAR. The appreciation distribution payable on the exercise of a SAR will be not greater than an amount equal to the excess of (A) the aggregate Fair Market Value (on the date of the exercise of the SAR) of a number of shares of Common Stock equal to the number of Common Stock equivalents in which the Participant is vested under such SAR, and with respect to which the Participant is exercising the SAR on such date, over (B) the aggregate strike price of the number of Common Stock equivalents with respect to which the Participant is exercising the SAR on such date. The appreciation distribution may be paid in Common Stock, in cash, in any combination of the two or in any other form of consideration, as determined by the Board and contained in the Stock Award Agreement evidencing such SAR.
- (e) Transferability of Options and SARs. The Board may, in its sole discretion, impose such limitations on the transferability of Options and SARs as the Board will determine. In the absence of such a determination by the Board to the contrary, the following restrictions on the transferability of Options and SARs will apply:
- (i) Restrictions on Transfer. An Option or SAR will not be transferable except by will or by the laws of descent and distribution (or pursuant to subsections (ii) and (iii) below), and will be exercisable during the lifetime of the Participant only by the Participant. The Board may permit transfer of the Option or SAR in a manner that is not prohibited by applicable tax and securities laws. Except as explicitly provided in the Plan, neither an Option nor a SAR may be transferred for consideration.
- (ii) Domestic Relations Orders. Subject to the approval of the Board or a duly authorized Officer, an Option or SAR may be transferred pursuant to the terms of a domestic relations order, official marital settlement agreement or other divorce or separation instrument as permitted by Treasury Regulation 1.421-1(b)(2). If an Option is an Incentive Stock Option, such Option may be deemed to be a Nonstatutory Stock Option as a result of such transfer.
- (iii) Beneficiary Designation. Subject to the approval of the Board or a duly authorized Officer, a Participant may, by delivering written notice to the Company, in a form approved by the Company (or the designated broker), designate a third party who, upon the death of the Participant, will thereafter be entitled to exercise the Option or SAR and receive the Common Stock or other consideration resulting from such exercise. In the absence of such a designation, upon the death of the Participant, the executor or administrator of the Participant's estate will be entitled to exercise the Option or SAR and receive the Common Stock or other consideration resulting from such exercise. However, the Company may prohibit designation of a beneficiary at any time, including due to any conclusion by the Company that such designation would be inconsistent with the provisions of applicable laws.

- (f) Vesting Generally. The total number of shares of Common Stock subject to an Option or SAR may vest and therefore become exercisable in periodic installments that may or may not be equal. The Option or SAR may be subject to such other terms and conditions on the time or times when it may or may not be exercised (which may be based on the satisfaction of performance goals or other criteria) as the Board may deem appropriate. The vesting provisions of individual Options or SARs may vary. The provisions of this Section 5(f) are subject to any Option or SAR provisions governing the minimum number of shares of Common Stock as to which an Option or SAR may be exercised.
- (g) Termination of Continuous Service. Except as otherwise provided in the applicable Stock Award Agreement or other agreement between the Participant and the Company, if a Participant's Continuous Service terminates (other than for Cause and other than upon the Participant's death or Disability), the Participant may exercise his or her Option or SAR (to the extent that the Participant was entitled to exercise such Stock Award as of the date of termination of Continuous Service) within the period of time ending on the earlier of (i) the date two months following the termination of the Participant's Continuous Service (or such longer or shorter period specified in the applicable Stock Award Agreement, which period will not be less than 30 days if necessary to comply with applicable laws unless such termination is for Cause) and (ii) the expiration of the term of the Option or SAR as set forth in the Stock Award Agreement. If, after termination of Continuous Service, the Participant does not exercise his or her Option or SAR (as applicable) within the applicable time frame, the Option or SAR will terminate.
- (h) Extension of Termination Date. If the exercise of an Option or SAR following the termination of the Participant's Continuous Service (other than for Cause and other than upon the Participant's death or Disability) would be prohibited at any time solely because the issuance of shares of Common Stock would violate the registration requirements under the Securities Act, then the Option or SAR will terminate on the earlier of (i) the expiration of a total period of time (that need not be consecutive) equal to the applicable post-termination exercise period after the termination of the Participant's Continuous Service during which the exercise of the Option or SAR would not be in violation of such registration requirements, and (ii) the expiration of the term of the Option or SAR as set forth in the applicable Stock Award Agreement. In addition, unless otherwise provided in a Participant's Stock Award Agreement, if the sale of any Common Stock received upon exercise of an Option or SAR following the termination of the Participant's Continuous Service (other than for Cause) would violate the Company's insider trading policy, then the Option or SAR will terminate on the earlier of (i) the expiration of the period of time (that need not be consecutive) equal to the applicable post-termination exercise period after the termination of the Participant's Continuous Service during which the sale of the Common Stock received upon exercise of the Option or SAR would not be in violation of the Company's insider trading policy, and (ii) the expiration of the term of the Option or SAR as set forth in the applicable Stock Award Agreement.
- (i) Disability of Participant. Except as otherwise provided in the applicable Stock Award Agreement or other agreement between the Participant and the Company, if a Participant's Continuous Service terminates as a result of the Participant's Disability, the Participant may exercise his or her Option or SAR (to the extent that the Participant was entitled to exercise such Option or SAR as of the date of termination of Continuous Service), but only within such period of time ending on the earlier of (i) the date 12 months following such termination of Continuous Service (or such longer or shorter period specified in the Stock Award Agreement, which period will not be less than six months if necessary to comply with applicable laws unless such termination is for Cause), and (ii) the expiration of the term of the Option or SAR as set forth in the Stock Award Agreement. If, after termination of Continuous Service, the Participant does not exercise his or her Option or SAR within the applicable time frame, the Option or SAR (as applicable) will terminate.

- (j) Death of Participant. Except as otherwise provided in the applicable Stock Award Agreement or other agreement between the Participant and the Company, if (i) a Participant's Continuous Service terminates as a result of the Participant's death, or (ii) the Participant dies within the period (if any) specified in the Stock Award Agreement for exercisability after the termination of the Participant's Continuous Service (for a reason other than death), then the Option or SAR may be exercised (to the extent the Participant was entitled to exercise such Option or SAR as of the date of death) by the Participant's estate, by a person who acquired the right to exercise the Option or SAR by bequest or inheritance or by a person designated to exercise the Option or SAR upon the Participant's death, but only within the period ending on the earlier of (i) the date 18 months following the date of death (or such longer or shorter period specified in the Stock Award Agreement, which period will not be less than six months if necessary to comply with applicable laws unless such termination is for Cause), and (ii) the expiration of the term of such Option or SAR as set forth in the Stock Award Agreement. If, after the Participant's death, the Option or SAR is not exercised within the applicable time frame, the Option or SAR (as applicable) will terminate.
- (k) Termination for Cause. Except as explicitly provided otherwise in a Participant's Stock Award Agreement or other individual written agreement between the Company or any Affiliate and the Participant, if a Participant's Continuous Service is terminated for Cause, the Option or SAR will terminate immediately upon such Participant's termination of Continuous Service, and the Participant will be prohibited from exercising his or her Option or SAR from and after the date of such termination of Continuous Service.
- (I) Non-Exempt Employees. If an Option or SAR is granted to an Employee who is a non-exempt employee for purposes of the Fair Labor Standards Act of 1938, as amended, the Option or SAR will not be first exercisable for any shares of Common Stock until at least six months following the date of grant of the Option or SAR (although the Stock Award may vest prior to such date). Consistent with the provisions of the Worker Economic Opportunity Act, (i) if such non-exempt Employee dies or suffers a Disability, (ii) upon a Corporate Transaction in which such Option or SAR is not assumed, continued, or substituted, (iii) upon a Change in Control, or (iv) upon the Participant's retirement (as such term may be defined in the Participant's Stock Award Agreement, in another agreement between the Participant and the Company, or, if no such definition, in accordance with the Company's then current employment policies and guidelines), the vested portion of any Options and SARs may be exercised earlier than six months following the date of grant. The foregoing provision is intended to operate so that any income derived by a non-exempt employee in connection with the exercise or vesting of an Option or SAR will be exempt from his or her regular rate of pay. To the extent permitted and/or required for compliance with the Worker Economic Opportunity Act to ensure that any income derived by a non-exempt employee in connection with the exercise, vesting or issuance of any shares under any other Stock Award will be exempt from the employee's regular rate of pay, the provisions of this Section 5(1) will apply to all Stock Awards and are hereby incorporated by reference into such Stock Award Agreements.
- (m) Early Exercise of Options. An Option may, but need not, include a provision whereby the Optionholder may elect at any time before the Optionholder's Continuous Service terminates to exercise the Option as to any part or all of the shares of Common Stock subject to the Option prior to the full vesting of the Option. Subject to the "Repurchase Limitation" in Section 8(1), any unvested shares of Common Stock so purchased may be subject to a repurchase right in favor of the Company or to any other restriction the Board determines to be appropriate. Provided that the "Repurchase Limitation" in Section 8(1) is not violated, the Company will not be required to exercise its repurchase right until at least six months (or such longer or shorter period of time required to avoid classification of the Option as a liability for financial accounting purposes) have elapsed following exercise of the Option unless the Board otherwise specifically provides in the Option Agreement.

- (n) Right of Repurchase. Subject to the "Repurchase Limitation" in Section 8(l), the Option or SAR may include a provision whereby the Company may elect to repurchase all or any part of the vested shares of Common Stock acquired by the Participant pursuant to the exercise of the Option or SAR.
- (o) Right of First Refusal. The Option or SAR may include a provision whereby the Company may elect to exercise a right of first refusal following receipt of notice from the Participant of the intent to transfer all or any part of the shares of Common Stock received upon the exercise of the Option or SAR. Such right of first refusal will be subject to the "Repurchase Limitation" in Section 8(1). Except as expressly provided in this Section 5(0) or in the Stock Award Agreement, such right of first refusal will otherwise comply with any applicable provisions of the bylaws of the Company.

6. PROVISIONS OF STOCK AWARDS OTHER THAN OPTIONS AND SARS.

- (a) Restricted Stock Awards. Each Restricted Stock Award Agreement will be in such form and will contain such terms and conditions as the Board will deem appropriate. To the extent consistent with the Company's bylaws, at the Board's election, shares of Common Stock underlying a Restricted Stock Award may be (i) held in book entry form subject to the Company's instructions until any restrictions relating to the Restricted Stock Award lapse; or (ii) evidenced by a certificate, which certificate will be held in such form and manner as determined by the Board. The terms and conditions of Restricted Stock Award Agreements may change from time to time, and the terms and conditions of separate Restricted Stock Award Agreements need not be identical. Each Restricted Stock Award Agreement will conform to (through incorporation of the provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions:
- (i) Consideration. A Restricted Stock Award may be awarded in consideration for (A) cash, check, bank draft or money order payable to the Company, (B) past services to the Company or an Affiliate, or (C) any other form of legal consideration (including future services) that may be acceptable to the Board, in its sole discretion, and permissible under applicable law.
- (ii) Vesting. Subject to the "Repurchase Limitation" in Section 8(l), shares of Common Stock awarded under the Restricted Stock Award Agreement may be subject to forfeiture to the Company in accordance with a vesting schedule to be determined by the Board.
- (iii) Termination of Participant's Continuous Service. If a Participant's Continuous Service terminates, the Company may receive through a forfeiture condition or a repurchase right, any or all of the shares of Common Stock held by the Participant that have not vested as of the date of termination of Continuous Service under the terms of the Restricted Stock Award Agreement.
- (iv) Transferability. Rights to acquire shares of Common Stock under the Restricted Stock Award Agreement will be transferable by the Participant only upon such terms and conditions as are set forth in the Restricted Stock Award Agreement, as the Board will determine in its sole discretion, so long as Common Stock awarded under the Restricted Stock Award Agreement remains subject to the terms of the Restricted Stock Award Agreement.

- (v) Dividends. A Restricted Stock Award Agreement may provide that any dividends paid on Restricted Stock will be subject to the same vesting and forfeiture restrictions as apply to the shares subject to the Restricted Stock Award to which they relate.
- (b) Restricted Stock Unit Awards. Each Restricted Stock Unit Award Agreement will be in such form and will contain such terms and conditions as the will Board deem appropriate. The terms and conditions of Restricted Stock Unit Award Agreements may change from time to time, and the terms and conditions of separate Restricted Stock Unit Award Agreements need not be identical. Each Restricted Stock Unit Award Agreement will conform to (through incorporation of the provisions hereof by reference in the Agreement or otherwise) the substance of each of the following provisions:
- (i) Consideration. At the time of grant of a Restricted Stock Unit Award, the Board will determine the consideration, if any, to be paid by the Participant upon delivery of each share of Common Stock subject to the Restricted Stock Unit Award. The consideration to be paid (if any) by the Participant for each share of Common Stock subject to a Restricted Stock Unit Award may be paid in any form of legal consideration that may be acceptable to the Board, in its sole discretion, and permissible under applicable law.
- (ii) Vesting. At the time of the grant of a Restricted Stock Unit Award, the Board may impose such restrictions on or conditions to the vesting of the Restricted Stock Unit Award as it, in its sole discretion, deems appropriate.
- (iii) Payment. A Restricted Stock Unit Award may be settled by the delivery of shares of Common Stock, their cash equivalent, any combination thereof or in any other form of consideration, as determined by the Board and contained in the Restricted Stock Unit Award Agreement.
- (iv) Additional Restrictions. At the time of the grant of a Restricted Stock Unit Award, the Board, as it deems appropriate, may impose such restrictions or conditions that delay the delivery of the shares of Common Stock (or their cash equivalent) subject to a Restricted Stock Unit Award to a time after the vesting of such Restricted Stock Unit Award.
- (v) Dividend Equivalents. Dividend equivalents may be credited in respect of shares of Common Stock covered by a Restricted Stock Unit Award, as determined by the Board and contained in the Restricted Stock Unit Award Agreement. At the sole discretion of the Board, such dividend equivalents may be converted into additional shares of Common Stock covered by the Restricted Stock Unit Award in such manner as determined by the Board. Any additional shares covered by the Restricted Stock Unit Award credited by reason of such dividend equivalents will be subject to all of the same terms and conditions of the underlying Restricted Stock Unit Award Agreement to which they relate.
- (vi) Termination of Participant's Continuous Service. Except as otherwise provided in the applicable Restricted Stock Unit Award Agreement, such portion of the Restricted Stock Unit Award that has not vested will be forfeited upon the Participant's termination of Continuous Service.
- (vii) Compliance with Section 409A of the Code. Notwithstanding anything to the contrary set forth herein, any Restricted Stock Unit Award granted under the Plan that is not exempt from the requirements of Section 409A of the Code shall contain such provisions so that such Restricted Stock Unit Award will comply with the requirements of Section 409A of the Code. Such restrictions, if any, shall be determined by the Board and contained in the Restricted Stock Unit Award Agreement evidencing such Restricted Stock Unit Award. For example, such restrictions may include, without limitation, a requirement that any Common Stock that is to be issued in a year following the year in which the Restricted Stock Unit Award vests must be issued in accordance with a fixed pre-determined schedule.

(c) Other Stock Awards. Other forms of Stock Awards valued in whole or in part by reference to, or otherwise based on, Common Stock, including the appreciation in value thereof (e.g., options or stock rights with an exercise price or strike price less than 100% of the Fair Market Value of the Common Stock at the time of grant) may be granted either alone or in addition to Stock Awards provided for under Section 5 and the preceding provisions of this Section 6. Subject to the provisions of the Plan, the Board will have sole and complete authority to determine the persons to whom and the time or times at which such Other Stock Awards will be granted, the number of shares of Common Stock (or the cash equivalent thereof) to be granted pursuant to such Other Stock Awards and all other terms and conditions of such Other Stock Awards.

7. COVENANTS OF THE COMPANY.

- (a) Availability of Shares. The Company will keep available at all times the number of shares of Common Stock reasonably required to satisfy then-outstanding Stock Awards.
- (b) Securities Law Compliance. The Company will seek to obtain from each regulatory commission or agency having jurisdiction over the Plan such authority as may be required to grant Stock Awards and to issue and sell shares of Common Stock upon exercise of the Stock Awards; provided, however, that this undertaking will not require the Company to register under the Securities Act the Plan, any Stock Award or any Common Stock issued or issuable pursuant to any such Stock Award. If, after reasonable efforts and at a reasonable cost, the Company is unable to obtain from any such regulatory commission or agency the authority that counsel for the Company deems necessary for the lawful issuance and sale of Common Stock under the Plan, the Company will be relieved from any liability for failure to issue and sell Common Stock upon exercise of such Stock Awards unless and until such authority is obtained. A Participant will not be eligible for the grant of a Stock Award or the subsequent issuance of cash or Common Stock pursuant to the Stock Award if such grant or issuance would be in violation of any applicable securities law.
- (c) No Obligation to Notify or Minimize Taxes. The Company will have no duty or obligation to any Participant to advise such holder as to the time or manner of exercising such Stock Award. Furthermore, the Company will have no duty or obligation to warn or otherwise advise such holder of a pending termination or expiration of a Stock Award or a possible period in which the Stock Award may not be exercised. The Company has no duty or obligation to minimize the tax consequences of a Stock Award to the holder of such Stock Award.

8. MISCELLANEOUS.

- (a) Use of Proceeds from Sales of Common Stock Proceeds from the sale of shares of Common Stock pursuant to Stock Awards will constitute general funds of the Company.
- (b) Corporate Action Constituting Grant of Stock Awards. Corporate action constituting a grant by the Company of a Stock Award to any Participant will be deemed completed as of the date of such corporate action, unless otherwise determined by the Board, regardless of when the instrument, certificate, or letter evidencing the Stock Award is communicated to, or actually received or accepted by, the Participant. In the event that the corporate records (e.g., Board consents, resolutions or minutes) documenting the corporate action constituting the grant contain terms (e.g., exercise price, vesting schedule or number of shares) that are inconsistent with those in the Stock Award Agreement or related grant documents as a result of a clerical error in the papering of the Stock Award Agreement or related grant documents, the corporate records will control and the Participant will have no legally binding right to the incorrect term in the Stock Award Agreement or related grant documents.

- (c) Stockholder Rights. No Participant will be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares of Common Stock subject to a Stock Award unless and until (i) such Participant has satisfied all requirements for exercise of, or the issuance of shares of Common Stock under, the Stock Award pursuant to its terms, and (ii) the issuance of the Common Stock subject to the Stock Award has been entered into the books and records of the Company.
- (d) No Employment or Other Service Rights Nothing in the Plan, any Stock Award Agreement or any other instrument executed thereunder or in connection with any Stock Award granted pursuant thereto will confer upon any Participant any right to continue to serve the Company or an Affiliate in the capacity in effect at the time the Stock Award was granted or will affect the right of the Company or an Affiliate to terminate (i) the employment of an Employee with or without notice and with or without cause, (ii) the service of a Consultant pursuant to the terms of such Consultant's agreement with the Company or an Affiliate, or (iii) the service of a Director pursuant to the bylaws of the Company or an Affiliate, and any applicable provisions of the corporate law of the state in which the Company or the Affiliate is incorporated, as the case may be.
- (e) Change in Time Commitment. In the event a Participant's regular level of time commitment in the performance of his or her services for the Company and any Affiliates is reduced (for example, and without limitation, if the Participant is an Employee of the Company and the Employee has a change in status from a full-time Employee to a part-time Employee or takes an extended leave of absence) after the date of grant of any Stock Award to the Participant, the Board has the right in its sole discretion to (x) make a corresponding reduction in the number of shares subject to any portion of such Stock Award that is scheduled to vest or become payable after the date of such change in time commitment, and (y) in lieu of or in combination with such a reduction, extend the vesting or payment schedule applicable to such Stock Award. In the event of any such reduction, the Participant will have no right with respect to any portion of the Stock Award that is so reduced or extended.
- (f) Incentive Stock Option Limitations. To the extent that the aggregate Fair Market Value (determined at the time of grant) of Common Stock with respect to which Incentive Stock Options are exercisable for the first time by any Optionholder during any calendar year (under all plans of the Company and any Affiliates) exceeds \$100,000 (or such other limit established in the Code) or otherwise does not comply with the rules governing Incentive Stock Options, the Options or portions thereof that exceed such limit (according to the order in which they were granted) or otherwise do not comply with such rules will be treated as Nonstatutory Stock Options, notwithstanding any contrary provision of the applicable Option Agreement(s).
- (g) Investment Assurances. The Company may require a Participant, as a condition of exercising or acquiring Common Stock under any Stock Award, (i) to give written assurances satisfactory to the Company as to the Participant's knowledge and experience in financial and business matters and/or to employ a purchaser representative reasonably satisfactory to the Company who is knowledgeable and experienced in financial and business matters and that the Participant is capable of evaluating, alone or together with the purchaser representative, the merits and risks of exercising the Stock Award; and (ii) to give written assurances satisfactory to the Company stating that the Participant is acquiring Common Stock subject to the Stock Award for the Participant's own account and not with any present intention of selling

or otherwise distributing the Common Stock. The foregoing requirements, and any assurances given pursuant to such requirements, will be inoperative if (A) the issuance of the shares upon the exercise or acquisition of Common Stock under the Stock Award has been registered under a then currently effective registration statement under the Securities Act, or (B) as to any particular requirement, a determination is made by counsel for the Company that such requirement need not be met in the circumstances under the then applicable securities laws. The Company may, upon advice of counsel to the Company, place legends on stock certificates issued under the Plan as such counsel deems necessary or appropriate in order to comply with applicable securities laws, including, but not limited to, legends restricting the transfer of the Common Stock.

- (h) Withholding Obligations. Unless prohibited by the terms of a Stock Award Agreement, the Company may, in its sole discretion, satisfy any federal, state or local tax withholding obligation relating to a Stock Award by any of the following means or by a combination of such means: (i) causing the Participant to tender a cash payment; (ii) withholding shares of Common Stock from the shares of Common Stock issued or otherwise issuable to the Participant in connection with the Stock Award; provided, however, that no shares of Common Stock are withheld with a value exceeding the minimum amount of tax required to be withheld by law (or such lesser amount as may be necessary to avoid classification of the Stock Award as a liability for financial accounting purposes); (iii) withholding cash from a Stock Award settled in cash; (iv) withholding payment from any amounts otherwise payable to the Participant; or (v) by such other method as may be set forth in the Stock Award Agreement.
- (i) Electronic Delivery. Any reference herein to a "written" agreement or document will include any agreement or document delivered electronically or posted on the Company's intranet (or other shared electronic medium controlled by the Company to which the Participant has access).
- (j) Deferrals. To the extent permitted by applicable law, the Board, in its sole discretion, may determine that the delivery of Common Stock or the payment of cash, upon the exercise, vesting or settlement of all or a portion of any Stock Award may be deferred and may establish programs and procedures for deferral elections to be made by Participants. Deferrals by Participants will be made in accordance with Section 409A of the Code. Consistent with Section 409A of the Code, the Board may provide for distributions while a Participant is still an employee or otherwise providing services to the Company. The Board is authorized to make deferrals of Stock Awards and determine when, and in what annual percentages, Participants may receive payments, including lump sum payments, following the Participant's termination of Continuous Service, and implement such other terms and conditions consistent with the provisions of the Plan and in accordance with applicable law.
- (k) Compliance with Section 409A of the Code. To the extent that the Board determines that any Stock Award granted hereunder is subject to Section 409A of the Code, the Stock Award Agreement evidencing such Stock Award shall incorporate the terms and conditions necessary to avoid the consequences specified in Section 409A(a)(1) of the Code. To the extent applicable, the Plan and Stock Award Agreements shall be interpreted in accordance with Section 409A of the Code. Notwithstanding anything to the contrary in the Plan (and unless the Stock Award Agreement specifically provides otherwise), if the shares of Common Stock are publicly traded, and if a Participant holding a Stock Award that constitutes "deferred compensation" under Section 409A of the Code is a "specified employee" for purposes of Section 409A of the Code, no distribution or payment of any amount that is due because of a "separation from service" (as defined in Section 409A of the Code without regard to alternative definitions thereunder) will be issued or paid before the date that is six months following the date of such Participant's "separation from service" (as defined in Section 409A of the Code without regard to alternative definitions thereunder) or, if earlier, the date of the Participant's death, unless such distribution or payment can be made in a manner that complies with Section 409A of the Code, and any amounts so deferred will be paid in a lump sum on the day after such six month period elapses, with the balance paid thereafter on the original schedule.

(I) Repurchase Limitation. The terms of any repurchase right will be specified in the Stock Award Agreement. The repurchase price for vested shares of Common Stock will be the Fair Market Value of the shares of Common Stock on the date of repurchase. Unless otherwise specified in another agreement, the repurchase price for unvested shares of Common Stock will be the lower of (i) the Fair Market Value of the shares of Common Stock on the date of repurchase or (ii) their original purchase price. However, the Company will not exercise its repurchase right until at least six months (or such longer or shorter period of time necessary to avoid classification of the Stock Award as a liability for financial accounting purposes) have elapsed following delivery of shares of Common Stock subject to the Stock Award, unless otherwise specifically provided by the Board.

9. ADJUSTMENTS UPON CHANGES IN COMMON STOCK; OTHER CORPORATE EVENTS.

- (a) Capitalization Adjustments. In the event of a Capitalization Adjustment, the Board will appropriately and proportionately adjust: (i) the class(es) and maximum number of securities subject to the Plan pursuant to Section 3(a), (ii) the class(es) and maximum number of securities that may be issued pursuant to the exercise of Incentive Stock Options pursuant to Section 3(c), and (iii) the class(es) and number of securities and price per share of stock subject to outstanding Stock Awards. The Board will make such adjustments, and its determination will be final, binding and conclusive.
- (b) Dissolution or Liquidation. Except as otherwise provided in the Stock Award Agreement, in the event of a dissolution or liquidation of the Company, all outstanding Stock Awards (other than Stock Awards consisting of vested and outstanding shares of Common Stock not subject to a forfeiture condition or the Company's right of repurchase) will terminate immediately prior to the completion of such dissolution or liquidation, and the shares of Common Stock subject to the Company's repurchase rights or subject to a forfeiture condition may be repurchased or reacquired by the Company notwithstanding the fact that the holder of such Stock Award is providing Continuous Service, provided, however, that the Board may, in its sole discretion, cause some or all Stock Awards to become fully vested, exercisable and/or no longer subject to repurchase or forfeiture (to the extent such Stock Awards have not previously expired or terminated) before the dissolution or liquidation is completed but contingent on its completion.
- (c) Corporate Transaction. The following provisions will apply to Stock Awards in the event of a Corporate Transaction unless otherwise provided in the instrument evidencing the Stock Award or any other written agreement between the Company or any Affiliate and the Participant or unless otherwise expressly provided by the Board at the time of grant of a Stock Award. In the event of a Corporate Transaction, then, notwithstanding any other provision of the Plan, the Board may take one or more of the following actions with respect to Stock Awards, contingent upon the closing or completion of the Corporate Transaction:
- (i) arrange for the surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company) to assume or continue the Stock Award or to substitute a similar stock award for the Stock Award (including, but not limited to, an award to acquire the same consideration paid to the stockholders of the Company pursuant to the Corporate Transaction);

- (ii) arrange for the assignment of any reacquisition or repurchase rights held by the Company in respect of Common Stock issued pursuant to the Stock Award to the surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company);
- (iii) accelerate the vesting, in whole or in part, of the Stock Award (and, if applicable, the time at which the Stock Award may be exercised) to a date prior to the effective time of such Corporate Transaction as the Board determines (or, if the Board does not determine such a date, to the date that is five days prior to the effective date of the Corporate Transaction), with such Stock Award terminating if not exercised (if applicable) at or prior to the effective time of the Corporate Transaction; provided, however, that the Board may require Participants to complete and deliver to the Company a notice of exercise before the effective date of a Corporate Transaction, which exercise is contingent upon the effectiveness of such Corporate Transaction;
 - (iv) arrange for the lapse, in whole or in part, of any reacquisition or repurchase rights held by the Company with respect to the Stock Award;
- (v) cancel or arrange for the cancellation of the Stock Award, to the extent not vested or not exercised prior to the effective time of the Corporate Transaction, in exchange for such cash consideration (including no consideration) as the Board, in its sole discretion, may consider appropriate; and
- (vi) make a payment, in such form as may be determined by the Board equal to the excess, if any, of (A) the value of the property the Participant would have received upon the exercise of the Stock Award immediately prior to the effective time of the Corporate Transaction, over (B) any exercise price payable by such holder in connection with such exercise. For clarity, this payment may be zero (\$0) if the value of the property is equal to or less than the exercise price. Payments under this provision may be delayed to the same extent that payment of consideration to the holders of the Company's Common Stock in connection with the Corporate Transaction is delayed as a result of escrows, earn outs, holdbacks or any other contingencies.

The Board need not take the same action or actions with respect to all Stock Awards or portions thereof or with respect to all Participants. The Board may take different actions with respect to the vested and unvested portions of a Stock Award.

(d) Change in Control. A Stock Award may be subject to additional acceleration of vesting and exercisability upon or after a Change in Control as may be provided in the Stock Award Agreement for such Stock Award or as may be provided in any other written agreement between the Company or any Affiliate and the Participant, but in the absence of such provision, no such acceleration will occur.

10. PLAN TERM; EARLIER TERMINATION OR SUSPENSION OF THE PLAN.

(a) Plan Term. The Board may suspend or terminate the Plan at any time. Unless terminated sooner by the Board, the Plan will automatically terminate on the day before the 10th anniversary of the earlier of (i) the date the Plan is adopted by the Board, or (ii) the date the Plan is approved by the stockholders of the Company. No Stock Awards may be granted under the Plan while the Plan is suspended or after it is terminated.

(b) No Impairment of Rights. Suspension or termination of the Plan will not impair rights and obligations under any Stock Award granted while the Plan is in effect except with the written consent of the affected Participant or as otherwise permitted in the Plan.

11. EFFECTIVE DATE OF PLAN.

This Plan will become effective on the Effective Date.

12. CHOICE OF LAW.

The laws of the State of Delaware will govern all questions concerning the construction, validity and interpretation of this Plan, without regard to that state's conflict of laws rules.

- 13. DEFINITIONS. As used in the Plan, the following definitions will apply to the capitalized terms indicated below:
- (a) "Affiliate" means, at the time of determination, any "parent" or "majority-owned subsidiary" of the Company, as such terms are defined in Rule 405. The Board will have the authority to determine the time or times at which "parent" or "majority-owned subsidiary" status is determined within the foregoing definition.
 - (b) "Board" means the Board of Directors of the Company.
- (c) "Capitalization Adjustment" means any change that is made in, or other events that occur with respect to, the Common Stock subject to the Plan or subject to any Stock Award after the Effective Date without the receipt of consideration by the Company through merger, consolidation, reorganization, recapitalization, reincorporation, stock dividend, dividend in property other than cash, large nonrecurring cash dividend, stock split, reverse stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure, or any similar equity restructuring transaction, as that term is used in Statement of Financial Accounting Standards Board Accounting Standards Codification Topic 718 (or any successor thereto). Notwithstanding the foregoing, the conversion of any convertible securities of the Company will not be treated as a Capitalization Adjustment.
- (d) "Cause" will have the meaning ascribed to such term in any written agreement between the Participant and the Company defining such term and, in the absence of such agreement, such term means, with respect to a Participant, the occurrence of any of the following events: (i) such Participant's commission of any felony or any crime involving fraud, dishonesty or moral turpitude under the laws of the United States or any state thereof; (ii) such Participant's attempted commission of, or participation in, a fraud or act of dishonesty against the Company; (iii) such Participant's intentional, material violation of any contract or agreement between the Participant and the Company or of any statutory duty owed to the Company; (iv) such Participant's unauthorized use or disclosure of the Company's confidential information or trade secrets; or (v) such Participant's gross, or material deliberate, misconduct. The determination that a termination of the Participant's Continuous Service is either for Cause or without Cause will be made by the Company, in its sole discretion. Any determination by the Company that the Continuous Service of a Participant was terminated with or without Cause for the purposes of outstanding Stock Awards held by such Participant will have no effect upon any determination of the rights or obligations of the Company or such Participant for any other purpose.

(e) "Change in Control" means the occurrence, in a single transaction or in a series of related transactions, of any one or more of the following events:

- (i) any Exchange Act Person becomes the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities other than by virtue of a merger, consolidation or similar transaction. Notwithstanding the foregoing, a Change in Control will not be deemed to occur (A) on account of the acquisition of securities of the Company directly from the Company, (B) on account of the acquisition of securities of the Company by an investor, any affiliate thereof or any other Exchange Act Person that acquires the Company's securities in a transaction or series of related transactions the primary purpose of which is to obtain financing for the Company through the issuance of equity securities or (C) solely because the level of Ownership held by any Exchange Act Person (the "Subject Person") exceeds the designated percentage threshold of the outstanding voting securities as a result of a repurchase or other acquisition of voting securities by the Company reducing the number of shares outstanding, provided that if a Change in Control would occur (but for the operation of this sentence) as a result of the acquisition of voting securities by the Company, and after such share acquisition, the Subject Person becomes the Owner of any additional voting securities that, assuming the repurchase or other acquisition had not occurred, increases the percentage of the then outstanding voting securities Owned by the Subject Person over the designated percentage threshold, then a Change in Control will be deemed to occur;
- (ii) there is consummated a merger, consolidation or similar transaction involving (directly or indirectly) the Company and, immediately after the consummation of such merger, consolidation or similar transaction, the stockholders of the Company immediately prior thereto do not Own, directly or indirectly, either (A) outstanding voting securities representing more than 50% of the combined outstanding voting power of the surviving Entity in such merger, consolidation or similar transaction or (B) more than 50% of the combined outstanding voting power of the parent of the surviving Entity in such merger, consolidation or similar transaction, in each case in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such transaction; or
- (iii) there is consummated a sale, lease, exclusive license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries, other than a sale, lease, license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries to an Entity, more than 50% of the combined voting power of the voting securities of which are Owned by stockholders of the Company in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such sale, lease, license or other disposition.

Notwithstanding the foregoing definition or any other provision of this Plan, (A) the term Change in Control will not include a sale of assets, merger or other transaction effected exclusively for the purpose of changing the domicile of the Company, and (B) the definition of Change in Control (or any analogous term) in an individual written agreement between the Company or any Affiliate and the Participant will supersede the foregoing definition with respect to Stock Awards subject to such agreement; *provided, however*, that if no definition of Change in Control or any analogous term is set forth in such an individual written agreement, the foregoing definition will apply.

- (f) "Code" means the Internal Revenue Code of 1986, as amended, including any applicable regulations and guidance thereunder.
- (g) "Committee" means a committee of one or more Directors to whom authority has been delegated by the Board in accordance with Section 2(c).

- (h) "Common Stock" means the common stock of the Company.
- (i) "Company" means Grove Collaborative, Inc., a Delaware corporation.
- (j) "Consultant" means any person, including an advisor, who is (i) engaged by the Company or an Affiliate to render consulting or advisory services and is compensated for such services, or (ii) serving as a member of the board of directors of an Affiliate and is compensated for such services. However, service solely as a Director, or payment of a fee for such service, will not cause a Director to be considered a "Consultant" for purposes of the Plan
- (k) "Continuous Service" means that the Participant's service with the Company or an Affiliate, whether as an Employee, Director or Consultant, is not interrupted or terminated. A change in the capacity in which the Participant renders service to the Company or an Affiliate as an Employee, Director or Consultant or a change in the Entity for which the Participant renders such service, provided that there is no interruption or termination of the Participant's service with the Company or an Affiliate, will not terminate a Participant's Continuous Service; provided, however, that if the Entity for which a Participant is rendering services ceases to qualify as an Affiliate, as determined by the Board in its sole discretion, such Participant's Continuous Service will be considered to have terminated on the date such Entity ceases to qualify as an Affiliate. For example, a change in status from an Employee of the Company to a Consultant of an Affiliate or to a Director will not constitute an interruption of Continuous Service. To the extent permitted by law, the Board or the chief executive officer of the Company, in that party's sole discretion, may determine whether Continuous Service will be considered interrupted in the case of (i) any leave of absence approved by the Board or chief executive officer, including sick leave, military leave or any other personal leave, or (ii) transfers between the Company, an Affiliate, or their successors. Notwithstanding the foregoing, a leave of absence will be treated as Continuous Service for purposes of vesting in a Stock Award only to such extent as may be provided in the Company's leave of absence policy, in the written terms of any leave of absence agreement or policy applicable to the Participant, or as otherwise required by law.
- (1) "Corporate Transaction" means the consummation, in a single transaction or in a series of related transactions, of any one or more of the following events:
- (i) a sale or other disposition of all or substantially all, as determined by the Board in its sole discretion, of the consolidated assets of the Company and its Subsidiaries;
 - (ii) a sale or other disposition of more than 50% of the outstanding securities of the Company;
 - (iii) a merger, consolidation or similar transaction following which the Company is not the surviving corporation; or
- (iv) a merger, consolidation or similar transaction following which the Company is the surviving corporation but the shares of Common Stock outstanding immediately preceding the merger, consolidation or similar transaction are converted or exchanged by virtue of the merger, consolidation or similar transaction into other property, whether in the form of securities, cash or otherwise.
 - (m) "Director" means a member of the Board.

- (n) "Disability" means, with respect to a Participant, the inability of such Participant to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment that can be expected to result in death or that has lasted or can be expected to last for a continuous period of not less than twelve (12) months as provided in Sections 22(e)(3) and 409A(a)(2)(c)(i) of the Code, and will be determined by the Board on the basis of such medical evidence as the Board deems warranted under the circumstances.
- (o) "Effective Date" means the effective date of this Plan, which is the earlier of (i) the date that this Plan is first approved by the Company's stockholders, and (ii) the date this Plan is adopted by the Board.
- (p) "Employee" means any person employed by the Company or an Affiliate. However, service solely as a Director, or payment of a fee for such services, will not cause a Director to be considered an "Employee" for purposes of the Plan.
 - (q) "Entity" means a corporation, partnership, limited liability company or other entity.
 - (r) "Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.
- (s) "Exchange Act Person" means any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act), except that "Exchange Act Person" will not include (i) the Company or any Subsidiary of the Company, (ii) any employee benefit plan of the Company or any Subsidiary of the Company or any trustee or other fiduciary holding securities under an employee benefit plan of the Company or any Subsidiary of the Company, (iii) an underwriter temporarily holding securities pursuant to a registered public offering of such securities, (iv) an Entity Owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their Ownership of stock of the Company; or (v) any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act) that, as of the Effective Date, is the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities.
- (t) "Fair Market Value" means, as of any date, the value of the Common Stock determined by the Board in compliance with Section 409A of the Code or, in the case of an Incentive Stock Option, in compliance with Section 422 of the Code.
- (u) "Incentive Stock Option" means an option granted pursuant to Section 5 of the Plan that is intended to be, and that qualifies as, an "incentive stock option" within the meaning of Section 422 of the Code.
 - (v) "Nonstatutory Stock Option" means an option granted pursuant to Section 5 of the Plan that does not qualify as an Incentive Stock Option.
 - (w) "Officer" means any person designated by the Company as an officer.
 - (x) "Option" means an Incentive Stock Option or a Nonstatutory Stock Option to purchase shares of Common Stock granted pursuant to the Plan.
- (y) "Option Agreement" means a written agreement between the Company and an Optionholder evidencing the terms and conditions of an Option grant. Each Option Agreement will be subject to the terms and conditions of the Plan.

- (z) "Optionholder" means a person to whom an Option is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Option.
- (aa) "Other Stock Award" means an award based in whole or in part by reference to the Common Stock which is granted pursuant to the terms and conditions of Section 6(c).
- (bb) "Other Stock Award Agreement" means a written agreement between the Company and a holder of an Other Stock Award evidencing the terms and conditions of an Other Stock Award grant. Each Other Stock Award Agreement will be subject to the terms and conditions of the Plan.
- (cc) "Own," "Owner," "Owner," "Ownership" A person or Entity will be deemed to "Own," to have "Owned," to be the "Owner" of, or to have acquired "Ownership" of securities if such person or Entity, directly or indirectly, through any contract, arrangement, understanding, relationship or otherwise, has or shares voting power, which includes the power to vote or to direct the voting, with respect to such securities.
- (dd) "Participant" means a person to whom a Stock Award is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Stock Award.
 - (ee) "Plan" means this 2016 Equity Incentive Plan.
 - (ff) "Restricted Stock Award" means an award of shares of Common Stock which is granted pursuant to the terms and conditions of Section 6(a).
- (gg) "Restricted Stock Award Agreement" means a written agreement between the Company and a holder of a Restricted Stock Award evidencing the terms and conditions of a Restricted Stock Award grant. Each Restricted Stock Award Agreement will be subject to the terms and conditions of the Plan.
- (hh) "Restricted Stock Unit Award" means a right to receive shares of Common Stock which is granted pursuant to the terms and conditions of Section 6(b).
- (ii) "Restricted Stock Unit Award Agreement" means a written agreement between the Company and a holder of a Restricted Stock Unit Award evidencing the terms and conditions of a Restricted Stock Unit Award grant. Each Restricted Stock Unit Award Agreement will be subject to the terms and conditions of the Plan.
 - (jj) "Rule 405" means Rule 405 promulgated under the Securities Act.
 - (kk) "Rule 701" means Rule 701 promulgated under the Securities Act.
 - (II) "Securities Act" means the Securities Act of 1933, as amended.
- (mm) "Stock Appreciation Right" or "SAR" means a right to receive the appreciation on Common Stock that is granted pursuant to the terms and conditions of Section 5.
- (nn) "Stock Appreciation Right Agreement" means a written agreement between the Company and a holder of a Stock Appreciation Right evidencing the terms and conditions of a Stock Appreciation Right grant. Each Stock Appreciation Right Agreement will be subject to the terms and conditions of the Plan.

- (00) "Stock Award" means any right to receive Common Stock granted under the Plan, including an Incentive Stock Option, a Nonstatutory Stock Option, a Restricted Stock Award, a Restricted Stock Unit Award, a Stock Appreciation Right or any Other Stock Award.
- (pp) "Stock Award Agreement" means a written agreement between the Company and a Participant evidencing the terms and conditions of a Stock Award grant. Each Stock Award Agreement will be subject to the terms and conditions of the Plan.
- (qq) "Subsidiary" means, with respect to the Company, (i) any corporation of which more than 50% of the outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether, at the time, stock of any other class or classes of such corporation will have or might have voting power by reason of the happening of any contingency) is at the time, directly or indirectly, Owned by the Company, and (ii) any partnership, limited liability company or other entity in which the Company has a direct or indirect interest (whether in the form of voting or participation in profits or capital contribution) of more than 50%.
- (rr) "Ten Percent Stockholder" means a person who Owns (or is deemed to Own pursuant to Section 424(d) of the Code) stock possessing more than 10% of the total combined voting power of all classes of stock of the Company or any Affiliate.

GROVE COLLABORATIVE, INC.

STOCK OPTION GRANT NOTICE (2016 EQUITY INCENTIVE PLAN)

GROVE COLLABORATIVE, INC. (the "Company"), pursuant to its 2016 Equity Incentive Plan (the "Plan"), hereby grants to Optionholder an option to purchase the number of shares of the Company's Common Stock set forth below. This option is subject to all of the terms and conditions as set forth in this notice, in the Option Agreement, the Plan and the Notice of Exercise, all of which are attached hereto and incorporated herein in their entirety. Capitalized terms not explicitly defined herein but defined in the Plan or the Option Agreement will have the same definitions as in the Plan or the Option Agreement. If there is any conflict between the terms in this notice and the Plan, the terms of the Plan will control.

	Optionholder:	«Optionee»
	Date of Grant:	«Grant_Date»
	Vesting Commencement Date:	«VestComDate»
	Number of Shares Subject to Option:	«NoofShares»
	Exercise Price (Per Share):	«ExercisePrice»
	Type of Option:	«Type»
	Expiration Date:	«ExpDate»
Exercise Sc	hedule: ⊠ Same as Vesting Schedule □ E	Early Exercise Permitted
Vesting Sch	nedule: «VestSchedule»	
Payment:	By one or a combination of the following items ((described in the Option Agreement):
	 ☑ By cash, check, bank draft or money order pay ☐ Pursuant to a Regulation T Program if the shay ☐ By delivery of already-owned shares if the shay ☐ If and only to the extent this option is a Nonst exercise" arrangement 	res are publicly traded
Option Agre modified, an Notice, the O supersede al	ement and the Plan. Optionholder acknowledges at mended or revised except as provided in the Plan. C Option Agreement, and the Plan set forth the entire	vieldges receipt of, and understands and agrees to, this Stock Option Grant Notice, the agrees that this Stock Option Grant Notice and the Option Agreement may not be Optionholder further acknowledges that as of the Date of Grant, this Stock Option Grant understanding between Optionholder and the Company regarding this option award and or representations on that subject with the exception of (i) options previously granted and ally.
Отн	HER AGREEMENTS:	

By accepting this option, you consent to receive such documents by electronic delivery and to participate in the Plan through anon-line or electronic system established and maintained by the Company or another third party designated by the Company.

GROVE COI	LLABORATIVE, INC.	OPTIONHOLDER:	
By:			
-	Signature	Signature	
Title:		Date:	
Date:			

ATTACHMENTS: Option Agreement, 2016 Equity Incentive Plan and Notice of Exercise

GROVE COLLABORATIVE, INC.

2016 EQUITY INCENTIVE PLAN

OPTION AGREEMENT (INCENTIVE STOCK OPTION OR NONSTATUTORY STOCK OPTION)

Pursuant to your Stock Option Grant Notice ("Grant Notice") and this Option Agreement, GROVE COLLABORATIVE, INC. (the "Company") has granted you an option under its 2016 Equity Incentive Plan (the "Plan") to purchase the number of shares of the Company's Common Stock indicated in your Grant Notice at the exercise price indicated in your Grant Notice. The option is granted to you effective as of the date of grant set forth in the Grant Notice (the "Date of Grant"). If there is any conflict between the terms in this Option Agreement and the Plan, the terms of the Plan will control. Capitalized terms not explicitly defined in this Option Agreement or in the Grant Notice but defined in the Plan will have the same definitions as in the Plan.

The details of your option, in addition to those set forth in the Grant Notice and the Plan, are as follows:

- 1. VESTING. Your option will vest as provided in your Grant Notice. Vesting will cease upon the termination of your Continuous Service.
- 2. NUMBER OF SHARES AND EXERCISE PRICE. The number of shares of Common Stock subject to your option and your exercise price per share in your Grant Notice will be adjusted for Capitalization Adjustments.
- 3. EXERCISE RESTRICTION FOR NON-EXEMPT EMPLOYEES. If you are an Employee eligible for overtime compensation under the Fair Labor Standards Act of 1938, as amended (that is, a "Non-Exempt Employee"), and except as otherwise provided in the Plan, you may not exercise your option until you have completed at least six months of Continuous Service measured from the Date of Grant, even if you have already been an employee for more than six months. Consistent with the provisions of the Worker Economic Opportunity Act, you may exercise your option as to any vested portion prior to such six month anniversary in the case of (i) your death or disability, (ii) a Corporate Transaction in which your option is not assumed, continued or substituted, (iii) a Change in Control or (iv) your termination of Continuous Service on your "retirement" (as defined in the Company's benefit plans).
- **4. EXERCISE PRIOR TO VESTING ("EARLY EXERCISE").** If permitted in your Grant Notice (*i.e.*, the "Exercise Schedule" indicates "Early Exercise Permitted") and subject to the provisions of your option, you may elect at any time that is both (i) during the period of your Continuous Service and (ii) during the term of your option, to exercise all or part of your option, including the unvested portion of your option; *provided, however*, that:
- (a) a partial exercise of your option will be deemed to cover first vested shares of Common Stock and then the earliest vesting installment of unvested shares of Common Stock;
- **(b)** any shares of Common Stock so purchased from installments that have not vested as of the date of exercise will be subject to the purchase option in favor of the Company as described in the Company's form of Early Exercise Stock Purchase Agreement;

- (c) you will enter into the Company's form of Early Exercise Stock Purchase Agreement with a vesting schedule that will result in the same vesting as if no early exercise had occurred; and
- (d) if your option is an Incentive Stock Option, then, to the extent that the aggregate Fair Market Value (determined at the Date of Grant) of the shares of Common Stock with respect to which your option plus all other Incentive Stock Options you hold are exercisable for the first time by you during any calendar year (under all plans of the Company and its Affiliates) exceeds \$100,000, your option(s) or portions thereof that exceed such limit (according to the order in which they were granted) will be treated as Nonstatutory Stock Options.
- **5. METHOD OF PAYMENT**. You must pay the full amount of the exercise price for the shares you wish to exercise. You may pay the exercise price in cash or by check, bank draft or money order payable to the Company or in any other manner *permitted by your Grant Notice*, which may include one or more of the following:
- (a) Provided that at the time of exercise the Common Stock is publicly traded, pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board that, prior to the issuance of Common Stock, results in either the receipt of cash (or check) by the Company or the receipt of irrevocable instructions to pay the aggregate exercise price to the Company from the sales proceeds. This manner of payment is also known as a "broker-assisted exercise", "same day sale", or "sell to cover".
- (b) Provided that at the time of exercise the Common Stock is publicly traded, by delivery to the Company (either by actual delivery or attestation) of already-owned shares of Common Stock that are owned free and clear of any liens, claims, encumbrances or security interests, and that are valued at Fair Market Value on the date of exercise. "Delivery" for these purposes, in the sole discretion of the Company at the time you exercise your option, will include delivery to the Company of your attestation of ownership of such shares of Common Stock in a form approved by the Company. You may not exercise your option by delivery to the Company of Common Stock if doing so would violate the provisions of any law, regulation or agreement restricting the redemption of the Company's stock.
- (c) If this option is a Nonstatutory Stock Option, subject to the consent of the Company at the time of exercise, by a "net exercise" arrangement pursuant to which the Company will reduce the number of shares of Common Stock issued upon exercise of your option by the largest whole number of shares with a Fair Market Value that does not exceed the aggregate exercise price. You must pay any remaining balance of the aggregate exercise price not satisfied by the "net exercise" in cash or other permitted form of payment. Shares of Common Stock will no longer be outstanding under your option and will not be exercisable thereafter if those shares (i) are used to pay the exercise price pursuant to the "net exercise," (ii) are delivered to you as a result of such exercise, and (iii) are withheld to satisfy your tax withholding obligations.
 - **6. WHOLE SHARES.** You may exercise your option only for whole shares of Common Stock.
- 7. SECURITIES LAW COMPLIANCE. In no event may you exercise your option unless the shares of Common Stock issuable upon exercise are then registered under the Securities Act or, if not registered, the Company has determined that your exercise and the issuance of the shares would be exempt from the registration requirements of the Securities Act. The exercise of your option also must comply with all other applicable laws and regulations governing your option, and you may not exercise your option if the Company determines that such exercise would not be in material compliance with such laws and regulations (including any restrictions on exercise required for compliance with Treas. Reg. 1.401(k)-1(d)(3), if applicable).

- **8.** TERM. You may not exercise your option before the Date of Grant or after the expiration of the option's term. The term of your option expires, subject to the provisions of Section 5(h) of the Plan, upon the earliest of the following:
 - (a) immediately upon the termination of your Continuous Service for Cause;
- (b) two months after the termination of your Continuous Service for any reason other than Cause, your Disability or your death (except as otherwise provided in Section 8(d) below); provided, however, that if during any part of such two month period your option is not exercisable solely because of the condition set forth in the section above relating to "Securities Law Compliance," your option will not expire until the earlier of the Expiration Date or until it has been exercisable for an aggregate period of three months after the termination of your Continuous Service; provided further, that if (i) you are a NonExempt Employee, (ii) your Continuous Service terminates within six months after the Date of Grant, and (iii) you have vested in a portion of your option at the time of your termination of Continuous Service, your option will not expire until the earlier of (x) the later of (A) the date that is seven months after the Date of Grant, and (B) the date that is two months after the termination of your Continuous Service, and (y) the Expiration Date;
 - (c) 12 months after the termination of your Continuous Service due to your Disability (except as otherwise provided in Section 8(d)) below;
- (d) 18 months after your death if you die either during your Continuous Service or within three (3) months after your Continuous Service terminates for any reason other than Cause;
 - (e) the Expiration Date indicated in your Grant Notice; or
 - (f) the day before the 10th anniversary of the Date of Grant.

If your option is an Incentive Stock Option, note that to obtain the federal income tax advantages associated with an Incentive Stock Option, the Code requires that at all times beginning on the Date of Grant and ending on the day three months before the date of your option's exercise, you must be an employee of the Company or an Affiliate, except in the event of your death or Disability. The Company has provided for extended exercisability of your option under certain circumstances for your benefit but cannot guarantee that your option will necessarily be treated as an Incentive Stock Option if you continue to provide services to the Company or an Affiliate as a Consultant or Director after your employment terminates or if you otherwise exercise your option more than three months after the date your employment with the Company or an Affiliate terminates.

9. EXERCISE.

(a) You may exercise the vested portion of your option (and the unvested portion of your option if your Grant Notice so permits) during its term by (i) delivering a Notice of Exercise (in a form designated by the Company) or completing such other documents and/or procedures designated by the Company for exercise and (ii) paying the exercise price and any applicable withholding taxes to the Company's Secretary, stock plan administrator, or such other person as the Company may designate, together with such additional documents as the Company may then require.

- (b) By exercising your option you agree that, as a condition to any exercise of your option, the Company may require you to enter into an arrangement providing for the payment by you to the Company of any tax withholding obligation of the Company arising by reason of (i) the exercise of your option, (ii) the lapse of any substantial risk of forfeiture to which the shares of Common Stock are subject at the time of exercise, or (iii) the disposition of shares of Common Stock acquired upon such exercise.
- (c) If your option is an Incentive Stock Option, by exercising your option you agree that you will notify the Company in writing within 15 days after the date of any disposition of any of the shares of the Common Stock issued upon exercise of your option that occurs within two years after the Date of Grant or within one year after such shares of Common Stock are transferred upon exercise of your option.
- (d) By exercising your option you agree that you will not sell, dispose of, transfer, make any short sale of, grant any option for the purchase of, or enter into any hedging or similar transaction with the same economic effect as a sale with respect to any shares of Common Stock or other securities of the Company held by you, for a period of 180 days following the effective date of a registration statement of the Company filed under the Securities Act or such longer period as the underwriters or the Company will request to facilitate compliance with FINRA Rule 2711 or NYSE Member Rule 472 or any successor or similar rules-or regulation (the "Lock-Up Period"); provided, however, that nothing contained in this section will prevent the exercise of a repurchase option, if any, in favor of the Company during the Lock-Up Period. You further agree to execute and deliver such other agreements as may be reasonably requested by the Company or the underwriters that are consistent with the foregoing or that are necessary to give further effect thereto. In order to enforce the foregoing covenant, the Company may impose stop-transfer instructions with respect to your shares of Common Stock until the end of such period. You also agree that any transfere of any shares of Common Stock (or other securities) of the Company held by you will be bound by this Section 9(d). The underwriters of the Company's stock are intended third party beneficiaries of this Section 9(d) and will have the right, power and authority to enforce the provisions hereof as though they were a party hereto.
- (e) By exercising your option you agree that you will execute such additional documents as the Company may then require including, without limitation, any voting agreement (with a drag along right), right of first refusal and co-sale agreement, or other agreement between the Company and certain of its stockholders. You further acknowledge and agree that the Company's receipt of your signature pages to such agreements will be a condition precedent to you being issued the shares as to which you are exercising your option.
- 10. TRANSFERABILITY. Except as otherwise provided in this Section 10, your option is not transferable, except by will or by the laws of descent and distribution, and is exercisable during your life only by you.
- (a) Certain Trusts. Upon receiving written permission from the Board or its duly authorized designee, you may transfer your option to a trust if you are considered to be the sole beneficial owner (determined under Section 671 of the Code and applicable state law) while the option is held in the trust. You and the trustee must enter into transfer and other agreements required by the Company.
- (b) Domestic Relations Orders. Upon receiving written permission from the Board or its duly authorized designee, and provided that you and the designated transferee enter into transfer and other agreements required by the Company, you may transfer your option pursuant to the terms of a domestic relations order, official marital settlement agreement or other divorce or separation instrument as permitted by Treasury Regulation 1.421-1(b)(2) that contains the information required by the Company

to effectuate the transfer. You are encouraged to discuss the proposed terms of any division of this option with the Company prior to finalizing the domestic relations order or marital settlement agreement to help ensure the required information is contained within the domestic relations order or marital settlement agreement. If this option is an Incentive Stock Option, this option may be deemed to be a Nonstatutory Stock Option as a result of such transfer.

- (c) Beneficiary Designation. Upon receiving written permission from the Board or its duly authorized designee, you may, by delivering written notice to the Company, in a form approved by the Company and any broker designated by the Company to handle option exercises, designate a third party who, on your death, will thereafter be entitled to exercise this option and receive the Common Stock or other consideration resulting from such exercise. In the absence of such a designation, your executor or administrator of your estate will be entitled to exercise this option and receive, on behalf of your estate, the Common Stock or other consideration resulting from such exercise.
- 11. OPTION NOT A SERVICE CONTRACT. Your option is not an employment or service contract, and nothing in your option will be deemed to create in any way whatsoever any obligation on your part to continue in the employ of the Company or an Affiliate, or of the Company or an Affiliate to continue your employment. In addition, nothing in your option will obligate the Company or an Affiliate, their respective stockholders, boards of directors, officers or employees to continue any relationship that you might have as a Director or Consultant for the Company or an Affiliate.

12. WITHHOLDING OBLIGATIONS.

- (a) At the time you exercise your option, in whole or in part, and at any time thereafter as requested by the Company, you hereby authorize withholding from payroll and any other amounts payable to you, and otherwise agree to make adequate provision for (including by means of a "same day sale" pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board to the extent permitted by the Company), any sums required to satisfy the federal, state, local and foreign tax withholding obligations of the Company or an Affiliate, if any, which arise in connection with the exercise of your option.
- (b) If this option is a Nonstatutory Stock Option, then upon your request and subject to approval by the Company, and compliance with any applicable legal conditions or restrictions, the Company may withhold from fully vested shares of Common Stock otherwise issuable to you upon the exercise of your option a number of whole shares of Common Stock having a Fair Market Value, determined by the Company as of the date of exercise, not in excess of the minimum amount of tax required to be withheld by law (or such lower amount as may be necessary to avoid classification of your option as a liability for financial accounting purposes). If the date of determination of any tax withholding obligation is deferred to a date later than the date of exercise of your option, share withholding pursuant to the preceding sentence shall not be permitted unless you make a proper and timely election under Section 83(b) of the Code, covering the aggregate number of shares of Common Stock acquired upon such exercise with respect to which such determination is otherwise deferred, to accelerate the determination of such tax withholding obligation to the date of exercise of your option.

 Notwithstanding the filing of such election, shares of Common Stock shall be withheld solely from fully vested shares of Common Stock determined as of the date of exercise of your option that are otherwise issuable to you upon such exercise. Any adverse consequences to you arising in connection with such share withholding procedure shall be your sole responsibility.
- (c) You may not exercise your option unless the tax withholding obligations of the Company and/or any Affiliate are satisfied. Accordingly, you may not be able to exercise your option when desired even though your option is vested, and the Company will have no obligation to issue a certificate for such shares of Common Stock or release such shares of Common Stock from any escrow provided for herein, if applicable, unless such obligations are satisfied.

13. TAX CONSEQUENCES. You hereby agree that the Company does not have a duty to design or administer the Plan or its other compensation programs in a manner that minimizes your tax liabilities. You will not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates related to tax liabilities arising from your option or your other compensation. In particular, you acknowledge that this option is exempt from Section 409A of the Code only if the exercise price per share specified in the Grant Notice is at least equal to the "fair market value" per share of the Common Stock on the Date of Grant and there is no other impermissible deferral of compensation associated with the option. Because the Common Stock is not traded on an established securities market, the Fair Market Value is determined by the Board, perhaps in consultation with an independent valuation firm retained by the Company. You acknowledge that there is no guarantee that the Internal Revenue Service will agree with the valuation as determined by the Board, and you will not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates in the event that the Internal Revenue Service asserts that the valuation determined by the Board is less than the "fair market value" as subsequently determined by the Internal Revenue Service.

14. RIGHT OF REPURCHASE.

- (a) Shares of Common Stock that you acquire upon exercise of your option are subject to any right of repurchase that may be described in the Company's bylaws in effect at such time the Company elects to exercise its right; *provided, however*, that if there is no right of repurchase described in the Company's bylaws at such time, the right of repurchase upon termination of your Continuous Service for Cause described below will apply. The Company's right of repurchase will expire on the first date upon which any security of the Company is listed (or approved for listing) upon notice of issuance on a national securities exchange or quotation system.
- (b) The Company may elect (but is not obligated) to repurchase all or any part of the shares of Common Stock that you acquire upon exercise of your option (the Company's "Repurchase Right"). If, from time to time, there is any stock dividend, stock split or other change in the character or amount of any of the outstanding Common Stock which is subject to the provisions of your option, then in such event any and all new, substituted or additional securities to which you are entitled by reason of your ownership of the shares of Common Stock acquired upon exercise of your option will be immediately subject to this Repurchase Right with the same force and effect as the shares subject to the Company's Repurchase Right immediately before such event.
- (c) The Company's Repurchase Right will be exercisable only within the ninety (90) day period following the Repurchase Event (as defined below) (or such longer period as may be required to avoid classification of the option as a liability for financial accounting purposes), or such longer period as may be agreed to by the Company and you (the "Repurchase Period"). The "Repurchase Event" occurs upon termination of your Continuous Service for Cause. The Repurchase Period will commence on the date of termination of your Continuous Service (or in the case of a post-termination exercise of your option, the date of such exercise).
- (d) The Company will exercise its Repurchase Right only for cash or cancellation of purchase money indebtedness for the shares of Common Stock and will give you written notice (by registered or certified mail) accompanied by payment for the shares of Common Stock within ninety (90) calendar days after the Repurchase Event or, if later, ninety (90) calendar days after a proper purchase of shares following such Repurchase Event (i.e., upon exercise of the option), including after any extension of the Repurchase Period for financial accounting purposes.

(e) The repurchase price will be equal to the shares' Fair Market Value on the date of repurchase; provided that if the Company terminates you for Cause, the repurchase price will be equal to the lower of (a) the shares' Fair Market Value on the date of repurchase, and (b) your original purchase price.

- 15. ESCROW OF SHARES. To ensure that the shares subject to the Company's Repurchase Right will be available for repurchase by the Company, the Company may require you to deposit the certificates evidencing the shares that you purchase upon exercise of your option with an escrow agent designated by the Company under the terms and conditions of an escrow agreement approved by the Company. If the Company does not require such deposit as a condition of exercise of your option, the Company reserves the right at any time to require you to so deposit the certificates in escrow. As soon as practicable after the expiration of the Company's Repurchase Right, the agent will deliver to you the shares of Common Stock and any other property no longer subject to such restriction. In the event the shares and any other property held in escrow are subject to the Company's exercise of its Repurchase Right, the notices required to be given to you will be given to the escrow agent. Within thirty (30) days after payment by the Company for the shares, the escrow agent will deliver the shares that the Company has purchased to the Company and will deliver the payment received from the Company to you.
- 16. NOTICES. Any notices provided for in your option or the Plan will be given in writing (including electronically) and will be deemed effectively given upon receipt or, in the case of notices delivered by mail by the Company to you, five days after deposit in the United States mail, postage prepaid, addressed to you at the last address you provided to the Company. The Company may, in its sole discretion, decide to deliver any documents related to participation in the Plan and this option by electronic means or to request your consent to participate in the Plan by electronic means. By accepting this option, you consent to receive such documents by electronic delivery and to participate in the Plan through an on-line or electronic system established and maintained by the Company or another third party designated by the Company.
- 17. GOVERNING PLAN DOCUMENT. Your option is subject to all the provisions of the Plan, the provisions of which are hereby made a part of your option, and is further subject to all interpretations, amendments, rules and regulations, which may from time to time be promulgated and adopted pursuant to the Plan. If there is any conflict between the provisions of your option and those of the Plan, the provisions of the Plan will control.

GROVE COLLABORATIVE, INC.

2016 EQUITY INCENTIVE PLAN

RESTRICTED STOCK UNIT AGREEMENT

Pursuant to your Restricted Stock Unit Grant Notice ("Grant Notice") and this Restricted Stock Unit Agreement (this "Agreement"), GROVE COLLABORATIVE, INC. (the "Company") has granted you restricted stock units (your "Award") under its 2016 Equity Incentive Plan (the "Plan") with respect to the number of shares of the Company's Common Stock indicated in your Grant Notice. The Award is granted to you effective as of the date of grant set forth in the Grant Notice (the "Date of Grant"). If there is any conflict between the terms in this Restricted Stock Unit Agreement and the Plan, the terms of the Plan will control. Capitalized terms not explicitly defined in this Restricted Stock Unit Agreement or in the Grant Notice but defined in the Plan will have the same definitions as in the Plan.

The details of your Award, in addition to those set forth in the Grant Notice and the Plan, are as follows:

- 1. VESTING. Your Award will vest as provided in your Grant Notice. Except as otherwise provided for in the Grant Notice, vesting will cease upon the termination of your Continuous Service.
 - 2. NUMBER OF SHARES. The number of shares of Common Stock subject to your Award will be adjusted for Capitalization Adjustments.
- 3. FORM OF SETTLEMENT. Your Award may be settled only in whole shares of Common Stock; provided, however, prior to the first date upon which any security of the Company is listed (or approved for listing) upon notice of issuance on a national securities exchange or quotation system, the Committee, in its sole and absolute discretion, may elect to settle the Award in cash based on the fair market value of a Share on the applicable vesting date, with fair market value in good faith determined by the Committee.
- 4. SECURITIES LAW COMPLIANCE. In no event may your Award be settled in shares unless the shares of Common Stock issuable upon such settlement are then registered under the Securities Act or, if not registered, the Company has determined that the issuance of the shares would be exempt from the registration requirements of the Securities Act. The settlement of your Award also must comply with all other applicable laws and regulations governing your Award, and your Award may not be settled if the Company determines that such settlement would not be in material compliance with such laws and regulations.
- 5. ISSUANCE OR DELIVERY OF SHARES. Except as otherwise provided for herein, within 60 days after the vesting of any portion of the Award, the Company shall issue or deliver, subject to the conditions of this Agreement, the vested shares of Common Stock to you (or, if elected by the Committee, a cash payment in accordance with Section 3); provided, however, that if the Liquidity Event (as defined in the Grant Notice) is an initial public offering or the consummation of a merger, consolidation or similar transaction in which the shares of Common Stock outstanding immediately preceding the merger, consolidation or similar transaction are converted or exchanged by virtue of the merger, consolidation or similar transaction into securities that are publicly-traded on an established exchange, then any shares of Common Stock that become vested prior to the lapsing of any lock-up period required in connection with such transaction shall be, to the extent permitted by Section 409A of the Code, issued or delivered to you on the first to occur of (i) the date on which such lock-up period lapses and (ii) a

date determined by the Board, which shall be no later than the 15th day of the third month following the year in which the substantial risk of forfeiture with respect to the Common Stock lapses. Such issuance or delivery shall be evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company. The Company shall pay all original issue or transfer taxes and all fees and expenses incident to such issuance or delivery, except as otherwise provided in Section 9. Prior to the issuance to you of the shares of Common Stock subject to the Award or, if applicable, the cash settlement of the Award, you shall have no direct or secured claim in any specific assets of the Company or in the shares of Common Stock subject to the Award, and will have the status of a general unsecured creditor of the Company.

- 6. TRANSFERABILITY. Except as otherwise provided in this Section 6, your Award is not transferable, except by will or by the laws of descent and distribution
- (a) Certain Trusts. Upon receiving written permission from the Board or its duly authorized designee, you may transfer your Award to a trust if you are considered to be the sole beneficial owner while the Award is held in the trust. You and the trustee must enter into transfer and other agreements required by the Company.
- (b) Domestic Relations Orders. Upon receiving written permission from the Board or its duly authorized designee, and provided that you and the designated transferee enter into transfer and other agreements required by the Company, you may transfer your Award pursuant to the terms of a domestic relations order, official marital settlement agreement or other divorce or separation instrument that contains the information required by the Company to effectuate the transfer. You are encouraged to discuss the proposed terms of any division of this Award with the Company prior to finalizing the domestic relations order or marital settlement agreement to help ensure the required information is contained within the domestic relations order or marital settlement agreement.
- (c) Beneficiary Designation. Upon receiving written permission from the Board or its duly authorized designee, you may, by delivering written notice to the Company, in a form approved by the Company and any broker designated by the Company to handle the settlement of restricted stock units, designate a third party who, on your death, will thereafter be entitled to receive the Common Stock or other consideration in connection with the settlement of this Award. In the absence of such a designation, your executor or administrator of your estate will be entitled to receive, on behalf of your estate, the Common Stock or other consideration resulting from such settlement.
- 7. AWARD NOT A SERVICE CONTRACT. Your Award is not an employment or service contract, and nothing in your Award will be deemed to create in any way whatsoever any obligation on your part to continue in the employ of the Company or an Affiliate, or of the Company or an Affiliate to continue your employment. In addition, nothing in your Award will obligate the Company or an Affiliate, their respective stockholders, boards of directors, officers or employees to continue any relationship that you might have as a Director or Consultant for the Company or an Affiliate.
- **8. RIGHTS AS A STOCKHOLDER.** You shall not be entitled to any privileges of ownership with respect to the shares of Common Stock subject to the Award (including, without limitation, any voting rights or rights to receive dividends) unless and until, and only to the extent, such shares become vested pursuant to your Grant Notice and you become a stockholder of record with respect to such shares.
- **9. WITHHOLDING OBLIGATIONS.** As a condition precedent to the issuance or delivery of the shares of Common Stock upon the vesting of the Award, you shall, upon request by the Company, pay to the Company such amount as the Company may be required, under all applicable federal, state, local or other laws or regulations, to withhold and pay over as income or other withholding taxes (the "*Required*"

Tax Payments") with respect to the Award. If you shall fail to advance the Required Tax Payments after request by the Company, the Company may, in its discretion, deduct any Required Tax Payments from any amount then or thereafter payable by the Company to you. You may elect to satisfy his or her obligation to advance the Required Tax Payments by any of the following means: (1) a cash payment to the Company; (2) if permitted by the Company, delivery to the Company (either actual delivery or by attestation procedures established by the Company) of previously owned whole shares of Common Stock having an aggregate Fair Market Value, determined as of the date on which such withholding obligation arises (the "Tax Date"), equal to the Required Tax Payments; (3) if permitted by the Company, authorizing the Company to withhold whole shares of Common Stock which would otherwise be delivered to you having an aggregate Fair Market Value, determined as of the Tax Date, equal to the Required Tax Payments; (4) to the extent permitted by applicable law, a cash payment by a broker-dealer acceptable to the Company to whom the participant has submitted an irrevocable notice of same-day sale or (5) any combination of (1), (2), (3) and (4). Shares of Common Stock to be delivered or withheld may not have a Fair Market Value in excess of the minimum amount of the Required Tax Payments (or such higher withholding amount permitted by the Committee). Any fraction of a share of Common Stock which would be required to satisfy any such obligation shall be delivered until the Required Tax Payments have been satisfied in full. Any determination by the Company with respect to the tendering or withholding of shares of Common Stock to satisfy the Required Tax Payments shall be made by the Board or the Committee if you are subject to Section 16 of the Securitas Exchange Act of 1934, as amended.

10. COMPLIANCE WITH SECTION 409A OF THE CODE. This Award is intended to be exempt from or comply with Section 409A of the Code, and shall be interpreted and construed accordingly, and each payment hereunder shall be considered a separate payment. To the extent this Award is nonqualified deferred compensation, within the meaning of Section 409A of the Code, and any agreement between you and the Company or any of its Affiliates provides that this Award shall become vested and be settled upon your termination of employment, the applicable shares of Common Stock shall be transferred to you or your beneficiary upon your "separation from service," within the meaning of Section 409A of the Code; provided that if you are a "specified employee," within the meaning of Section 409A of the Code, then to the extent the Award constitutes nonqualified deferred compensation, within the meaning of Section 409A of the Code, such shares of Common Stock shall be transferred to you or your beneficiary upon the earlier to occur of (i) the six-month anniversary of such separation from service and (ii) the date of your death, to the extent required to comply with Section 409A of the Code.

11. RIGHT OF REPURCHASE.

- (a) Shares of Common Stock that you acquire upon the settlement of your Award are subject to any right of repurchase that may be described in the Company's bylaws in effect at such time the Company elects to exercise its right; provided, however, that if there is no right of repurchase described in the Company's bylaws at such time, the right of repurchase upon termination of your Continuous Service for Cause described below will apply. The Company's right of repurchase will expire on the first date upon which any security of the Company is listed (or approved for listing) upon notice of issuance on a national securities exchange or quotation system.
- (b) The Company may elect (but is not obligated) to repurchase all or any part of the shares of Common Stock that you acquire upon the settlement of your Award (the Company's "Repurchase Right"). If, from time to time, there is any stock dividend, stock split or other change in the character or amount of any of the outstanding Common Stock which is subject to the provisions of your Award, then in such event any and all new, substituted or additional securities to which you are entitled by reason of your ownership of the shares of Common Stock acquired upon the settlement of your Award will be immediately subject to this Repurchase Right with the same force and effect as the shares subject to the Company's Repurchase Right immediately before such event.

- (c) The Company's Repurchase Right will be exercisable only within the ninety (90) day period following the Repurchase Event (as defined below) (or such longer period as may be required to avoid classification of the Award as a liability for financial accounting purposes), or such longer period as may be agreed to by the Company and you (the "Repurchase Period"). The "Repurchase Event" occurs upon termination of your Continuous Service for Cause or such other event as specified in the Company's bylaws. The Repurchase Period will commence on the date of termination of your Continuous Service or such other date as specified in the Company's bylaws.
- (d) The Company will exercise its Repurchase Right only for cash or cancellation of purchase money indebtedness for the shares of Common Stock and will give you written notice (by registered or certified mail) accompanied by payment for the shares of Common Stock within ninety (90) calendar days after the Repurchase Event, including after any extension of the Repurchase Period for financial accounting purposes.
- (e) The repurchase price will be equal to the shares' fair market value on the date of repurchase, as determined in good faith by the Committee; provided that if the Company terminates you for Cause, the repurchase price will be equal to the lower of (a) the shares' fair market value on the date of repurchase, as determined in good faith by the Committee, and (b) your original purchase price.
- 12. ESCROW OF SHARES. To ensure that the shares subject to the Company's Repurchase Right will be available for repurchase by the Company, the Company may require you to deposit the certificates evidencing the shares that you receive upon the settlement of your Award with an escrow agent designated by the Company under the terms and conditions of an escrow agreement approved by the Company. If the Company does not require such deposit as a condition of the settlement of your Award, the Company reserves the right at any time to require you to so deposit the certificates in escrow. As soon as practicable after the expiration of the Company's Repurchase Right, the agent will deliver to you the shares of Common Stock and any other property no longer subject to such restriction. In the event the shares and any other property held in escrow are subject to the Company's exercise of its Repurchase Right, the notices required to be given to you will be given to the escrow agent, and any payment required to be given to you will be given to the escrow agent. Within thirty (30) days after payment by the Company for the shares, the escrow agent will deliver the shares that the Company has purchased to the Company and will deliver the payment received from the Company to you.
- 13. NOTICES. Any notices provided for in your Award or the Plan will be given in writing (including electronically) and will be deemed effectively given upon receipt or, in the case of notices delivered by mail by the Company to you, five days after deposit in the United States mail, postage prepaid, addressed to you at the last address you provided to the Company. The Company may, in its sole discretion, decide to deliver any documents related to participation in the Plan and this Award by electronic means or to request your consent to participate in the Plan by electronic means. By accepting this Award, you consent to receive such documents by electronic delivery and to participate in the Plan through an on-line or electronic system established and maintained by the Company or another third party designated by the Company.
- **14. GOVERNING PLAN DOCUMENT.** Your Award is subject to all the provisions of the Plan, the provisions of which are hereby made a part of your Award, and is further subject to all interpretations, amendments, rules and regulations, which may from time to time be promulgated and adopted pursuant to the Plan. If there is any conflict between the provisions of your Award and those of the Plan, the provisions of the Plan will control.

GROVE COLLABORATIVE, INC.

RESTRICTED STOCK UNIT GRANT NOTICE (2016 EQUITY INCENTIVE PLAN)

GROVE COLLABORATIVE, INC. (the "Company"), pursuant to its 2016 Equity Incentive Plan (the 'Plan"), hereby grants to Participant restricted stock units (this "Award") with respect to the number of shares of the Company's Common Stock set forth below. This Award is subject to all of the terms and conditions as set forth in this notice, in the Restricted Stock Unit Agreement and the Plan, all of which are attached hereto and incorporated herein in their entirety. Capitalized terms not explicitly defined herein but defined in the Plan or the Restricted Stock Unit Agreement will have the same definitions as in the Plan or the Restricted Stock Unit Agreement. If there is any conflict between the terms in this notice and the Plan, the terms of the Plan will control.

Participant:	«Participant»
Date of Grant:	«Grant_Date»
Vesting Commencement Date:	«VestComDate»
Number of Shares Subject to Award:	«NoofShares»
Expiration Date	5th anniversary of Date of Grant

Vesting Schedule: Subject to the terms of the Restricted Stock Unit Agreement, the vesting of each Restricted Stock Unit requires the satisfaction of both of two conditions: an event condition and a service condition. The event condition will be satisfied if a Liquidity Event occurs prior to the Expiration Date so long as the Participant's Continuous Service continues through the date of such Liquidity Event. The service condition is satisfied with respect to 1/4th of the shares subject to the Award on the Company's Standard Quarterly Vesting Date (February 15th, May 15th, August 15th and November 15th of each year; provided, that if such date occurs on a weekend or federal holiday, vesting shall occur on the next business day) that occurs in the same calendar quarter as the first anniversary of the Vesting Commencement Date, with 1/16th of the shares subject to the Award vesting on each of the Company's Standard Quarterly Vesting Dates thereafter so long as the Participant's Continuous Service continues through the applicable vesting date, except as otherwise provided for in this Restricted Stock Unit Grant Notice. Any Restricted Stock Unit for which both conditions are satisfied shall become vested.

"Liquidity Event" means the first to occur, if any, of: (i) the closing of the initial underwritten public offering of securities of the class of equity securities then subject to the Award pursuant to an effective registration statement filed under the Securities Act; (ii) the time immediately prior to the consummation of a Change in Control and (iii) the consummation of a merger, consolidation or similar transaction in which the shares of Common Stock outstanding immediately preceding the merger, consolidation or similar transaction are converted or exchanged by virtue of the merger, consolidation or similar transaction into securities that are publicly-traded on an established exchange.

Additional Terms/Acknowledgements: Participant acknowledges receipt of, and understands and agrees to, this Restricted Stock Unit Grant Notice, the Restricted Stock Unit Agreement and the Plan. Participant acknowledges and agrees that this Restricted Stock Unit Grant Notice and the Restricted Stock Unit Agreement may not be modified, amended or revised except as provided in the Plan. Participant further acknowledges that as of the Date of Grant, this Restricted Stock Unit Grant Notice, the Restricted Stock Unit Agreement, and the Plan set forth the entire understanding between Participant and the Company regarding this Award and supersede all prior oral and written agreements, promises and/or representations on that subject with the exception of (i) Awards previously granted and delivered to Participant, and (ii) the following agreements only.

OTHER AGR	REEMENTS:		
	ward, you consent to receive such documents and maintained by the Company or another thi	by electronic delivery and to participate in the Plan through	h anon-line or electronic
GROVE COLLABORATIVE, INC.		PARTICIPANT:	
By:			
	Signature	Sig	nature
Title:		Date:	
Date:			

ATTACHMENTS: Restricted Stock Unit Agreement and 2016 Equity Incentive Plan

ATTACHMENT I

RESTRICTED STOCK UNIT AGREEMENT

ATTACHMENT II

2016 EQUITY INCENTIVE PLAN



June 23, 2022

Office of the Chief Accountant

Securities and Exchange Commission 100 F Street, NE Washington, D.C. 20549

Ladies and Gentlemen:

We have read Grove Collaborative, Inc.'s (formerly known as Virgin Group Acquisition Corp. II) statements included under Item 4.01 of its Form8-K dated June 23, 2022. We agree with the statements concerning our Firm under Item 4.01, in which we were informed of our dismissal on June 16, 2022. We are not in a position to agree or disagree with other statements contained therein.

Very truly yours,

WithumSmith+Brown, PC

Withem Smith + Brown , PC-

New York, New York

WithumSmith+Brown, PC 1411 Broadway, 9th Floor, New York, New York 10018-3496 T (212) 751 9100 F (212) 750 3262 withum.com

AN INDEPENDENT MEMBER OF HLB = THE GLOBAL ADVISORY AND ACCOUNTING NETWORK

List of Subsidiaries

of

$Grove\ Collaborative\ Holdings,\ Inc.$

1. Grove Collaborative, Inc.

UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

Introduction

The unaudited pro forma condensed combined financial information has been prepared in accordance with Article 11 oRegulation S-X as amended by the final rule, Release 33-10786 "Amendments to Financial Disclosures about Acquired and Disposed Businesses" and presents the combination of the historical financial statements of Virgin Group Acquisition Corp. II ("VGAC II") and Grove Collaborative, Inc. ("Grove"), adjusted to give effect to the Mergers and other related events contemplated by the Merger Agreement. VGAC II and Grove shall collectively be referred to herein as the "Companies." The Companies, after giving effect to the Mergers, shall be referred to herein as "New Grove."

The unaudited pro forma condensed combined balance sheet as of March 31, 2022 combines the historical unaudited condensed balance sheet of VGAC II as of March 31, 2022 with the historical unaudited condensed balance sheet of Grove as of March 31, 2022 on a pro forma basis as if the Mergers and the other events contemplated by the Merger Agreement, summarized below, had been consummated on March 31, 2022.

The unaudited pro forma condensed combined statement of operations for the three months ended March 31, 2022 combines the historical unaudited condensed statement of operations of VGAC II for the three months ended March 31, 2022 and the historical unaudited condensed statement of operations of Grove for the three months ended March 31, 2022 on a pro forma basis as if the Merger and other events contemplated by the Merger Agreement, summarized below, had been consummated on January 1, 2021.

The unaudited pro forma condensed combined statement of operations for the year ended December 31, 2021 combines the historical audited statement of operations of VGAC II for the period from January 13, 2021 (date of inception) through December 31, 2021 and the historical audited statement of operations of Grove for the year ended December 31, 2021 on a pro forma basis as if the Mergers and other events contemplated by the Merger Agreement, summarized below, had been consummated on January 1, 2021.

The unaudited pro forma condensed combined financial information was derived from and should be read in conjunction with the following historical financial statements and the accompanying notes:

- the (a) historical audited condensed financial statements of VGAC II as of December 31, 2021 and for the period from January 13, 2021 (inception) to December 31, 2021, included in VGAC II's Annual Report on Form 10-K filed with the SEC on February 24, 2022 and incorporated by reference and (b) historical unaudited condensed financial statements of VGAC II as of and for the three months ended March 31, 2022 included in VGAC II's Quarterly Report on Form 10-Q filed with the SEC on May 16, 2022 and incorporated by reference;
- the (a) historical audited financial statements of Grove as of and for the years ended December 31, 2019, 2020 and 2021 included in the
 proxy statement/consent solicitation statement/prospectus, and (b) historical unaudited condensed financial statements of Grove as of and for
 the three months ended March 31, 2022 included as Exhibit 99.1 to this Current Report on Form 8-K and are incorporated herein by
 reference;
- other information relating to Grove and VGAC II included in the proxy statement/consent solicitation statement/prospectus, including the Merger Agreement and the description of certain terms thereof set forth under the section titled "Business Combination Proposal."

The unaudited pro forma condensed combined financial statements have been presented for illustrative purposes only and do not necessarily reflect what New Grove's financial condition or results of operations would have been had the Mergers occurred on the dates indicated. Further, the unaudited pro forma condensed combined financial information also may not be useful in predicting the future financial condition and results of operations of New Grove. The actual financial position and results of operations may differ significantly from the pro forma amounts reflected herein due to a variety of factors. The unaudited pro forma adjustments represent management's estimates based on information available as of the date of these unaudited pro forma condensed combined financial statements and are subject to change as additional information becomes available and analyses are performed.

The unaudited pro forma condensed combined financial information should also be read together with "VGAC II's Management's Discussion and Analysis of Financial Condition and Results of Operations", "Grove's Management's Discussion and Analysis of Financial Condition and Results of Operations" and other financial information included elsewhere in the proxy statement/consent solicitation statement/prospectus.

Description of the Transactions

Pursuant to the Merger Agreement (as entered into on December 7, 2021 and amended and restated on March 31, 2022), the following transactions occurred:

- (i) On June 15, 2022, one day prior to the Closing Date, VGAC II became a Delaware public benefit corporation (the <u>Domestication</u>") and, in connection with the Domestication,
 - (A) VGAC II's name was changed to "Grove Collaborative Holdings, Inc." ('New Grove"),
 - (B) each then-issued and outstanding Class A ordinary share of VGAC II converted automatically into one share of Class A common stock of New Grove (the "New Grove Class A Common Stock"),
 - (C) each then-issued and outstanding Class B ordinary share of VGAC II converted automatically into one share of New Grove Class A Common Stock,
 - (D) each then-issued and outstanding warrant to purchase Class A ordinary shares of VGAC II ("Public Warrant") converted automatically into one warrant to purchase one share of New Grove Class A Common Stock, and
 - (E) each then-issued and outstanding units that had not previously separated into the underlying Class A ordinary shares and underlying Public Warrants were canceled and entitled the holder thereof to one share of New Grove Class A Common Stock and one-fifth of one warrant to purchase one share of New Grove Class A Common Stock.
- (ii) On the Closing Date, VGAC II Merger Sub I merged with and into Grove (the <u>finitial Merger</u>"), with Grove as the surviving company in the merger and, after giving effect to such merger, continuing as a wholly-owned subsidiary of New Grove (the "<u>Initial Surviving Corporation</u>"), and immediately following the Initial Merger, and as part of the same overall transaction as the Initial Merger, the Initial Surviving Corporation merged with and into VGAC II Merger Sub II (the "<u>Final Merger</u>" and, together with the Initial Merger, the "<u>Mergers</u>"), with VGAC II Merger Sub II as the surviving company in the Final Merger and, after giving effect to the Final Merger, continuing as a wholly-owned subsidiary of New Grove. The Mergers and the other transactions contemplated by the Merger Agreement other than the Domestication are hereinafter referred to as the "Business Combination."
- (iii) Upon the consummation of the Business Combination, Grove equity holders received or have the right to receive shares of class B common stock of New Grove (the "New Grove Class B Common Stock"), as set forth below, at a deemed value of \$10.00 per share after giving effect to the exchange ratio as determined pursuant to the Merger Agreement ("Exchange Ratio"). Accordingly, based on the Exchange Ratio of approximately 1.18, 133,390,456 shares of New Grove Class B Common Stock were immediately issued and outstanding (excluding those shares of New Grove Class B Common Stock that certain holders elected to be converted to New Grove Class A Common Stock at the time of issuance), 31,332,454 shares were reserved for the potential future issuance of New Grove Class B Common Stock upon the exercise of New Grove stock options, the vesting of New Grove restricted stock units, and the exercise of New Grove warrants, based on the following transactions contemplated by the Merger Agreement:
 - (A) the conversion of each issued and outstanding share of Grove's common stock (other than the Backstop Tranche 1 Shares) and Grove's preferred stock (on an as-converted to Grove common stock basis) (other than dissenting shares) into (i) a number of shares of New Grove Class B Common Stock as determined pursuant to the Exchange Ratio and (ii) a number of restricted shares of New Grove Class B Common Stock that will vest upon the achievement of certain earnout thresholds prior to the tenth anniversary of Closing (as further described below, the "Earnout Shares");

- (B) the conversion of all outstanding Grove warrants, excluding 275,631 warrants to purchase Grove common stock that were net settled prior to the consummation of the Business Combination, into (i) warrants exercisable for shares of New Grove Class B Common Stock with the same terms except for the number of shares exercisable and the exercise price, each of which was adjusted using the Exchange Ratio, and (ii) a number of Earnout Shares;
- (C) the conversion of all outstanding vested and unvested Grove stock options into (i) New Grove stock options exercisable for shares of New Grove Class B Common Stock with the same terms except for the number of shares exercisable and the exercise price, each of which was adjusted using the Exchange Ratio, and (ii) a number of Earnout Shares;
- (D) the conversion of all outstanding Grove restricted stock units into (i) New Grove restricted stock units with the same terms that each represent the right to receive the number of shares of New Grove Class B Common Stock adjusted using the Exchange Ratio and (ii) a number of Earnout Shares; and
- (E) the conversion of each issued and outstanding Backstop Tranche 1 Share (as defined below) into the right to receive a number of shares of New Grove Class B Common Stock equal to the Exchange Ratio (which were immediately exchanged on a one-for-one basis for shares of New Grove Class A Common Stock).

Notwithstanding the foregoing, pursuant to elections made shortly before the Closing, certain holders of Grove common stock and Grove preferred stock elected to receive, and at the Closing were issued, an aggregate of 5,711,098 shares of New Grove Class A Common Stock on a share-for-share basis in lieu of the New Grove Class B Common Stock they would have otherwise been entitled to receive.

The holders of New Grove Class A Common Stock have one vote per share and the holders of New Grove Class B Common Stock have ten votes per share. The New Grove Class B Common Stock is subject to automatic conversion to New Grove Class A Common Stock upon any transfers (except for certain permitted transfers).

Other Related Events in Connection with the Business Combination

Other related events that took place in connection with the Business Combination are summarized below:

Backstop Subscription Agreement

On March 31, 2022, VGAC II entered into a subscription agreement (the 'Backstop Subscription Agreement') with Corvina Holdings Limited, an affiliate of the Sponsor (the "Backstop Investor") and Grove, pursuant to which, among other things:

- (i) The Backstop Investor purchased, and Grove sold to the Backstop Investor, on the date of the Subscription Agreement, a number of shares of Grove common stock equal to the quotient of \$27,500,000 and \$11.70, for an aggregate purchase price of \$27,500,000 (such shares of Grove common stock, together with any other shares of Grove common stock issued to the Backstop Investor prior to the closing of the Business Combination, the "Backstop Tranche 1 Shares").
- (ii) The Backstop Investor agreed to subscribe for and purchase, on the Closing Date, certain shares of New Grove Class A Common Stock at a purchase price of \$10.00 per share (the "Backstop Tranche 2 Shares"), for aggregate gross proceeds in an amount equal to (x) \$22,500,000 minus (y) the amount of cash available, as of immediately prior to the Closing, to be released from the trust account (after giving effect to all payments to VGAC II shareholders that exercise their redemption rights in connection with the Business Combination).

As further described in the proxy statement/consent solicitation statement/prospectus, the Backstop Subscription Agreement also provides:

(i) For an adjustment to the number of Backstop Tranche 1 Shares held by the Backstop Investor immediately prior to the closing of the Business Combination to reflect any differences between the estimate of the Exchange Ratio used for purposes of determining the purchase price per share for the Backstop Tranche 1 Shares and the final Exchange Ratio calculated pursuant to the terms of the Merger Agreement.

- (ii) That immediately prior to the closing of the Business Combination, to the extent the amount of cash available, as of immediately prior to the Closing, to be released from the trust account (after giving effect to all payments to VGAC II shareholders that exercise their redemption rights in connection with the Business Combination) exceeds \$22,500,000, the Backstop Investor will have the right to redeem all or a portion of the Backstop Tranche 1 Shares in cash for a purchase price per share equal to (x) the final Exchange Ratio calculated pursuant to the Merger Agreement multiplied by (y) \$10.00.
- (iii) That immediately following the closing of the Business Combination:
 - (A) each share of New Grove Class B Common Stock issued to the Backstop Investor pursuant to the Merger Agreement as consideration for the Backstop Tranche 1 Shares will be exchanged for one share of New Grove Class A Common Stock (the "Backstop Share Exchange"), and
 - (B) New Grove will issue to the Backstop Investor a number of warrants to purchase New Grove Class A Common Stock (each warrant exercisable to purchase one share of New Grove Class A Common Stock for \$0.01) (such warrants, the "Backstop Warrants"). The number of warrants to be issued is the product of (i) the sum of (x) 0.75% and (y) the product of (1) 1.25% multiplied by (2) the quotient of (A) the number of Tranche 1 and Tranche 2 Backstop Shares divided by (B) 5,000,000 and (ii) the number of VGAC Common Shares, determined on a fully diluted basis, as of immediately following the Business Combination Closing. Such warrants shall be exercisable by Subscriber at any time for a period of five years from the date of issuance.
- (iv) For additional shares of New Grove Class A Common Stock to be issued to the Backstop Investor if the volume weighted average price of New Grove Class A Common Stock is less than \$10.00 during the 10 trading days commencing on the first trading day after New Grove's first quarterly earnings call for the fiscal quarter ending June 30, 2022 ("Additional Shares"). The Backstop Investor shall be entitled to receive additional VGAC Class A Common Shares equal to the lesser of (i) the product of (x) the sum of (1) the VGAC Class B Common Shares issued to Subscriber at the Business Combination Closing pursuant to the Business Combination Agreement as consideration for the Tranche 1 Shares and (2) the Tranche 2 Shares, if any, multiplied by (y) a fraction, (A) the numerator of which is \$10.00 (as adjusted for any stock split, reverse stock split or similar adjustment following the Business Combination Closing) minus the Measurement Period VWAP, and (B) the denominator of which is the Measurement Period VWAP and (ii) the number of Post-Combination VGAC Shares outstanding as of immediately following the Business Combination Closing.

On June 16, 2022, New Grove exchanged or issued the following:

- 2,338,352 shares of Backstop Tranche 1 shares were exchanged for 2,750,000 shares of New Grove Class B Common Stock, which was
 immediately exchanged on a one-for-one basis for shares of New Grove Class A Common Stock.
- Backstop Investor purchased 1,671,524 shares of Backstop Tranche 2 Shares for aggregate gross proceeds in an amount equal to \$16,715,524;
- Backstop Investor received 3,875,028 Backstop Warrants.

PIPE Subscription Agreements

On December 7, 2021, VGAC II entered into the Subscription Agreements with respect to the PIPE Financing. Pursuant to the Subscription Agreements, certain accredited and strategic investors have purchased 8,707,500 shares of New Grove Class A Common Stock ("PIPE Shares"), for a purchase price of \$10.00 per share and an aggregate purchase price of \$87,075,000, of which only 8,607,500 shares of New Grove Class A Common Stock have been issued for aggregate proceeds of \$86,075,000. The Company expects to issue the balance of these shares of New Grove Class A Common Stock and collect the remaining proceeds of the PIPE Financing on a subsequent date.

Grove Earnout Shares

At the closing of the Merger, Grove shareholders (including Grove stock option, restricted stock unit, and warrant holders) (other than the Backstop Investor) were issued 13,999,960 shares of New Grove Class B Common Stock ("Grove Earnout Shares"). Such shares are subject to vesting and forfeitures based upon certain triggering events that can occur during a period of ten years following the closing of the Business Combination (the "Earnout Period"). The triggering events that will result in the vesting of the Grove Earnout Shares during the Earnout Period are the following:

- 7,000,173 shares earned if the share price of New Grove Class A Common Stock is greater than or equal to \$12.50 over any 20 trading days within any consecutive 30 trading day period during the Earnout Period;
- 6,999,787 shares earned, including the shares subject to the \$12.50 threshold if not previously vested, if the share price of New Grove
 Class A Common Stock is greater than or equal to \$15.00 over any 20 trading days within any 30 consecutive trading day period during the
 Earnout Period; and
- If, during the Earnout Period, there is a Change of Control Transaction (as defined in the Merger Agreement), then all remaining triggering events that have not previously occurred and the related vesting conditions shall be deemed to have occurred.

If, at any time prior to the expiration of the Earnout Period, any holder of Grove Earnout Shares forfeits all or any portion of such holder's converted Grove options and restricted stock units, all unvested Grove Earnout Shares issued to such holder with respect to any such awards shall be automatically forfeited to New Grove and distributed to the other holders of Grove securities as of immediately prior to the closing of the Business Combination on a pro rata basis.

Grove Class A Share Issuance

Immediately after the Closing, New Grove issued an aggregate of 32,300 shares (<u>'Employee Stock Grants'</u>) of New Grove Class A Common Stock, net of taxes, to certain of its employees of the Company's fulfillment centers, none of which are officers or directors of New Grove or any of its predecessors.

Expected Accounting Treatment of the Business Combination

The Merger will be accounted for as a reverse recapitalization in accordance with GAAP. Under this method of accounting, VGAC II will be treated as the "acquired" company for financial reporting purposes. Accordingly, for accounting purposes, the financial statements of New Grove will represent a continuation of the financial statements of Grove with the Business Combination treated as the equivalent of Grove issuing stock for the net assets of VGAC II, accompanied by a recapitalization. The net assets of VGAC II will be stated at historical cost, with no goodwill or other intangible assets recorded. Operations prior to the Business Combination will be those of Grove in future reports of New Grove.

Grove has been determined to be the accounting acquirer based on evaluation of the following facts and circumstances:

- Current Grove stockholders will have a relative majority of the voting power of New Grove;
- The New Grove Board will have nine members (such members currently consisting of eight members and one vacant board seat) of whom
 one individual shall be designated by VGAC II and of whom eight individuals shall be designated by Grove;
- · Grove's senior management will comprise the senior management roles of New Grove and be responsible forthe day-to-day operations;
- New Grove will assume the Grove name; and
- The intended strategy and operations of New Grove will continue Grove's current strategy.

Grove is in process of assessing the accounting related to the Business Combination and the treatment related to the:

• Grove Earnout Shares – The Grove Earnout Shares, excluding those allocated to the unvested Grove options and restricted stock units, are expected to be accounted for as a long-term liability classified as equity-linked contracts that are earned upon achieving the triggering events, which include events that are not indexed to the common stock of New Grove. The Company has preliminarily concluded that liability classification for the Grove Earnout Shares is appropriate and as such, the liability will be recognized at fair value upon the Merger closing and remeasured in future reporting periods through the statement of operations. The preliminary fair value of the Grove Earnout Shares was determined using the most reliable information currently available. The Grove Earnout Shares attributed to the unvested Grove options and restricted stock units are expected to be accounted for as stock-based compensation due to the continued service requirement and will be equity-classified. Any compensation expense related to such Grove Earnout Shares will be recognized over the respective service periods in future reporting periods and has not been reflected in the unaudited pro forma condensed combined statement of operations.

- Public Warrants and Private Placement Warrants The Company has preliminarily concluded that liability classification for the Public Warrants and
 Private Placement Warrants is appropriate and as such, the liability will be recognized at fair value upon the Merger closing and remeasured in future
 reporting periods through the statement of operations.
- Direct and Incremental Transaction Costs Estimates are necessary to finalize the allocation of direct and incremental transaction costs between
 instruments issued or assumed in the Business Combination. The Company has preliminarily allocated such costs on a relative fair value basis
 between the VGAC II ordinary shares, PIPE Shares, Public Warrants, Private Placement Warrants, Grove Earnout Shares, Downside Protection
 feature and Backstop Warrants based on estimates that are available. Direct and incremental transaction costs allocated to equity-classified
 instruments have been preliminarily recorded within equity in the unaudited pro forma condensed combined financial statements. Direct and
 incremental transaction costs allocated to liability-classified equity instruments were expensed in the unaudited pro forma condensed combined
 financial statements.
- Backstop Subscription Agreement The Company has determined there is a freestanding instrument related to the Downside Protection feature
 under the Backstop Subscription Agreement. Such feature is required to be bifurcated and classified as a liability. A separate valuation has not yet
 been completed and therefore the fair value of this freestanding instrument has not been reflected in the unaudited pro forma condensed combined
 financial statements. The Company has preliminarily classified the entire proceeds from the Tranche 1 financing within the equity section of the pro
 forma balance sheet as of March 31, 2022.

The final accounting related to the Business Combination, including the Grove Earnout Shares, Public Warrants, Private Placement Warrants, Backstop Subscription Agreement shares, Backstop Warrants and transaction costs will be finalized by New Grove and reported in the Company's financial statement as of and for the six months ended June 30, 2022.

Basis of Pro Forma Presentation

The unaudited pro forma condensed combined financial information has been prepared in accordance with Article 11 of Regulations-X. The adjustments in the unaudited pro forma condensed combined financial information have been identified and presented to provide relevant information in accordance with GAAP necessary for an illustrative understanding of New Grove upon consummation of the Business Combination. Assumptions and estimates underlying the unaudited pro forma adjustments set forth in the unaudited pro forma condensed combined financial information are described in the accompanying notes.

The unaudited pro forma condensed combined financial information has been presented for illustrative purposes only and is not necessarily indicative of the operating results and financial position that would have been achieved had the Business Combination occurred on the dates indicated, and does not reflect adjustments for any anticipated synergies, operating efficiencies, tax savings, or cost savings. The cash proceeds remaining after the consummation of the Business Combination and the other related events contemplated by the Merger Agreement are expected to be used for general corporate purposes. The unaudited proforma condensed combined financial information does not purport to project the future operating results or financial position of New Grove following the completion of the Business Combination. The unaudited proforma adjustments represent management's estimates based on information available as of the date of these unaudited proforma condensed combined financial information and are subject to change as additional information becomes available and analyses are performed. VGAC II and Grove have not had any historical relationship prior to the Business Combination. Accordingly, no proforma adjustments were required to eliminate activities between the companies.

The unaudited pro forma condensed combined financial information contained herein reflects VGAC II stockholders' approval of the Merger on June 14, 2022, and that VGAC II Class A ordinary shareholders holding 39,672,045 shares elected to redeem their shares prior to the Closing and the purchase of 1,671,524 shares of Backstop Tranche 2 Shares by the Backstop Investor. At the closing of the Merger, there were 29,412,877 outstanding shares of New Grove Class A Common Stock and 133,390,456 outstanding shares of New Grove Class B Common Stock, after giving effect to the Employee Stock Grants and certain employees' elections to receive New Grove Class A Common Stock in lieu of New Grove Class B Common Stock.

The following summarizes the New Grove Class A Common Stock and New Grove Class B Common Stock issued and outstanding immediately after the Merger (after giving effect to the Employee Stock Grants):

		%
	Number of Shares	Ownership
Former VGAC II shareholders(1)(2)	667,955	0.4%
Sponsor	9,972,500	6.1%
PIPE Investors	8,607,500	5.3%
Grove Shareholders(3)	127,851,594	78.6%
Employee Stock Grants	32,300	— %
Tranche 2 Backstop Shares	1,671,524	1.0%
Grove Earnout	13,999,960	8.6%
Total shares of New Grove Common Stock outstanding at closing of the		
Merger	162,803,333	100.0%

- The Exchange Ratio is calculated as of the close of the Merger. The number of shares outstanding for both VGAC II and Grove are calculated as of the close of the Merger.
- (2) Includes 90,000 of Class B ordinary shares which were transferred from the Sponsor to three independent directors, each receiving 30,000 Class B ordinary shares.
- (3) Includes Backstop Tranche 1 Shares issued in accordance with the Backstop Agreement.

The following summarizes the New Grove Class A Common Stock and the New Grove Class B Common Stock issued and outstanding immediately after the Business Combination, inclusive of the dilutive effects of the Grove Earnout, Public and Private Placement Warrants, Backstop Warrants, and the rollover of Grove equity awards, after giving effect to the Exchange Ratio (after giving effect to the Employee Stock Grants):

	Number of Shares	% Ownership
Former VGAC II shareholders(1)(2)	667,955	0.3%
Sponsor	9,972,500	4.7%
PIPE Investors	8,607,500	4.0%
Grove Shareholders(3)	127,851,594	60.2%
Employee Stock Grants	32,300	— %
Tranche 2 Backstop Shares	1,671,524	0.8%
Grove Earnout	13,999,960	6.6%
Private Placement Warrants	6,700,000	3.1%
Public Warrants	8,050,000	3.8%
Grove equity holders (options)	24,691,735	11.6%
Grove equity holders (RSUs)	5,716,862	2.7%
Grove equity holders (warrants)	923,857	0.4%
Backstop Warrants	3,875,028	1.8%
Pro forma shares outstanding, diluted	212,760,815	100.0%

- (1) The Exchange Ratio is calculated as of the close of the Mergers. The number of shares outstanding for both VGAC II and Grove are calculated as of the close of the Mergers.
- (2) Includes 90,000 of Class B ordinary shares which were transferred from the Sponsor to three independent directors, each receiving 30,000 Class B ordinary shares.
- (3) Includes Backstop Tranche 1 Shares issued in accordance with the Backstop Agreement.

UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET As of March 31, 2022 (in thousands)

		in Group II istorical)	Grove (Historical)	Transacti	ion Accounting Adjustments (Note 3)	Notes	Pro Fo	rma Combined
ASSETS								
Current assets:								
Cash and cash equivalents	\$	80	\$ 74,428				\$	147,664
					402,566	A		
					86,075	В		
					(14,088)	C		
					(21,392)	D		
					16,715	M		
					(396,720)	F		
Inventory, net		_	50,559					50,559
Prepaid expenses and other current								
assets		615	11,264					11,879
Total current assets		695	136,251		73,156			210,102
Cash and Investments held in trust								
account		402,566	_		(402,566)	A		_
Property and equipment, net		_	16,095					16,095
Operating lease right-of-use assets		_	20,471					20,471
Other long-term assets			5,550		(4,163)	D		1,387
Total assets	\$	403,261	\$ 178,367	\$	(332,573)		\$	248,055
STOCKHOLDERS' EQUITY (DEFICIT) Current liabilities:								
Accounts payable	\$	_	\$ 31.822	\$	(131)	D	\$	31.691
Accrued expenses	φ	2.814	24,208	Ψ	(2,424)	D	Ψ	24,598
Due to related party		32	24,200		(2,424)	D		32
Deferred revenue			11,426					11,426
Operating lease liabilities, current		_	3,724					3,724
Other current liabilities		_	894					894
Debt, current		_	16,720					16,720
Promissory note - related party		1,000						1,000
Total current liabilities		3,846	88,794		(2,555)			90.085
Debt, noncurrent			50,034		(2,333)			50,034
Operating lease liabilities, noncurrent		_	19,090					19,090
Other long-term liabilities		_	3,924		(2,901)	K		1,023
Warrant liability		5,686			(2,501)			5,686
Earnout liability		_	_		66,265	G		66,265
Backstop derivative liability		30,234			(30,234)	0		,
Deferred underwriters' discount		14,088	_		(14,088)	C		_
Total liabilities:		53,854	161.842		16.487			232,183
Convertible preferred stock			487,918		(487,918)	I		
Convertible common stock			27,473		(27,473)	Ĺ		_
VGII Class A ordinary shares subject to			.,		(.,)			
redemption		402,500	_		(402,500)	E		_
*					` ' '			

	Virgin Group II (Historical)	Grove (Historical)	Transaction Accounting Adjustments (Note 3)	Notes	Pro Forma Combined
Stockholders' equity (deficit):		<u> </u>			
Grove Common stock	_	1	(1)	J	_
Preferred stock (VGII)	_				_
VGII Class A Ordinary Shares	_				_
VGII Class B Ordinary Shares	1		(1)	E	_
New Grove Class A common					
stock	_		1	В	2
			5	E	
			(4)	F	
New Grove Class B common					
stock	_		1	J	11
			10	I	
Additional paid-in capital	_	38,660	86,074	В	558,596
			(17,790)	D	
			402,496	E	
			(396,716)	F	
			(66,265)	G	
			(53,094)	Н	
			487,908	I	
			2,901	K	
			27,473	L	
			16,715	M	
			30,234	N	
Accumulated deficit	(53,094)	(537,527)	53,094	H	(542,737)
			(5,210)	D	
Total stockholders' (deficit) equity	(53,093)	(498,866)	567,831		15,872
Total liabilities, convertible preferred stock and stockholders' equity					
(deficit)	\$ 403,261	\$ 178,367	\$ (333,573)		\$ 248,055

 $See\ accompanying\ notes\ to\ the\ unaudited\ pro\ forma\ condensed\ combined\ financial\ statements.$

Unaudited Pro Forma Condensed Combined Statement of Operations For the Three Months Ended March 31, 2022 (in thousands, except share and per share data)

	For the Three Months ended March 31, 2022 VGII (Historical)	For the Three Months ended March 31, 2022 Grove (Historical)	Transaction Accounting Adjustments (Note 3)	Notes	Pro Forma Combined
Revenue, net	\$ —	90,479	\$ —	110103	90,479
Cost of goods sold	_	47,742	_		47,742
Gross profit		42,737			42,737
Operating expenses:		,			<u> </u>
Advertising	_	32,793	_		32,793
Product development		6,240			6,240
Selling, general and administrative	_	50,970	_		50,970
Formation and operating costs	1,008				1,008
Operating loss	(1,008)	(47,266)	_		(48,274)
Interest income earned on investments held in Trust Account	(36)		36	EE	_
Change in fair value of warrant liabilities	(7,654)		_		(7,654)
Interest expense	_	2,087	_		2,087
Other expense (income), net	_	(1,992)	(1,886)	FF	(3,878)
Initial measurement of derivative liability	30,234		(30,234)	GG	
Interest and other expense (income), net	22,544	95	(32,084)		(9,445)
Income (Loss) before provision for income taxes Provision for income taxes	(23,552)	(47,361) 23	32,084		(38,829) 23
Net income (loss)	(22,552)		22.084		(38,852)
· /	(23,552)	(47,384)	32,084		(38,832)
Net loss attributable to common stockholders, basic and diluted	(23,552)	(47,384)	32,084		(38,852)
Weighted average shares outstanding of New Grove Class A and B Common Stock – basic and diluted				нн	152,218,639
Net loss per share of New Grove Class A and B Common Stock – basic and diluted					(0.26)
Weighted average shares outstanding of Grove common stock – basic and diluted		7,981,994			
Net loss per share of Grove common stock – basic and diluted		\$(5.94)			
Weighted average shares outstanding of VGII Class A ordinary shares – basic and diluted	40,250,000				
Net loss per share of VGII Class A ordinary shares – basic and diluted	10,062,500				
Weighted average shares outstanding of VGII Class B ordinary shares – basic and diluted	\$ (0.47)				
Net loss per share of VGII Class B ordinary shares – basic and diluted	\$ (0.47)				
					

 $See\ accompanying\ notes\ to\ the\ unaudited\ pro\ forma\ condensed\ combined\ financial\ statements.$

Unaudited Pro Forma Condensed Combined Statement of Operations For the Year Ended December 31, 2021 (in thousands, except share and per share data)

		the period from January 13, (inception) to December 31, 2021 VGII (Historical)	For th	e Year Ended December 31, 2021 Grove (Historical)	Transaction Accounting Adjustments (Note 3)	Notes	Pro Forma Combined
Revenue, net	\$	_	\$	383,685	\$ —		\$ 383,685
Cost of goods sold				195,181			195,181
Gross profit		_		188,504	_		188,504
Operating expenses:				105.212			107.212
Advertising Product		_		107,313	_		107,313
development		_		23,408	_		23,408
Selling, general and				23,100			23,100
administrative		_		186,638	5,210	AA	191,848
Formation and							
operating costs		3,573					3,573
Operating loss		(3,573)		(128,855)	(5,210)		(137,638)
Interest income earned on investments held in Trust Account		(31)			31	ВВ	
Offering costs allocated		(31)			31	DD	_
to warrants		570		_	_		570
Change in fair value of							
warrant liabilities		(6,811)		_	_		(6,811)
Interest expense		_		5,202	_		5,202
Loss on extinguishment of debt				1.027			1.027
Other expense		_		1,027	_		1,027
(income), net		_		760	(1,234)	CC	(474)
Interest and other				700	(1,25.1)		
expense							
(income), net		(6,272)		6,989	(1,203)		(486)
Income (Loss) before provision for income							
taxes		2,699		(135,844)	(4,007)		(137,152)
Provision for income				50	_		52
taxes Net income (loss)		2,699		(135,896)	(4,007)		<u>52</u> (137,204)
Net loss attributable to common stockholders, basic and diluted		2,699		(135,896)	(4,007)		(137,204)
Weighted average shares outstanding of New Grove Class A and B Common Stock – basic and diluted		2,002		(100,070)	(1,000)	DD	151,402,644
Net loss per share of New Grove Class A and B Common Stock – basic and diluted							<u>\$ (0.91)</u>
Weighted average shares outstanding of Grove common stock – basic and diluted				7,288,145			
Net loss per share of Grove common stock – basic and diluted			\$	(18.65)			
Weighted average shares outstanding of VGII Class A ordinary shares – basic and diluted		32,705,669		_			
Net loss per share of VGII Class A ordinary shares – basic and diluted	\$	0.06					
Weighted average shares outstanding of VGII Class B ordinary shares – basic and diluted	<u>v</u>	10,062,500					
basic and undted		10,002,300					

\$ 0.06

 $See\ accompanying\ notes\ to\ the\ unaudited\ pro\ forma\ condensed\ combined\ financial\ statements.$

NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS

Note 1—Description of the Business Combination

On December 7, 2021, VGAC II entered into the Merger Agreement with Grove and on March 31, 2022, VGAC II entered into an Amended and Restated Agreement and Plan of Merger with Grove. All references to the Merger Agreement in these notes shall be interpreted to refer to the Merger Agreement as amended and restated, unless the context otherwise requires.

Pursuant to the Merger Agreement, at the closing on June 16, 2022, Grove merged with a wholly owned subsidiary of VGAC II, with Grove being the surviving corporation as a wholly owned subsidiary of VGAC II and II changed its name to Grove Collaborative Holdings, Inc. ("New Grove"). Following the closing, (a) New Grove owns all the equity interests of Grove and (b) the former equity holders of Grove holds all of the outstanding New Grove Class B Common Stock.

Note 2-Basis of the Pro Forma Presentation

The Business Combination will be accounted for as a reverse recapitalization in accordance with GAAP. Under this method of accounting, VGAC II will be treated as the "acquired" company for financial reporting purposes. Accordingly, for accounting purposes, the financial statements of New Grove will represent a continuation of the financial statements of Current Grove with the Business Combination treated as the equivalent of Current Grove issuing stock for the net assets of VGAC II, accompanied by a recapitalization. The net assets of VGAC II will be stated at historical cost, with no goodwill or other intangible assets recorded. Operations prior to the Business Combination will be presented as those of Current Grove in future reports of New Grove

The unaudited pro forma condensed combined balance sheet as of March 31, 2022 assumes that the Merger and the other related events, summarized below, occurred on March 31, 2022. The unaudited pro forma condensed combined statements of operations for the three months ended March 31, 2022 and for the year ended December 31, 2021 present pro forma effect to the Merger and the other related events, summarized below, as if it had been completed on January 1, 2021, (the date of inception for VGAC II was January 13, 2021).

The unaudited pro forma condensed combined financial information was derived from and should be read in conjunction with the following historical financial statements and the accompanying notes, which are included elsewhere in this proxy statement/consent solicitation statement/prospectus:

- the (a) historical audited condensed financial statements of VGAC II as of December 31, 2021 and for the period from January 13, 2021 (inception) to December 31, 2021, included in VGAC II's Annual Report on From 10-K filed with the SEC on February 24, 2022 and (b) historical unaudited condensed financial statements of VGAC II for the three months ended March 31, 2022 included in VGAC II's Quarterly Report on Form 10-Q filed with the SEC on May 16, 2022 and incorporated by reference;
- the (a) historical audited financial statements of Grove as of and for the years ended December 31, 2019, 2020 and 2021 included in the proxy statement/consent solicitation statement/prospectus, and (b) historical unaudited condensed financial statements of Grove for the three months ended March 31, 2022 included as Exhibit 99.1 to this Current Report on Form 8-K and are incorporated herein by reference.
- other information relating to Grove and VGAC II included in the proxy statement/consent solicitation statement/prospectus, including the Merger Agreement and the description of certain terms thereof set forth under the section titled "Business Combination Proposal."

Management has made significant estimates and assumptions in its determination of the pro forma adjustments. As the unaudited pro forma condensed combined financial information has been prepared based on these preliminary estimates, the final amounts recorded may differ materially from the information presented.

The unaudited pro forma condensed combined financial information does not give effect to any anticipated synergies, operating efficiencies, tax savings, or cost savings that may be associated with the Business Combination.

The proforma adjustments reflecting the consummation of the Business Combination are based on certain currently available information and certain assumptions and methodologies that New Grove believes are reasonable under the circumstances. The unaudited proforma adjustments, which are described in the accompanying notes, may be revised as additional information becomes available and is evaluated. Therefore, it is likely that the

actual adjustments will differ from the pro forma adjustments and it is possible the difference may be material. New Grove believes that its assumptions and methodologies provide a reasonable basis for presenting all of the significant effects of the Business Combination based on information available to management at this time and that the pro forma adjustments give appropriate effect to those assumptions and are properly applied in the unaudited pro forma condensed combined financial information.

The unaudited pro forma condensed combined financial information is not necessarily indicative of what the actual results of operations and financial position would have been had the Business Combination taken place on the dates indicated, nor are they indicative of the future consolidated results of operations or financial position of the post-combination company. They should be read in conjunction with the historical financial statements and notes thereto of VGAC II and Grove.

Note 3-Pro Forma Adjustments

The unaudited pro forma condensed combined financial information has been prepared to illustrate the effect of the Business Combination and has been prepared for informational purposes only.

The unaudited pro forma condensed combined financial information has been prepared in accordance with Article 11 oRegulation S-X as amended by the final rule, Release No. 33-10786 "Amendments to Financial Disclosures about Acquired and Disposed Businesses." Release No. 33-10786 replaces the existing pro forma adjustment criteria with simplified requirements to depict the accounting for the transaction ("Transaction Accounting Adjustments") and present the reasonably estimable synergies and other transaction effects that have occurred or are reasonably expected to occur ("Management's Adjustments"). Given such Management Adjustments, if any, would not enhance an understanding of the pro forma effects of the Transaction, VGAC II has elected not to present any Management's Adjustments and will only be presenting Transaction Accounting Adjustments in the following unaudited pro forma condensed combined financial information.

VGAC II and Grove have not had any historical relationship prior to the Business Combination. Accordingly, no pro forma adjustments were required to eliminate activities between the companies.

The pro forma combined provision for income taxes does not necessarily reflect the amounts that would have resulted had the post-combination company filed consolidated income tax returns during the periods presented. VGAC II has a full valuation allowance on any of its net deferred assets and accordingly, has not recorded any provision for income taxes in its historical statements of operations. Therefore, the proforma adjustments assume a 0% effective interest rate.

Transaction Accounting Adjustments to Unaudited Pro Forma Condensed Combined Balance Sheet

The adjustments included in the unaudited pro forma condensed combined balance sheet as of March 31, 2022 are as follows:

- (A) The reclassification of \$402.6 million of cash and investments held in the VGAC II Trust Account that became available at closing of the Business Combination. See adjustment note (F) for actual redemptions in connection with the Merger closing
- (B) In connection with the signing of the Merger Agreement, VGAC II entered into subscription agreements with certain investors (the <u>PIPE Investors</u>"). Pursuant to the Subscription Agreements, the PIPE Investors purchased, and VGAC II issued and sold to such investors 8.6 million shares of New Grove Class A Common Stock with par value of \$0.0001, resulting in gross proceeds of \$86.1 million. The costs related to the issuance of the PIPE Financing are adjusted against additional paid in capital (see adjustment D below).
- (C) Reflects the cash settlement of the \$14.1 million liability for VGAC II's deferred underwriting commissions related to its initial public offering.

- (D) Represents the preliminary estimated direct and incremental transaction costs incurred by Grove related to the Business Combination, including underwriting/banking, legal, accounting and other fees, of which \$17.8 million is allocated to equity-classified instruments consisting of the VGAC II ordinary shares, PIPE Shares, and the Backstop Warrants, and reflected in the unaudited pro forma condensed combined balance sheet as a direct reduction to New Grove's additional paid-in capital and are assumed to be cash settled. The direct and incremental costs related to liability-classified instruments totaling \$5.2 million is expensed as of January 1, 2021 (see adjustment AA below). As of March 31, 2022, Grove had deferred transaction costs incurred of \$4.2 million, of which \$2.6 million was unpaid.
- (E) Reflects the recapitalization of VGAC II Class A ordinary shares subject to possible redemption and VGAC II Class B ordinary shares into New Grove Class A Common Stock at \$0.0001 par value.
- (F) Reflects the cash disbursed to redeem 39,672,045 shares of VGAC II common stock for \$396.7 million at a redemption price of \$10.00 per share and allocated to New Grove Class A Common Stock and additional paid-in capital using par value of \$0.0001 per share
- (G) Reflects the preliminary estimated fair value of contingently returnable Grove Earnout Shares, excluding those allocated to the unvested Grove options and restricted stock units, that are expected to be accounted for as liability classified equity instruments that are earned upon achieving the triggering events, which include events that are not indexed to the common stock of New Grove. The preliminary fair value of the Grove Earnout Shares was determined using the most reliable information currently available. Refer to Note 5 for more information. Subsequent to the consummation of the Business Combination, this liability will be remeasured to its fair value at the end of each reporting period and subsequent changes in the fair value will be recognized in New Grove's statement of operations within other income/expense.
- (H) Reflects the elimination of VGAC II's historical accumulated deficit with a corresponding adjustment to additional paid-in capital for New Grove in connection with the reverse recapitalization upon closing of the Business Combination.
- (I) Reflects the conversion of Grove convertible preferred stock into shares of Grove common stock, and such shares were converted into the right to receive shares of New Grove Class B Common Stock pursuant to the Exchange Ratio concurrent with the closing of the Business Combination.
- (J) Reflects the conversion of Grove Common Stock into shares of New Grove Class B Common Stock pursuant to the Exchange Ratio concurrent with the closing of the Business Combination.
- (K) Reflects the reclassification of Grove warrants from liability to equity classification as the warrants became exercisable for shares of New Grove Class B Common Stock rather than Grove convertible preferred stock upon closing of the Business Combination.
- (L) Reflects the reclassification of 2,750,000 shares of Tranche 1 Shares from mezzanine equity to permanent equity and allocated to New Grove Class A Common Stock and additional paid-in capital using par value of \$0.0001 per share
- (M) Reflects the cash proceeds of \$16.7 million for the purchase of 1,671,524 shares of Backstop Tranche 2 consisting of New Grove common stock by the Backstop Investor
- (N) Reflects the elimination of the backstop derivative liability as recorded on VGAC II historical balance sheet

Transaction Accounting Adjustments to Unaudited Pro Forma Condensed Combined Statements of Operations

The pro forma adjustments included in the unaudited pro forma condensed combined statements of operations for the year ended December 31, 2021 are as follows:

- (AA) Represents the preliminary estimated direct and incremental transaction costs incurred by Grove related to the Business Combination, including underwriting/banking, legal, accounting and other fees, of which \$5.2 million is allocated to the Downside Protection Feature, Grove Earnout, Public Warrants and Private Placement Warrants that are expected to be liability-classified and expensed.
- (BB) Reflects the elimination of interest income related to the investments held in the trust account for the period from January 13, 2021 through December 31, 2021
- (CC) Reflects the elimination of remeasurement losses on the Grove convertible preferred stock warrant liability.

(DD) As the Business Combination is being reflected as if it had occurred on January 1, 2021, the calculation of weighted average shares outstanding for pro forma basic and diluted net income per share assumes that the shares issuable in connection with the Business Combination, the PIPE Financing, Tranche 1 and Tranche 2 of the Backstop Subscription Agreement and the Backstop Warrants have been outstanding for the entirety of the period presented. The calculation is retroactively adjusted to eliminate all VGAC II Class A ordinary shares redeemed as if such redemption occurred on January 1, 2021.

The pro forma adjustments included in the unaudited pro forma condensed combined statements of operations for the three months ended March 31, 2022 are as follows:

- (EE) Reflects the elimination of interest income related to the investments held in the VGAC II trust account for the three months ended March 31, 2022.
- (FF) Reflects the elimination of remeasurement losses on the Grove convertible preferred stock warrant liability for the three months ended March 31, 2022
- (GG) Reflects the elimination of initial measurement of the Backstop Derivative Liability on the VGAC II income statement for the three months ended March 31, 2022
- (HH) As the Business Combination is being reflected as if it had occurred on January 1, 2021, the calculation of weighted average shares outstanding for pro forma basic and diluted net income per share assumes that the shares issuable in connection with the Business Combination, the PIPE Financing, Tranche 1 and Tranche 2 of the Backstop Subscription Agreement and the Backstop Warrants have been outstanding for the entirety of the period presented. The calculation is retroactively adjusted to eliminate all VGAC II Class A ordinary shares redeemed as if such redemption occurred on January 1, 2021.

Note 4—Loss per Share

Represents the net loss per share calculated using the historical weighted average shares outstanding, and the issuance of additional shares in connection with the Business Combination, the PIPE Financing, Tranche 1 and Tranche 2 of the Backstop Subscription Agreement and the Backstop Warrants, assuming the shares were outstanding since January 1, 2021. The calculation is retroactively adjusted to eliminate all VGAC II Class A ordinary shares redeemed as if such redemption occurred on January 1, 2021.

The unaudited pro forma condensed combined financial information is as follows (in thousands, except share and per share data):

	Year Ended ember 31, 2021	ee Months Ended Tarch 31, 2021
Pro Forma net loss	\$ (137,204)	\$ (38,852)
Basic weighted average shares outstanding	151,402,644	152,218,639
Pro forma net loss per share, basic and diluted(1)	\$ (0.91)	\$ (0.26)

 The pro forma basic and diluted shares include 3,875,028 shares underlying the Backstop Warrants (each warrant exercisable to purchase one VGAC II class A common share for \$0.01) Following the closing of the Business Combination, the following outstanding shares of common stock equivalents were excluded from the computation of pro forma diluted net loss per share for all the periods presented above because including them would have had an anti-dilutive effect:

	Year Ended	Three Months Ended
	December 31, 2021	March 31, 2022
Private Placement Warrants(1)	6,700,000	6,700,000
Public Warrants	8,050,000	8,050,000
Grove common stock options(2)	27,882,520	25,691,329
Grove restricted stock units(2)	1,777,183	3,224,181
Grove common stock warrants	1,424,108	1,148,477
Grove common stock issued upon early exercise of options	81,750	16,539
Grove Earnout Shares (3)	13,999,960	13,999,960
Total	59,915,521	58,830,486

- (1) One whole warrant entitles the holder to purchase one share of New Grove Class A Common Stock at a price of \$11.50 per share. New Grove's warrants are anti-dilutive on a pro forma basis and have been excluded from the diluted number of New Grove's shares outstanding at the time of closing.
- (2) All outstanding Grove options and restricted stock units at the closing, whether vested or unvested, will convert into options or rights to purchase a number of shares of New Grove Class B Common Stock, determined in accordance with the Exchange Ratio. Additionally, holders of Grove options and restricted stock units will receive a pro rata share of the Grove Earnout Shares.
- (3) Grove Earnout Shares are subject to certain vesting and forfeiture conditions as described herein.

Note 5—Grove Earnout Shares

The Grove Earnout Shares, excluding those related to unvested Grove options and restricted stock units, are expected to be accounted for as liability classified equity instruments that are earned upon achievement of the Share Price Milestones, which provide for settlement provisions that are not indexed to New Grove's Class A common stock. The preliminary estimated fair value of the Grove Earnout Shares is \$66.3 million.

The estimated fair value of the Grove Earnout Shares was determined by a Monte Carlo simulation valuation model using a distribution of potential outcomes on a rolling basis over the ten-year earnout period. The preliminary estimated fair value of the Earnout Shares was determined using the most reliable information currently available. Assumptions used in the preliminary valuation, which are subject to change at the closing of the Business Combination, were as follows:

Current stock price: the current stock price was set at the current value per share for the Class A ordinary shares.

Expected volatility: the volatility rate was determined using an average of historical volatilities of selected industry peers deemed to be comparable to Grove's business, corresponding to the expected term of the awards.

Risk-free interest rate: the risk-free interest rate is based on the U.S. Treasury yield curve in effect at the time of issuance for zero-coupon U.S. Treasury notes with maturities corresponding to the expected ten-year term of the earnout period.

Expected term: the expected term is the ten-year term of the earnout period.

Expected dividend yield: the expected dividend yield is zero as Grove has never declared or paid cash dividends and has no current plans to do so during the expected term.

The actual fair values of the Grove Earnout Shares are subject to change as additional information becomes available and additional analyses are performed and such changes could be material once the final valuation is determined at the closing of the Business Combination.